Instrumental music teaching with dyslexic students: An action research project

Kristl Kirk PhD University of York

Arts and Creative Technologies April 2024

Abstract

The complexities of dyslexia can create challenges for music educators and their dyslexic students. Although there is a substantial body of dyslexia research related to literacy remediation, there is limited research concerning instrumental music teaching and dyslexia. Much of the existing information is deficit-based, focuses on skill discrepancies and highlights limitations. There is little understanding of the value of relationships between parent-teacher-student, exploration of the perspectives of dyslexic music students or consideration of how their strengths might be identified and utilised in the lesson context.

The purpose of this thesis is to improve and enhance instrumental music teaching for dyslexic students. Cycles of action research were informed by a review of the literature and a survey investigating the perceptions of music exams amongst dyslexic students. Data collection also included semi-structured interviews with students, parents and teachers in addition to reflective observations with two case studies in the context of my piano teaching practice. Data from an evaluative focus group brought insights from the wider inclusive music education perspective.

Thematic analysis of the data suggests that there are a number of barriers in music education for dyslexic students and their families across ecological systems and highlights the need for high-quality pedagogical training for teachers. A shift from a deficit-focused approach to an individually tailored strengths-focused approach has potential benefits for the teacher and parent perspectives as well as promoting the student's self-confidence and reducing anxiety in the learning environment. The findings led to the development of an innovative risk-resilience framework for a better understanding of how to support dyslexic learners in the music education context. Analysis of the data also underpins the creation of a teacher knowledge and practice standards framework to form the basis of a future teacher training programme.

Contents

Chapter 1 INTRODUCTION	
1.1 Introduction	19
1.2 Context	19
1.3 Research aim and research questions for this study	22
1.4 Outline of the thesis	22
1.5 An expanded overview of each chapter	24
Chapter 2 LITERATURE REVIEW	
2.1 Introduction	27
2.2 Context and role of researcher	27
2.3 Dyslexia: Definition and characteristics	29
2.3.1 Defining dyslexia	29
2.3.2 Underlying causes	31
2.3.3 Co-occurring conditions	33
2.3.4 Secondary characteristics	34
2.3.5 Support factors	35
2.4 Music training as an intervention for dyslexia	38
2.4.1 Connections between music and language	
2.4.2 Rationale for the use of music as an intervention for dyslexia	40
2.4.3 Music training for remediation of impairments associated with dyslexia	42
2.4.4 Summary	44
2.5 Music teaching and dyslexia literature	46
2.6 Analysis of music and dyslexia materials	51
2.6.1 Introduction	51
2.6.2 Comparison of the layout and accessibility of the books	53

2.6.3 Identifying features of dyslexia	55
2.6.4 The label of dyslexia	57
2.6.5 Learning modalities	58
2.6.6 Strategies	
2.7 Inclusive music teaching and access to provision	70
2.7.1 Inclusive music teaching approaches and considerations	70
2.7.2 Access to provision	74
2.7.3 Summary	76
2.8 The research questions	77
REFLECTIVE STATEMENT I: MOTIVATIONS AND CHALLENGES	80
RS1 Bronfenbrenner's Ecological Systems theoretical model (1979)	80
RS1.1 Motivation	81
RS1.2 Challenges	82
RS1.3 Theory and outputs	85
RS1.4 Conclusion	85
REFLECTIVE STATEMENT 2: INTRODUCTION TO THE STUDENTS	87
RS2.1 Introduction: Process-Person-Context-Time Model	87
RS2.2 Process	87
RS2.3 Person	
RS2.3.1 Introduction to Ben	90
RS2.3.2 Introduction to Alex	
RS2.4 My observations of these lessons	
RS2.5 Time	92
RS2.6 Context	93
RS2.7 Conclusion	94

Chapter 3 RESEARCH DESIGN	
3.1 Theoretical framework – Social constructivism	95
3.2 Discussion of positivist, interpretivist and pragmatic approaches	96
3.3 Action research methodology	98
3.3.1 Benefits of action research	99
3.3.2 Limitations of action research	103
3.4 Methods of data collection	103
3.4.1 Lesson planning and observations	105
3.4.2 Questionnaire	106
3.4.3 Interviews	107
3.4.4 Video-recorded lessons	108
3.4.5 Reflective journal	110
3.4.6 Evaluative focus group	111
3.5 Data analysis	112
3.5.1 Thematic analysis	114
3.5.2 Steps of thematic analysis	115
3.6 Establishing validity	116
3.7 Ethical issues	117
3.8 Conclusion	121
Chapter 4 THE ROLE OF THE RESEARCHER	122
4.1 The role of the researcher	122
4.2 Consideration of being both an insider as well as an outsider researcher	122
4.3 Advantages to being an insider-researcher	123
4.4 Challenges for the insider-researcher	124
4.4.1 Conflict of dual roles	124

4.4.2 Navigating power imbalances	125
4.4.3 Overcoming bias	
4.4.4 Identifying preconceptions and beliefs	
4.5 Ethical considerations of insider-research	
Chapter 5 FINDINGS FROM INITIAL SURVEY	130
5.1 Overview	
5.2 Contextual information	131
5.3 Student challenges	132
5.4 Obtaining information about the exams	
5.4.1 Accessibility of exam board's website	
5.4.2 Documentation required	
5.5 Choosing to take exams	
5.5.1 Adjustments/accommodations taken during exams	
5.5.2 A question of equity	
5.5.3 Recommendations for improvement	
5.6 Limitations, conclusion and recommendations for the future	
5.6.1 Limitations	
5.6.2 Summary of key findings	
Chapter 6 FINDINGS FROM TEACHER INTERVIEWS: BARRIERS AND CHALLENGES.	138
6.1 Introduction	138
6.2 Contextual information	
6.2.1 Participants	
6.2.2 Summary of the semi-structured interview questions	
6.3 Teacher perceptions of barriers in teaching dyslexic students	
6.3.1 Barriers: Gaining expertise	141

6.3.2 Barriers: Communication with school and parents	148
6.4 Challenges in the teaching context	152
6.4.1 Identifying dyslexic students	153
6.4.2 Recognising whether a strategy is working	154
6.4.3 Lack of control over the learning environment	155
6.4.4 Recognising when a student is experiencing stress	156
6.4.5 Developing the student's understanding of dyslexia and their own met processes	acognitive 157
6.5 Challenges faced by teachers with dyslexia/neurodivergence	158
6.5.1 Limitations caused by the inaccessibility of material	159
6.5.2 Challenges related to the particular learning difference	159
6.6 Summary of findings	160
6.6.1 Barriers related to research, training and resources	160
6.6.2 Barriers in communication with schools and parents	161
6.6.3 Challenges	162
6.7 Conclusion	162
REFLECTIVE STATEMENT 3: MY BARRIERS AND CHALLENGES	164
RS3.1 Barriers: Lack of knowledge	164
RS3.2 Challenge: The term 'dyslexia'	164
RS3.3 Challenge: My own assumptions	165
RS3.4 Challenge: Choosing teaching strategies	165
RS3.5 Challenge: Recognising strategies of resilience and support that enable f	lourishing
	166
RS3.6 Conclusion	167
Chapter 7 FINDINGS FROM TEACHER INTERVIEWS: STUDENT CHALLENGES AND STRATEGIES	168

7.1 Introduction	168
7.2 Student challenges identified by teachers	168
7.2.1 Slow speed of processing	170
7.2.2 Executive function deficits	170
7.3 Strategies designed to mediate slow processing and executive function deficits	173
7.4 Low self-esteem and mediating strategies	175
7.5 Coordination and spatial awareness issues	178
7.6 Visual processing issues	179
7.7 Strategies specific to sight-reading	181
7.8 Summary of findings	187
7.9 Conclusion	189
REFLECTIVE STATEMENT 4: POWER AND KNOWLEDGE DYNAMICS	190
RS4.1 Resisting control and valuing unpredictability	190
RS4.2 Metacognition	192
RS4.2.1 Metacognition: Letting the student lead	192
RS4.2.2 Metacognitive development	194
RS4.2.3 Metacognitive development: Asking the student how they learn	195
RS4.2.4 Physical proximity and balance of verbal/non-verbal communication	196
RS4.3 Factors affecting motivation	197
RS4.4 Conclusion	201
REFLECTIVE STATEMENT 5: JOURNEY TO TACIT KNOWLEDGE	204
RS5.1 Introduction	204
RS5.2 Analysis of strategies from the literature on music and dyslexia	205
RS5.3.1 Providing options for perceiving the information, language and symbols	206
RS5.3.2 Providing options for comprehension	209

RS5.3.3 Providing options for physical action, expression and communication	210
RS5.3.4 Providing options for executive functions	212
RS5.4 Conclusion	214
Chapter 8 FINDINGS FROM TEACHER INTERVIEWS: STRENGTHS	216
8.1 Introduction	216
8.1.2 Creativity	218
8.1.3 Verbal strengths	221
8.1.4 Aural strengths	222
8.1.5 Memory strengths	224
8.1.6 Ability to follow detailed verbal instructions	226
8.1.7 Strong performers	226
8.1.8 Persistence	228
8.1.9 Summary of findings	228
8.1.10 Conclusion	230
REFLECTIVE STATEMENT 6: TRANSITION TO A STRENGTHS-FOCUSED APPROACH	232
RS6.1 Introduction	232
RS6.2 Strengths-based approaches in literature	233
RS6.3 Positive psychology in music pedagogy	235
RS6.3.1 Ben	237
RS6.3.2 Alex	242
RS6.4 Conclusion	246
Chapter 9 FINDINGS FROM PARENT INTERVIEWS	
9.1 Introduction	247
9.2 Context and participants	247
9.3 Findings	248

9.3.1 Dyslexic parents' experiences and perspectives	249
9.3.2 Musically trained parents	251
9.3.3 Communication with the teacher	253
9.3.4 Parental communication with schools and classroom teachers and im	plications for
the relationship with the instrumental teacher	256
9.3.5 Positive communication with teachers	258
9.3.6 Pride in their child's achievements and the importance of role models	259
9.4 Summary of findings	261
9.5 Conclusion	262
Chapter 10 FINDINGS FROM STUDENT INTERVIEWS	
10.1 Introduction	264
10.2 Methods	264
10.2.1 Participant recruitment	264
10.2.2 Data collection	265
10.2.3 Data analysis	267
10.3 Findings	267
10.3.1 Family incidence and environment	267
10.3.2 Assessment and diagnosis process	268
10.3.3 Challenges and strategies	270
10.3.4 The use of technology	276
10.3.5 Coloured overlays	279
10.3.6 Strengths	280
10.3.7 Positive and negative teacher experiences	282
10.3.8 Table of student-reported useful resources	284
10.3.9 Student perspectives for envisioning change	286

10.4 Summary of findings	287
10.5 Conclusion	290
REFLECTIVE STATEMENT 7: EMPATHIC UNDERSTANDING AND COMPASSIO	ON FATIGUE
	292
RS7.1 Introduction	292
RS7.2 Context	292
RS7.3 In my relationship with parents	294
RS7.4 Externalising and internalising behaviours	295
RS7.5 Masking behaviour and consequences	297
RS7.6 Developing empathic understanding	298
RS7.7 The importance of preventing compassion fatigue	298
Chapter 11 DISCUSSION OF FINDINGS	302
11.1 Introduction	
11.2 Research question 1	
11.2.1 RQ1 Barriers and challenges	
11.2.2 RQ1 Teacher observations of students' challenges and correspond	ling strategies
	307
11.3 Research question 2	311
11.4 Research question 3	312
11.4.1 Strengths identified	
11.4.2 Teachers' utilisation of strengths in the lesson context	315
11.5 Research question 4	316
11.5.1 Dyslexic student voice in the music education literature	
11.5.2 The importance of and rationale for investigating student voice	
11.6 Research question 5	318
11.7 A new risk-resilience model of dyslexia: Music education	323

11.7.1 Knowledge and practice standards for teachers of music: Rationale
11.7.2 Knowledge and practice standards: Adapted from IDA (2018)
11.8 Feedback on thesis findings from an evaluative focus group
11.8.1 The utility of the term 'dyslexia'335
11.8.2 Managing expectations through a variety of outcomes
11.8.3 Implications for students from diverse backgrounds
11.8.4 Potential routes for dissemination
Chapter 12 SUMMARY OF THE STUDY, LIMITATIONS, IMPLICATIONS AND
RECOMMENDATIONS
12.1 Summary of the study
12.2 Reviewing the research questions340
12.2.1 (RQ1) What are the perceptions of music teachers regarding their experiences
and utilisation of pedagogical practices, including strategies, methods and material, in
teaching dyslexic students?340
12.2.2 (RQ2) Do adjustments and accommodations which music exam boards allow
offer dyslexic students unbiased inclusion? What are the perspectives of teachers and
students on this topic?
12.2.3 (RS3) Do teachers, parents and students perceive dyslexic students to have
specific strengths? How might these strengths be utilised in the music lesson context?
12.2.4 (RS4) What are dyslexic students' perceptions of their music learning
experiences? How might their voices be nurtured and amplified in the pedagogical
process?
12.2.5 (RS5) What are the teacher, student and parental perceptions of the pedagogical
process and roles in the music lesson context? How can these roles collaborate to
collectively enable positive outcomes?342
12.3 Summary of my development as a teacher and researcher
12.3.1 Development through analysis of reflective observations

12.3.2 Development of professional practice	344
12.4 Limitations	347
12.5 Implications and recommendations for practice and future research	349
12.5.1 Implications for practice	350
12.5.2 Recommendations for future research	351
12.6 Conclusions	352
Reference List:	355
List of appendices	404
Appendix A	405
Appendix B	408
Appendix C	412
Appendix D	416
Appendix E	424
Appendix F	430
Appendix G	433
Appendix H	434
Appendix I	435
Appendix J	436
Appendix K	438
Appendix L	440
Appendix M	443
Appendix N	445
Appendix O	450
Appendix P	454

List of Tables

Table 2.1 Summary of strategies 5	1
Table 2.2 Comparison of chapter headings in key texts	5
Table 6.1 Contextual information about participants 14	0
Table 9.1 Contextual information about participants 24	8
Table 10.1 Contextual information about participants 26	6
Table 10.2 Resources recommended by students	5
Table 11.1 International Dyslexia Association Knowledge and Practice Standards for	
Teachers of Reading (IDA, 2018)33	0
Table 11.2 Knowledge and Practice Standards for Teachers of Music: Dyslexia specific33	3

List of Figures

Figure 2.1 Universal Design for Learning guidelines (CAST, 2018)72
Figure 2.2 Key questions to consider when planning lessons (UDL, 2020)73
Figure RS1.1 Adapted from Bronfenbrenner's ecological systems theory (1979)81
Figure RS2.1 Action research cycles87
Figure 3.1 A timeline of the research process showing cycles of action research
Figure 5.1 Difficulties reported by dyslexic students132
Figure 5.2 Adjustments in practical exams
Figure 5.3 Adjustments for music theory exams135
Figure RS4.1 Card from Alex's parents202
Figure RS5.2 Multisensory activity to form treble and bass clefs from clay207
Figure RS5.3 Music symbols code, created by Ben (2022-5-19)207
Figure RS5.4 Multisensory technique for reinforcing learning (Ben 2022-2-17)210
Figure RS5.5 Alex's composition from 2019211
Figure RS5.6 Alex's composition from 2021211
Figure 8.1 A comparison of the music literature on dyslexic strengths
Figure RS6.1 Key questions to consider when planning lessons (UDL, 2020)236
Figure 11.1 Framework for the assessment of resilience factors (Daniel & Wassell, 2002, p.
11)
Figure 11.2 A cumulative risk and resilience model of dyslexia (Catts & Petscher, 2022, p.
173)
Figure 11.3 A new risk-resilience model of dyslexia for musical training

Acknowledgements

I am especially grateful to God who has sustained me and whose Love is the ultimate mediator between my weakness and His strength in my life. He has met me graciously in countless *Kairos* moments to show me glimpses of who He really is.

But this I call to mind; therefore, I have hope. It is because of the Lord's loving kindnesses that we are not consumed, because His compassions never fail. They are new every morning; great and beyond measure is Your faithfulness. Lamentations 3:21-23

I would like to acknowledge my husband, Matt, and my children, Hannah and Noah (my sonin-law), Nate and Ben, who have been encouraging and supportive throughout this parttime journey of research. A special thanks to Ben who shared his wealth of video editing knowledge and skills so generously. I am grateful to my mum and dad, the late Dr Lawrence Killian and Lois Killian, for their dedication to education and support in all my endeavours in life. Thank you to my sister, Erin Johnson, for her sense of humour and many messages of warm support.

Thank you to 'Alex' and 'Ben', my piano students in this research. You and your families have inspired this journey. I am full of admiration for your perseverance, especially during the Covid-19 pandemic, and for the resilience with which you face the challenges of school and life. Lessons with you brought me much joy.

I want to express my deepest thanks to my friends, Rachel and Nigel Poulton, MBE. There are so many instances when your friendship and kindness have given me the courage and strength to keep going in difficult times. I will be forever grateful for your prayers and the many ways you have shown love to our family. Thank you for your encouragement.

I wish to express my heartfelt gratitude to my supervisor, Dr Liz Haddon. From our very first meeting at an interview for the MA course until the completion of this research, she has been an endless source of inspiration, encouragement, patient support and guidance. Her thought-provoking and wise words have enabled me to develop as a critical thinker and

keen observer. I admire her dedication to the field of music education and feel deeply grateful to have her as my mentor.

I am grateful to Mark Hutchinson who showed wise advice and patient understanding whilst also challenging me at every stage of the journey. I am thankful for the British Dyslexia Association music committee who have shown their support in many ways. Sally Daunt, former chair of the committee, graciously shared her wealth of experience and encouraged me from our first meeting. I am also grateful for the academic and administrative support from the School of Arts and Creative Technologies at the University of York. I have had many wonderful experiences being a part of this department and I am thankful for how they have helped me to develop as a teacher and researcher.

The summary table from the International Dyslexia Association's Knowledge and Practice Standards for Teachers of Reading (Table 11.1 in Chapter 11) has been included under the UK fair dealing copyright exception for the purposes of examination.

Author's declaration

I declare that this thesis is a presentation of original work and I am the sole author. This work has not previously been presented for a degree or other qualification at this University or elsewhere. All sources are acknowledged as references. This work received approval from the University of York Arts and Humanities Ethics Committee.

Chapter 1 INTRODUCTION

1.1 Introduction

This thesis presents the findings from an action research project with two case studies in the context of my piano teaching practice, informed by literature and data collected from a questionnaire, interviews with teachers, parents and dyslexic students, analysis of reflective observations from my piano teaching as well as an evaluative focus group. Despite ongoing debates regarding the definition of dyslexia (Elliot & Grigorenko, 2014; Kirby, 2020), a rise in social media groups with an emphasis on dyslexia, and the existence of a number of organisations (British Dyslexia Association¹, Helen Arkell Trust², Made by Dyslexia³, PATOSS⁴), it seems remarkable that research evidence to provide a better understanding of music education for dyslexic students is still quite limited.

In this chapter, the context (1.2), research aims and questions (1.3), an outline of the thesis (1.4) and an overview of each chapter (1.5) are presented.

1.2 Context

Dyslexia is characterised as a learning disorder (WHO, 1993; DSM-5, 2013). Depending on the criteria used, the prevalence of dyslexia is believed to be around 10% of the general population (BDA, 2024; Wagner et al., 2020). The most common disadvantage is demonstrated in literacy skills (Reid, 2016), but research has shown that dyslexia may involve difficulties relating to speech, writing abilities, mathematical abilities, motor coordination, visuospatial and attentional abilities (Habib, 2021). Dyslexic children showed impairment in executive functions, such as planning, organising and working memory (Akyurek & Bumin, 2019). Secondary traits of dyslexics include low self-esteem and poor self-concept (Livingston et al., 2018; Ridsdale, 2005) potentially leading to what is known as a state of 'learned helplessness' (Peterson et al., 1993, p. 8).

¹ https://www.bdadyslexia.org.uk/

² https://helenarkell.org.uk/

³ https://www.madebydyslexia.org/

⁴ https://www.patoss-dyslexia.org/

Learning to play a musical instrument is a complex undertaking which uses a network of neurological processes. Music involves features which are concrete and abstract: pitch, timbre, dynamics and rhythm. Furthermore, the visual analysis of a music score initiates activation in a global neural network of attention and memory whilst task-specific networks enable a musician to separate and analyse distinct elements in the music (Baeck, 2002; Stenberg & Cross, 2019; Tallal & Gaab, 2006). Using this information to inform limbic, motor and sensory systems, the brain coordinates movement with emotional response, concentration and discipline (Bresin & Friberg, 2011; Gudmundsdottir, 2010; Jäncke, 2008). Whilst reading a music score, the brain is required to recognise and process patterns, to anticipate and manage time intervals, to rank and to select which information is most relevant (Gudmundsdottir, 2010; Fournier et al., 2019; Wristen, 2005).

Practising the instrument involves focused attention, understanding musical symbols and the ability to keep sequences in the working memory (Couvignou & Kolinsky, 2021; Kim-Boyle, 2017; Williamon, 2012). These skills normally become automatic through practising (Schlaug et al., 2005; Stewart, 2008) but a dyslexic student⁵ may require more time for these skills to become automatic. A difficulty with dyscalculia, or numbers, may affect the understanding of rhythm values and fingering numbers (Oglethorpe, 2008). Executive functions, affected in dyslexic students, are necessary for successful consolidation of information in music learning, but also for practical aspects such as remembering music books and time management. With so many of these skills corresponding to the processing difficulties associated with dyslexia, students and instrumental music teachers may find the experience challenging.

Musical activities have been employed by researchers as an intervention aimed at improving literacy skills (Cancer & Antonietti, 2022; Cogo-Moreira et al., 2013; Flaugnacco et al., 2015; Habib et al., 2016; Overy, 2003). Music learning has the potential to assist dyslexic students with global cognitive development, in addition to being a gratifying space to create and perform music (Habib, 2021). Recent research has suggested that dyslexic individuals might have a number of strengths connected with creativity (Cancer et al., 2016; Gobbo, 2020;

⁵ There is some debate between using the term 'dyslexic students' as opposed to 'students with dyslexia' related to identity versus person constructs of disability. For the sake of consistency, the term 'dyslexic students' is used in this thesis.

Jantzen, 2009; Schneps et al., 2012), reasoning abilities (Eide & Eide, 2011) and entrepreneurial skills (Logan, 2009). Adaptable teaching methods and greater awareness among teachers may allow dyslexic students to enjoy their educational experiences and may enhance their motivation to engage in more self-directed learning (Reid, 2016; Rolka & Silverman, 2015).

The British Dyslexia Association⁶ provides a handbook (Daunt, 2012) and printable resources on their website. Currently, there are two main texts which support further understanding: Instrumental music for dyslexics: A teaching handbook (Oglethorpe, 2008) and Music and dyslexia: A positive approach (Miles et al., 2008), as well as a small number of journal articles (Flach et al., 2016; Ganschow et al., 1994; Heikkila & Knight, 2012; Hubicki & Miles, 1991; Jaarsma et al., 1998; Macmillan, 2004; Morrow, 2023; Nelson & Hourigan, 2016; Reifinger, 2019; Vance, 2004). In 2019, the Associated Board of Royal Schools of Music printed a booklet of teaching strategies and information for music teachers entitled 'Making music accessible' (ABRSM, 2019). However, there is more to be explored and understood about dyslexic students and their views on the music learning process including their experiences of graded music exams. Further investigation is needed on the perceptions of instrumental music teachers as to how they acquire knowledge about dyslexia, develop their teaching approaches for dyslexic students, prepare students for graded music exams and cultivate relationships with dyslexic students and their parents. The area of strengths has been addressed from the dyslexic musicians' perspective (Ganschow et al., 1994; Nelson & Hourigan, 2016) and from teachers' observations (Hubicki & Miles, 1991; Oglethorpe, 2008), but there is scope to understand more about these strengths from multiple perspectives (parent/teacher/student) and to consider how they might be utilised in the learning process to encourage the student. The importance of parental involvement has been noted (Oglethorpe, 2008), but there is a need for greater awareness related to parents' needs, how they might support their dyslexic child and factors that promote positive collaboration with the teacher.

⁶ https://www.bdadyslexia.org.uk/advice/children/music-and-dyslexia

1.3 Research aim and research questions for this study

In light of a lack of research evidence contributing to an understanding of music education for dyslexic students, relevant teaching strategies and approaches and the interactions of the roles of parent, teacher and student, the aim of this research was to improve and enhance instrumental music education for dyslexic students through contributing to a body of research-informed knowledge which could support instrumental teachers. The following research questions (RQs) were articulated:

- What are the perceptions of music teachers regarding their experiences and utilisation of pedagogical practices, including strategies, methods and material, in teaching dyslexic students?
- 2. Do adjustments and accommodations which music exam boards allow offer dyslexic students unbiased inclusion? What are the perspectives of teachers and students on this topic?
- 3. Do teachers, parents and students perceive dyslexic students to have specific strengths? How might these strengths be utilised in the music lesson context?
- 4. What are dyslexic students' perceptions of their music learning experiences? How might their voices be nurtured and amplified in the pedagogical process?
- 5. What are the teacher, student and parental perceptions of the pedagogical process and roles in the music lesson context? How can these roles collaborate to collectively enable positive outcomes?

1.4 Outline of the thesis

This thesis consists of twelve chapters and seven reflective statements. Cycles of research were flexible and iterative representing the spiral nature of action research (Kemmis & McTaggart, 1988). In order to achieve a cohesive and clear structure, reflective statements are interspersed with the main chapters, thereby providing evidence of how the findings were embedded in and impacted my teaching practice. The reflective statements are also designed to give context to my experiences as a teacher (RS 1-3), but also to describe transitions which characterised my development (RS 4-7). The following outline of this thesis is composed of three main sections and depicts how chapters are paired with reflective statements.

Part 1: Introduction, Reflective Statements 1-2 and Methodology

- Chapter 1: Introduction to the thesis
- Chapter 2: Literature review
 - Research Statement 1: Presenting the researcher
 - Research Statement 2: Presenting the students
- Chapter 3: Methodology of the research
- Chapter 4: Positionality of the researcher
- > Chapter 5: Initial survey of dyslexic students' experiences of music exams

Part 2: Findings and Reflective Statements 3-7

- > Chapter 6: Teacher perceptions of barriers and challenges
 - Reflective Statement 3: Understanding the barriers and challenges encountered by the researcher
- Chapter 7: Teacher perceptions of dyslexic students' challenges and strategies
 - Reflective Statement 4: Understanding power and knowledge dynamics between teacher and student
 - Reflective Statement 5: Developing tacit knowledge as a teacher
- > Chapter 8: Teacher perceptions of dyslexic students' strengths
 - Reflective Statement 6: Developing a strengths-focused approach
- Chapter 9: Findings from student interviews
- > Chapter 10: Findings from parent interviews
 - Reflective Statement 9: Developing empathic understanding

Part 3: Discussion and Conclusion

- Chapter 11: Discussion of the findings
- Chapter 12: Summary, limitations and conclusion

1.5 An expanded overview of each chapter

The following section expands on the previous outline by offering a more detailed description of each chapter.

Chapter 1 discusses the context and rationale for beginning the research, as well as the aims and research questions.

Chapter 2 reviews literature related to music and dyslexia, including an overview of inclusive music education in the UK, the challenges faced by dyslexic students in learning music, strategies advocated by researchers and practitioners as well as an analysis of existing pedagogical materials.

Reflective Statement 1 addresses the motivations and challenges experienced during the research related to the nature of investigating my own teaching practice as well as challenges associated with the Covid-19 pandemic and introduces Bronfenbrenner's (1979) ecological systems theory to evidence my development through the course of the research.

Reflective Statement 2 introduces Bronfenbrenner and Morris' (2006) bioecological model which focuses on the person, place, context and time (PPCT) and explains the importance of proximal processes to drive development. An introduction to the students in my teaching practice, the teaching environment and context of our lessons follows.

Chapter 3 provides information on the research design, including the theoretical framework of social constructivism and the methodology of action research, methods of data collection, data analysis procedures, validity and reliability of the research and ethical issues.

Chapter 4 discusses the role of the researcher, considering the positionality of being an insider-researcher as well as the challenges and ethical issues which arise from that role.

Chapter 5 reports data analyses from a questionnaire investigating dyslexic students' perceptions of graded music exams, the results of which informed the semi-structured interviews with teachers and students.

Chapter 6 discusses the findings from teacher interviews with themes of barriers and challenges experienced in their teaching of dyslexic students.

Reflective Statement 3 presents my reflections on the themes of 'challenges' and 'barriers', which arose from the teacher interviews, in light of my own teaching experiences and how they provided motivation for this research.

Chapter 7 presents findings from teacher interviews with themes of student challenges observed by teachers and corresponding strategies used by the teachers. The following reflective statements (RS 4 and RS 5) correspond to these findings and give evidence of the researcher's observations of student challenges and the decisions made regarding strategy use in their teaching practice, as well as reflecting on other themes such as teaching approaches, lesson planning, motivation and students' self-regulated learning.

Reflective Statement 4 describes my transition from a master-apprentice paradigm to a new paradigm in which I reconsider the balance of power and knowledge between teacher and student. I reflect on the impact of this changing dynamic on the students' development of metacognitive skills and their motivation.

Reflective Statement 5 consists of the analysis of the framework of Universal Design for Learning (CAST, 2018) used in conjunction with strategies identified by researchers and practitioners to inform my lesson planning, and the growth in my understanding of the role of tacit knowledge in my teaching practice.

Chapter 8 presents the findings from teachers' perceptions of dyslexic students' strengths with implications for how they might be utilised in the teaching process.

Reflective Statement 6 addresses the researcher's transition from a deficit perspective of dyslexia to a strengths-focused approach. Also discussed in this statement is the rationale for this approach as well as practical implications for teaching and the student.

Chapter 9 presents the findings from parental perspectives and discusses implications for teachers and parents.

Chapter 10 reports analyses of findings from student perspectives, including their experiences, challenges, strategies, strengths and ways they envision change in future music education.

Reflective Statement 7 presents my development in considering the role of empathic understanding with students and their families, as well as reflections on the importance of knowing how to prevent compassion fatigue as a teacher.

Chapter 11 discusses analyses of the findings overall against the backdrop of the research questions and relevant literature, leading to the creation of two tools: an innovative risk-resilience model for musical training with dyslexic students and an adapted Knowledge and Practice Standards document to provide the foundation for a training course for teachers.

Chapter 12 provides a summary of the study, the limitations of this research, implications for practice and recommendations for future research.

The thesis concludes with a reference list and appendices.

Chapter 2 LITERATURE REVIEW

2.1 Introduction

This chapter begins by describing the context and role of the researcher (2.2) and discusses the definitions (2.3.1), underlying causes (2.3.2) co-occurring conditions (2.3.3), secondary characteristics (2.3.4) and support factors (2.3.5). This is followed by an overview of music and dyslexia research with a focus on the use of music as an intervention (2.4), analysis of music teaching and dyslexia materials (2.5 and 2.6) and accessible teaching practices (2.7). Finally, the research questions are identified (2.8).

2.2 Context and role of researcher

This research has emerged from my experiences as a piano teacher working with students of all ages, both with and without dyslexia, in York, UK. I found my initial attempts to work with dyslexic students disheartening. Assumptions made based on past learning and teaching experiences meant that I experienced frustration when teaching dyslexic students. Becoming more aware of the complexities and co-occurring conditions of dyslexia, my professional and personal interest developed, leading to my pursuit of doctoral research on the topic.

My contribution to this research was as an active participant within my roles as teacher and researcher: the teacher in the instrumental lessons, designer of the research project, analyser of relevant data and collator of results. I found teaching piano lessons rewarding and discovered that building a relationship with students who differ vastly in terms of personality, interest and capability is a unique privilege.

One teaching example propelling my interest in the research topic is as follows: a student did not progress at a level which corresponded to her peers. Significant difficulty in locating notes and coordinating hands, memory issues, struggles differentiating between left and right and a fragile self-concept were the key indicators I observed. We tried many approaches based on knowledge gained through my time as a student and my experience of teaching, but they were not successful. While the student's reading and writing skills appeared average for her age (9 years old), I learned that she had never learned to ride a bike or participate in sports which seemed to suggest that she may have coordination

difficulties. I wondered if she was practising sufficiently, but also considered that I might not be equipped to teach her well.

Research suggests that parental involvement is key to positive results in instrumental learning (Creech & Hallam, 2003; Macmillan, 2004) with Creech and Hallam (2003) describing the communication process between parents, teachers and students as 'circular' in nature (p. 39). A discussion with my student's mother provided some confirmation that specific learning difficulties might be involved. As I later came to learn, a substantial number of characteristics pointed to dyslexia, and to dyspraxia, suggested by poor motor coordination. At that point, I lacked an understanding of how adaptive teaching strategies might improve the learning experience and outcomes for this student. The analogy I found most constructive for viewing the situation from the student's perspective was when I compared it to my own experiences of moving to a new country where I did not speak the language. Although I wanted to communicate, my attempts to do so were met with frustration until I learned the new language or unless people knew mine. I imagined that my student might feel the frustration of being approached during piano lessons with strategies that were like a 'new language' for her.

Developing a better understanding of how to teach this student was a primary motivating factor for embarking on an MA in Music Education: Instrumental and Vocal Teaching at the University of York in 2016. I undertook the course requirements: producing lesson commentaries, essays and analysis of my video-recorded piano lessons. I was challenged by discussions with my supervisor Dr Elizabeth Haddon and through reviewing the academic literature on the topic. This enabled a pivotal shift in my understanding of the key factors of engaging in reflective practice and focusing on the student's voice in the learning process. As I gained experience, continued to analyse my recorded lessons and incorporated feedback from others, I began to develop as a 'reflective practitioner' (Schön, 1992) and continued to encounter obstacles which increased my curiosity about how to improve my teaching with dyslexic students. I realised there was more scope to research the interactions between student and teacher and to consider the role of parents within the pedagogical process. From a personal and professional point of view, my interest in the topic was stimulated. However, whilst there was a substantial amount of information about the use of music as an intervention to improve literacy, there was a limited amount of available

literature on instrumental music teaching and this was the motivation behind beginning this research project as my doctoral research.

This literature review begins with a discussion of how dyslexia is defined, the socioemotional impact of dyslexia and support factors (2.3). Following this, the connections between music and language are considered (2.4), including studies on the use of music training to transfer positive benefits to language and literacy domains, with a focus on recent research in this area. Pedagogical materials and research pertaining to music teaching for dyslexic students are then presented (2.5 and 2.6) and areas for further research are considered. There is a brief overview of inclusive music teaching and access to provision (2.7). Finally, the research questions for this research are discussed in light of the material which has been reviewed (2.8).

2.3 Dyslexia: Definition and characteristics

The following sections discuss the definitions (2.3.1), underlying causes and implications for education (2.3.2), co-occurring conditions (2.3.3), secondary characteristics (2.3.4), and intrinsic and extrinsic support factors (2.3.5) including the strengths associated with dyslexia (2.3.5.3).

2.3.1 Defining dyslexia

Dyslexia is a complex condition to define, but resulting implications for research, education and government policy are immense. This section explores diverse definitions of dyslexia and considers their implications. Depending on the criteria used, the prevalence of dyslexia ranges from 3-7% (Rutter et al., 2004, Snowling & Melby-Lervåg, 2016; Wagner, 2020) and 17-21% (Ferrer et al., 2015) of the English-speaking population. The breadth of variability in the characteristics and degrees to which dyslexia presents mean that a high proportion of dyslexics may not be identified (Shaywitz & Shaywitz, 2005).

Although the first definition mentioned uses the term 'disability', due to the negative connotations, the use of the word 'condition' is recommended. The following definition of dyslexia is widely recognised:

Dyslexia is a specific learning disability that is neurobiological in origin. It is characterized by difficulties with accurate and/or fluent word recognition and by poor spelling and decoding abilities. These difficulties typically result from a deficit in the phonological component of language that is often unexpected in relation to other cognitive abilities and the provision of effective classroom instruction. Secondary consequences may include difficulties in reading comprehension and reduced reading experience that can impede growth of vocabulary and background knowledge. (Lyon et al., 2003, p. 2)

In 2009, the UK government commissioned an independent report, *Identifying and Teaching Children and Young People with Dyslexia and Literacy Difficulties* (Rose, 2009), which resulted in a number of recommendations for teacher training and specialist support for dyslexic students. In the Rose Report (2009), the definition of dyslexia includes the addition of co-occurring difficulties:

Dyslexia is a learning difficulty that primarily affects the skills involved in accurate and fluent word reading and spelling. Characteristic features of dyslexia are difficulties in phonological awareness, verbal memory and verbal processing speed. Dyslexia occurs across the range of intellectual abilities. It is best thought of as a continuum, not a distinct category, and there are no clear cut-off points. Cooccurring difficulties may be seen in aspects of language, motor coordination, mental calculation, concentration and personal organisation, but these are not, by themselves, markers of dyslexia. A good indication of the severity and persistence of dyslexic difficulties can be gained by examining how the individual responds or has responded to well-founded intervention. (Rose, 2009, p.11)

The British Dyslexia Association utilises the Rose report definition on their website but adds the following:

In addition to these characteristics, the British Dyslexia Association acknowledges the visual and auditory processing difficulties that some individuals with dyslexia can experience, and points out that dyslexic readers can show a combination of abilities and difficulties that affect the learning process. Some also have strengths in other

areas, such as design, problem solving, creative skills, interactive skills and oral skills. (BDA, 2024)

Without assessment and diagnosis, adequate care provision plans may be delayed and students may not receive reasonable adjustments and accommodations shown in many cases to improve outcomes for them. Some researchers believe that the assessment process and labelling individuals as 'dyslexic' is unnecessary and may create disadvantages for individuals from low socio-economic situations, suggesting instead that teacher training and specific responses to intervention would be more helpful for students struggling to read (Elliott & Grigorenko, 2014; Gibbs & Elliott, 2020). At the time this thesis was being written, a new definition of dyslexia was in the process of being drafted from a Delphi study to clarify areas of confusion and provide a new framework for assessment (Snowling & Hulme, 2024).

2.3.2 Underlying causes

A number of theories have arisen to explain the causes of dyslexia and these will be briefly discussed using a causal modelling framework which takes into account biological, cognitive and behavioural factors as well as environmental factors influencing dyslexia at each of the first three levels (Frith, 2002). How dyslexia is conceptualised and understood has implications for educational practice and there is agreement that both practitioners and researchers must collaborate on evidence-based approaches so that interventions and remediations might be improved (Hulme et al, 2015; Snowling & Hulme, 2024; Stein, 2018). Snowling and Hulme (2024) note that cognitive and behavioural measures are less costly, more reliable and more effective at assessing students and informing interventions than methods using neuroimaging, stating that 'we do not see the rationale for including in a definition of dyslexia a reference to its likely neurobiological causes' (p. 357).

There is evidence that dyslexia may be heritable (Almahrag, 2022; Kere, 2014; Wolf et al., 2024) and brain imaging studies have identified structural and functional differences in the brains of dyslexic individuals (Ozernov-Palchik & Gaab, 2016; Norton et al., 2014). As Wolf et al. (2024) note, these differences in brain structure have been widely understood in terms of deficits or impairments but not in terms of the potential strengths that they might confer, concluding that 'research into educational practices to support the strengths associated with dyslexia is still limited' and that 'asset-based teaching pedagogies for students with

dyslexia' deserves further investigation (Wolf et al., 2024, p. 315). Furthermore, Vaughn et al. (2024) express concern that an emphasis on neurobiological factors might contribute to the view of dyslexia being an 'immutable condition' and lessen openness to the potential of remediation and support factors to positively impact dyslexic individuals (p. 327).

Reading is a complex activity which draws on auditory and visual processing systems in the brain. The visual ability to recognise words and symbols (orthography) is an important factor in learning to read and write. Speech, language and visual systems rely on executive function processes like working memory, attention and concentration which influence and are influenced by aspects of emotion and reflection (Wolf et al., 2024). The ability to process information quickly and automatically affects the speed and comprehension of reading. Specific deficits in these processes have been identified in an attempt to explain the causes of dyslexia and to guide remediation strategies. The following theories are briefly discussed: phonological deficit theory, double deficit theory, cerebellar deficit theory, magnocellular deficit theory and temporal sampling theory. Although there is ongoing debate in the field of theoretical causes, the phonological deficit theory has received the most support. This represents a cause at the cognitive level, but other theories have attempted to address phonological, visual and motor deficits at the biological level.

The phonological deficit theory, representing a low ability to detect and process sound segments (phonemes) which form words has been the most prominent theory at the cognitive level (Snowling, 2000; Vellutino et al., 2004) and has driven many of the interventions for the remediation of dyslexia. However, the phonological deficit does not explain some of the other factors which may be experienced by dyslexic individuals in terms of auditory (other than the phonological deficit), visual and motor impairments (Ramus et al., 2003). The double deficit theory (Wolf & Bowers, 1999) suggests that the phonological deficit and the rapid automatized naming⁷ deficit (Denckla & Rudel, 1976) together might be the cause of reading difficulties indicating that interventions may need to address phonological skills as well as processing speed. The cerebellar deficit theory (Nicolson & Fawcett, 2005) postulates a cause of dyslexia at the biological level with effects on phonological skills and speed of processing at the cognitive level leading to challenges with

⁷ Rapid automatized naming refers to the ability to quickly name a series of items (Wolf & Bowers, 1999).

language, reading and motor coordination at the behavioural level. The magnocellular deficit theory (Stein, 2018) suggests that there is reduced sensitivity in magnocellular pathways leading to visual processing problems which may impact the ability to read and write fluently. The temporal sampling theory (Goswami et al., 2013) links auditory processing difficulties, particularly the ability to process the sounds of speech, with poor phonological awareness leading to difficulties with reading. This is discussed in greater detail (see section 2.4.1). These theories highlight for educators the importance of being aware that dyslexia may manifest itself in a variety of ways, that interventions may not work in each particular case and that although there is a substantial amount of evidence to support the phonological deficit theory, this alone does not account for all the difficulties experienced by dyslexic individuals. This has led to the growth of multifactorial models (Catts & Petscher, 2022; Haft et al., 2016) to explain the interactions between risk factors (for example, phonological deficit, co-occurring conditions, familial risk) and other environmental factors such as support factors (see section 2.3.5).

It is now believed that genetic, environmental, brain structural and perceptual/cognitive factors could all be involved in causing dyslexia (Catts & Petscher, 2022; Haft et al., 2016; Ozernov-Palichik & Gaab, 2016; Peterson & Pennington, 2012). Multiple causal factors (McGrath et al., 2020; Snowling, 2020) and co-occurring conditions may explain the vast diversity in the characteristics and cognitive abilities of individuals with dyslexia. This presents particular challenges for educators in terms of the variability that may be expected in dyslexic learners. However, evidence for the effectiveness of interventions, appropriate instruction and other factors has shown the potential of neuroplasticity to aid remediation. The following section discusses co-occurring conditions in greater detail (2.3.3).

2.3.3 Co-occurring conditions

Research has focused on the differences between dyslexic and non-dyslexic individuals to increase understanding of the causes of dyslexia and to provide effective interventions. Snowling et al. (2020) note that despite dyslexia being seen primarily as a difficulty with literacy, 'the history of dyslexia captures a sense of complexity' (p. 501), not to mention the many dimensions of co-occurring conditions which may also be implicated. Individuals with dyslexia may have additional challenges with language, numbers, attention or motor

coordination, as detailed earlier by Rose (2009). Co-occurring conditions include attentiondeficit hyperactivity disorder (between 20-40%), difficulties with maths or dyscalculia (around 40%) and around 30% with autism spectrum disorder (Hendren et al., 2018). Developmental coordination disorder is also correlated with dyslexia (Brimo et al., 2021). Research suggests that there may be a correlation between amusia⁸ and dyslexia. A study suggests that as many as 30% of adults with dyslexia have amusic traits, whilst 25% of adults demonstrated characteristics of dyslexia (Couvignou et al., 2019). In a sample of 38 dyslexic children, 34% were found to have traits of amusia, which is substantially higher than the rate of amusia (1.5-4%) for the general population (Couvignou & Kolinsky, 2021).

2.3.4 Secondary characteristics

Reading difficulties, along with other characteristics associated with dyslexia, may mean that dyslexics express unwillingness to go to school, may avoid challenging tasks or demonstrate poor behaviour as a way of coping with stress (Livingston et al., 2018). A lack of parental or educational support may negatively impact students, resulting in low selfesteem and anxiety (Francis et al., 2019; Novita, 2016; Ridsdale, 2004; Singer, 2005). Research (Leitão et al., 2017; Wilmot et al., 2023) indicates that this may increase stress for the family, with dyslexic children internalising their worries and feeling a sense of shame at causing parental anxiety. Washburn et al. (2014) note that mothers tend to have the role of 'investigators' (p. 123) and that the level of their knowledge of dyslexia plays a part in their ability to act as advocates for their children (Reid, 2016). Negative educational experiences have been found to have consequential effects on motivation and emotional well-being (Wilmot et al., 2023) with reports of isolation and bullying common (Humphrey & Mullins, 2002). The result of these persistently negative experiences may lead to a state of 'learned helplessness' in which the individual with dyslexia may lose confidence in their capability of learning (Maier & Seligman, 2016).

By considering support factors (see Section 2.3.5) in education, it may be possible to reduce or lessen the development of secondary characteristics and subsequent negative impact on dyslexic individuals, the learning process and their families. The next two sections consider

⁸ Amusia is defined as the inability to perform specific tasks including recognising a melody without the lyrics, distinguishing incorrect notes or perceiving music that is not in tune (Ayotte et al., 2002).

support factors, intrinsic and extrinsic, which may contribute to better educational and emotional outcomes, as well as assisting the dyslexic individual to reach the full potential of their capabilities (2.3.5).

2.3.5 Support factors

In light of the serious impact of secondary characteristics on socio-emotional well-being, there are a number of support factors identified in the dyslexia literature. These may encourage resilience and mitigate the negative impact of the secondary characteristics (Catts & Petscher, 2022). Factors may be intrinsic, related to the individual, or may be extrinsic, related to their environment.

2.3.5.1 Intrinsic support factors

Support factors related to the individual include positive perceptions of the label of dyslexia, feelings of hope, self-determination, growth mindset and a sense of personal agency (Catts & Petscher, 2022; Haft et al., 2016; Singer, 2005; Wilmot et al., 2023). Shaywitz et al. (2016) state that 'self-awareness and self-knowledge, gained by an accurate diagnosis of dyslexia, brings in the light and allows the person to understand himself' (p. 283). Strengths, which may be considered an intrinsic support factor, are discussed in more detail in section 2.3.5.3. This emphasises the importance of an accurate understanding of dyslexia and how it affects each person individually, as well as an understanding of the role of positive mindsets and approaches in promoting dyslexic students' resilience.

2.3.5.2 Extrinsic support factors

Extrinsic factors may be related to the family and the student's educational environment. Of particular importance is the role of parental involvement (Catts & Petscher, 2022; Haft et al., 2016; Wilmot et al., 2023). Communication between the parent and their child and parental attitudes toward dyslexia were found to be valuable factors in effective support (Reid, 2016), with parents often acting as advocates between the school and their child, negotiating better terms on their behalf. Parental support, in terms of encouragement and advocacy, may involve minimising stigma and managing expectations (Ross, 2019). However, Glazzard (2010) suggests that both a diagnosis and parental support are beneficial, but notes that in some cases 'negative influences from both teachers and peers negated the positive support provided by parents' (p. 68).

By promoting their child's strengths and encouraging resilience in the face of challenging situations with school or teachers, parents may not only act as advocates but also teach their children to advocate for themselves (Alexander-Passe, 2008). However, research shows that parents of dyslexic students do not always feel supported or able to access the information necessary to make informed decisions about their child's education; they may lack the necessary trust in the professionals involved and may not feel confident in their knowledge of dyslexia or the options available to them (Ross, 2019; Wilmot et al., 2023). In light of the role of heritability and the possibility that parents of dyslexic students might themselves be dyslexic, this may create challenges for them as to how they acquire knowledge about dyslexia, advocate on the student's behalf and the accessibility of written policy statements or communication from the school.

Catts and Petscher (2022) affirm this, suggesting that factors such as 'instruction, growth mindset, task-focused behaviour, adaptive coping strategies, family and peer support' might offset risk factors associated with dyslexia (p. 173). There may be challenges for students concerning their awareness of dyslexia, and how it affects them individually and their learning environment, but an awareness of dyslexia and self in addition to parental and community support may mitigate these. Wolf et al. (2024) also emphasise the role of appropriate instruction, stating that:

Dyslexia can change over time, particularly when the strengths and advantages of its unique differences in brain organization are fostered alongside preventative factors that emphasize resiliency, persistence of effort, and respect for the potential of every individual. (Wolf et al., 2024, p. 317)

2.3.5.3 Strengths associated with dyslexia

Another support factor may be the more positive view of dyslexia and related strengths which is emerging (Eide & Eide, 2011; Malpas, 2017; Reid, 2016; Rooke, 2015). Some of these strengths may have developed as a coping mechanism for the difficulties created by dyslexia (Everatt et al., 2008). Creativity, high-level reasoning strengths and character strengths have been explored within research, showing that dyslexics are prevalent in creative fields such as art and design, engineering and architecture (Alden & Pollock, 2011; Cancer et al., 2016; Lemon & Shah, 2014; Rooke, 2015; Wolff & Lundberg, 2002). Research
has identified a high number of dyslexic individuals among entrepreneurs, suggesting creative innovation and social skills (Logan, 2009). An understanding of visual representations and intuitive thinking may be associated with differing brain structure and functioning in dyslexics with both strengths seen as important factors for creative thinking (Cancer et al., 2016; Casanova et al., 2002; Shaywitz & Shaywitz, 2016; Schneps et al., 2012). Kapoula et al. (2016) found that children and adolescents with dyslexia scored higher on creativity tests than non-dyslexic controls, although they suggest that the learning environment also plays a role in supporting these strengths through personalised instruction. Conversely, in a meta-analysis of dyslexia and creativity, Majeed et al. (2020) found that higher levels of creativity were demonstrated in dyslexic adults only.

In addition to creativity, Eide and Eide (2011) propose that dyslexics may display high-level reasoning strengths which they describe as material, interconnected, narrative and dynamic. Material reasoning is described as 'three-dimensional spatial reasoning'. One example of supporting research for this involves dyslexic adolescents who outperformed non-dyslexics in the reconstruction of 3-D images using virtual reality (Attree et al., 2009). Interconnected reasoning is the ability to make connections between diverse concepts. Research to support this includes reference to a dyslexic advantage of employing peripheral rather than central vision (Schneps et al., 2012). Narrative reasoning refers to the use of story-telling or visual images to remember concepts and processes. Dynamic reasoning involves the ability to predict through past pattern recognition what might happen in future settings. Character strengths such as the ability to network and perform well as part of a team (Rowan, 2010), determination and resilience are also noted as dyslexic strengths (Kannanga et al., 2018). Some research suggests a strength in the area of empathy (Sanderson-Mann & McCandless, 2006), and research with dyslexic children found them to have greater emotional reactivity, which is believed to be associated as a social strength (Sturm et al., 2021).

Positive psychology has the potential to inform educational philosophy as an approach which acknowledges areas of weakness but seeks to identify and leverage the use of strengths to promote learning (Magyar-Moe, 2009). Frederickson's (2004) theory suggests that positive emotions broaden an individual's openness to new ideas and interactions. Furthermore, Seligman's (2011) positive psychology approach focuses on positive emotions,

traits and the creation of positive communities to facilitate human flourishing. Unfortunately, this strengths-focused approach is not commonly utilised when it comes to dyslexia, which most commonly has been correlated with deficits.

The next section of this chapter focuses on the connections between music and language and how this has driven music interventions for improving literacy in dyslexic students. An overview of music teaching and dyslexia literature follows.

2.4 Music training as an intervention for dyslexia

The connections between music and language, particularly in the area of early childhood development, have been investigated for a number of purposes. This field has evolved in the last twenty-five years from a pedagogical focus on instrumental teaching to the use of music as an intervention. Research on music and dyslexia is now predominantly focused on the use of music training as an intervention to improve aspects of literacy but also includes some research on the related topic of music abilities as a potential early predictor for dyslexia or reading difficulties. There is far less research and pedagogical material concerned with music teaching for dyslexic students in comparison. The music teaching and dyslexia research is focused on the challenges experienced by dyslexic musicians and their teachers with descriptions of effective compensatory strategies and approaches. This highlights the need for the provision of high-quality and evidence-based information for music teachers with dyslexic students.

This literature review begins with a brief overview of the connections between music and language (2.4.1), the theories which underpin the use of music as an intervention for literacy remediation (2.4.2) and evidence for specific types of music training (2.4.3) followed by a summary (2.4.4). The second half of this literature review explores an overview of music teaching and dyslexia literature (2.5) including the challenges reported by dyslexic musicians and music teachers and the strategies they correlated with these difficulties. Next, a discussion of three books on music and dyslexia includes an analysis of the content and accessibility (2.6). The next section considers general inclusive teaching approaches and provision to access (2.7). Finally, the literature is summarised and discussed leading to the formulation of the research questions (2.8).

2.4.1 Connections between music and language

Music and language are both methods of expression and communication, aural and written, which rely on similar hierarchical structures: sequences of acoustic melodic and rhythmic patterns which occur over time and are based on smaller units such as phonemes or notes. These smaller units are represented by symbols: letters and speech marks in language correlate to music notes and articulation marks in music. Sounds are distinguished by duration, frequency, intensity and timbre. In both music and speech, there are elements of timing and prediction (Patel & Iverson, 2014; Tillmann, 2012). The rhythm of speech is understood through changes in duration and pitch of sound (intonation, stress) which convey meaning and guide the listener's understanding (prosody). Rhythm, in speech and music, is organised into metrical structures. An ability to map the symbol to sounds is foundational for reading and writing in both language and music. In music, notes and intervals are combined to create melodic and harmonic phrases in a piece of music; similarly, in language, phonemes and morphemes are combined to create words and sentences. This requires phonological awareness or the ability to recognise and manipulate sounds (Hulme et al., 2015). These sounds are organised into words and phrases which relate to one another structurally in language through what is known as syntactical processing; the meaning of these words and phrases is understood through semantic processing (Patel, 2007). In terms of purpose and structure, language and music have many similarities.

Dyslexia may affect both visual, auditory and motor processing. Auditory processing refers to peripheral (reception of sound) and central auditory systems (analysis of sound). The peripheral system consists of the auditory nerve and auditory pathways in the brain, whilst the central system includes the brainstem, thalamus and cortex. Temporal processing is the processing of sound over time (Patel, 2007). The processing of music and language requires the use of memory and motor systems in the brain, which may be affected by dyslexia. As mentioned in section 2.3.3, the prevailing theory of dyslexia is based on a phonological deficit which may affect the processing of rhythm in speech and music (Boll-Avetisyan et al., 2020; Goswami, 2019). Both music and language consist of stressed and unstressed sounds. In music, these sounds are regulated by pulse and predict metrical structures. Speech rhythm is understood by the amplitude envelope, which contains information through

patterns of duration, rhythm, tempo and stress information (Goswami, 2019; Meyer et al., 2019). Both the ability to process speech and to acquire language skills come from an accurate processing of the amplitude envelope. Tierney and Kraus (2013) identify several sub-skills relating to reading: 'phonological awareness, speech-in-noise perception, rhythm perception, auditory working memory and an ability to learn sound patterns and show that they are correlated to music processing abilities' (p. 209).

Some studies have shown that dyslexic music difficulties may include synchronising with a metronome (Colling et al., 2017; Overy, 2003; Thomson & Goswami, 2008; Wolff et al., 2002), identifying rhythm or metre changes (Goswami et al., 2013; Huss et al., 2011), pitch discrimination and pitch in the short-term memory (Besson et al., 2007; Ziegler et al., 2012; Weiss, 2014) and tone deafness (Couvignou et al., 2019; Couvignou & Kolinsky, 2021). Research has indicated that there may be difficulty in dyslexia with synchronising to the speech envelope (Leong & Goswami, 2014; Molinaro et al., 2016). Ladányi et al. (2020) suggested that poor rhythm perception was linked to neurodevelopmental disorders, including dyslexia.

These challenges in rhythm for dyslexics correlate with poor phonological awareness and reading difficulties (Cancer & Antonietti, 2018; Flaugnacco et al., 2015; Forgeard et al., 2008; Goswami, 2011; Huss et al., 2011; Ozernov-Palchik et al., 2018; Thomson & Goswami, 2008). Studies have appeared to confirm the TSF theory, the impairment of rise-time processing in dyslexics as a result of poor entrainment to neural oscillations (Goswami et al., 2016; Huss et al., 2011). These findings emphasise the connection between mechanisms of rhythm and language processing and support the rationale for music training as an intervention. This is described in greater detail in the next section.

2.4.2 Rationale for the use of music as an intervention for dyslexia

A number of theories have been put forward to explain the connections between music training and language processing. Research indicates that music and language may share some neural resources (Albouy et al., 2020; Koelsch, 2011; Schön et al., 2010; te Rietmolen et al., 2024). However, there is also evidence that they may have separate and distinct areas of processing (Boebinger et al., 2021; Chen et al., 2023; Blank & Fedorenko, 2020). Based on the theory that some neural systems are shared, music training has been investigated as a

means of improving language processing (Besson et al., 2011; Flaugnacco et al., 2015; Francois et al., 2013; Kraus and Chandrasekaran, 2010; Tallal & Gaab, 2006). The theories which underpin the rationale for music training are discussed in conjunction with the PRISM framework (Fiveash et al., 2021). These theories are mentioned in the next paragraph in preparation for discussion of the PRISM framework.

Dynamic attending theory (Jones, 1976) suggests that 'tone sequences presented at a regular rhythm entrain attentional oscillations and thereby facilitate the processing of sounds presented in phase with this rhythm' (p. 324). Tallal and Gaab (2006) posit that in dyslexia there may be problems with rapid auditory processing. In the temporal sampling framework, Goswami et al. (2002) recognised that an impairment in the rise-time perception of sound may indicate difficulties with the ability to segment syllables and phonological processing in dyslexia, but suggests that interventions which draw on the entrainment of syllables to rhythm and metre in music may be effective mediators. Patel (2012) suggested that music training might benefit language subject to the following conditions in the OPERA hypothesis: 'overlap, precision, emotion, repetition and attention' (p. 125). Another theory in support of music training is that music requires greater precision in auditory processing and the impact of music training would enhance processing for speech in a transfer of skills based on shared neural circuits, or the precise auditory timing hypothesis (Tierney & Kraus, 2014). The SEP (sound envelope processing and synchronisation to pulse) hypothesis (Fujii & Wan, 2014) provided further support that due to overlapping speech and language domains, an intervention using rhythmic activities would potentially benefit sound envelope processing and entrainment to a pulse leading to improved speech and language processing. These theories gave rise to an expanded understanding of the role of rhythmic activities as an intervention for speech and language difficulties and led to the proposal of the following framework.

Fiveash et al. (2021) developed a framework, based on the theories discussed in the preceding paragraph, of processing rhythm in speech and music (PRISM) which emphasises 'fine-grained auditory precision, neural oscillations and sensorimotor coupling' in a combination of these previous theories to highlight specific aspects of music rhythm training which would benefit speech processing (p. 773). These findings emphasise the importance of accurate speech rhythm processing to underpin phonological awareness and language

development in children with dyslexia. Zuk et al. (2018) conducted the first neuroimaging study which showed that music training may enable dyslexics to develop a neural network which compensates for the impaired regions for reading ability. Studies with dyslexic musicians have identified similar auditory sensitivity and rhythmic perception compared with non-dyslexic musicians (Bishop-Liebler et al., 2014) and improved amplitude processing as compared to the general dyslexic population without music training (Zuk et al., 2017). This supports the use of musical training for dyslexia as an effective way to improve difficulties with speech and language processing.

2.4.3 Music training for remediation of impairments associated with dyslexia

Music has the potential to be an enjoyable and social experience which is fulfilling and meaningful to those who participate in it. Early studies investigating the use of music as an intervention for language development highlighted the benefits of music as a multisensory and engaging activity and demonstrated benefits to phonological awareness (Overy, 2003, 2008; Register et al., 2007). Musical activities may have an impact on dyslexic students' confidence and self-esteem (McCarthy & Ditchfield, 2008; Oglethorpe, 2008; Overy, 2008). There appears to be valuable potential for improvement in auditory processing, sequencing and memory as well as motor coordination skills through activities such as singing, movement and playing musical instruments (Forgeard et al., 2008; Oglethorpe, 2008; Overy, 2008).

2.4.3.1 Studies of music training to support language development

Based on the importance of temporal processing to the development of phonological awareness, language and literacy skills, rhythm has been an important focus. Although several experimental studies supported the use of rhythmic activities to improve phonological awareness, Cogo-Moreira et al. (2012) found no evidence for music interventions as a remediation for dyslexia as none of the studies reviewed had used randomised control trials. However, Flaugnacco et al. (2015) conducted the first randomised control trial supporting the role of rhythm in improving temporal processing abilities which are important for the development of language and reading skills. In another review, Rolka and Silverman (2015) analysed 23 studies of music with dyslexic children and adults and suggested that amongst the intervention literature, there was a positive correlation between music training and literacy development. A more recent review of music rhythm

interventions with poor readers demonstrated further support for improved phonological awareness (Cancer & Antonietti, 2022).

Research involving the use of rhythmic training approaches in dyslexic children has demonstrated improvements in phonological awareness (Bhide et al., 2013; Bonacina et al., 2015; Caccia & Lorusso, 2021; Cancer & Antonietti, 2022; Flaugnacco et al., 2015; Frey et al., 2019; Goswami et al., 2013; Habib et al., 2016; Huss et al., 2011; Overy, 2003; Register et al., 2007; Thomson et al., 2013). The use of an applied video game ('Mila-Learn') to improve rhythmic abilities has demonstrated potential in dyslexic children (Vonthran et al., 2024), also indicating that music training might take place in a variety of contexts and formats both formal and informal.

Tallal (2012) demonstrated that improved attention, listening and reading skills correlated to music training directed at auditory processing skills. This is extended to levels of higherlevel language understanding (syntax and grammar) in studies using rhythmic musical primes (Chern et al., 2018; Przybylski et al., 2013). Loui et al. (2011) found evidence of a correlation between poor pitch processing and reduced phonemic awareness in dyslexic children; this appears to confirm studies which support a connection between dyslexia and amusia (Couvignou, 2019; Couvignou & Kolinsky, 2021). Studies comparing dyslexic and non-dyslexic musicians indicate that music training may improve auditory processing skills (Bishop-Liebler, 2014). Drawing away from the temporal processing theory, Rathke and Lin (2021) propose that difficulties with phonemic awareness, possibly related to poor auditory short-term memory, are the cause of the difficulties with rhythm. More research is needed to understand the distinct mechanisms that disrupt music, speech and language processing in dyslexia so that interventions through music training in particular may improve brain circuitry for both music and language skills (Gordon et al., 2015, Groß et al., 2022).

In a study of infants (birth to six months old) at familial risk of dyslexia, Paula et al. (2023) found that passive listening to vocal music was sufficient to benefit speech and language processing. Research using repeated reading with vocal music masking (RVM) remediation with dyslexic children has shown improvements in attention and phonological skills (LeLoup et al., 2021). This suggests that apart from temporal processing, there may be other shared

mechanisms that promote music and language development. These findings support the evidence for music training to be an effective remediation for impairments. There is scope for further research to investigate the use of different types of music interventions, to consider at which age interventions may most affect neuroplasticity and the consequent impact on language development.

2.4.3.2 Studies of music skills as early identifiers of language and literacy difficulties

Investigations of musical aptitude, using tonal and rhythmic measures, suggest that low musical aptitude might predict low phonological awareness and would support early intervention before the child begins to struggle with reading (Bégel et al., 2022; Culp, 2017; Lundetræ & Thomson, 2018; Strait et al., 2011; Turker & Reiterer, 2021). Studies involving event-related potentials to speech appear to indicate that even in infancy there are identifiable markers correlated to language difficulties (Cantiani et al., 2016; Kailaheimo-Lönnqvist et al., 2020). In a study including both preschoolers and older children, aged 8, both the ability to reproduce rhythms and perceive pitch was a predictor of phonological awareness (Steinbrink et al., 2019). Politimou et al. (2019) observed that being able to perceive melody showed only an effect for grammar development in language. However, in a recent longitudinal study with pre-readers, measures of music ability (melodic and rhythmic perception, memory) in children at risk of dyslexia did not appear to predict their problems with literacy (Couvignou et al., 2024).

2.4.4 Summary

Thus far, the motivations for this research which arose from experiences in my teaching practice have been discussed. To gain a greater understanding of dyslexia, definitions, characteristics and support factors have been explored. From a music teaching perspective, this has enhanced my comprehension in terms of the complexity of dyslexia and the variability with which it may affect individuals and has given me a novel view of the role of dyslexic strengths and support factors in my teaching. These findings suggest the importance of basic knowledge about dyslexia for music teachers. Understanding the overlap in co-occurring conditions enables an awareness of potential challenges for the student, not just in the area of text reading, but also in the areas of motor coordination, attention, organisation, sequencing, pitch discrimination and memory. It also suggests that

teaching that is tailored to the individual might account for and accommodate the diversity that is to be found amongst dyslexic individuals. In particular, this suggests that facilitating an awareness of strengths and other support factors may also promote students' selfesteem, and mental and emotional well-being which, according to the literature, may be an area of vulnerability.

In section 2.4, the relationships between music and language have been considered as a means of understanding how music training may benefit areas which show impairment for speech processing in dyslexic individuals. Music requires more precise processing than speech. In particular, processing the rhythm of speech may be disrupted in dyslexia and this affects the ability to segment syllables which may explain the phonological deficit which has been theorised to cause reading difficulties in dyslexics. The research discussed above in section 2.4 highlights some of the difficulties that dyslexic students may have in learning music: maintaining a pulse, accurate rhythmic and melodic perception and the ability to discriminate between pitches and intervals.

However, there is a growing body of evidence that music training does improve these aspects which are foundational for learning to play a musical instrument through a process of transfer. Music teachers may need to give more focus at the beginning to multisensory activities to improve beat and rhythm processing, as suggested by Overy (2008). This area of research also highlights the importance of rhythm training, not just to build foundational knowledge or correct errors, but to be an enjoyable and social activity which may have extended benefits for dyslexic students. My own experiences as a piano teacher and preliminary discussions with fellow teachers suggest that teachers may not feel that they have enough training or knowledge in understanding dyslexia and effective strategies for successfully teaching students with dyslexia. The lack of appropriate training and resources for teachers may prove an obstacle to dyslexic students developing an understanding of their learning potential and achieving their musical goals.

The next section (2.5) focuses on the music teaching and dyslexia literature beginning with an overview of recent literature, an analysis of music and dyslexia materials (2.6) and inclusive music teaching approaches (2.7). A discussion of the research questions concludes this chapter (2.8).

2.5 Music teaching and dyslexia literature

The following sources related to music teaching and dyslexia are presented in a general timeline. Analysis of some of the key texts follows (2.6) and these are discussed in relation to other dyslexia educational materials and research.

One of the earliest articles was written by Margaret Hubicki, a teacher at London's Royal Academy of Music, and Professor Tim Miles, one of the founders of the British Dyslexia Association. They were some of the earliest members of the British Dyslexia Association music committee. Hubicki and Miles (1991) reported on Hubicki's experiences with students who showed characteristics of dyslexia in the music lesson setting. These included difficulties with sight-reading music and spatial awareness. Strategies were based on effective literacy principles for remediating dyslexia but were incorporated into the music teaching setting. For difficulties with specific spatial concepts, for example, correlating horizontal left and right directions on the keyboard with low to high sounds, Hubicki and Miles (1991) note the importance of students having the opportunity to manipulate physical materials in order to promote comprehension of various concepts. Hubicki described ideas of how to incorporate multisensory, structured and individualised activities into music learning during lessons. Most notably, Hubicki developed a colour staff to help students distinguish the lines on the music score and to be able to recognise recurring patterns through the use of colour (Hubicki & Miles, 1991).

Ganschow et al. (1994) present profiles of six adult musicians (with difficulties associated with dyslexia but without formal assessments) who described their challenges with learning music and described compensatory strategies which they had developed. These were compared with the reported case of a professional pianist with dyslexia (Backhouse, 2001). All seven musicians reported challenges with sight-reading music, speed of processing, detecting and maintaining a pulse in addition to other rhythmic difficulties. Other areas presenting challenges were mixed in the profiles of the musicians: attention, fluency, reproducing music from memory and spatial awareness. Ganschow et al. (1994) refer specifically to musicians with dyslexia and their approaches to learning notation through the use of aural and kinaesthetic memory and in the case of one pianist, by learning a piece hands together rather than hands separately as they needed a more holistic understanding

of the piece before separating components. In a discussion of musical strengths and weaknesses, compensatory strategies included a holistic sense of the score and music, clapping rhythms, kinaesthetic memory, reliance on aural memory and multisensory approaches such as listening, feeling the melody line contours and visualising the music.

Sheila Oglethorpe's *Instrumental music for dyslexics: A teaching handbook* was first published in 1996, and depicts her experiences with dyslexic students in her piano teaching practice, challenges they presented with and strategies to support the students with, not only music learning but also to promote their self-concept and encourage their well-being. Multisensory instruction, in particular Dalcroze and Kodály methodologies, was suggested as well as an understanding of the student's learning profile. Noting that dyslexic students are normally anxious and vulnerable to fragile self-esteem, Oglethorpe points out the weighty responsibility of the teacher to encourage and support, stating that teacher attributes such as 'humility, adaptability, imagination, awareness, and genuine interest' are crucial (Oglethorpe, 2008, p. 28).

Oglethorpe (2008) notes that parents may choose teachers for a variety of reasons which do not always align with students' needs and describes the lack of training in educational psychology and dyslexia for instrumental music teachers as a 'dangerous scenario' (p. 25). Oglethorpe (2008) found parents of dyslexic students to be generally supportive in assisting their child with practising but also noted that some found it challenging not to pass on their own anxieties. Some parents were reluctant to reveal information about their child which could relate to assessment needs, although Oglethorpe (2008) pointed out that it might enable the teacher to better organise and prepare instruction for the student. Several sources refer to a parental denial of dyslexia that can be an obstacle to students receiving an assessment and assistance (Bajaj & Bhatia, 2020; Bonifacci et al., 2014; Oglethorpe, 2008). Anecdotal experiences describe parents learning of their own dyslexia by observing their child's challenges and subsequent diagnosis (McCarthy & Ditchfield, 2008; Oglethorpe, 2008; Nelson & Hourigan, 2016). Oglethorpe's (2008) book is analysed in more detail in section 2.6.

In a small-scale pilot study on learning music notation, Jaarsma et al. (1998) note that the speed and accuracy of dyslexic children was affected in reading music. Suggesting that

alternative notation systems using colour or codes would not be effective, they emphasise an approach to music reading that focuses on making music and relating that to notation gradually in an implicit learning approach. Rather than learning the names of notes and symbols first, playing the instrument and engaging with music would be the starting point. *Music and dyslexia: Opening new doors* (Miles & Westcombe, 2001) is a compilation of essays written by dyslexic musicians, music teachers and researchers; the book highlights the benefits of music learning and gives practical suggestions for teachers. This was the prequel to *Music and dyslexia: A positive approach* (Miles et al., 2008) which, as one of the key texts on this topic, is discussed in greater detail in section 2.6. Vance (2004), a classroom music teacher, discussed the challenges experienced by dyslexic learners: sight-reading, organisation, sequencing, concentration, behaviour, following directions, motor coordination and spatial awareness. Adaptation to the music classroom consisted of advising instrument choice carefully, simplifying tasks, repetition and the use of colour paper, colour coding of materials and recordings.

Macmillan (2004), a piano teacher with Suzuki training, highlighted the Suzuki approach as beneficial for dyslexic students due to the emphasis on structured instruction, an emphasis on listening and imitating, repetition and group lesson settings. Flexible and adaptable teaching was advised with a focus on promoting students' understanding of their own paths to learning. A discussion (Heikkila & Knight, 2012) of dyslexia in the music classroom highlights the valuable role of the music teachers and suggests adaptations to the learning environment, colour coding on music scores, the use of tangible materials and multisensory teaching such as embodied pulse and rhythm activities. To understand specific music notation reading difficulties, Flach et al. (2014) compared dyslexic with non-dyslexic children to consider how best to adapt music to reduce these challenges. They found that enlarging the music and drawing the stems of the music in the same direction was beneficial, but that the use of colour to represent stave lines was not supported.

In a comparative case study (Nelson & Hourigan, 2016) investigated the perspectives of five adult musicians with dyslexia. One surprising finding was that not all of the participants struggled with reading music. The study confirms the benefits of multisensory and structured instruction, small group lessons, breaking tasks and concepts down into smaller units and the use of technology for composing. Jazz and popular music, both genres which

lend themselves to improvising and simple chord structures, may be useful for those who find reading music a challenge. Four of the participants correlated positive factors in their careers as musicians with dyslexia. These included holistic ways of interpreting music, problem-solving abilities, strong aural memory and empathy. Empathy skills were described in relation to performance abilities and interpersonal relationships. Participants highlighted the importance of the teacher-student relationship. Nelson and Hourigan (2016) suggest that future research should investigate the preparedness and qualifications of music teachers for teaching students with dyslexia.

Morrow (2023), a cello teacher with a dyslexia instructor certification, suggests that most music teaching is not appropriate for students with reading disabilities and supports the need for more logical and structured learning to be applied. She identified music reading as the main challenge but mentioned confusion with the use of terminology or concepts such as duration and pitch. Morrow (2023) explains her process of moving from the simplest to the most complex concepts in a series of steps designed to build understanding, rather than promote rote memorisation. Using multisensory activities, visuals and repetition, Morrow (2023) encourages the student to build connections between concepts and introduces material in a cumulative way. Morrow (2023) concludes by asking 'Have music education specialists thoroughly researched this issue?' and 'What researched-based solutions have been provided to teachers so they know exactly how to help their struggling students?' (p. 26). Interestingly, both Nelson and Hourigan (2016) and Morrow (2023) suggest that there is a need for high-quality training for music teachers so that they may effectively teach dyslexic students.

In order to prevent repetition and for purposes of further analysis, strategies suggested by these sources are summarised in the following table (Table 2.1).

Strategies	Literature			
General Strategies				
Multisensory approaches	Hubicki & Miles (1991), Oglethorpe (2008), Miles			
	et al. (2008), Reid (2016), Morrow (2023)			
Overlearning or repetition	Vance (2004), Oglethorpe (2008), Miles et al.			
	(2008), BDA (2022), Morrow (2023)			
Systematic or structured	Vance (2004), Oglethorpe (2008), Miles et al.			
	(2008), BDA (2022), Morrow (2023)			

Personalised or individualised	Hubicki & Miles (1991), Vance (2004),			
instruction	Oglethorpe (2008), Miles et al. (2008)			
Visualisation	Ganschow et al. (1994), Oglethorpe (2008), Miles			
	et al. (2008)			
Scor	e modifications			
Enlargement of score	Vance (2004), Heikkila & Knight (2012), Flach et			
	al. (2016), Oglethorpe (2008)			
Simplifying or reducing parts	Oglethorpe (2008), Morrow (2023)			
Use of colour to highlight challenging	Hubicki & Miles (1991), Oglethorpe (2008)			
parts				
Drawings or visual aids to highlight	Ganschow et al. (1994), Vance (2004),			
specific music instructions	Oglethorpe (2008), Miles et al. (2008)			
Isolating elements and learning them	Vance (2004), Oglethorpe (2008), Nelson &			
separately	Hourigan (2016), Morrow (2023)			
Identifying common rhythm patterns and practising them	Oglethorpe (2008), Miles et al. (2008)			
Alternative notation systems (for	Hubicki & Miles (1991)			
example, Figurenotes ⁹)				
Modifying notes (changing stem direction)	Flach et al. (2016)			
Aural I	earning strategies			
Listening to recordings or making	Hubicki & Miles (1991), Ganschow et al. (1994),			
recordings	Vance (2004), Nelson & Hourigan (2016),			
	Oglethorpe (2008)			
Demonstration	Vance (2004), Oglethorpe (2008)			
Singing	Vance (2004), Oglethorpe (2008)			
Using aural skills to construct the	Oglethorpe (2008), Jaarsma et al. (1998), Nelson			
melody	& Hourigan (2016), Miles et al. (2008)			
Rhythm	learning strategies			
Demonstration	Ganschow et al. (1994), Oglethorpe (2008)			
Clapping or tapping the rhythm	Vance (2004), Oglethorpe (2008), Miles et al. (2008)			
Kinaesthetic learning – remembering	Oglethorpe (2008), Ganschow et al. (2004)			
the feel of the music in the body				
Following the music with a finger	Oglethorpe (2008), Hubicki & Miles (1991)			
Specific music methods				
Kodály method – sound before	Ganschow et al. (1994), Oglethorpe (2008), Miles			
symbol, embodiment of rhythm and	et al. (2008)			
pitches				
Suzuki method – aural learning,	Macmillan (2004), Vance (2004), Oglethorpe			
repetition, involvement of parents	(2008), Miles et al. (2008)			
Dalcroze method – embodiment of	Miles et al. (2008), Oglethorpe, 2008, Vance			
rhythm and pulse	(2004)			
The use of technology				

⁹ https://drakemusicscotland.org/figurenotes/

-

Arranging and composing music	Nelson & Hourigan (2016)	
Editing music scores	Oglethorpe (2008), Nelson & Hourigan (2016),	
	BDA (2022)	
Recordings	Oglethorpe (2008), Nelson & Hourigan (2016),	
	BDA (2022)	

Table 2.1 Summary of strategies

These studies identify the need for further research in the area of music notation adaptations. Although children participated in the studies on music reading and notation (Flach et al., 2014; Jaarsma et al., 1998), there is scope for further research on the student perspective of their music learning experiences. Future research might explore aspects of strengths and compensatory mechanisms which have been described (Ganschow et al., 1994; Oglethorpe, 2008; Nelson & Hourigan, 2016) and how these might be utilised in music lessons to support students' learning. Further studies are needed to examine the types and subtypes of dyslexia including their relationship to music learning (Nelson & Hourigan, 2016; Flach et al., 2014), although the diversity across the dyslexic population makes such categorisations difficult. There is scope for a deeper understanding of the relationships between parent-teacher-student and how this may impact music learning and other factors. The need for an understanding of each individual student's learning profile as a starting point in tailoring instruction for their lessons is emphasised. Examining teachers' perceptions of their experiences with dyslexic students may give insight to relevant themes for teacher training, the need of which was emphasised by at least two of the sources (Morrow, 2023; Nelson & Hourigan, 2016).

2.6 Analysis of music and dyslexia materials

2.6.1 Introduction

As mentioned previously in this thesis, the topic of music and dyslexia has been discussed in a fairly small number of books. The aim of this section is to critically appraise available material and to suggest the need for current and high-quality information for instrumental music teachers. They have been analysed for accessibility and content. The content is critiqued according to the evidence given for accessibility, research-based strategies, considerations of student strengths, discussion on memory issues, the inclusion of multiple perspectives, safeguarding/data protection concerns and the use of technology. Two of the three are in the category of a teacher's handbook: Oglethorpe (2008) Instrumental music for dyslexics: A teaching handbook and Aloba (2022) How to teach dyslexics music. Because Aloba (2022) has not had any editorial or peer-review process, unlike the other two books, it cannot be considered a rigorous source, though it has been included here due to its potential value as a guide written by a teacher with experience of working with dyslexic pupils. The third is a compilation of edited essays entitled *Music and dyslexia: A positive approach* (Miles et al., 2008). Sheila Oglethorpe (1933-2020) was a piano teacher who held the first conference on music and dyslexia at the Royal Academy of Music in 2010 when she was chair of the British Dyslexia Association music committee (Oglethorpe, 2020). Deborah Aloba is a singer and vocal teacher (Aloba, 2021) working in the UK. Professor Tim Miles, one of the editors of Miles et al. (2008) and a founding member of the British Dyslexia Association, began his work at a time when dyslexia was not recognised as a specific learning difference and was instrumental in changing perceptions of dyslexia (Fawcett, 2009). Editors John Westcombe and Diana Ditchfield are music educators with an interest in dyslexia.

Oglethorpe (2008) was first published in 1996 by Whurr Publishers whilst Aloba's (2022) handbook is self-published. Both handbooks (Aloba, 2020; Oglethorpe, 2008) are based on the authors' experiences teaching instrumental or vocal music lessons with dyslexic students. Oglethorpe (2008) has been reprinted in three editions and was advised by dyslexia experts Professor Margaret Snowling and Professor Margaret Hubicki (p. xi). Oglethorpe (2008) includes in-text citations in the main body of the book whilst Aloba (2022) chose not to include them in the text consistently. Both Oglethorpe (2008) and Miles et al. (2008) reference a number of sources.

Aloba (2022) does include a bibliography at the end of the book, but reference entries are inconsistent. Positive features of Aloba (2022) are the colourful diagrams and practical examples of the strategies suggested as well as practical application for teachers through questions at the end of each chapter. She also refers to percentages of her students without clarifying any characteristics of the sample population in terms of total numbers or specific learning differences. According to the introduction, the purpose in writing the book was so that other teachers would not have to go through as much difficulty compiling information,

stating that what the author did find was based on 'anecdotal evidence' and was not easily accessible in one place (Aloba, 2022, p. 1).

Music and dyslexia: A positive approach (Miles et al., 2008), was published by Whurr (now acquired by Wiley) and contains an edited compilation of essays from eighteen contributors (Oglethorpe contributes to more than one chapter) which are divided into the following four sections: 'Tackling problems', 'In and around the classroom', 'Strategies and successes' and 'Science takes us forward' (p. vi). The book is a sequel to a 2001 edition, with the addition of some new material. It is punctuated with personal stories. The book considers practical issues and solutions but also notes directions that future research may take. This book was written for instrumental and music classroom teachers, but might also be a useful resource for parents or dyslexic adult musicians. In-text citations and a reference list are included at the end of each essay.

2.6.2 Comparison of the layout and accessibility of the books

The following section contains a comparative table of the content of the books, as well as a discussion on the accessibility of the content. None of the books meet current published dyslexia accessibility standards (BDA, 2023). Whilst these books are written primarily for teachers working with dyslexic students rather than as self-help books for dyslexic students, the lack of attention to accessibility suggests wider issues. There appears to be a lack of consideration for diverse readers and users of the material. Both Oglethorpe (2008) and Miles et al. (2008) are printed using serif fonts and are text-based apart from minor diagrams. Serif fonts have been found to be less accessible to dyslexic readers than a non-serif font, whilst the use of visual diagrams and layouts in place of or in addition to text is considered to be useful for dyslexics (BDA, 2023). Whilst the layout is simpler and there is less text in Aloba (2022), there is a lack of consideration in the area of accessibility as it is written using low contrast ratio between fonts and background in highlighted sections and contains columns (BDA, 2023).

Though the narrower columns do not meet the accessibility guidelines of the British Dyslexia Association (BDA, 2023), Rello and Baeza-Yates (2017) suggest that column width did not change the readability of text for dyslexics in their study, and therefore they do not indicate recommended column widths. On the other hand, Miniukovich et al. (2017) suggest in their

guidelines for dyslexic web accessibility that wider-length columns should be avoided. Some research (Kous & Polančič, 2019; Rello & Baeza-Yates, 2017; Schneps et al., 2013) suggests that a singular universal dyslexia design is an ineffective concept due to the vast diversity of the dyslexic population; therefore, the recommendation is for books to be available on websites or e-readers which allow for a high level of customisation.

The following table (Table 2.2) provides comparison of chapter headings:

Oglethorpe (2008) 178 pages	Aloba (2022) 181 pages	Miles et al. (2008) 176 pages
Chapters:	Chapters:	Chapters:
1. Recognising dyslexia	1. Introduction	1. Dyslexia and other
2. Communication - the	2. Defining dyslexia	developmental
pupil and the teacher	3. Does the student	differences
3. Auditory	absorb information	2. Things that can go
considerations	visually, aurally or	wrong
4. Visual challenges	kinaesthetically?	3. In and around the
5. Motor problems	4. Assessing if a	classroom
6. Memorising and	student has a	4. Classroom rhythm
sight-reading	learning difficulty	games for literacy
7. Music theory -	5. Visual	support
coping with writing	processing/reading	5. Early years: Deidre
music	difficulties	starts to learn piano
8. The lesson and the	6. Visual stress	6. Winning over the
wider world	7. Dysgraphia	reluctants
Appendix: Repertoire for	8. Kinaesthesia and	7. Can music lessons
pianists	dyslexia	help the dyslexic
Glossary	9. Lack of focus	learner?
Useful addresses	10. Multi-sensory	8. Parallels between
Bibliography	approaches	the teaching of
Suggested further reading	11. Coping with visual	musical and
	processing	mathematical
	difficulties and	notation
	phonological	9. The paperwork
	methods with deep	10. Sight-reading
	dyslexia	11. Sight-reading and
	12. Assistive technology	memory
	tor exam	12. Ten top tips and
	preparations	thoughts
	13. Causes of dyslexia	13. Can computers help?
	14. Auditory processing	Matching the inner
	disorder	with the outer ear
	15. Conclusion	

Bibliography	14. Positive connections across the generations
	15. Similarities and differences in the dyslexic voice
	16. Thirty-seven oboists
	17. Suzuki benefits for children with dyslexia
	18. Dyslexia: No problem
	19. Insights from brain imaging
	20. Music reading: A cognitive neuroscience
	approacn There is no bibliography at the end, but each chapter contains a reference list.

Table 2.2 Comparison of chapter headings in key texts

Miles et al. (2008) include a chapter on neuroscientific research related to dyslexia (Chapter 20 in Miles et al., 2008). Oglethorpe's (2008) book has a glossary of important terms, and signposts the reader to dyslexia resources and further reading resources; these are helpful resources for anyone not familiar with the topic of dyslexia. Whilst Oglethorpe (2008) and Aloba (2022) approach the topic from the viewpoint of teachers, Miles et al. (2008) include a multiplicity of perspectives from dyslexic musicians, researchers in the field of dyslexia and instrumental and school teachers.

2.6.3 Identifying features of dyslexia

Each book refers to the definition and identifying characteristics of dyslexia. Oglethorpe (2008) and Aloba (2022) are written from their perspective of experiences with dyslexic students in instrumental and vocal lessons. Oglethorpe (2008) does refer to spatial difficulties more than Aloba (2020); this is most likely due to the difference of challenges experienced between singing lessons and using a physical instrument. The key difference in these books (Aloba, 2022; Oglethorpe, 2008) is that only the teacher's narrative is included; whereas in Miles et al. (2008), first-person perspectives from teachers, students, musicians and researchers are incorporated.

Oglethorpe (2008) notes that most of the issues stem from poor sound and sight connections (p. 3) and that the student shows 'difficulty that is not typical of their general level of performance' (p. 4). The author refers to research being done in order to understand the different brain characteristics of dyslexics and characterises the challenging areas as auditory, visual, spatial and memory along with various secondary characteristics, noting that auditory challenges refer to slow aural processing, difficulty recognising rhyme, and phonological difficulties (Oglethorpe, 2008, p. 5). She recognised students' difficulty in maintaining a fixed focus, struggling with glare or tracking the music and differentiating between lines and space notes (p. 11), and Oglethorpe (2008) was able to identify patterns related to dyslexia. Spatial concerns which were observed by Oglethorpe include challenges with left and right, near and far, up/down and high/low. Struggles relating to memory consist of poor short-term, auditory, visual and kinaesthetic memories, but a permanence of material once the material is committed to long-term memory (p. 6). Secondary characteristics include disorganisation, poor concentration, anxiety, low self-esteem, frustration and inconsistent behaviour (p. 7).

In his contribution to the co-edited book, Miles (In Chapter 1 of Miles et al., 2008, p. 4) identifies some of the key characteristics of dyslexia as being a hereditary disorder, more common in males and originating from distinct brain characteristics. The author notes that this may cause difficulty with left and right, following instructions, learning times tables, spelling, short-term memory and symbols (p. 5), referring to related conditions of dyspraxia, ADHD, ADD, dysgraphia, dyscalculia, and autism spectrum under the main category of neurodiversity. Miles (2008) describes how awkward social situations, poor time management and school stress may also result from the related challenges of being dyslexic (Chapter 2 of Miles et al., 2008, pp. 12-15).

Aloba (2022) refers to the Rose Report (2009) definition of dyslexia with the addition of information about strengths from the British Dyslexia Association. If previously diagnosed and assessed, Aloba (2022) suggests using the formal assessment as a starting point. However, only Aloba (2022) notes that the cost of private assessment can be a potential barrier for families whilst also suggesting that additional support in the UK at the higher educational level is better than that of the primary and secondary school level. Indicators of dyslexia, noted by Aloba (2022), include avoidance of written music and desire to learn

aurally, challenges recalling melody or lyrics, lack of understanding of or misinterpreting instructions, lack of focus and attention, hesitation when there is a rest in the music, avoiding writing music, showing signs of anxiety or frustration, tense or frozen body language, singing incorrect but similar lyrics, preferring to chat through lessons, difficulty clapping a rhythm or 'trying too hard' (Aloba, 2022, p. 9).

Whilst Aloba (2022) and Oglethorpe (2020) refer to how dyslexic traits might be identified in the undiagnosed music student, Miles (Chapter 1 of Miles et al., 2008, p. 8) promotes the use of a diagnosis by a qualified assessor. Additional school support in the form of reasonable adjustments and accommodations is often the intention of an official diagnosis; however, Aloba (2022) notes the cost may be prohibitive for some families. Aloba (2022) and Oglethorpe (2008) highlight various challenges that they have identified with dyslexic students in instrumental or vocal lessons. It might have been useful to have included the student perspective as they may be able to narrate their own unique challenges and report strategies that have helped them whilst also avoiding misinterpretations or bias that may happen by solely including the teacher's narrative. An analysis of these viewpoints might enable teachers to have a better understanding of the needs of their dyslexic students.

2.6.4 The label of dyslexia

There is considerable debate about the advantages or disadvantages of the diagnostic label 'dyslexic' (Elliott & Grigorenko, 2014; Elliott & Nicholson, 2016). There is also discussion about how dyslexia should be classified, with Miles (2008) preferring to call it a 'difference' instead of a 'disability' (p. 3). Whilst Oglethorpe (2008, p. 14) suggests that the label of dyslexia is not necessary to be able to effectively teach a dyslexic student, Miles (Chapter 1 of Miles et al., 2008) disagrees, stating that 'The correct diagnostic label is essential if the student is to be adequately helped' (p. 8). All three books agree about the need for personalisation and individualisation in teaching dyslexic students.

Students, or their parents, might see the label of dyslexia as a negative thing, therefore Oglethorpe (2008) and Aloba (2022) stress the importance of adapting teaching for students, with or without a diagnosis. Oglethorpe (2008) refers to labels as potentially 'dangerous' considering the psychological damage they may inflict on the student, but also as potentially useful for the information they might provide to the teacher (p. 16). Aloba

(2022) also emphasises the ethical importance of confidentiality, particularly in the safekeeping of assessment records and notes, as a legal requirement for teachers under general data protection guidelines (ICO, 2021). There is scope for revision in future editions (Miles et al., 2008; Oglethorpe, 2008) to include information on current teaching guidelines for aspects such as data protection and safeguarding.

2.6.5 Learning modalities

Aloba's (2022) material on learning modalities and strategies lacks a robust research base. She states that 'In the course of absorbing the information required to learn a new skill, it has been established that everyone has a preference in how they learn, be it visually, aurally or kinaesthetically' (p. 6). To support this statement, Aloba (2022) refers to an outdated study on learning styles (Rose & Nicholl, 1997). This study is not included in the bibliography of the book but appears to be a book published by Dell in 1997 by Rose and Nicholl entitled Accelerated learning for the 21st Century: The six-step plan to unlock your master-mind. Aloba (2022) then refers to two more recent research studies (Andreou & Vlachos, 2013; Stampoltzis et al., 2010) which relate directly to dyslexic learners. Stampoltzis et al. (2010) compared 20 dyslexic and 40 non-dyslexic university students' preferred learning styles and concluded that dyslexic students preferred the kinaesthetic learning style whilst nondyslexic students preferred visual learning styles. Andreou and Vlachos (2013) compared 43 dyslexic and 86 non-dyslexic adolescents and confirmed that dyslexic students prefer a variety of learning styles, which may justify a multisensory approach to teaching. Whilst Aloba (2020) appears to believe that students will demonstrate a 'preferred learning style' and seeks to justify this position through the literature in the hope that it will inform her teaching, such a view is not universally supported (Kirschner, 2017; Westby, 2019). Nancekivell et al. (2020) suggest that a potential consequence of those who hold to this belief is that they risk a lack of understanding that students may prefer different learning styles at various stages and in different environments.

2.6.6 Strategies

Based on the definition and characteristics of dyslexia reviewed in this literature review, there are numerous areas likely to be affected in music learning for dyslexic students, as visual, auditory, coordination, processing, sequencing, automaticity skills, spatial awareness,

working memory and organisation are implicated in learning to read and perform music (Miles et al., 2008: Oglethorpe, 2008; Nelson & Hourigan, 2016). This is likely to affect the ability to sight-read at speed. From observations of dyslexic students, Oglethorpe (2008) noted difficulty with left-right differentiation, inability to discern if notes on the stave were on lines or in spaces, lack of recall of lesson topics, disorganisation, poor coordination and low self-esteem. Another challenge is the fear of stigma or lack of confidence dyslexic musicians might face from fear of disclosure relating to components of music they find challenging (Ganschow et al., 1994; Oglethorpe, 2008; Miles et al., 2008; Vance, 2004).

The Orton-Gillingham approach to reading instruction provides the foundation for many programmes and teaching methods for students with reading difficulties and has been used for over 80 years; strategies include the use of multisensory activities, repetition, personalised instruction that is tailored to the student and their needs, with learning developed logically in a systematic and structured manner which enables students to progress (Reid, 2016; Phillips & Kelly, 2016). According to the Orton-Gillingham Institute for Multi-Sensory Education, a multisensory approach uses 'sight, hearing, touch and movement to help students connect and learn the concepts' (Institute for Multi-Sensory Education, 2020). The use of visual material may be useful for some, as well as opportunities to review and build on past knowledge with the teacher's assistance through a process known as scaffolding (Reid, 2016). Other strategies advocated for dyslexic students use of technology for reminders and recording lectures, being conscious of removing distractions, 'embedding' in memory things that need to be recalled (using mnemonics, acrostics or other memory assists), mind mapping and allowing space for breaks and relaxation (Saunders & Eastap, 2018, pp. 140-142). Reinforcement of knowledge might take place through verbalisation techniques, connecting to larger frameworks and simplified concepts (Reid, 2016). The following sections discuss strategies related to multisensory and structured teaching (2.6.6.1), visual (2.6.6.2), auditory (2.6.6.3), memory (2.6.6.4), motor coordination (2.6.6.5), using assistive technology (2.6.6.6), strengths (2.6.6.7) and summarise the key points (2.6.6.8).

2.6.6.1 Multisensory and structured teaching approaches

Some general teaching strategies are advocated by dyslexic professionals as an intervention for literacy: multisensory, structured and the use of repetition (Reid, 2016; IDA, 2018;

Phillips & Kelly, 2016). However, these same principles might be applied to music. Multisensory teaching is advocated by all three books with Oglethorpe (2008) stating that this is the 'foremost piece of advice' for teachers (p. 12). Oglethorpe (2008, p. 55) suggests the use of a floor stave to help with interval recognition and fingering in a kinaesthetically appealing approach. Aloba (2022, p. 106) suggests using stairs or steps to teach intervals, breathing exercises and clapping or jumping games to learn rhythms. Aloba (2022, p. 108) also suggests breaking Italian terms into syllables and then inventing a corresponding movement to go along with the term as a multisensory activity. By asking a student to put together a jigsaw puzzle made of various 'pieces of a score of music' (i.e. elements such as the key and time signatures, double bar), Aloba (2022, p. 109) encourages the student to consider the different components and order of a piece of music.

Structured instruction (Miles et al., 2008, p. 56; Oglethorpe, 2008, p. 157) is also emphasised. Oglethorpe (2008) describes the need for each lesson to be structured in advance, but with 'flexibility' and adjustments when needed (p. 157). Aloba (2022, p. 111) mentions detailed and structured ways of teaching students with severe, or what she refers to as 'deep dyslexia' which involves using multi-modal teaching methods as well as simplification of the music to avoid overwhelming the student with information. Miles (Chapter 8 of Miles et al., 2008, p. 72-73) suggests that musical notation should be taught 'slowly and systematically', particularly in the case of musical symbols. Oglethorpe (2008) describes a systematic approach to piano pedalling designed to offset challenges by coordinating hands and feet (p. 95). Aloba (2022) details how using a 'music tree' is a structured way to explain the relationship of note values and refers to them using the American terms 'whole, half, quarter note' rather than 'semibreve, minim, crotchet note'. A visual aid of a tree-like structure is constructed through her layout of note values: starting with semibreve notes, she then adds the equivalent minim, crotchet, quaver and semiquaver notes underneath (Aloba, 2022, p. 102).

Repetition is a key teaching strategy for dyslexia and is advocated by each of the books (Aloba, 2020, p. 78; Oglethorpe, 2008, p. 43; McRitchie-Pratt, 2008, p. 43 (Chapter 3 of Miles et al., 2008). Oglethorpe (2008, p. 43) notes that the keyboard provides multiple opportunities to 'reinforce motor function, memory training, rhythmic and melodic discrimination'. The purpose of repetition as a teaching strategy for dyslexia is to improve

automaticity and to reinforce what has already been learned (Reid, 2016). The following sections compare the use of strategies in the three books regarding challenges that dyslexic music students might face: visual, auditory, memory, motor skills, composition or sight-reading.

2.6.6.2 Visual

Oglethorpe (2008, p. 5) and Aloba (2022, p. 4) refer to visual disturbances or stress. Aloba (2022) states that:

It is accepted that many people who suffer with dyslexia encounter visual problems: headaches and eyestrain when reading, text blurring, text going in and out of focus, text appearing double, text alternating between single and double, difficulty keeping place in text, difficulty tracking across lines of text, difficulty because of brightness of the page, difficulty because of contrast between text and background, text appearing to shimmer or flicker (pp. 4-5).

Oglethorpe (2008) is careful to point out that there are often unhelpful assumptions made about visual issues and dyslexia; she refers to genuine issues she has observed: poor binocular control (p. 54), glare from black print on white paper (p. 56) 'unreliable eye-brain relationship' (p. 57), challenges with focusing and 'tracking from left to right' (p. 66), and 'the score stimulating the brain to perceive movement' (p. 71). Oglethorpe (2008) does not refer to any research but suggests using coloured lenses, overlays or coloured paper as a solution (Oglethorpe, 2008, p.56). Aloba (2022) points to the benefits of covered overlays with students in her teaching practice (p. 23).

Advice given by SASC (the Specific Learning Difficulties (SpLD) Assessment Standardisation Committee in the UK) in 2018 points out that 'visual stress should not be used as a catch-all term for visual difficulties, there is no strong evidence between visual stress and dyslexia/SpLD and that assessors should not conduct coloured overlay tests or suggest or diagnose visual stress' (SASC, 2018, p. 3). In a systematic review, Suttle et al. (2018) concluded that there was not enough evidence to suggest that coloured overlays or lenses are a solution for visual discomfort. However, other research suggests that coloured filters may be useful to reduce visual stress in poor readers (Stein, 2022). SASC (2019) suggest that

anyone describing symptoms of visual stress or distortion should be referred to their optometrist and GP for assessment.

Enlarging the score

Enlarging the score is suggested in all three books: Oglethorpe (2008, p. 49), Aloba (2022, p. 14), and the following chapters from Miles et al. (2008): Chapter Ten 'Sight Reading' (Oglethorpe, 2008, p. 88) and in Chapter Sixteen 'Thirty-seven oboists' (King, 2008, p. 134); this correlates with the evidence that enlarging the score may reduce errors based on a study of the readability of music in dyslexic children (Flach et al., 2016).

Note stems in the same direction

Both Aloba (2022, p. 51) and Oglethorpe (2008, p.48) suggest modifying scores so that stems on note heads are in the same direction, or erasing the stems as a means of simplifying the score for a student with dyslexia. This was supported as an error reduction strategy in a study of dyslexic children (Flach et al., 2016).

Use of coloured notes or stave lines

Research (Flach et al., 2016) indicates that the use of multi-coloured lines did not reduce errors in dyslexic students and in some cases produced confusion. However, using multicoloured notes, lines and other aspects of the score is a strategy advocated by each of the books. Miles (In Chapter 8 of Miles et al., 2008, p. 73) refers to the coloured staff developed by Hubicki (2001) as an effective tool. Oglethorpe (2008) suggests that the use of colour might be a way of highlighting various aspects of a score for a dyslexic student who may find the contrast of black and white visually unhelpful (p. 51) whilst Aloba (2022, p. 25) refers to anecdotal information about the use of colour and states that 50% of her students found using colour to be a supportive strategy. However, she does not refer to the specific type of use of colour or the number of students in her sample. Recognising the individuality of each student and their preferences seems to point again to the importance of customisation and personalisation of teaching materials.

Alternative notation systems have been devised. For example, Figurenotes¹⁰ (Drake Music Scotland, 2018) is a system of notation which uses colours to depict pitch, shapes to denote register and length to indicate rhythm. Kivijärvi (2019) found that the system was useful for reducing the amount of processing needed and could be an accessible tool for inclusive music teaching. More research is needed to clarify these strategies and to study the effectiveness of their incorporation to music learning strategies for dyslexic students. Additional testing and knowledge on implementation and learner responses would be useful; this could include further research on the effectiveness of colours or shapes for note heads, coloured staves, coloured paper, which colours are preferred, how to simplify or adapt scores using technology and specific strategies for different age groups.

Due to similarities in terms of challenges with reading a score of music, it is worth considering strategies used with visually impaired students. The Amber Sound Trust website (Amber Sound Trust, 2024) offers a number of strategies for instrumental teachers working with visually impaired students including developing aural skills, using demonstration and physical guidance, starting with simple pieces and building awareness of tonality gradually, carefully choosing repertoire and using improvisation. As Braille music differs from notated music, serving as a memory aid or scheme to guide the learning process with students unable to read Braille with their fingers and play an instrument at the same time, a reliance on aural memory skills is quite important. Teachers and students might find it difficult to sustain motivation if many repetitions are needed to learn music. These insights offer useful considerations for dyslexic students by focusing on music learning which is less reliant on a score of music and more focused on aural skills and memorisation.

2.6.6.3 Auditory

Overy (Chapter 4 of Miles et al., 2008, p. 26) refers to temporal processing and timing deficits which affect the auditory and motor domains of the brain in dyslexia. Through the use of rhythm and music games, Overy (2008) developed a programme based on music education, music therapy and literacy resources to improve dyslexic students' timing skills whilst also taking into account challenges of 'concentration, sequencing, motor coordination and memorisation' (p. 28). Some of the activities were based on the Kodály method of

¹⁰ https://figurenotes.org/

music education with an emphasis on rhythm and movement. The games were designed to be repetitive and to structure learning in a cumulative way, but also to be facilitated in a sensitive and engaging way. Students were encouraged to join at their own pace. The result was that students made improvements in the areas of rhythm, phonological awareness and spelling skills (Overy, 2003). Learning a musical instrument improved a dyslexic student's 'auditory perception' according to Oglethorpe (2008, p. 58), who stated that whilst some dyslexic students do have poor auditory memory for speech, this does not affect their music abilities (p. 88). McCarthy and Ditchfield (Chapter 5 of Miles et al., 2008, pp. 49-50) state that auditory difficulties, among other potential areas of challenge, may be referred to as potential strengths and weaknesses.

2.6.6.4 Memory

Whilst Aloba (2022) briefly refers to issues with 'remembering rhythms and note values' and describes the importance of matching physical movements as a way of assisting the student to develop recall (p. 81), Oglethorpe (2008, p. 122) has a much more research-based analysis of short and long-term memory systems and how a poor working memory might impact on sight-reading and memorising music for dyslexics. The advantages and disadvantages to kinaesthetic memory approaches to learning are discussed including the use of memory aids in the form of mental or physical pictures and consideration of the 'architecture of the music' as a building block for memorisation (Oglethorpe, 2008, pp. 121-123).

In Miles et al. (2008) challenges with memory are frequently referred to in personal accounts (Lea, 2008; Wood, 2008; Ditchfield, 2008; Oglethorpe, 2008, p. 60). McRitchie-Pratt et al. (Chapter 6 of Miles et al., 2008, p. 55) observe that some dyslexics display a preference for learning or memorising a piece of music aurally. Lea (Chapter 11 of Miles et al., 2008, p. 93) considers how playing different instruments might activate the memory systems, illustrated by his difficulty with memorising music for the cello, compared with his ease at memorising pieces with guitar. This experience led him to question whether there was a difference between the direct impact of his fingers on the guitar strings compared to the use of the bow on the cello strings; the need for further research is noted.

Because of the impact of working memory on learning, research on dyslexia has focused on confirming a link between dyslexia and poor working memory (Alloway, 2009). More recent research on working memory profiles of children with dyslexia (Gray et al., 2019) indicates that poor working memory and dyslexia are not always correlated; they stress the importance of working memory assessments to build an individual profile of the student. They also note that with co-occurring conditions, a wider profile of memory deficits might be apparent. This means that teachers may observe that a dyslexic student who also has attention or coordination issues might have more severe or broader difficulties with memory than those who do not.

2.6.6.5 Motor coordination

Apart from a brief section on the challenges of writing music for a student with dysgraphia, Aloba (2022, p. 74) does not make much reference to dyspraxia/developmental coordination disorder as a co-occurring condition with dyslexia. This may be because she is a vocal teacher and does not encounter some of the issues which instrumental teachers may face. She suggests multi-sensory strategies such as clapping, tracing notes, making models of them from Plasticine and using notes made from felt to help overcome some of the challenges of dysgraphia (pp. 73-79).

In contrast, Oglethorpe (2008) dedicates an entire chapter to the issue of motor coordination and notes the overlap between dyslexia and dyspraxia identifying challenges with balance, posture, coordination of arms, hands, fingers and feet, left/right or high/low orientation, and fingering (p. 72). She suggests the importance of stressing good posture and making allowances for a need to move or change position to maintain concentration (p. 74). Further suggestions include rhythmical, cross body warm-ups, practising away from the piano or with eyes closed, the visualisation of motor tasks, tapping short rhythms from flashcards, and making allowance for breaks as eye/muscle coordination can be very tiring (pp. 75-78). Oglethorpe (2008) suggests that specific issues with piano fingering might be improved by writing the numbers on the fingers (p. 86), 'crab walking' using fingers 1 and 2 to practise using the length of the keyboard and turning the thumb under (p. 87). Scales might be practised not by synchronising hands together at first but by playing one hand at a time right after the other (Oglethorpe, 2008, p. 87).

Miles et al. (2008) refer to dyspraxia but state that it 'infrequently' co-occurs with dyslexia (p. 6). It is possible that whilst Oglethorpe (2008) observed motor coordination issues in students, the research connecting DCD/dyspraxia (developmental coordination disorder) as a co-occurring condition with dyslexia was not yet evident. Even now, the research is not conclusive. Stoodley and Stein (2013) seem to suggest that there are many inconsistencies across dyslexic profiles and that not all dyslexics struggle with motor coordination issues. However, Marchand-Krynski et al. (2017) suggest that motor coordination impairments do correlate with dyslexia and attention deficit disorders and suggest that targeted training may be effective in improving coordination in both populations. Dan (2019) suggests that 'current data do not allow us to say whether a child with mild, moderate, or severe DCD will present with a particular cluster of motor and cognitive issues' (p. 1126). Overy (2008, Chapter 4 of Miles et al., p. 26) connects auditory and motor coordination timing skills and targets them in the development of musical games with the aim of improving reading and coordination. Teaching approaches such as Suzuki and Dalcroze are recommended for their emphasis on activities which use the whole body as a means of improving motor coordination, as suggested by Macmillan (Chapter 17 of Miles et al., 2008). Stewart (Chapter 20 of Miles et al., 2008) notes the connections between rhythmic listening tasks and the activation of the motor (areas related to balance and timing) and auditory regions of the brain in brain imaging research (p. 154).

2.6.6.6 Using assistive technology

Oglethorpe (2008, p. 123) speculates that computers might 'one day' assist dyslexics with writing music, whilst Aloba (2022) offers a comprehensive review of how MuseScore (MuseScore, 2021) might be used to customise scores. This includes how to use features of MuseScore such as creating a score, magnification and how to create specific notes, clef, time signature and background colours. Aloba (2022) includes several screenshots to illustrate how to use MuseScore to customise scores with students and to create worksheets for them to practise music theory skills in preparation for exams (pp. 121-156); this appears to illustrate for teachers what is possible with this specific technology. The notational aspects of MuseScore are free and the software is continually refined through the 'open-source developer community' (MuseScore, 2021).

Ditchfield (Chapter 9 of Miles et al., 2008, p. 79) notes that assistive technology may be pivotal for dyslexic students whilst recognising that the cost may be a barrier for some. Apostoli (Chapter 13 of Miles et al., 2008) describes their experiences as a dyslexic musician with using technology but also refers to specific hindrances such as complex software interface systems or complex text-based manuals used at that time (p. 105). The pace of technology is rapidly moving forward to make interface with programs simpler. The increase in the use of smartphones may enable recordings for practice or demonstration, but ethical and data protection issues should be carefully considered. Ward et al. (2017) highlight how distracting smartphones are in a lesson setting and suggest that although the rise in their use is inevitable, caution should be used in their application to education.

Software which focuses on isolating specific tasks or information such as identifying notes in different clefs, identifying intervals and rhythmic practice may prove useful for dyslexic students (McCord, 2004). An application such as Practice First^{™11} (Music First, 2024), which analyses and offers feedback on rhythm and pitch aspects, might be beneficial for assisting students when they practise at home. Della Ventura (2019) notes that the challenge teachers face in using artificial technology is mainly that of leveraging the 'right balance between providing direct instruction (or texts, notes, examples, exercises, ...) to improve areas of weakness and the use of AIT to compensate for the identified learning difficulties' (p. 15). However, Della Ventura (2019) highlighted the need for teachers to continue to monitor dyslexic students throughout the learning process, including their at-home use of technology as well as during lessons.

Technology is a rapidly changing field which has the potential to be an effective tool for teachers 'to challenge the world, challenge certainties, increase awareness, shape attitudes, and foster communication' as long as the focus is on music (Himonides, 2012, p. 455). Although there is only a small amount of research available on the use of technology with dyslexic students, the literature suggests that it may be useful for listening to or making recordings of music (Ditchfield, 2008; Ganschow et al., 1994; Oglethorpe, 2008), for composing (Apostoli, 2001; Nelson & Hourigan, 2016), and through using assistive technology which translates text to audio formats (Ditchfield, 2008). Notation software may

¹¹ https://www.musicfirst.com/software/practicefirst

provide the advantage to users of composing a notated score with MIDI playback, facilitating auditory feedback in the development of a composition (Nelson & Hourigan, 2016). In conclusion, technology may provide some useful ways of assisting dyslexic students in learning music, but it should not be regarded as a replacement for quality, structured teaching.

2.6.6.7 Strengths

Of the three books, only Aloba (2020) does *not* refer to strengths, instead framing dyslexia as a weakness and disadvantage. Oglethorpe (2008) states that the best advice for teachers is to 'teach to the pupil's strengths' (p. 13) and emphasises how important it is to know the student's strengths and weaknesses and to incorporate lesson planning around them. Considering dyslexia may be an advantage, the author (Oglethorpe, 2008) suggests that there are compensatory strengths in dyslexic music learners: resilience and creative problem solving (p. 8) tactile (p. 50), spatial (p. 83), kinaesthetic memory (p. 100), aural (p. 113) and musical composition (p. 123).

In Miles et al. (2008) strengths are referred to in a variety of contexts, including personal stories, teacher analysis and future research suggestions. In terms of specific strengths, McCarthy and Ditchfield (Chapter 5 of Miles et al., 2008, p. 49) discuss a young dyslexic piano student whose strengths were 'auditory and motor, family support and intelligence' (p. 49). Miles (Chapter 8 of Miles et al., 2008, p. 71) notes reasoning strengths and creativity, as well as pattern recognition. Ditchfield (Chapter 9 of Miles et al., 2008, p. 144) describes a musician using their strengths to compensate for areas they find challenging. Bishop-Liebler (Chapter 15 of Miles et al., 2008, p. 123) refers to the necessity of working with dyslexic students' strengths by 'mapping the areas that the student finds easy and difficult'. Overy (Chapter 19 of Miles et al., 2008, p. 157) contributes to the discussion by referring to research which indicates that dyslexics may have unique visuospatial abilities and discusses the need for dyslexic strengths to be investigated further through neurological research. Overy (Chapter 19 of Miles et al., 2008) indicates that with music performance using so many different areas of the brain, this may 'explain why dyslexic musicians can exhibit extraordinary strengths in some areas of musicality while experiencing difficulties in other areas' (Overy, 2008b, p. 158).

As general dyslexia research has increasingly focused on a strengths-based perspective (Davis, 2010; Eide & Eide, 2011; Everatt et al., 2008; Gilger, 2017; Gobbo, 2020; Malpas, 2017; Rooke, 2015; Shaywitz, 2003; West, 1990/2009), music education might also benefit from further analysis and research in this area. Rolka and Silverman (2015) note the importance of focusing on strengths for the following reasons: 'to promote self-confidence, academic self-efficacy, the joy of learning and to teach strategies to improve frustration tolerance' (p. 24). Useful research directions for the future may be examining how dyslexic strengths are perceived by students and instrumental/vocal teachers, how instrumental teachers identify strengths in dyslexic students and how these may be developed and used in music education.

2.6.6.8 Summary of key points for educational materials

In conclusion, there are some key points to take forward from these materials that would inform future educational outputs. These include the significance of having materials, digital or in print, which meet current dyslexia-friendly guidelines. Material should be referenced according to an accepted standard formal referencing technique and it would be advantageous to include a glossary of the terminology used, resources and other material to signpost readers to quality sources of information about dyslexia.

The strategies that are advocated must be based on current research, and assumptions regarding dyslexia or dyslexics as a whole group should be avoided. Evidence-based teaching strategies for dyslexics which have been established in literacy teaching should be utilised in future materials; these include multisensory, structured, systematic and overlearning or repetition. Incorporating material regarding data protection and safeguarding of students is particularly important from an ethical, and legal, point of view and is foundational in building a relationship of mutual trust and respect with a student and their family.

In the case of visual stress, it is important to emphasise to teachers, parents and students that a student who complains of visual distress should be referred to an optometrist as many symptoms of visual distress overlap with refractive and oculomotor issues which might be treated with special glasses or exercises. There should be careful consideration before an instrumental music teacher recommends a SpLD assessment for an undiagnosed

student. This is beneficial only if they believe that it might serve the student to accomplish their goals; for instance, in the case of evidence needed for music exams. An underdeveloped area in these books is that of dyslexic students and their strengths in the music learning process. I have learned about and observed strengths in the areas of performing, learning music aurally, recognising patterns, using problem-solving abilities to learn musical concepts, improvising and composing, both in research (Oglethorpe, 2008; Miles et al., 2008; Rolka & Silverman, 2015; Reifinger, 2019) and in my teaching. However, the views of the learners themselves warrant further attention.

The next section (2.7.1) considers some general inclusive music teaching approaches and discusses the issue of access to provision in music education in both the classroom and individual settings (2.7.2).

2.7 Inclusive music teaching and access to provision

2.7.1 Inclusive music teaching approaches and considerations

Music education pedagogy in the UK has developed with the philosophy that 'music should be available to all pupils' (Paynter, 1982, p. xiii). However, provision in mainstream schools has not been consistent (Bate, 2020; Savage, 2021; Zeserson et al., 2014). Research has also highlighted a lack of music opportunities for students in special schools, particularly through a report which provided the impetus for the development of the Sounds of Intent music framework (Welch et al., 2009). This framework was relevant for establishing an understanding of music development, engagement, curriculum and assessment from the most basic starting points to musical maturity intending to enrich musical education, particularly for those with special educational needs (Ockelford & Voyajolu, 2020).

It is crucial for instrumental music teachers to be aware of general strategies recommended for use with dyslexic students. However, it is likewise important to note that the use of these strategies and skills will need to be adapted to the individuals and the teaching context. Variants of these skills and strategies might become important depending on the individual learner and flexibility is needed. As the teacher observes the student and their individual strengths and weaknesses, they can begin to put into practice the strategies that may promote optimal learning. This means that lesson planning and curriculum can be tailored to unique student profiles and is well-suited to inform the teacher's approach for dyslexic students who may vary greatly in their profile of strengths and weaknesses. Bishop-Liebler (2008) refers to this process as 'building up a detailed picture of the student's learning history, mapping the areas that the student currently finds easy and difficult, and identifying their preferred learning styles' (p. 123). Advice from dyslexia general educational material suggests that establishing a learning profile may help the teacher to better understand the student and their needs, should consider the students' interests and may also help support the student's understanding of themselves (Mortimore & Dupree, 2010; Reid, 2016).

This correlates with concepts in the framework of Universal Design for Learning (CAST, 2018) which suggest that students' learning barriers might be addressed through the use of planning that focuses on engagement, adaptable use of materials, and accessible instruction. This might better allow for the flexibility needed throughout the various stages of the student's learning process as well as responding to the variability in the dyslexia population. UDL consists of guidelines for teaching which consider the students' barriers and how to overcome them. It was developed from an awareness that buildings designed to be accessible for disabled individuals were beneficial and more suitable for everyone who used them. The framework of UDL (CAST, 2018) provides checkpoints to design and plan materials for a range of students (See Figure 2.1)¹². A more accessible, text-only version of the checkpoints is provided in Appendix J. The 'Key Questions to Consider When Planning Lessons' (CAST, 2020) document (See Figure 2.2) has a series of questions designed to assist teachers in reflecting on their lesson planning to consider multiple options in planning.

¹² Both Figures 1 and 2 are used with permission.



Figure 2.1 Universal Design for Learning guidelines (CAST, 2018)
Key Questions to Consider When Planning Lessons

Think about how learners will engage with the lesson.



Does the lesson provide options that can help all learners:

- regulate their own learning?
- sustain effort and motivation?
- engage and interest all learners?

Think about how information is presented to learners.



Does the information provide options that help all learners:

- reach higher levels of comprehension and understanding?
- understand the symbols and expressions?
- perceive what needs to be learned?

Think about how learners are expected to act strategically & express themselves.



Does the activity provide options that help all learners:

- act strategically?
- express themselves fluently?
- physically respond?

From: Universal Design for Learning: Theory and Practice Available at <u>udltheorypractice.cast.org</u> For print and accessible EPUB, contact <u>publishing@cast.org</u> or any book retailer.

CAST | Until learning has no limits[®] | cast.org

Figure 2.2 Key questions to consider when planning lessons (UDL, 2020)

Whilst UDL encourages many inclusive and accessible practices (the use of feedback, structured instruction, praise for effort, scaffolding, collaborating and adaptive materials), criticism of the model suggests that it overreaches in its claims to be based on neuroscientific evidence and that it is too much like an updated version of the learning styles theory (Boysen, 2024). Whilst there is an emphasis on learning materials and

instruction being offered in different formats, this may become an unrealistic and burdensome task for teachers. Parameters should be established so that the teacher's preparation time and any costs incurred are reasonable from a practical standpoint.

Concerning planning lessons, Reid (2016) suggests that 'it is important to view teaching approaches and programmes in relation to the individual and not to the syndrome, or label' (pp. 175-176) and to deliver constructive feedback in a positive and encouraging manner, particularly with sensitivity to the potential that the student may have had previous negative educational experiences (Eide & Eide, 2011; Reid, 2016). Darrow and Adamek (2018) consider that several areas can be modified or adapted for inclusive teaching, including the time given to complete a task, the degree of difficulty, the level of participation, the student's learning goals, instruments and the teaching space. Reid (2016) suggests that the learning environment including how the room is arranged, where the student and teacher sit, the lighting level, the amount of free space and type of chairs and the level of noise and stimulation should be carefully considered for dyslexic students. By asking students how to solve their difficulties, using self-reflection and working in collaboration (Vance, 2004), the teacher and student might experience more of a mentorfriend (Lehmann et al., 2007) as opposed to a master-apprentice (Folkestad, 2006; Schiavo et al., 2020) teacher-student relationship. Considerations may need to be given to young children who may not be able to communicate their needs and to adult learners who may have developed unhealthy coping mechanisms that may be barriers to communication. The setting of an instrumental music lesson provides opportunities for a number of adaptations to the learning environment, instruction and materials.

2.7.2 Access to provision

In 2011, the National Plan for Music Education (Department of Education, 2011) established that music education should be available to all students in England and that Music Hubs should act as coordinators of specific expertise in certain areas, including that of inclusive teaching, and should provide recommendations for the use of assistive technology or adapted instruments as beneficial to students with specific needs. However, Gall (2021) notes that neither the National Curriculum nor the National Plan for Music Education

mention 'effective teaching strategies for Disabled pupils or ways in which assessment systems should or could take account of their additional needs' (p. 49).

A report of data on music hubs indicates a low number of students with SEN statements participating in Whole Class Ensemble Teaching (WCET) or taking part in ensembles or choirs in England in 2017-18 (Fautley & Whittaker, 2018). There may be many factors influencing this low uptake of provision for SEN students, but it is suggested that music teachers may not be adequately prepared to teach students with special needs (Gall, 2021). Many music teachers come from a conservatoire performance background and may not have had specific teacher training before they begin a teaching career; others may perform to an advanced level but have learned music with individual teachers rather than in an institutional setting (Mills, 2004; Boyle, 2020). In a survey of 330 teachers, factors such as 'quality, relevance, availability and access' related to their views of CPD (continuing professional development), and the cost of the training in addition to a loss of earnings was seen as a potential barrier (Boyle & ISM, 2020, p. 55). Some research indicates that training for music teachers which offers them the opportunity to work with students who have special needs not only increases their sense of confidence in this area but also inspires them to make adaptations that are beneficial to their work with other students (Hourigan, 2007; Van Weelden & Whipple, 2005).

A report published by Arts Council England highlighted a paucity of research on disabilities in those working in classical music, apart from concerns about limited work opportunities related to disclosure (Cox & Kilshaw, 2021). This report did not differentiate between disabilities, but it seems particularly clear that those with hidden disabilities might link disclosure with potentially adverse consequences. Students without an official assessment may be at a disadvantage. A child who wants to learn a musical instrument but has not been assessed, or even identified as having a specific learning disability, may begin lessons with a music teacher who has not been trained to identify characteristics of dyslexia. Their selfesteem might be affected negatively if their education takes place in an environment that does not consider their learning needs or attributes their difficulties to other causes. This outcome might be avoided with proper support for families and training for teachers.

This point has relevance for the implications of the current research, as it raises concerns relating to the training of instrumental teachers and how they develop their knowledge of inclusive teaching. Instrumental teachers may face a sense of isolation depending on their working arrangements; however, they may hope that certain channels available to them could offer training and information (Burwell et al., 2019). As it can be difficult for instrumental teachers to access information (Boyle & ISM, 2020), there is a role for more high-quality training and more sharing of good practice; this thesis contributes to the specific realm of piano pedagogy, with transferable value to other instrumental teaching and music teaching contexts. Teachers might benefit from opportunities to reflect on and embed these practices in their teaching with the availability of ongoing support. It is worth considering which channels might enable the greatest impact and have the furthest reach, for example, through music hubs, music unions, music exam boards, music educational organisations and educational establishments.

2.7.3 Summary

The literature reviewed in this chapter indicates that there is scope for the development of learner support, strategies and pedagogy that supports a positive orientation to learning. Moreover, the need for more inclusive teaching practice is becoming an increasingly urgent matter with music hubs being required to demonstrate their commitment through the development of inclusive music strategies and the appointment of music hub inclusion leads. The Power of Music to Change Lives: A National Plan for Music Education (DfE, 2022) states that schools and music teams 'should consider how they can bring in disabled musicians, music leaders and music educators, disability specialists or disabled-led organisations, to help train their staff and build an inclusive musical offer' and encourages 'collaboration' between schools and organisations with this expertise (p. 43). Indeed, without opportunities to engage with music, DfE (2022) notes that young people with SEND will be denied the 'chance to experience the joys of the subject and to progress their learning in it' (p. 42). The development of knowledge and practice standards would be of great benefit for teachers, many of whom may not be aware of the impact of dyslexia on students and their families' lives, and who will then have the power to improve the music learning experience for dyslexic students. A greater understanding of the risk factors for

dyslexic students, as well as promotive factors which might help students' build resilience, would offer insights as to how best to support them.

2.8 The research questions

The results of this literature review highlight the limitations of research in this area and support the need for a better understanding of how teachers conceptualise the pedagogical process for dyslexic students and what their experiences might reveal about this. Thus, the first research question emerges from these considerations:

RQ1: What are the perceptions of music teachers regarding their experiences and utilisation of pedagogical practices, including strategies, methods and material, in teaching dyslexic students?

Due to a paucity of research related to how dyslexic students and their teachers view graded music exams, there is little understanding of their support needs in this process and how they perceive the use of reasonable adjustments in the exam setting. The second research question is a development of this theme:

RQ2: Do adjustments and accommodations which music exam boards allow offer dyslexic students unbiased inclusion? What are the perspectives of teachers and students on this topic?

An underdeveloped area is that of dyslexic students and strengths in the music learning process. Creativity (Gobbo, 2020; Rooke, 2015), character strengths (Kannangara et al., 2018; Rowan, 2010; Sanderson-Mann & McCandless, 2006) and high-level reasoning strengths (Eide & Eide, 2011) have been recognised in dyslexic individuals. Research on dyslexic children has found them to have greater emotional reactivity, vivid imaginations and intuitive thinking skills (Rolka & Silverman, 2015; Sturm et al., 2021). Although strengths are somewhat underrepresented in the music and dyslexia literature, aspects such as learning music aurally, recognising patterns, using problem-solving abilities to learn musical concepts, improvisation and performing have been discussed (Oglethorpe, 2008; Miles et al., 2008; Rolka & Silverman, 2015). A resource that is strengths-focused might be a useful means of encouraging the identity and positive self-concept of dyslexic students. A study of

teachers and dyslexic schoolchildren found a significant correlation between teachers' expectations of dyslexic students and students' achievements (Hornstra et al., 2010). This may affect teachers' perceptions of dyslexic students; focusing on strengths may also imbue both teacher and learner with hope and endurance through challenging times. In light of this, the third question is formulated:

RQ3: Do teachers, parents and students perceive dyslexic students to have specific strengths? How might these strengths be utilised in the music lesson context?

Including multiple perspectives, particularly those of dyslexic students and musicians, may further support the development of instrumental pedagogy as well as offer valuable insights from music pedagogy towards working with dyslexic learners in other contexts. Although there is a body of research on the dyslexic student voice in other areas, little has been done to explore their views on their music learning experiences. Pursuing an understanding of these may bring about an enhanced awareness of their needs and clarity as to how teachers might support them better in the learning process. Thus, the fourth question is articulated:

RQ4: What are dyslexic students' perceptions of their music learning experiences? How might their voices be nurtured and amplified in the pedagogical process?

The risk-resilience literature (Catts & Petscher, 2022) identifies several factors which may mediate risks for dyslexic students. The importance of parental involvement in the lives of dyslexic students as a promotive factor against risk is well-documented. However, little is understood about how the instrumental music teacher, parent and student interactions might support the development of resilience in the dyslexic student, or how these relationships might be mutually beneficial. The fifth research question emerges from this reflection:

RQ5: What are the teacher, student and parental perceptions of the pedagogical process and roles in the music lesson context? How can these roles collaborate to collectively enable positive outcomes?

In summary, I return to the questions asked by Morrow (2023): 'Have music education specialists thoroughly researched this issue?' and 'What researched-based solutions have

been provided to teachers so they know exactly how to help their struggling students?' (p. 26). Teaching students with dyslexia involves more than a set formula of strategies, but rather consists of a complex relationship between teacher and student interwoven with other relationships: parents, schools and exam boards. This deserves to be studied in theory but also in practice. By examining the literature, investigating student-teacher-parent relationships from multiple perspectives and conducting cycles of action research in my teaching practice, the aim is to bring about transformative understanding and change whereby instrumental music teaching for dyslexic students may be improved and enhanced.

REFLECTIVE STATEMENT I: MOTIVATIONS AND CHALLENGES

RS1 Bronfenbrenner's Ecological Systems theoretical model (1979)

In this first reflective essay, I explore my motivations for conducting action research and the challenges I faced. Secondly, I briefly describe the theory and outputs of my research. I draw on Bronfenbrenner's (1979) ecological systems theory which suggests that the interactions between individuals and their environment affect their development on a number of concentric levels. The model (See Figure RS1.1) explains these levels as concentric circles moving out from the individual at the centre, with the next immediate circle the microsystem where close (proximal) interactions take place with family, friends and teachers. Winter et al. (2000) highlight the researcher-practitioner position: 'Because I am at the centre of my research process, any accounts produced must necessarily show the way in which I have come to understand myself and some of the distorted lenses through which I view my practice' (p. 30). In the microsystem, an individual's behaviour, beliefs and values are shaped. According to Bronfenbrenner (1979), the next level is the area where these microsystems interact: the mesosystem, with the implication that these microsystems influence one another. The exosystem explains the context in which the person lives socially and includes factors like government policies, local community and media. Broader societal forces are involved in the macrosystem, the next concentric circle, and include areas like culture, traditions and values. The outermost levels represent the chronosystem, or development over a chronological period.

I also utilise Bronfenbrenner and Morris' (2006) bio-ecological system theoretical model, described in greater detail in Reflective Statement 2, which explains development through analysis of a 'process, person, context and time model (PPCT)' (p. 798). Some important aspects in the investigation of this model are that any 'components' to be studied must be 'maximally relevant' to the research questions being studied, that the PPCT model should include 'more than one proximal process' (or interactions) in order to have maximum effectiveness and that these interactions are bidirectional (p. 808). The research questions for this research (Chapter 1) relate to dyslexic students in the instrumental music teaching context and lessons (proximal processe) which took place over a four-year period. Bronfenbrenner & Morris' (2006) bio-ecological system theoretical model provides a more

detailed explanation of development at the microsystems level. The students are introduced in Reflective Statement 2. The next section details my motivation for the research.



Figure RS1.1 Adapted from Bronfenbrenner's ecological systems theory (1979)

RS1.1 Motivation

The aim of this research is to improve and enhance the quality of teaching for dyslexic students through a process of 'reflective learning' which Moon (2004) describes as 'a purposeful framing and reframing of material in internal experience with the intention of learning' and which is also guided by relevant literature and from data collected in the course of my research. I quote McNiff's (2004) fundamental questions of action research here:

- How do I improve what I am doing?
- How do I help you to learn?

• How do I improve what I am doing for our mutual benefit? (p. 19)

These questions relate to the microsystem or interactions primarily between myself and students but also include their parents and my academic supervisor. Kember (2002) notes that other outcomes gained through action research might include the 'development of skills, changes in attitudes and the development of revised practices' (Kember, 2002, p. 92). My professional development, as a reflective practitioner, researcher and teacher, is detailed in Chapter 12. Beyond the motivations mentioned by McNiff (2004), I was also keen to provide resources which might benefit others through conferences, workshops or publications and which, I hoped, might bring about positive transformational change across levels of society, including educational institutions, music education organisations and music exam boards (Nofke, 1997). These resources and opportunities represent my development at the exosystem and macrosystem levels. At the same time, challenging my own beliefs and perceptions about teaching was, at times, uncomfortable.

RS1.2 Challenges

In this section, I will discuss two distinct challenges which I faced during the course of my research: the Covid-19 pandemic and lockdown period and the vulnerabilities associated with researching my own teaching practice. When I began my research in 2018, I could never have foreseen that the Covid-19 pandemic would change so much of the landscape of my teaching practice, my personal life and my PhD journey.

RS1.2.1 The Covid-19 pandemic

This was a historic event at the chronosystem level, affecting my life in many ways, but also that of my students and their families. Viewing the videos of lessons from this time period reminded me of the feelings I had at that time of being in a type of 'survival mode'. The research I was doing was based on interacting closely with participants in the lesson context, and I felt that online interactions limited our engagement and challenged patterns of behaviour and rapport established in our previous work together. This meant that I needed to adapt to new ways of interpreting their responses and of collaborating with them (Cornejo et al., 2023). A 2020 survey of 621 teachers reported that teachers felt 'exhausted beyond measure' and struggled to keep up with the ever-changing social distancing requirements, finding also that the 'sudden shift to online required teachers to self-manage

production and delivery of online teaching and learning materials, without adequate training and resourcing' (Phillips & Cain, 2020, p. 3,5). There were concerns about students and their families in relation to IT capability with Zoom lessons, including the devices, software and camera angle; this often left me feeling a lack of control (Hall et al., 2021). I addressed safeguarding concerns by ensuring that a parent was available throughout the lesson. At the same time, it enabled me to see into student's practice spaces and to interact with their families in novel ways.

From a researcher perspective, my experience correlates with Cornejo et al. (2023) which found that women taking part in graduate studies were especially impacted during the pandemic as they sought to balance boundaries between home responsibilities and work. At that time, I had a daughter preparing to take A-level exams, a son preparing to take GCSE exams and a son in Year Five of primary school; I was teaching piano lessons to fifteen students weekly and was a graduate teaching assistant on an MA programme at the University of York. This was in addition to my doctoral research.

Campbell (2021) found that participant recruitment for their research was impacted by the pandemic, and this correlated to my challenges with finding more students to teach and more interview participants. Phillips and Cain (2020) found that many teachers felt overwhelmed and exhausted during that time, and this may have influenced their willingness to participate. There was a great deal of uncertainty about the future and loneliness as normal social interactions were interrupted. Even with disruptions in our personal lives, difficulties with the transition to online lessons and interruptions from the setting change, it was a testament to my students and their parents that they persevered with lessons. Although there were many unplanned factors which were out of my control, I also continued to be present for the students and to do the best that I could under the circumstances. Over time, I began to see benefits to online lessons regarding the balance of power in our interactions; this is discussed further in Reflective Statement 4.

RS1.2.2 Researching my teaching practice

Whilst the pandemic had an impact on teaching, there were also areas of discomfort and vulnerability involved with researching my own practice (see Chapter 4). Dadds (1993) describes this in the following:

...we may be entering into processes by which we deconstruct some basic, historically rooted views of ourselves. In such processes our existing images of the professional self will be challenged, questioned, re-thought and reshaped in some degree. These processes are necessary if change and development are to occur and self-study is to lead to new learning. We cannot escape them, nor the discomfort they may bring if we value our commitment to professional development. (p. 288)

One example of this was acknowledging the need to change my proximity to the student and reduce verbal instructions; although this was a new way of teaching for me, I recognised that this had benefits for the student's metacognitive development. These uncomfortable feelings are what Mockler and Groundwater-Smith (2015) describe as 'the unwelcome truths' of action research, meaning that a practitioner's feelings of certainty and selfconfidence may be disrupted in the process and this may lead to a 'catalyst for rethinking and recasting practice' (p. 606). Another example was when a student refused to do an activity: instead of reacting defensively, I learned to develop sensitivity to their lives outside of our lessons and to recognise the impact that might have on their ability to concentrate (Reflective statements 4, 7). Dadds (2003) cautions that 'feelings generated from rigorous and close self-study can verge on the traumatic when one does not recognise or appreciate the "self" revealed by the analysis of one's practice' and further warns that 'this can lead to feelings of personal insecurity, loss and grief' which have 'the potential to be destructive' (p. 269).

For action research to be truly empowering in the sense of bringing about change, uncertainty and chaos must be accepted (Heikkinen et al., 2012). Instead of feeling that I was getting clear answers to my questions, often further questions emerged. Action research is risky (Winter, 1989) and cycles can be messy and unpredictable (Townsend, 2010). Often, I was not sure how to address the complexities that arose, nor how to make sense of them when looking at the 'supposedly' orderly cycles of action research which are displayed in models. Reflective analysis has led me to consider some shifts in my thinking which could be described as coming from 'contradictions, doubts, dilemma and possibilities' (Cunliffe, 2004, p. 38) experienced during the course of the research. As Voltaire wrote: 'Doubt is not a pleasant condition, but certainty is an absurd one' (Tallentyre, 1919, p. 232); by embracing uncertainty, I enabled myself to consider new ways of working. This is the

phronesis which Aristotle described, the ability to be familiar enough with a situation that I can act with practical wisdom, or as Elliott (2009) states: 'The greater the particularisation of situations the more they take the complexity of making wise judgements and decisions into account ... by throwing light on possibilities for action in other situations' (p. 8). The challenges and uncertainty of the Covid-19 pandemic and researching my own practice were unsettling at times, but as I adapted to them and received support from my supervisor, family and friends, I allowed those things to be precursors for substantive change.

RS1.3 Theory and outputs

From these accounts of my teaching and development which I chart in the reflective statements of this thesis, I began to build a theory of education which culminated in a Knowledge and Practice Standards framework (see Chapter 11, Table 11.2), a foundational document which could underpin a teacher training programme. As another output, I extended Catts and Petscher's (2022) cumulative risk-resilience model of dyslexia to a new risk-resilience model for music training with dyslexic students (see Chapter 11, Figure 11.3). The purpose of this model is to enhance the understanding of promotive and protective factors which might mediate adversity and vulnerability factors in dyslexic students in the music education context. Factors may be influenced by decisions made by government policies, music hubs, music exam boards and the chronological changes that students experience as they journey through the UK educational system. The construction of a theory of education is, as I understand from Elliott (2009), 'a provisional summary of the common features of good practice across a given range of contexts' (p. 35) and Stenhouse (1979) notes that it is a 'tradition of understanding' drawn from 'an articulation of teachers' shared practical understanding' (pp. 19-20). This gives rise to the understanding of knowledge as 'systems of possibilities of action' (MacMurray, 1957, p. 198).

RS1.4 Conclusion

Thus, I developed in my interactions with the student, our relationship and the strategies used at the microsystem level and in my interactions with the students' parents, my supervisors and the ethics committee; the interactions between these factors are represented at the mesosystems level. There was growth at the macrosystem and exosystem level in my shifting views on dyslexia, the early means of disseminating my

findings through presenting at the higher educational level and in workshops and conferences (see Chapter 12), as well as the production of a new risk-resilience model for musical training and a Knowledge and Practice Standards framework (see Chapter 11). This development has taken place since my PhD began in 2018 in terms of a period of time (chronosystem level). Smith (1969) refers to another Greek term for time, *kairos*, or 'opportune events', describing the 'significance and purpose of events and... constellations of events yielding results which would not have been possible at other times and under other circumstances' (p. 2). In my reflective statements, I point to specific moments, *kairos*, which have signified my growth as a teacher-researcher, and which are investigated through the 'process-person-context-time' model introduced in Reflective Statement 2 (Bronfenbrenner & Morris, 2006, p. 808).

REFLECTIVE STATEMENT 2: INTRODUCTION TO THE STUDENTS

RS2.1 Introduction: Process-Person-Context-Time Model

In the following reflective essay, my focus is on the components of the Process-personcontext-time (PPCT) model (Bronfenbrenner & Morris, 2006) through which interactions occur as a means of accounting for developmental growth; these 'proximal interactions' or 'engines of development' (p. 118) are positive when they produce 'competence' which may also 'serve to reduce and act as a buffer against effects of disadvantaged and disruptive environments' (p. 805). Primarily, these interactions occurred in our lessons and with the student's parents, and as Bronfenbrenner and Morris (2006) note, those at the microsystems level influence and are influenced by one another. In this reflective essay, I draw on the PPCT model (Bronfenbrenner & Morris, 2006) as a lens through which to view contextual understanding of the process of lessons and research, my relationship with the students and their parents, frequency and duration of lessons and the lesson environment and my relationship with the students and their parents.

RS2.2 Process

The proximal processes of development, or interactions, occurred during lessons and with parents, as described in section 1.1, whilst the extended process began with teaching experiences prior to the research. My research process had four larger cycles of action research with many smaller cycles in between; I reflected on interactions (proximal processes) and then re-examined literature and interview findings in order to implement them into my teaching practice. The following figure (Figure RS2.1) shows the four main cycles (Somekh, 2006):

Action research cycles

Reconnaissance Reflecting and Planning Implementing and Observing Evaluating and Reflecting

Figure RS2.1 Action research cycles

Oglethorpe (2008) states that teachers need 'humility, adaptability, imagination, awareness and genuine interest' in the student (p. 28). My initial understanding of dyslexia affecting music learning and also of reflective practice came from what I learned as a student in the MA Music Education: Instrumental and Vocal Teaching programme at the University of York. In reflecting on my experiences teaching dyslexic students as well as another student with dyspraxia/developmental coordination disorder (DCD), I had not felt that my attempts were always successful, and this correlated with other accounts. Oglethorpe (2008) reported that after unsuccessfully teaching some dyslexic students, she suggested to their parents that they try a different instrument or quit altogether (p. 19). After hearing this approach confirmed by another teacher, I made a note in my reflective journal:

I recall feeling dissatisfied when others mentioned that there were times when you had to tell dyslexic students that 'learning the piano was not for them'. I have no doubt that this was grounded in desiring the best outcome for students, but it also made me wonder how that conversation might affect a dyslexic students' sense of self-concept and even feeling of control over which instrument they should play. I question if this is my role as a teacher.

Drawing on the literature review in this thesis (Chapter 2), I understood that dyslexia teaching strategies such as multisensory, overlearning, personalised, systematic and structured, may be effective for dyslexic students. As these were drawn originally from literacy contexts, I wondered how they might be utilised appropriately in the music lesson context as music is already, by nature, a multisensory activity. Based on my experiences and my understanding of the literature, the following questions began to emerge:

- What is the best approach for planning their lessons? What sort of balance should there be between number and duration of activities? How do I make sure they are learning in a systematic and structured way that suits them best?
- 2. How much repetition would they be able to tolerate and still maintain their interest and motivation in learning? What type of components should be repeated, how often and how should this be incorporated into the lesson?
- 3. How might I identify and build on the strengths that I saw in my students?

- 4. How would I build profiles and collaborate with the student to establish the best ways of working? Where are the boundaries between my roles and responsibilities and theirs? How do I help them to practise at home?
- 5. How might technology be used to support the learning process?
- 6. Was it wise to encourage them to take exams and should we use access arrangements? How would I prepare them for using those in the exam context?
- 7. How do I support the student's emotional well-being in the learning process? How do I manage my own emotions as the teacher-researcher in these lessons?

In the reflective statements of this thesis, I give evidence for how I addressed these questions. My experiences correlated with findings from interviews with instrumental and vocal teachers in my research who reported that they found a lack of relevant information in relation to appropriate training for music teachers or practical resources (see Chapter 6). I drew on literature suggesting that the development of learner profiles might assist a teacher in creating a more individualised approach (Reid, 2016) and as a starting point, I created profiles based on my observations of the students. The following section introduces the students with the learning profiles which I developed when the research commenced.

RS2.3 Person

Bronfenbrenner and Morris (2006) categorise person characteristics as 'disposition, resource and demand'; these might influence immediate interactions but also may act as predictors of future development (pp. 795-796). According to this model, they can be seen on a spectrum as either developmentally 'generative' or 'disruptive' characteristics; examples of disruptive characteristics range from 'impulsivity' to 'apathy' whilst generative characteristics relate to 'curiosity', perseverance and eagerness to learn (p. 810). Essentially, an interest in interacting with the environment then leads to engaging in more complex activities, the outcomes of which, along with environmental factors, affect what a person believes about themselves, according to Bronfenbrenner and Morris' (2006) conception of 'developmental generative disposition' (p. 810). With my students, a resource characteristic was dyslexia which could be seen as both a limitation and an asset. I was aware of the tremendous responsibility that my attitudes and actions would have as an influence on how the students viewed themselves and their capabilities in our lessons. The students viewed themselves through lenses created by parental and societal influences, positive and negative, as well as their past experiences at school. Demand characteristics relate to those which influence developmental change by the way in which they cause a reaction in other people or the environment around the person. In the following sections, I introduce my students. Future reflective essays describe some of the developmental growth which occurred throughout the research, but this reflective essay is presented as an introduction and a starting point.

RS2.3.1 Introduction to Ben

Ben (a pseudonym) started taking lessons in 2017, at the age of nine. He was still in primary school at the time and our relationship has continued through to supporting him with GCSE music in secondary school. I was acquainted with Ben's family in various social contexts and I taught other family members before Ben began having his lessons. Ben's family had observed some discrepancies in his school work and suspected a SEN issue; however, he did not receive a formal assessment until 2019. From our first lesson, Ben was eager to learn, curious, responsive in lessons and willing to attempt practising at home. He had a very supportive home environment with parents who had professional backgrounds which lent themselves to an understanding of diverse needs in children, negotiating with authorities and a sensitivity to the challenges Ben may face. They were willing to offer support during practice in the form of reminders or encouragement, but neither parent had previous musical experience.

I observed that Ben had a good sense of pitch, good aural memory, excellent coordination skills (he rides a unicycle) and was quite confident when performing for others. The challenges that I observed in Ben had to do with low confidence at certain times; this seemed to manifest itself in an apparent lack of effort in activities which he normally solved by himself or in a reluctance to attempt a task in which he might not be guaranteed success. The other challenge I noticed was that he frequently forgot his music books. He also had some difficulty remembering the starting place on the keyboard although he would simply

transpose the song and play it from memory wherever his hands tended to land. I include the following excerpt (Video excerpts are not available to readers due to GDPR and participant confidentiality) of one of our earliest recorded lessons as a means of introducing his personality, to give an understanding of our rapport together and to illustrate his approach to the task of playing his pieces.

RS2.3.2 Introduction to Alex

Alex began having lessons in 2018, at the age of eight. Lessons began whilst he was in primary school, with early educational experiences having been guite negative until he moved to another primary school with a more supportive ethos. I was acquainted with Alex's family socially for a number of years before we began having lessons together. His parents are supportive in ensuring that he attends lessons regularly, but neither have musical experience so assistance with practising at home is somewhat limited. His mother has dyslexia and was aware early on that he had similar traits. According to Alex's mother, she often relied on his father, who is not dyslexic, to fill in forms and support in other ways to compensate in areas she finds challenging. Her early educational experiences, which took place outside of the UK, were overwhelmingly negative as she described a system in which there was little awareness or understanding of dyslexia. His mother understood the importance of 'drilling' (repetition) for reading skills and had worked extensively with Alex outside of school. I observed that Alex was confident, resilient, exuberant, ambitious and eager to demonstrate what he had practised or learned. He often shared with me experiences from school, positive and negative. Some behavioural characteristics included impulsivity displayed by pressing the pedals, interrupting or changing topic quickly. Other challenges which I observed were related to processing rhythm and pitch playing together, synchronised to a pulse. Usually notes were correct, but rhythm might be read or remembered inaccurately. In this excerpt (Video excerpts are not available to readers due to GDPR and participant confidentiality), I wish to give a glimpse into our interactions with each other and to introduce you to his personality.

RS2.4 My observations of these lessons

In my observations of these lesson excerpts, I observe that Ben rarely looks at the score, although he can play his pieces successfully once he remembers where to begin. This is after

we have done a substantial amount of work on identifying notes on the keyboard and the stave, as well as quite a number of directional games. By asking him to play the duet more quickly, I am trying to evoke the feeling of Hungarian dances where there is an increase in tempo as the piece progresses, and at the same time I am curious as to whether or not increasing speed might test his memory and skill. In reviewing the video, I think it would have been more useful to play it at a tempo where he was comfortable without adding any additional pressure for speed. I also note that he was not looking at either piece throughout this excerpt, but was focused mainly on the keyboard.

In Alex's lesson, I observe that he is even more fidgety than I recall during our lessons. I let students know that I have a rule not to stand on the pedals or press on them when they are not playing a piece and he had to be reminded of this frequently. This was quite irritating to me, as I was concerned that the pedals might become damaged. In the future, I might have him stand to play pieces or have him turn away from facing the pedals so as to remove the temptation. At the same time, I am touched that he was excited and wanted to share with me about recognising a C chord at his church. I also think he was trying to convey that he was glad that I am not the kind of teacher who expects perfection and is flexible. Taking the time to explain patiently when there are questions about exams is sometimes a time consuming but important process. It is easy to forget how much of that knowledge might be taken for granted. Alex, who enjoys video games, is highly motivated by attaining levels and achieving points.

RS2.5 Time

The lessons included in this study for research purposes took place over a four-year period. I taught the students prior to commencing research (with Ben starting lessons with me in 2017 and Alex in 2018), so neither were complete beginners when we started the research process. Part of the reconnaissance cycle of the research process involved implementing information I had learned from previous teaching experiences in order to refine the research questions and consider specific foci for future cycles of research. Bronfenbrenner and Evans (2000) refer to the term 'exposure' which describes the contact between myself and the students in terms of 'frequency, duration, timing and intensity' (p. 118). Throughout most of the research, we had 30-minute lessons. With Ben, we had some lessons which

were like guided practice sessions for 15 minutes two or three times per week. Generally, the lessons took place weekly during the thirty-eight weeks of school term. Many of the lessons were video recorded, following prior discussion with the students and informed consent from their parents (as both students were minors), in order to help both myself and the student feel comfortable with the feeling of being recorded. I usually had weekly contact with the parents, in the form of short chats after lessons or in WhatsApp message form. The next section describes the context (1.2) in which the lessons took place.

RS2.6 Context

The context was the environment in which the processes took place. Initially, these lessons took place in the context of my home, although once lockdown was instituted during the pandemic, they took place in a digital context on an online meeting platform using the application Zoom until face-to-face meeting was allowed again. In my home, I am fortunate to have a room which is dedicated for music lessons and it contains bookshelves and pedagogical resources, an upright piano, boxes containing instruments and games, a small desk and a guitar which hangs on the wall. The lighting is soft. One wall is dark blue, and the other walls are light grey. Normally, the piano is kept regularly in tune, although during lockdown I was not able to have a tuner come to my home. Initially, I sat in a chair next to the student who sat on the piano stool. However, during the pandemic, physical proximity was limited due to online lessons and when we returned to the same teaching space we had varying social distancing requirements in the room. When life resumed more normally, I found that there were advantages to allowing more physical space between us which I detail more in Reflective Statement 4. I have an adjustable piano stool, but also a backed chair for students who need additional support. Within these lessons, as with all of my teaching, I tried to keep the atmosphere encouraging and positive, so that even if they have not practised much during the week, students still feel they have achieved something significant during the lesson. I tried to convey to my students the sense that I am interested in their lives beyond piano lessons, and I was privileged to be able to hear about experiences they enjoy and their challenges at school.

RS2.7 Conclusion

Developmental growth, according to Bronfenbrenner and Morris (2006), occurs through 'proximal processes' leading to either 'competence' or 'dysfunction' (p. 118). The notion of competence implies that the learner has acquired knowledge and can integrate specific skills and conduct their behaviour in a way that produces growth in other areas as well. Dysfunction, on the other hand, would indicate that growth is impeded by ongoing challenges with skill acquisition and behaviour management which also affects different areas of activity. These concepts, of course, relate to my learning and pedagogical development as a teacher as well as to the learning of the students. The aim of this research was to enhance and improve the quality of piano teaching for these two students in such a way that they would be able to use areas of compensation and areas of strength as a buffer against their challenges and that this might, in turn, also positively affect their self-belief and promote metacognition and self-regulation leading to competence.

Chapter 3 RESEARCH DESIGN

This chapter outlines the philosophical position of social constructivism as the theoretical framework underpinning this research. The justification for the methodology of action research and flexible cycles of obtaining data are summarised. Qualitative methods to gather data include semi-structured interviews, observations, video evidence from lessons and a questionnaire. The means and procedures for data analysis are discussed and issues of reliability and validity are considered.

The research aims are the focal point for beginning a discussion on research design and methodology. As mentioned in Chapter 1 (1.2), the overarching purpose of this research is to improve and enhance the quality of instrumental teaching for dyslexic students. By developing a better understanding of dyslexic students and the approaches which aid them in learning music, including through researching my own practice as a piano teacher working with dyslexic students, a secondary goal would be to generate resources which may be beneficial for students, parents, teachers, educational establishments and music exam boards. Some of these resources are presented in Chapter 11. Of particular importance in this research is the empowerment of the student voice which might inform their collaboration with the teacher and support the teacher's understanding of the student in a mutually beneficial process.

3.1 Theoretical framework – Social constructivism

To understand how research will have meaning and how the data will be interpreted, it is important to consider the researcher's paradigm. A consideration of the meaning of 'paradigm' and its importance in the methodological framework of my research begins with examining the definition. A paradigm is described as 'a loose collection of logically related assumptions, concepts and propositions that orient thinking and research' (Bogdan & Biklen, 1998, p. 22) or a way of seeing the world that 'frames a research topic' and consequently influences our thinking and approach to the topic (Hughes, 2010, p. 35). Fraser and Robinson (2004) interpret the paradigm as a 'set of beliefs about the way in which particular problems exist and a set of agreements on how such problems can be investigated' (p. 59). Paradigms are belief systems and constructs, according to Guba and Lincoln (1994, p. 108), who developed three defining questions to distinguish types of

inquiry: ontological question ('What is the form and nature of reality?), epistemological question ('What is the nature of the relationship between the knower or would-be knower and what can be known?') and the methodological question ('How can the inquirer go about finding out whatever he or she believes can be known'?) (p. 108). The philosophical perspective that knowledge is socially constructed, described by Crotty (1998) as the understanding that 'truth, or meaning, comes into existence in and out of our engagement with the realities in our world' (p. 5), establishes the underlying theoretical framework for my research.

3.2 Discussion of positivist, interpretivist and pragmatic approaches

My approach was a combination of interpretivist and pragmatic, corresponding to my interest in people and how they relate to each other, in which I seek to understand 'how their worlds are constructed' (Thomas, 2009, p. 75). Whilst the following section compares the interpretivist with the positivist approach, the pragmatic approach refers to an 'individual decision maker within an actual real-world situation' gaining an understanding of 'the multiple factors involved in people's actions in a given situation' (Salkind, 2010, p. 1073), which resonates with my position as teacher-researcher.

The positivist approach views the world as being based on universal laws which explain everything that occurs around them (Hughes, 2020). The theory of positivism originated with French philosopher Auguste Comte who sought a scientific methodology which would define laws of human behaviour in the same way as other sciences (Tekin & Kotaman, 2013). Positivists believe that obtaining an understanding of these universal laws comes through observing and describing phenomena and causes; they use scientific research methods to measure their outcomes. The limitations to this position include the fact that measuring human 'intentions, attitudes and thoughts' and producing an absolute theory of knowledge about them is not achievable given the diversity in human behaviour (Hammersley, 2013, p. 24). Social phenomena are too complex to be described in absolutes and quantitative research alone does not allow for the intricacies and complications of social situations. However, the relativist ontology of interpretivism does allow for multiple meanings rather than scientifically measured truths and is a way of gaining a deeper understanding about a unique setting rather than generalising society (Cresswell, 2007). An

interpretivist position allows that there are multiple perspectives and that situations are 'fluid' and are 'richly affected by context' (Cohen et al., 2011, p. 17). In this case, rather than scientific measures being used to reduce the research to generalised theorems, 'thick descriptions' which allow for the complexity of these situations are most useful (Lincoln & Guba, 1986, p. 77); the emphasis is on 'Verstehen' – the researcher seeking to understand the situations and feelings of others (Khalifa, 2019). Interpretivism also places an emphasis on individualisation as opposed to generalisation, and this is vital considering the context of this research and the diversity that occurs within the dyslexic population.

The concept of situating theory within a practice, or praxis, was drawn from Aristotle's concept of phronesis or 'practical wisdom'; this way of thinking challenged teachers 'to realise their educational values in complex practical situations' (Elliott, 1989, p. 85). Schön (1992) describes competence as being able to perceive and take the right course of action in a particular context. In this way, the knowledge or truth is embedded in personal situations and open to being reinterpreted in other contexts. Salkind (2010) indicates four guiding principles in pragmatic research: 'accept chaos in interrelationships among variables; seek an understanding based on human experience; view a problem as a complex problematic situation; and promote activism, democracy, and policy formulation' (p. 1073). Practicebased research allows for the application of these interpretations to bring about change. Salkind (2010) also notes that pragmatic researchers are not looking for fixed solutions, but rather recognise 'problematic situations' and attempt to improve them through their research (p. 1074). In my research setting, I recognised that the resources for instrumental teachers regarding teaching dyslexic students were sparse; this led to the objectives of improving my understanding of dyslexia in the music educational context, encouraging discussion and generation of new resources within the field, and supporting developments to my own teaching practice. In the following sections, I will explore the action research as a methodology (3.3) benefits (3.3.1) and constraints of action research (3.3.2), detail the methods of data collection (3.4) and data analaysis (3.5), describe how the reliability and validity of the research will be evaluated (3.6) and discuss ethical issues (3.7).

3.3 Action research methodology

This section details my design frame of action research in greater depth. The most important consideration in choosing research methods is that they must answer the research questions (Denscombe, 2008). According to Whitehead (1989), the key question for initiating action research is 'How do I improve my practice?' (p. 41). The definition of a practice is 'a form of socially established cooperative activity that involves characteristic forms of understanding (sayings), modes of action (doings), and ways in which people relate to one another and the world (relatings), that hang together in a distinctive project ... and answers the question "What are you doing?" while they are engaged in practice' (Kemmis et al., 2014, p. 155). The purpose of my research is defined as two-fold: the improvement and enhancement of the quality of instrumental teaching for dyslexic students in the context of my piano studio practice, and the generation of resources which may benefit teachers, parents, students, organisations and institutions. The choice of action research as the methodology for this study stems from the benefits and flexibility offered. Somekh (2005) described action research as occurring through a 'series of flexible cycles' which:

...consist of partnerships between participants, transpire in a natural setting, initiated by a desire for transformative change, emphasises self-reflection, requires the researcher to have a broad understanding of the existing literature but also situates the action research in a comprehensive frame of reference, promotes influential development and change. (Somekh, 2005, pp. 6-8)

Thus, qualitative research approaches were the most appropriate research method for obtaining data which would enable a depth of understanding of the practical music educational contexts and concerns involving dyslexic students, their parents and teachers. These were from a variety of sources including my lesson plans, observations, reflective journal, questionnaire, video evidence and semi-structured interview transcripts as well as data collected from an evaluative focus group. Following a discussion of the rationale for using action research including benefits and limitations of the methodology, I outline the methods of data collection (3.4) in greater detail.

Action research cycles



Figure 3.1 A timeline of the research process showing cycles of action research

3.3.1 Benefits of action research

Action research methodology has been used in many fields but is particularly suited to professions such as education and healthcare as it is characterised by 'reflexivity, collaborative sense-making, and finding meaning in feeling' (McLeod, 1999, p. 20). There are many benefits to action research. Laprise (2017) writes that action research 'challenges assumptions we may have about our teaching and our students' learning and, through the collection and analysis of data, can either show us that what we are doing is working or prompt us to find an alternative approach' (p. 28). Action research can help teachers develop their reflective practice (Sowa, 2009) and overcome barriers in their teaching practice. Nofke (1997) describes three potential motivations for teacher-researchers: the desire to understand and improve their teaching, to disseminate the knowledge with others and to transform society through positive changes. It can be seen to 'improve a teaching practice' (Elliott, 1991, p. 3), but also to develop a 'capacity for discrimination and making judgements in complicated human situations' (Carr & Kemmis, 1986, p. 223). It seems particularly appropriate to use this methodology, as an understanding of dyslexia in the music learning context is limited, as discussed in the literature review. Dyslexia is a complex condition with many variable factors which occur across a spectrum in individuals. Action research allows for informed plans to be tested directly in the teaching context in order to

establish if the theories work in a practical setting; a benefit is that it allows for personalisation and therefore specific relevance.

Action research allows researchers to document progress using records. Kemmis et al. (2014) identify the following changes that can be recorded: 'changing activities and practices; changes in the language and discourse in which we describe, explain and justify our practices; changes in the social relationships and forms of organisation which characterise and constrain our practices; and redevelopment in our mastery of action research' (p. 14). Denscombe (2014) refers to action research as having four characteristics: 'practical, change producing, cyclical and participatory' (p. 176). The following sections explore the integration of these characteristics in this research.

3.3.2.1 Practical aspects of action research

The practical nature relates to issues with dyslexia and music learning which I observed in my teaching practice. Geiger (2016) describes action research as 'a practitioner's research that is performed in order to improve one's own performance', which 'involves a group of people collecting evidence and making decisions regarding their performance, attitudes, and beliefs, in order to understand them better and to improve them' (p. 12). Through dilemmas in my teaching practice, I realised that there were limited practical applications available to music educators from the extensive research in related interdisciplinary fields: educational, cognitive, neurobiological, psychological and general dyslexia research. The issue was that I saw a practical need for my professional development regarding dyslexia and related factors. As a teacher, I believe that there could be increased understanding and an improvement in the resources available to instrumental teachers about dyslexia. I also believe that there have been advances in related fields which could impact music education for dyslexic students. Variables which can impact effectiveness include the teacher's connection with the student, the student's responsiveness and psychological state, cooccurring conditions, curriculum content and structure, flexibility, manner of delivery and a strengths-focused approach as discussed in Chapters 2 and 3. These are practical areas with great significance for dyslexic students.

3.3.2.2 Change-producing aspect of action research

Action research is intended to produce 'change (action) and understanding (research)' (Dick, 1997). It is intended to be a bridge between practical aspects and theoretical research (Somekh, 2006). The following changes were anticipated from this research: to enhance my teaching practice through greater awareness of the needs of dyslexic students, research findings and available resources; to improve music lessons for dyslexic students, and to provide a basis of understanding leading to potentially transformative impact in the field of music learning and dyslexia.

Stenhouse (1984) distinguishes reflective teaching from action research by describing the latter as 'a systematic enquiry made public' (p. 77). This research is informed by an extensive literature review (Chapters Two and Three), a qualitative study investigating the views and practices of other instrumental teachers, parents and dyslexic music students, and observations made in the context of teaching two students with dyslexia in my piano teaching practice. Swanwick (2012) describes acquaintance knowledge as 'a way of knowing' as opposed to distinct knowledge and states that acquaintance knowledge is 'multilayered, intuitive and attitudinal' (p. 330). Having a relationship over a period of years with the two students in my teaching practice was a distinct advantage in the development of this 'way of knowing' across seasons of development in their lives and mine which also included the Covid-19 pandemic.

My actions were initially taken from knowledge of the literature and past teaching experiences, but as I began to interview teachers and teach the students, the complexity of dyslexia, when it came to music learning, became even more apparent. As this developed, I became more aware of the importance of observing and teaching my students as individuals and avoiding assumptions about dyslexia. Based on my interviews with dyslexic music students, as well as discussions with my students about negative school experiences, I tried to develop more responsiveness to the individual in my teaching but also began to acknowledge limitations in my knowledge, or acquaintance, of the students. Having a strong relationship with the parents in each case enabled other perspectives when responses from the students were ambiguous. Having the opportunity to review lessons showed me where my responsiveness and flexibility might be improved, and I attempted to improve this in later lessons. This knowledge has informed the application of scholarship to my own

teaching practice through the method of action research. Lewin (1946) visualised action research as cyclical steps of action and reflection which need to be adaptable to respond to needed change. In my research, this continuous loop of feedback enabled a flexible yet structured approach to investigate dyslexia and music learning and to collect robust data that would evidence the improvement of my teaching practice (see Figure 3.1 for a detailed timeline of the research).

3.3.2.3 Cyclical aspect of action research

Action research is described as having 'cycles within cycles'; for example, one interview may raise new lines of inquiry which may need to be fed into future interviews (Dick, 1997, p. 1). Planning includes the literature review, interviews and knowledge which feed directly into the action of teaching; this produces observations of results which feed into the next cycle of planning. Following the next cycle of 'action', additional literature may need to be examined and included in the literature review. Kemmis and McTaggart (1988, p. 5) developed a model of the 'action research spiral' which illustrates the iterative process of planning leading to action with reflection leading to revised plans in a continuous evolution. Gibbs' (1988) model of reflection on experiences also informs my research. This model elaborates reflection through a cycle of 'description, feelings, evaluation, analysis, conclusion and an action plan' (Dye, 2011, p. 230). This allowed for more personal reflections and a deeper awareness of the feelings of participants in the research. As I evaluated my teaching experiences, this led to an analysis of what might improve the situation and allowed for reconstructed plans to bring change in future cycles. Critical reflection on past cycles brought clarity to future planning in a flexible, yet concrete way.

3.3.2.4 Participatory aspect of action research

Action research has an important element of being participatory. My role was to participate as initiator, director, teacher and researcher. Building trust, connection and credibility with other stakeholders, especially students, was essential. With the desire to foster a rich and creative learning environment, I gave careful attention to students' concerns, ensured that our interactions were consistent and centred the lessons on students and their goals. Quality of relationships was very important, as was sensitivity to ethical issues, clear role boundaries and the flexibility to adapt and change. These are discussed in section 3.7 in more detail. Other participants included teachers, parents and dyslexics with music learning

experiences and those with expertise in inclusion and music education as contributors and shareholders in the study. Academic supervisors contributed through advising and refining the research strategy and output.

3.3.2 Limitations of action research

Action research may raise questions about wider contexts in the ecological strata of the student's life, for example, in their home, with teachers at school, and concerning music exam board practices, existing research and government policy. This may cause disruption to the status quo by suggesting that changes are needed. However, if authenticity is the aim, discomfort and disruption are necessary over a teacher's desire to prove their own success. Fielding (2004) suggests that teachers ask themselves the following questions, which are quoted below:

- 1. How confident are we that our research does not redescribe and reconfigure students in ways that bind them more securely into the fabric of the status quo?
- 2. How clear are we about the use to which the depth and detail of data is likely to be put? Is our more detailed knowledge of what students think and feel largely used to help us control them more effectively?
- Are we sure that our positions of relative power and our own personal and professional interests are not blurring our judgements or shaping our advocacy? (pp. 302–304).

Whilst these were not comfortable questions to ask throughout the research, for the sake of authenticity and trustworthiness, they should be considered. Some limitations in action research may arise from a teacher's need to feel certainty and to justify their actions, rather than interrogating accounts with rigour. In Chapter 4, these assumptions are articulated as a means of considering their impact on my teaching accounts. By acknowledging and reflecting on critical incidents, support from advisors and seeking advice from the student's parent, I also sought to mitigate these limitations.

3.4 Methods of data collection

According to Whitehead and McNiff (2006) the motive for collecting data 'is to generate evidence to support and test a claim to knowledge (theory)' (p. 63). Drawing a connection to the aims of this research, I sought to collect data which showed how I developed my teaching practice with dyslexic students. I also sought to collect data from multiple sources as a means of triangulation; my aim was a robust evidence base from multiple perspectives (Elliott, 1991). Hopkins (1993) emphasises collaboration in the collection of data, believing that teacher-researchers' relationships with others should be mutually trustworthy and supportive. The key methods of data collection were a questionnaire, observations and reflective journal entries, video recorded lessons, interview and focus group transcripts.

The Reconnaissance stage was the review of key journal articles, books and other sources of information about dyslexia and music education (Chapter 2). As there was no research available on students with dyslexia taking music exams, I produced a questionnaire for dyslexic students to collect data on their perspectives of music exams, including their experiences obtaining information about examination board accommodations and use of access arrangements and reasonable adjustments for which they might qualify. Following ethical approval (section 3.7) I began video recording my piano lessons with two students with dyslexia. The purpose of recording these lessons was primarily to establish where my practice was situated at the beginning to be able to show how it improved over the research period. The next stage involved interviewing instrumental teachers who had experience teaching dyslexic students. Incorporating the results of the interviews and literature reviews, I sought more participants for lessons. However, possibly due to the onset of the Covid-19 pandemic which necessitated online lessons, I was unable to expand my sample of students with dyslexia in my teaching practice beyond the two initial students. Understanding and nurturing the student's voice was a key component of my research in my own teaching, but also as reflected through the interviews with dyslexic participants who had music learning experiences. Throughout the PhD, I kept detailed written notes on supervisory meetings, casual discussions with other teachers, discussions with parents, notes made when observing my lesson videos, and notes on lesson plans and reflective observations of teaching. In the following sections, the data collection methods are examined in greater detail.

3.4.1 Lesson planning and observations

Lesson plans were a natural starting point for the research, as they detailed methods and strategies which I wanted to use and provided indications of repertoire and other material used. As the cycles progressed and change occurred, it was helpful to refer to the record of these lesson plans and to make reflective notes. Using the Universal Design for Learning framework (CAST, 2018) which accounts for recruiting student interest and motivation, adapting curriculum and materials, as well as supporting self-regulation and executive functions (aspects of particular interest for students with special educational needs). This framework enabled an examination of inclusive teaching practices against the backdrop of the literature on instrumental music teaching strategies for dyslexic students. Teaching observations were recorded in a journal which provided the basis for critical reflection of lessons (see sample reflective journal entry in Appendix J); these included ongoing notes on each lesson as well as observational insights from studying videos of my teaching (See section 3.4.4)

Research observations consisted of written notes in the following categories: 'my own perceptions, others' perceptions and behaviour' (Smith & Rolledo, 2018, p. 53). Behavioural observations comprise those made of video-recorded lessons, lesson planning and assessment of lessons. These included teaching notes and reflective journal entries which include these perceptions as well as the pedagogical process with which they are intertwined (see 3.4.5). Some observations provided the opportunity for first-hand accounts of what occurred in the setting of the natural social situation of the piano lesson. Other observations were unstructured and consisted of descriptions of 'conversations, settings, participants, events, behaviours and participant activities' (Cohen et al., 2011, p. 553). My own perceptions included my written notes on reflections on teaching and those which detail casual conversations with other teachers and supervisory staff. Feedback from research presentations was also incorporated. The notes were typed and were written as soon as possible after the observed events. Each note contained the date, the time, the setting and the participants. Participant's comments were indicated by their name and quotation marks. The perceptions of others include student's reflections, interviews, responses to questionnaires and casual conversations with others. My approach was inductive and therefore started with general observations from the data which emerged

into more focused themes. Thomas (2006) describes the purpose of an inductive approach as aiding 'an understanding of meaning in complex data through the development of summary themes or categories from the raw data' with this analysis derived from iterative review and my interpretation of the data (p. 3).

Morrison (1993) refers to the following observable settings which provide evidence: 'physical, human, interactional and programme settings' (p. 80). The physical setting refers to the environment in which the observations take place: in this context, the setting refers to piano lessons. The human setting refers to the things that make participants unique and how they are categorised in the setting: in my research, this was myself as the teacher, with students, parents, experts and critical friends as part of the other categories. Interactional settings refer to the 'formal, informal, planned, unplanned, verbal and non-verbal' interplay between teacher and student (Cohen et al., 2011, p. 543). The programme setting refers to the curriculum applied in the lesson setting. The lesson plans enabled me to establish a chart for structured observations during lessons. In addition, the questions from the Universal Design for Learning 'Key questions to consider when planning lessons' document (See Chapter 2) helped me to reflect on more inclusive practices for my lesson planning and delivery (CAST, 2020). When video material (See 3.4.4) was reviewed, it was useful to compare the lesson plan (See sample lesson plan in Appendix K) and the actual lesson, to link to objectives and to consider how my students were responding.

3.4.2 Questionnaire

A questionnaire was used to collect data from a wider population of dyslexic students, beyond the scope of the piano students in my research, in order to obtain information regarding their perceptions and experiences of music exams. The key objective of this research was to ascertain firstly whether or not students were able to obtain access to information about music exam reasonable adjustments and accommodations, and secondly, whether they felt that the accommodations and adjustments provided them with an equitable exam experience. Prior to beginning the questionnaire, respondents were required to acknowledge that they were informed about the purpose of the research, ethical issues of confidentiality and anonymity, and how the data would be stored and used. The questions asked are included in Appendix E. There are advantages and disadvantages to

the use of questionnaires, with the possibility of collecting data from a large number of participants being the foremost advantage whilst the main disadvantages are that they lack flexibility and cannot be adapted to individual participants, they may be difficult to design and the rate of response may be low (Patten, 2016).

Geographical constraints justified the choice of using questionnaires. I trialled a pilot version of the questionnaire with teaching colleagues, a PhD student with dyslexia and with my supervisor, revising aspects which were suggested in their feedback. The questionnaire could be read with a screen reader and would allow for contrasts and backgrounds to be adjusted by the user, but it is also possible that the layout or the length might nevertheless have been a barrier for someone with dyslexia. I followed web accessibility guidelines (WCAG 2.0, 2024) for the questionnaire, but in some cases that may not have been sufficient to enable ease of completion by all prospective participants. The electronic questionnaire was created in Qualtrics and was distributed via an anonymous link distributed to dyslexic students via a call for participants through the British Dyslexia Association music committee¹³, the Curious Piano Teachers group¹⁴, the local European Piano Teachers Association network ¹⁵ and through social media. The distribution period was designed to follow music performance examination periods so that experiences were fresh in students' minds. My telephone number and email address were available for those who desired further information or preferred a telephone or in-person interview. In either case, anonymity was assured.

3.4.3 Interviews

As a means of expanding my knowledge of experiences, teaching techniques and strategies which could be utilised with dyslexic students, I conducted interviews with instrumental and vocal teachers possessing experience of teaching students with dyslexia, parents of students with dyslexia and individuals with dyslexia who had experienced music lessons. There are several advantages and disadvantages to interviews as a data collection method. The main advantages include the opportunity to record and analyse participants' perspectives and to

¹³ https://www.bdadyslexia.org.uk/advice/children/music-and-dyslexia

¹⁴ https://thecuriouspianoteachers.org/

¹⁵ https://epta-uk.org/

allow for flexibility in the dialogue between the researcher and participant, with the chance for follow-up and clarification questions. The main disadvantages concern the demands of time in recruiting participants, arranging and recording the interview and transcribing it (Hopf, 2004).

I sought participants from the University of York Music Department's instrumental and vocal teaching staff, the European Piano Teachers Association, the British Dyslexia Association music committee network and via relevant social media networks. Often participants would recommend other potential participants; this was usually due to an established, interconnected relationship and the referral lent credibility to my research. This non-probability sampling technique is referred to as 'snowball sampling' and is often used when respondents are difficult to locate or may struggle to trust the researcher (Check & Schutt, 2012).

Using my research questions as the basis for the questions asked in the semi-structured interviews, I sought to learn more about their perceptions of and experiences with dyslexia. I also asked about background information, general life and educational experiences to provide context and history in a developing narrative. I gave each interviewee the opportunity to review their interview transcript in order to ensure that meanings had not been misinterpreted or misunderstood. Participants were advised that they had two weeks to review the transcript and or it would be presumed that they approved. This technique of 'member checking or participant validation' helps to guard against researcher bias, and also encourages the 'co-construction of knowledge' as the respondent has the opportunity to participate by changing or adding to their accounts (Birt et al., 2016, p. 1802). As a way of improving rigour and enhancing trustworthiness in research, member checking is one of the validation methods used in qualitative research (Lincoln & Guba, 2015; Doyle, 2007). The interviews were transcribed, anonymised and coded for analysis (See 4.5 Data analysis; interview questions are located in Appendix H).

3.4.4 Video-recorded lessons

Video-based fieldwork was used to record and analyse piano lessons in this research. Jewitt (2012) states that 'just as becoming a researcher requires learning how to undertake 'observational research' even though that person can 'see', it is necessary to consider how
to use video effectively for research purposes even though a person may be able to use a video camera' (p. 2). In this section, the benefits and disadvantages of video recording are discussed, as well as considerations for its effective use as a research tool. Video recording has benefits and constraints. It has the advantage of preserving data that can be analysed later: verbal and non-verbal communication, the materials used in the lesson and the teaching styles employed (Marsh & Mitchell, 2014). Another advantage is that interactions might be understood in the natural context of the piano lesson. In addition, the video recording is a tool which can be edited and shared with others, allowing them to enter into the experience as it occurred. It allows the researcher to review the experience so that fresh insights might be obtained and enables reflective observations and the development of new insights after multiple viewings (Marsh & Mitchell, 2014).

The disadvantages include the significant amount of time required to record, view, edit and analyse the video recording. However, this limitation may be overcome by refining and using focused data that has been sifted through cycles of research (Jewitt, 2012); for example, pertaining to specific themes. A video recording is also somewhat restricted due to what it shows and what it does not show. As a lens can only show one angle, there might be elements which are not captured in the recordings. Goldman (2014) discusses four criteria for addressing issues of partiality. There must be a sense of presenting the whole situation in detail 'wholeness/particularity', a feeling of 'being there and being with' the participants in the video, events must be shown in an honest representation of 'chronological verisimilitude' and the video needs to acknowledge the 'perspectivity' of the person responsible for filming it (Goldman, 2014, p. 32). By recording as many lessons as possible with my students, I aimed to avoid a fractured picture of their experience but also to acclimatise them (and myself) to the feeling of being filmed. This might allow us both to feel less self-conscious than if we were simply having 'special' filmed lessons which occurred infrequently.

Considering the issue of reliability, I asked students to share their observations during lessons as to specific strategies we had used and to give their feedback. This provided an important opportunity to allow the student's voice to be heard in the analysis and to consider the meaning that they may have drawn from events and situations in the natural setting of the lesson context. I used a laptop or a phone on a fixed stand for in-person

lessons and I used Zoom cloud recording for online lessons. Following the lessons, I downloaded the lesson to a secure password-protected Google Drive folder to maintain safe storage. This method was relatively unobtrusive and was utilised only after discussion with the students and their parents in advance and after written consent was obtained. Data analysis of the videos is described in greater detail in section 3.5.

My goal was to have the participants and the piano keyboard central in the video. I arranged the camera so that body language could be easily observed. I turned the camera on after we greeted each other, and turned the camera off in order to allow us to have a few moments without being recorded at the end. Access to participants was negotiated by giving them an information sheet and consent form well before the lessons were due to be recorded. They had the opportunity to ask questions and discuss any issues which concerned them before they gave their consent to video recording. As discussed in section 3.7, parents or guardians of minor students gave their permission following the provision of an information sheet and an opportunity to ask questions about the research. The videos were stored on password-protected programs on Google Drive and the University of York Filestore. As the researcher and teacher, I was responsible for recording the lessons. To manage the data, I created a log of significant moments (video and timestamp) to summarise the narrative and context of the student's development and my growth as a teacher (as evidenced in reflective statements in this thesis).

3.4.5 Reflective journal

An important aspect of the cycle of action research is reflection. A journal may provide not only the opportunity for researchers to develop self-awareness but also qualitative data to be used within action research. Journals may also enable teacher-researchers to develop self-awareness (Phelps, 2005) and may serve as a bridge between theory and practice enabling a documentation of improved learning (Dyment & O'Connell, 2011). Bashan and Holsblat (2017) maintain that reflective journals are 'valuable to student teachers for developing metacognitive abilities and for promoting their self-orientation and responsibility for the processes of their personal and collaborative learning' (p. 2). The following factors were considered in the reflective journal: considerations of motivations for the behaviour of myself and the students, interpersonal dynamics, new questions for

teaching and research and how they might be implemented into plans of future action. New connections and links were made when examining and reflecting on the evidence collected. Schön (1992) described how practitioners:

...frame the problem of the situation, they determine the features to which they will attend, the order they will attempt to impose on the situation, the directions in which they will try to change it. In this process, they identify both the ends to be sought and the means to be employed (p. 165).

The process of reflection enabled a deeper processing of events and the opportunity to make connections in order to build explanations or theories.

3.4.6 Evaluative focus group

Focus groups may be a beneficial way of considering the transferability of research (Lincoln & Guba, 1986) and may consist of an expert panel who bring insights to the theories produced by the research in an evaluative way (Chioncel et al., 2003). A focus group is a small group of participants chosen by the facilitator, who moderates the discussion, and the purpose is to examine aspects of research and evaluate them (Stewart et al., 2007). This might enable feedback which adds additional insights into how findings might be utilised or disseminated. There are some benefits and drawbacks to using a focus group. The advantages include the openness and interaction with participants which enables the collection of valuable data and allows new ideas to emerge (Stewart et al., 2007). However, some of the challenges include participant recruitment and the dynamic of the group which may limit an open and authentic discussion from developing (Chioncel et al., 2003).

In the evaluative focus group for this research, participants were identified for their expertise in inclusive music education. The participants were sent information and consent forms in advance of the meeting; once those were returned, they were sent a brief which identified the research aims and questions. At the meeting, participants were presented with some of the findings from the research and the risk-resilience model. Following this presentation, they were asked to evaluate the model and to consider its relevance to inclusive music education in the UK and how these findings might be disseminated, or

transferred to other settings (for example, music classrooms or music hub environments). The findings of this focus group are discussed in Chapter 11.8.

3.5 Data analysis

Data analysis provides the means of organising and interpreting the collected data, in order to understand meaning from the data. The process for this in an action research context is described by Coghlan and Brydon-Miller (2014) as 'mining', 'interrogating' and 'interpreting' the data to report the 'contribution to knowledge and theory' (p. 2).

Another consideration in action research relates to the question of when it is appropriate to begin data analysis. In conventional research forms, data is compiled and analysed at the end of the collection period. According to Burns (2010), action research data must be analysed from the start of the research project. In Kemmis and McTaggart's (1988) spiral model, data is examined throughout the process. The data was analysed by referring back to the research questions and asking whether or not they were being answered by the data that was collected. If not, the challenge was to consider how to change the data collection to collect the evidence needed. This is described as 'iterative sampling' and 'is valuable for its sensitivity to the richness and variability of data and for ensuring data address the study's objectives' (Mills et al., 2010, p. 504).

The starting point was the literature review, the student questionnaire and the initial lessons with my students. From the literature review, I understood the complexities of dyslexia, emerging dyslexia research, recommended teaching approaches and strategies, the importance of the identity-construct of individuals with dyslexia and parental involvement, as well as developing a comprehensive understanding of the music and dyslexia research and suggested strategies for teaching music to dyslexic students. This enabled me to identify gaps in the literature and to formulate the research questions for this study. It also underpinned my choice of the use of Universal Design for Learning (CAST, 2018) as a framework to support inclusive lesson planning and delivery and informed my choice of strategies for initial lessons.

The video data were analysed in the following way. After recording the video, fresh observations about the lesson were recorded in my reflective journal. I used those as points of comparison when I reviewed the video again and included timestamps for critical incidents, students' reactions, opinions and body language and I identified strategies and methods which worked well, ways in which the lesson might be improved and how I might take action to make those changes (Marsh & Mitchell, 2014). I viewed the lesson again in a database which enabled me to make links from the timestamps enabling me to crossreference specific materials used or keywords to identify commonalities or differences between lessons. The lessons were coded by date and labelled so that I could quickly locate them. This was helpful in correlating reflective observations with specific videos and the development of complex themes emerging as the videos were watched a number of times.

Results from the student questionnaire enabled me to see that further enquiry was needed in the area of music exams and I extended my knowledge of this by including questions for teachers and students in the interview schedules. I wanted to enhance my understanding of teachers' and students' views. Initial teacher interviews (Participants 1-11) took place before the Covid-19 pandemic. I reviewed the data that I gathered following these initial interviews, the initial cycle of teaching and literature review; and this information informed further teaching cycles and future interviews and lessons. For example, I familiarised myself with the material and identified emerging themes, then selected materials and strategies based on the development of my understanding in those areas. I carefully recalled situations, contexts, behaviours and participants from rich and detailed descriptions; this enabled the creation of more informed action plans in the following cycles. As previously mentioned, an inductive analysis means that I expected themes to emerge during or after I collected the data, being open to see what the data revealed and using robust processes for data analysis in which I considered aspects concerning my own potential bias as a teacher/researcher (see Chapter 4). In this way, I hoped to prevent pre-formed assumptions from manipulating the direction of the analysis.

The next step after critical reflection in the data analysis process was to identify emerging patterns to make sense of them and to theorise how to inform future action. My standards of judgment are my deep beliefs about the values I consider important. Articulating those standards of judgement enables me to draw meaning by selecting data which evidences

how I have demonstrated my values in my teaching (See 3.5.1), but also to scrutinise data where I did not demonstrate my values, as these moments further informed my reflection and analysis (See Reflective Statements 1-7). I value the student's voice, collaboration, flexibility and spontaneity in the learning process. In addition, I value the facilitation of strengths-based learning, inclusivity and equitable learning environments.

3.5.1 Thematic analysis

In qualitative research, thematic analysis is a means of recognising and analysing patterns in the data. Braun and Clarke (2006) state that an advantage of using thematic analysis is its inherent 'flexibility' whilst the challenge lies in maintaining a clear and transparent process of analysing the themes for the purposes of trustworthiness (p. 78). Another benefit to using thematic analysis is the possibility of finding perspectives that were not anticipated (King, 2004). Braun and Clark (2006) identify six phases in thematic analysis: 'familiarising yourself with the data, generating initial codes, searching for themes, reviewing themes, defining and naming themes and producing the report' (p. 87). Nowell et al. (2017) state that this is an 'iterative and reflective process that develops over time and involves a constant moving back and forward between phases' (p. 4). This is particularly true for action research as each cycle reveals new data that needs to be analysed and compared with the previous analysis. This is then applied to teaching and re-assessed. An example of a section of coded material which was analysed after the first eleven interviews is available in the examiner's folder. There are examples of this type of re-assessment in my Reflective Statements (see Reflective Statement 6). I acknowledge my role as a researcher in collecting the data and in organising the data in ways which demonstrate how experiences, events and perspectives operate in a socially constructed environment. I aim to answer the research questions by interpreting the experiences of teachers, dyslexic students and parents and analysing this data, along with my experiences of teaching dyslexia students, to inform my teaching practice and support ongoing pedagogical practice. This theoretical approach is held in tension with an inductive approach of creating initial codes and allowing themes to emerge from the data.

3.5.2 Steps of thematic analysis

By immersing myself in the data collected, I was able to identify descriptive codes and from these allow themes and subthemes to emerge (Coghlan & Brydon-Miller, 2014). The first step was to transcribe transcripts, read and re-read all text data, or in the case of the recorded lessons review the recording, to ensure that I had a good understanding of the material and context. Through this iterative process, themes were refined continually in order to validate their importance to the research. As described in section 3.5, videos of lessons were reviewed and reflective observations were recorded, including initial impressions and later reflections with specific action points as well. I sought to identify themes which captured the nuances within the data and tried to avoid overly simplifying complex issues. I then sought to bring the data together and to review and refine the themes that were emerging.

I also used MAXQDA software to compare frequency amongst themes. I do recognise that in qualitative analysis, frequency is not the only indicator of the importance of a theme, and I used my 'researcher judgement' to make the final selection (Braun & Clarke, 2006, p. 82). It was important to consider areas of contradiction within my findings and to consider how these might be resolved. Due to the diversity of a condition like dyslexia, it was important not to make generalisations, but instead to consider how findings related to the aim of improving my teaching practice by acknowledging variation of the condition and diversity in the context of individual circumstances. I continued to return to the raw data as I established a hierarchical structure and connections between the codes. These subsets also relate to the questions asked in the interviews (Section 4.4.3). I determined which themes had the greatest number of subsets; I also considered how to make use of categories with limited data.

The next step was to write a detailed analysis (See Chapters 6-8) based on the refined themes. In the reporting of my research narrative, I was careful to delineate the three following categories: 'the events, the voices of the participants and the interpretations of the researcher' (Coghlan & Brydon-Miller, 2014, p. 6) to allow readers to form their own objective analysis.

In addition, as mentioned in Chapter 1, I wrote a series of reflective statements to be included in my reporting of the research at specific points in the process. The aim of these reflective statements was to demonstrate transparency in the incorporation of analysis of the research into my own teaching practice.

3.6 Establishing validity

Whilst validity from a positivist perspective comes from that of a value on 'truth', standards for the interpretivist and pragmatic perspective extend to terms such as 'credibility and workability' and 'outcomes that go beyond knowledge generation' (Herr & Anderson, 2005, p. 59). Lincoln and Guba (1986) refer to this as trustworthiness. According to McNiff and Whitehead (2006), validity in action research relates to the 'trustworthiness of a claim' to knowledge whilst 'legitimacy' is to do with 'getting the claim accepted in the public domain' (p. 155). One of the aims of action research is to give evidence for development in my teaching practice.

Wood (2010) reiterated the need to demonstrate an improved practice, but 'rather than being a "recipe" of how to do things, it is a sharing of what worked for a particular researcher working in a particular context' whilst also exploring applications to wider contexts (p.115). These contexts might be applied to show validity through 'innovative insights into practice' which might be relevant for 'other practitioners to improve their performance' and may be used by the researcher to demonstrate 'professional development and innovation' as well as to detail their roles as practitioner-researcher by describing their accounts with 'a very high level of professional creativity, sensitivity and responsibility' (Winter et al., 2000, p. 32). Part of the process of validation involves producing authentic accounts; this is supported by internal critique taking place through regular supervisory and thesis advisory meetings in addition to self-reflection, as detailed in section 4.4.5.

Dadds (2008) refers to an empathetic validity in practitioner research which may refer not only to the development of compassion and understanding between the teacher, the student and their family, but also to promote change through dissemination in extended spheres of influence. Related specifically to my research, the development of empathic awareness is discussed in greater detail in Reflective Statement 7, and my presentations for

conferences and workshops, in addition to the resources presented in Chapter 11, show my commitment to enhancing empathy for students with dyslexia and their families to a wider audience.

Herr and Anderson (2005) refer to five validity criteria which are linked to the goals of action research. For example, the goal of generating new knowledge corresponds to the criteria of 'dialogic and process validity', whilst a focus on 'action-oriented outcomes is related to 'outcome validity' (Herr & Anderson, 2005, p. 54). The development of learning for both the researcher and the participants corresponds with 'catalytic validity' and findings that are relevant to the setting being studied correspond to 'democratic validity' criteria; finally, the goal of a 'sound and appropriate research methodology' corresponds to 'process validity' (p. 54). In my research, conducting the research in the natural setting of the lessons, gathering data from multiple sources and using processes that meant I captured data immediately, in addition to participant validation of accounts, were means by which I enhanced the validity (Dadds, 2008) in Reflective Statement Seven. This mixture of validity measures is designed to enhance robust scholarship in action research.

3.7 Ethical issues

There are many ethical considerations to be made in action research. Human participants require particular considerations which are set out in the British Educational Research Association's guidelines (BERA, 2018). The following ethical factors had to be considered in my application for ethical approval from the Arts and Humanities Ethics Committee (AHEC) at the University of York: potential conflicts of interest, methods of data collection, participants, informed consent, vulnerabilities, risk analysis, data protection, data storage or transfer, and anonymity. In this research, participant minors and dyslexic students were subject to more stringent guidelines from AHEC to protect their vulnerabilities. In this research, the term 'participants' refers not only to students participating in the teaching aspect of the action research but also to parents, teachers, dyslexic students and dyslexia experts who were interviewed or participated by completing a questionnaire. This also includes stakeholders who took part in the focus group.

As a piano teacher, I have an Enhanced Disclosure and Barring (DBS) certificate¹⁶ which is proof that I do not have a criminal record in the UK. Additionally, I am registered with the General Data Protection Register¹⁷ as I have records of some student and family details on file. In my teaching practice, guardians of minor students are required to fill in a form which declares that I do not obtain or store any information which I do not use for a specific purpose; I have maintained a rigorous and responsible attitude to my research as well as to these aspects of my teaching practice. I hold memberships with ISM¹⁸ (International Society for Musicians) and EPTA¹⁹ (European Piano Teacher's Association) and I follow their regulations for safeguarding students and guidelines for best practices in teaching. An example of that can be seen in the EPTA risk assessments that were done before resuming face-to-face teaching following the initial lockdown period in the UK as the result of the Covid-19 global pandemic (see Appendix L).

As stated above, one of the initial steps of my research project was to obtain ethics approval from the University of York's Arts and Humanities ethics committee. 'Informed consent, privacy and confidentiality, protection from harm' are the three tenets of ethical research (Norton, 2009, p.181). In order to give informed consent, participants must have knowledge of the research project, and the potential risks and benefits. The researcher must make it clear that consent can be withdrawn at any time and that confidentiality is assured. Guardians of young participants must also give their informed consent.

Interview (See Appendix C) and focus group participants and piano lesson participants (Appendix B) received a relevant information sheet (available in Appendices A-D) and consent was obtained by their written permission (in the case of minors, written permission was obtained from the parent/guardian). It was made clear in each case that consent was entirely voluntary. In my teaching practice, parents/guardians were given the relevant information/consent forms and were given time to read them, consider the information and

¹⁶ https://www.gov.uk/government/organisations/disclosure-and-barringservice/about#:~:text=An%20Enhanced%20DBS%20check%20is,working%20in%20the%20Gambling%20Comm ission.

¹⁷ https://ico.org.uk/for-organisations/uk-gdpr-guidance-and-resources/data-protection-principles/a-guide-to-the-data-protection-principles/

¹⁸ https://www.ism.org/

¹⁹ https://epta-uk.org/

ask any questions. It was made clear that a decision to participate did not need to be made immediately. If they gave permission, the student was approached with an age-appropriate information sheet and consent form. Students were given time to discuss the research with their parents and no pressure was placed on them to come to a decision or to agree to participate. I needed to ensure I was not creating what Norton (2009) describes as a 'social penalty' or pressure to take part. My ethics applications are included in the examiner folder. Likewise, no pressure was placed on the interview and focus group potential participants to take part but they were given the information sheet and consent form and given an opportunity to ask me questions prior to the meeting.

Concerning the piano pupil participants and their parents, information should be accurate regarding both risks and benefits, as downplaying the risks or overemphasising the benefits might lead to the participant feeling a lack of trust in the researcher (BERA, 2018). The students were given an information sheet that was relevant to their age and understanding (Appendix M). Potential risks included personal safety, risk of accusation of harm/impropriety or potential conflicts of interest (BERA, 2018). By maintaining ethical and safe boundaries in the teaching setting these risks were greatly reduced. These good practice guidelines were continually informed by my professional memberships in the Incorporated Society for Musicians and the European Piano Teacher's Association. In addition, content from previous lectures that I attended on safeguarding and ethics as part of the University of York MA in Music Education: Instrumental and Vocal Teaching informed me of foundational principles of safeguarding and ethical issues involved in instrumental teaching.

By maintaining openness and being transparent about the use of personal information, guaranteeing anonymity, safely storing data and the reassurance of trust between the investigator and other participants, these risks were mitigated (BERA, 2018). The risk of adverse publicity to the University of York was minimised by the information sheets provided to participants which detailed the measures taken to protect their data and anonymity and through which it is made clear that consent is completely voluntary and may be withdrawn at any point in the research (BERA, 2018). It must be clear to the participant how long the data will be stored, how it will be stored, who has access to it and where it will be kept (BERA, 2018). Data protection was secured by the use of pseudonyms and safe

storage in password-protected files. As previously noted, data for this research was stored on Google Drive and University of York Filestore, and will be retained for ten years in accordance with the University of York protocol. The data consists of text documents, video recordings and audio recordings. Teachers were given access to their own interview transcript, and were informed that my supervisor and the thesis examiners would have access to the data for guidance and assessment purposes. This was part of the informative phase of action research, which was intended to enhance the validity and reliability of the results.

Consent forms were edited for storage purposes so that the participants' names were not viewable. In the research data, pseudonymised names prevented the identification of the participants. The video material was edited using Adobe Premiere Pro²⁰ and Capcut²¹ to blur identifying characteristics of participants using facial recognition anonymising software to preserve their anonymity. Discussion of the research with parents and students was done with sensitivity to minimise any risk of embarrassment or anxiety. Utmost care was taken to extend empathy and understanding of each individual case.

The protection of participants, in interviews and in teaching, from harm referred mainly to self-esteem or embarrassment. I was aware that some students might be uncomfortable talking about their dyslexia diagnosis, or about possible learning challenges which might have revealed difficulties indicative of dyslexia. In my piano teaching studio, I was careful to establish a safe learning environment for the students. If participants were uncomfortable being filmed in the context of a lesson, an open discussion of the potential benefits of the research, in conjunction with the knowledge that consent could be withdrawn at any time, ameliorated discomfort, though I was of course prepared to stop recording at any point. As the teacher, I have a duty of care to ensure that students are safe in the learning environment. This includes thinking about how to minimise any risks to them and also to myself and my reputation.

Obtaining informed consent from participants without any sense of coercion or consequence if refused and maintaining confidentiality were two of the ways in which these

²⁰ https://www.adobe.com/uk/

²¹ https://www.capcut.com/

risks were minimised. Students were made aware that they were able to withdraw their consent at any time for any question or activity with which they did not feel comfortable. During interviews, it was explained that they did not have to answer any question they chose not to, did not have to offer an explanation for this and would not face any consequences. Clarification of how anonymity was preserved included discussions of video editing to prevent recognition and the use of pseudonyms in writing. In addition, the benefits of the research were explicitly stated, but participants also understood that they could withdraw their consent at any time up until the submission of the thesis.

Benefits to participants included piano lessons taught in a collaborative way which incorporated their needs and strengths, were guided by their goals and aspirations and considered findings from the most recent dyslexia research. Participants also benefited from the knowledge that this research may provide future contributions to the field of music and dyslexia.

3.8 Conclusion

McTaggart (1994) states: 'Action research is a form of self-reflective enquiry undertaken by participants in social situations in order to improve the rationality, justice, coherence and satisfactoriness of (a) their own social practices, (b) their understanding of these practices, and (c) the institutions, programmes and ultimately the society in which these practices are carried out' (p. 317). In Chapter 3, the theoretical framework of social constructivism which underpins this research through an interpretivist and pragmatic approach in cycles of action research has been examined. Action research is practitioner-research which takes place in social contexts. Villacañas de Castro and Banegas (2020) describe how 'practitioners can learn from academic forms of scientific research, but above all they must learn from (and transform) each other and themselves' (p. 10). The benefits and limitations of action research and the rationale for using action research methodology have been considered. Qualitative methods of data collection and data analysis, and measures for establishing validity and ethical issues involved in the research conclude this chapter.

Chapter 4 THE ROLE OF THE RESEARCHER

4.1 The role of the researcher

In this chapter, the role of the researcher in the research process will be examined; this includes the challenges, benefits and ethical issues, going beyond the considerations provided in Chapter 3. Literature on qualitative research has emphasised the importance of establishing the positioning of the researcher in the research process (Cresswell, 2007; Robson, 2011) with Robson (2011) describing the role of the researcher as an 'instrument' (p. 133). Consideration of the researcher's role reflects a growing societal 'consciousness of situational identities' and 'the perception of relative power' (Angrosino, 2005, p. 734), particularly when this involves participants with vulnerabilities. In my research, participants included minors and those with Specific learning differences (SpLD) which underscored the importance of carefully considering ethical concerns and potential power imbalances in order to avoid any negative effects from participation.

In undertaking this research, I acknowledge that personal experiences in my piano teaching practice influenced my decision to research music education and dyslexia in the instrumental lesson context. During the research process, I positioned myself as an insider-researcher, a teacher, an observer and a facilitator of both the teaching practice and research interviews. Further considerations of these positions are described in the next sections.

4.2 Consideration of being both an insider as well as an outsider researcher

Being an insider and an outsider may offer differing perspectives and advantages whilst also entailing specific risks. The concept of 'insider' was identified as a structural concept of knowledge production (Merton, 1972) involving unique and privileged knowledge of a group, whereas an 'outsider', as a non-member of the group, does not have this access (p. 36). Others suggest that these positions exist across a spectrum and depend on boundaries that might change, and that the researcher tends to fluctuate between insider-outsider throughout the course of the research (Griffith, 1998). I was situated as an outsider due to being non-dyslexic and in section 4.4, some of the preconceptions and potential bias that I

disclosed before the start of the research are considered. I was positioned in the research process as an insider as the teacher within the lesson environment and in the instrumental pedagogy context. As an insider I might become too close to the situation to be objective, and, as the outsider I might miss the nuances and details that an insider would capture. I participated 'inside' the research as the teacher, and, as a 'friendly outsider' (Greenwood & Levin, 2007, p. 124), I facilitated data collection and analysis and produced reports of my research. In the next sections, the benefits and challenges of my role as researcher will be explored.

4.3 Advantages to being an insider-researcher

Three key advantages of being an insider-researcher include a better understanding of the setting, involvement in interactions that would naturally occur and a closeness to the setting (Bonner & Tolhurst, 2002). Piano pupils were accustomed to my role as the teacher, and this was a natural environment for them, and for myself due to many years of teaching experience. I had access to the pupils' lived experiences as they shared them with me. There were clear expectations and patterns of work during lessons; I had developed an understanding of the needs of the individual students.

My teaching took place in a one-to-one setting either in person or online (due to the constraints of the Covid-19 pandemic) in which students often shared experiences from their lives that were unrelated to music learning. This insider knowledge helped me to empathise with them and their challenges in navigating life, and musical instrument learning, with dyslexia. Students and their parents trusted me as the teacher in the relationship. Palmer (2006) suggests that insider-researchers are less likely to cause distress or disruption to participants. Because of this rapport, I was familiar with them and their families, including sensitive information relating to their learning and in some cases, family lives (Sike & Potts, 2008).

These relationships were built over a period of years in which camaraderie and trust had been established; an outsider would not have had the same depth of understanding and insight into the student's lives. This enabled me to immerse myself in their worlds and to collect rich data. Geiger (2016) posits that 'close relationship and ongoing long-term interaction with the research participants' may be an indication that the conclusions

reached are 'authentic and reflect the participants' perspective' (p. 17). In the next section, mitigation of the challenges based on the researcher's positioning in the research will be discussed.

4.4 Challenges for the insider-researcher

Robson (2011) accentuates the necessity of being a quality 'investigator' with experience in the following skills: 'question asking, good listening, adaptiveness and flexibility, grasp of the issues and lack of bias' (pp. 133-134). The relationship between the researcher and participants is an important aspect of action research. I was operating as both instrumental teacher and researcher and had to navigate the tension of those two roles in a productive way. Maiter et al. (2008) elaborate on the importance of reciprocity in this relationship, defining it as 'an ongoing process of exchange with the purpose of creating and maintaining equality between parties' (p. 305). The challenge of building reciprocity with participants, in addition to ongoing personal reflection in my teaching practice, was an important part in helping to uncover and understand any biases I may have. Trust and confidence must be carefully nurtured between participants. Having taught piano lessons for several years, experience has helped me to develop awareness of how I operate in this role and my relationships with pupils, and with their parents are foundational to the research process.

4.4.1 Conflict of dual roles

The qualitative researcher is in a paradoxical situation needing 'to be acutely tuned-in to the experiences and meaning systems of others' and yet to be aware of how 'biases and preconceptions may be influencing what one is trying to understand' (Maykut & Morehouse, 1994, p. 123). These biases might be overcome by making accounts of the research open to participants and dissemination through research presentations at conferences, to BDA, and through the use of a focus group consisting of experts in the field of dyslexia at the final stages of the research. By using the phrase 'working the dialectic' Cochran-Smith and Lytle (2009) refer to the 'reciprocal, recursive, and symbiotic relationships of research and practice' with the boundaries between teacher and researcher obscured; they describe this method of producing knowledge as 'improvising dance moves', rather than 'climbing stairs' (p. 43). McNiff (2016) concurs, referring to action research as a

'dialogue', not a specific technique; the researcher is responsible for reflecting, making choices and being held accountable for the outcome of those choices (p. 40).

Others consider that trying to balance these dual roles may lead to conflict, particularly in the data analysis process (Brannick & Coughlin, 2007). Dwyer and Buckle (2009) resist a focus on duality by stating that 'the core ingredient is not insider or outsider status but an ability to be open, authentic, honest, deeply interested in the experience of one's research participants, and committed to accurately and adequately representing their experience' (p. 59). In regard to my insider status as teacher, I found that at times I was aware of the recording process during lessons and was possibly less spontaneous than I might be in a normal lesson setting. Sometimes I found there to be an internal conflict when I wanted to try a new strategy or method during the lesson, but could see that the student was overloaded, and it felt as though by accepting this, I was 'limiting' the research that might be done. However, as stated earlier (see Chapter 3, section 3.7), I would always prioritise the quality of the pupil's experience and their comfort within this setting. Flexibility was needed to maintain a good relationship with the student and adapt to their needs.

4.4.2 Navigating power imbalances

In addition to reciprocal trust, there were the challenges of navigating power imbalances between teacher and student, interviewee and interviewer, adult and child, parent (employer) and teacher (employee). Geiger (2016) indicates that the researcher moves from the initiator of the project to someone who is dependent on the contributions of the participants for the success of the project; this 'transfer of ownership' is important for effective data and interpretation (p. 14). Interview participants held the power through the extent and content of what they shared, but the power to analyse and interpret that information was held by the researcher. In relation to the parents of pupils in my research, different factors came into play. Parental support, even when parents do not have extensive knowledge of music, can affect students' practising and development (Davidson et al., 1995).

4.4.3 Overcoming bias

Each researcher seeks to account for and overcome their own personal bias in their research process (Herr & Anderson, 2005). In my research, critical friends, academic staff, peer

students, other instrumental teachers and dyslexia experts offered a balance to my role as the insider-researcher. As outsiders, without emotional connection to the teaching context, they were able to be aware of how my personal bias might affect the research. They provided feedback on lessons and written material, offered advice and signposted to specific resources. In addition, developing narratives of the research helped me to look more objectively at situations where I have insider knowledge. This may permit the selfreflective awareness of areas where my preconceived thoughts and ideas hinder me from perceiving situations accurately. I acknowledge, for example, that I view music reading through musically literate lenses, which may make it difficult to understand and empathise with my dyslexic students.

Luft and Ingham (1955) developed a framework of understanding self-awareness in interpersonal relationships: the Johari Window. The aim of the model is to enhance transparency and openness in relationships and consists of four areas: arena, blind spot, hidden and unknown. By increasing the open areas (i.e. the 'arena') in relationships through self-analysis, seeking feedback from others and collaboration with participants, the blind, hidden and unknown areas of the relationship might be reduced. In my piano teaching practice, there is a certain amount of information which is known and shared equally with the student; this represents the 'arena'. However, a student's perspectives of repertoire and lessons, of their capabilities and of me as their teacher may, for various reasons, be hidden to me. Building a foundation of trust and acceptance may create an environment where the student will feel liberty to reflect on and offer explanations for barriers that might occur in the learning process.

For many dyslexic students, previous educational experiences have been negative and any new learning environment may cause a certain amount of anxiety and stress; this might cause students to try to disguise their difficulties (Riddick, 2012). I sought to mitigate this in my teaching practice by collaborating with the individual students and offering as much of the decision-making to the students as possible: choice of repertoire, order of lesson activities and choice of teaching methods. Sharing the power and ownership of the direction of the lessons was intended to address any teacher-student power imbalances. I was committed to a mentor-friend relationship (Lehman et al., 2007) with the student for the purpose of promoting learner autonomy, reflection and independent thinking. I sought to

view them as 'co-researchers' and not as 'subjects' by honouring their voices in the lessons (Reynolds, 2014, p. 349). Additionally, I tried to demonstrate sensitivity and empathy in our relationship so that if they were overloaded, they knew they could ask for a break, for further explanation or a change of activity without any negative reaction or defensiveness on my part.

4.4.4 Identifying preconceptions and beliefs

Flexibility allows a researcher to acknowledge that new questions may need to be asked and to allow the cycles of action and reflection to adapt when the need arises. Teaching and research rely on many variables, and degrees of flexibility are needed to navigate these variables appropriately. By regular reflection and by maintaining openness to change, these challenges may be negotiated (Kemmis & McTaggart, 2005). Atkinson (1994) points out that in real-life action research, the cycles are not always 'neat spirals', that teachers are prone to making decisions intuitively instead of in a planned way and that documenting research requires sustained effort (p. 399). I developed a list of guidelines in order to mitigate any preconceptions or bias which I might have; this 'bracketing' process (Tufford & Newman, 2012, p. 83) led to the following principles.

- It is important that I do not make assumptions about each person's experience as a dyslexic individual based on previous participants' experiences or what I have read in academic or pedagogical literature.
- Remaining impartial when analysing my own piano teaching might be difficult, but it is important to be aware of my own potential emotional attachment and egoistic attitudes in discerning how the data should be reflected upon and interpreted.
- 3. I must remember that as I am non-dyslexic, I must be sensitive not to take certain abilities (for example: reading, calculating numbers, sight-reading music, ease in writing) for granted in dyslexic participants and that I should seek as much information as possible from them or their parents in advance in order to plan lessons, aiming to avoid any humiliation or sense of embarrassment.

- It is imperative that I maintain my roles as a teacher and researcher and understand that these roles do not qualify me as an assessor of dyslexia or other learning differences.
- 5. I will make use of outside relationships such as those with my university supervisor and piano teaching colleagues along with reflective journals in order to limit bias and enhance the research process.

In this process, the researcher identifies preconceptions and beliefs about the research participants and then 'suspends or brackets them' prior to the research study so that they understand more objectively how they might affect their interpretation of the results (Cresswell & Miller, 2000, p. 127). Bracketing is meant to be a process of 'self-discovery' for the researcher throughout the research process and there is no uniform definition of how it should be achieved; this flexibility is beneficial for allowing the researcher to implement bracketing with flexibility in accordance with their own research needs (Tufford & Newman, 2012, p. 93). I used the points above as a starting point but frequently returned to them to remind myself throughout my research in order to attempt to separate my preconceptions or bias from my research process. During participant interviews, particularly those with dyslexic participants, it was clear that each individual has experienced dyslexia in their own way and this confirmed the points made above in brackets (1, 3, 4). My reflective process and research reading continued to support my awareness of bracket 3. I reviewed these often during my teaching and data collection. Reviewing the lesson videos helped me to observe and reflect as an outsider, rather than as the teacher thinking and acting in the moment. As detailed above, I also sought feedback from others in my teaching and research in order to attempt to eliminate possible blind spots.

4.5 Ethical considerations of insider-research

In terms of the ethical considerations of making sure that there is no harm to participants, being an insider allowed me to develop sensitivity to each participant in order to avoid embarrassment or shame throughout the research process (Berger, 2015). Insider knowledge of individual personalities in my teaching practice enabled an advantage in this sense as each situation and student are unique. Conversely, I was also aware that this familiarity might cause me to develop assumptions about the students or create the feeling

that they could not disagree with or express negative opinions or beliefs about me, lesson structure or activities. Students might feel pressured to agree to participate in the research, perhaps even to please me. I sought to explain to each student the purpose of the research and to reiterate that there was no expectation or obligation to participate; additionally, as noted above, my own guidelines were frequent points of reference to remind me of my aims and priorities.

As detailed in Chapter 3, educational research guidance such as the British Educational Research Association (BERA, 2018) and the University of York ethics guidelines were followed as a means of protecting the participants and making sure that they knew their rights, before and during the research, and afterwards in terms of how I presented and stored data relating to them, and how I ensured their anonymity. Informed written consent was a significant part of the ethics process, as participants were assured anonymity, voluntary participation and were aware of how the research findings would be shared. I reassured students and parents that our relationship would not suffer or be changed if they decided not to participate. In conclusion, the role of insider-researcher requires consideration of a number of factors, both positive and negative. In order to mitigate the challenges, I sought to nurture openness with participants, to share the power in lessons through collaboration and a mentor-friend teaching style, to maintain flexibility, solicit feedback and consider prior assumptions or preconceived bias. I deployed these approaches on an ongoing basis, continuing to review them as the research progressed.

Chapter 5 FINDINGS FROM INITIAL SURVEY

5.1 Overview

Although some literature examines challenges and strategies for music dyslexic students (see Chapter 2), there is a lack of research relating to students' experiences with music exams. A questionnaire was conducted in order to understand the views and experiences of graded music performance and theory exams for dyslexic students. The online questionnaire was distributed from April 2019 and was open until December 2019 to cover the time spanning at least two exam periods. The information sheet and consent form are located in Appendix A, whilst the questions are included in Appendix E of this thesis. The participants were recruited through contact with organisations which have dyslexia and music overlap; for example, the British Dyslexia Association music committee mailing list, music teacher organisations and social media for music teachers interested in teaching dyslexic students.

At the beginning of the questionnaire, participants were asked to consent to the following:

- They have read and understood the following content and that as the guardian to a minor child they may assist them in completing this questionnaire;
- They understood that the purpose of this questionnaire is to better understand dyslexic students' perceptions of music exam access arrangements and reasonable adjustments;
- They understood that their responses were anonymous and would only be used for the purposes of this research and
- 4. They voluntarily consented to participate in this research study which they understood involved completing this questionnaire.

The questionnaire was designed so that it would not progress to the questions unless the participant gave consent to these statements.

5.2 Contextual information

The questionnaire consisted of open and closed questions and asked students/parents/guardians about the following:

- age
- diagnosis of dyslexia
- instruments played
- difficulties
- how they obtained information about the exams
- required documentation
- their views on the accessibility of exam board websites
- choices related to opting to take exams and selecting an exam board
- reasonable adjustments or accommodations used
- the perceived equity in the use of accommodations and recommendations for improvement

The questions were designed to be as accessible as possible for dyslexic students. Fourteen participants completed the questionnaire. The range of ages indicates that there was one participant five and under, one participant six to ten years old, five participants age eleven to eighteen and seven participants age nineteen or above. Of these participants, thirteen had been assessed with dyslexia or a specific learning difficulty. As mentioned in the consent form statements, it was made clear that parents might complete the questionnaire for their children.

Participants reported playing a wide range of instruments, usually more than one. Thirteen participants listed the piano as their primary instrument, whilst other instruments played included woodwind, string and brass instruments. Three students had vocal training. The range of experiences varied widely as some reported learning their instruments for two years whilst other participants had studied their instruments over decades. In describing their motivation to take music exams, building on existing skills and boosting confidence were the primary reasons given, with some evidence that stimulating a commitment to practise, demonstrating a level of competence to others and exploring repertoire were also motivating factors.

5.3 Student challenges

In describing their difficulties, participants were asked to identify which, if any, of the following options presented in Figure 5.1 applied to them.



Figure 5.1 Difficulties reported by dyslexic students

Poor working memory (11) and difficulties with reading (10) were the most common, but difficulties with attention (8), difficulties with writing (9) and sequencing (9) were also noted. Seven participants reported 'visual distraction', and five students reported difficulties with numbers. Six students reported difficulty with determining left from right. Participants also noted poor motor coordination (3) and poor hand independence (2). One participant responded with another characteristic, stating that they had 'difficulties understanding verbal instructions, especially when under pressure'.

5.4 Obtaining information about the exams

Participants reported that their primary means of obtaining information about access arrangements was through the exam board's website, with others reporting that their instrumental teacher, a parent or contact with the British Dyslexia Association provided assistance in understanding the guidelines. Three participants were not aware there were special arrangements for dyslexic students.

5.4.1 Accessibility of exam board's website

Five students felt their exam board's website was friendly for dyslexic readers, whilst three participants disagreed. Four participants replied 'I don't know' to the question on the ease of use of the exam board's website, which suggests that they may not feel confident in recognising what constitutes a dyslexia-friendly website.

Factors affecting the readability of the website were reported to include the font type and lack of customisable options, as well as font size, colour combinations or choices, spacing of text or images and content. The design and absence of a site map were not indicative of factors affecting their ability to access the website. They were also divided regarding whether or not they felt access arrangements were made clear to them by the website.

5.4.2 Documentation required

Seven participants had been asked by exam boards to provide supporting evidence to document their dyslexia, whilst three respondents indicated that documentation was not required. Six participants had provided evidence from an official assessment and one participant provided a letter from their school's SEN coordination officer. The evidence had to be provided at least four weeks before the exam with three participants reporting that they had to provide the evidence more than one month in advance.

5.5 Choosing to take exams

Eight participants reported taking both music theory and practical exams, with two participants taking only the practical exams. One participant had not taken any exams. Unsurprisingly, younger participants reported taking practical graded exams only. The majority of exams had been taken with ABRSM (Associated Board of the Royal Schools of Music), Trinity, Rockschool and LCM (London College of Music) less well represented. Eight participants reported choosing not to take music exams at some stage in their musiclearning journey; of these, four chose not to take music theory exams and four chose not to take practical exams. In describing the challenges with music theory, one participant 'struggled to represent the information I had learnt' while those who chose not to take a

practical exam did so because of sight-reading and memory issues. One student reported that they took a Grade Three practical exam in piano, but it 'took far too much time so [I] didn't do the next level'.

5.5.1 Adjustments/accommodations taken during exams

The following figure describes the adjustments used during practical exams.



Figure 5.2 Adjustments in practical exams

The participant who responded with 'Other' described 'using colour overlays and tinted glasses for sight reading' as their adjustment. Music theory exams required different accommodations, with Figure 7 detailing those used by participants:



Figure 5.3 Adjustments for music theory exams

Of those who chose 'Other', three reported that they chose not to take any adjustments and one reported using plain paper instead of manuscript paper, stating that it 'allowed me to work without the manuscript lines moving around and distracting' during the exam. However, this is not listed as a music exam adjustment but is something which each person taking the music theory exam might elect to do.

5.5.2 A question of equity

Participants were asked if they felt that these access arrangements were effective in giving them a fair opportunity in the exam; only three affirmed this for the music theory exam whilst four felt that the arrangements contributed to equity in the practical exam. Three participants noted that the extra time during sight-reading was helpful in terms of processing the information, and one participant noted that it was useful to have 'repetition of scales without penalty'. This is not listed as an adjustment but was reported by the participant as an open-ended response indicating that arrangements might be flexible depending on the exam board and the needs of the student.

5.5.3 Recommendations for improvement

When asked for recommendations to make music exams more accessible for dyslexic students, one participant responded that they would like more 'consideration around how

dyslexic students take aural tests in practical exams' whilst another suggested extra support with music theory would be helpful. Describing an uncomfortable experience with an examiner who stood behind them during the extra time, one participant suggested that 'examiners should be given more training in [their] manner' during exams.

5.6 Limitations, conclusion and recommendations for the future

This questionnaire was completed before the widespread use of online music theory exams during the Covid-19 pandemic; however, data related to how those exams are perceived by dyslexic students were collected during interviews with instrumental teachers and dyslexic music students (Chapters 6-8, Chapter 10).

5.6.1 Limitations

Whilst this study is limited to a small sample from the population of dyslexic music students (14 participants), the responses do enable an understanding of their experiences and give insight into ways in which the process might be improved in terms of accessibility and equity. The small number of respondents may also suggest that few take music exams.

5.6.2 Summary of key findings

The following are the key findings from the questionnaires:

- Challenges for dyslexic students include reading and writing text, working with numbers as well as a range of executive functions, such as working memory, attention and sequencing. There were indications of spatial awareness and coordination issues.
- Visual distractions were also mentioned by half of the participants.
- Dyslexic students are motivated to take graded exams, but they experience a number of barriers.
- The exam board website is often the first point of contact for students when searching for information about access arrangements and reasonable adjustments, but participants found that the accessibility of the website made it difficult to obtain the relevant information.

- Teachers need additional time and resources, such as training, in order to prepare students to utilise the reasonable adjustments effectively and to their advantage.
- The specific requirements of the exam might be a bottleneck to dyslexic students' progression.
- Examiners may need training to better understand how to support students with specific learning differences during exams.

In conclusion, findings suggest a number of recommendations to improve the accessibility and quality of the exam experience for dyslexic students. Exam board websites, as the gateway to information about reasonable adjustments and access arrangements, might need clearer signposting and would benefit from incorporating changes based on suggestions from users with special educational needs. Although some dyslexic students do take exams for a variety of reasons, there are also a number of barriers to their participation which the exam boards would do well to consider.

Chapter 6 FINDINGS FROM TEACHER INTERVIEWS: BARRIERS AND CHALLENGES

6.1 Introduction

The literature review (Chapter 2) of this thesis emphasises the complexities of dyslexia and reveals potential challenges experienced by music educators and their dyslexic students (Livingston et al., 2018; Oglethorpe, 2008; Rolka & Silverman, 2015). Some handbooks (Oglethorpe, 2008), narratives and case studies (Miles et al., 2008; Nelson & Hourigan, 2016) have established a foundation for understanding dyslexia and music learning. However, research is needed that further explores the teacher's perspective on how they gain expertise, identify dyslexic students, utilise strategies, develop individual student profiles, collaborate with the student and parents, select method books, employ technology and use music exams. This chapter and the following two chapters set out the findings from teacher interviews. Chapter 6 provides contextual information on the participants and then identifies barriers and challenges reported by teachers. In Chapter 7, teacher's perspectives of the areas of challenge for dyslexic students and corresponding teaching strategies are explored and Chapter 8 provides an analysis of teacher's views on the strengths of dyslexic students.

6.2 Contextual information

Participants who had experience teaching dyslexic students were sought to participate in this research. Twenty-seven instrumental or vocal teachers participated in semi-structured interviews. Further details regarding the processes of recruitment, data collection and analysis are found in Chapter 3.

6.2.1 Participants

The participants were instrumental and vocal teachers, and all had experience with instrumental teaching for dyslexic students. The interviews were conducted either face-to-face (before March 2020) or on Zoom and were recorded for the purpose of making a transcription. To ensure that the transcript represented an accurate written statement of the participants' experiences, participants were offered the opportunity to read and revise their transcripts. Some of the participants also taught in other educational contexts; these

are listed in Table 6.1 below. Twelve participants self-reported a neurodivergence during the interview, which they felt brought both advantages and challenges to their roles as teachers. Although this information was not solicited from participants, this may suggest that others among the remaining participant teachers may also be neurodivergent.

Of the 27 participants, two held dyslexia assessor or AMBDA (Associated member of the British Dyslexia Association) status, one held a postgraduate dyslexia specialist certification and two were working towards British Dyslexia Association (BDA) Level 5 certification, which is an accreditation in literacy, support and intervention with dyslexic students. Level 5 (BDA) enables those who complete it to achieve dyslexia specialist teacher and practitioner status, although this level does not allow individuals to carry out diagnostic assessments (BDA, 2020). Some of the participants were involved in their own research related to music and dyslexia.

Twenty of the participants were in the United Kingdom, four in the United States, two teachers were based in European countries, and one in Australia. Table 6.1, below, presents contextual details of the participants including the instruments taught (Column 1), educational contexts²² (Column 2), dyslexia-specific training or research background (Column 3), location (Column 4), and self-reported neurodivergence (SR-N) (Column 5).

No.	Instrument	Context	Training	Location	SR-N
1	Piano	IT, HE	Level 5 BDA	UK	
2	Piano	IT, SS, HE		UK	
3	Piano	IT		UK	
4	Piano	IT, PS	Level 7 BDA	UK	
5	Piano	IT, PS	Working to Level 5 BDA	UK	Dyslexia
6	Flute	IT, HE		UK	
7	Piano	IT, HE		UK	
8	Cello	IT, HE, PS	PG Cert.	USA	
9	Piano/ Organ	IT, SS		UK	
10	Piano	IT		UK	

²² Instrumental Teaching (IT), Higher Education (HE), Secondary school (SS) and Primary school (PS)

11	Voice	IT, HE		UK	
12	Brass	IT, PS	Working to Level 5 BDA	UK	Dyslexia
13	Piano, Guitar, Voice	IT, PS, SC		Australia	Dyslexia
14	Piano	IT		UK	
15	Voice	IT, HE, PS	Postgrad. research	Sweden	Dyslexia
16	Piano, Voice	IT, PS		UK	Dyslexia
17	Piano	IT		UK	Autism
18	Piano	IT		UK	Dyspraxia/
					Autistic
19	Piano, Voice, Clarinet	IT		Switzerland	Dyslexia,
					Dyscalculia
20	Piano, Voice	IT, SS, HE		USA	
20 21	Piano, Voice Piano	IT, SS, HE IT		USA	Dyslexia
20 21 22	Piano, Voice Piano Piano	IT, SS, HE IT IT, HE	Postgrad. research	USA USA	Dyslexia Dyslexia
20 21 22 23	Piano, Voice Piano Piano Woodwind	IT, SS, HE IT IT, HE IT, SS	Postgrad. research	USA USA USA UK	Dyslexia Dyslexia
20 21 22 23 24	Piano, Voice Piano Piano Woodwind Guitar, Percussion	IT, SS, HE IT IT, HE IT, SS IT, PS, SS	Postgrad. research	USA USA USA UK UK	Dyslexia Dyslexia Dyslexia
20 21 22 23 24 25	Piano, Voice Piano Piano Woodwind Guitar, Percussion Piano	IT, SS, HE IT IT, HE IT, SS IT, PS, SS IT, PS	Postgrad. research Research	USA USA USA UK UK UK	Dyslexia Dyslexia Dyslexia
20 21 22 23 24 25 26	Piano, Voice Piano Piano Woodwind Guitar, Percussion Piano Piano	IT, SS, HE IT IT, HE IT, SS IT, PS, SS IT, PS IT, HE	Postgrad. research Research	USA USA UK UK UK UK	Dyslexia Dyslexia Dyslexia Dyslexia
20 21 22 23 24 25 26	Piano, Voice Piano Piano Woodwind Guitar, Percussion Piano Piano	IT, SS, HE IT IT, HE IT, SS IT, PS, SS IT, PS IT, HE	Postgrad. research Research	USA USA UK UK UK UK	Dyslexia Dyslexia Dyslexia Dyslexia/ Autism

 Table 6.1 Contextual information about participants

6.2.2 Summary of the semi-structured interview questions

The interviews were semi-structured, and the interview schedule is included in Appendix F. Participants were asked to describe their experiences with dyslexic students including how they identified dyslexic students, challenges that were faced, strategies they applied and their use of technology or specific method books. They were asked to relate their observations related to strengths in dyslexic students and to share their views on parental involvement. Participants were asked about any resources which had informed them regarding their teaching of dyslexic students. Finally, participants were offered an opportunity to share any other information which they felt might be relevant. Chapter 6 focuses specifically on the barriers and challenges, whilst Chapters 7 and 8 address teachers' perspectives of student challenges, strategies and strengths.

6.3 Teacher perceptions of barriers in teaching dyslexic students

From these interviews, the following themes emerged under the larger subtext of challenges and barriers faced by teachers in their experiences teaching dyslexic students. In this chapter, findings relating to barriers include a lack of training (6.3.1.1), research (6.3.1.2), knowledge and resources (6.3.1.3) and poor communication with schools and parents (6.3.2). In the next section (6.4), a number of challenges were identified by teachers in terms of their identification of dyslexic students (6.4.1), recognition of when the student was learning (6.4.2) and when they were experiencing stress (8.4.3) as well as encouraging the student's metacognitive processes (6.4.5).

6.3.1 Barriers: Gaining expertise

An understanding of dyslexia has been steadily gaining traction since the 1960s, as described in the literature review of this thesis (see Chapter 2). Despite this, teachers reported difficulties in gaining expertise due to their perceived lack of research, training, knowledge and resources for dyslexia and instrumental music teaching. Inadequate communication with parents and schools relating to disclosure and support was also identified as a barrier. Some teachers felt that these factors led to misunderstandings, assumptions and poor strategy selection which then contributed to detrimental outcomes for dyslexic students.

6.3.1.1 Lack of research

Consistent with the literature review (Chapter 2) which highlighted the paucity of relevant literature, participants (P4, P10, P17, P20, P21, P22, P25, P25) struggled to find research on music and dyslexia. Welcoming the opportunity to participate in this research, a teacher stated: 'I'm glad to hear about the research as it's nice to have someone who has faced the same issue' (P21), which suggests that isolation may create difficulties in gaining knowledge of issues and good practice. One limiting factor is that the majority of information seems to be passed on informally, focusing on general experiences: 'There just weren't good examples for music; it was mostly anecdotal stuff that people have from their own lessons' (P22). Reiterating the importance of practical solutions, a piano teacher stated: 'It's okay to say these children have these problems, but what are we doing to help them navigate through it?' (P5). Additionally, online sources on the topic were not seen as up-to-date,

relevant or sufficiently pragmatic. These findings also emphasise research gaps, and potentially, the accessibility of the findings of research for instrumental teachers. They also indicate that due to the isolation instrumental teachers may face, they may explore or seek informal ways of learning from other teachers and opportunities to embed this knowledge into their teaching practices.

6.3.1.2 Lack of training

Nine teachers (P1, P4, P9, P10, P14, P17, P21, P20, P25) reported that they believed there was a lack of regulated pedagogical training or continuing professional development (CPD) to prepare instrumental and vocal teachers for dealing with complex issues like dyslexia. A teacher noted:

If there was one thing that I would summarise so far from what I've seen, I would like to create more awareness in curriculum music, in schools and in private teaching. I'm conservatoire trained and there was nothing about dyslexia. It was about how we performed or taught. (P25)

The impact of this absence of information could mean that the dyslexic student may be misunderstood or struggle within the context of the instrumental or vocal lesson. Although one might expect that there would be better support and training structures in place within a school setting, even instrumental teachers who worked in school settings recognised the need for better understanding. Assumptions made by other teachers who did not recognise the concept of dyslexia as a spectrum disorder might mean uneven levels in areas of difficulties are potentially misinterpreted and misunderstood by teachers; this might have a negative impact on the teacher's relationship with the student.

There were a variety of challenges for teachers in identifying the correct strategies to use based on their knowledge of dyslexia and its effects on music learning as well as the individual student. As dyslexia research is seen as complex, and at times, contradictory (see Chapter 2), it is not difficult to imagine that teachers might not find current information which is relevant to their teaching practice. This suggests that it might be beneficial for teachers to have access to high-quality, recent research and information which might be embedded into their teaching practices through ongoing continual professional development courses (CPD) or programmes of teacher support offered through music hubs.

6.3.1.3 Assumptions

Findings suggest that a lack of training and information sometimes leads to assumptions and misconceptions. For example, P25 noted that 'sometimes the music teachers think "it only affects reading and not music" (P25). Teachers referenced the challenges of avoiding assumptions in the process of getting to know the student: 'The more different a student is from our own way of working, the harder we have to work to understand them' (P2). Supporting students when dyslexia 'varied from student to student' was described as 'the most challenging aspect' (P7).

Even assuming that a particular strategy will be useful might lead to adverse consequences unless the teacher has a grounded understanding of dyslexia, co-occurring conditions and the individual pupil. A teacher gave the example of two different dyslexic students in their approach:

If I give them a piece of music that is quite complicated but is chock full of patterns, they will get it like "that". If I give it to another dyslexic student with sequencing issues, they will think it's horrible. (P25)

This suggests that teachers need sensitivity when considering how specific challenges from dyslexia might affect a student's preferences for curriculum, strategies and repertoire.

6.3.1.4 Misconceptions

Three teachers (P5, P7, P25) expressed the fact that there were times when they were unaware which strategies might be effective, or how to utilise them appropriately with dyslexic students. As an example, the use of colour is a recommended dyslexia strategy (Hubicki & Miles, 1991; Oglethorpe, 2008); however, a teacher expressed caution because 'If someone has synaesthesia then the colours [that a teacher chooses] may not work, then the child should come up with their own colour or choose their prompts' (P5).

There were also misconceptions about dyslexia which were utilised overconfidently as a 'cure' in educational settings. One example of this is the use of specific-coloured overlays or coloured paper which has been discussed in previous chapters of this thesis.

I've seen teachers identify dyslexic students and ask for a yellow overlay or background for them. You would ask them 'Is it really helping them?' and they would say 'I don't know, I was just told to give them a yellow background'. (P25)

A teacher who had done considerable research on the use of coloured paper or overlays reported that they were really 'bucking the trend' and caused an uproar when they gave a presentation to music educators, referring to the 'myth about dyslexia, [that] it's not visual, that using coloured lenses or coloured overlays is not an appropriate accommodation' (P8). This suggests that teachers may be aware of the variance of information available even in more supported, regulated contexts such as school settings; challenges could be even greater for instrumental teachers working in isolation.

Further misconceptions were shown by a vocal teacher who reported that their student's learning support tutor offered suggestions which were not applicable in the music lesson context, giving the student a table of 'learning techniques to say the words, speak the rhythm, say the word with the rhythm [and then] putting it together' (P11). However, the student did not find this helpful and the teacher felt that this negatively impacted their confidence, as well as reminded them of past educational experiences. Teachers' frustrations at these limitations may mean that they must explore information outside of their formal training and both informal and formal contexts may not always offer well-researched practical solutions.

This reinforces the need for current research to be made available to teachers as well as an understanding of dyslexia, specific music learning techniques and an awareness of when it is appropriate to use them. This also highlights the importance of personalising the learning process and collaborating with the student to find their best ways of learning. Despite the learning support tutor being a professional trained to assist the student (who was an adult), this emphasises that teachers should encourage students to be open and honest about what approach works best for them.

6.3.1.5 Lack of knowledge and resources

Findings highlight the barriers that teachers perceive in terms of a lack of knowledge and experience regarding how to teach dyslexic students. Teachers attributed challenges in
supporting dyslexic students to several factors including a lack of expertise and resources. Findings suggest that some participants did not feel that they had achieved much in teaching dyslexic students and correlated this with the student's difficulties as well as gaps in their own experience as teachers. Some teachers suggested feeling out of their depth in teaching dyslexic students, for example, P7 who stated: 'To be absolutely honest, I'm not sure that I did anything very successfully'. Another teacher stated: 'It is difficult to know what to do because we are still on a Grade One level because they can't handle anything more challenging; that is frustrating for me as a teacher' (P17). This is consistent with Oglethorpe (2008) who found that despite putting forth her best effort, some students did not appear to make much progress.

A dyslexic piano teacher wondered if teachers were put off by the challenges of working with dyslexic students and 'just want students who read normally and don't struggle' (P21). Teachers experienced a tension in wondering whether to attribute the lack of progress to the students' lack of practising or to a lack of preparation in knowing how to practise. Perhaps teachers may have felt unsupported in understanding these challenges and therefore reluctant to take on dyslexic pupils or they may have been unsuccessful teaching a student only to later realise they might be dyslexic. Teachers sought help from SEN specialists in school settings, organisations like the Independent Society for Musicians (ISM), professional development courses, the British Dyslexia Association (BDA) website, European Piano Teachers' Association (EPTA) and informal discussions with other teachers. A small number of participants mentioned books as resources, Oglethorpe (2008) and Miles et al. (2008), which have been mentioned previously along with a piano pedagogy book (Polischuk, 2018) and a general book on teaching students with dyslexia and dysgraphia (Berninger & Wolf, 2009).

6.3.1.6 Limited expertise in teaching dyslexic students

Some teachers recognised that they lacked proficiency in choosing and devising appropriate strategies for dyslexic students, for example P2 who stated that 'I haven't had enough experience teaching dyslexic students to be able to follow through and see how things can go' (P2). As a result of this, a number of teachers (P1, P4, P5, P8, P12, P25) sought dyslexia literacy training when they observed that certain students were not responding to traditional music teaching methods and demonstrated specific patterns of difficulties:

My journey into this was due to some pupils who were struggling and... Someone said they might be dyslexic so I thought 'Oh, I will do some research' but couldn't find much and then I went to do my Masters in supporting learners with dyslexia, a two-year course. It was about literacy and numeracy, not about music, but it was within the course that I decided to do a project on music and dyslexia. So, some of the ideas I got from literacy and numeracy have fed into what I do. (P4)

Teachers with relevant training understood that teaching was more likely to be successful when incorporating multisensory, repetition, systematic and structured learning in a personalised way. It can be burdensome for teachers to accurately estimate the amount of preparation time to prepare these strategies: one stated that the greatest challenge 'relates to the extra time it takes to devise appropriate individual strategies as all pupils are different' (P9). This underscores the need for dedicated music and dyslexia training for teachers, as dyslexia literacy applications may have limitations.

6.3.1.7 Lack of well-designed method books and awareness of appropriate repertoire

Teacher findings indicate that no single method book is useful for all dyslexic students. Teachers reported using a variety of method books to tailor repertoire to the student's needs. Numerous issues were found to be barriers, including the use of stave lines and stems for beginners, the length of pieces, small print music, disproportionate printing of music (for example, with some bars longer than others), lack of patterns, not encouraging use of the full keyboard, lack of emphasis on connecting keyboard to stave, a focus on finger numbers for note finding and pieces with counterpoint. Books and methods which focused on using colour or shape as indicators were considered to be dyslexic-friendly with some pupils (discussed further in Chapter Eight), but the eventual transfer to a conventional format of score was seen as a challenging factor.

Identifying a lack of emphasis on coordination in method books, one teacher felt that:

There aren't any method books specifically for dyslexics. I think they all assume that the coordination will develop on its own and with dyslexia, you have to make that your first priority. If you don't, it won't develop. (P21)

Finding appropriate repertoire was emphasised as teachers felt it was important for dyslexic pupils in particular to experience success when learning a piece as a means of keeping up their motivation. With some of the hurdles relating to sight-reading, repertoire choice may be limited and teachers may have to make sensitive decisions:

They want to play Einaudi or a film theme or a video game theme and get loads of 'street cred', or maybe tackle the reading and we know that's a challenge and so we have to have a different set of expectations for the music we can play. But it's still valuable and we need to do them both concurrently (P4).

Therefore, teachers may need to be resourceful and creative in their approaches in negotiating a balance between the student's choices, their capabilities, the difficulty level and the accessibility of the music. This also highlights the usefulness of and need for potential repertoire lists which may cater to dyslexic students' strengths but would also interest and challenge them.

6.3.1.8 Limited knowledge on the availability and utilisation of technology

Findings suggest variance amongst teachers' perceptions of the value of technology in lessons. A teacher considered that while technology might provide some motivation as a novelty, it was 'just not something I use very much in my teaching anyway' and reflected whether 'it would have really achieved any worthwhile teaching goals' (P7). According to one piano teacher, training apps were not helpful because they lacked a multisensory element (P5). Another teacher also indicated there were limitations and benefits with technology:

Technology can be a distraction, it can also be a tool. But if you are trying to teach a kid, where's 'A'? You don't need bells and whistles, it's simple and structured. (P8)

Due to the rapid pace of change, one teacher stated: 'There is so much technology that comes out now that I can't keep my finger on the pulse of it' (P12). One participant reflected that they 'never used it with dyslexic students, because I was completely unaware of what was helpful' and did not have access to 'lists of recommended apps' (P10). This indicates a lack of informed knowledge on which apps are available and relevant for dyslexic learners and a concern about the students' engagement with them.

6.3.1.9 Limited guidance for graded music performance and theory exams

Teachers reported a variety of perspectives on the use of music exams, practical and theory, with their dyslexic students. Some felt that dyslexic students were not interested in taking exams because they could not cope with the amount of repertoire that needed to be learned (P17, P25), they did not want the stigma of using accommodations (P5) or had difficulty with sight-reading elements (P1, P10, P13). Some teachers were not aware of the reasonable adjustments or accommodations which might be used in exams, nor of the preparation and practice students needed to utilise them in an effective way during exams; they also felt that there needed to be more flexibility in the use of reasonable adjustments, with further research on their effectiveness needed (P6, P12, P17). In two cases, teachers noted experiences where the examiner demonstrated a lack of tact and sensitivity when it came to allowing reasonable adjustments in the exam setting (P6, P12). Examples of accommodations used were extra time for sight-reading, aural and transposition sections, as well as note-taking, enlarged music and the use of tinted paper. Another example of this was the replacement of Grade Five music theory exams with the alternative Practical Musicianship exam in the Associated Board of Royal Schools of Music (ABRSM) syllabus as discussed by P7 in relation to a student who struggled with handwriting.

Others reported the challenges involved in searching out information and selecting exam boards which were more tailored to the specific instrument, the students' needs and their capabilities, citing a lack of clear signposting and accessibility of information (P1, P7, P25, P26. This raises the question whether it was the parent, teacher or the exam board who should be responsible for educating about specific learning difficulties and available accommodations (P1). Parental expectations regarding exams were a tension which teachers had to navigate, with one teacher explaining that they dealt with this by explaining to parents that the goal of playing should be measured by enjoyment, rather than preparing for exam deadlines (P25).

6.3.2 Barriers: Communication with school and parents

Several sub themes arose in relation to the issue of disclosure and an understanding of the student. Teachers reported that poor communication with parents and schools might prevent them from effectively supporting the student. For a variety of reasons, students and

their parents might be reluctant to disclose information related to their difficulties or access arrangements at school.

6.3.2.1 Communication with schools

One teacher referred to a supportive collaboration with schools where they worked: 'I was very fortunate to have that information to hand and that was shared with all the instrumental teachers and all of the staff on some kind of document' (P7); however another teacher was not 'comfortable searching out information about students unless it was definitely necessary' (P2). Several teachers were left to discover the student's challenges by themselves (P1, P6, P15, P16, P23, P25). Similar to the literature (Chapter 1) which suggests that schools need to be incentivised to support students with special educational needs, one teacher believed that the reluctance of primary schools to have a child assessed, or even to acknowledge difficulties, was related to funding issues:

I spoke to her mum and she said 'No, I spoke to the school and the school says nothing is wrong' and I've had this happen so many times ... actually dyslexia is not something 'wrong'. The schools have a vested interest in this because if the student is diagnosed then they will need support, which the school has to fund. (P25)

These findings suggest that teachers may be faced with a variety of scenarios when it comes to communication with schools. If schools do not see the benefits of sharing information about students with the instrumental teacher, this may be a barrier to the teacher's ability to provide the student with effective support; this may, in turn, have negative ramifications for the student. This also suggests a need for examples of good practice in collaboration between parents, schools and instrumental teachers.

6.3.2.2 Communication with parents

Findings suggest the value and the beneficial impact of parental involvement: 'A lesson is just one part of the learning process, in fact it's just the tip of the iceberg as most of the learning is actually going on at home' (P4) and that parents should be 'as well informed as possible about the condition so that they can be the child's advocate' (P9). However, parental knowledge and belief systems about dyslexia, expectations for their child and other factors may create barriers for teachers in developing their understanding of the student and supporting their practice at home.

6.3.2.3 Inaccurate knowledge and belief systems about dyslexia

Several teachers acknowledged that the parent's lack of knowledge about the effects of dyslexia on learning music might hinder them from sharing information as they might not view it as relevant. As a teacher noted:

It took a lot of courage to speak to one mum of a student I suspected was dyslexic. When I explained what I thought, she said 'Oh yes, we know he is dyslexic!', then I said 'You never told me?' (P25)

Teachers felt that parents' belief systems about dyslexia might influence disclosure and the value of a label is subjective (P10). Another participant referred to the fact that there are 'lots of triggering opportunities with the parents if they are dyslexic and they are undiagnosed or ashamed that their child has problems' (P5). Some teachers recognised that 'Dyslexic pupils often have dyslexic parents, and that can be a problem in itself as the students might not be getting the help to remember' (P4); this may mean that lessons or practice sessions might be missed or music is forgotten. Being aware that parents may react defensively or become upset, recall personal negative educational experiences or perhaps believe that their own dyslexia might potentially affect the learning of their child suggests that teachers need sensitivity and tact when communicating with parents. The importance of letting parents lead in any discussion pertaining to students' difficulties was highlighted by a teacher who stated:

I will ask them if they have noticed that their student is having issues at school. If they say the student is doing fine, but I'm seeing a learning issue, then I will not pursue it. If they describe problems with reading, then I will mention that I have noticed the same thing in piano. (P21)

These findings suggest that parents need high-quality information about music learning and dyslexia and that instrumental teachers need sensitivity and care when supporting parents, especially those with dyslexia.

6.3.2.4 Pressure through expectations

Teachers found that navigating the expectations of parents could be challenging. However, the importance of the quality of the relationship between teacher and parent was highlighted by the following participant:

The parent who is keen for a child to do an exam, you have to talk to them and explain. If they don't pass [the graded exam], it will have a massive effect on their self-esteem and I am not keen to put my name to it. (P10)

This also suggests there can be considerable pressure on teachers to meet parental expectations. Parental expectations can have an impact on the student's engagement and self-confidence, but teachers noted that sensitive communication and quality relationships helped them to navigate these challenges.

Parental expectations may come from a desire to see their child succeed in spite of their challenges, as P18 noted:

Sometimes, if the student is having lots of problems, the parents are glad the child can do anything at all. But, if they don't accept that the child has problems or they think that music is going to be the answer to everything, then their expectations can be incredibly high. (P18)

One teacher recognised the phenomena of intense competition amongst families and spoke about the challenge of 'deflating this in the musical setting' by reiterating that everyone is on their own musical journey and the need to 'really praise the things that student has done, why you are really proud of that and what skills have been involved' (P4). These findings suggest that teachers might need to take the initiative to prepare and present information regarding expectations in a way that sets up both the student and parent for positive outcomes.

6.3.2.5 Lack of supportive collaboration

It was noted by several participants (P5, P18, P24, P25) that parents might not be involved for a variety of reasons:

[Students need] help with their organisation, help with their practise and someone sitting alongside them. Not in a negative sense, but in an encouraging way: 'I love it when you play that' or 'Could you do that again?' They need encouragement or they get dispirited and distracted, but a lot of parents just don't give it the time. That is such a pity. I ask them through emails or in the notebook, but it often doesn't happen. (P18)

When asked why parents might not be involved in their student's practice, one teacher stated: 'I think it's because it can be a bloody great fight to work with your own children' (P18). There were a number of challenges which teachers faced regarding relationships with parents. Some found parents were overbearing: 'If a mother tells me something her child likes, I will listen politely, but they may be used to speaking for the child' (P10). Other participants found parental interference was not helpful: 'Especially if a parent is musical, I find they often want to get involved too much in the teaching, but usually with no teaching experience' (P27).

These findings indicate the importance of clear guidance from the teacher on what they expect from the parents and how parents might support the student whilst also remaining sensitive to the challenges of parenting, parental expectations, the busyness of modern family life and family dynamics.

6.4 Challenges in the teaching context

Findings suggest there are challenges in the development of individual profiles of students, for example, in recognising whether the student comprehended a task or information, recognising when they experienced stress and developing the student's metacognition. Some of these challenges included 'working out the individual strengths and weaknesses' (P1). According to P27, the challenges vary due to the age of the student:

With younger students they aren't always sure what they are 'meant' to see, so find it hard to describe what they struggle with, and what they find ok. It's a case of working with them to help them, and then work with what works for their individual needs going forward. In the case of adult learners, they tend to just know that standard notation is too complicated for them. (P27)

6.4.1 Identifying dyslexic students

Whilst teachers tended to recognise that it was not their remit to suggest that a child might have dyslexia, some believed that it might be in the student's best interest long-term to signpost them to formal testing, with one teacher stating: 'If you see patterns of difficulties, then you start to think it might end up helping them more in other educational aspects' (P1). Some teachers identified undiagnosed students by comparing their challenges with diagnosed students in their teaching practice:

He [the student with the assessment] would crack something one week and it would be gone by the next. And he did some other unusual things which made me think I had other students with the same problems. I ended up having a further six [dyslexic] students then at school. (P5)

Wanting to avoid speculation, another teacher noted that 'I think there are quite a few who are dyslexic who haven't told me; it is a difficult thing to ask because they might think I see them as [being] "thick"' (P17). Raising awareness of the issue might have wider ramifications for the entire family and for the student's support at school; as one teacher 'suggested [the student] was screened; in the end, they turned out to be dyslexic as did several other family members' (P10).

As mentioned previously, this may be a sensitive topic and the outcome of the discussion might range from parental acceptance to defensiveness, all of which need to be dealt with by the teacher, who may often be unsupported and struggling to find relevant information and useful resources. This highlights the tensions that teachers face between a sense of responsibility to the student and potential guilt for feeling unsure or under-equipped to know how best to handle parental relationships. Having identified that the school, parents or students themselves may or may not notify the teacher of a diagnosis, these findings indicate that there may be clear reasons why the teacher might choose not to seek out specific information about an assessment. This also suggests the importance of the relationship between the teacher and the individual student, including knowing the student and demonstrating sensitivity to their views about themselves and dyslexia.

6.4.2 Recognising whether a strategy is working

A teacher's need for observational skills and experience were seen as two important issues when recognising if strategies are working or not. Resisting an overly critical approach was a challenge: 'I start evaluating the teaching process of taking the student from point A to point B; but it is very hard not to put a value on what they do as being right or wrong' (P19). In considering why the student avoids an activity, or is compensating in some way, the relationship between teacher and student is crucial: 'It would take a long time to build up the trust for them to say they don't understand' (P5). Teachers recognised challenges in building up the student's 'self-confidence, because they know they are messing up but they don't quite know how to get out of it' (P18).

Recognising the student's difficulty threshold was not a simple task, according to participants, and the student's ability to deal with risk and frustration had to be managed carefully. Whilst emphasising the need for fairly easy successes in order to build 'selfconfidence', another participant stated 'I think you've got to be careful that you don't help them to avoid things they find difficult all the time' (P5). Distinguishing difficulties related to dyslexia from those from a lack of instrumental practice was highlighted by several participants: 'You often wonder "Is the student practising or is there a dyslexia issue?"' (P21). This raises several issues including decisions about whether pupils should avoid 'harder' instruments and how much of accessibility should be about making the tasks themselves easier or adding scaffolding around them to support them, and how much about equipping students with the toolkit they will need to build their own scaffolding. Another possibility relates to dyslexic students not practising because of underlying issues which are preventing their practice from being meaningful and productive.

Some students, especially older students, were able to articulate the things they found useful or not useful, as illustrated by this teacher's experience with note making on the score: 'One girl says "It's just too much information on the score and my eyes don't like it"' (P11). At other times, a student might not be able to articulate what they see or what is not working for them:

I think a lot of students with processing issues of any kind are used to going through school life sort of assuming that they can't expect to understand anything

completely in the way that they see others doing; this is particularly the case for those who remain undiagnosed. (P2)

In summary, a teacher's observational skills and experience were significant factors in recognising approaches that might be effective. Dealing sensitively with the student's selfesteem, recognising the student's difficulty threshold, carefully choosing informed strategies and setting realistic expectations of progress were key elements of navigating the learning process in a way that was rewarding for the teacher and student.

6.4.3 Lack of control over the learning environment

A teacher's lack of control over the learning environment may be an obstacle to the student's progress. A teacher mentioned an adult student's concern that 'certain lights really upset them, the glare; they had to move their place in the room, and adjust the lighting so it's not too [bright]' (P11). These difficulties might affect teachers who are teaching in schools, without much natural light or in noisy or distracting environments; equally, a student's own home might not be set up in an ideal way to support their learning.

Online lessons presented difficulties for teachers in terms of the limitations of technology, with a teacher reporting that an online lesson was held with the mother holding a phone for the student which was moving constantly and made it difficult to track what the student was doing (P17), and describing how it was much more difficult to point out notes on the piano (P21). The instrument at home might be poorly tuned which could make it difficult to assess the student's performance (P17). Another challenge with online lessons was that it might increase the student's self-consciousness if their family overheard them playing or singing as the lesson may be situated in a more public space within the home (P16). These findings highlight the challenges faced by teachers who may be aware that certain environments are not conducive to learning for their dyslexic students, with online lessons creating a number of challenges, and underscores the tensions between a desire to support the student and circumstances out of the teacher's control. This also links to the importance of adaptability as teachers and students find ways of overcoming these challenges together.

6.4.4 Recognising when a student is experiencing stress

Teachers experienced challenges with understanding perplexing behavioural outcomes in lessons, although teachers recognised that it was also a challenge not to make assumptions based on student's responses or behaviour (P10, P13, P20). Being either 'self-driven' or 'apathetic' can be an indicator of a dyslexic student experiencing stress; as one teacher noted, 'they're just so used to failing' (P22). P13 reported that 'Often, I find that the students are really shy and they think they can't do anything' (P13). Noting a behavioural difference between genders, one teacher stated:

Some of the students who struggle with self-esteem walk in and they can be totally 'bolshy' ... It seems to me that girls will come in and verbalise their problems and frustration, but boys come in and act like 'Whatever, yeah' and that may also be because they can't do it. (P18)

A piano teacher called attention to the issue of 'masking', referring to 'a child that's extremely good at pretending everything is okay when it's not at all' (P5). This highlights the need for teachers to exercise caution in assuming that students have understood what has been taught. The uncertainty that arises when a student avoids an activity or presents with confusing behavioural clues highlights the need for teachers to develop a supportive relationship with each student individually, but also the importance of adequate training and knowledge. Findings suggest that stress might emerge in a variety of behaviours. P11 observed a lack of confidence being shown through students' 'poor posture and low energy' as well as the following behaviours:

Frustration comes out in the lesson, a lot of self-castigation, or apologising. And I would ask them why they feel the need occasionally to apologise or to be so upset and try and get to the bottom of what their belief is about it, and then try to help them. Show them what they can do when they approach things differently. (P11)

This indicates that teachers may need to gently probe students to find out what is behind their behaviour as well as offer reassurance. A teacher with dyslexia described how her dyslexic students 'are sharp at what they can get away with, covering up' suggesting that

she recognised the emotion behind this, stating that 'we've done it our whole lives because we don't want to be embarrassed' (P21). Other indicators include visual stress:

A reluctance to look at the book, and complaints about their eyes hurting, the music jumping around or rubbing their eyes or taking their glasses off and cleaning them; you recognise that the student is getting quite stressed with the book. (P25)

Dyslexic students with co-occurring conditions might also exhibit stress in different ways. A teacher related their experience of teaching a dyslexic student with ADHD, recalling that 'if I try to challenge them too much, it becomes too much of a conflict for them' (P12). These findings suggest that indicators of stress might be seen as avoidance or reluctance to perform specific tasks and might be displayed in a variety of ways from low energy, negative body language, and visual complaints to self-castigation. A good relationship and honest communication with the student may enable the teacher to understand the backdrop of the behavioural issues which are present.

6.4.5 Developing the student's understanding of dyslexia and their own metacognitive processes

Teachers found that students did not always have accurate information about dyslexia and the challenges it might create for them as well as their metacognitive approach to music learning. At times, this meant giving reassurance: 'I explained to them that it is normal for them to have difficulty with the scores and most have never heard this before; lots of them think they are alone' (P15). A piano teacher described their experiences with a dyslexic student who struggled with sight-reading:

I said to him [that] his brain is working fine, it is just the way that music is written down is not working for him. If he had devised a written music system, he would find it easy to read and I would find it impossible! (P2)

These responses demonstrate empathy as well as an awareness of the importance of encouraging self-reflection. This emphasis of a student examining their own learning and playing was seen as crucial, as a teacher recalled of his teenage student that 'If you could self-reflect and actually understand where the errors are in your own performance, then you are sorted' (P12), though this might be more challenging with younger students. When asked about their own performance, students always focused on negative aspects according to a teacher who found a way to counter this: 'I start with looking at the good things, and then we look at the bits that didn't quite work' (P2). Teachers' responses indicate that reassurance and positive feedback, as well as encouraging the student to assess their own playing and develop their understanding of dyslexia, contributed to growth in a student's self-awareness.

6.5 Challenges faced by teachers with dyslexia/neurodivergence

In this study, self-reported neurodivergent teachers referred to their own past negative educational experiences as a motivation for improving the learning environment for students and participating in this research. Of the twelve teachers who disclosed their own dyslexia or neurodivergent conditions, the main challenges centred on the inaccessibility of pedagogical materials for dyslexic teachers, and difficulties related to their specific neurodivergence, including short-term memory recall and sight-reading during lessons. However, these teachers recognised an advantage in that they could relate to their students' frustrations based on their own experiences. One teacher found it useful to empathise with them: 'I say "Do not apologise, we learn by making mistakes" and I still make mistakes when I am sight-reading their pieces... I point it out and try to encourage them' (P13).

Teachers found that it was useful to share their own experiences with dyslexia for the purposes of reassuring the student and also as a means of promoting awareness of the positive aspects of dyslexia. Conveying a sense of advocacy was also seen as an important challenge, with one vocal teacher stating it was 'once I realised there was a difference in the way I perceive things, I started very consciously to develop strategies' (P19). A dyslexic teacher who had negative educational experiences reported that 'I don't want anyone else within my educational sphere of influence to suffer or go through what I had to go through' (P12). Being able to impart this concept to dyslexic students might be a valuable way of helping them to overcome negative attitudes or emotions. At the same time, a dyslexic teacher may need to exercise caution in managing assumptions and maintaining a flexible approach by being aware that their dyslexic students may have different learning profiles, personalities or past experiences.

6.5.1 Limitations caused by the inaccessibility of material

Most of the research and pedagogical literature is text-based, and this impacts the accessibility for some dyslexic teachers, as one participant noted, stating that 'the BDA book about Performing Arts²³ was very good, but I'm not a good reader, which is my problem' (P12); their solution to this issue was found in verbally communicating with another teacher: 'I do talk to a SENCo [Special Education Needs Coordinator] who is also dyslexic' (P12). Another participant stated that 'Some of the books are too long and I need some pictures as my own issues with dyslexia affect me' (P21). Therefore, further consideration is needed in relation to the accessibility of material provided to inform and support all teachers, not just those who are neurotypical.

6.5.2 Challenges related to the particular learning difference

Teachers related negative experiences in either teaching or in training which created challenges for them. One teacher related her feelings, stating that: 'to begin with it was the most terrifying thing I have ever done because I thought "I have to keep up with this student"' (P13). Referring to their own coordination issues when they attended a Dalcroze training session, a teacher stated that:

The rhythm games are very physically complex and if you are mildly dyspraxic, it is horrific. I remember being at a session, sitting at the back and wanting to cry because I wasn't very good at it. (P5)

This sense of shame was also referred to by another piano teacher who felt that they were often put under pressure to sight-read in various capacities. As one piano teacher noted, 'I have learned to say "no" because it is so embarrassing when I'm trying to read something pretty elementary and I just can't do it' (P21).

Preparing students for exams was another area of stress, with a dyslexic teacher relating: 'That's when I get anxious; I don't feel like I can support students through sight-singing to the level that is expected in that environment of the exam culture' and 'Sometimes I will even say to them, "Why don't you go to see a piano teacher for a little bit and do a bit of sight-singing with them as well?"' (P16). This suggests though that there seems to be

²³ Daunt, S. (2012). *Music, other performing arts, and dyslexia*. Published by the British Dyslexia Association.

reluctance or shame on the part of some teachers to acknowledge that another teacher might be better able to support the student in a specific area.

Other participants referred to similar challenges: 'My ability to recall music was probably affected by my dyslexia' (P12). Sight-reading issues (as stated in 7.6) especially affect teachers who may need to demonstrate for students, as one participant described: 'If they are having issues, especially with the early pieces, I will play for them [but] as they get into intermediate, I will play through *as I can*' (P21).

While some dyslexic teachers see themselves in an empathic role with dyslexic students, students' difficulties may remind dyslexic teachers of past negative experiences, as one participant recalled: 'What I get a lot from these students with issues is anxiety and a sense of failure; I recognise that from my time in school [as] I constantly felt that I was failing and not good enough' (P25). Having experienced many of the frustrations which their dyslexic students face, neurodivergent teachers appear able to empathise, but they reported some barriers in carrying out tasks as teachers and performing musicians.

6.6 Summary of findings

In summary, teachers reported several barriers related to research, training and resources, as well as in communication with parents and schools. A summary of the challenges faced by teachers relates to the teaching context, behaviour, learning environment and understanding of the student's way of learning.

6.6.1 Barriers related to research, training and resources

- Findings suggest that although teachers report seeking research and information
 related to music and dyslexia, they struggled to find it and recognised the need for
 more formal research, training, and practical resources. This includes well-designed
 method books which are accessible to dyslexic students, as well as research and
 information related to technology and its application for dyslexic students.
- Teachers were aware of the variances in information related to coloured overlays and felt more research was needed to clarify their usefulness.

- A recurrent theme in the interviews was that as instrumental teachers often work in isolation, learning tends to be informal and non-research-led, and this may lead to assumptions and misconceptions about dyslexia.
- Assumptions reported by participants include the belief that dyslexia only affects text, that every person with dyslexia is the same and that the same strategies will work with all dyslexic students.
- Teachers might be reluctant to take dyslexic students for a variety of reasons.
- Teachers recognised several barriers and challenges related to preparing students for exams, including selecting an exam board, obtaining information, preparation for exam requirements, coaching needed for using reasonable adjustments and managing parental expectations.
- An important theme was that whilst a minority of teachers had dyslexia training, there was a consensus that this minority sought to apply dyslexia literacy approaches in their music teaching practices.
- A few interviewees reported a need for resources and materials that are accessible to all teachers, not just those who are neurotypical. Although they were able to empathise with dyslexic students, neurodivergent teachers attributed their difficulties with certain tasks in their teaching practice to their neurodivergence.

6.6.2 Barriers in communication with schools and parents

- Some schools were reluctant to acknowledge students' challenges with learning; participants suggested that limited finances and resources might be the root cause.
- Concerns regarding poor communication with parents emerged, and whilst most participants agreed collaboration with parents was important, they recognised challenges in recruiting their involvement and managing their expectations.
- Findings suggest that parents may not always believe that disclosure is relevant for their child's music teacher.
- Teachers diverged in their belief that they should report concerns to parents, as some felt that it might lead to better outcomes for the student whilst others felt that

it might create tension with parents. The latter considered that parents should be the ones to lead any such discussion.

• Participants indicated that parents, especially parents with dyslexia, may react defensively if the teacher addresses their concerns about the student.

6.6.3 Challenges

- Teachers reported that whilst it was challenging to identify undiagnosed dyslexic students, the students might be reluctant to disclose. Lesson observations as well as comparison of their difficulties with those of assessed students were common starting points.
- An important theme was that behavioural issues might be an indication of student stress. Teachers found this challenging to interpret at times and one participant noted that there might be gender differences in the way it was displayed.
- A subtheme of behavioural issues was the idea that dyslexic students may 'mask' or hide their difficulties.
- Teachers found balancing the difficulty level of repertoire and knowing how to support, whilst also maintaining the student's self-esteem, to be a challenge.
- Participants noted that developing the student's metacognition was a challenge which they attempted to overcome through reassurance, explanation of how dyslexia affects individuals, encouraging self-reflection and focusing on the positive aspects of the student's efforts.

6.7 Conclusion

Findings indicate that teachers experience barriers related to a lack of training, research and expertise, as well as those linked to communication with schools, parents and students regarding disclosure and support. One interesting finding is that teachers tended to correlate poor strategy selection to assumptions and misconceptions about dyslexia, based on a lack of knowledge and experience. Most striking was the substantial difference between teachers who had dyslexia literacy training and those who did not in terms of an

awareness of strategies likely to benefit dyslexic students. The results of this study show that teachers face significant challenges in identifying dyslexic students, selecting appropriate strategies, recognising student stress and encouraging the development of metacognition. Further challenges were reported in communication with schools and parents, as teachers sought to navigate their expectations sensitively. Some parents reported to music teachers that primary schools seemed unwilling to acknowledge issues with dyslexic students; this may pertain to funding. On the other hand, some parents may welcome the information, act as advocates for the student and have a greater understanding of their own or other family members' dyslexia. Choosing relevant method books or repertoire required teachers to be discerning about student's needs; there appears to be a lack of appropriately structured material designed with dyslexic students in mind. Some teachers expressed concern about the lack of information guiding them in the use of appropriate technology for dyslexic students, particularly training apps, and there was concern that screens and devices could not replace quality teaching approaches. Motivated by their own learning experiences and feeling able to deliver lessons with empathy, a group of self-reported neurodivergent teachers described challenges which might enable insights to better support them as colleagues and in workplace environments such as schools or music hubs.

REFLECTIVE STATEMENT 3: MY BARRIERS AND CHALLENGES

RS3.1 Barriers: Lack of knowledge

Bronfenbrenner and Morris (2006) refer to the chronosystem of the bioecological model and like tracing footprints on a journey, this reflective statement is meant to give some sense of time to my development as a teacher. I started this journey much like many of the teachers I interviewed. Prior to studying on the MA Music Education: Instrumental and Vocal Teaching course at the University of York, I had never received an explanation or had much understanding of dyslexia in my educational experiences. My primary and secondary school years took place in the 1970s and 1980s and I do not recall hearing the term 'dyslexia' ever being used. During my years in university, in the early nineties, I cannot recall a single discussion on dyslexia. Until the MA, my instrumental music learning experiences were mainly in the form of tuition with a single teacher, and as I am not dyslexic, the issue was never raised. When I began teaching piano lessons in 2012, apart from some continual professional development courses through the local music service, most of my growth as a teacher was from informal conversations or through my experiences of teaching.

This lack of awareness contributed to my feelings of unease when I had a student who seemed to struggle with specific patterns of difficulties, and this created a discrepancy when compared with their understanding and knowledge overall. It was during my master's dissertation project in 2017 that I began to look for literature on the topic and found, just as the teacher participants in Chapter 6, that there was very little information or practical strategies available. I read Oglethorpe (2008) and Miles et al. (2008) and whilst that developed my understanding further, it also served to underscore the complexity of dyslexia and the different ways in which it might affect individuals. Most of the literature seemed to focus on anecdotal experiences rather than foundational principles for teaching music to dyslexic students.

RS3.2 Challenge: The term 'dyslexia'

I realised that although I was looking for answers to some of my questions and hoping to find direct solutions, dyslexia is far too complicated to allow for simplistic answers. I recognised that this may be the reason for the lack of available information. There was also

a dilemma as I came to realise that the term 'dyslexia' was inadequate as an explanation for the differences that might be seen. The terminology used is important and has ramifications for the way dyslexic students view themselves, are viewed by their teachers and parents and whether they qualify for support and accommodations.

RS3.3 Challenge: My own assumptions

A greater understanding of dyslexia has certainly developed in the past twenty to thirty years, and the shift from a medical model of disability to a social model implies that we need to change the way we teach and the materials we use to be more effective teachers for dyslexic students. Correlated to my lack of awareness of Oglethorpe's (2008) book prior to this research, I found that many teachers I interviewed for this research or met in workshop settings were unaware of this book. Teachers appeared to be looking for quick fixes and easy solutions to teaching dyslexic students, such as offering students coloured overlays. I recognised that my lack of understanding and knowledge caused me to misinterpret some of the challenges I faced with my students, believing their challenges to result from a lack of practising or insufficient effort on the student's part.

My own experiences with dyslexic students showed me how easy it was to make assumptions about their work ethic when the problem might have been a concentration or processing issue. I realised that I needed to be more sensitive to dyslexic students and that their previous learning experiences may have been rooted in failure or encouragement to 'try harder' when in fact, they were trying their best. I tried to develop more empathy through questions about their experiences. This led me to value the uniqueness of each student, including their needs and strengths.

RS3.4 Challenge: Choosing teaching strategies

In the Reconnaissance cycle of my research, I was aware of several strategies which had been shown to be beneficial in working with dyslexic students from the literature. These included the general strategies which had been shown to be effective in literacy training: multisensory, repetition, systematic, structured and personalised. Strategies which I used with my student, prior to my research, incorporated more aural pattern work and enlarging music for sight-reading. We increased duet playing, and this seemed to build some confidence. Duet playing seemed to deflate the sense of comparison or competition they felt when peers were playing more advanced pieces at a recital. I wondered whether there were teaching strategies and methods to help dyslexic students overcome those challenges and whether the next stage of my research would provide insight as to how to put them into practice.

Like the teachers interviewed for my research, I was not aware of applications or specific software which might be beneficial for dyslexic students. With experience of submitting students for music exams since 2012, I was not aware of the process of applying for reasonable adjustments nor how to prepare the student for using these in a music exam.

RS3.5 Challenge: Recognising strategies of resilience and support that enable flourishing

Listening to Anna Devin's session 'Thoughts from a dyslexic opera singer' at the British Dyslexia Association Music and Dyslexia conference in 2018 highlighted the resilience and determination needed as a musician with dyslexia. My notes indicate that Anna underscored the importance of her parents' and teachers' support. She described her strategy of walking out the storyline of a piece to stimulate her memory recall. This was like the narrative reasoning strengths described by Eide and Eide (2011). I was curious about her mention of having a teacher who was Froebel trained. Friedrich Froebel was a German educator who believed in the uniqueness of each individual student, that learning develops from a sense of curiosity and play, and the need for knowledgeable and supportive teachers, families and communities (Watts, 2021). There are similarities in this approach to the Universal Design for Learning framework (CAST, 2018) which is described in greater detail in the literature review (Chapter 2), and which is not well-known amongst the music teachers I interviewed and worked with in workshop environments. I became aware of UDL (which is discussed further in RS5) through an internship in digital accessibility at the University of York and could see the potential of using it for lesson planning for students, especially for my dyslexic students.

RS3.6 Conclusion

Just like the teachers in my research, I faced barriers and challenges in teaching dyslexic students, from my starting point of a complete lack of knowledge and experience relating to dyslexia to my search for literature which would help me to become a more knowledgeable and effective teacher. I was fortunate to have two influences which deepened my understanding: the community of the music department at the University of York and the British Dyslexia Association music committee. Through this, I interacted with teachers who had more extensive training and knowledge in working with dyslexic students. Bronfenbrenner and Morris (2006) draw on Elder's (1998) life course theory in their bioecological model, which traces 'historical time and place, the timing of lives, linked or interdependent lives, and human agency' (p. 4) in their framework for individual development.

This reflective statement has identified the implications of time in the rationale for my research project by highlighting the limitations in the form of barriers and challenges which I faced at the beginning and factors and relationships which influenced my understanding. The following list describes the next four reflective statements which evidence how these findings are embedded in my teaching practice and are interspersed with the findings chapters. I discuss transitions in my development in the following areas:

- Reflective Statement 4: Transitions in power and knowledge dynamics between teacher and student
- Reflective Statement 5: Shifting from a deficit-lens to a strengths-focused teaching approach
- Reflective Statement 6: The transition from novice approaches to the use of research-informed strategies developed within the framework of Universal Design for Learning to guide lesson planning thus enhancing my tacit knowledge of students and effective teaching approaches

Reflective Statement 7: The development of greater empathic awareness with my students and their families as well as the wider community of dyslexic individuals; I also reflect on the impact on teacher well-being and importance of preventing compassion fatigue

Chapter 7 FINDINGS FROM TEACHER INTERVIEWS: STUDENT CHALLENGES AND STRATEGIES

7.1 Introduction

This chapter describes the following themes of student challenges and corresponding pedagogical strategies that emerged. Subthemes connect the primary difficulties teachers associated with dyslexia (7.2), including slow processing speed (7.2.1) and executive function deficits (7.2.2) with corresponding general strategies (7.3). The characteristics of low self-esteem and meditating strategies (7.4), coordination and spatial awareness issues and strategies (7.5), visual processing difficulties and strategies (7.6) and specific sight-reading strategies (7.7) are explored in the second half of this chapter.

Chapter 2 of this thesis provides an in-depth overview of the academic and pedagogical material available regarding dyslexic student challenges and teacher interventions. Published literature (Miles et al., 2008; Oglethorpe, 2008; Nelson & Hourigan, 2016; BDA, 2024) and self-published material (Aloba, 2020) largely consist of anecdotal experiences of instrumental teachers or dyslexic music students. Studies on adaptive music notation suggest that the use of score enlargement (Flach et al., 2014) and the use of a colour stave (Hubicki & Miles, 1991) have been beneficial to some dyslexic students. Many strategies draw on recommended literacy teaching techniques for dyslexic students struggling with reading; these include multisensory teaching, simplification, highlighting important information, carefully considering the learning environment, overlearning, assistive technology, the use of explicit instruction strategies, structuring the material systematically and adapting to individual learners (Phillips & Kelly, 2016). For some dyslexic students, previous educational experiences have been negative and learning environments may be a cause of stress (Livingstone et al., 2018) with teachers encouraged to display sensitivity (Oglethorpe, 2008, p. 28).

7.2 Student challenges identified by teachers

Consistent with the literature, teachers reported several areas of challenge with their dyslexic students. Although findings suggest that these difficulties commonly result in issues

with sight-reading as it tests multiple skills simultaneously, in some cases (P7, P8, P14, P17), teachers reported dyslexic students who did not struggle with sight-reading. P8 states:

I don't think that all dyslexic kids have problems reading music. That's just my anecdotal observation. And not all kids who have problems reading music are dyslexic. But I think there is some crossover and I think it's more about brains that are just wired differently and for that particular skill. (P8)

This may suggest that some students have either mild dyslexia or less difficulty reading symbols than text, although pinpointing the exact causes or understanding the exact effects of remediation strategies is not always possible.

Sight-reading difficulties ranged from an inability to read or understand music notation to confusion with the lines and spaces to an inability to connect the score to the instrument. Musical concepts and lyrics also presented text-based or memory recall problems. Several teachers observed that not all dyslexic students struggled with rhythms (P2, P11, P26) and one teacher reported having dyslexic students who were 'excellent at rhythm' (P15). However, some dyslexic students had difficulty 'internalising the pulse' (P5, P25), recalling the rhythm (P4, P6, P13), processing or distinguishing the rhythm symbols (P24, P25, P26), feeling the duration of a note (P4, P22) and combining pitch and rhythm together (P4, P26). Teachers reported that students struggled to recognise when a melody line was going up or down and coordination issues sometimes prevented dyslexic students from being able to move the piece forward without compromising the pulse.

As might be expected with a spectrum condition, teachers reported variances in the types and severity of the difficulties experienced by their students. This suggests that there is overlap between many of these areas; for example, rhythmic difficulties might result from coordination problems, but could also be an aural processing or sequencing issue. High rates of co-occurring conditions like autism, attention deficit/hyperactivity disorder (ADHD), dyspraxia or developmental coordination disorder (DCD) (Hendren et al., 2018; Moll et al., 2020) amongst individuals with dyslexia complicate the process of identifying the cause of difficulties. As Snowling et al. (2020) observed, 'A central theme in the history of dyslexia is the tension between the specificity of the disorder and its complex association with other

forms of learning disability' (p. 501). The difficulty of distinguishing specific challenges highlights the importance of teacher training and an understanding of co-occurring conditions, the development of student profiles and letting a personalised approach drive lesson preparation.

7.2.1 Slow speed of processing

A participant recognised that the 'speed of processing is a primary challenge' (P4) for dyslexic students. Drawing a comparison to reading, another teacher stated:

Musical notation is a desperately complicated code anyway. There can be difficulty in recognising a pattern and I think there is an analogy to be drawn between looking at notes on a page and looking at letters and words on a page. I think when a child is learning to read, they will sound out letters individually in a phonic way to make the sound of words. I think the same kind of thing can happen with things like scales and arpeggios, seeing a group of notes on a page and having to process every single note and go through all the ponderous stages, can sometimes indicate that there is a difficulty. (P23)

Note and rhythm reading, alongside other information like tempo, dynamic and pedalling markings must be assimilated in time, in addition to the placement of hands and selection of the correct notes. Other instruments might have different processing demands, for instance with embouchure or maintaining pressure on valves or a bow whilst also reading a score of music. Thus, the combination of physical and mental processing was reported to be a limiting factor for some dyslexic students.

7.2.2 Executive function deficits

Executive functions are core skills which affect many aspects of student development; they consist of 'inhibition and interference control', 'working memory' and 'cognitive flexibility' and provide the foundations for 'reasoning, problem solving and planning' (Diamond, 2013, p.2). Teachers reported limitations in dyslexic students' executive function skills, including concentration, disorganisation, cognitive load, working memory and sequencing.

7.2.2.1 Concentration

There was some variance in how teachers viewed students' concentration abilities. Several teachers observed that dyslexic students struggled with focus and attention (P1, P2, P3, P4, P6, P7, P11, P18, P24, P26). Another teacher felt that the attention issues might be related to weak memory skills: 'One minute they might see a note and say 'This is D' and the next bar, they do not know what the note is; that is quite frustrating' (P26). Two teachers (P21, P26) debated whether this was a co-occurring issue like ADHD or the students' frustration tolerance levels.

7.2.2.2 Organisation

Teachers noted that disorganisation (P1, P4, P6, P10, P23, P25) was a challenge with students forgetting instruments, notebooks, music or lesson timetables. The variance in how dyslexic students might cope with executive function challenges is highlighted by one teacher's reflection of secondary school music students:

There are difficulties with organisation, absolutely. Not always, [but] some of the students have built these coping mechanisms of trying to be hyper organised and you look in their folder, it is beautifully arranged and labelled but the content of the work might not be very good to an absolute standard. (P23)

Being mindful not to blame a student's struggles with organisation on being lazy or not caring was also described as an important aspect of offering support.

7.2.2.3 Cognitive load and poor working memory

Teachers recognised that dyslexic students might become overwhelmed with too much information. A participant reported that students appeared to have varying levels of ability possibly based on competing demands on cognitive ability from stress or poor memory issues:

Students come to me they say, 'Oh, but I could play it this morning' and I know that scenario, but this is something [else], the difference between really being almost ready to perform to standard one lesson and then sight-reading and tripping over notes and rhythms the next lesson. (P6)

This could again be a useful indicator of dyslexia as well as another opportunity for a teacher to adjust content and teaching approaches to align with the student's cognitive and emotional state.

Despite findings suggesting strong long-term memory strengths (see Chapter 8) for some dyslexic students, poor working memory was described as a significant challenge:

The working memory, or the bit that you would use if I told you something and then asked you to write it down, can get stretched and gets overloaded quite quickly. So, you can give shorter instructions, short clear instructions and break things into chunks, and it can be very much more effective. (P4)

Therefore, challenges can emerge both within the work during a lesson as well as between and across lessons due to cognitive load and working memory issues.

7.2.2.4 Sequencing

Teachers reported challenges for dyslexic students with sequencing across a variety of contexts (P2, P3, P4, P5, P22, P26). A piano teacher would ask their students a series of questions to establish their sequencing abilities: 'Can they follow a sequence forward and backwards? Do they know what day or month it is? Do they know what time it is?' (P4). Another piano teacher identified this as the challenge of 'working out what has to happen in which order' (P2). Sequencing and pattern recognition are closely related; the following excerpt explains how these difficulties intertwine to slow the speed of sight-reading with scales and patterns in pieces:

Sequencing is a problem. If you are learning a scale, you tend to spot the bottom note and the top note and then play the scale. I've had quite a few dyslexic students who just can't do that. They read every single note and cannot seem to sequence between them. They also can't sequence or recognise similar notes in the music. For example, the music drops to a 'B' but they don't recognise that it was the same as three notes back. It's that recognition of patterns and shapes and remembering the note names that are huge issues. Note names have to be 'written in' all the time. (P26)

An interesting contradiction is noted as patterns are described as strengths for some dyslexic students (see Chapter 8) and as a weakness for others. A teacher suggested a strategy for reminding students of the sequence of intervals in scales:

Using a visual representation of intervals was a strategy developed by one participant in order to help dyslexic students distinguish the tone and semitone interval patterns of scales: 'The dot is the note you play and the T [tone] and S [semitone] are the interval to the next note; it's particularly useful for melodic minors' (P2).

The visual represented here is for the major scale but represents the sequential process:

• T • T • S • T • T • T • S •

Students would write the visuals on notecards and memorise them, and for them this was a more effective strategy than reading the scales on a score.

7.3 Strategies designed to mediate slow processing and executive function deficits

7.3.1 General dyslexia literacy teaching strategies

A minority of teachers were aware of dyslexia literacy teaching strategies and applied them to their music teaching. Another participant with dyslexia training described how multisensory teaching combined 'expressive and receptive teaching', adding that 'you don't just teach them from one direction' (P8) as a means of reinforcing learning. Several substrategies or combinations were described by teachers; for example, one teacher described a strategy of explicit instructions (structured and systematic) which were consolidated through repetition (overlearning) on the piano, stating: 'We worked by mapping out every single step she needed, identifying notes and then moving to them and working out in which order moves had to happen in' (P2).

7.3.2 Modelling or physical demonstration

One multisensory teaching strategy that was described by teachers was modelling. Teachers described the effectiveness of modelling by singing or playing the music for the student. Similar to strategies suggested for students with visual impairments (see Chapter 2), physical demonstration could involve physically moving the students' hands into position.

There are important safeguarding aspects to consider with this approach such as consulting parents or guardians if the student is a minor, making sure that the lesson takes place in an area visible to others and ensuring that the student understands the purpose and is comfortable with it before proceeding.

7.3.3 Simplification of concepts or material

Participants reported that breaking concepts and instructions down into smaller units, or chunking, was a frequently used strategy. A teacher noted: 'We try breaking things down into bite-sized units; this is a good teaching strategy for all students but particularly for students who struggle with processing' (P23). Highlighting and presenting a smaller amount of information, where elements are separated out and dealt with singularly, are strategies which are advocated by the dyslexia literacy literature (Phillips & Kelly, 2016). The teacher reported that 'if I take the processing overload out, then I can get them to see the pattern, but they can't see the pattern if they have everything else swimming around as well' (P5). A piano teacher reported that dyslexic students struggled to remember acronyms like 'All Cows Eat Grass', so they reduced the information needed in order to enable a starting point for understanding:

I just teach them the bottom line in bass clef is the Ground and they can remember that. Then we work up by skipping on the piano. The bottom line on the treble clef is Earth, so you only have to remember two words. Then you can work out all the lines and spaces from that. (P18)

7.3.4 Kinaesthetic learning

The combination of touching notes on the keyboard, hearing them and connecting them to the score (multisensory – visual, aural and kinaesthetic) in addition to reducing the information load were seen as effective simplification strategies. In comparison to teaching non-dyslexic students, one teacher noted that 'The other things that I would notice were that general concepts had to have a much greater number of steps' (P5). Taking time for the student to process information slowly and carefully enables a greater opportunity for accuracy, particularly when identifying note names or patterns. Simplification by slow process was seen as an important way to handle limited automatic processing issues:

I do an awful lot of things slowly because a lot of automatic processing happens by the students and they take the bits they need to and so for a dyslexic student or a student with processing they need some help to find that list of instructions. (P2)

Reducing complexity, highlighting, linking and reinforcing were identified as valuable strategies to support student learning; these strategies might help the student to tolerate frustrations that arise due to task difficulty and primary challenges (Phillips & Kelly, 2016).

7.4 Low self-esteem and mediating strategies

Teacher experiences correlated similarly to research into impacts on dyslexic students' social, behavioural and mental health (Wilmot, 2023; Livingstone et al., 2018; Novita, 2016). Low self-esteem was observed in dyslexic students with 'poor posture and frustration coming out in the lesson, a lot of self-castigation or apologising' (P11). Referring to one pupil who also struggled with coordination issues, a teacher noted:

That was sad actually. He became really grumpy and shrugged his shoulders and got a bit [frustrated]. I mean typical sort of teenage boy, but it was a bit more sort of extreme. (P7)

Another participant noted that psychological issues were 'variable' and some that students were proud of their successes with some 'really good at playing the piano and good at performing and they do concerts and get a boost from that to all sorts of things with low self-esteem' (P7). Another participant stated:

There are all sorts of different responses from students about their own abilities. Some are quite sanguine about it and think "That's just the way I am" and others beat themselves up and need a lot of reassurance. (P23)

This indicates that teachers may encounter varied responses from dyslexic students based on factors such as self-concept, disposition and past learning experiences. The following section discusses general strategies which were used by teachers to counteract issues of low self-concept amongst dyslexic students.

7.4.1 Reducing stress for the student

A teacher reported that when the student was overloaded and needed a respite from verbal instructions, they would use games:

We use 'Music theory Wrap-ups'²⁴, which are excellent for giving them something to do with their hands while reinforcing learning at the same time ... dyslexic students seem to especially love them because they are puzzles. (P21)

The teacher suggested that these were valuable for students as they were a multisensory tool which enabled them to check their own understanding .

7.4.2 A focus on praise, encouragement and activities that give a sense of success

Teachers referred to the importance of praise, encouragement and activities that give a sense of success as a means of reducing stress for dyslexic students. A piano teacher recognised the value of awareness and reflection on how students respond to praise: 'Do they positively glow with pride when you say "Well done"? Do you get the sense that they are not used to having that kind of praise?' (P4). In order to support successful engagement, P25 preferred a simplification strategy rather than attempting to avoid the student's areas of difficulty:

I wouldn't skip the score completely, I would rather simplify the score than take it out of the situation completely. Otherwise, they will struggle more with it (P25).

7.4.3 Considering the pace of the lesson and the students' needs

The pace of the lesson was seen as another important factor, although teachers varied in their responses. Keeping the student's concentration and attention was this participant's goal:

I find myself planning loads and loads of different activities because of their concentration levels, anything that has to do with clapping, standing up, pulse, rhythms, flash cards, and massive cards with rhythms. Otherwise, without breaks, even for half an hour, it just would not work. (P25)

²⁴ https://learningwrapups.com/category/music-theory

Another teacher noted that it was not helpful for the pace to be 'frenetic' for students with attention problems and highlighted the importance of balance in the amount and type of lesson activities (P4).

7.4.4 Creating a positive learning environment

Teachers saw the importance of creating an atmosphere where students could comfortably make errors without feeling a sense of failure. This was described by a vocal teacher:

They've got that feeling of anxiety, 'Is this going to be right or wrong?' I'll get them to sing a tune, so the first one is 'Happy birthday' and I'll be the out-of- tune aunt at the party. I'll sing out the wrong tune over them and they'll cringe and then we swap. They think 'That was horrible' but then they recognise how hard it is to sing out of tune because your brain just wants to follow the pattern. (P16)

Using these strategies as a means of building confidence was reported by a teacher preparing their student for aural test exams:

I encourage them to just guess the melody because it will always end on a perfect cadence and will always end on the tonic. So, I play a little game with them where I don't play the second half and they have to make it up. Then I show them the ending and they are surprised that it is often very close to what they have made up even though they haven't heard it. That gives them quite a lot of confidence that they can work out the second half even if they can't remember it. (P26)

Teachers used a variety of general strategies as a means of overcoming the potential primary challenges of slow processing, weak executive function skills and low self-esteem which were potential markers of dyslexic students. These strategies included multisensory, overlearning, personalised, systematic and structured instruction, modelling, demonstration and the simplification of concepts or materials. Strategies reported by teachers to reduce stress for dyslexic students focused on praise and encouragement, building on their previous achievements, pacing the learning according to the student's need and ability and creating a positive learning environment.

7.5 Coordination and spatial awareness issues

Coordination issues include overlapping difficulties with independence of hands and fingers, coordination of hands and feet, spatial awareness and poor handwriting skills. Judging spatial distances was also considered to be an indicator, as one teacher reported: 'When a person can't judge distances between notes, that might mean they are dyspraxic' (P18). Dyspraxia/DCD (developmental coordination disorder) is described by the NHS (2024) as 'a condition affecting physical coordination' which 'causes a child to perform less well than expected in daily activities for their age and appear to move clumsily'. Teachers recognised that there was complex overlap with different learning differences amongst students, as P26 described 'There were some [issues] with coordination and a big crossover between dyslexia and dyspraxia and ADHD and autism' (P26).It was recognised that beginner pianists might be able to hide some of these difficulties, but when they came to playing more complex pieces 'there might be problems with playing the pedal, coordinating feet or coordinating hands to play together' (P25).

Another participant related coordination to the student being able to control specific movements: 'What is consistent [as a challenge] with dyslexic students is their coordination; it is their eye-hand coordination as well as their independence' (P21). One teacher recalled a game they would use if the student seemed confused about left or right:

With little ones I play this 'Can you pick up my pencil?' game, and you play it quite a lot. 'Can you pick it up with your right hand with fingers one and three?' Make it a game, but if they are struggling with finger numbers or with left and right, that will tell you something. (P4)

While the Dalcroze method was reported to be an effective strategy for students with rhythmic challenges, it was viewed as less helpful for those with coordination issues, according to one participant who stated that: 'You have to make it a much simpler thing and you have to limit the amount of coordination that's needed, and always ask them what they are comfortable with' (P5).

Teachers recognised a lack of spatial awareness with dyslexic students:

Spatial awareness was quite a revelation to me actually. Because high and low (if you think about it) or what we are asking pupils at the piano to do is translate high notes and low notes on the piano in a horizontal axis. Then, to a vertical axis which is high and low ... terminology can be quite confusing. They are taking the words literally. (P4)

In referring specifically to the piano, one teacher noted that students struggled to recognise 'when the melody line went up or down', 'direction on the score and on the keyboard' and with 'keyboard geography' (P25).

Another characteristic which might be used to identify dyslexia in a student was 'poor handwriting' (P23), also known as dysgraphia. It is interesting that when dysgraphia was noted by a teacher, the student did not have difficulty with sight-reading (P7). Another teacher noted that the dyslexic student 'could not hold a pencil; it was excruciating for him and he could not write' (P20). The consequence of this was highlighted by another teacher who stated:

The difficulty with writing [dysgraphia] has a big significance when you are doing theory because they cannot pen control enough to write the notation some of the time; so, instantly, you have to adapt. I did ABRSM Practical Musicianship with those students. (P5)

In these cases, the dysgraphia might prove a barrier for the student. Without an assessment, teacher awareness of reasonable adjustments and accommodations and ways to prepare the student for an exam (for example, with an amanuensis), dysgraphia might limit their ability to complete any written assessment.

7.6 Visual processing issues

Some dyslexic students were reported by their teachers to have visual processing difficulties (P13, P23, P24, P25). One teacher recalled a casual conversation between students, 'One [student] asked the other "Don't the notes jump about on the page for you?" (P23).

7.6.1 Glare or visual stress from the score

This might also include being disturbed by 'fluorescent lights' or the 'glare' of a plastic file pocket, as well as 'lines on the notebook' page (P13). The issue of visual stress has been previously discussed at length in this thesis (Chapters 2 and 7). It is recommended that teachers signpost parents and students to specialist ophthalmologists to determine the cause of visual stress.

7.6.2 Overlays or coloured paper

Teachers' beliefs diverged regarding the use of coloured overlays or coloured paper; research regarding covered overlays has been discussed previously in this thesis in Chapter 2 and in Chapter 7.

I've got students who work on pale green or pale blue paper. Another uses coloured lenses for reading music, a pale yellow. It seems that different colours work for different people. (P23)

Several teachers advocated for using coloured lenses or paper, and most took the approach that even if they were aware that the evidence for them was limited, they would encourage the student to use them if they found them helpful. There was a recognition that most students do not want to use things that make them stand out from others or feel different in any way:

It does make a student stand out to use a pink overlay. It's not nice and makes them feel different. They don't want to feel different; they just want to fit in. (P26)

Others found that parents might not support the student's use of overlays:

I have used overlays and seen them work. Sometimes it is like a magic solution. I gave it to the parents to use at home, but they did not use it, which was such a shame. (P26)

Another teacher reported that students generally stopped using the coloured overlays:

I've had very few of them [pupils] with a colour overlay and they want to use it, I don't have an issue at all. But I never suggest it, because the pupils who tried it have
stopped using it after a while. It comes in from the parental and educational side and then it disappears. (P10)

This correlates with the findings from Chapter 6 which suggests that there is a need for more research to confirm the efficacy of coloured paper or overlays, and that there is a variance amongst teachers' perspectives on their use.

7.7 Strategies specific to sight-reading

Sight-reading involves several potentially challenging aspects for dyslexic students, including the development of fluent pitch and rhythm reading. Teachers reported using flashcards and games or apps as overlearning and simplification strategies, reinforcing note recognition and aural skills. Technology was used to modify scores: enlarging, simplifying or highlighting areas of difficulty. Although the use of colour is referred to as an important strategy in the literature (Hubicki & Miles, 1991; Oglethorpe, 2008), teachers appeared to weigh up the efficacy of the use of colour with the potentially negative effects of a chaotic score on dyslexic students. Even if alternative notation strategies were somewhat effective, teachers were made aware of the challenges of the potential stigma and complexities of transition from alternative to standard notation. A consistent theme amongst participants was that sensitivity and adaptability were needed for each individual learner.

7.7.1 Modified notation

Teachers reported enlarging scores, simplifying them and using colour as means of reducing the processing burden or highlighting trouble spots for dyslexic students. Several teachers found that enlarging scores was a useful strategy for students (P4, P5, P9, P10, P22, P23). Two teachers describe their use of colour annotations:

For dyslexic students, if there are some adjustments on the score, I don't find that they need to struggle that much. I play around with the scores, making bigger gaps between right and left and adding colours, removing things that are not as important, like ledger lines. (P25)

One of my students has a system, I don't prescribe a particular system, but we do talk it through. She has one colour for fingering, one for dynamics, one for notes that

she is getting wrong and another colour for something else. Another student I had highlighted whatever was not working. (P26)

Teachers referred to using software which helped them to modify the score, using apps like *Piascore* (Piascore, 2022) (P9) or *forScore* (forScore, 2022) (P11) to allow them to enlarge the music, annotate it, and highlight it with colours. Software which allows for MusicXML files to be played back to the student was also seen as effective:

I record it with *Flat.io* (Tutteo, 2022) – a web-based notation app – and I send him the web-link and he has it forevermore. It works within Google chrome so it's a really great application as far as I am concerned. You can change note head shapes and colours so it can be used for several different applications. (P12)

Another participant referred to the value of an application playing the sound of the music for the students: 'We have worked with an app and we take a photo of the score, so that the app will play it for him' (P15). One consideration of these strategies is whether they will support the student to learn in an environment where this support is not provided.

7.7.2 Concerns about the use of colours

There were two areas of concern with the use of colours. One participant reported the need for an awareness of synaesthesia and how that might influence the use of colours (P5). Another participant noted that some students only want to mark the music temporarily:

And with my dyslexic adult student we use erasable highlighters because sometimes it's the anxiety of 'If I mark this, it's not going to go away'... So, then we highlighted the whole line green and we did that every time. And then she stopped missing it and then we stopped having to highlight it. (P22)

This suggests that students might use these highlighted sections to reduce their processing burden and to act as reminders when they are learning a piece, but their use might become redundant once the student no longer needs the reminder.

7.7.3 Figurenotes

Teachers reported that alternative notation systems, particularly those using colour and shape were helpful for dyslexic students (P20, P27). The use of colours and shape might

reduce the processing burden of reading music. Only one participant had experience with Figurenotes (Drake Music Scotland, 2024), an alternative notation system:

The literal nature of the design of Figurenotes as it is colour and shape based, along with the note being as long as it looks, is a fantastic tool for learning music. It also lends itself to creative play, which is ideal for younger pupils. I find that the colours of Figurenotes in stage 3 alone, can help a learner with standard notation identify pitch easier. (P27)

Figurenotes is designed to transition students from Figurenotes notation (Stage 1) to Figurenotes on a stave (Stage 2) and the use of coloured noteheads on a stave (Stage 3) (Drake Music Scotland, 2024). This teacher related their varied experiences of moving students from Figurenotes to standard notation systems using these transitional stages:

I personally haven't noticed any problems transitioning from stage 1 to stage 2 and had a 7-year-old pupil who really took to this very well indeed. I've had less experience with transitioning in my teaching from stage 2 to stage 3 with pupils with dyslexia. One of my dyslexic pupils refused to move and didn't want to attempt stage 3. (P27)

Another teacher was sceptical about modified notation systems, stating:

You get quite a lot of people saying that they have a notation system with dyslexics. And I just smile and think, 'Well, you are still going to have to learn to read music'. My own view is to keep it as near as possible [to standard notation]. (P5)

This suggests that not many teachers are aware of alternative notation systems and that more research should be done on their effectiveness for dyslexic students and to reduce stigma related to their use. It also indicated that teachers should be attuned to the individual preferences of students and ensure that they work with them accordingly.

7.7.4 Games, flashcards and worksheets

Teachers referred to a variety of methods to establish note recognition with dyslexic students including flashcards, worksheets and apps. Teachers used 'flashcard work to try to

get her to read the notes' (P17) or made individualised flashcards (P2). A dyslexic teacher expressed caution in relation to the use of flashcards, stating that:

One of the issues dyslexic people can have is that [the letters] p, b, d, and q can all look the same depending on how you're feeling so [the notes] e, g, b, d, and f might all look the same on the G clef (P22).

A teacher described using a series of worksheets containing either line or space notes:

Especially with dyslexic students, they don't understand the movement on the staff in comparison with the movement on the keyboard because the movement on the keyboard is so much bigger than the movement on the staff. It doesn't help the time it takes to transfer what they are seeing to where it is on the keyboard, but it reinforces that one note at a time. (P21)

This area of confusion between the spatial understanding of high and low as related to the score and the keyboard is mentioned further in the section on spatial awareness (Section 8.5).

Note-reading applications to build note recognition skills included '*FlashNote Derby* [Bartolomeo, 2019] and *Piano Tutor*' [SmileyApps, 2020] (P2). Flexibility and cost were key factors for teachers, as described by the following participant:

Music tutor [JSplashApps, 2021) app is brilliant because they can take it home and do it on their parent's phone. You can limit and adjust ranges, so it is great for all ranges of students. It's really good on a giant iPad and it's also free. (P24)

These points suggest that teachers are keen to support the ongoing development of reading skills within students' individual practice sessions as well as during lessons.

7.7.5 Interval recognition

Interval recognition (P5, P10, P14, P17, P18) was seen as an effective method of counteracting the challenge of identifying each particular note in a score of music. This was reiterated by a piano teacher who noted that finding the first note was the starting point: 'I try and teach using the chord shapes and intervals and I found that my dyslexic pupils have

great success with a melody in one hand and chords in the other' (P10). A piano teacher created individual note-reading using flashcards where 'the top of the card has one note and then underneath is a four-note pattern beginning with that note' (P17). Another piano teacher related how the Kodály method of going from the known to the unknown was useful for systematically building up interval and pattern recognition (P5).

7.7.6 Aural interval recognition

Learning to recognise pitches aurally was seen as an effective method of improving aural test skills:

I was taught creative play and that helps you to learn things a lot easier. Making a game; finding a song the student knows and identifying the interval in it. Most people switch off to aural training or theory, but we will play 'Name this song' and we do that for ages until they can name every song comfortably. Then I change the name of the song to the interval and they've done it. (P24)

A vocal teacher referred to a numeric system of teaching intervals which they felt had been effective in teaching sight-singing to dyslexic students:

I teach by number, so the students learn the intervals relationally. I don't talk about theory too much; it just seems to make more sense to them if I discuss numbers. So, they sing, whichever key it is, on 1-2-3-4-5-6-7-1, not on solfège, but on numbers. (P20)

Caution and sensitivity in using this strategy with students who are dyscalculic might be an important consideration, but the main principle of developing a sense of the relationships between notes in a key remains the same.

7.7.7 Specific rhythm strategies

Strategies for rhythmic difficulties include using the Kodály approach to introduce rhythm symbols, utilising kinaesthetic approaches like Dalcroze to internalise the pulse and rhythm, subdividing the beat or using rhythm syllables to reduce, and games or applications. As one teacher related:

The Kodály approach has been life changing for me, the way it introduces rhythm and teaching. I always use practical things as it has to be hands-on. I've got a box full of pom poms, sticks and boards and stickers and we can put things on lines and spaces. (P25)

Some participants used clapping to establish the pulse (P6, P14) but another participant highlighted the importance of working with the student in order to see which action they preferred (P5). A teacher who was trained in the Dalcroze method described how the visual and kinaesthetic approaches help the student to delineate note durations:

I use tambours in rhythm work - that was from Dalcroze. It was quite a revelation to me actually, when I was tapping a rhythm before. I would tap, say three crotchets and then a three-beat note (Ta-ta-ta-TAAAA). I would have just played tap, tap, tap taaaaaaaaa and I wasn't showing the duration of the three-beat note. Dalcroze teaching is very strict in that you have to show the full duration of the note when you are tapping a note, you have a certain size circle and when you are tapping a longer note, you have to have a bigger size circle. As you are tapping, that is the beginning of your circle and you go around and show the full duration by making that size of a circle. (P4)

Vocalising the rhythm was advocated by P4:

He plays in a soul band and they often 'talk' the rhythm with the drummer, so he is used to listening for beats like that ... It's versatile because you can apply them to any music you like. Try it every which way: speak the rhythm, tap the rhythm, step the rhythm, then you can play it. (P4)

Teachers recognised that there was some variance in the students' and their own preferences regarding the use of syllables or numbers to sub-divide the beat.

The French rhythm names 'Ta-Ta' 'tiki-tiki', I've used them in the past and found them helpful but then I went on to the Dalcroze way of 'walk, jog, stride'. Some people love a numerical way of counting the beat or subdividing it 1 and 2 and 3. (P4) What I don't do is use ta-ta and ti-ti's because I struggle with them; I use numbers. I only use words very rarely, only if there are song lyrics, although sometimes they don't always match the rhythm! For triplets against duplets, I will use 'three little frogs' because that's a really nice one. Apart from that I tend not to use words. (P26)

Games, books and applications

Teachers recommended flashcards, games and apps as a means of developing proficiency with rhythm. The *Flip-a-rhythm* (Nelson, 1998) series allows the student to become familiar with a rhythm whilst also introducing new variations (P4).

The other one I use is a book called *Basic Timing for the Pianist* by Allan Small. It is a sequence of four-bar rhythms in five-finger position, starting with crotchets, minims, semibreves and moving on to quavers and dotted notes, that is really helpful for students who have a problem with rhythm (P2)

An app which built the knowledge systematically and uses an interactive and multisensory approach to embedding the information was described by one participant:

The *Rhythm trainer* [GuitarTabs LLC, 2022) app is also brilliant. They have to tap on the phone to the beat and they adore it. It helps them to gain a sense of pulse and it then gives you small progressive sight-reading sections to follow. (P24)

A piano teacher described using Lego to give students a visual picture of note duration: 'I also found with rhythms, playing games with Lego have been really effective' (P22). Strategies to assist with rhythm difficulties included the use of syllables or numbers, visual depictions of beat duration or rhythmic patterns, vocalising rhythms and utilising flashcards, books and games to build skills cumulatively.

7.8 Summary of findings

• Co-occurring conditions and the spectrum nature of dyslexia create challenges for teachers in identifying the specific cause of a difficulty.

- Although sight-reading is reported to be a key difficulty in the music and dyslexia literature, teachers reported that not all dyslexic students struggled in this area and that there was variance in challenges related to pitch or rhythm reading.
- Slow processing and difficulties with concentration, organisation, working memory and sequencing skills (executive function skills) were primary challenges identified by teachers.
- Teachers reported using multisensory, repetition, cumulative and personalised teaching strategies, adapted from the dyslexia literacy literature. Other general strategies included physical demonstration, simplification of concepts or material and kinaesthetic approaches to learning.
- Low self-esteem was mediated by reducing stress, encouraging the student with praise, considering the pace of the lesson and creating a positive learning environment.
- Coordination issues and spatial awareness created several challenges with translating score to instrument and independence between hands or fingers.
 Dysgraphia might affect the ability to write efficiently for an exam and might affect exam choice or indicate a need for reasonable adjustments. Strategies included sensitivity to the student and an awareness of the amount of time it may take them to overcome the challenges with coordination.
- Visual stress was reported by teachers. The use of coloured overlays varied amongst teachers. Best practice is to signpost the student to relevant professionals to assess their visual difficulties.
- Depending on the areas of difficulty with sight-reading, teachers utilised a variety of strategies including modifying scores, using colours to highlight specific aspects, alternative notations systems, repetition through the use of games or other materials, and the recognition of note intervals either aurally or on a score of music.

 Rhythm strategies included the use of Dalcroze and Kodály methodologies, visual strategies to denote the duration of notes or rhythms, rhythmic syllables or vocalisation and embodiment of the pulse or rhythm.

7.9 Conclusion

Findings from teacher interviews indicate that teachers tended to trial a variety of strategies until finding effective solutions, respecting that each student was a unique individual. The need for sensitivity to the student and an awareness of their past educational experiences was key to helping reduce stress or a sense of failure in the student's learning experiences. Teachers reported primary challenges dealing with students' processing and executive function skills which reflect previous dyslexia research literature. Challenges with poor executive function skills, disorganisation, poor concentration levels, reduced cognitive load and poor working memory were described by teachers.

Teachers recognised student challenges with low self-esteem and confidence. A small number of teachers utilised teaching strategies originally designed to help with literacy issues (Phillips & Kelly, 2016) but applied in a music teaching setting. Many of the strategies reported involved breaking concepts up into small steps, slowing down and simplifying activities or scores of music. Teachers described the importance of developing an encouraging, supportive atmosphere where challenging activities might be included on a carefully measured basis. Specific challenges in the areas of sight-reading, visual stress, coordination and sequencing were noted by teachers with correlating strategies.

REFLECTIVE STATEMENT 4: POWER AND KNOWLEDGE DYNAMICS

RS4.1 Resisting control and valuing unpredictability

This transition came from my perspective of desiring the student's progress in becoming an active learner but needing to acknowledge the challenges of resisting my impulse to control and valuing unpredictability in lessons. As my student experiences had been rooted in a master-apprentice tradition where the teacher directs the lesson and is in control, I became aware of how transferring this control to the student caused me to feel discomfort even as I became aware of the benefits to the student. Kemmis (2006) notes that practitioner-researchers generally aim to 'improve existing techniques rather than question them in any critical manner' and 'to enhance the efficiency of practices rather than evaluate them in terms of their consequences' in learning environments (pp. 460-461). Teachers' attitudes also affect this process, as Southcott and de Bruin (2022) note, stating that 'the need to be right and in charge drives many teachers' and that they expect students not to think for themselves but to 'just comply with the unspoken pattern' (p. 3). In my analysis of lessons, I could sense that I felt justified in approaching lessons based on my past student and teaching experiences.

Drawing again from Bronfenbrenner and Morris' (2006) bioecological model:

When a proximal process involves interaction with another person, the power of the bioecological model is substantially enhanced by including in the research design a measure of the other person's focus of attention on the particular aspects of the behavior of the subject that are presumed, on theoretical and empirical grounds, to be most closely related to the developmental outcome. (p. 813)

This related to my responsiveness in transferring power from myself to the student, and considering factors that might affect both my development and that of the students, with the 'research design' being equated to our one-to-one lesson context. These proximal processes are bi-directional, according to Bronfenbrenner and Morris (2006, p. 800). From a teaching position of sitting in close proximity to the student prior to the Covid-19 pandemic and subsequent lockdown, I recognised the benefit of having more physical space between us and also offering space for them to think or act with minimal verbal instructions in the

lesson, thereby exploring a lessening of my control and influence. Questions that arose for me thus included: 'Who takes responsibility for the learning? What can I learn from the student when I let them teach me?' I recognise that there is a tension between wanting to support the student and trying to invite them to use the space to explore and to try their own strategies for learning. As Liddicoat (2022) states:

In preparing to engage with diversity, teachers need to understand themselves in relation to that diversity and to adopt a critical perspective on their own situatedness, on the situatedness of their learners and on the situatedness of the curriculum they teach and pedagogies they use. (p. 189)

At times, I recognised a tone of defensiveness in my voice when I could see areas that needed to be improved, but the student had other ideas for the lesson. My positionality of having experienced master-apprentice teaching in my own learning caused a sense of striving for 'results' in my teaching, and challenges in letting go of this directed focus in my teaching. My other position as a researcher meant that I felt compelled to document their learning and to try various strategies. These attitudes could impede the exploration and discovery needed for them to engage in deeper learning about music which might need to include times of flexibility and adaptation. Once I took a critical perspective of my own situatedness, I began to see that greater openness and freedom allowed other aspects which were of importance to emerge, for example to learn about the student, their experiences and how I might understand them better.

Dyslexic students have specific vulnerabilities to stress, based on their educational experiences, peers and family support (Alexander-Passe, 2008). Kershner (2021) suggests a connection between dyslexia, learning and stress, noting that excessive stress may have an effect on reducing both neuroplasticity and attentional control. I recognised that school demands or activities needing a high level of concentration depleted my dyslexic students, sometimes resulting in a reduced willingness to participate in certain tasks. I also recognised that our activities during the lesson might be a source of stress for them, and unpicking this could be a substantial demand on my reserves of patience and energy.

Students with learning disabilities may struggle with self-regulation and it is believed that this stems from the student's view of themselves (Goldfus, 2012). Essentially, they must first believe that they can achieve and have confidence in their ability to learn, and I was conscious of needing to convey this in my feedback to the student. I acknowledged the importance of affirming effort and focusing on what was done well before speaking to them about areas needing correction. By giving them control over lesson activities and repertoire, they seemed to grow in confidence in their abilities to learn. By giving them opportunities to share their achievements in recitals or by recording a mastered piece for parents, I desired to see an increase in their self-belief. My hope was that by choosing repertoire and activities based on recruiting their interest, this would serve to motivate them and enhance their learning. By acting in ways that support students' autonomy and in developing more sensitivity to their stress levels, the outcome of this transition appeared to show an increase in their engagement and willingness to utilise metacognitive strategies.

RS4.2 Metacognition

A critical incident occurred during a lesson with Alex. This occurred two months after a national lockdown had been declared in the UK during the Covid-19 pandemic. I was still getting to grips with recording lessons via Zoom and this clip is recorded in speaker mode, rather than a side-by-side mode which I later used.

RS4.2.1 Metacognition: Letting the student lead

I began by asking Alex to set a goal. In doing this, I wanted to give him autonomy in the choice and control over this decision and additionally to observe what he believed was achievable. Alex stated 'I would like to learn a piece ... but by myself' (Video excerpts are not available to readers due to GDPR and participant confidentiality). I felt quite surprised and my first response was positive. However, instead of using this opportunity to observe how he would proceed, I instead suggested several strategies: clapping the rhythm, demonstrating the piece and proposing that he listen to recordings. For me, the most meaningful moment arose when I reviewed the video and began to question why I felt I had to have that type of control over the student's learning process, and why I was somewhat resistant to their request to learn it by themselves. This might be due to the fact that I felt I was doing my duty as his teacher in guiding and supporting him. Alternatively, this could be

related to a reluctance to relinquish control, although this would be contrary to my values of wanting to learn from the student and to promote their metacognition. Having experienced this incident, I now question if I was also contributing to a sense of helplessness, or disempowerment, that some dyslexic students may face when they are treated as if they are not capable of learning on their own. Kerr (2001) notes that whilst a dyslexia diagnosis may be beneficial for funding and education 'the diagnosis may also act as a profoundly maladaptive attribution, inducing learned helplessness in student and teacher alike' (p. 84).

Equally, my reaction might have come from my desire for him to succeed and to have 'quick wins' so that he would not become discouraged. I wonder if I should have praised him simply for wanting to have a go on his own, as that demonstrated motivation and selfconfidence on his part, regardless of the outcome. As these are strengths he has demonstrated in past tasks, this might also reinforce these positive traits and encourage future initiative-taking. From this experience, I learned to be more careful when students assert themselves with a desire to take ownership of their own learning and remind myself to step back when appropriate. This knowledge is useful to me as a practitioner because I want to see my students move from being passive learners to active ones. It is positive that he expresses self-efficacy in instances such as this in our learning journey, and I want to build on that confidence and his desire to be involved, rather than hold him back in any way. As a next step, I asked Alex what his approach might be in learning the piece on his own and encouraged him to verbalise the process of doing this (Video excerpts are not available to readers due to GDPR and participant confidentiality). His suggested strategy was to write some of the note names on the music. Even though he experienced some frustration, I held back from making suggestions and praised him as he moved forward in his own way to play a challenging section of 'Minuet in C' hands together. Although this was uncomfortable for me, I interpreted this as progress for us both, with implications for resisting control in my lessons with other students, not just dyslexic students. For the student, this indicated that he had taken the initiative by articulating his next steps. My next step was to consider what the student might need at this stage in order to feel supported in the next section of the piece.

RS4.2.2 Metacognitive development

Working with students who had different dispositions meant that I had to have a variety of approaches as some students were not as comfortable as others in verbalising their opinions and feelings. According to Craft (2017), one of the root Latin words for education is *educere* which means 'to lead out' (p. 9). I adopted this approach in my teaching as I prefer to find out what is within the student and bring that out than try to train the student into a mould, particularly in light of the diverse ways of learning which might be expressed by a student with dyslexia. In this lesson (Video excerpts are not available to readers due to GDPR and participant confidentiality), I began by offering Ben autonomy in choosing if he would like to warm up with a well-known piece or some improvisation. He chose to warm up with 'Scary Stuff' which he enjoyed playing and was proud of playing as it was an ABRSM Grade Three level piece.

Nicolson and Fawcett (2019) refer to difficulties with procedural learning and suggest that individuals with dyslexia are limited in their ability to build the neural wiring necessary to combine procedural and declarative learning systems; they call this the 'delayed neural commitment' framework. The result is that building automaticity in skills may take longer than with non-dyslexic children, and that sub-skills must be built cumulatively so that the foundation of learning is secure. This is similar to Morrow (2023) who suggests that the traditional methods of teaching music do not allow for skills to be presented and reinforced in such a way as to be sufficiently effective for dyslexic students to build the necessary neural wiring for thorough comprehension. However, Gabay (2021) suggests that 'shifting the load from procedural to declarative processing can enable intact learning in dyslexia' and that interventions 'to help people with dyslexia should focus on encouraging declarative memory engagement' as a means of compensating for the procedural learning difficulties (p. 9).

Though I worked on skills like note recognition and rhythm reading (declarative learning) separately to try to build this automaticity, there were still challenges connecting the note on the score to the keyboard, for example, recognising a G note on the stave to finding the correct G on the keyboard (procedural learning). As a result, Ben would often begin a piece

by sight-reading as much as he could and then commit as much as possible to his excellent aural memory. At times in the learning process, however, there were little bottlenecks to his working memory which could usually be sorted with scaffolded reminders, such as simply saying 'Thumb to E-flat' until he did not need the reminder. I tried to allow space for Ben to think through and to remember these on his own at first, and (at 2:50) I also encouraged him to find his place in the score and take as many cues as possible from the score. When a student with dyslexia recognises that they need help, Goldfus (2012) describes this as 'metacognitive development' and states that 'it acknowledges the fact that the inability to do something is not shameful, but part of the learning process' and demonstrates 'the ability to self-regulate learning' (p. 61). This signified growth in our rapport and dialogue and emphasised the importance of my ability to self-regulate as a teacher by allowing the student space and time to process their own ways of learning.

RS4.2.3 Metacognitive development: Asking the student how they learn

Later in 2020, I offered a number of strategies to support Alex learning the piece, 'Rudolph'; for example, making short videos of sections (Video excerpts are not available to readers due to GDPR and participant confidentiality) and simplifying the score. Even though he persevered to play through the piece, at 1:04 in this video (Video excerpts are not available to readers due to GDPR and participant confidentiality) the result of intense concentration shows him slumping his head forward on the keys. After acknowledging his effort, I asked Alex what would help him to learn this piece (1:14) and his suggestion is 'practising'. We negotiate the number of times he will rehearse it during practice and he suggests making a list where he ticks off each time he has played. This preference for a visual practice reminder seems to be helpful for him as a means of externalising the organisation and motivation for practice. Here he has taken ownership for the solution to learning (practising), the procedure (repetitions of practice) and for the way in which he will keep account for his learning (by a visual practice account). The next step was monitoring and encouraging him to consider if this method achieved the goal and if not, to consider what could we change so that he might be able to master the piece. I felt confident that he would attempt to practise, although he might encounter some challenges, and would likely be able to achieve his goal.

RS4.2.4 Physical proximity and balance of verbal/non-verbal communication

In some of the videos, I am not in the camera frame due to social distancing regulations from the Covid-19 pandemic. The lockdown brought about new ways of working and both the student and I had to adapt. As can be seen in the introductory videos to the students (Reflective Statement 2) and in this lesson (Video excerpts are not available to readers due to GDPR and participant confidentiality) from 2019, prior to the pandemic, I sat or stood near the student and was able to point to keys or places in the music as a form of guidance. Close physical proximity was limited first by the Covid-19 lockdown and then through social distancing restrictions. When lessons began on Zoom, I instinctively felt that my students needed to see me and not be overwhelmed with multiple camera angles as some teachers were using. There were challenges, however, because I could not point to a place on the keyboard or physically demonstrate as easily, and more verbal instructions had to be used. We tried to overcome these challenges from Zoom by making short videos which were shared on a WhatsApp channel with their parents.

Over time I could see the benefit for the student as they relied more on their problemsolving skills and less on me as their teacher in the immediate moment. In later lessons, even after social distancing, I sat or stood several steps away from the student. I also tried to reduce the number of verbal instructions I was giving as I could see that more time was needed to process instructions. In this lesson (Video excerpts are not available to readers due to GDPR and participant confidentiality) Ben is sight-reading his part of the 'Capriccio' duet. I have given him the physical and thinking space to accomplish this, with only a reminder about the B-flat from the key signature and confirmation about his starting place on the keyboard. He took a while to identify the starting note but he immediately corrects his initial answer as soon as he realises it is wrong. After that, he does quite well with following the shape of the music even when it leads to a chord which is correct but does not sound right to him, because it lacks the harmonisation of the left hand. Later in the same lesson (Video excerpts are not available to readers due to GDPR and participant confidentiality), he chose to try to play hands together and made minimal errors and required reassurance from me rather than any correction. I believe that the allowance of physical space and less verbal communication supported the student's metacognitive development, in addition to scaffolded reminders which enabled him to reach this point.

RS4.3 Factors affecting motivation

There are a number of factors which affect students' motivation and resilience, but in this section, I will focus on the students' stress levels as a vulnerability factor and autonomy in repertoire choice and achievement as promotive factors. At times, school pressures meant concentration and attentional levels were low and I tried to be sensitive to that in our lessons.

RS4.3.1 Student stress and teacher flexibility

In this lesson (Video excerpts are not available to readers due to GDPR and participant confidentiality), Alex was very clear that he did not want to do a music theory game or drumming rhythms, even though they were activities he seemingly enjoyed in the past. When I asked him what he would like to do, he said (at 0:50) 'Just mess around and make some cool tunes' and proceeded to improvise with some chords. He mentioned three tests at school and recovering from illness and clearly he needed a non-demanding activity which the improvisation provided. I was glad that he felt he could be honest with me, and I felt that this openness also demonstrated self-awareness and a form of behavioural selfmonitoring. Research findings (Alkhadim, 2022) indicate that dyslexic students experience a physiological state of stress when faced with situations in which they are judged by normative criteria as opposed to non-normative criteria, affecting their cognition and emotions. If Alex made the choice to work on theory, drum to a rhythm pattern or to sightread a piece of music, he would need to act according to the standards of knowledge and skill needed for those activities. However, with improvisation, I speculate that he was choosing an activity which was not necessarily subject to those standards and which released him from a sense of being measured or compared. It was unusual for Alex to choose improvisation and I often found myself encouraging him to 'give it a try' during our lessons. Improvisation may have been giving him a creative release from negative or anxious emotions, helping him to relax. From this experience, I recognised the importance of laying aside my lesson plans and following the student's lead. I also began to include other lower pressure activities in our bank of activities. Music theory Wrap-ups²⁵ which were recommended by a teacher participant, were an example of this and enabled the student to

²⁵ https://learningwrapups.com/category/music-theory

learn at their own pace. I also encouraged the students to reflect on their achievements and to revisit pieces which had given them a sense of accomplishment.

RS4.3.2 Student choice in repertoire

Student choice of repertoire was another important motivational factor. I used repertoire selection as a point of dialogue with the student, wanting to find a balance between expanding their musical interests and learning about the students and their preferences. 'Learning music through diverse repertoire may open self-introspection as well as awareness of others in the world' according to Rotjan (2021, p. 29). Several factors had to be considered when choosing repertoire, for example the difficulty level and the aim in learning the piece. At times, it was uncomfortable to resist the urge to control the students' repertoire choices and to easily accept their use of the repertoire. Some pieces were not learned to a standard for a recital but were played for enjoyment. At times, it was difficult for me to resist the urge to perfect these pieces, which would indicate being in control of the process, but seeing the students' obvious sense of pride and accomplishment with learning these pieces made it easier for me to accept the imperfections.

Alex had a keen interest in video games and in one series of games, Black Flag, a number of sea shanties were played. Students were encouraged to think creatively about their pieces, and Alex chose to put several video game songs into a 'Pirate's Metaverse Medley' (Video excerpts are not available to readers due to GDPR and participant confidentiality) for our summer 2022 recital. We made this video during a lesson to help him with practice at home. Maintaining a steady pulse was always a challenge, and this is sometimes hampered by the time it takes him to find notes or chords, but the piece does move forward and he was very proud of it. The element of creativity in bringing the pieces together seemed to motivate him.

Examples of achievements include an online recital during lockdown where Ben played 'A Shave and a Haircut' (Video excerpts are not available to readers due to GDPR and participant confidentiality) and proudly beams at the camera at the end of the piece. Ben chose to play 'Enemy' (Video excerpts are not available to readers due to GDPR and participant confidentiality) by Imagine Dragons and JID, and coped very well with the large chords and greater amount of movement on the keyboard, something which was fairly new

and challenging for him. I believe the amount of effort he put into practising the piece correlated with his choice to play it and subsequently, this motivated him further. Alex enjoyed James Bond movies and was pleased to be able to play the melody with chords in the piece 'Skyfall' (Video excerpts are not available to readers due to GDPR and participant confidentiality). He found the chords being shown in a pattern on the keyboard useful to his ability to bring them together with the melody. Being able to share these achievements with parents and friends and to perform in front of peers enabled a move on the 'continuum of self-assessment' to 'involvement' and 'confidence' leading to more active learning (Goldfus, 2012, p. 62). As I learned to release control, which was not always comfortable, I learned that I would see an impact from this in my students, increasing their confidence and motivation.

RS4.3.3 The use of technology

Other factors influencing motivation include the modality of strategies used. Harrar et al. (2014) found that comparing dyslexic and control groups of participants, 'results demonstrate that dyslexics distribute attention asymmetrically between auditory and visual modalities, more so than controls; it is difficult for dyslexics to disengage their attention from visual stimuli and shift it to auditory stimuli' (p. 533). Knoop-van Campen et al. (2018) found that children with dyslexia spent more time on multimedia learning than control users overall and would benefit from more audio support to compensate for slow reading speeds. Both of my students enjoyed video games and technology, and I integrated as much as I could into our lessons in a way that I felt would bring about deeper learner engagement and goals. I chose activities that did not include a great deal of text and I tried to give simple instructions or gentle reminders during the activity if they were struggling. It was different to the way that I taught in past lessons and had been taught as a student, but as I saw the students engage and enjoy the content, I could see how it was engaging their interest.

In this example, (Video excerpts are not available to readers due to GDPR and participant confidentiality), Alex identifies the majority of treble clef note names correctly. This program²⁶ is web-based and means that parents do not need to download an application on their computer or phones; however, it does have a timer element which I was aware might

²⁶ https://www.richmanmusicschool.com/products/name-that-note

be a potential problem for some students. I asked Alex about that in this clip (Video excerpts are not available to readers due to GDPR and participant confidentiality) but he stated (at 1:22) 'I think it's better because it makes you make the decision quicker'. I am aware that might not always be the case with other students, and I have other note recognition games which do not contain the element of time. We moved on to a game: Chasing Chase: A Treble Clef Mystery, which utilised his strengths in problem solving, where Alex had to choose the correct word or phrase to match a sequence of letters on the stave (Video excerpts are not available to readers due to GDPR and participant confidentiality). By building from simple note recognition to reading patterns of notes, I hoped to see an improvement in sightreading. Ben reported that he enjoyed playing this Mandalorian note recognition game (Video excerpts are not available to readers due to GDPR and participant confidentiality), but I recorded in my reflective journal that I felt it was too easy for him and needed to have more of a component of challenge. Responses seemed to vary at times and concepts which appeared to have been grasped sometimes needed a great deal more reinforcement. This was frustrating for me as a teacher as I struggled to discern whether this was due to my teaching methods or something I was missing; I also wondered if it related to the students' tiredness and depletion of mental energy. I felt it did help to have some activities which were low-demand and allowed the student to have feelings of success and accomplishment. I dealt with this by checking responses over time and if they were persistently incorrect in a certain area, then I began to explore other methods or reasons why this might be.

Reviewing and explaining time signatures seemed to be an on-going issue, as can be seen in this game, Original Time Signature: Among Us, which we played at the end of a lesson (Video excerpts are not available to readers due to GDPR and participant confidentiality). This caused me to question the most effective way of explaining the foundation of metre and subdivision of beats. According to Morrow (2023), dyslexic students need a visual example of how a whole note is divided into units within a bar. I used the rhythm pyramid idea with both students, but they continued to struggle with the concept. Hammel et al. (2016) suggest introducing macrobeats, microbeats and rhythm through chants and music with movement, allowing the student to internalise these concepts before they see them on a score, and playing a number of games where the student and teacher swap parts. I felt that my students were able to recognise and subdivide beats, but struggled with the

conceptual element of time signatures. I recognised that part of my role was to give students the vocabulary and framework for understanding what they already felt in the metre and rhythm. But, Larsen (2019) states 'it is always worth asking the question whether or not the things "learned " in the first place were worth learning' (p. 1). I observed that the Figurenotes adaptive notation does not include time signatures, but instead places the music within a grid. This provides the same information through a visual representation, rather than a more abstract numerical representation of the time signature.

I chose to introduce Music Mania Mats²⁷ (Video excerpts are not available to readers due to GDPR and participant confidentiality) during our lessons, which are a series of worksheets with no more than five or six music theory questions. These were useful as I would ask the student to complete whatever they could and we would work together on anything they did not understand. This allowed me to observe what they recalled from previous lessons and areas where we needed to reinforce learning. By addressing these questions in this 'bite-size' way, the students progressed as we continued to reinforce the information about time signatures. It was sometimes challenging to determine which areas needed further reinforcement and which areas were secure, as responses might be variable depending on their emotional and behavioural state in the lesson. Although I found this frustrating, over time if I managed to obtain correct answers around 80% of the time then the other 20% likely accounted for days when fatigue and depletion were evident. Learning is complex, so I could only infer if the instruction was successful by looking at a variety of components over time.

RS4.4 Conclusion

In conclusion, this reflective statement has discussed factors relating to metacognition and motivation and reflects transitions in my teaching approach toward reducing control and creating more space for students to actively participate in their learning. With the aim of promoting their resilience by encouraging them and highlighting their achievements, I sought to develop their self-belief and confidence in learning and performing music. Alex's parents gave me a card which indicated that they had seen growth in that area:

²⁷ https://www.piano-together.com/blog



Figure RS4.1 Card from Alex's parents

And Ben's parents passed on a promising report from his school music teacher:

Ben has impressed me with the speed at which he picks up what we're doing in lessons – I understand he has piano tuition outside of school – and that is making a huge difference to the rate of progress he is making.

These outside reports were confirmations to me of what I felt I was seeing in the lessons. Returning again to Bronfenbrenner and Morris' (2006) bioecological model, the microsystem is described as:

... a pattern of activities, social roles, and interpersonal relations experienced by the developing person in a given face-to-face setting with particular physical, social, and symbolic features that invite, permit, or inhibit, engagement in sustained, progressively more complex interaction with, and activity in, the immediate environment (p. 814)

The students were gaining confidence in their music abilities, increasing in motivation and in finding their own methods for learning. We became quite deeply engaged during our lessons, so I had to be very conscious about remembering to stop what we were doing near the end so that the student had time to think through the goals for the following week. This was one of the most challenging aspects. I felt that having them verbalise their goals was a

more efficient way of reinforcing the information than having them write the information in a notebook.

I tried to remedy this by having them set the goals when we finished, for example a scale or section of a piece, and then I added it to the shared WhatsApp chat later. This meant that parents were also aware of the students' goals that week. Sometimes I asked them to send me a video through the week to show me their progress or for the student to ask questions to check their understanding. Regarding the use of technology, I was aware of the limitations of using smartphones during lessons and had to consider safeguarding and practical issues very carefully. I only used this method with parental agreement. Smartphones are a potential source of distraction in the lesson setting and may affect attentional control (Ward et al., 2017).

Though there were times of student stress when initial plans had to be altered, allowing the students this flexibility appeared to offer the breathing space they needed to let their physiological stress levels recover. By being sensitive and responsive to these clues, I sought to adapt and to put the student in control of the order of the lesson, and the intensity or challenge level of many of the lesson activities. The development and growth in this area of releasing control and allowing for space, both physical and mental, enabled growth in my rapport with the students, the students' engagement with lessons and the development of their metacognitive skills.

REFLECTIVE STATEMENT 5: JOURNEY TO TACIT KNOWLEDGE

RS5.1 Introduction

Polanyi (1966) described tacit knowledge in the following way: 'we know more than we can tell' (p. 4); therefore, this type of knowledge can be difficult to verbalise. Drawing again on Bronfenbrenner and Morris (2006), who hold that proximal processes (or the interactions between teacher and student) are the 'engines of development' and are 'bidirectional', repeated interactions lead to development of both the teacher and the student (p. 798). Drawing on an example of tacit knowledge from my interviews with teachers and students (Chapters 6-8, 10), a conservatoire-based teacher (P11) described the inadequacy of literacy-based strategies provided to a student by a learning support tutor:

... he'd been given some kind of table of different things he could do, when he was learning from the music, different kinds of learning techniques like to say the words, speak the rhythm, say the word with the rhythm, putting it together. He said that he hadn't found it that helpful ... Partly because it involves a lot of looking at notation still ... In fact, he does say that the phonetics are helpful and it breaks it down because obviously it's one symbol to one sound system. But my experience with him learning phonetics is we get this slightly robotic sound by sound approach. (P11)

In this case, the goal of achieving a fluent musical performance was partially hampered by the ineffectiveness of the strategies applied and partly due to the student's challenges with reading notation. Fortunately, in this case, the teacher adopted flexible approaches based on the embodiment of sound with gestures, aural patterns, and encouraging memorisation of the music as quickly as possible. The student described by this teacher was interviewed for this research and offered his perspective to the teacher's approach:

I had extra theory at [institution name] ... but they didn't understand how dyslexia works and how to make links to the material. I felt I would forget the key, I didn't want to, I would try, but it just wouldn't stick at all. That's why [name of teacher] is great, he plays silly imaginative games with pictures which link things and that really helps me. With theory I didn't have anything to link to it. I have a good imagination, pictures, I can make links myself, but [teacher name] is fantastic at making links and enabled me to be creative with the links that I'm making to remember. (Ned)

This example illustrates the challenges that arise as learning support tutors may understand dyslexia, but lack an understanding of appropriate music learning strategies, whilst some music teachers might lack knowledge of dyslexia and how to tailor learning in a responsive way to the individual student. However, the teacher's tacit knowledge of music and their reciprocal interactions with the student guided their approach, benefitting the student.

Just as someone may use a physical map to determine the landmarks leading to their destination, I consulted literature and the Universal Design for Learning (CAST, 2018) as a framework to guide lesson planning, but my interactions with students and their parents also led to the development of tacit knowledge. In describing how this tacit knowledge can be 'brought to the surface', Mitchell et al. (2022) suggest that this occurs through 'reflections of the subject on their own practice by explaining what they do and why they do it' (p. 1668) and that tacit knowledge is acquired through 'feedback and the correction of self, or by others' in a 'process of trial and error' (p. 1670). These reciprocal processes with the student led to my development of tacit knowledge which informed future lesson planning and benefitted the work with my students.

RS5.2 Analysis of strategies from the literature on music and dyslexia

From the literature (Chapter 2), I identified general evidence-based strategies for teaching dyslexic students and collated various music-specific learning strategies for dyslexic students (see Chapter 2, Table 2.1). The general strategies have been developed from the Orton-Gillingham (Phillips & Kelly, 2016) approach to teaching literacy and applied to the music teaching context.

In Chapter 2, Figure 2.1 provides a more detailed overview of the Universal Design for Learning framework with checklists to guide students through a process of accessing information, building upon and internalising that knowledge (CAST, 2018). The use of the framework aids in flexible strategy selection for lesson planning; observations from lessons fed back into the checklist to consider ways of differentiating instruction as well as accounting for students' self-regulatory and motivational needs. Figure 2.2 provides a list of

questions to guide the teacher in lesson planning. In the sections that follow, I connect the strategies in Table 2.1 (pages 49-51) with the checkpoints from the UDL framework (Figure 2.1, page 72) to give evidence of using these guidelines in my teaching. A more accessible version of this table is available in a text-only format in the Appendices of this thesis (see Appendix J).

RS5.3.1 Providing options for perceiving the information, language and symbols

In looking for ways to engage and sustain my piano students' (Ben and Alex) interest (UDL Checkpoint 7.1) my focus initially was to provide some simple pieces which they could easily master without notation. I chose *Piano Adventures Accelerated Level One* book (Faber & Faber, 1998) for several reasons: a systematic introduction to the keyboard and notation, enlarged well-spaced notation, as advocated by Flach et al. (2016), as well as tuneful melodies with easily identifiable patterns and harmonious duet parts (UDL Checkpoint 1.1). Students had autonomy in the choice of pieces from the book (UDL Checkpoint 7.1). Pieces learned at an early stage were learned aurally (UDL Checkpoint 1.3), following teacher demonstration and imitation, as advocated by Oglethorpe (2008).

I introduced the keyboard to students through a form of discovery learning whereby the students were encouraged to identify patterns and to explore the registers of the piano. Alfieri et al. (2010) recognises benefits of discovery learning, as students 'interact with materials, manipulate variables, explore phenomena, and attempt to apply principles' and they begin to 'notice patterns, discover underlying causalities, and learn in ways that are seemingly more robust' (p. 1).

The stave and notation were introduced through explicit instruction (UDL Checkpoint 2.1) and through the use of mnemonic devices which students were encouraged to devise themselves (UDL Checkpoint 3.4). McCleskey et al. (2017) notes that 'teachers use explicit instruction when students are learning new material and complex concepts and skills' (p. 23). The Rose report (2009) advocated multisensory teaching as best practice for dyslexic students. Music is a multisensory activity by nature and uses aural, visual and kinaesthetic senses to engage the learner (Phillips & Kelly, 2016). The use of physical manipulative aids is a suggested multisensory means of reinforcing memory (Byrne et al., 2023). I invited

students to use pearl clay (leaves no residue on hands or surface) to re-create the treble and bass clefs (UDL Checkpoint 5.1). I recognise that this would have to be used with caution depending on the sensory sensitivities of individual students (See Figure RS5.2) for an example).



Figure RS5.2 Multisensory activity to form treble and bass clefs from clay

Students' knowledge of music symbols was reinforced through tracing, drawing and games (UDL Checkpoint 3.4). The students used music symbols and notes on a stave to create a code (See Figure RS5.3) which they would then use to write short messages which I decoded (UDL Checkpoint 2.5). I used the Whiteboard function on Zoom to ask them to practise writing symbols (UDL Checkpoint 5.1) when teaching online and a physical whiteboard in face-to-face settings.



Figure RS5.3 Music symbols code, created by Ben (2022-5-19)

The learning was reinforced by overlearning techniques, as advocated by dyslexia educational literature (Phillips & Kelly, 2016), through simple games, for example, naming, locating and playing all of the C notes on the keyboard, locating and playing notes in short sequences, recognising notes on the stave through flashcards and games (UDL Checkpoints 2.2; 2.5).

I trialled notation based on finger numbers, note name tablature and colour representations of pitch (UDL Checkpoints 1.2; 2.3). I modified scores to reduce distractions or to simplify them, as advocated by multiple sources and the use of recordings, as advocated by Nelson and Hourigan (2016). Through a process of elimination, I observed that Alex preferred standard notation with reminders, such as note names written in or sharp notes circled, whilst Ben preferred to learn pieces aurally with scaffolded reminders although he would consult the score until the piece was memorised. Both students used scores involving melody lines and chordal accompaniment, with Alex using a graphic representation of chords on a keyboard diagram and Ben using simple chord tablature, as chord theory developed (UDL Checkpoints 1.1; 2.3). In Reflective Statement 6, I note how the use of an alternative notation system (See Section RS 6.3.2) was effective for Alex, although he had concerns that this might prevent him from progressing with reading standard notation in the future.

I introduced scales through aural patterns, as advocated by Oglethorpe (2008), and then offered a variety of options to help with recall: using note names of the scale pattern, graphic representations of the scale on a keyboard, kinaesthetic pattern on the keyboard and standard notation (UDL Checkpoint 2.3). Alex preferred a graphic representation of scales²⁸, whilst Ben chose to recall scales through aural memorisation and kinaesthetic pattern. Neither student was interested in modifying scores using colour, although I invited them to do so on a number of occasions.

From my observations, I could see that Ben had excellent aural repetition, was able to maintain a steady pulse, needed slower pacing for verbal directions and coped well with duets when he felt confident in his ability to place his part by himself. My reflections of

²⁸ https://www.hoffmanacademy.com/blog/c-major-scale-on-piano/

Alex's early lessons were that he needed more time to establish pulse and rhythm but demonstrated persistence in learning. I used a variety of techniques, isolating the rhythm, as advocated by Nelson and Hourigan (2016), and practising by demonstration and imitation away from the piano, as advocated by Oglethorpe (2008). Their pride in being able to play pieces was an important way to sustain their motivation and often parents were treated to mini recitals at the end of lessons as well as non-compulsory group recitals at the end of term (UDL Checkpoint 7.1).

RS5.3.2 Providing options for comprehension

As students mastered the basic knowledge, they set goals for themselves (UDL Checkpoint 6.1). Choosing new pieces enabled them to have the opportunity to connect what they learned previously and apply it in a new setting (UDL Checkpoint 3.1). I encouraged them to verbalise their thinking in approaching a new piece, which also enabled me to observe their level of understanding, and I used prompts or questions to remind them of elements they might have overlooked (UDL Checkpoint 3.2). Sometimes this involved prompting or returning to key concepts and linking current challenges with what had been learned in the past (UDL Checkpoint 3.4). In this video example (Video excerpts are not available to readers due to GDPR and participant confidentiality), Ben approaches a fairly new piece and finds his starting point. He had written a note to himself at the top of the page 'RH third finger to F' and 'LH thumb to A and little finger to D'. Throughout the video clip, I asked him various questions about how he remembered to start the piece, about the movement of notes on the score and about articulation. The answers to these questions indicated areas needing further reinforcement. Upon further reflection, I can see that asking these questions and discussing the answers together was also a means of reinforcing the learning during the lesson.

Ben demonstrated his knowledge of the concept of 'texture' in preparation for his GCSE music course as demonstrated in this time lapse video (Video excerpts are not available to readers due to GDPR and participant confidentiality) and in Figure RS5.4.



Figure RS5.4 Multisensory technique for reinforcing learning (Ben 2022-2-17)

In our previous lesson, I introduced the concepts through explicit instruction (UDL Checkpoint 2.1). I asked him to recreate the textures using textiles (UDL Checkpoint 5.1). As he is someone who enjoys doing practical things with his hands, he seemed relaxed throughout and took his time to ensure the result was neatly presented. I recognise that not every adolescent would feel comfortable doing this activity, particularly in a group setting. In later lessons, we reviewed the concept by listening to music excerpts and I assessed his ability to identify texture in the music (UDL Checkpoint 3.4). This was a concept which he seemed to recall and understand with a good degree of accuracy.

RS5.3.3 Providing options for physical action, expression and communication

From early pieces, we used 'chunking' strategies (Checkpoint 3.3) to break music down into smaller sections. Ben and Alex used these strategies frequently: playing more slowly, hands separately and working to master smaller sections at a time (Checkpoint 4.1). Neither Ben nor Alex felt comfortable with singing, so I did not ask them to interact with the music in that way. I invited them to improvise and to compose using manuscript paper at first, and later, notation software, Flat.io (Checkpoint 4.2; 5.1). Alex's first composition 'Mose and the Cat' was created from simple note names with numbers to denote repetition (See Figure RS5.5).

the QF Cd り事 h Eb

Figure RS5.5 Alex's composition from 2019

And later, he demonstrated his ability to draw the notes on the stave (See Figure 7.7):



Figure RS5.6 Alex's composition from 2021

Ben enjoyed improvising freely. Although Alex was generally reluctant to improvise, occasionally he did show a preference for it as an alternative to activities requiring more concentration. And in their interview, both students mentioned that they enjoyed composing with Flat.io (UDL Checkpoint 5.2). They were able to choose instruments, clefs, tempo, key signature, time signature and Alex particularly enjoyed using the playback function as feedback (Video excerpts are not available to readers due to GDPR and participant confidentiality). These pictures and videos demonstrate the development of these composing skills and increased confidence over time (UDL Checkpoint 5.3). Ben used notation software to simplify the score of a piece he was playing (Video excerpts are not available to readers due to GDPR and participant confidentiality). This was a useful tool as knowledge was used from a different angle and this sometimes revealed areas that needed reinforcement (UDL Checkpoints 5.1, 5.2, 5.3).

Even though I felt we did quite a lot of repetitive activities, particularly related to rhythm and note recognition, after longer school holiday periods these concepts needed to be revisited and rehearsed until they were secure again. With UDL as my framework and as my tacit knowledge of the students developed, I reflected on the use of repetitive activities and students' motivation. In future lessons, instead of feeling disappointed about this, I was careful to ensure that the student understood it was fine if they needed to refresh their memory in certain areas. I recognised that the balance in lesson activities needed to be weighted toward achieving their goals of playing specific pieces interspersed with brief, focused sessions of repetition.

RS5.3.4 Providing options for executive functions

At the beginning of each term, I asked students to describe their goals and these were also shared with parents; I gave the students feedback at the end of the term and included this on our shared WhatsApp chat groups with their parents (UDL Checkpoint 6.1). If the goal chosen was unrealistic, we discussed ways to simplify or modify the piece, as has been demonstrated in previous sections. In advance, I would plan questions or prompts to encourage them to think about specific aspects of a piece (UDL Checkpoint 6.2). For example, this excerpt is from a lesson plan (2020-6-16):

- Practise this piece as it is written and then improvise changes to the rhythm or pitches. Draw pictures or a diagram to show me how you would change it.
- Go through the piece and tell me how many one-beat notes there are in the right hand in the whole piece? (Answers on the whiteboard) How many half-beat notes

are there in the right hand? How many beats does the first note in the left hand have? Tap the rhythm with the drumsticks.

- Left hand can be simpler than the one we played last week. Use chords C-G and F-C or you can just play C or F. See which Ben likes better.
- In bars, 9 and 10, there are some new chords. Challenge for this week: 'Using what you know about the lines and spaces, see if you can work out what notes these are'. Write them down and try to play them.

In earlier lessons and during the pandemic, I planned a number of activities for each lesson. However, as the students progressed, I tailored the lesson to their specific needs during that lesson, particularly related to stress levels, as well as encouraging them to identify any barriers. Resources were kept on a Google drive shared with their parents during the pandemic, but this was not used by the parents or students once face-to-face lessons resumed. To do what worked practically, I provided recordings, PDF copies of pieces and brief notes on the shared WhatsApp group. I sent recordings of achievements from lessons as well, whether that be a challenging line or an entire piece. These provided a way for students to self-reflect on their playing and to observe how they incorporated feedback. We discussed this process in subsequent lessons. Parents were also aware of the students' goals and progress.

RS6.3.5 Providing options for recruiting interest, sustaining effort, persistence and selfregulation

Students had individual choice about their repertoire and the order of lesson activities, and they were aware that lesson plans and goals were flexible (UDL Checkpoint 7.1). Generally, they were aware of what they needed to do for practising through our discussion and shared notes, but at times, parents asked clarifying questions throughout the week. This dialogue helped me to observe whether students indeed understood what they needed to practise, and assisted parents in feeling knowledgeable and supported in helping their child. As mentioned in previous reflective statements, the environment of the lesson was designed to be calming and low arousal, and our lesson times and days remained the same so that the students could have a sense of predictability and reliability.

By optimising their own choice of repertoire and considering their interests, particularly those related to video games, this provided the students with a variety of different ways to explore and to participate in their own learning (UDL Checkpoint 7.2). Performances which exhibited effort were affirmed and acceptable, even if that meant the piece played was a close approximation of the actual piece (UDL Checkpoint 8.2). They were invited to play at recitals with other students in my teaching practice and both students performed well in front of an audience and seemed to find it a motivating experience. As mentioned in Reflective Statement 5, I strove to give students feedback in a way that emphasised their efforts, persistence and improvement. Recognising their reluctance to identify and acknowledge their strengths, I took opportunities to recognise what they had achieved and looked for ways to build on that growth (UDL Checkpoint 8.4).

In this thesis, I have already identified a positive correlation between my development through transferring power to the student and focusing on students' strengths with their confidence to develop metacognitive strategies (UDL Checkpoint 9.1) and to take breaks as necessary or to choose low pressure activities as a means of stress management (UDL Checkpoints 9.2; 9.3). Breaks were offered if I sensed that they were tired or depleted (UDL Checkpoint 7.3). By developing an empathic understanding of their challenges at school, I learned to recognise when they needed emotional support by talking through their challenges in a non-judgemental atmosphere. The aim of this empathic awareness was to reduce student stress, a factor which had implications for their self-regulation and emotional well-being (see Reflective Statement 7).

RS5.4 Conclusion

Mitchell et al. (2022) describe how 'the role of tacit knowledge cannot be underestimated because it gives meaning to explicit knowledge' (p. 1667). Using reflective observations and students' feedback, both during and after lessons, enabled the development of flexible student profiles which served to inform my future lesson planning and actions. Parental feedback, usually in the form of clarifying questions about practising or factors related to the student's school experiences or emotional state, was an integral part of this process and allowed me to correct errors in my judgement or misunderstanding of situations. In returning to the analogy of a map in this reflective statement, my aim was to articulate how

the landmarks of strategies from academic and pedagogical literature and the principles of the Universal Design for Learning (CAST, 2018) along with my interactions with students and their parents, led to my development of tacit knowledge; I saw the growth of my students and development in my own teaching. This growth in the awareness and use of specific strategies as well as the collaboration with students led to an increase in my teaching confidence to utilise these strategies and to plan lessons in a more accessible way for all students in my teaching practice.

Chapter 8 FINDINGS FROM TEACHER INTERVIEWS: STRENGTHS

8.1 Introduction

A focus on remedial and deficit-based strategies has previously driven music education approaches for dyslexic students (Oglethorpe, 2008; Miles et al., 2008; Morrow, 2023). Participants in the present research reflected this, being overwhelmingly focused on student challenges and corresponding strategies as compared with student strengths. However, a more positive view of dyslexia and related strengths is emerging (Eide & Eide, 2011; Malpas, 2017; Reid, 2016; West, 2009). Some research suggests that these strengths are compensatory mechanisms to mediate the deficits caused by dyslexia (Yu et al., 2018; Geschwind & Galaburda, 1985), others posit that strengths are related to the use of alternative methods of processing information during formative years (Tafti et al., 2009). A difficulty in making skills automatic (discussed in Reflective Statement 4) is likely caused from the use of declarative, rather than procedural learning, but this dependence on declarative learning may convey a benefit in allowing for an openness to new possibilities, or an explorative strength (Nicholson, 2015).

Literature on strengths related to dyslexia, including creativity, high-level reasoning strengths and character strengths, can be found in the literature review (Chapter 2) of this thesis. In the music education literature, strengths have been acknowledged, although this is somewhat limited. In the following table, a comparison of dyslexic strengths in the music literature indicates that visual memory and problem-solving skills were noted amongst three out of four of the sources. Aural memory, aural skills and kinaesthetic memory were each described by two out of the four sources. The remaining categories indicate some overlap, for example tactility may be connected to kinaesthetic memory and motor skills/strengths. Spatial aptitude and the ability to perceive multi-dimensionality may be related to visual memory skills. Just as there are complexities in determining what might be the cause of a challenge for a dyslexic student, this is mirrored in understanding the causation for strengths.
Strengths reported in the	Oglethorpe	Miles, Westcombe	Rolka &	Nelson &
music literature	(2008)	& Ditchfield (2008)	Silverman	Hourigan
			(2015)	(2016)
Kinaesthetic memory	х	X		Х
Spatial aptitude	х			
Aural memory	х	x		Х
Aural skills	х	X		
Music composition	х	X		Х
Improvisational skills		X		Х
Visual memory	х		Х	Х
Verbal skills		X		
Motor skills/strengths		X		
Problem solving	х	x	Х	Х
Recognising patterns		x		
Intuitive/Insightful				
Vivid imagination	х		Х	Х
Ability to perceive multi-	Х		Х	
dimensionality				
Empathy				Х
Determination	Х	X		
Performance skills		X		Х
Creativity	Х			

Figure 8.1 A comparison of the music literature on dyslexic strengths

A study of teachers and dyslexic schoolchildren found a significant correlation between teachers' expectations of dyslexic students and the student's achievements (Hornstra et al., 2010). Teaching that is focused on a student's strengths might change the teacher's perspective of the student and the student's perspective of themselves and may motivate them to continue during times of frustration. Untangling any bias or assumptions that we as teachers hold about dyslexic students or their capacity for learning music, or our ability to help them learn, and instead re-focusing on their strengths might allow us to reframe our expectations for their progress. The following sections contain themes which emerged in my interviews with instrumental and vocal teachers: creativity (8.1.2), verbal strengths (8.1.3), aural strengths (8.1.4), memory strengths (8.1.5), an ability to follow detailed instructions (8.1.6), strong performance skills (8.1.7) and persistence (8.1.8). Teachers' views aligned with the strengths identified in the literature.

8.1.2 Creativity

Teachers reported creativity as a strength for dyslexic students (P1, P3, P4, P7, P9, P14, P15, P21, P24, P25, P26, P27). Some teachers referred to benefits associated with creative thinking such as problem solving and a vivid imagination. Teachers noted that there was a uniqueness to dyslexic students, with P26 stating that 'There is a creativity and, I hate to use the cliché, but a "thinking outside the box" really'. P15 described this, stating 'It's about solving problems, and being creative, thinking big and globally, inventing things, seeing solutions, and new ways of doing things'. This can be a disadvantage as well as a strength when it comes to a musical score, as P21 noted:

They are highly creative and three-dimensional thinkers. This is one of the problems because the score is two dimensional. The keyboard is three dimensional and makes much more sense than the score. (P21)

This suggests that interaction with the kinaesthetic and aural patterns of the keyboard might prove useful as part of the students' learning process. Suggesting that this creativity was a marker of dyslexic students, the teacher remarked that 'I can see who is dyslexic without knowing how they read or write; I can feel how they innovate or are creative in a certain way' (P15). This seems to indicate that teachers with more experience of teaching dyslexic students recognise specific aspects in the students. Another teacher related an example of this multi-dimensional thinking:

There was a student who came in one day and said 'Do you realise that the letter names of your scales on the Circle of Fifths, if you add them together, they always add up to seven. For example, C major has no sharps or flats, C# has seven or B major has five and B-flat major has two.' I had never noticed that pattern before. So, now we have the 'Seven Rule'. (P21)

P5 also identified students' original thinking: 'I found with my dyslexic students, they really do view things in a very quirky way'. Teachers likewise reported students' use of imaginative improvisatory and composing skills which were also used as memory aids for recalling pieces.

8.1.2.1 Vivid imagination and the ability to visualise abstract images

Grant (2010) describes the ability to use visualisation as a compensation for poor memory. Some of the interviewed teachers in this PhD research would deliberately invite pupils to come up with a story or scenario or image to recall or improve their interpretation of phrases or sections of a piece of music. By doing this, they may be more likely to recall the dynamics and articulation of a piece. It seemed that teachers recognised this as an effective strategy for students and encouraged its use in different contexts. P25 described this vivid visual imagination as the student having 'an image and story in their head' (P25). Another teacher related how a dyslexic student came up with the story behind a piece, describing:

She's started thinking about what this story might be behind it. She came back the next week and it was like a sort of film script. 'He's left me, woe is me, but now I'm strong'. (P6)

Another student would use their imagination to draw connections in the score:

I had this student who would connect dots on the score and make pictures. I would give him extra copies. We would work on the shape of the melody while he was adding shape and colours. It was brilliant. I find the imagination and the 'thinking outside of the box' so creative. (P25)

Using composition and inviting students to verbalise their thoughts after listening to music were strategies used by P15:

I think that by creating music and doing it early, the dyslexic students are quite good at that. Composing their own tunes. I think that's important. Also, thinking in pictures ... Dyslexic students are usually good at that, but it is very good for nondyslexic students to practise this as well. (P15)

This engagement with the music in a way that the student enjoyed enabled them to create their own music, practice using the score in a way that supported their creative process and encouraged them to think about things like where pitches are placed and why rhythm and time signatures are important. This also served as a means of giving the pupil more of a sense of ownership which helped to motivate them.

8.1.2.2 Improvisatory abilities

Leveroy (2013) found that dyslexic actors excelled at improvising, feeling free from the constraints of a script. Further to creative thinking and problem-solving abilities, teachers reported that improvisatory strengths were also observed in their dyslexic students (P1, P2, P3, P4, P5, P14, P19, P23, P24, P25). A participant mentioned a contrast between dyslexic and non-dyslexic students and the use of parameters in improvising tasks:

I like working on improvisation with dyslexic students as well, because they are not scared of it. The rest of the students are happy with sight-reading, but they are scared of improvisation, but it is different with dyslexic students. We start with pictures, create the whole story, and then I usually accompany what they play. We do set some parameters, 'You can use these keys in any octave' or 'How is the character of the song going to be according to the picture? (P25)

Another teacher observed 'They tend to be better at memorising and improvisation and more musically creative than those who have always been wedded to the score' (P9). An instrumental teacher reiterated the ease they observed in dyslexic students when using improvisation:

Some of my students, after they have approached a scale in an improvisational way, when they approach a piece in that key they will start adding bits in. They learn it [in order] to use it. (P24)

Musical improvisation was viewed similarly to the actors described by Leveroy (2013). P23 reported that:

From time to time, we get students who, because they really struggled with notation, have a real gift for improvisation. The fact that they have to trust their ear

more than their visual processing has led them to approach it in that way. It's not universal, but some children have had the most difficult challenges, but because they have got this facility in handling sounds imaginatively, they are able to produce something quite sophisticated. (P23)

Another teacher reported the use of improvisation as the student created a composition:

So, his composition was purely done by him improvising and then finding those patterns on Sibelius that he played around with. That's how we did it. We did it via improvisation, recording, and then we put it on Sibelius. (P5)

Describing their use of improvisational activities to create a more relaxed learning environment and the impact on students, P10 stated:

I knew that jazz was an improvised style, and if it's improvised then it doesn't matter if you get it quite right. This took away some of the stress. I could start working with jazz pieces at [private school] and I could say 'It's not what is written, but I like it' or 'what do you think about that bit?' and I didn't worry one bit if it didn't match the written page, unless it was for exam purposes of course (P10)

P16 described how improvisation was used to '... soften any panic or performance anxiety that can develop ...' (P16). This reinforces the idea that dyslexic students might choose improvisatory music-making over an activity like sight-reading in which the numerous processing demands may cause them to feel overwhelmed. Again, they may experience more control in this type of activity, as well as ownership, motivation and a sense of achievement.

8.1.3 Verbal strengths

Teachers reported that some dyslexic students had strong verbal abilities. A piano teacher described their dyslexic students as 'chatterboxes; they will come up with a story for the whole piece of music and give it character' (P25). P7 felt that their student would have been able to respond to verbal questions, rather than text, to pass a written music theory exam:

He just couldn't express himself in written form. I think probably if you sat him down and asked him the questions verbally for Grade Five theory, he'd be able to do it; but he just wasn't really able to sit down and complete the exam for two hours. (P7)

A teacher referred to the importance for some students in vocalising or verbalising what they are learning as they touch the score or the keys as a multisensory strategy (P8). Mills (2018) states that 'This kinesthetic approach in conjunction with hearing, saying, and cognitively processing causes students to use both sides of the brain to process (verballinguistic and visual-spatial) rather than rely on just one neurological route' (p. 39). A teacher described how encouraging a student to use their verbal skills enabled them to get a sense of the student's level of understanding: 'I often get kids to repeat back to me what they think I've asked them to do' (P4). It might be suggested that strong verbal abilities are one of the compensatory mechanisms used by some dyslexic students. Several studies (van Viersen et al., 2015; van Viersen et al., 2016; Moojen et al., 2020) suggest that higher verbal abilities are a compensatory and protective factor which may allow dyslexics to compensate for their difficulties with text. Other research (Foley Nicpon et al., 2011; Berninger & Abbot, 2013) suggests that a mismatch between exceptional verbal abilities and a specific learning difficulty may be the cause of the misidentification or lack of assessment in some dyslexic students, resulting in poor educational accommodations as well as an increase in negative psychosocial factors. This underscores the importance of training to inform teachers as to possible indicators of dyslexia.

8.1.4 Aural strengths

Several teachers identified aural strengths in dyslexic students, including aural memory and good pitch perception (P10, P12, P14, P17, P18, P20, P21, P22, P23, P25, P26). One participant noted 'I would find the student's quickest way to learn; with a dyslexic student, it tends to be the ears or the body versus the eyes' (P20). Another teacher suggested that the strong aural memory was a compensatory device, stating that 'They may develop an aural strength, but it is probably memory based' (P10). Illustrating how an emphasis on aural strengths might influence teaching strategies, a participant stated:

If you have a processing difficulty, rather than trying to memorise it by reading it, it might be very hard. They might prefer to memorise it aurally. (P23)

It was noted that some dyslexic students had accurate pitch recall:

He's got incredible pitch. If I gave him a B-flat and asked him to sing a D, he would. He can visualise that note, so he has very good relative pitch. (P12)

A teacher described their student's impressive ability to learn a piece after hearing it:

They will bring me a piece of music, and I will think in my head 'Not possible'. I had one lad who came to me at the age of nine and taught himself to play Chopin's *Fantaisie Impromptu* from watching it on YouTube. Okay, it wasn't a perfect rendition, but it was pretty good! He had dyslexia but he had an amazing ear, good dexterity and good memory. (P18)

This process may have been supported by the student's visual attention to the movement of the hands on the keyboard as well as through aural learning. However, teachers were keen to connect to students' aural strengths. A vocal teacher referenced the use of aural abilities to detect and correct errors:

Instead of analysing in terms of F sharps, G [notes], key signatures and notation, we will meaningfully make the correct sounds a few times and then insert that meaningfully correct bit of sound-making into their music rather than thinking of the notation and they seem to be quite happy doing that. (P11)

Highlighting the use of aural skills over sight-reading abilities with a particular student, a piano teacher stated: 'I think when he had an aural picture he was able to quickly recreate that on the piano in a slightly more successful way' (P7).

P18 found that there needed to be a balance between relying on aural memory and more challenging skills:

I would teach them pieces by rote, but we would do sight-reading as well but as a different issue and eventually the two will overlap. We have to build on what they can do. Some teachers will say 'I will not play it for you first'. You have to make the most of whatever gift they've got and then work on the things they are not. It's difficult, of course. I will usually alternate; when they first come in I will ask them to

play whatever they are bursting to play for me. Then, I will tell them how wonderful it is and we will tweak it. Then, we alternate that with something they might not want to do, scales or whatever, and then they can play me something else. For a short time, we will do sight-reading. But maybe not a whole piece, just a line of notes or just working out the notes. (P18)

This suggests that teachers recognise aural memory strengths but realise the need to encourage the student not to avoid technical work or sight-reading which they may find more difficult. Based on these findings, there appears to be a close connection between aural and long-term memory strengths in dyslexic students which may point to this as another compensatory mechanism to protect against difficulties with sight-reading.

8.1.5 Memory strengths

Long-term memory and kinaesthetic memory strengths were identified by teachers in this research. Teachers recognised students' challenges with working memory and the impact that had on recalling the labels of symbols and notes when needed.

8.1.5.1 Long-term memory strengths

Some teachers (P10, P12, P17, P18, P26), noted their dyslexic student's strengths: 'Long term memory is one of them; once something is known, it is known' (P26). Another teacher described how their student's 'memory for music is really good; the more they play a piece of music, the less reliant they are on the score' (P12). However, a piano teacher noted that this was not always the case, and felt that complexity and number of pieces should be limited:

You can't have pieces that take them months to learn because they are unlikely to remember them. You can't do more than one piece at a time, although it depends on the person, but you should be prepared to concentrate on one piece. Two can be too many sometimes. (P10)

Findings suggest that this may indicate that there are some considerations for teachers related to their choices of repertoire and lesson/curriculum planning. The concept of a memory strength as a compensatory technique related to aural strengths caused a participant to reflect that 'They tend to be much better aurally; they don't read well, but

they can play well by ear or can memorise well' (P18). There are positive as well as negative benefits of this, as two teachers (P7, P10) referred to the difficulty they had in helping students unlearn errors which had been committed to long-term memory systems.

She was this really bright, really hard-working girl who was really smart and a really good oboist, good singer, good pianist. But the most frustrating thing about her piano playing was that she would really sort of 'learn in' mistakes (again it was this relationship to the score) which you could never extract. (P7)

This correlates with Oglethorpe (2008) who stated about dyslexic students that 'once something has been committed to the long-term memory it is almost impossible to erase it' (p. 6). These findings suggest that although many dyslexic students appear to have strong long-term memory systems, the importance of supervised learning and strategies which strengthen their metacognition might prevent the ingrained errors which can occur.

8.1.5.2 Kinaesthetic memory strengths

A strong memory for the kinaesthetic shape made by the student's hands was observed, particularly by those teaching piano:

You might show them a chord on the piano and show them the shape of the music and they associate the two. I have a dyslexic student aged seven, and she has all the classic signs. As long as she knows what one of the notes is, she can find any chord shapes with her left hand, which is mostly where the chord shapes are. (P10)

I quite frequently will get pupils to play with their eyes closed. 'Let's have a challenge'. You are simply doing it by feel. (P3)

This correlates with the spatial aptitude strength in the literature (Oglethorpe, 2008). A piano teacher had a different view of this kinaesthetic memory or 'muscle memory', as they described:

This is very important in my teaching, that 'muscle memory is not a thing'. Providing you are capable of playing the passage, when you do it 'using muscle memory' (i.e. when it feels like you are on automatic pilot) it is because there is a neural instruction set off by something. (P2)

For a teacher who themselves may not rely on this form of memory, it might be difficult to appreciate how important and useful it is to a dyslexic student. Time and patience might be required as the student becomes familiar with the feel and shape of the movements in their hands and fingers.

8.1.6 Ability to follow detailed verbal instructions

Teachers (P2, P5, P11) reported finding that dyslexic students were receptive to verbal instructions. One participant described this:

It was absolutely phenomenal how many of these specific instructions she could take on board. And I think there's a combination of recognising that some of these instructions are things that people can have and it comes more easily, they just naturally put together this group of instructions, just piecing it out step by step. (P2)

This may suggest that the teacher was scaffolding instruction to mediate the student's working memory skill limitations. These points suggest the importance of knowing the student in order to achieve effective learning without overloading them, and the value of clear and targeted information for the student. It also suggests that teachers need sensitivity and patience to know how to assess the student's ability to perform without reminders.

8.1.7 Strong performers

Nelson and Hourigan's (2016) research sought to establish an understanding of the experiences of adult performers with dyslexia; the fact that they were professional musicians presupposes some strength in the area of performance. In this research, some dyslexic students were reported by teachers to be strong performers (P3, P4, P7, P11, P23, P26). It was observed that 'They will have an intense creativity with the music; with interpreting the music' (P26). P4 considered that:

Whether they can clap a rhythm back or if everything else is a struggle, but once they learn something they can nail it and never forget it or maybe they perform with real panache. When they've got something, no matter how difficult it is, they can perform it, playing with a lot of energy. (P4)

Connecting memorisation skills to effective performance, P11 stated:

It's been very important to me that I feel that they can deliver great musical performances. So, I get them to memorise as soon as possible. I just really think I've seen that some of these singers, they can really sing brilliantly. And they can start to learn much more quickly so it's great to try and get them past the dyslexia. To unlock what they can do. I just don't, I don't really feel the end result of singing (which is actually all that I really know about) needs to be influenced by the learning from a score. (P11)

One teacher reported that 'If you had a line-up of our typical students performing music, you would not be able to tell who was dyslexic and who was not' (P23), indicating that the barriers for dyslexic students to learning music do not necessarily have to limit their performance of the music.

Additionally, a student's enjoyment of the music meant more than high marks on a graded exam and gave them a sense of pride in their achievements, as a participant noted:

I think sometimes as long as they are enjoying the process, that's the main thing. She's always performed in concerts and shown off what she has achieved. (P3)

This was seen in students' ability to connect with an audience. A teacher would ask their student to visualise their interpretation:

If you are playing for a concert, what you are doing when you are performing a piece is you are painting a picture. What is the picture that you see? And some of them will go to great lengths to paint or to draw a picture and present what the music means to them. (P4)

Often, they are actually kind of musical, very intuitively musical and instinctive. Creative. Good at performing, playing in a concert and communicating to the audience, all those strengths. (P7)

The strengths of creative and vivid imaginations, empathy and the ability to connect with an audience enable some dyslexic students to perform successfully. These findings suggest that

performance seems to encourage the students' self-concept and increase their motivation for learning.

8.1.8 Persistence

Persistence and resilience were character traits which teachers observed in some dyslexic students, with P12 reporting: 'He will practise every single day, regardless of the conflicts that he has ... he has got good discipline' (P12). This ability to endure in the face of difficulty was highlighted by another teacher who stated that their student 'was just such a hard worker' and that 'she could have done with extra time in exams and all of that, but she just kept at it' (P7). Another teacher admired their student's persistence in the face of challenge, stating that 'She does find sight-reading hard, but she is determined to do it; I read somewhere that dyslexic people are more determined than other people to get going' (P17). Teachers were aware that this need to work diligently might be a means of compensating for other challenges like slower processing speeds and poor working memory. Motivated by a goal to play a specific piece or perform, P27 described their student, saying that 'Determination has been a strength with dyslexic students, and they want to achieve the same despite the challenge' (P27).

8.1.9 Summary of findings

A number of themes and subthemes emerged from the teachers' interview data in relation to strengths that were observed in dyslexic students. The following section is a summary of these findings:

- Creativity was seen in dyslexic students' unique approaches to learning and connecting ideas in a novel way, which teachers described as three-dimensional thinking.
- This suggests that interaction with the kinaesthetic and aural patterns of an instrument might prove useful as part of the students' learning process, as opposed to working with a two-dimensional music score.
- A vivid imagination was another creative strength which teachers utilised by encouraging students to make a visual narrative of the piece to further engage with it and to aid their memory recall.

- Teachers recognised that dyslexic students enjoyed creating and composing their own music and were encouraged by teachers to verbalise their understanding of music which they heard as a way of engaging with it.
- Dyslexic students showed an eagerness to improvise and teachers recognised that this may stem from it requiring fewer processing demands, as opposed to an activity like sight-reading. Teachers generally began to introduce this with some parameters such as a scale or with an emphasis on specific musical concepts.
- Teachers recognised that an improvisatory skill might have been developed as a means of using aural strengths to compensate for poor engagement with the score.
- Improvisation was used in conjunction with notation software as an initial pathway for composition projects.
- Verbal strengths were identified in dyslexic students and teachers utilised this in a multisensory approach, similar to literacy strategies, of vocalising note names or pitch in conjunction with touching the score or the keys. Students were asked to summarise verbally what they had learned as a means of reinforcing their learning.
- Students who find writing difficult might benefit from teaching and assessment approaches that focuses more on a verbalisation of understanding.
- Teachers recognised that aural strengths, aural memory and pitch perception, were most likely a compensatory mechanism used to overcome difficulties with sightreading.
- Some teachers were keen to build on and develop these aural skills but noted the need to balance learning between more challenging activities, such as sight-reading, with those that gave a sense of achievement.
- Despite a recognition of challenges related to poor working memory, teachers highlighted the long-term and kinaesthetic memory strengths of dyslexic students which they also recognised might be a compensation.
- This long-term memory strength was a factor for consideration when planning lessons and choosing repertoire. Teachers were aware of the time and effort in

terms of repetition required from the student that may be needed to commit a piece to long-term memory, and adjusted repertoire amount and difficulty level accordingly.

- Two teachers noted that dyslexic students found it challenging, if not impossible, to 'unlearn' errors once they were committed to long-term memory storage.
- Supervised learning and strategies which strengthen students' self-reflection in the learning process might prevent the ingrained errors which can occur.
- Spatial and kinaesthetic strengths were described by teachers as a means of remembering music for dyslexic students. This suggests that teachers may need sensitivity to the time it takes for students to become familiar enough with the feel of the music to recall it effectively. This also emphasises the need to focus on the student and their needs instead of the teacher's preferred approaches for learning.
- The strength of being able to recall and process detailed verbal instructions might be a compensatory mechanism to deal with working memory skill limitations. This suggests the importance for teachers to have an awareness of how to use these effectively without overloading the student and an ability to assess when the student may no longer need the reminders.
- Teachers highlighted the strengths of dyslexic students in terms of their abilities to perform creatively and to communicate well to an audience. Quality performance experiences led to a sense of achievement and accomplishment which might positively affect their self-concept and motivation.
- A sense of resilience and persistence was noted by teachers who recognised that students' desire to achieve a goal or to perform often motivated their intense determination.

8.1.10 Conclusion

Teacher perceptions of dyslexic student strengths were similar to earlier findings in the literature. Creativity, vivid imagination and improvisatory skills were noted as useful building blocks for student's devising their own learning strategies or to which they might

incorporate previously learned material. The use of problem-solving abilities to find unique solutions was incorporated into creativity skills. Verbal, aural and memory strengths were noted, as were the abilities of dyslexic students to follow detailed instructions. Dyslexic students were reported to have strengths in terms of performing, including the ability to communicate well with an audience, and were described by their persistence and determination to continue to put forth effort to achieve their goals. Suggesting that the benefits of strengths-focused teaching might extend into other areas of students' lives, Rappolt-Schlichtman et al. (2018) state 'If we instead build the capacity of students with dyslexia to both improve on their areas of weakness, as well as build on their individual and unique areas of strength, we begin to create the foundation for thriving in learning and life' (p. 872).

REFLECTIVE STATEMENT 6: TRANSITION TO A STRENGTHS-FOCUSED APPROACH

RS6.1 Introduction

Dyslexia is widely considered to be a disorder or disability, and current definitions are rooted in deficit-thinking, although some recent research reflects an awareness of strengths in individuals with dyslexia. I reflected on my conversations with professional musicians in internationally recognised orchestras or ensembles who, even at such acclaimed levels, declined to disclose their dyslexia. This was due to a fear of facing stigma at the mesosystem, exosystem and macro-system levels (Bronfenbrenner, 1979). The mesosystem levels involved their relationships with colleagues and the fear that disclosure might bring about a lack of trust. At the exosystem level they were concerned about losing employment opportunities and feeling shame in the workplace. At the macrosystem level are the cultural and societal influences which may be formed from misconceptions about dyslexia and musicians with dyslexia. I felt a sense of injustice about this, and it motivated me to want to make a change in my teaching and approach to dyslexic students. As Dixon et al. (2016) state: 'Deficit-based questions lead to deficit-based conversations, which in turn lead to deficit-based patterns of action' (p. 7). These deficit-based patterns of action culminate in a sense of shame in individuals with dyslexia and a feeling of stigma in society.

In reflecting on this, I considered how I might be described as disorganised, a procrastinator and highly opinionated; equally, I might be described as a loyal friend, a caring mother and a good cook. Whilst all of these descriptions might be true, depending on the context, it is reasonably certain that either description would greatly influence how I might be perceived by others, what I might be considered to be capable of or entrusted with in any given situation. Who among us would choose or desire to be defined by our less positive attributes? This might also influence how I view myself and have a profound impact on my mental health. A poor sense of identity and self-concept for individuals with dyslexia may be the result of repeated criticism (Wilmot et al., 2023). Singer et al. (2013) describe this connection, stating that 'the capacity for meaning-making in response to narrated experience is generally predictive of psychological health, well-being and capacity for growth' although they suggest that it must be 'coherent', 'flexible' and 'accurate' otherwise the individual may develop a fixed mindset or unrealistic expectations (p. 575).

RS6.2 Strengths-based approaches in literature

Literature related to dyslexic strengths (Eide & Eide, 2011; Gobbo, 2020; Malpas, 2012; Nelson & Hourigan, 2016; Rappolt-Schlichtmann et al., 2018; Rolka & Silverman, 2015) and observations from my teaching practice led to a transition from my initial perceptions of dyslexia as a deficit-based condition to an understanding that 'dyslexia' represented a diverse array of individuals with patterns of strengths and areas of challenge. I began to understand more about how the differences in brain structure and functioning might convey specific advantages, and that a reliance on declarative learning over procedural learning may impart an explorative strength (Nicholson, 2015). This suggests that although a person with dyslexia may recall symbols or facts, their ability to utilise these in an automatic way may be impaired. However, this may mean that they are able to use their understanding of the declarative information in novel ways.

Maier (2014) describes this paradigm shift from a 'deficit lens' to a 'capacity lens' and suggests that this will impact the way the teacher speaks about the student, how support needs are determined, how collaboration is valued and the teachers' expectations for the student. The benefits to students from this approach can be considered from the positive psychology literature. Seligman's (2011) theory of human flourishing, based on the PERMA model of well-being, includes the consideration of 'positive emotions, engagement, relationships, meaning and accomplishment' (p. 16). Fredrickson's (2004) broaden-and-build theory links positive emotions to an openness to new ideas and the development of resilience and agency, believing that this might create reserves for dealing with high pressure or demands. Using an analogy from strengths-based sports coaching, Dixon et al. (2016) suggests that although 'the natural response may be to focus on fixing problems and eradicating weaknesses', it is more important to acknowledge and to question why something was *successful* in order to cultivate a positive mindset.

This literature suggests that by focusing on the lenses of students' strengths and capabilities, rather than on their deficits, we might imagine new ways of thinking about and collaborating with them. This also represents a shift at the macrosystems level of inclusive

music teaching where limiting belief systems might give way to capability-focused paradigms in which the students' needs are at the centre. Armstrong (2012) describes this as 'positive niche construction' and suggests that there are seven components, quoted below:

- 1. Strength awareness
- 2. Environmental modifications
- 3. Assistive technology/Universal Design for Learning
- 4. Strength-based learning strategies
- 5. Human resources
- 6. Positive role models
- 7. Positive career aspirations (p. 14)

Implementing these areas into teaching may have an impact on dyslexic students' self-belief and feelings of agency, as described in Reflective Statement 3, and may encourage them to move from being passive to more active learners. Whilst I sought to identify my students' strengths, created a low arousal learning environment for face to face lessons and as much as was in my control for online lessons and employed technology, strengths-based strategies and the Universal Design for Learning Framework (CAST, 2018) to guide lesson planning, aspects of the 'positive niche construction' components which I did not emphasise were positive role models and career aspirations. Many of the examples of musicians with dyslexia are not well documented and any example would need to be in line with the individual student's aspirations, otherwise they would not see it as being relevant to them. Furthermore, there might be many reasons why an individual with dyslexia has become successful, although this idea of success tends to be narrowly defined in terms of wealth and fame. I preferred instead to focus on my students as individuals with their aims and goals (or what success looked like for them) in learning to play the piano as worthy of my attention and resources and viewed these as sufficient in themselves rather than warranting the possible distraction of using valuable lesson time in discussion of famous individuals with dyslexia.

Another aim of this thesis is to raise awareness of the importance of the development of human resources around the student, so that teachers, parents and other professionals are equipped with accurate knowledge about dyslexia and an awareness of approaches which promote the student's metacognition. In that sense, human resource development has occurred in my own growth as a teacher, but I hope to make further contributions to those who teach, parent and work with dyslexic students.

RS6.3 Positive psychology in music pedagogy

It was clear from dyslexia literature (Chapter 2) that despite a fair amount of research, there is still much to be understood in relation to the defining characteristics and cause of dyslexia. I became aware that the music and dyslexia literature were largely focused on deficit-based thinking. In reflecting on my own experiences as a student which were rooted in the master-apprentice tradition, the focus was on direct instruction, limited student autonomy and critical feedback. I distinctly recall the few positive comments and praise which were made many years ago. Drawing on music pedagogy literature, I found that Patson and Waters (2015) refer to a model of positive instruction in the music teaching context called PIMS (Positive Instruction in Music Studios), and suggest four steps in this process: 'positive priming exercises, reminding students of their strengths, pausing to praise, praising the process, not just the results' (p. 4).

Louis (2011) points out that mere 'talent identification' may result in fixed mindsets with students, similar to the identification of weaknesses, but suggests that a 'strengths development intervention' might be more beneficial as it was 'designed to encourage students to consider novel approaches for mobilizing and strategically developing their existing talents into strengths, emphasizing students' capacity for growth' (p. 209). In lesson planning, I used the following questions in Figure RS6.1 to think about how lessons might be tailored to the individual student, their strengths and their needs:

Key Questions to Consider When Planning Lessons

Think about how learners will engage with the lesson.



Does the lesson provide options that can help all learners:

- regulate their own learning?
- sustain effort and motivation?
- engage and interest all learners?

Think about how information is presented to learners.



Does the information provide options that help all learners:

- reach higher levels of comprehension and understanding?
- understand the symbols and expressions?
- perceive what needs to be learned?

Think about how learners are expected to act strategically & express themselves.



Does the activity provide options that help all learners:

- act strategically?
- express themselves fluently?
- physically respond?

From: Universal Design for Learning: Theory and Practice Available at <u>udltheorypractice.cast.org</u> For print and accessible EPUB, contact <u>publishing@cast.org</u> or any book retailer.

CAST | Until learning has no limits[•] | cast.org

Figure RS6.1 Key questions to consider when planning lessons (UDL, 2020)

I recognised that views that are rooted in deficit-based thinking might be expressed even in subtle ways in my teaching. This might be in the form of lowered expectations or conversely, I might develop unrealistic expectations for the student which might place a burden on them and cause them to feel a sense of inadequacy. I sought each student's views on our lessons and checked with parents often to see if there was a mismatch between my perspective from what I was hearing and seeing in the lessons and what the student was expressing at home. Asking these questions made me aware of aspects of the students' lives which might not have been made available to me otherwise; for example, I began to understand the impact of school exams, friendship difficulties, frustrations with teachers or the school setting, house moves and the subsequent effects on their emotional and mental well-being. By continuing to focus on what was working and by taking opportunities to pause and praise them for their efforts and to remind them of their strengths, the following examples demonstrate evidence of how I sought to establish a positive learning environment.

RS6.3.1 Ben

RS6.3.1.1 Introduction and early self-reflection

Based on my experiences teaching Ben (See Reflective Statement 2), some of the emerging strengths I had observed were strong pitch perception and aural memory, excellent coordination skills and persistence. Dixon et al. (2016) describe the purpose of inviting self-reflection as 'asking 'how' and 'why' type questions, in order to build on positives and replicate success' (p. 13). The following examples demonstrate evidence of pausing to self-reflect on his progress and to praise the effort he was putting forth in the process. In the following example (Video excerpts are not available to readers due to GDPR and participant confidentiality) Ben used the whiteboard feature on Zoom to draw an emoji and treble clef and was asked to reflect on the things he had achieved in the lesson. Upon reflection, I can see that he minimised his own effort and achievements from the lesson. Whether this was just the result of tiredness at the end of the lesson or not being used to acknowledging positive feedback, it was difficult to tell. I continued to focus on what was working and took time to pause and praise them for their efforts and to remind them of their strengths.

RS6.3.1.2 Praise

I took the time to praise him for creative solutions to a problem when, during an online lesson I did not have access to a copy of the music, Ben took a screenshot of the score and set it as his virtual backdrop on Zoom (Video excerpts are not available to readers due to GDPR and participant confidentiality). He did seem very proud of this and it reminded me that this was a good example of the use of dynamic and interconnected reasoning skills

described by Eide and Eide (2011). I praised him for his sight-reading efforts when learning 'The Clown' (Video excerpts are not available to readers due to GDPR and participant confidentiality). A few months later, I reminded Ben of his excellent coordination skills when he played 'The Clown' again (Video excerpts are not available to readers due to GDPR and participant confidentiality). He seemed to acknowledge what was said, but replied 'Well, it took a long time'. I found this was relevant because it signified to me how deeply rooted his lack of confidence was and I felt sad that he did not recognise the importance of these strengths in the area of aural memory and strong coordination skills. These are highly valuable talents for any musician. I reflected on my comment 'that's a gift, though' referring to his good coordination skills and my intention was to explain that others might find coordination much more difficult, although I try to avoid comparison. So often, it seems that people with dyslexia are made to feel small because of their challenges and my desire was to remind Ben of his areas of strength. However, I understand the importance of the words that I use and how this might have undermined his effort to learn the piece. He may have felt that he had to work very hard to accomplish this result (in a piece which he had only begun to sight-read two months earlier), but perhaps he needed this effort acknowledged as well. His persistence did pay off, as this recording shows that by January, 13, 2022 he was able to play through the piece fairly well to the end with a few reminders, mainly provided to support the transitions between sections (Video excerpts are not available to readers due to GDPR and participant confidentiality).

RS6.3.1.3 Strengths-based approaches

Building on Ben's strong aural skills and wanting to practise sight-reading in short segments, I chose this <u>Melody Among Us game</u> (Video excerpts are not available to readers due to GDPR and participant confidentiality), where the user has to sight-read a short phrase and remember the sound and sight of the phrase. If they encounter that phrase again, it is the 'imposter' and the student should identify it and choose not to play it. I predicted he would recognise the sound and perhaps recall the sight of the Imposter, which he did on each occasion it appeared. I hoped that this would be a way of developing his strength in this area, as well as motivating him to work on other aspects. There were some challenges, with finding the note on the keyboard and some directional confusion, but what was interesting was how he made progress in recognising the pitch, location on the keyboard and direction

throughout the course of the game. Over time, there was growth as he developed skills in areas that had previously been quite difficult for him, for example, sight-reading and following the direction of the music on the keyboard. Morrow (2023) suggests having the student draw a series of high and low notes on a blank piece of paper and then covering that with an overlay of the grand stave to show the student how right-left on the keyboard correlates to high-low on the score. In the future, I would use this strategy for a beginner student who struggled with directional orientation. However, it may be that repetition over a period of time might be needed to establish directional accuracy, and that a one-time exercise might not be sufficient.

Scales were another area where his aural strengths were developed (Video excerpts are not available to readers due to GDPR and participant confidentiality). I include this because it shows our use of aural skills in learning scale patterns at an early lesson. However, upon reviewing it, I can see that the student might have benefited from more thinking space and less verbal communication from me. He might have been encouraged to play slowly as well. Also, upon reflection, I would not have let him know to expect a 'sharp note' in the G scale. Because I did, he was anticipating it before it happened, and more likely than not, would have figured it out on his own if I had not said anything. This would have allowed him that opportunity to discover the scale as he developed his aural skills and knowledge of major scale patterns. From this, I learned to allow more time and to hold back on information which the student might be able to solve on their own.

Later, the aural pattern of a scale was revisited with a <u>diagram</u>²⁹ of the Circle of fifths in which Ben filled in the major scale tonic note names, to incorporate a greater understanding of keys and key signatures (Video excerpts are not available to readers due to GDPR and participant confidentiality). I asked Ben a series of questions so that he would become aware of the relationship between keys, but he took the responsibility for playing the scales and completing the worksheet. In this clip, I praised him for his effort, but also emphasised his aural strengths: good pitch recognition and ability to recall aural patterns. I expressed appreciation for the sight-reading effort earlier in the lesson, an area in which he was

²⁹ https://www.myfunpianostudio.com/music-theory/circle-of-fifths-worksheet/

showing improvement. My aim in doing this was that this was to help him to see his capacity for growth in all areas, not just in areas of natural aptitude.

RS6.3.1.4 Later self-reflection

In 2022, I asked Ben to reflect on his experiences (S3 Transcript 2022):

R: What are your strengths in learning music?

Ben: The best way is listening to it. I remember songs. Then I practise and practise it.

R: Without a recording, what would you do?

Ben: I would try to hum it. I would write the note names down.

R: But you don't often do that, actually.

Ben: Yes, but if I was on my own and I don't know the piece ... But usually you give me examples, and we make recordings.

R: Any other activities you enjoy in lessons?

Ben: Composing on Flat.io

R: What do you enjoy most about playing the piano?

Ben: It helps me to relax and is something fun to do when I'm bored.

R: What do you feel is your greatest achievement in terms of playing a piece?

Ben: 'Enemy', I think it was my best piece yet. I'd like to learn 'Megalovania', too. I learned a bit of it.

R: How would you learn it on your own?

Ben: I would search for it and look for a tutorial on YouTube, not with Synaesthesia, but if someone talked me through it.

R: Are there any strategies that we've used that you think haven't been helpful for you?

Ben: No. I think I could practise more maybe.

R: Do you feel that you know what to practise when you are on your own?

Ben: Sometimes I don't know how to practise, so I just play some scales. I've tried to make flashcards to practise sight-reading. When I'm tired my mind changes things up, and I see notes and I think they are different to what they are. I think I need to get better at that.

R: Have you ever started with finding the first note and then just watch whether it goes up or down?

Ben: Yes, but what's on the page just doesn't click with me. I don't know why.

R: Do you think it would be good if we talked through how to practise more at lessons, like getting more specific about how to do it. What are your greatest strengths?

Ben: I can improvise well, listen to a piece and try to play it. Not perfectly, but I can get round what it sounds like.

R: Do you think learning pieces without music is better for you?

Ben: We've played some pieces without any music and I think it's gone well.

R: Is it better hands together or hands separately?

Ben: I feel like I need to know what each hand has to do separately before I put them together.

Ben's responses indicate to me that he is aware of some of his strengths, perseverance and good aural recall, and I think it is positive that he recognises that more practice might help him to improve. We can also use these reflections to consider how to prepare him more effectively for practice times in future. One way which we did this was to schedule mini guided practices throughout a week, especially if he was at the beginning stages of learning a piece.

RS6.3.2 Alex

RS6.3.2.1 Introduction and praise

From my observations, Alex demonstrated strength in the areas of problem-solving, visuospatial strengths, pattern recognition, resilience and perseverance. He displayed confidence and excellent verbal communication skills. Here are two examples with Alex where I pause to praise him and point out the positive aspects of his playing and the effort involved in playing hands together in Train Ride (Video excerpts are not available to readers due to GDPR and participant confidentiality) and for his sight-reading skills and the effort he put forth to plan and organise his fingering (Video excerpts are not available to readers due to GDPR and participant confidentiality). At other times, I praised Alex for remembering major or minor chord formations, for concentration and for his strength at recognising patterns.

RS6.3.2.2 Self-reflection leading to strengths-based strategies

I encouraged him to self-reflect and invited him to share which strategies he felt we were using were working well (Video excerpts are not available to readers due to GDPR and participant confidentiality). In my reflective journal entry, I described some interesting insights from this conversation. He recognised that simple sight-reading was achievable, but that more complex sight-reading needed some alternative strategies. On the topic of how he learned that piece, using aural skills, I reminded him of a challenge I had given him earlier in the lesson, which was to learn 'Musette in D' through demonstration and aural recall. His reaction on seeing the score was somewhat telling, as he described it looking more complicated than it was to play it, and he removed the score as he clearly did not want to be using notation.

One of these pieces was 'He's a Pirate' which was a piece which he was keen to learn in late 2019. He decided to write some note names on the score, and learned the melody line, albeit with his left hand (Video excerpts are not available to readers due to GDPR and participant confidentiality). This was a strategy that he used occasionally in the earlier days of learning music (2018-2020). He would learn the melody line in his left hand and then transfer it to his right hand. Alex is right handed, but both he and his mother, who is also dyslexic, mention that they are able to do specific tasks better with specific hands. For

example, I wrote in my reflective notes that his mum prefers to paint with her left hand, even though she is right-handed also. I chose not to challenge this at that time, but waited to see how he would approach this when he wanted to play hands together. He chose to take a break after learning the melody, although he did return to the piece much later to add the chords to it. He played it with his hands together in the medley of pirate songs which he developed (See Reflective Statement 4). I think it can be a good step to take a break from something that is challenging and return to it at another time. Wilmot et al. (2023) suggest that being able to have breaks may be beneficial for the mental well-being of dyslexic students. Energy and concentration levels may need renewing and the break from the piece can give them a chance to do that.

He also mentioned 'We will sail together', one of a series of sea shanties from the Sea of Thieves³⁰ video games. This was adapted by reading the melody line and adding simple chords; he performs it here (Video excerpts are not available to readers due to GDPR and participant confidentiality). This was a piece that seemed to give him great pride to be able to play as he felt successful in playing hands together and using the sustain pedal.

RS6.3.2.3 Adaptive notation strategies

At other times, I praised Alex for remembering major or minor chord formations, for concentration and for his strength at recognising patterns. Often, when Alex would play a piece, there were a number of areas which needed correction, but I tried to focus on what he had done well first. In choosing repertoire, we would adapt pieces which might have complex or challenging sections so that they were able to master them and achieve their goal. Alex referred to this as 'editing pieces' in our discussion of what was working well. One example of an adaptation that we used included simplifying and enlarging score parts from the original 'Jurassic World'; we adapted it to a section of the right hand first and then to this version. After a period of time, Alex was able to play it hands together (Video excerpts are not available to readers due to GDPR and participant confidentiality). There is some instability in the rhythm of the piece. This video was recorded in 2019 and although we had

³⁰ https://www.seaofthieves.com/

done a substantial amount of pulse and rhythm work, it was an area of challenge and one that we continued to develop.

Based on what I had observed in terms of visuospatial skills, I invited him to use Figurenotes³¹ alternative notation system as the use of colours and shapes might reduce the processing burden of sight-reading. I provided him with a key to the system. What we found was that he understood the system of colours and shapes, intuitively and quickly read 'Ode to Joy' which might be more difficult in standard notation. After a few lessons using Figurenotes notation, he sight-read 'Au Clair de la Lune' for the first time (Video excerpts are not available to readers due to GDPR and participant confidentiality). He does well to keep the pulse stable. Whilst the use of the pedal was not recommended, I believe that it gave him a sense of security to use it as it sustained the previous note whilst he was looking for the next one. However, his thoughts and feelings on using Figurenotes are discussed in Section SR6.3.2.5 of this reflective statement. Sensitivity to feelings of stigma or 'otherness' which might arise when alternative notation or adaptations are used is an area which I have learned to become more aware of since beginning this research.

RS6.3.2.4 The use of problem-solving games to assess sight-reading and musical terms knowledge

Particularly during the period of lockdown when we could not meet face-to-face, I tried to include a short game in each lesson. In this game, Elements Among Us (Video excerpts are not available to readers due to GDPR and participant confidentiality), his knowledge of musical terms was tested. He was very proud to report that he had come fourth in a class of 30 students in a music test, and I praised him for his achievement at school and for being 'one of the highest people in my class'. Responses in the game indicated that it was good to have some review of the terms; however, it is possible that the emphasis on text and some writing may have been a challenge. We used musical terms when practising scales as well so that the student had the opportunity to hear them and to apply them to their playing in a practical way. In our lessons, I tried to avoid conveying any sense of blame when there were difficulties and tried not to draw attention to areas of challenge, but to work on them with the student in a supportive manner. I chose games that I thought he would find enjoyable,

³¹ https://drakemusicscotland.org/figurenotes/

but which also used problem-solving strengths as a way of buffering the challenges, such as this 'Virtual escape room' where you had to choose clues, problem-solve and perform a sight-reading task to progress (Video excerpts are not available to readers due to GDPR and participant confidentiality).

RS6.3.2.5 Later self-reflection

In 2022 (S2 Transcript 2022), I asked him to reflect again on the following aspects of our lessons:

R: What are the best approaches for you in learning music?

Alex: Listening to the music first, or having you play it.

R: If you had to do it completely on your own, what would you do?

Alex: Write the note names in, at least at the beginning.

R: Do you remember you used Figurenotes, but you decided that you preferred to practise with standard notation?

Alex: Let's say I'm with someone else and I want to learn a piece of music, but I don't know how to read notes like that. It's better that I know how to use the regular notes.

R: What has helped you to learn?

S2: Listening to the song a few times, so I know the pace. Just trying and failing until I get it right.

R: How do you know if you've got it right?

S2: I use recordings, we practise together.

R: What things other than recording have helped you?

S2: The composing app (Flat.io) that was fun.

R: Do you have any suggestions for how to make our lessons better?

S2: You make it interesting, I don't like to be bored. I like music and I'm good at it. We do different activities and that is good, not just playing scales but composing, learning new pieces, improvising. I think I'm good at improvising on the piano.

From these responses and from discussion in lessons, I was further able to develop my profile of Alex, and this helped to inform future lesson planning. I learned that Alex was not comfortable using Figurenotes but did use other score adaptations with ease. He understood there was a need for persistence, or 'just trying and failing until I get it right,' if he wanted to learn. The use of technology was only mentioned in relation to recordings and notation software which we had used for composing.

RS6.4 Conclusion

I viewed this knowledge as foundational building blocks for the future with the desire that the students would know that their input allowed for enhanced collaboration between us. Their thoughts and feelings were important to me. In these previous sections, I have demonstrated evidence that I praised the students for specific strengths which I observed, for effort in the process and sought to develop a better understanding of each student. I prioritised their voices by encouraging them to reflect on their learning and to consider what had worked well and why. I considered some of the components of 'positive niche instruction' (Armstrong, 2012, p. 14): an awareness of students' strengths, an environment that was comfortable for them, the use of adapted notation and multiple means of representation, the use of strategies and repertoire which were chosen based on students' strengths and the development of expertise built from collaborating with the students and trialling resources and varied approaches. As I made the transition from a deficit-based mindset to a strengths-focused approach, it enabled me to value the students as individuals with diverse and unique profiles. Being informed by students' self-reflections as well as my observations enabled me to make effective use of the questions from the UDL model for future lesson planning by considering students' engagement, how to present material and how students might express themselves or respond. This has had positive implications for the ways in which I taught other students as well, by considering their strengths and being collaborative rather than having a top-down approach to my teaching.

Chapter 9 FINDINGS FROM PARENT INTERVIEWS

9.1 Introduction

Parental support is an important aspect of a student's instrumental music education (Creech, 2010; Creech & Hallam, 2010; Upitis et al., 2017) and has been identified as one of the social factors affecting students' motivation and self-regulatory behaviours (McPherson & Renwick, 2011). Positive parental roles might include observing, guiding, teaching, encouraging, offering support and praise; conversely, overbearing parents with unrealistic expectations without considering a student's feelings or aspirations might have a negative effect (Upitis et al. 2017). Parental support is seen as a protective factor for children with dyslexia that supports their development of resilience (Catts & Petscher, 2022; Haft et al., 2016; Wilmot et al., 2023).

9.2 Context and participants

Although research on parental roles with dyslexic students and genetic, emotional, behavioural and educational risk factors have been examined by the literature, the role of parents of dyslexic children in the instrumental music education context has not been widely explored. In this research, participants were recruited through my teaching practice as the parents of my dyslexic students (See Appendix G for interview schedule). In addition, of the teachers interviewed for this research, four were mothers of children with dyslexia (Abigail (P5), Rosemary (P14), Hannah (P20), Sally (P21), and one student participant was a father of children with dyslexia (Simon), and their experiences, described during their teacher interviews, add additional insights to the relationship between parents, their children with dyslexia and teachers. Two of the parent participants were from the United States (Sally, Hannah), but the others were based in the UK. All but one of their children had been assessed as dyslexic through formal assessment. As it is a potentially sensitive topic, participants were invited to share their experiences as a parent of a dyslexic child and were reassured that they were not required to discuss anything that might make them feel uncomfortable. The participants have been given pseudonyms in the analysis.

Two participants (Erin and Polly) were mothers of dyslexic students in my teaching practice; these were the only non-musician parents in the study. Four participants were themselves

dyslexic, and in line with anecdotal evidence in the literature, two of these participants discovered they had similar challenges when their child was assessed and diagnosed with dyslexia. The data were analysed and coded using iterative, thematic processes as detailed in Chapter 3; this enabled patterns and themes to emerge. Table 9.1 below shows the pseudonym name of the participants, self-reported dyslexia (D) or dyslexic tendencies (DT), level of music training, location, number of children with dyslexia and the gender of those children.

Participants	D/	Music training	Location	Number of children	Gender of children
	DT			with dyslexia	
Simon	D	Professional	UK	2	М, М
Abigail (P5)	D	Professional	UK	2	M, F
Rosemary (P14)	N/A	Professional	UK	1	F
Erin	D	None	UK	1	М
Polly	N/A	None	UK	1	М
Hannah (P20)	N/A	Professional	US	1	М
Sally (P21)	D	Professional	US	1	М

Table 9.1 Contextual	information	about	participants
----------------------	-------------	-------	--------------

9.3 Findings

Parental support may be influenced by several factors. Several themes emerged from analysis of the data, which are briefly noted here before being further detailed in subsequent sections. Findings suggest that dyslexic parents may be impacted by their past experiences and perspectives of dyslexia, but also perceive benefits to their role as a person with dyslexia and as a parent to a child with dyslexia (9.3.1). A musically trained parent with dyslexia reported the benefits of using the Suzuki method as a means of developing music learning based on a focus on aural skills and repetition, which was suited to their mode of learning (9.3.2.1). Some musically trained parents reported that they assumed the role of teacher for their dyslexic students, feeling that they had specific insight to the student's learning process (9.3.2.2). Parents took a role as mediator between the student and their school and teachers, with mothers taking this role predominantly (9.3.4). Parents also had to investigate assessment needs and communicate with schools about behavioural difficulties, best practice in teaching and learning approaches and the use of reasonable adjustments or accommodations. Findings suggest that positive relationships and communication between the teacher and parent may be mutually beneficial (9.4). Parents reported great pride in their child's achievements and were aware of the determination and effort invested by their children in pursuing their goals. Findings suggest that parents also sought to promote positive role models with dyslexia (9.5).

9.3.1 Dyslexic parents' experiences and perspectives

For those parents who were dyslexic or had dyslexic tendencies, their own school experiences and music learning journeys contributed significantly to their view of dyslexia, of themselves and that of their child with dyslexia. Dyslexic parents recognised that they had been labelled, stigmatised or had hidden their difficulties in order to cope. For some, this helped them to identify strategies that they recognised that their children were also using. Conversely, this suggests that parents with dyslexia may assume that a strategy will work for their child because it worked for them, only to discover that it is not effective for their child because of the complex nature of interacting effects of dyslexia on each individual.

9.3.1.1 Impact on dyslexic parents

A complex mixture of relief and sadness were some of the emotions identified when their children were diagnosed. Simon recalled how it felt to be labelled and misunderstood during his own educational experiences when his two children were diagnosed as dyslexic. Previously undiagnosed, Sally described masking her difficulties in school and noticing that her son had similar traits:

I recognised it in my son. I thought 'You know, I'm probably dyslexic'. He had the same symptoms. He had enormous trouble reading text and would mis-read. His writing skills were bad as well ... I was reading sixth grade [Primary school Year 6 in the UK educational system] level reading material and just not telling anybody. (Sally)

However, despite their own negative experiences and without minimising the negative consequences, another dyslexic parent had an awareness of positive aspects of her child's diagnosis, stating:

You feel stupid because you can't read. It's so difficult. You can't make out big words or understand what was going on. I don't think my friends knew because I was quite clever. But you lose a lot of confidence ... I was sad for Alex because I know how debilitating it can be, but I wasn't that upset because I think it can teach you skills that a lot of other people don't have (Erin).

However, societal stigma did affect parents' view of their identity, according to Sally, who stated that 'Both as a parent and as a dyslexic person, I have seen the kind of stonewalling, labelling and stereotyping that occurs when a person declares that they are dyslexic'. In conclusion, dyslexic parents might recognise their own dyslexic tendencies when observing their children encounter similar difficulties. Reactions and emotions might vary from relief to shame to an awareness of how their difficulties were masked to manage challenging situations.

9.3.1.2 Perceived advantages to being a dyslexic parent

A dyslexic parent mentioned how important it is to encourage a growth mindset in their dyslexic children: 'Encouragement is one of the main things, and empathy, and to say, "Mistakes are fine, they help us to learn"' (Simon). This participant perceived an advantage over his non-dyslexic partner when it came to understanding and empathising with their dyslexic children. However, he noted that his partner was the first to recognise signs of dyslexia in their children. Some dyslexic parents referred to the non-dyslexic spouse as the 'investigator'; this may be because a large amount of information about dyslexia is text-based and the non-dyslexic spouse might feel more confident about their reading and comprehension skills. Confirming a perceived advantage in terms of empathy, one dyslexic mother described that being dyslexic 'helps as I understand him and can help him navigate life, be more patient and, I guess, less frustrated' (Erin). Shared experiences and empathy with their difficulties in school were seen as advantages for parents with dyslexia in their relationship with their dyslexic child.

9.3.1.3 Similar compensatory mechanisms

Dyslexic parents with musical learning experience noted similar compensatory mechanisms with their dyslexic children in their approaches to learning music, including overlearning and a strong memory. A dyslexic parent with musical training recognised challenges in learning

scales similar to those of their dyslexic child, stating that 'the level of overlearning that I had to do was just quite phenomenal' and she saw this mirrored in her daughter who required several months of practise to learn one scale. A dyslexic parent who is a professional musician describes how their children used coping mechanisms like their own with sightreading, stating that 'They are very similar in learning to me as they learn by listening' (Simon). Another dyslexic parent emphasised her strong memory as a way of learning music from their time in school, stating:

I'm sure you've seen that with Alex as well. You kind of fake your way through life. When I learned to read music, I couldn't make out the notes, I just memorised the song. Things like that were my coping mechanisms. (Erin)

In summary, parents with dyslexia might perceive that they have a greater understanding, empathy and patience with their dyslexic children based on shared experiences and similarities in their use of coping mechanisms. Some dyslexic parents identified the same challenges in learning music as their dyslexic children and were able to highlight strategies such as overlearning, strong memory, aural skills and methods which were suited to their strengths. This suggests that parental communication with the teacher may provide useful information related to how their child learns.

9.3.2 Musically trained parents

Parents reported music learning methods which they recognised were beneficial to dyslexic students. Some parents reported that they took the role of teaching their child and other ways in which they involved their child in music. The Suzuki method has been a suggested method for dyslexic students based on parental involvement in observing lessons, collaborating in the learning process and nurturing the child (Macmillan, 2004; Vance, 2004).

9.3.2.1 Suzuki method

A musically trained parent noted the benefits of the Suzuki method for dyslexic students based on an emphasis on aural learning in accordance with students' music learning preferences. Simon stated:

That's why the Suzuki method really suited me. It was just listening to the sounds and reproducing it. That is how I learn. At some point, you have to start learning to read notations and that starts quite late in Suzuki. (Simon)

This suggests that these parents may be aware of positive approaches and can reframe their expectations, for example, in regard to the point at which notation is dealt with.

9.3.2.2 Parents as teachers

Three of the mothers were music teachers who also gave instrumental lessons to their own children. They assumed the role of 'tutor' in this case as they felt they had teaching expertise and a knowledge of their child and the child's needs. Hannah stated that 'As a mother, I think I could just connect what he needed with creative ways to help him, and to make things easier'. Hannah described how she helped her son learn to play the piano:

If you show [son] how to do something, he can do it. He's tactile. A template helps him. When I was teaching him piano, I would put my hands on his hands, just so he could feel how my fingers were moving. (Hannah)

Hannah then passed this information on to her son's piano teacher. Similar to strategies for visually impaired students discussed in Chapter 2, this teaching style might be difficult to manage in another context given the sensitivities around safeguarding and the use of touch in a teaching setting. Rosemary considered that it might be better for her daughter to experience working with another teacher, stating that:

I was her instrumental teacher. When she got to a certain age, I thought someone else should do this, it's not good for it all to be with me. She had three other teachers and didn't want to stay with any of them. So, we just resumed our relationship and that was it. (Rosemary)

This might suggest that the other teachers were not adequately equipped to understand her needs as a student with dyslexia, whether they were aware of an assessment or not. Hannah also describes how she would provide her son with methods of organisation to use in music learning, but also in other areas of life:
I showed [son] how I organise things when he moved to university. I told him 'You can do things however you want, but this is the system that I use and I find it really helpful', and he adapted it to suit him. It was helpful to give that template of organisation, but to allow him to find his own way of doing things. (Hannah)

One mother described trialling different method books and strategies which likely informed their teaching with other dyslexic students (Abigail). Parents with music training might be an asset to their children with dyslexia as they may be aware of a variety of method books, repertoire and approaches and may also provide support during practice sessions.

9.3.3 Communication with the teacher

Parental communication with the instrumental teacher is important to allow the teacher to develop knowledge of the student so that they can work with them effectively and avoid asking them to do things which might cause them shame or embarrassment. Sharing the assessment report may be a valuable way for the teacher to begin to understand the student's particular profile.

9.3.3.1 Sharing assessment profiles

Assessment profiles give a narrative results summary which describes the student's general ability, processing speed, working memory and reading/writing abilities along with suggestions for working with the student. The British Dyslexia Association³² quotes an assessment with a specialist teacher at £660 or an assessment with a psychologist at £840 as of April 2024 and indicates that assessors are booked for months in advance. This might indicate that families may need to wait some time before an assessment can be made. Without a formal assessment and diagnosis, parents noted that their children would not be eligible for reasonable adjustments/accommodations in school or exam settings. Simon noted that 'the cost of assessment is expensive' as well as 'the cost of extra tuition', indicating potential inequality for families in low socioeconomic circumstances. Hannah confirmed this and stated that for her son to enter university and qualify for extra support, they had to pay for a full report at considerable expense. Oglethorpe (2008) notes that the assessment report may contain valuable information for an instrumental teacher, referring

³² British Dyslexia Association 'Diagnostic Assessments for Dyslexia or Dyscalculia' Retrieved on September 4, 2024 from https://www.bdadyslexia.org.uk/services/assessments/diagnostic-assessments/overview-2

to parents as the 'vital link' between the child, the school and other professionals in sharing this information (p. 14). Regarding the student's response to assessment, Polly stated:

Ben seemed quite unperturbed by the assessment and was interested in the results. He was reassured that he now understood why he had found certain areas of his schooling difficult and was especially pleased that having dyslexia was different from being 'stupid' or 'slow' which is how he had often regarded himself before he was assessed as having dyslexia. (Polly)

The assessment report included useful information relating to pacing, cognitive and working memory load, all of which would be helpful for the instrumental teacher's lesson planning and delivery. Although an assessment may be useful, parents noted the cost of assessment as a limiting factor to receiving an educational plan and reasonable adjustments/accommodations. It is possible that parents may not realise the relevance of assessments to music learning and that instrumental teachers without dyslexia-specific knowledge might not be aware of their significance for informing teaching approaches. However, parents can play a valuable role in supporting the work of the instrumental teacher through sharing the assessment report.

9.3.3.2 Observing the individual profiles of strengths and weaknesses

A parent of a student in my teaching practice described their son as a strong problem solver who was easily able to grasp visual explanations of music:

I guess that's why music is nice because it's visual ... He likes patterns and order. As a child, everything had to be neat and lined up. Piano is very much like that. Mathematical, almost. He likes problem solving, working things out. I think he does struggle with rhythm a bit. (Erin)

Even the use of compensatory mechanisms, such as strong aural skills and memory, did not always prevent frustration in other areas like processing or coordination when attempting to sight-read music. Despite noting strengths with memorisation and aural skills, one parent described the challenges and resulting tension they and their son faced: [Son's name] has always struggled reading music. He can read one line of music, no problem. But when he had to read two clefs at the piano and then add the tactile dimension, he was the most frustrated that I have ever seen him. He would throw things, and get so frustrated with himself. Now, he's better, but it takes him so long to figure it out. (Hannah)

However, even though coordination and processing appeared to be causing difficulties, Hannah reported strengths in areas such as resilience, exceptional aural memory and transposition, particularly when score modifications like enlargement were used:

[The teacher] said that he was one of her favourite students because he is such a hard worker, persevering. She did ear training with him. His challenges were sightreading. If he went through it twice, he had it memorised. Reading the piece is the hardest. Transposition is easy for him. We would enlarge scores for him. (Hannah)

Co-occurring conditions related to concentration were noted by Polly, who stated:

Ben undoubtedly has a short attention span and he is very easily distracted. He seems to learn best through what he hears, rather than visually ... Ben is extremely practical and gains great satisfaction from completing a physical task such as constructing a bookcase or making a model. He has particular strengths in IT, science, maths and technology-based subjects. (Polly)

Rosemary's daughter was undiagnosed until she was in her third year of conservatoire. She was able to achieve a good level of sight-reading after years of practise, with her mother as her teacher. Rosemary stated that 'I've made a note that the reading was not that good [at the beginning], but by the time she got up to Grade five and six, she got quite good'. Rosemary's strategies to support her daughter's sight-reading included a consistent focus on recognising intervals; for example, 'noticing how fourths and fifths look different'. This may have been effective because it was more of a visual strategy or focus on the interval shapes rather than a processing of each distinct note.

Challenges were noted across varying student profile patterns and at times the parents observed that their children had poor tolerance for frustration. Parents observed that their

children's self-esteem and confidence were affected in a variety of ways by being diagnosed and in relation to the dyslexic child's self-belief when practising and performing. They recognised their child's strengths, including problem-solving, strong aural memory and memorisation, in addition to resilience and the ability to develop musical skills, like sightreading, over time. This is valuable for teachers in recognising the need for balancing expectations and quality communication with parents to avoid tensions or additional pressure.

9.3.3.3 Parental role in identifying the effects of low self-esteem

Parents recognised that issues of self-esteem affected their dyslexic child's learning in general, and in music this was sometimes seen in difficulties with practising and confidence to perform. A combination of self-esteem problems and difficulties at the piano might create possible 'tensions' for parents and their children when it comes time to practise (Abigail). Hannah concurred with this, referring to how 'dyslexia usually goes hand in hand with a lack of confidence and that 'it doesn't take much for [son's] confidence to snap if he is doing something like piano' (Hannah). Rosemary related how her daughter has a beautiful singing voice, but that following her diagnosis with dyslexia, she lost the confidence to perform. This has important implications for family and instrumental teachers in terms of sensitivity and utilising approaches which include encouragement and positive feedback.

9.3.4 Parental communication with schools and classroom teachers and implications for the relationship with the instrumental teacher

In this study, parents referred to a variety of experiences relating to their communication with their child's school and teachers. This was seen in the process of getting the school to acknowledge the student's difficulties, managing behavioural issues and advocating for dyslexic friendly strategies to be incorporated into the student's learning at school.

9.3.4.1 Pursuing assessment when schools are reluctant

In the case of the mother of a dyslexic piano student in my teaching practice, Polly recognised her son's reading difficulties, but had to tenaciously pursue assessment:

We flagged this up with the school several times but unfortunately no action was taken until a very cursory and inconclusive assessment was carried out during the final weeks of Year 6 when he was nearly 11 years old. When he started at secondary school, we once again raised the issue. This time a more detailed and in-depth assessment was performed almost immediately, and the results showed clearly that Ben has moderate dyslexia. (Polly)

This assessment contained a number of important points relating to pacing and manner of lesson delivery and Ben's needs in the learning context. It was a valuable asset when planning lessons and to my approach in teaching Ben.

9.3.4.2 Managing behavioural difficulties and relating them to dyslexia

One participant described how the process of communicating with school caused a great deal of frustration, particularly when the student began exhibiting behavioural issues:

Alex hates school and finds it hard to sit still and concentrate and then gets into trouble quickly whenever he isn't coping. I have to keep reminding them [the school teachers] that he has dyslexia. They don't seem to be able to pick up his struggling as his 'dyslexia' and then his behaviour gets worse because he is struggling and their solution is to exclude him instead of helping. (Erin)

These findings indicate the importance of parental and teacher communication in determining the root causes of a student's behaviour, whether that be that they are deflecting from their difficulties, anxious about events at school or simply have not taken the time to practise. The parental role in managing this behaviour, attempting to establish the cause and communicating that to the teacher, plays an important part in the development of the teacher's understanding of both the parents' experiences but also a sensitivity to the causes of the student's behaviour.

9.3.4.3 Ensuring the use of dyslexia-friendly strategies

Parents referred to a need for teachers to learn how dyslexia may affect their children and to employ specific strategies in their approach. Abigail described communicating with a teacher about her daughter's problems with organisation, stating that 'I'm going to be very challenging now, you keep telling her she just has to "sort it out" but she ... has got to do more learning and have more teaching in order to be able to do that'.. Hannah confirmed this, stating that her son's greatest difficulty in school was a 'disorganised teacher' and that

some teachers simply did not understand the importance of explicit instructions for dyslexic students; therefore, she had to be quite proactive to achieve support:

I have always advocated for him. I made sure that the teachers knew that he was dyslexic. I had to go to his art teacher, because she did not understand that she was not giving him concrete enough information in order for him to be able to do what she wanted him to do. (Hannah)

Participants related how educational experiences varied across schools and by teacher, and suggested that although there are levels of support they are not always appropriate, and that behavioural and organisational issues are not always considered in the context of a learning difficulty.

9.3.4.4 Ensuring the application of reasonable adjustments and accommodations

As part of their advocacy role, parents face a number of challenges in negotiating for correctly applied reasonable adjustments as they are also mindful of preventing situations where the student feels stigmatised. Polly described her son's school:

I think Ben is moderately well supported at school but not all the teachers are fully understanding of his dyslexia and how it affects his work or the extra time he may need for some tasks. The use of a laptop in some lessons has been helpful but I think Ben also finds this somewhat stigmatising as he has to collect it before each of the relevant lessons and recently he seems more keen to try to complete tasks on paper. (Polly)

It is understandable that students also struggle when teachers and schools lack an understanding of the application of reasonable adjustments/accommodations.

9.3.5 Positive communication with teachers

A lack of dyslexia knowledge is a barrier to effectiveness for many instrumental teachers (Oglethorpe, 2008; Ganschow et al., 1994; Nelson & Hourigan, 2016), and a major source of motivation for my research. As has been demonstrated within my data, instrumental teachers can make a positive difference to their dyslexic students' learning. Simon recalled a teacher's kindness and that she 'would offer extra help after classes'. I was encouraged by this response from Polly in relation to my student, Ben:

Our understanding of Ben's dyslexia, the effects that it has on his learning and above all the range of techniques possible to help him learn has benefited hugely from his piano teacher. We have greatly valued the individual tailored approach taken with Ben. His resulting progress fills him with satisfaction and helps his confidence. It also has the effect of teaching us new things in our understanding of Ben, how his dyslexia affects his learning and how his abilities may be maximised. (Polly)

This suggests that teachers might influence and benefit the parents, as well as the student, in understanding the way that the child learns, or recognising their potential for learning. Music may well be the area in which a dyslexic student excels and finds an important outlet. If the instrumental teacher can highlight the student's abilities, it may not just promote their self-confidence but also change family members' perceptions of their capabilities.

9.3.6 Pride in their child's achievements and the importance of role models

Perhaps aware of the immense hurdles that some dyslexic students must face, parents described their pride in their children's achievements. Parents recognised the tremendous amount of effort that went into students' achievements whether that was a recital, graded music exams or degree in higher education (Simon, Rosemary, Abigail, Hannah, Erin, Polly).

She learned the piano to Grade Eight [ABRSM], singing and cello. She went to [conservatoire] and read music. She was always very diligent, and she got a 2:1 degree so she didn't do badly. But she had to do a lot more work than everybody else. (Rosemary)

Erin noted Alex's sense of pride in his ability to play the piano, stating:

When he [Alex] sees some of the girls playing piano in class, he says 'Well, I can do so much better than that'. He likes boasting about it ... He is loving piano and is getting the affirmation that he needs and that means he is engaged with it. It's something he enjoys and feels well doing. (Erin)

Focusing on positive aspects was important for one mother, who stated that 'I was never ashamed at all to say that he has dyslexia. He can do so many things' (Hannah). Parents also indicated further approaches which were effective in helping their children to succeed at music learning and in other areas of life. The parents of dyslexic students in my piano teaching practice, Ben and Alex, reflected on how their child's motivation and selfdetermination were key to maintaining their engagement in learning:

He [Alex] is good with discipline and routine, when he knows the boundary, he will keep it. He likes ownership of things, feeling that he can make decisions. As long as he is interested, he will be engaged. Anything visual, or technical like recording (he's very vain), then he really likes it. He's very proud of what he has accomplished so far. He works very hard and is very persistent. He doesn't practise [piano] a lot, but he does it every day without complaint. (Erin)

I think it helps Ben when someone is 'on his wavelength' and able to communicate with him in such a way that his confidence is enhanced and he is able to achieve. If he is taught in small chunks he is more able to carry out instructions, retain information and achieve manageable goals. (Polly)

These statements also indicate that parents are aware of the strengths that their children with dyslexia possess. Parents emphasised the importance of positive role models for their children as a means of motivation and encouragement (Abigail, Erin). However, it is possible that students may also compare themselves negatively to others who are successful when they find themselves struggling.

One parent mentioned non-music role models:

Richard Branson, Jamie Oliver, others. They had to learn how to keep up and make plans to succeed. I've seen that in Alex, he has to compensate and do that extra bit of work to keep up. I think it helps to motivate you further, but I wouldn't wish it [dyslexia] on anybody. You have to look at the positives because you can't get away from it. (Erin) Parental support roles include offering praise and encouragement for efforts and celebrating the achievements of their dyslexic children. To avoid the perception of being condescended to or a feeling of unfair comparison, students might benefit from a balance of quality information about dyslexia and the meaning of 'success' as well as a focus on other factors that might influence good outcomes. These factors include parental and community support. With the promotion and discussion of positive dyslexic role models in music and in other areas, parents might encourage their children to believe that barriers can be overcome with persistent effort and appropriate support.

9.4 Summary of findings

- Dyslexic parents might recognise similar challenges and coping strategies as seen in their dyslexic children. These may serve as reminders of past negative educational experiences with an emotional impact on the parent. This suggests a need for sensitivity and tact on the part of the music teacher who may also need to consider using either verbal or short written forms of communication with dyslexic parents.
- Some parents with dyslexia felt that they had an advantage of empathic understanding with their dyslexic children. However, this suggests that a parent with dyslexia might inadvertently assume the same strategies that worked for them will work for their child with dyslexia, although individual profiles of dyslexia, even within families, may be very different and require different approaches.
- Musically trained parents were able to support their child with dyslexia, by taking the role of tutor or by supporting guided practice sessions, although personality differences or the child's unwillingness to be taught by the parent might be limiting factors. The benefits of the Suzuki method for children with dyslexia, with an initial focus on aural training and later development of notation, were recognised by a father with dyslexia who had been trained using the method.
- The cost of assessment, the extent to which parents believe an assessment is relevant for the music and their willingness to share reports can be limiting factors to the teacher's access to relevant information. Findings suggest that in sharing their

child's assessment profiles, parents might enhance the instrumental music teachers' ability to support their child.

- Parents' knowledge of their child, including their strengths and weaknesses profile, in conjunction with assessment information, is an asset for the teacher in planning and delivering lessons more effectively for their child. Areas of sensitivity related to the individual child's identity and self-esteem or behavioural patterns may help the teacher to develop an empathic understanding of their needs.
- Findings suggest that support through schools might only be obtained through persistent advocacy on the part of parents. Poor communication within schools meant that teachers were not always aware of the students' challenges, and behavioural challenges were not viewed consistently through the lens of dyslexia.
- Additional stress and burdens were placed on parents as they had to negotiate with schools to ensure that teachers understood their child's needs, used appropriate strategies and established that reasonable adjustments were in place and used effectively for their child.
- Findings suggest that positive communication between parents and the teacher might enhance familial perceptions of dyslexia and promote their understanding of their child's capabilities, thus supporting the student's self-confidence.
- Parents of children with dyslexia reported a great sense of pride in their child's achievements, particularly considering the effort and persistence a child with dyslexia may invest in attaining their goals.
- Parents felt that positive role models with dyslexia served to motivate and encourage their children that limitations might be overcome with persistent effort.

9.5 Conclusion

Parental roles have the potential to be strong protective factors in the development of resilience in dyslexic children. These findings correlate to research that indicates that low socioeconomic status is a risk factor, that it tends to be mothers who identify, advocate for and tutor their children and who tend to report concerns regarding their child's internalising

behaviours, though it is acknowledged that only one father participated in this research, so further research is needed in this area concerning music learning. Mothers also reported stress and frustration in their roles as communicators or negotiators with teachers and schools. By sharing assessment information and relevant communication about the child, parents might develop a mutually beneficial partnership with instrumental music and/or other teachers. This may also give insight to teachers as to how they perceive students' behavioural issues and how to manage them. Instrumental teachers are privileged to be able to focus on individual students in one-to-one, or even small group lessons and may feel more able to tailor their approach and to develop a mutually supportive relationship with parents.

These findings suggest that careful thought and sensitivity should be given as to how instrumental teachers communicate with parents, particularly dyslexic parents, by avoiding dense terminology or long emails. Similarly, rather than adding to the advocacy burden or anxiety the parent may be feeling about their dyslexic child, instrumental teachers may be able to encourage them as their student's capabilities, effort and strengths become evident during music lessons. This may enable the parent to view the dyslexic child differently and become more aware of how their potential might be maximised in other areas. As teachers recognise the internalising and externalising behaviours that dyslexic students may exhibit at times, open communication with parents might enable greater empathy and understanding of the root causes and management practices. Communication with teachers is an important aspect of parental support; the instrumental teacher might benefit from the sharing of the assessment report, parental observations about the student's profile, self-esteem and parents' challenges in communicating with schools and classroom teachers. Positive communication can benefit all stakeholders in the process.

Chapter 10 FINDINGS FROM STUDENT INTERVIEWS

10.1 Introduction

Dyslexic students' music learning experiences, feelings and attitudes about dyslexia, including their views on teachers and educational environments, represent their reality through which our understanding of their needs can be developed. As Cook-Sather (2006) states, the term 'voice' implies 'having a legitimate perspective and opinion, being present and taking part, and/or having an active role' (p. 362). Instead of passive participation in research, 'learner voice' research is described as 'considering the perspectives and ideas of learners, respecting what everyone has to say, taking risks, sharing and listening, engaging and working together in partnership' (Walker, 2008, p. 5). Different levels of learner engagement in learner voice research include 'informing, consulting, involving, collaborating and empowering' indicating a move from mere participation to 'learner control' (Rudd et al., 2006, p. 13) with the aim of the student bringing about transformational change to their own learning process. This perspective has the benefit of providing a new understanding which can help teachers, and parents, support learners. The perspectives of teachers (Chapters 6-8) and parents of dyslexic children (Chapter 9) have been examined previously in this thesis; in this chapter, the focus is on the dyslexic music student voice.

10.2 Methods

Music learning experiences from multiple age and vantage points (beginner instrumental to higher education) might enable an understanding of how individual dyslexic students view specific elements of music learning, their perspectives on how dyslexia may affect them, beneficial learning experiences and also their views on characteristics of effective teachers and teaching practices. The following sections describe the participant recruitment (10.2.1), data collection (10.2.2) and data analysis procedures (10.2.3).

10.2.1 Participant recruitment

Inclusion criteria extended to those with lived experiences of music lessons as a dyslexic student. Participants were recruited from the British Dyslexia Association music committee mailing list, by word of mouth, through Facebook groups relating to music and/or dyslexia

and through my piano teaching practice. Potential participants (and, if appropriate, their parents) were provided with information forms and the opportunity to ask questions about the research project in advance prior to signing informed consent forms. Nineteen dyslexic participants who had experiences as music students, in addition to two dyslexic students in my piano teaching practice (Alex and Ben), were interviewed in order to collect data.

10.2.2 Data collection

Data collection took place through semi-structured interviews with 19 participants, and interviews and reflective observations of lessons with my two dyslexic pupils, Alex and Ben (See Chapter 3 for full details of methods). Participants were encouraged to speak about their experiences to highlight the issues that they felt were most important in relation to each question. Participants were offered the opportunity to review their transcripts. All participants were assured anonymity through the use of a pseudonym. Participants, and parents of minor children, were provided with information sheets which detailed the processes of data collection. They were offered an opportunity to discuss any concerns they might have and were provided with a consent form. In addition to interviews, data was collected from feedback which occurred naturally in the course of the lessons with my two dyslexic students Ben and Alex, as well as these students' reflections on their learning process. Comments made by the two students during lessons were noted in my reflective journal on a regular basis, both as a form of data collection, and to support my own reflection and teaching development so that the findings might be embedded in my teaching practices. The aim of this study was to understand students' perspectives on dyslexia, the effectiveness of specific strategies used in the lessons, their perspectives on how they learn music and any positive or negative learning experiences. Students were also encouraged to consider whether or not their preferred strategies empowered them to learn music effectively on their own.

The following table (Table 10.1) introduces the participants, their age range, instrument and level of proficiency.

Participant	Age	Instrument	Level on instrument
Aaron	Adult	Brass	Advanced
Laura	Adult	Guitar, piano	Advanced
Ned	Higher education	Vocal	Advanced
Rose	Higher education	Vocal, piano, cello	Advanced
Meg	Secondary school	Vocal	Intermediate
Grace	Higher education	Vocal, piano	Advanced
Katie	Higher education	Vocal, piano, flute	Advanced/ Intermediate/
			Beginner
Rory	Secondary school	Guitar, piano	Intermediate/ Beginner
Ruth	Adult	Flute, piano	Advanced
James	Secondary school	Vocal, piano	Advanced
Simon	Adult	Viola	Advanced
Esther	Higher education	Vocal, clarinet	Advanced/Intermediate
Victoria	Higher education	Vocal, flute	Intermediate
Mel	Adult	Piano, French horn	Advanced
Julie	Higher education	Piano	Advanced
Peter	Adult	Guitar, percussion,	Advanced/ Intermediate
		piano	
Nathan	Adult	Guitar	Advanced
Jane	Higher education	Piano, Guitar	Intermediate
Keith	Adult	Oboe, flute	Advanced
Alex	Secondary school	Piano	Beginner/ Intermediate
Ben	Secondary school	Piano	Beginner/ Intermediate

Table 10.1 Contextual	information a	about	participants
-----------------------	---------------	-------	--------------

10.2.3 Data analysis

The data was analysed according to the process of thematic analysis (Braun & Clarke, 2006) which began with the interviews and transcription process as a way to become familiar with the data (this process is described in greater detail in Chapter 3 of this thesis). Reflective observations from my lessons with Alex and Ben were in the form of notes made during the lesson and further reflection made after the lesson, following Gibbs' (1988) reflective model. These reflections enabled me to evaluate actions made in a cyclical manner and to re-evaluate their effectiveness, students' responses and my thoughts and feelings.

10.3 Findings

The following themes emerged: family environments, familial incidence of dyslexia, motivational factors and attitudes toward assessment, challenges and strategies identified, strengths, positive and negative teacher experiences, the use of technology, points for change and beneficial resources.

10.3.1 Family incidence and environment

Participants described their experiences prior to formal assessment and diagnosis, as well as the motivational factors for beginning the process. Family and educational experiences clearly had an impact on their lives. Ten participants mentioned their mother's support as an important protective factor in their development and well-being:

I was very lucky in that before I started at primary, my mum had already taught me to read, write and do maths. In my assessment, they felt that could have been a protective factor for me in school. (Keith)

James felt that he owed his success in music to the parental support he received from his musician parents, with his father being dyslexic. Only two of the participants had parents with specialist dyslexia training (Meg, Katie), whilst non-specialist but supportive attitudes were identified in other parents.

Sixteen participants reported having family members with dyslexic tendencies or a formal diagnosis of dyslexia. This correlates with research indicating a familial risk for dyslexia.

Grace describes how this created a sense of solidarity: 'For me, it was very important that my mother understood, and was in the same position'. Some participants indicated how the understanding and attitude of their parents linked to advocacy for their needs to be taken into consideration in educational settings. Despite personal experiences and knowledge gained, parents were reported by participants to have encountered barriers when approaching schools for support. Ruth noted how the school tried to blame her parents for her poor reading skills, and Meg's mother raised issues about her struggles with the school but the school decided not to pursue assessment.

Whilst Ben and Alex referred to their families as supportive with parents who were actively involved in their learning, neither set of parents had completed specialist dyslexia training. Their mothers were the main investigators and advocates in their assessment, and the main source of communication with me, in my work with them as their piano teacher.

These findings confirm the results from interviews with parents (Chapter 10) that there is importance in family support, particularly that of mothers, in preparing their dyslexic children for school, gaining knowledge, advocating and supporting them. The results also suggest that schools or teachers may resist parents' attempts to advocate for their children or may blame the parents for student's difficulties. For example, there is potential for an instrumental teacher to place blame on the parents or student for a lack of practice or effort or to grow impatient with the student if they are not aware of dyslexia and how it may affect the student's music learning.

10.3.2 Assessment and diagnosis process

Of the nineteen participants, four were diagnosed with dyslexia as children (under 11 years old), five at secondary school level (12-17 years old) and seven participants received a diagnosis after the age of 18. Three participants were also diagnosed with dyscalculia. Those assessed as children reported fewer negative educational experiences than those who were assessed later in life. This is consistent with literature that emphasises the importance of early identification and assessment (Snowling, 2013). Four participants (Aaron, Keith, Nathan, Mel) who attended primary and secondary school prior to the 1990s experienced

more negative educational experiences than those who attended school in more recent years, suggesting a positive shift in the understanding and support of dyslexic students.

Participants also noted that, at university level, the barrier of the cost toward the assessment was reduced or eliminated through financial support and this enabled them to seek a formal diagnosis; this choice was mainly motivated by the need to prove eligibility for academic accommodations and to receive computer and software equipment. Several participants felt that the accommodations had a positive effect on their academic performance (Rose, Meg, Anna, Victoria, Julie, Peter, Emily, Keith, Ned).

Three participants (Simon, Esther, Mel) did not have a dyslexia diagnosis, but recognised dyslexic tendencies in themselves. In two cases (Simon, Mel) this had been identified in their children (who were diagnosed with dyslexia) or in the case of one participant (Esther) discovered through their own research. In each case, they felt that the cost of assessment was too prohibitive and that it would not be beneficial at their age. They were employed in a professional capacity as musicians and music teachers, and yet were adamant that they would not share their dyslexic tendencies with colleagues or employers:

There is a part of me that wants to justify myself in front of them. 'Yes, sorry I am dyslexic and that is why that happened'. But I know I can't really do that. It's not acceptable to share it. When you are with musicians, a huge aspect is trust. (Esther)

One of these three participants noted how she would simply avoid jobs which required sight-reading, whilst the other two used complex overlearning or aural listening strategies to achieve better sight-reading skills.

Participants reported a sense of ambiguity about the assessment and diagnosis process, reporting that they experienced less self-blame following a formal diagnosis. However, two participants (Meg, Victoria) described having emotional breakdowns during assessment sections that were challenging to complete. Other interpretations of the assessment process included a recognition of past coping mechanisms, defensiveness, fears and sensitivities. Positive reactions included a reassurance that they could attribute their difficulties to

dyslexia and not to being lazy or lacking in effort, as their teachers or parents might have suggested (Victoria, Laura, Keith, Nathan, Katie, Ned).

Both Alex and Ben had a formal assessment and diagnosis. Their responses to being diagnosed with dyslexia were very similar. Alex stated that 'It didn't really change anything, but during primary school I got some extra help from teachers'. Ben noted that he was glad for the extra time and use of a laptop in secondary school. This suggests that having had the assessment earlier on in their school experiences has provided them with additional support and prepared them in a sense for the challenges that they face.

The cost of assessment may be a barrier for some dyslexics to be able to provide proof of eligibility for reasonable adjustments and exam accommodations which they perceive will improve their academic record. Some older participants did not perceive a benefit for themselves in a diagnostic assessment, as they were established in their careers and did not want to be stigmatised by employers or colleagues. There may be benefits to early assessment for dyslexic students. Sharing sensitive information like an assessment report may also require parents and the student to have some degree of trust that the teacher will use the information appropriately; teachers should be aware of data protection issues and the importance of safe data storage and confidentiality.

10.3.3 Challenges and strategies

General challenges reported by participants include a slow pace and misreading of text, challenges concerning writing, processing, visual disturbances, basic maths, sequencing, motor coordination, time management, being overwhelmed by information, spatial awareness issues, following instructions, working memory, stress and anxiety during tests or in a classroom.

10.3.3.1 General challenges

Text reading speed was described as 'painfully slow' (Rose). Participants reported frustration that they had to work harder than their peers to achieve the same results (Meg, Simon, Katie, Ruth) with another participant lamenting the fact that they had tried to use the same strategies as others but had not been successful with them (Esther). Participants

noted that they had to discover their own strategies and coping mechanisms. James stated: 'I have always had to find my own way of learning; it was the same with music'. Others attributed their challenges to poor teaching and felt that if material had been presented in a different manner, their results would have improved (Peter, Meg, Katie, Aaron, Rose). Aaron noted that due to the differing assessment standards of exam boards, he was able to pass his Music GCSE after having failed it the first time. These findings suggest that although dyslexic students recognise their challenges and weaknesses, they also recognise that appropriate instruction and more flexible educational systems would likely produce better outcomes for them. There was an emphasis on the importance of being enabled to develop their own ways of learning.

Similar to the interview participants, my piano student Ben reported that his challenges at school were very much dependent on the way classes were taught. For instance, when asked to describe what was challenging at school he said 'Everything, but sometimes there are fun things I can do' (Reflective notes 2021-6-17). He described being frustrated when 'It doesn't come together the way I want it to' in ICT [information and communication technologies] or textiles, suggesting a possible difficulty with following complex instructions. In ICT, he found 'typing and coding' a struggle. Ben reported other challenges:

English. Writing PEEL³³ paragraphs. I can't get them to the length the teacher wants. Reading is harder for me than writing. I find using things like 'simile' and 'juxtaposition' really hard. I just don't get it.

Alex reported his challenges to be in the subject areas of 'Maths, English, art, geography and being good [at school]'. In terms of memory, he stated that 'I don't ever remember things; I have to try to put it in the top of my brain and I just keep reminding myself'. Alex noted that reading and writing were both challenging:

I find both quite difficult to be honest. So, I usually type on my iPad or my teachers print out extra stuff for me because they like me to practise writing sometimes.

³³ PEEL paragraphs refer to the methods of argument construction: point, evidence, explain and link.

Alex also described difficulty with learning foreign languages, and was relieved that he had been exempted from the requirement at GCSE level, reporting 'In my last French test, I got 3 out of 50!'.

A slow pace of reading, spelling difficulties, poor executive function, visual disturbances and poor working memory are some of the general challenges faced by dyslexic students. An important theme is that students preferred to find their own way of learning and could be frustrated that it took them longer than peers to complete similar tasks. The importance of a more flexible educational system was highlighted as a way to reduce the stress they might face when learning foreign languages or when taking exams.

10.3.3.2 Music-related challenges and strategies – findings from interview participants

Challenges with aspects of reading music were reported by all participants ; this is consistent with the literature (Oglethorpe, 2008; Nelson & Hourigan, 2016; Ganschow et al., 1994). As one participant stated:

I think it has to do with one step between knowing where the note is and starting to make music. The step between analysing where everything is and starting to be fluent. (Esther)

Some participants noted that to a certain extent they were able to improve their sightreading by using various strategies; other participants felt that learning to sight-read at speed was not attainable for them despite years of practice (Mel, Victoria) or even repeated attempts to learn to read music at all (Nathan, Rory). This is consistent with the idea that dyslexia occurs on a spectrum and that challenges associated with it may persist despite attempts at remediation. Reading music was described as most challenging when it involved more than one stave of music, or for singers when they had the additional burden of reading lyrics as well as pitch and rhythm. General strategies for reading music included using non-white paper, enlarging the score and highlighting troublesome areas as well as using methodical and structured approaches which might include recognising patterns or harmonic analysis. Several participants reported the strategy of obtaining a recording of the music and listening to it prior to attempting to read it.

Rhythm

Thirteen participants described a difficulty with reading rhythm, including processing music symbols and feeling the pulse and rhythm in the music. Effective strategies noted by participants were to analyse the music, isolate the rhythm, break it down to the smallest component, use phonetic symbols, write the beats into the music, to clap or tap the rhythm and to attempt to play it very slowly at first. Practice strategies that focused on recognising commonly found rhythmic patterns were deemed helpful (Julie). One participant noted the effectiveness of the Dalcroze approach, using movement to embody the feeling of the rhythm (Katie). Victoria devised a system of colours and shapes to denote music rhythm in a visual way.

Pitch recognition

Challenges with pitch recognition involved correctly identifying the specific pitch as well as matching the pitch to the position of the body relative to the correct location of the note on the instrument. Isolating and practising pitch recognition through intense overlearning in addition to recognising the shape of intervals between notes were reported as useful approaches. The challenge that complex key signatures create was described by participants who reported that they found it easier to write each flat or sharp note into the music, rather than remembering the key signature and applying it to the piece (Keith, James). Learning, practising and visualising specific scales helped some participants with remembering key signatures. Meg reported writing the pitch names on a score to help with sight-reading. A strategy to focus on poor bass clef reading, reported by Keith to be 'enjoyable', was to practise pieces by Bach which had interesting bass line movement. The use of solfège, mainly for vocal students, was identified as a way of distinguishing intervals when sight-singing (Meg).

Music theory

Other challenges reported by dyslexic students included remembering the meaning of foreign terms, recalling and processing aural tests in time to achieve a response in exams, identifying sequences of terms (for example, *tonic, submediant, dominant*), or understanding and applying concepts (for example, the Circle of Fifths). Participants

reported that using visual memory tools, Memrise³⁴ as an example, helped them to learn and recall foreign music terminology. Memrise, or other applications like Quizlet³⁵, benefit the learner by allowing them to set their own pace by choosing the number and type of questions in a given learning set. Developing a visual strategy for learning the Circle of Fifths, one participant reported another self-teaching concept:

I found a music theory book in Oxfam and thought I would just read it. Within a few days, I decided to teach myself the information. I made Post-it notes for the circle of fifths and put them on the wall. (Meg)

Music exams

There was a variance in responses regarding experiences with music exams. Victoria was not aware of reasonable adjustments and accommodations that are available. In describing her experiences, Rose stated:

In terms of music, I've done fourteen ABRSM exams in piano, singing and cello. I always struggled with aural tests, especially the memory ones. My working memory is terrible. So, the playing and singing back was not good, and I compensated by doing as much as I could with scales and pieces just so it levelled out. (Rose)

She considered how accommodations might have changed her exam experiences:

I wonder what it would be like to have adjustments on music exams, I didn't even think they were available. I think it would have helped if I could have made notes during aural exams, during sight-reading, and extra time too.

Katie took a more relaxed approach, describing how 'I'm not really bothered about what I get ... I view it as "I'm going to give you a couple of pieces that I'm pretty pleased with and I'm going to try to keep with you while sight-singing"'. Peter struggled with the listening part of aural exams but noted that he overcame these challenges with targeted training. Meg described challenges with pitching when preparing for her Grade 8 singing exam, stating

³⁴ Memrise: https://www.memrise.com/

³⁵ Quizlet: https://quizlet.com/gb

that the 'teacher was worried about me doing the aural and sight-reading parts of the exam'.

The teacher plays an important role of supporting students, but without appropriate training may struggle to do this effectively. Julie was given extra time and a separate location but felt that teachers were not aware of her needs as an individual with dyslexia. Keith stated that 'accommodations in music exams need to be improved, and teacher's awareness of them needs to be better as well', whilst Ned credited a teacher who offered extra support after school with helping him to pass Grade 5 theory.

These findings suggest the importance of the teacher's role in supporting students with specific strategies, but also the need for an awareness of the implications of dyslexia on students' learning. Teachers would benefit from knowledge relating to preparation and support for students who wish to use accommodations and reasonable adjustments during exams.

Other challenges

In some contexts, performers are required to memorise music and for some dyslexic students this was reported to be a challenge, while for others the use of aural memory was preferable to sight-reading a score of music. Julie reported that she endured a variety of exhausting and time-consuming self-imposed strategies to memorise piano music for an audition including handwriting the score, harmonic analysis, practising away from the piano, using different rhythms and articulation and the use of memory checkpoints. Meg described a strategy of learning to play a piece on the piano with hands together from the start, as she felt she was learning kinaesthetically, knowing a piece 'in her fingers'. Mel approached this differently, aurally memorising a piece, orienting it to the piano keyboard and then working backwards to understand and connect this to the score. Visual strategies helped another participant with memorising, stating that 'I do memorise music, because I use visualisation and learning through mental work ... it is very effective for me; I memorise a picture of the music' (Jane).

Sight-reading, rhythm and pitch reading at speed can be a challenge for dyslexic music students. By devising their own strategies, using recordings or other aural strategies and repetitive overlearning, most are able to improve to some degree (Nelson & Hourigan, 2016). However, for some students, music scores remained inaccessible as the primary means of learning music; a potential future research direction might explore how to normalise the use of more visual representations of pitch and note durations but in a way that reduces stigma or shame and unobtrusively converts from standard notation. Like the interview participants, my student Alex reported his challenges as reading the score of music, but noted that he had improved through the last four years. Alex stated about music that 'It seems like a blank piece of paper to me. It's better if it's all spaced out. I would use an iPad but then I would have to scroll'. My student Ben also described his main challenges as 'sight-reading notes'.

Dyslexic music students reported that their main challenges were in the areas of sightreading, feeling rhythms or establishing a steady pulse, coordination, memorisation and aspects of music theory learning. Participants reported that the use of aural learning strategies, often through recordings or demonstration, was a useful strategy for learning music. They also suggest that the use of repetitive overlearning as a means of improving automaticity in response to slow processing may be beneficial. However, this was not always the case, and some participants preferred aural learning with frequent repetition.

10.3.4 The use of technology

Thirteen participants referred to the use of technology as a music learning, composing and producing tool, and also as a means of support through finding a like-minded community who share similar challenges. General technology applications such as mind mapping, speech-to-text, screen readers and text editing (for example, Grammarly³⁶) were found to be beneficial by participants.

³⁶ https://www.grammarly.com/

10.3.4.1 Recordings

The majority of participants reported using technology for music recordings, suggesting that the rise of applications like YouTube, Spotify and Apple Music in recent years have been useful for dyslexic music students. Meg recalled the value of obscure recordings with isolated individual choral parts as a tool to improve accuracy in part singing, when finding that attempting to learn the part by playing it on the piano was too time-consuming and slow. Ned and Meg made recordings of their vocal lessons to help them remember various aspects. However, the use of recordings as a means of learning music could be seen as controversial by some teachers, as Jane described:

I use recordings way too much. I know that using recordings is not a great tool for learning music. It's frowned upon. But I am definitely guilty of listening to them to know how to do or play something. (Jane)

This suggests that there is limited flexibility of approach to learning music in music education systems which may lead, not only to inequalities for dyslexic students, but feelings of inadequacy.

10.3.4.2 Composing and score modification tools

Nathan, a student guitarist, recording musician and music producer recalled that he has never been able to read music successfully, but used technology to record and manipulate music. Other participants indicated that music scoring programs (for example, Sibelius³⁷, Finale³⁸, Dorico³⁹, Flat.io⁴⁰) were effective ways of obtaining immediate feedback when composing by using the playback option. These programs also allow for creative approaches with the use of multiple, layered instrumentation and a means of simplifying the composing process. Participants reported modifying music scores by adding colour or written notes even temporarily when learning the piece of music, as well as removing lyrics or other aspects with a white highlighter tool (ForScore , Flat.io).

³⁷ https://www.avid.com/sibelius

³⁸ https://www.finalemusic.com/

³⁹ https://www.steinberg.net/dorico/

⁴⁰ https://kirk-piano-tuition.flat.io/

10.3.4.3 Games and social media accounts

Katie and Meg described their approach to learning foreign language terms by using the application Memrise, a visual and flashcard-based approach to language learning which allows students to tailor the amount of and type of information they learn at a time. Meg created an Instagram account where she shared her journey of learning the ABRSM Grade 5 music theory exam material which became a source of inspiration and encouragement for other dyslexic students and their parents. This also suggests that accounts like these might serve as a means of informing others about the challenges dyslexic music students face and as a way of de-stigmatising perceptions about the label of 'dyslexia'. There is a sense that some students may feel guilty for using tools like recordings or modifying scores and this suggests that perhaps teachers and educational systems might be more flexible in finding approaches which support the learner whilst also developing music reading skills to their full potential.

Both of my piano students, Alex and Ben, reported that they used recordings of pieces. They described enjoying the use of composing and arranging tools like Flat.io or GarageBand⁴¹. I asked Alex to create a drum accompaniment on GarageBand with the idea that this might encourage a more even tempo when he was playing a piece. Ben and I created a game online using Jeopardy Labs⁴² as a way of testing his GCSE music knowledge. A game is created by making categories with different degrees of difficulty. When a category and level are chosen, it provides the user with an 'answer' and they supply the 'question'. We also used technology in facilitating online lessons during the Covid-19 pandemic, and WhatsApp as a tool for sharing recordings, communication and brief notes with parents.

In conclusion, students reported the use of technology for obtaining recordings of pieces or their lessons, using applications which allowed them to isolate and practise core skills, arranging or editing music, and modifying scores by highlighting specific areas, or as a means of communication with teacher and parents.

⁴¹ https://www.apple.com/uk/mac/garageband/

⁴² https://jeopardylabs.com/

10.3.5 Coloured overlays

Although coloured paper or overlays are often referred to as a strategy to help dyslexic students, students expressed some concern that coloured paper or overlays were being proposed as some sort of magic fix for their dyslexia. One participant recalls:

In grad school I was told [by lecturers] that 'Well your dyslexia won't be an issue if I print your music on coloured paper'. And I was like "What?!" So, all my assignments were printed on various colours of paper. I thought "This is the same typewriter font - that isn't helpful!" With a serif font! Well, they said the files were so old they could not convert them. (Julie)

Julie went on to say that she had tried coloured overlays without success, stating that 'they weren't especially helpful to me because I still had to do the same amount of decoding and everything still moves [around on the page] the same amount'. She suggested that there were so many variables between colours and text sizes that might make a difference in reading that it was difficult to know how to establish what worked best. Coloured overlays were seen to be helpful, but a sense of stigma when using them was expressed, as Meg described: 'I should use overlays more often, but I don't like to because I don't like to draw attention to the fact that I'm using them. I could use my tinted glasses, but I don't' (Meg). Rose had also used coloured paper and tinted glasses with text, but not with music scores:

It was recommended to use different colours of paper and I was given a pair of tinted glasses. I believe it helped, but I didn't use them when looking at scores. I'm not sure why that is. I suppose a lot of music books aren't printed on pure white paper, and maybe that helped? (Rose)

A number of participants referred to the use of colour to highlight specific elements of their scores, such as highlighting parts played by different hands (Mel), differentiating between note values (Victoria), open or closed phonetic sounds (Ned), rather than using an overlay or coloured paper. Laura preferred the use of colours over the use of numbers to separate beats:

So, where in piano you have both hands, I will put a [vertical] line. Blue is beat one, yellow is beat two, red is beat three and green is beat four and that way I don't get lost in a sea of numbers. (Laura)

These findings from students indicate that the use of colour to highlight specific elements of the score tends to be more useful for them than an overlay or coloured paper. This may mean that highlighting reduces the processing burden connected with certain elements, for example rhythm and phonetic sounds, and also helps to distinguish between specific elements, for example separate hands or parts.

10.3.6 Strengths

The most common self-reported dyslexic strength was the ability to find their own way of learning. Sixteen participants reported aural strengths: playing by ear, strong relative or perfect pitch, an ability to mimic or copy aurally, and recognising or producing vocal harmonies. Several participants referred to themselves as good listeners in the verbal and interpersonal sense, although this was often followed by words like empathy, collaboration and the ability to relate to others and particularly those with dyslexia (Simon, Victoria, Julie, Ruth, Rose). Some participants referred to themselves as good performers and this suggests that these self-reported sensitivities to people and the environment may assist them as performers in communicating to an audience; a sense of flair for being dramatic was also identified as a personal strength (Ned, Simon, Esther, Nathan, Peter). In correlation, Julie described how she learned to communicate in sign language and attributed that to an ability to easily replace words with physical actions. Verbal communication skills were also noted by some participants (Aaron, Ned, Meg, Grace, Rose). One participant observed that one of his strengths was 'distracting people from my weaknesses' (Ned).

Resilience, determination, leadership and tenacity were among the words used by participants in identifying their own strengths. Despite research on the negative effects of dyslexia on executive functions such as working memory (Gray et al., 2019; Fostick & Revah, 2018), some participants reported exceptional memory for longer pieces of music. Having taught himself to play the guitar at age 16, Nathan described a dichotomy between his challenges and strengths:

The bizarre thing about dyslexia is that it cripples you for doing particular things and in other things, it doesn't seem to have much of an effect. I can remember long pieces of music, but I can't remember things I have to hold in my short-term memory. (Nathan)

Problem-solving skills, lateral thinking, inventing, creativity as displayed in composing, improvising and arranging music and practical skills were reported by participants (Aaron, Laura, Grace, Rory, Ruth, Nathan, Peter, James). Ruth described this as 'finding unique connections' which she felt was helpful to her as a leadership skill as well as the ability to 'imagine, improvise and find solutions' to problems.

When I asked my piano student Ben to finish the statement 'In school, I am really good at ...' he replied 'Maths, science and RE'. He described making the choice to take the Music GCSE course at his school and also felt that he had strengths in design technology classes, describing how he enjoyed using diagrams to create electrical circuits. In terms of learning music, he stated that 'I can improvise well, listen to a piece and try to play it, maybe not perfectly, but I can get around what it sounds like'.

In response to the same question, my student Alex reported having strengths in 'PE, music and maths'. Interestingly, when asked what he perceived his strengths to be, he replied that 'being told off' in school was a strength. Alex expressed confidence, independence and a strong sense of self-efficacy, making statements like 'I'm a quick learner'. After being able to play a difficult section, he stated: 'It's satisfying when you get something right when it's really, really hard'. This attitude of self-determination characterises his learning approach, and has been a promotive factor in his resilience to continue with piano lessons.

In summary, aural strengths, resilience and determination, creativity with improvising or composing music, verbal communication, problem-solving and performing strengths were reported by participants. In reference to the risk-resilience model (Catts & Petscher, 2022) described in Chapter 2 of this thesis, these strengths might be seen as protective and promotive factors for dyslexic music students when faced with frustration due to challenges with slow sight-reading, poor coordination or executive functioning when learning music.

10.3.7 Positive and negative teacher experiences

Attitudes and approaches used by teachers were significant factors in how students engaged with and enjoyed learning. Participants reported that teachers who were flexible, but employed a systematic approach, were helpful. Participants appreciated creative methods which helped to guide them to find their own ways of learning. Peter described this when learning Grade 8 theory material:

I had an amazing theory teacher when I was working on jazz in Grade 8 theory. With Bach chorales, there is such a long list of what to do and I found that intimidating. The way I found to do it was to create jazz Bach chorales; the idea being that you break as many rules as you can and that allowed me to remember the rules. (Peter)

Ned attributed his achievements in music to a secondary school teacher who would give him one-to-one sessions after school, as well as to the following teacher who had a creative, dramatic approach:

Everything is taken to the extreme, when he demonstrates something, for example dynamics, he will do it really loud and then take it very quiet and be very animated in the way he would do it. Extreme colours, extreme emotions, extreme animations and that's what sticks for me. (Ned)

A relaxed, collaborative approach was valued by Victoria: 'When I had my first flute teacher, he would sit beside me and he would be playing the flute or his other instruments – it was great!'. Teachers who had knowledge of dyslexia and recognised the difficulties students might have could lead them to feel supported and encouraged rather than blamed for their weaknesses. Beyond that, teachers with knowledge of appropriate instructional strategies for dyslexic students and those who were able to recognise when students were overwhelmed demonstrated kindness and understanding. James recalled:

Every so often, my piano teacher would tell me to close my eyes and then ask me random questions like 'Who is your favourite composer?' and I would answer and then we would go back to the piano and I would play a lot better. (James)

Participants expressed gratitude for teachers who went above and beyond their required duties by offering the student extra hours of tuition when needed for no additional costs. Meg described her teacher's 'unrelenting faith' in her despite her shortcomings: 'For me, the right teacher made all the difference'.

Participants also reported some negative experiences with teachers, particularly those who refused to use accommodations for which students were eligible. Parents and disability support tutors had to be drafted in to advocate for students in these cases. In the 1980s, Nathan and his parents were told that he was educationally 'sub-normal' despite also being told that he had a high IQ. Julie described having to educate her teachers on dyslexia as they had never encountered a dyslexic student in their graduate music classes previously, noting that they often used inaccessible methods of teaching or assessing.

Participants reported a lack of engagement when teachers scolded them for trying to use strategies such as writing in different colours or writing on the score of music (Rose) or forced them to work on pieces or material which were too difficult for them and not approached in an appropriate way (Peter). Meg described that while her disruptive behaviour caused her teacher to become very angry, in retrospect, she recognised that through her behaviour she was trying to hide the fact she was struggling. This suggests that had the teacher been able to recognise her behaviour as a cry for help, they might have responded with more understanding. As Keith described, as students we may make 'progress, but [it might] not be that perceptible on someone else's timetable'. Some participants reported abusive behaviours such as being ridiculed or misunderstood and that teachers were frequently frustrated or in a bad mood with them. There is no excuse for abusive behaviour directed at students. However, this may suggest that teachers are under pressure from educational systems or parents to demonstrate the student's progress through exams and high marks, and they may transfer this pressure to their students. It also suggests that there continues to be a lack of knowledge of dyslexia and how it affects dyslexic music students and a lack of understanding of appropriate music instruction for dyslexics as well as a lack of flexibility in approach to individual students.

My student Ben described enjoying RE [religious education] lessons as the teacher would show the class 40-minute videos and then ask pupils to answer one or two simple questions. This suggests that learning that was mainly focused on watching and listening, with a minimum emphasis on reading and writing, was preferable for him. Alex and his mother reported numerous punishments due to disruptive behaviour in school. Alex described good teachers as those who were 'fun and silly', and 'not strict'.

In summary, dyslexic music students reported preferring teachers who are flexible, relaxed and willing to try a variety of learning approaches. As they may require extra time to learn, dyslexic students were appreciative when a teacher offered extra time and patience in order to support them. Negative experiences related to scolding, punishment or abuse were reported by participants; this may suggest that some teachers do not have sufficient knowledge about dyslexia or the effect it may have on a student's learning process.

10.3.8 Table of student-reported useful resources

Table 10.2 lists resources which dyslexic students recommended as a means of informing their understanding of dyslexia or flexible approaches to learning music. Several participants noted that available resources on dyslexia were usually text-based and thus not optimal for them. This suggests the need for accessible resources in the area of dyslexia and music learning. Meg's mother taught the *Toe by toe reading approach* (see Table 10.2) and that is where she became familiar with the methods. The resources on dyslexia were located independently by participants, but some of the books on flexible teaching practices or performing psychology were recommended by teachers to the interviewees. The only book not directly related to dyslexia was Berg (2019) which focuses on components of foundational skills and good practice technique whilst making a case against common errors and mindsets which prevent these from being established.

Participant name	Resource
Aaron, Esther,	British Dyslexia Association resources
Julie	
Meg	<i>Toe by toe reading approach</i> ⁴³ , a highly structured phonics-based approach
Katie	<i>Playing by colour</i> music tuition book (Goodey, 2013)
Grace	In the mind's eye: Visual thinkers - gifted people with dyslexia and other
	<i>learning difficulties</i> (West, 2009)
Grace, Mel,	The gift of dyslexia (Davis & Braun, 2003)
Esther	
Grace, Julie, Keith	Music and dyslexia: A positive approach (Miles et al., 2008)
Victoria	The mind map book: Unlock your creativity, boost your memory, change your
	<i>life</i> (Buzan, 2009)
Mel	Dyslexia: A beginner's guide (Brunswick, 2012) Oneworld
	The secret life of the dyslexic child (Frank with Livingston, 2002) Rochdale
	Overcoming dyslexia (Shaywitz, 2003)
	The dyslexic advantage (Eide & Eide, 2011)
Keith	Instrumental music for dyslexics: A teaching handbook (Oglethorpe, 2008)
Peter	Zen guitar (Sudo, 1997) Simon and Schuster
Jane	Practicing music by design: Historic virtuosi on peak performance (Berg, 2019)
	Routledge

 Table 10.2 Resources recommended by students

My piano students Alex and Ben reported that their parents were their main source of information about dyslexia. Ben described a school assembly and a leaflet that he was given about dyslexia. He recognised similar struggles as those mentioned in the leaflet but realised that he was not as severe as others who cannot read at all. Ben stated that he did not look for resources on the internet about dyslexia, but felt that a series of short videos might be useful for parents who might have unrealistic expectations of their dyslexic children.

⁴³ https://toe-by-toe.co.uk/

In summary, participants reported obtaining information from books, parents or from online resources. However, some highlighted the lack of accessible materials, with text-based materials being the most widely available resource.

10.3.9 Student perspectives for envisioning change

According to participants, there were a number of areas in need of transformation across systems of individual, family, school and society. From a societal point of view, several participants noted that more people need access to assessments in order to receive accommodations which they believe may empower them to succeed. They called for more openness in workplace and institutional environments regarding neurodiversity and for more incorporation of training on assumptions made about dyslexic people and their challenges. At an educational level, participants reported the need for teachers to have awareness of the signs of dyslexia and felt that educational institutions needed to use a less text-based approach and improve their adaptations for different learners. This suggests that educational institutions might benefit from incorporating frameworks like the Universal Design for Learning (CAST, 2018) in their curriculum and teaching approaches so that they consider a variety of ways to engage the student, present information or materials and allow the student to express their learning (See Reflective Statement 6 for further explanation). One participant suggested a change in the mindset of what 'ability' means in academia (Nathan), whilst Keith suggested that musicians should be able to use diverse and non-traditional music-learning strategies without fear of repercussion, shame or any other sense of barrier.

On the music education level, Ned described a need for specialist *music* support, similar to the role of academic learning support tutors at his university. Participants noted that there were too many hurdles to accessing music exam accommodations and that this may prevent them from being more widely used. This also suggests that a lack of awareness of accommodations may prevent teachers and their dyslexic pupils from using them effectively. Exams and online learning packages which use 'click and drag' technology, multiple choice questions and which are timed are all seen as problematic for dyslexic

students; participants suggested that practical assessments were better ways of measuring their skill levels.

Information on music and dyslexia is largely text-based, and content on the internet is not always accessible for those with dyslexia. Participants reported difficulties finding information with web or application-based technologies, describing a sense of frustration in trying to navigate content hidden under layers of web-links. Peter suggested that a series of brief, engaging videos aimed at dyslexic music students might also be relevant to adults if the content and information were presented in an appropriate way. On an individual level, several participants expressed a desire to champion dyslexia and produce resources which would inspire and inform others. Grace had created a number of books and webinars for this purpose, whilst Julie encouraged other dyslexic students to pursue assessment and a confidence in developing their own learning strategies.

Both my students Alex and Ben referred to the wish for virtual reality technology that might help them learn to read music. Ben asked 'Is there a way to get rid of it [dyslexia]?'. This suggests that he views dyslexia as a disadvantage, and also indicates that there may be gaps in his knowledge of the nature and causes of dyslexia. These findings support the need for the availability of engaging and accessible resources for dyslexic music students. Short videos, like the type described by one of the interview participants, might highlight similar challenges and strategies dyslexic students might find beneficial. These might be used as a way to present the positive aspects of dyslexia as well as a way to reduce common myths. These findings also indicate that dyslexic students can envision ways in which technology might complement their music learning skills and reduce barriers for them.

10.4 Summary of findings

- Students recognised the importance of family support, particularly that of mothers, in preparing them for school, gaining knowledge about dyslexia, advocating and supporting them.
- Schools or teachers may resist parents' attempts to advocate for their children or may blame the parents for the student's difficulties. The findings suggest that

schools may be reluctant to acknowledge a student's difficulties due to financial implications.

- Without adequate training and knowledge, instrumental teachers may place blame on parental lack of involvement or students' lack of practice if they are not aware of how dyslexia may be affecting them.
- Findings suggest that an assessment of dyslexia may provide increased support to some students, especially at a young age, although older professionals did not perceive a benefit as they were established in their careers or did not intend to disclose in the workplace. The cost of an assessment may be weighed up against perceived benefits, such as additional support.
- Assessment information provides useful information for the instrumental music teacher. These findings suggest that the instrumental teacher's awareness of data protection issues and the importance of safe data storage and confidentiality is an important aspect of building and maintaining trust with parents and students.
- General challenges reported by participants include a slow pace and misreading of text, challenges concerning writing, processing, visual disturbances, basic maths, sequencing, motor coordination, time management, being overwhelmed by information, spatial awareness issues, following instructions, working memory, stress and anxiety during tests or in a classroom.
- Dyslexic students highlighted the importance of finding their own ways of learning.
 They recognise their strengths and weaknesses but also note that appropriate instruction and flexible education might produce better outcomes for them.
- Findings suggest that challenges with sight-reading music occur across a spectrum with some cases persisting despite attempts at remediation.
- Reading music was described as most challenging when it involved more than one stave of music, different clefs, or for singers when they had the additional burden of reading lyrics as well as the pitch and rhythm of the music.
- General strategies for reading music included using non-white paper, enlarging the score and highlighting troublesome areas as well as using methodical and structured approaches which might include recognising patterns or harmonic analysis. A number of students used recordings of lessons or sections of pieces as means of learning them.
- Some participants reported difficulties with rhythm, relating to reading rhythm, processing music symbols and feeling the pulse or rhythm in the music. Effective strategies noted by participants were to analyse the music, isolate the rhythm, break it down to the smallest component, use phonetic symbols, write the beats into the music, to clap or tap the rhythm and to practise it slowly.
- Challenges were reported with recognising pitches on the stave and relating them to the instrument, distinguishing intervals and recalling key signatures. Suggested strategies were to isolate and drill note recognition, interval recognition exercises, highlighting sharp or flat notes, visualising scales and using solfège to practise sightsinging.
- Challenges related to music theory include terminology, aural test preparation and information requiring sequencing. Students reported using online quiz applications as well as visual materials to develop recall.
- These findings suggest the importance of the teacher's role in supporting students with specific strategies, but also the need for an awareness of the implications of dyslexia on students' learning. Teachers would benefit from knowledge and training relating to preparation and support for students who wish to use accommodations and reasonable adjustments during exams.
- Students reported the use of technology for making recordings of pieces or their lessons, software which allowed them to isolate and practise core skills, arranging or editing music, modifying scores by highlighting specific areas or as a means of communication with teacher and parents.

- These findings from students indicate that the use of colour to highlight specific elements of the score tends to be more useful for them than an overlay or coloured paper. Highlighting may reduce the processing burden connected with certain elements, for example rhythm and phonetic sounds, and may help to distinguish between specific elements, for example separate hands or parts.
- Aural strengths, resilience and determination, creativity with improvising or composing music, verbal communication, problem-solving and performing strengths were reported by participants. With reference to some of these skills being developed compensatory mechanisms, they may be seen as protective and promotive factors when dyslexic students are faced with frustration due to challenges with slow sight-reading, poor coordination or executive functioning when learning music.
- Desirable teacher traits include flexibility, adaptability and a relaxed approach.
 Students recognised, with great appreciation, those teachers who had given them extra time and patience.
- Negative experiences related to scolding, punishment or abuse were reported by participants; this may suggest that some teachers do not have sufficient knowledge about dyslexia or the effect it may have on a student's learning process.
- Information was obtained from a number of sources: parents, books and online resources, although with the majority of materials based on text this limited their accessibility for dyslexic students.

10.5 Conclusion

The dyslexic music student voice may help us as researchers and teachers to better understand their priorities, challenges, strengths and views on the learning process. This may enable us to provide them with better music education experiences and prevent or reduce feelings of low self-esteem. Participants reported the significance of family support, particularly that of mothers, in helping and advocating for them through their educational journey and their assessment with dyslexia, as well as in the role of negotiating with schools or teachers. Early identification and an openness and positivity in the family's attitude toward dyslexia was seen as significant for participants, who specifically felt that dyslexic parents were able to empathise with their challenges. Participants reported a number of general and music-related challenges for which they had devised strategies, often with the guidance of a teacher or parent, to compensate for their difficulties. Technology was often used as a means of mediating aspects of learning which they found frustrating, with music recordings seen as particularly valuable. In light of dyslexic music students' reported strengths with aural skills, this is not surprising. Experiences with teachers were reported to be both positive and negative; but dyslexic music students reported that flexibility, adaptability, support and patience were appreciable traits. Whilst some resources relevant to dyslexia or flexible music-learning approaches were described by participants, the need for more accessible resources specifically related to dyslexic music students, their parents and instrumental or vocal teachers was highlighted.

REFLECTIVE STATEMENT 7: EMPATHIC UNDERSTANDING AND COMPASSION FATIGUE

RS7.1 Introduction

Dadds (2008) considers the role of 'empathetic validity' in practitioner research, distinguishing 'internal empathetic validity' as concerned with relationships at the microsystems level of the research and 'external empathetic validity' relating to the dissemination of the research at the exosystem and macrosystem levels (p. 279). As Bronfenbrenner and Morris (2006) noted, proximal processes at the microsystems level are bidirectional and by being made aware of the experiences of my students and their parents in our interactions, I felt compassion and empathy for them and have grown in my respect for their sustained determination in the face of the challenges that they face. My hope was that they would sense my understanding and concern for them, even if there were issues which I did not completely understand, and for them to be aware that I wanted our connection to be positive, authentic and hopeful.

In this reflective essay, I consider the emotional transition I have undergone in relation to my role as a teacher-researcher in developing greater empathy at the microsystem level toward my dyslexic students and their parents; this sense of compassion and desire to advocate for people with dyslexia has motivated me to bring about transformational societal change at the exosystem and macrosystem levels of development (Bronfenbrenner, 1979). An additional area of growth has been the recognition of the toll that this compassion may take on me as the teacher, and my development in recognising the signs and preventing compassion fatigue, or burnout, by maintaining a healthy perspective and professional boundaries (Rosefield, 2023).

RS7.2 Context

Coming from a family background without any incidence of dyslexia, I was unaware of the inequalities which are faced by individuals with dyslexia. I was also unaware of the challenges of obtaining assessments, ensuring reasonable adjustments and accommodations were being used effectively and of the toll and cost of energy and finances

that this takes on families. Dyslexic students may face stigma and misunderstanding in the school system and with their peers. In my reflective journal, I noted my irritation when my student Ben told me about asking a teacher at school not to speak so quickly because he struggled to take notes in time. The teacher never responded to his request. Although I understood that the teacher was most likely constrained by the need to meet curricular requirements in a timely manner, I imagined the frustration I would feel if my requests for help were ignored or minimised, and the powerlessness that would likely cause me to feel.

My student Alex described how when he misunderstood directions in a lesson at school, the teacher became angry with him and made him do the entire assignment over again. His mother, Erin, related how 'Alex hates school and finds it hard to sit still and concentrate. Then gets into trouble quickly whenever he isn't coping'. I asked Erin, Alex's mother, if there were any positive relationships with teachers at school and the one-word answer 'No' revealed the extent of her frustration. There were no practical solutions offered nor understanding as to the root causes of the behavioural difficulties. This correlates with Dahle et al. (2011) which suggests that teachers might not attribute behavioural difficulties to dyslexia or may feel that they would be labelling the child if they did; parents in my research appear to have to persist in communication with schools and teachers to attempt to achieve better understanding of their child and their needs. In the literature review of this thesis (Chapter 2) the secondary characteristics of dyslexia (Livingston et al., 2018) have been discussed. National statistics provide a bleak educational picture, as the BDA (2019) report that compared to non-dyslexics, dyslexic students are 'twice as likely to fail to achieve grade 4 or above in English and maths at GCSE' (p. 5) and are 'three and a half times more likely to be temporarily or permanently excluded' from school (p. 6). These statistics highlighted the challenges that my students faced and I felt compassion for them and their parents. I recognised the need for teachers (at school) to be better supported in terms of training and with additional support staff and I felt discouraged that due to financial constraints, this was not always a possible solution.

RS7.3 In my relationship with parents

Two critical incidents occurred with parents, both of which resulted in an increase in my empathic understanding which led to changes in my teaching practice. In the first incident, I became aware of a slightly defensive response when I followed up a request for the student to have access to an app which would help him build his recognition of notes on the stave.

I have noticed that I have advised his mum to get these apps (which are free or very low cost) and I'm not sure it has happened. Perhaps I haven't stated my case strongly enough. However, I feel that much more progress would be made if he was able to practise these skills more regularly. Note: at the next lesson, his mum asked me to send an email with the app names. I need to remember that my dealings with her must be dyslexia friendly as well – just telling her something at the end of a lesson is not enough. Also, I need to show her how the apps benefit her son's learning.

An instrumental teacher may not always know the reasoning for a perplexing lack of cooperation from a parent, particularly a dyslexic parent, in certain areas. The teacher may not always know that a parent is dyslexic. We cannot underestimate the trauma of some of the negative experiences they may have had in the past. As they observe their child struggle in similar areas, parents may feel exposed and responsible, and this may provoke unpleasant emotions (Leitão et al., 2017). Rather than speculate, the most productive thing we can do is to try to create an environment of open and authentic communication so that even vulnerabilities might be safely shared (Wilmot et al., 2023). My other action point after this incident was to highlight to parents the importance of reinforcing learning by applying specific practice techniques or repetition at home. This seemed to work best with some notice in advance that resources should be expected, checking in with them during the week and then following up with appreciative messages for their support and feedback on the student's progress.

In another situation, the mother related that due to the structure of the school day, their dyslexic child did not receive extra time during significant exams. This was the student's usual way of working, and extra time had been noted as a beneficial reasonable adjustment

in his dyslexia assessment. When I expressed my surprise about this, the parent's immediate response was to express guilt and regret for not having been able to advocate more effectively for him. I never intended for that to be the result of my words. From my point of view, I felt the school had been negligent. However, I had a discussion with a specialist for dyslexic students who related that this was common and that schools were not obliged to provide dyslexic students with reasonable adjustments and accommodations which did not work within their timetables.

There are value judgments that must be made by schools and exam boards in determining whether reasonable adjustments are awarded. This exposes some of the shortcomings within educational systems in providing dyslexic students with their established rights under the Equality Act 2010 and further highlights the difficulties schools might have with implementing reasonable adjustments. As an instrumental teacher, it expanded my understanding of how my legitimate concerns for students may contribute to an increased sense of burden for parents in a situation which is out of their control, and this taught me to exercise more caution and sensitivity in future communication. It also expanded my awareness of the significance of the challenges faced by parents in attempts to advocate and negotiate with schools.

RS7.4 Externalising and internalising behaviours

I was aware of the differences in confidence levels between my dyslexic students and how this manifested itself in their outlook on trying new approaches or pieces of music. Behavioural issues also affected my students in different ways. One student displayed a loss of concentration and would rub his eyes, whilst another student would be restless and would fidget with the piano keys and pedals. During an online lesson, a student behaved so poorly that a parent had to intervene at one point, but I felt this externalising behaviour, demonstrated through hyperactivity and a loss of concentration, might also be attributed to the social isolation and uncertainty of the Covid-19 pandemic as well as a family house move. I recorded a conversation with one mother who described some uncharacteristically angry episodes at home during a time of significant stress at school. We discussed whether this might be a form of 'letting off steam' in a safe space as a way of dealing with the pressure from a specific teacher who had reprimanded the student. I was aware in our lessons that I needed sensitivity and care in my manner, choice of words and tone of voice and that I was privileged to have information about their experiences shared with me. Singer (2005) notes that children with dyslexia may be reluctant to discuss their problems with their parents and teachers. I knew that they did not want to be treated with pity or condescension. At the same time, I was aware that in my desire for them to achieve their full potential, I could overestimate what level of challenge they had the emotional and mental capacity to handle, along with school and other life demands, and my intentions could backfire. I recall at least one example of this where the student put up a wall of defence, and it led me to feelings of guilt, frustration and disappointment in myself.

It was only through reflecting on how the student coped with the situation that I realised they had been internalising how they really felt. As much as I wanted the students to be able to communicate freely and openly with me, sometimes this led to ambiguities, and I had to look past them for deeper meanings. Beyond hiding their true feelings, I also observed an indecision about goals and the choices they made. During one period, we seemed locked in a cycle where the student would choose a piece to play, I prepared resources and recordings, they changed their mind and chose another piece, and the pattern went on for several months. I felt irritated at the wasted hours of preparation and I felt frustrated because we did not seem to be making progress in meeting any goals as the goalposts were changing so often. I observed that excuse-making might be a way of masking a difficulty with the tasks themselves. Yet, when I asked questions about what might be more suitable or how they might like to approach this situation, the answers were vague, and it became apparent that there were other issues at play. I observed them deflecting questions as a coping mechanism when faced with challenging situations.

As a result of understanding that a change was needed in my emotions and approach, especially when we were looking for new pieces to play, I took another approach. I described to them how, as a child, I would go to the local library and choose a huge stack of books. I would come home and read a few pages of each book to see which one I wanted to read through first and then climb a tree in my front garden and read it. Some of the books were never read to the end. In telling them this story, I hoped to convey, or normalise, the idea that it was perfectly fine to sample pieces and that learning was still taking place even if they did not have a polished product at the end of a term. Also, this reinforced that it was fine for them to change their minds about a piece. This was a transition in my mindset and emotions from past teaching practices. Some pieces became favourites and others were discarded quickly. I observed a change in the students as it seemed to relax some of the tension or pressure they felt, and each trialled piece had something to teach us.

RS7.5 Masking behaviour and consequences

Singer (2005) refers to the strategies children with dyslexia may use when trying to protect their self-image as 'hiding, working hard, fighting back or explaining' (p. 421). Understanding that these are common reactions used to preserve self-esteem helped me to develop more empathy for their actions and to look for prompts or cues that might indicate how they were feeling. Masking strategies might include copying the behaviours of others, using prescripted responses and hiding uneasiness from a sensory overwhelm response (Kidwell et al., 2023). As a type of surface management, or emotional regulation, individuals may conceal or disguise their emotions (Grandey, 2000). This can be seen as a way to manage identity in different contexts and has been shown to lead to poor mental health and emotional well-being, including resource and ego depletion (Hülsheger & Schewe, 2011). Learning that ego depletion is linked to poor self-regulatory skills and resource depletion to poor mental health outcomes emphasised for me the importance of creating an atmosphere of empathic understanding for my students.

RS7.6 Developing empathic understanding

Empathy is not prescriptive or theoretical, instead it is communication which allows an 'identification, elaboration and transformation' of what someone else is feeling (Gardner, 2024, p. 81). Several factors may have been driving the students' behaviour based on their disposition, their needs, how they were coping at school or through the pandemic and the ways in which they viewed themselves. These factors were not static. Gardner (2024) suggests a process of focusing not only on the feelings, but the intentions of another person, and working with what they are willing to say, rather than what they may not be saying yet. Each student is unique and the dynamics in each relationship are different. By realising that my interpretation of each situation would not always be right, I allowed myself to sit with that uncertainty, adapt to the student's response and offer them the space to correct me. I tried to be aware of any words or phrases which might act as triggers for them. Even talking about dyslexia with them caused awkwardness at times and felt uncomfortable for us both. My own experiences and positionality were the lenses through which I viewed the students, and this affected my ability to empathise fully with them. Being in the roles of teacher and researcher and not having had the experience of dyslexia or learning music as a dyslexic person meant that I had to focus intensely on observing their behaviour. At times, this felt like an overwhelming and impossible task; in addition, factors such as the pandemic, being a mother, running a small business and working as a graduate teaching assistant along with other personal challenges meant that I was pulled in many directions and the situation took its toll.

RS7.7 The importance of preventing compassion fatigue

Observing the struggles that my students and their parents faced with teachers, schools and systems, as well as knowing that, unintentionally, I may have caused shame or discomfort for them was overwhelming at times. I felt exhausted from the intense focus of trying to unpick their feelings, perceptions and motives during lessons and from the level of observation that was required of me. I felt a sense of injustice at the unfair systems of assessment and inequalities in education that I was observing. There was another tension which developed from the multiple perspectives I was hearing through my research. Both of

my students had exceptional family support, and yet I could see how challenging things were for them and their families. How much more would this be exacerbated for a child from a low socioeconomic background who lacked parental or community support?

Throughout my life, I have been sensitive to injustice and feel deeply for others when they are treated unfairly. I recognise this is one of my dispositional characteristics (Bronfenbrenner & Morris, 2006), and it has benefits and drawbacks. What I viewed in a hospital for abandoned children with HIV in Romania in 1999 led me to establish a home for them which is still providing care to those who have survived to this day. But there were times when my identification with the horror of their situation was so deep that my emotional well-being was affected. Glaser (2014) notes that by observing others' pain, we also may have an experience of pain; an empathic response which can lead to compassion fatigue, or burnout (Rosefield, 2023). Maslach and Leiter (2016) describe burnout as having three elements: 'exhaustion', 'cynicism' and 'inefficacy' and suggest that prevention is better than dealing with complete burnout; they note that re-evaluating schedules, developing resilience, spending time with family and friends, relaxation, nutrition, exercise, spirituality, a positive mindset and greater self-awareness may help in dealing with stress (p. 103). It was those experiences which led me to understand that I must develop a professional stance in balancing care for myself as well as care for others and maintaining healthy boundaries. Signs of this were when I began to feel disinterested, unmotivated or found it difficult to disengage from thinking about challenging aspects of teaching or research.

I knew that I would be a better teacher if I took breaks and time to feel rejuvenated. However, there were tensions as I tried to balance highly competing demands, for example, wanting to reduce my teaching schedule with the need to pay household bills or wanting to commit to regular exercise but facing time constraints. The support of others, my friends and family, helped immensely to keep me grounded in a healthy perspective. I saw evidence of the valuable role of peer support in a visit to an inclusive practitioner's meeting with Music Education Hubs East Midlands (MEHEM Uprising) and at Music Mark conferences. In particular, the MEHEM meeting, with a setting which was not school or office-like, enabled a

feeling of retreat. Self-reflection was not always comfortable, but it played a role in allowing me to create a distance between events which occurred and the time needed to process my feelings and to consider new ways of moving forward (Dadds, 2008). Also, it helped me to show compassion to myself, particularly during the pandemic, when so many factors were out of my control. Still, I found it difficult to put a limit on the high expectations and standards I had set for myself, and which in the past I have been prone to hold on to stubbornly even to my own detriment at times.

One aspect which motivated me was the hope that I might be able to make a difference in someone else's life and bring about transformative change. However, I recognise that 'advocacy fatigue' may be an additional cause of burnout when facing interpersonal and systemic complexities (Maslach & Leiter, 2016). In myself, my students and their families' lives, I have shown some evidence of a development of internal empathetic validity. The following comments from the students' mothers give some indication of this, as Erin shared that 'He [Alex] is loving piano and is getting the affirmation that he needs and that means that he is engaged with it. It's something he enjoys and feels well doing' (Erin). And Ben's mum, Polly, stated:

Ben has been wonderfully taught in piano. This is probably the only area of his learning where I feel he has a teacher who really understands him and who is able to tailor and amend teaching techniques so that he can reach his potential. (Polly)

After delivering a workshop on dyslexia to teachers in Manchester in 2023⁴⁴, a teacher came to me and let me know that the workshop had helped them, as a person with dyslexia, to understand themselves better. Alongside the development of empathic understanding with my students and their families, I also seek to demonstrate external empathetic validity by improving music education for dyslexic students who would like to learn music, and for parents who may need support in their roles, and in preparing instrumental music teachers to have a greater awareness of dyslexia, risk-resilience factors affecting their students and a foundational knowledge of best practice in approaches and strategies. In sharing my

⁴⁴ Music Mark Northwest Music Hubs Conference, September 4, 2023

experiences more widely, my desire is for a greater recognition of the emotional labour costs of teaching dyslexic students, for teachers to be aware of the signs of burnout, to recognise systemic demands, to call for and enact change where possible and to demonstrate sustainable caring for others.

Chapter 11 DISCUSSION OF FINDINGS

11.1 Introduction

Previous chapters of this thesis (Chapters 5-10) presented the analyses of data throughout this study. This chapter provides an overview of the findings and Reflective Statements 1-7 and discusses these against the contextual backdrop of the literature review of music and dyslexia presented in Chapter 2. The literature reveals that dyslexia is a complex condition from a scientific standpoint, with many facets still to be understood. Additionally, there is limited music teaching and dyslexia literature available, with the majority consisting of anecdotal case studies from music teachers or dyslexic musicians. The lack of literature and research provided further impetus for this study which conceptualises music education and dyslexia from a pedagogical and holistic systems perspective in order to improve and enhance instrumental music teaching for dyslexic students.

The data used to support the findings discussed in this chapter include a questionnaire which was used to investigate the perceptions of music exams amongst dyslexic students. The data collected provided insights into the experiences and perspectives of these students. Interviews were conducted with students, parents and teachers. These gathered qualitative data and personal experiences related to instrumental teaching for dyslexic students. I also analysed reflective observations from two case studies within the context of my piano teaching practice. These observations provided firsthand insights into the challenges and successes of teaching dyslexic students.

In this chapter, the discussion is presented in relation to the research questions. Section 11.2 discusses the perceptions of music teachers on their experiences and pedagogical practices (including strategies, methods and materials) in teaching dyslexic students. The second section (11.3) addresses student and teacher perceptions of music exams, including the use of reasonable adjustments and accommodations. The third section (11.4) presents perceptions of the strengths of dyslexic students and how teachers might identify and utilise those in the lesson context. The fourth section (11.5) addresses the student voice in the learning process in view of findings. In light of the overall findings, the roles of student, parent and teacher interactions in the pedagogical process (11.6) are examined in relation to risk-resilience

literature, resulting in a new model (11.7) which conceptualises how promotive and protective factors at a number of systemic levels might enhance the development of resilience in dyslexic students. Section 11.8 presents the analysis of feedback in relation to the findings and the new model from an evaluative focus group of stakeholders with expertise in inclusive music education. Some of these topics are discussed from multiple perspectives in this chapter as there was considerable overlap between categories.

11.2 Research question 1

RS1 1. What are the perceptions of music teachers regarding their experiences and utilisation of pedagogical practices, including strategies, methods and material, in teaching dyslexic students?

Although some literature (Oglethorpe, 2008; Miles et al., 2008; Morrow, 2023) presents case studies of teachers and their approach to dyslexic students, there is limited research investigating the wider perceptions of music teachers' experiences, how they obtain information and put pedagogical approaches into practice as well as a lack of research on teachers' use of graded music exams with dyslexic students (discussed in section 11.3). Several themes emerged from the data collected from teachers' interviews. This study extends the understanding of teachers' barriers and challenges (11.2.1), their observations of dyslexic students' difficulties as well as the strategies utilised (11.2.2). Teachers' perceptions of students' strengths will be discussed in section 11.4 of this chapter.

11.2.1 RQ1 Barriers and challenges

This section discusses the importance of the teacher's role in the pedagogical process and the need for more research, training and resources related to music and dyslexia. Challenges faced by the teacher are discussed, as well as the means of mitigating some of these through the development of a positive teacher-student relationship. This highlights the value of the teacher-student relationship in facilitating effective learning and support for dyslexic students.

11.2.1.1 The impact of a lack of training and resources

These findings indicated that many teachers have not received adequate training nor have access to specific resources to effectively support dyslexic students. This can make it challenging for them to understand the unique learning needs of dyslexic students and to

implement appropriate instructional strategies, leading to stress and anxiety for the student, confirming the findings of Francis et al. (2019), Novita (2016) and Singer (2005). This limited awareness may lead to misconceptions and assumptions about the condition which can result in ineffective teaching approaches and a failure to address the specific challenges faced by dyslexic students. The impact on students' motivation and emotional well-being is documented by Wilmot et al. (2023).

Teachers in this research study face challenges in balancing the difficulty levels of instructional materials and tasks. They need to provide appropriate challenges to promote growth and learning whilst ensuring that tasks are not too overwhelming for dyslexic students. This can lead to poor behavioural outcomes for the student (Humphrey & Mullins, 2002). This can also create tensions if there is a mismatch between high parental expectations and appropriate support for the student, as identified by Oglethorpe (2008). Unless teachers are actively learning about dyslexia, they may not be able to help students develop a more positive and realistic understanding of their own abilities. Another challenge is navigating behavioural issues whilst maintaining a positive and inclusive lesson environment.

11.2.1.2 A need for inclusive schools and working environments

The literature review in Chapter 2 revealed a need for more inclusive music education practices (Fautley & Whittaker, 2018) and research on disability in the arts (Cox & Kilshaw, 2021). This suggests that peripatetic teachers may face challenges due to limited resources and support from the school, including a lack of specialised intervention programmes, assistive technology or access to additional support staff, making it more difficult to meet the needs of dyslexic students. The findings of this research indicate that teachers, especially those who are neurodivergent or have other forms of hidden disability, might be unwilling to disclose their condition due to stigma or shame; they may find aspects of teaching or administration challenging due to their dyslexia. This indicates a need for more inclusive practices in the workplace environment and with colleagues and highlights the importance of creating a supportive environment for all teachers along with the need for more research examining how this might be best implemented. It is important for schools and employers to provide ongoing professional development, resources and support to teachers to address these challenges and ensure effective instruction for dyslexic students, as confirmed by Hourigan (2007) and Van

Weelden & Whipple, 2005) training in inclusive teaching practices might increase the confidence of teachers to use inclusive teaching practices with all students.

11.2.1.3 Managing parental involvement

The findings of this research suggest that teachers may feel under-equipped or concerned about managing parental expectations when teaching dyslexic students. This might be seen in a reluctance to teach dyslexic students and suggests a need for training and information. Music exams are discussed in greater detail in Section 11.3, but the findings of this research indicate that knowledge is needed related to selecting examination boards, obtaining information about reasonable adjustments and preparing students for exam settings for teachers. Further emphasised is the importance of parental involvement and the management of expectations in building positive relationships between teachers and parents, which correlates to points made by Oglethorpe (2008) and Macmillan (2005). Relationships with parents are discussed in Section 11.6 in greater detail, but teachers' views confirm the importance of parental involvement whilst also acknowledging that much depends on the context of the relationship in terms of recruiting parental involvement and the management of expectations, aligning with Catts and Petscher (2022), Oglethorpe (2008) and Wilmot et al., (2023).

Challenges identified in this study are congruent with those described by Oglethorpe (2008) with variables of dyslexia disclosure linked to parental beliefs about dyslexia and their understanding of its relevance for the music teacher. Teachers observed the difficulties that parents encountered in advocating for their children at school, believing that schools lack the finances and resources to adequately support dyslexic students. This highlights again the systemic disadvantages and the need for schools to provide adequate support and resources for dyslexic students, which correlates to reports (DfE, 2023) that students face substantial delays in receiving educational and health care plans (EHCP) and that schools need to be incentivised to improve support for SEND students (National Audit Office Report, 2019). It also emphasises the importance of teachers using tact and compassion when speaking to parents, especially those with dyslexia, who may have negative educational experiences in their past and feel a sense of anxiety or stress, similar to other research findings (Leitão et al., 2017; Wilmot et al., 2023).

Whilst teachers tended to identify dyslexic students through general lesson observations or by comparison with other students, these methods may lead to incorrect assumptions about students. Singer (2005) described this as the student 'hiding, working hard, fighting back or explaining' to protect their self-image (p. 421). Similar to the literature on secondary characteristics (Livingston et al., 2018; Novita, 2016; Singer, 2005), teachers were aware of some students' low self-esteem, as manifested through body language and self-castigation. Teachers attempted to interpret students' behaviour, recognising that indications of stress might result in hiding their difficulties and emotions or practising a type of surface management (Grandey, 2000; Kidwell et al., 2023). Schwabe & Wolfe (2013) recognise the impact of stress on the student's capability to learn. However, these secondary characteristics might not relate so much to self-esteem as to the student's self-efficacy, self-determination and an appropriate learning environment (Burden, 2008).

11.2.1.4 Potential benefits of a positive teacher-student relationship

The findings of this research indicate a number of potential benefits from positive relationships in the pedagogical process, consistent with Oglethorpe (2008), Seligman (2011) and Armstrong (2012). When students feel valued, supported and connected to their teachers, they are more likely to actively participate in lessons, ask questions and take ownership in the learning process. Thus, their engagement in the learning process is enhanced. Positive relationships can foster a sense of belonging and intrinsic motivation in students. Teachers who establish positive relationships can inspire and encourage students to strive for their goals. When students feel secure, they are more likely to seek help, ask for clarification and take risks in the music lesson, aligning with (Singer, 2005) and Goldfus (2012). This can lead to improved understanding, retention of information and overall achievement. A positive relationship with the student can contribute to the student's emotional well-being. Teachers who create a supportive and caring learning environment may help students to feel safe, respected and understood. This can reduce stress, anxiety and feelings of isolation, promoting positive mental health and emotional development, which aligns with Seligman (2011) and Singer et al. (2013). Furthermore, positive relationships with students may contribute to the development of their social skills. Teachers who model positive communication, empathy, and respect can help students learn how to build and maintain healthy relationships. This can have

long-term benefits for students' interpersonal skills and their ability to navigate social interactions in various contexts.

It is important to note that these potential benefits are supported by a teacher having knowledge of dyslexia and research. The obvious conclusion is that when teachers feel supported and knowledgeable, they are better equipped to navigate the relationship with the student and the parent, as well as overcome some of the challenges which they reported. By providing reassurance and encouragement, teachers may help students to develop a positive self-concept and boost their confidence in their abilities, as confirmed by Riddick (1996) and Armstrong (2012). By offering appropriate modifications and instructional strategies teachers might empower students to progress. The recognition of effort reinforces students' belief in their own capabilities and motivates them to continue working hard, similar to the PIMMS model described by Patson & Waters (2015) in Reflective Statement 5. This gives further support to the strengths-focused approach which is discussed more in section 11.4. Overall, this emphasises the importance of the teacher's role in the pedagogical process and the need for ongoing support, training and resources to teach dyslexic students effectively.

11.2.2 RQ1 Teacher observations of students' challenges and corresponding strategies

Teachers observed patterns of challenges which were aligned with the literature described in Chapter 2. Insights from strategies recommended by teachers for dyslexic students may be valuable for music educators in different contexts as they work with diverse groups of students.

11.2.2.1 Teacher observations of dyslexic students

Consistent with Brimo et al. (2021) who highlighted the challenges created by co-occurring conditions, teachers recognised that identifying specific causes for difficulties was complex. The spectrum nature of dyslexia means that untangling the root causes to support the development of specific musical competencies may require patience and sensitivity on the part of the teacher. The primary challenges observed by teachers were related to slow processing, consistent with Peter et al. (2011), and difficulties with executive function skills, such as concentration, organisation, working memory and sequencing skills, again consistent with the literature (Smith-Spark, 2016; Akyurek & Bumin, 2019). Other areas of challenge reported by teachers include coordination, spatial awareness and visual stress, similar to

reports from the literature (Brimo et al., 2021; Kriss & Evans, 2005; Oglethorpe, 2008). The findings from teachers' observations agree with literature describing difficulties with sightreading (Ganschow et al., 1994; Jaarsma et al., 1998; Oglethorpe, 2008), with teachers reporting a divergence between students who struggled more with pitch or with rhythm reading. This may relate in part to the experiences of teachers of diverse instruments in this study. An important limitation to consider in this study is that teachers' responses may also be influenced by existing literature on the subject of music learning and dyslexia.

11.2.2.2 General strategies

General strategies reported by teachers included physical demonstration, simplification of concepts or materials and kinaesthetic approaches (for example, Dalcroze methods) to learning. A subset of teachers, with an awareness of dyslexia-literacy as well as an understanding of the music and dyslexia literature (Ganschow et al., 1994; Hubicki & Miles, 1991; Miles et al., 2008; Nelson & Hourigan; 2016; Oglethorpe, 2008; Vance, 2004) designed lessons with a greater understanding of the impact of dyslexia on the student and with an awareness of effective strategies.

11.2.2.3 *Multisensory teaching strategies*

Multisensory teaching approaches are emphasised as they engage visual, auditory and kinaesthetic neural networks. Goswami (2015) affirms the importance of multisensory teaching approaches, stating that 'Learning depends on neural networks distributed across multiple brain regions: visual, auditory and kinaesthetic' (p. 4). How teachers conceptualise the use of multisensory strategies varied greatly across the participants, with some understanding repetition to mean rehearsing the same information repeatedly until learned and others revisiting concepts from time to time, or a combination of these. Maintaining students' interest and motivation as a consideration within the Universal Design for Learning (UDL) (CAST, 2018) framework (see Chapter 2 and Reflective Statement 6), is of paramount importance, whilst focusing on overlearning may be challenging as there is potential for the student to lose motivation from boredom or disinterest. On the other hand, in Reflective Statement 6, I observed how my students were open to repetition and I wonder if perhaps it was useful for promoting a positive spiral in their learning due to their familiarity with the material.

Morrow (2023) describes multisensory teaching as a complex process, having the student 'interact tactilely with the page, touching while simultaneously naming each note, marking beats in the measure, and then vocalizing the rhythm with the metronome while touching the beats and naming the notes' (p. 25). Whilst this process is systematic and structured, there is a question as to how it would sustain the interest of a dyslexic student, particularly one with attention or concentration difficulties, who simply wants to achieve the success of playing a piece. Morrow (2023) states that 'Unfortunately, when music reading is relegated to the second tier of the learning process, it is forced to follow the logical instructional order of making music instead of reading music' (p. 21). However, it is difficult to see why music making and music reading could not be taught simultaneously. The students' perspective on Morrow's (2023) method would provide additional insights. Another consideration is how the emphasis on one mode of teaching can be reconciled with the knowledge that dyslexia occurs across a spectrum, with unique profiles and frequent co-occurring conditions. As reported in the findings from teacher interviews (Chapters 6-8), two dyslexic students in the same teaching practice may need substantially different approaches.

11.2.2.4 Approaches in relation to visual stress

Teachers should be aware of the importance of signposting students to relevant professionals for visual difficulty assessments. Teachers might benefit from being made aware of good practice in relation to visual stress; for example, advice from the Specific Learning Difficulties Assessment Standards Committee for practitioners on supporting students with visual difficulties (SASC, 2018/ 2019). However, as noted by Suttle et al. (2018) in Chapter 2, there is a need for greater research in the area of coloured overlays or lenses. This is an additional area where knowledge and training would be beneficial in order for teachers to have accurate knowledge about dyslexia and the area of visual stress.

11.2.2.5 Specific strategies for sight-reading challenges

Teachers articulated several specific strategies addressing challenges with reading music:

1. Using modified scores that highlight specific aspects of music, such as important notes or rhythms, to help dyslexic students focus on the essential elements of the piece.

- 2. Helping dyslexic students develop pattern recognition skills, which can aid in sightreading. By identifying these patterns in music, students can anticipate and read the notes more efficiently.
- 3. Incorporating repetition through flashcards, games or software to reinforce skills. This repetition helps dyslexic students become more familiar with musical notation and improves their ability to read music.
- 4. By listening to recordings and engaging in physical demonstrations to reinforce an understanding of rhythm and melody, teachers encourage students to learn through aural repetition and recall.
- 5. Using Dalcroze and Kodály methodologies, which emphasise movement and kinaesthetic learning. These help the student to engage with music through physical actions, enhancing their understanding and retention of the rhythm or pulse. This also includes the use of rhythmic syllables or vocalisation and repetitive practice with common rhythm patterns.
- 6. Only one teacher reported using Figurenotes (Drake Music Scotland, 2024), an alternative notation system, in their teaching. This suggests that teachers may not be aware of alternative systems and may need training in how to select and use them effectively with students. This visual strategy helps students to denote the duration of notes or rhythms, such as using different colours or symbols, and reduces the processing burden.
- 7. Technology was used to modify scores and to provide an aural representation of the score through the playback function, in addition to the games mentioned above.

Although teachers did utilise many of these strategies and applied elements of UDL in multiple ways of engaging the student, representing the music and expressing what they had learned, none of the teachers explicitly mentioned the use of a framework to incorporate and support a more inclusive approach to their teaching. In research with pre-service teachers, Lowrey et al. (2019) suggest that as a means of incorporating UDL into lesson planning, teachers might benefit from pre-designed templates and mentorship to promote their identification of obstacles to students' learning and note the need for an investment of time and resources to design purposeful lesson plans. Teachers may find this resource and a supportive collaboration with other teachers impactful.

11.3 Research question 2

RS2: Do adjustments and accommodations which music exam boards allow offer dyslexic students unbiased inclusion? What are the perspectives of teachers and students on this topic?

Due to a paucity of research regarding teachers' and dyslexic students' experiences of graded music exams, it is not possible to provide a comparison from literature with the findings from the questionnaire and interviews with teachers and dyslexic students. Whilst dyslexic students are motivated to take graded music exams, a variety of barriers were reported. Specific requirements of exams may act as a bottleneck to dyslexic students' progression; for example, the amount of repertoire needed for a recorded performance exam, the aural test or sight-reading components in face-to-face settings or the format of questions or tests on music theory written exams. A greater degree of flexibility might be welcomed in the length and type of exams offered, as students recognised challenges with representing the information they had learned in music theory exams or found that the preparation placed high demands on their time. Whilst extra time as an adjustment is clearly valued, the means by which scales, aural tests or sight-reading are presented in the practical exams might also be considered in terms of accessibility. Extra training and support for examiners who implement these arrangements might enable greater sensitivity to students' needs and ways to reduce anxiety in the exam setting.

Practical exams which used to be 'face to face' are now offered online, an initiative that was designed to support students' exam taking during the Covid-19 pandemic which has remained a permanent feature. For ABRSM (Associated Board of the Royal Schools of Music), students do not have to take the sight-reading, scales or aural test components of the exam but can play three pieces from the syllabus and an own-choice piece that is at the same grade level. Whilst this may make the exam more accessible in some respects by avoiding elements some dyslexic students might find challenging, this also raises concerns in terms of accessibility of recording space, quality of instrument and recording materials and a change in marking criteria with a potentially subjective emphasis on 'performance skills'. It might be useful to consider more flexibility in terms of the types of exams or the number of pieces required in order to make them more accessible to dyslexic students. Another factor of discussion might be

whether this format of exam could be offered as a live exam if that was the student's preference.

The accessibility of the exam board website is a factor which should be carefully considered for all users, as it should not be assumed it is only students who have dyslexia. Exam boards could enhance their understanding of these issues by soliciting feedback directly from the users. The choice of colours and spacing around images and text might also be more carefully considered, as well as flexibility in the customisable options. Additionally, accessibility of content and the extent to which it is 'nested' in the main menu may influence how accessible it is for individuals with dyslexia. In a systematic review of dyslexia and web accessibility, Enco-Jáuregui et al. (2023) note that many websites do not follow current web accessibility guidelines and suggest that dyslexic users need customisability in terms of font format and size as well as the possibility of customising other elements on a page.

Since it is often the instrumental teachers who access the website, exam boards might consider ways to promote the understanding and use of access arrangements. However, this also raises the question of roles and responsibilities between teachers, parents and exam boards and suggests a potential need for exam boards to recognise their influence and to commit to widening their support of teachers and parents in inclusive practices. Teachers might benefit from practical resources which show them how to prepare students to use the arrangements effectively in the exam setting. This indicates that there might be additional training or promotion of reasonable adjustments needed for parents and students, perhaps with a more accessible approach, for example, through a variety of social media platforms or in a series of webinars.

11.4 Research question 3

RS3: Do teachers, parents and students perceive dyslexic students to have specific strengths? How might these strengths be utilised in the music lesson context?

Dyslexia is viewed by society as a disorder or disability, and interviews with teachers and students in this research, reflect an emphasis on deficits. At each level of the bioecological model (Bronfenbrenner, 1979), from school to the workplace, experiences of failure leading to

a sense of shame and stigma were reported by students in this research, aligning with previous research (Livingston et al., 2018; Wilmot et al., 2023). In this study, one area of focus was the identification of strengths and how lessons for dyslexic students might be designed, not only to identify those areas of strength, but as suggested by Louis (2011), to develop them further and consider how students might build on them to generate resilience and a greater sense of agency (Frederickson, 2004). The rationale for this approach has been set out in Reflective statement 5, which discusses my development from a deficit-based perspective to a strengths-focused approach. The strengths identified by teachers and students (11.4.1) and the ways in which teachers might utilise these in the lesson context (11.4.2) are discussed.

11.4.1 Strengths identified

Literature related to the strengths of dyslexia related to music learning and performance is somewhat limited (See Chapter 2 and Chapter 8 of this thesis). This discussion examines the perceptions of teachers and students in relation to strengths and explores the similarities and differences in their views against the backdrop of the literature (Chapter 2 of this thesis). Literature suggests that strengths may be a means of compensating for difficulties (Abraham et al., 2012), whilst another view is that information is processed differently and this may convey certain benefits related to creativity (Tafti et al., 2009). Other research suggests that differing brain structure or functioning in dyslexics may lead to specific advantages with visuospatial learning (Nicholson, 2014; Schneps et al., 2012).

Overall, students' responses were much more limited than teachers' responses regarding dyslexic strengths. This suggests that individuals with dyslexia are more likely to be aware of and focused on their challenges, and unaccustomed to thinking about themselves from a strengths-perspective. This lends further evidence to the rationale for teaching which is driven by a strengths-development approach, consistent with the literature which highlights the benefits of positive relationships and engagement (Frederickson, 2004; Maier, 2014; Seligman, 2011).

A theme which emerged strongly from students was the concept that they had the ability to find their way of learning, and when their teachers connected with that way of learning, they were able to make progress. Conversely, when they were forced into systems of learning or assessment which did not correspond to their abilities, often the result was failure. This was

the case for a student, Aaron, who was interviewed and reported how he failed GCSE exams with one board but passed them with another which used different forms of assessment. A parallel example given in previous research concerned Dave Brubeck's difficulties with sightreading which were hidden by his masterful improvising skills; he was only allowed to graduate from university on the proviso that he would never teach music (Salmon, 1992). This lends further credence to the notion that failure for a person with dyslexia may be more tied to the types of assessment they encounter in the educational system, rather than their knowledge or abilities. This correlates to previous discussion (Section 11.3) on the types of assessment in graded music performance and theory exams which dyslexic students might find challenging and highlights the need for greater flexibility in the content and types of exams offered by exam boards.

Brubeck described frustrating experiences with Arnold Schoenberg, who was inflexible in his approach, but found inspiration and permission to compose more freely from another teacher, Darius Mihaud (Salmon, 1992). This also has implications for teachers and serves as further motivation to have a strengths-focus in their teaching, collaborating flexibly with the student to find what is successful and building on that as a foundation. Another theme that emerged was that students recognised hiding their difficulties as one of their strengths, or as Ned reported 'distracting people from my weaknesses'. Further implications for teachers include the importance of empathic awareness, as discussed in Reflective statement 7, as the longterm ramifications of masking emotions and difficulties may be detrimental to the students' ability to self-regulate and their mental well-being.

Overall, in the interview data and from analysis of my reflective observations, aural strengths were the most widely reported. This included good relative or perfect pitch, an ability to mimic or copy aurally, recognising or producing vocal harmonies and strong improvisatory skills. Teachers viewed aural memory, pitch perception and an ability to improvise as the student compensating for difficulties engaging with a score of music, and there was an emphasis on balancing lesson content for more challenging activities, such as sight-reading. Some teachers felt strongly that dyslexic students should not avoid areas of difficulty. Implications for teachers include the need for sensitivity to the students' reactions during activities, the student's frustration tolerance and view of themselves during activities they may view as remedial. This also has implications for teachers to examine their views of the importance of

sight-reading, which may have been influenced by exam requirements or their own past experiences as students and emphasise the need to maintain a flexible and adaptable approach.

Strengths in areas connected to creativity, for example, problem-solving, arranging music improvising and composing were identified. Creativity was seen by teachers in dyslexic students' unique approaches to learning and connecting ideas in a novel way, described by teachers as three-dimensional thinking, and through students' use of vivid imaginations to create narratives as memory aids for pieces. Strengths of resilience and perseverance were developed through an intense determination to achieve goals and often were viewed by students and teachers as a defensive response in the face of repeated failure. Verbal communication strengths were utilised as a social skill and as a way of creating narratives to develop recall of concepts or music. Dyslexic students reported themselves to be good listeners and described themselves as empathic and good collaborators, whilst this was not reported by teachers. This may suggest that many teachers view their roles with students less collaboratively and more from a master-apprentice perspective. Performance skills were noted, which may be linked to empathy and the ability to communicate with an audience, whilst students specifically recognised a strength in their flair for dramatic interpretation of music. This corresponds to the strengths identified in dyslexic actors (Leveroy, 2013).

Long-term memory strengths enabled the recall of pieces, although it was recognised as a dichotomy considering challenges with short-term or working memory. Music had to be learned and memorised carefully, with guidance, as errors might be difficult to unlearn. The ability to recall music through the kinaesthetic feel of the keyboard or instrument was identified as a strength. These strengths were seen as promotive and protective factors when students faced frustration due to challenges with slow sight-reading, poor coordination or executive functioning when learning music. The next section discusses teachers' use of these strengths in a strategic way to enhance the students' motivation and engagement.

11.4.2 Teachers' utilisation of strengths in the lesson context

Teachers incorporated a number of strategies in order to develop their students' strengths, including using kinaesthetic and aural patterns as part of the students' learning process, using parameters in improvisational activities to give the student some structure and support, using

notation software as an initial pathway for composition, encouraging students to verbalise their actions and responses to stimulate recall, supporting guided practice sessions and encouraging self-reflection in the learning process. Implications for teachers include sensitivity to the student and their chosen approach to activities, particularly if they are different to the teacher's preferred approach, and utilising a scaffolded approach, with an awareness of when detailed verbal instructions are appropriate. Students who enjoy performing should be given opportunities to have quality performing experiences which might heighten their sense of accomplishment resulting in increased motivation and a positive self-concept. By developing an awareness of the students' strengths, establishing a positive learning environment and collaborating with the student to eliminate barriers to learning, teachers might develop a strength-focused approach.

11.5 Research question 4

RS4: What are dyslexic students' perceptions of their music learning experiences? How might their voices be nurtured and amplified in the pedagogical process?

Though there is literature which examines the perspectives of teachers (Boardman, 2020), assessors (Reid, 2016; Bajaj & Bhatia, 2020) and parents (Griffiths et al., 2004; Washburn, 2014) on the topic of dyslexia, there is less research focusing on the student perspective. One of the earliest studies explored school experiences, support received, attitudes towards dyslexia and coping strategies of nine dyslexic students (Riddick et al., 1997), whilst other research investigated student viewpoints of the traits and characteristics of teachers who helped them to learn (Riddick, 1996; Pavey et al., 2013). Blackman (2011) investigated the views of dyslexic secondary school students on effective teaching strategies, finding that they preferred 'more detailed explanations, question and answer sessions, and other kinds of enquiry-based approaches to learning like project work' as well as 'demonstrations, drama and role play, storytelling and suggesting important points' (p. 184).

Struggles with low self-esteem, anger, and depression, largely due to poor educational experiences, are documented in the dyslexia literature (Alexander-Passe, 2008; Eissa, 2010; Durrant, 2022; Bajaj & Bhatia, 2020; Livingston et al., 2018). Tanner (2009) describes how

failure may be ascribed to dyslexics based on faulty constructs about dyslexia (p. 786). Exam pressures leading to stress, frustration with mis-applied or refused accommodations, feeling misjudged by teachers and comparison with non-dyslexic peers were some of the barriers identified by students (Camilleri et al., 2019; Denhart, 2008).

11.5.1 Dyslexic student voice in the music education literature

Dyslexic student perspectives of music learning experiences are even rarer, consisting mainly of anecdotal information (Ganschow et al., 1994; O'Brien Vance, 2004; Miles et al., 2005; Oglethorpe, 2008). In case studies of adult dyslexic musicians, findings indicate diverse profiles of difficulties with reading music notation and processing rhythm (Ganschow, 1994; Nelson & Hourigan, 2016). Both studies advocate the use of multisensory teaching techniques and isolating smaller components of the music as learning strategies, whilst Ganschow et al. (1994) also proposed that due to coordination and music reading challenges careful thought should be given when choosing an instrument. Further suggestions reported from a study of dyslexic musicians include expanding repertoire from traditional classical genres, the use of technology and an emphasis on small group or individual instruction (Nelson & Hourigan, 2016). Though there has been research into adult dyslexic musicians' perspectives, there does not appear to be any research which has considered a more diverse range of ages and context with regard to music education or, more specifically for the purposes of this research, instrumental teaching.

11.5.2 The importance of and rationale for investigating student voice

Considerations from the learner perspective may enhance the educational provider's understanding of their own assumptions and may promote trust within the learner/teacher relationship (Nightingale, 2006). Riddick (2009) argues that the motivation for listening to the dyslexic student voice should come from a desire to improve their learning experiences and thus enhance their psychological well-being. May (2005) cautions that there may be several conflicting motives for educational systems' interest in student voices, noting that schools may be concerned with league table performance but might also consider other reasons such as empowering students or improving the environment to promote their physical and psychological well-being. Teachers, researchers and parents may have differing incentives for asking students about their perspectives. Teachers may seek the validation of their own pedagogical approaches and there is the potential for manipulation or over-monitoring from parents, teachers and researchers.

Findings from this research confirm the importance of family support, particularly that of mothers, in helping and advocating for them through their educational journey and their assessment with dyslexia, as well as in the role of negotiating with schools or teachers. Early identification and an openness and positivity in the family's attitude toward dyslexia was seen as significant for participants, who specifically felt that dyslexic parents were able to empathise with their challenges. Findings indicate that the students' relationship with the teacher, use of technology and strengths were strategically used to mediate their challenges with learning in the music lesson context. Great value was placed on teachers who were flexible and patient. Students also recognised the need for more accessible resources related to music and dyslexia.

Leitão et al. (2017) argue that different ecological levels (Bronfenbrenner, 1979) including government, educational, teachers, family and individual systems, need to be examined in order to gain a holistic picture of the issues affecting dyslexics and their families. In a study of parents and their dyslexic children, children and their parents reported benefitting from a support community of friends and teachers which they felt helped reduce parental anxieties, and a sense of isolation and stress-related behaviours, both internal and external, in dyslexic children (Wilmot et al., 2022, p. 50). The next section discusses the importance of the interactions of these roles in the pedagogical process.

11.6 Research question 5

RS5: What are the teacher, student and parental perceptions of the pedagogical process and roles in the music lesson context and how can these roles collaborate to collectively enable positive outcomes?

Parental support for dyslexic students is explored in the context of risk-resilience literature with positive family support seen as a protective measure (Catt & Petscher, 2022; Jenson & Fraser, 2015; Reid, 2016; Riddick, 2009; Washburn, 2014). The roles that parents of dyslexic children assume are explored as a means of understanding parental agency. The impact of dyslexia on families has been presented from a whole-systems perspective which considers not only the relationships between the student, their family and educational environments but also policies which may affect them. Parental roles include maintaining their child's self-

esteem in conjunction with insensitivity from peers or teachers, reminding them of their strengths and encouraging them to persevere (Alexander-Passe, 2008; Reid, 2004). Washburn (2014) reported how mothers of dyslexic children journey through changing identity roles from 'investigator' to 'advocate' and finally, as 'tutor' (p. 123).

Findings from this research indicate that dyslexic parents may experience distress when they observe their child experiencing similar difficulties; however, they may also view their role as advantageous. Parents, dyslexic and non-dyslexic, with music training may be able to provide guided support during practice, depending on their relationship with the student. Parents also offer support through helping the student to organise material and practice times as well as celebrating their achievements. Communication with teachers is an important aspect of parental support; the instrumental teacher can benefit from the sharing of the assessment report, parental observations about the student's profile, knowledge of the student's emotional well-being, and parents' challenges in communicating with schools and classroom teachers.

Griffiths et al. (2004) utilises a conceptual framework to identify how mothers of SEN children form identities and obtain knowledge so that teachers might understand how best to work in partnership with them. There are a number of implications to be drawn from this framework. If parents are not satisfied with the response after initiating discussion with school or teachers, they might seek knowledge about dyslexia from sources which empower them to negotiate effectively with teachers and schools. However, Griffiths et al. (2004) found that teachers and parents may have different expectations, and that cultural aspects (for example, working class parents versus educated professionals) might give some parents a sense of disadvantage. Parents might struggle to obtain quality information and may feel obliged to pay for specialist advice if they can afford it. Dyslexic parents may struggle to read or comprehend reports or messages from the school, particularly if the terminology is difficult to understand. In navigating SEN reports, parents may be frustrated if teachers or the SEN process fails to find adequate solutions for their child. There may be issues of mistrust, with the parent believing that the teacher is not concerned about the child as an individual. In some cases, the child may ultimately be withdrawn from formal school settings.

Implications for instrumental teachers include an understanding of the potential imbalances between teacher and parent in terms of power and knowledge and learning how to transfer power to parents by taking their concerns seriously, communicating to the parent that the child is seen as an individual and signposting the parents to good quality resources to build accurate knowledge about dyslexia and how to support the student's learning effectively.

Research into family (Snowling et al., 2003; Torppa et al., 2010) and twin studies (Olson et al., 2014) has confirmed a heritable risk factor for dyslexia and identified other environmental factors which may negatively impact on dyslexic students' outcomes: misperceptions as to the extent of students' difficulties, inadequate teaching, absence of support in the school setting, poor home surroundings and breakdowns in family relationships (Muter & Snowling, 2009). Durrant (2022) refers to the importance of parents in a protective role, helping the student to mitigate any shame or low self-esteem, valuing their differences as well as empowering them to be involved in decisions which affect them.

Research in the field of child psychology has identified frameworks for considering why some children with higher risk factors have better outcomes in the face of adverse circumstances (Catts & Petscher, 2022; Daniel & Wassell, 2002). Resilience is a dynamic process which may be influenced by various factors over time; key protective factors are stability in the home as well as supportive parents and teachers (Jenson & Fraser, 2015). In addition to a secure base and sense of belonging, good self-esteem and a feeling of self-efficacy are fundamental to developing resilience (Gilligan, 1997). Daniel & Wassell (2002, p. 11) created a 'framework for the assessment of resilience factors' (see Figure 11.1) to show how protective factors might be used to overcome areas of adversity and vulnerability to promote resilience in vulnerable young children. A better understanding of resilience factors supports 'an alternative framework for intervention, the focus being on the assessment of potential areas of strength within the child's whole system' (Daniel & Wassell, 2002, p. 13).



Figure 11.1 Framework for the assessment of resilience factors (Daniel & Wassell, 2002, p. 11)

Used as an assessment tool, the framework is anchored around four dimensions: resilience, vulnerability, adversity and protective factors. By observing and analysing the interactive factors, a comprehensive picture of the individual develops and acts as a guide to indicate the measures needed to support them. The essential idea of the framework is to view the individual being assessed as moving from top to bottom and from left to right. The assessment should show which external areas of adversity or challenge may be obstacles to resilience and may need to be changed or removed and identify ways of bringing support to areas of intrinsic vulnerability.

Catts and Petscher (2022) apply this risk-resilience concept to literacy instruction for dyslexic students (See Figure 11.2)⁴⁵ and identify various 'protective and promotive factors' including appropriate instruction, higher verbal reasoning skills, working memory abilities, a student's belief in their capabilities or ability to learn, strong emotional coping skills, support and encouragement of a teacher, strong parent-teacher connections and nurturing family support (p. 173). They specify these as either 'promotive' factors which are beneficial for everyone despite risk levels, or 'protective' which mitigate the effects of the risk factors; some factors are described as both promotive and protective (Catts & Petscher, 2022, p. 173). An example of a factor which is both promotive and protective is parenting; all children benefit from supportive parents but for those who are at greatest risk, nurturing parents provide a protective measure enabling the child to adapt more successfully to challenges (Masten, 2018).

⁴⁵ Figure is used with permission.

Catts and Petscher (2022) postulate that by using this model teachers and parents might identify and facilitate dyslexic students' resilience factors as a means of mitigating or balancing the risks and putting into practice interventions which might be tailored individually for each student. Other protective factors relating to dyslexia in the family environment have been considered in relation to learning difficulties (Riddick, 2009; Yu et al., 2018; Catts & Petscher, 2022). Yu et al. (2018) state that brain development specific to the creation of compensatory mechanisms in dyslexics is influenced by environmental factors, noting that 'socioeconomic factors', the 'home literacy environment' and 'instructional approaches' may influence reading outcomes for dyslexic children (p. 244); socioeconomic aspects link to the availability of good nutrition and educational resources, both of which might affect a child's cognitive development.





Snowling et al. (2007) investigated the impact of learning difficulties on families of dyslexic adolescents and found that 74.3% of parents with a dyslexic child believed that there was a significant impact for the whole family and that mothers of children with learning difficulties demonstrated higher rates of anxiety and depression. In another study, Alias and Dahlan (2015) found that mothers of dyslexic children reported denial, guilt, anxiety, concern about their child's socialisation, academic skills and future plans; they might also have concerns about teachers who lack adequate knowledge or expertise corresponding to their child's

needs. Themes such as 'time constraints, lack of knowledge, increased financial demand, school issues and negative feelings towards the child's condition' in addition to the expectations placed on them in their roles as mothers emerged, along with the conclusion that mothers would benefit from support on a variety of levels (Alias & Dahlan, 2015, p. 109).

An ecological perspective (Bronfenbrenner, 1986) of the risk-resilience literature indicates that there are multiple systems and domains which may influence outcomes: the child's relationship with themselves, their family, school environment and governmental policies must be taken into consideration. Riddick (2009) notes that educators and other professionals have an opportunity to learn from mothers of dyslexic children; these were identified earlier in this thesis and also in Section 11.5.2 as being of particular value. The Rose Report (2009) recognised distinctly lacking provision for parents of children with dyslexia and proposed necessary changes to the UK educational system suggesting that parents should be more informed about dyslexia through the provision of published material and a telephone helpline. A more recent study of teachers, parents and school counsellors with dyslexic learners confirms the need for whole-systems support involving all stakeholders, as parents suffer as a result of a lack of knowledge, counsellors were impacted by parents' reluctance to have their child assessed, and both parents and counsellors felt that teachers were unsupportive; meanwhile, teachers reported a lack of training for working appropriately with dyslexic students and a sense of feeling overwhelmed with related responsibilities (Bajaj & Bhatia, 2020). Leitão et al. (2017) found that parents' and students' perceptions indicate the need for societal change in order to reduce the stigma of dyslexia. This finding is confirmed by Bonifacci et al. (2014) who state that 'in setting up educational intervention programmes it is important to consider parental needs and resources, thus moving, within a systemic perspective, from an individual-centred to a family-centred care viewpoint' (p. 188). The findings of the current study indicate that music education would likewise benefit from societal change and more visible, accessible systems for diagnosis and support.

11.7 A new risk-resilience model of dyslexia for musical training

The aim of this research is to improve and enhance instrumental teaching for dyslexic students. By considering multiple systems from an ecological perspective (Bronfenbrenner (1979), drawing on the risk-resilience literature and assessment framework (Daniel & Wassell,

2002) and incorporating the findings of this research, a new risk-resilience model of dyslexia for musical training (Figure 11.3) has been created.

The purpose of this model is to highlight the areas in which parents, teachers, exam boards and institutions might collaborate and support not only each other but also to facilitate the students' development of resilience, despite areas of risk. As with Catts and Petscher's (2022) model, this model is based on the image of a fulcrum with the idea that resilience factors may outweigh the risk factors to support the student in this process.



Figure 11.3 A new risk-resilience model of dyslexia for musical training

This model begins with the individual student and the potential risk factors they may experience in learning music (on the left-hand side of the model). The student may also have other vulnerability or adversity factors but for the sake of clarity in the model, only a few common risk areas have been identified. The right-hand side of the fulcrum shows the interactive nature of support between parents, students and the teacher with several resilience factors identified for each role. Finally, resilience factors from external support are
also identified to the right of the fulcrum. At the exosystem and macrosystem levels, external support might be provided in the form of training for teachers and support for students and families through music hubs, organisations and music exam boards.

One of the key resilience factors derived from this model is the need for teachers who have been equipped with high-quality pedagogical training. This would require an understanding of the knowledge and skills needed for teachers to improve their practice and enhance their teaching, not only for dyslexic students but for a diverse range of students. This resilience factor is amplified and expanded in a separate 'Knowledge and Practice Standards' document (Section 11.7.1) which collates the findings from the literature and this research study.

11.7.1 Knowledge and practice standards for teachers of music: Rationale

Dyslexia is a highly complex specific learning difficulty which affects from between 3-7% (Rutter et al., 2004, Snowling & Melby-Lervåg, 2016) and 17-21% (Ferrer et al., 2015) of the English-speaking population, depending on the criteria which are used. The breadth of variability in the characteristics and degrees to which dyslexia presents mean that a high proportion of dyslexics go undiagnosed (Shaywitz, 2005, Snowling et al., 2020). As a result of negative educational experiences and stigma, secondary traits of dyslexics include low self-esteem (Livingston, Siegel & Ribary, 2018; Ridsdale, 2005) and vulnerability to a state of 'learned helplessness' (Peterson, Maier & Seligman, 1993, p. 8). It is generally agreed that dyslexia is not caused by one specific variable, but is influenced by a number of factors, in a 'multi-deficit model (MDM)', or what might be called a 'multiple factors model' in a shift away from a deficit-focus (McGrath et al., 2020, p. 10).

A refined understanding of the multiple deficits associated with other neurodevelopmental disorders has the potential to support individually tailored interventions that may ultimately be more successful than "one-size fits all" approaches. (McGrath et al., 2020, p. 9)

Dyslexic students may also struggle with auditory and visual processing, sequencing, working memory, spatial awareness, motor coordination, organisation and symptoms of visual stress (Rose, 2009; BDA, 2024: Oglethorpe, 2002). With many of the skills needed to learn to read music corresponding to the difficulties associated with dyslexia, students and instrumental teachers may find the experience challenging. At the same time, instrumental music teachers

have the opportunity to create a more positive learning experience for dyslexic students. The findings of the research study of this thesis suggest that teachers perceive there is a lack of research, training and resources to support their teaching of dyslexic students (Chapter 6). This might lead to assumptions and misapplication of strategies which could mean detrimental outcomes for the student. It is crucial that instrumental music teachers have access to high-quality, accessible music training.

Music is inherently a multisensory activity, combining auditory, visual and kinaesthetic sensory information in a coordinated effort. With the complexities of dyslexia, any number of factors might disrupt the learning process from processing, attention or coordination, just to name a few. These difficulties might appear by themselves or in a cluster. This highlights the need for specialised instruction, which is tailored to the individual student, ideally supported by a framework such as Universal Design for Learning (CAST, 2018) as well as the new risk-resilience model created for this thesis (Figure 11.3) and by the Knowledge and Practice Standards document (Table 11.2). Along with an understanding of lesson preparation, teachers might benefit from an understanding of dyslexia and co-occurring conditions, current research and implications, effects on music learning, how to develop a student profile, strategies designed to mediate the challenges, approaches designed to support the student's metacognition, a strengths-focus to the planning and implementation of these strategies and most importantly, an emphasis on building rapport with the student. In supporting the interactions between students, teachers and their parents, a set of Knowledge and Practice Standards for Dyslexia: Music Education, adapted from IDA (2018) has been created and is presented in the next section (see Table 11.2).

11.7.2 Knowledge and practice standards: Adapted from IDA (2018)

These knowledge and practice standards (Table 11.1) have been adapted from the 'Knowledge and Practice Standards for Teachers of Reading' document which can be accessed at the 'International Dyslexia Association. (2018). Knowledge and Practice Standards for Teachers of Reading'. The complete document (Pages 1-41) is included for context in Appendix P of this thesis. The summary table (pages 9-11 of the IDA KPS document) is included here for reference and to show how the Knowledge and Practice Standards for Teachers of Music was adapted from it for this thesis⁴⁶. The summary table⁴⁷ consists of five main standards or content domains, and underneath these are numbered knowledge standards with a brief description of how they might be 'observed, tested or applied' (IDA, 2018, p. 8). The content domains include:

- Standard 1: Foundations of Literacy acquisition
- Standard 2: Knowledge of Diverse Reading Profiles, Including Dyslexia
- Standard 3: Assessment
- Standard 4: Structured Literacy Instruction
- Standard 5: Professional Dispositions and Practices

⁴⁷ International Dyslexia Association Knowledge and Practice Standards for Teachers of Reading. Retrieved on April 30, 2024 from <u>https://dyslexiaida.org/knowledge-and-practices/</u>

Knowledge and Practice Standards for Teachers of Reading		
Summary Table		
	Does Not Include Knowledge and Practice Examples	
	Standard I: Foundations of Literacy Acquisition	
1.1	Understand the (5) language processing requirements of proficient reading and writing: phonological, orthographic, semantic, syntactic, discourse.	
1.2	Understand that learning to read, for most people, requires explicit instruction.	
1.3	Understand the reciprocal relationships among phonemic awareness, decoding, word recognition, spelling, and vocabulary knowledge.	
1.4	Identify and explain aspects of cognition and behavior that affect reading and writing development.	
1.5	Identify (and explain how) environmental, cultural, and social factors contribute to literacy development.	
1.6	Explain major research findings regarding the contribution of linguistic and cognitive factors to the prediction of literacy outcomes.	
1.7	Understand the most common intrinsic differences between good and poor readers (i.e., linguistic, cognitive, and neurobiological).	
1.8	Know phases in the typical developmental progression of oral language, phoneme awareness, decoding skills, printed word recognition, spelling, reading fluency, reading comprehension, and written expression.	
1.9	Understand the changing relationships among the major components of literacy development in accounting for reading achievement.	
	Standard 2: Knowledge of Diverse Reading Profiles, Including Dyslexia	
2.1	Recognize the tenets of the (2003) IDA definition of dyslexia, or any accepted revisions thereof.	
2.2	Know fundamental provisions of federal and state laws that pertain to learning disabilities, including dyslexia and other reading and language disability subtypes.	
2.3	Identify the distinguishing characteristics of dyslexia.	
2.4	Understand how reading disabilities vary in presentation and degree.	
2.5	Understand how and why symptoms of reading difficulty are likely to change over time in response to development and instruction.	
	Standard 3: Assessment	
3.1	Understand the differences among and purposes for screening, progress-monitoring, diagnostic, and outcome assessments.	
3.2	Understand basic principles of test construction and formats (e.g., reliability, validity, criterion, normed).	
3.3	Interpret basic statistics commonly utilized in formal and informal assessment.	
3.4	Know and utilize in practice well-validated screening tests designed to identify students at risk for reading difficulties.	
3.5	Understand/apply the principles of progress-monitoring and reporting with Curriculum-Based Measures (CBMs), including graphing techniques.	
3.6	Know and utilize in practice informal diagnostic surveys of phonological and phoneme awareness, decoding skills, oral reading fluency, comprehension, spelling, and writing.	
3.7	Know how to read and interpret the most common diagnostic tests used by psychologists, speech-language professionals, and educational evaluators.	
3.8	Integrate, summarize, and communicate (orally and in writing) the meaning of educational assessment data for sharing with students, parents, and other teachers.	

	10	
	Standard 4: Structured Literacy Instruction	
Substandard A: Essential Principles and Practices of Structured Literacy Instruction		
4A.1	Understand/apply in practice the general principles and practices of structured language and literacy teaching,	
	including explicit, systematic, cumulative, teacher-directed instruction.	
4A.2	Understand/apply in practice the rationale for multisensory and multimodal language-learning techniques.	
4A.3	Understand rationale for/Adapt instruction to accommodate individual differences in cognitive, linguistic,	
	sociocultural, and behavioral aspects of learning.	
	Substandard B: Phonological and Phonemic Awareness	
4B.1	Understand rationale for/identify, pronounce, classify, and compare all the consonant phonemes and all the vowel phonemes of English.	
4B.2	Understand/apply in practice considerations for levels of phonological sensitivity.	
4B.3	Understand/apply in practice considerations for phonemic-awareness difficulties.	
4B.4	Know/apply in practice consideration for the progression of phonemic-awareness skill development, across age and	
	grade.	
4B.5	Know/apply in practice considerations for the general and specific goals of phonemic-awareness instruction.	
4B.6	Know/apply in practice considerations for the principles of phonemic-awareness instruction: brief, multisensory,	
	conceptual, articulatory, auditory-verbal.	
4B.7	Know/apply in practice considerations for the utility of print and online resources for obtaining information about	
	Substandard C: Phonics and Word Recognition	
4C.1	Know/apply in practice considerations for the structure of English orthography and the patterns and rules that inform	
	the teaching of single- and multisyllabic regular word reading.	
4C.2	Know/apply in practice considerations for systematically, cumulatively, and explicitly teaching basic decoding and spelling skills.	
4C.3	Know/apply in practice considerations for organizing word recognition and spelling lessons by following a structured	
40.4	prioritis residuation practice considerations for using multisensory routines to enhance student engagement and memory	
40.5	Know/apply in practice considerations for adapting instruction for students with weaknesses in working memory.	
	attention, executive function, or processing speed.	
4C.6	Know/apply in practice considerations for teaching irregular words in small increments using special techniques.	
4C.7	Know/apply in practice considerations for systematically teaching the decoding of multisyllabic words.	
4C.8	Know/apply in practice considerations for the different types and purposes of texts, with emphasis on the role of	
	decodable texts in teaching beginning readers.	
	Substandard D: Automatic, Fluent Reading of Text	
4D.1	Know/apply in practice considerations for the role of fluent word-level skills in automatic word reading, oral reading	
	fluency, reading comprehension, and motivation to read.	
4D.2	Know/apply in practice considerations for varied techniques and methods for building reading fluency.	
4D.3	Know/apply in practice considerations for text reading fluency as an achievement of normal reading development that can be advanced through informed instruction and progress-monitoring practices.	
4D.4	Know/apply in practice considerations for appropriate uses of assistive technology for students with serious	
	limitations in reading fluency.	
45.4	Substandard E: Vocabulary	
4E.1	written language comprehension.	
4E.2	Know/apply in practice considerations for the sources of wide differences in students' vocabularies.	
4E.3	Know/apply in practice considerations for the role and characteristics of indirect (contextual) methods of vocabulary instruction.	
4E.4	Know/apply in practice considerations for the role and characteristics of direct, explicit methods of vocabulary instruction.	
Substandard F: Listening and Reading Comprehension		
4F.1	Know/apply in practice considerations for factors that contribute to deep comprehension.	
4F.2	Know/apply in practice considerations for instructional routines appropriate for each major genre: informational text narrative text and argumentation	
4F.3	Know/apply in practice considerations for the role of sentence comprehension in listening and reading	
	comprehension.	

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

4F.4	Know/apply in practice considerations for the use of explicit comprehension strategy instruction, as supported by
	research.
4F.5	Know/apply in practice considerations for the teacher's role as an active mediator of text-comprehension processes.
	Substandard G: Written Expression
4G.1	Understand the major skill domains that contribute to written expression.
4G.2	Know/apply in practice considerations for research-based principles for teaching letter formation, both manuscript and cursive.
4G.3	Know/apply in practice considerations for research-based principles for teaching written spelling and punctuation.
4G.4	Know/apply in practice considerations for the developmental phases of the writing process.
4G.5	Know/apply in practice considerations for the appropriate uses of assistive technology in written expression.
	Standard 5: Professional Dispositions and Practices
5.1	Strive to do no harm and to act in the best interests of struggling readers and readers with dyslexia and other reading disorders.
5.2	Maintain the public trust by providing accurate information about currently accepted and scientifically supported best practices in the field.
5.3	Avoid misrepresentation of the efficacy of educational or other treatments or the proof for or against those treatments.
5.4	Respect objectivity by reporting assessment and treatment results accurately, and truthfully.
5.5	Avoid making unfounded claims of any kind regarding the training, experience, credentials, affiliations, and degrees of those providing services.
5.6	Respect the training requirements of established credentialing and accreditation organizations supported by CERI and IDA.
5.7	Avoid conflicts of interest when possible and acknowledge conflicts of interest when they occur.
5.8	Support just treatment of individuals with dyslexia and related learning difficulties.
5.9	Respect confidentiality of students or clients.
5.10	Respect the intellectual property of others.

11

Table 11.1 International Dyslexia Association Knowledge and Practice Standards for Teachers of Reading (IDA, 2018)

This framework has been adapted for music (Table 11.2) as, to the researcher's knowledge, no set of standards like this exists for teaching music to dyslexic students. The results of this study strongly indicate the need for teacher training in the knowledge of dyslexia, but also in the application of this knowledge to music pedagogy including the use of specific strategies. The standards are a means of equipping teachers in order to teach dyslexic students more effectively. Another result of this study has been the conceptualisation of music and dyslexia as viewed through a whole systems approach and this new framework, along with the risk-resilience model for musical training, supports the teacher's understanding of parental roles and the emotional well-being of the teacher, student and parent.

There are six levels or standards to this document (Table 11.2). Standard 1 addresses Foundational Dyslexia Knowledge, Standard 2 covers Accessible and Inclusive Teaching, whilst Standard 3 offers Music Pedagogy for Dyslexic Learners. Standard 4 addresses Specific Strategies and Practical Resources. Standard 5 covers Music Exams for Dyslexic Students and Standard 6 addresses the Ethical Issues related to teaching dyslexic students. These standards may be used as the underpinning for a teacher training course, and are designed so that each level could be extrapolated to specific learning objectives related to the standard. The standards are flexible and adaptable, and would need constant updating in order to keep pace with the research and technology changes. They could be developed further to support teachers of instrumental music teaching in a variety of contexts.

Knowledge and Practice Standards for Teachers of Music		
	Summary Table	
Standard 1 – Foundational Dyslexia Knowledge		
1.1	Explain the different definitions of dyslexia	
1.2	Describe and summarise dyslexia research in relation to theories of causality	
1.3	Identify the primary and secondary characteristics of dyslexia	
1.4	Recognise and describe co-occurring conditions	
1.5	Summarise relevant connections between music and language	
1.6	Explain the principles and purpose of assessments and reports	
1.7	Understand the Equality Act 2010 and implications for dyslexic people	
1.8	Describe the implications of the social model of disability (Barnes, 2019)	
1.9	Explain the assessment process and the components of an assessment report	
1.10	Understand principles of instruction which help dyslexic students to develop and	
	improve over time	
1.11	Summarise the importance of the learning environment for a dyslexic student	
	Standard 2 – Accessible and Inclusive Teaching	
2.1	Explain the concepts underpinning Universal Design for Learning (CAST, 2018)	
2.2	Explain the usefulness of a risk/resilience framework	
2.3	Understand the impact of dyslexia on families, specifically mothers, and ways that	
	teachers might support them	
2.4	Recognise the role of a supportive family environment, knowledge of and attitude	
	towards dyslexia as promotive factors	
2.5	Describe the benefits of accessible teaching for a dyslexic student	
2.6	Explain the relevance of having flexible student profiles	
Standard 3 – Music Pedagogy for Dyslexic Learners		
Substandard 3A: Variability in how dyslexia may affect music learning		

Dyslexia m	Dyslexia may affect music learning in various ways, and the impact can differ significantly from		
person to person. The following substandards identify some of the variable ways in which a person			
with dyslexia may be affected.			
3A.1	Provide an overview of the ways in which dyslexia might affect music learning		
3A.2	Summarise how rhythm and timing in music learning might be impacted		
3A.3	Explain the ways in which pitch and sound processing might be affected		
3A.4	Describe the challenges with reading music which may affect dyslexic students		
3A.5	Explain how executive function challenges might impact the learning experience:		
	working memory, cognitive flexibility, inhibitory control, sequencing, planning and		
	organisation		
3A.6	Describe how other memory systems may be affected		
3A.7	Recognise factors affecting coordination, spatial awareness and balance which may be		
	impacted		
3A.8	Explain how secondary characteristics of dyslexia may affect students' behaviour and		
	attitude toward learning		
3A.9	Describe symptoms indicative of visual stress and how to signpost student and parents		
	to appropriate support		
3.2	Understand how literacy instruction principles might be relevant to music instruction		
3.3	Understand the potential limits to literacy instruction approaches in the music learning		
	context		
3.4	Describe the benefits and use of UDL to plan music lessons		
3.5	Summarise the usefulness of music ed. methods (Suzuki, Dalcroze, Kodály)		
3.6	Summarise the benefits of a responsive intervention to challenges approach		
3.7	Explain the advantages of positive collaboration with students and parents		
3.8	Understand the importance of instruction as a protective factor		
3.10	Explain the importance and benefits of a strengths-focused teaching approach		
	Standard 4 – Specific Strategies and Practical Resources		
4.1	Understand general strategies related to music instruction		
4.2	Identify common strengths in dyslexic students		
4.3	Explain how instruction might strategically develop these strengths		
4.4	Understand how to support students with poor organisational skills		
4.5	Describe how to motivate students by offering them autonomy in repertoire and		
	activities		
4.6	Summarise the types of repertoire that might be useful for dyslexic students		

4.7	Describe how to create a bank of method books and repertoire that might appeal to
	students
4.8	Describe alternative notation strategies and their usefulness for specific students
4.9	Explain how to introduce and utilise alternative notation in lessons
4.10	Demonstrate the use of technology to stimulate learners' interest
4.11	Demonstrate the use of technology to build automaticity and reinforce learning
4.12	Demonstrate the use of technology to modify or adapt scores
4.13	Describe how to create a learning bank of multisensory activities
4.14	Summarise approaches to challenges with sight-reading
4.15	Identify challenges with rhythm and corresponding strategies
4.16	Identify challenges to pitch reading and corresponding strategies
4.17	Identify challenges with coordination and corresponding strategies
4.18	Understand ways of encouraging students' metacognition
	Standard 5 – Music Exams for dyslexic Students
5.1	Understand the motivations that students have for taking graded music exams
5.2	Describe the requirements of different exam boards
5.3	Explain various access arrangements for graded music exams
5.4	Describe the procedure of entering students who need reasonable adjustments
5.5	Understand how to support the student in preparing to utilise reasonable adjustments in
	the exam setting
5.6	Consider alternatives to graded music exams
5.7	Describe how to manage parental expectations in the area of exams
Standard 6 – Ethical Issues	
6.1	Seek to act in the best interests of the student and their family
6.2	Develop trust in your collaboration with the student and their family
6.3	Maintain confidentiality of records and assessments
6.4	Respect and maintain ethical considerations of the safe storage of data (GDPR
	requirements)
6.5	Seek communication and action that represents authenticity and openness to change
6.6	Describe the importance of empathic awareness
6.7	Explain the ethical issues involved in teacher advocacy
6.8	Discuss ways to manage lifestyle and working practices to prevent burnout

Table 11.2 Knowledge and Practice Standards for Teachers of Music: Dyslexia specific

These standards set forth competencies for becoming a more highly qualified teacher of dyslexic students. As has been discussed, when a teacher feels equipped and supported, this impacts their relationship with the student and their family. A teacher with relevant information about music exams will be able to support the student more effectively in the application and preparation processes involved. The connections between these standards and the risk-resilience model explain areas where communication between teachers, parents and music exam boards might be made more accessible. As parents and teachers are aware of risk factors and ways in which they might help to mediate some of those, the student benefits. These knowledge and practice standards, along with the risk-resilience model, are unique contributions that my research makes to music education through the findings of this research.

11.8 Feedback on thesis findings from an evaluative focus group

As a means of evaluating the thesis findings and generating further perspectives on dissemination, an evaluative focus group was held in 2024. Seven participants with expertise in inclusive music education were invited to participate and were given an information and consent sheet in advance of the meeting. Four individuals agreed to participate, but one was not able to attend the meeting, due to prior commitments. The following participants of the focus group are not anonymised and have given consent for this:

- Andrew Hird, Equality, Diversity and Inclusion Lead, Partnership Manager (Boston and South Holland) Lincolnshire Music Service
- Liz Horobin, chair of the BDA (British Dyslexia Association) music committee and a member of the music subject area group of UKAAF (the UK Association for Accessible Formats), professional practice lead at PATOSS (Professional Association of Teachers of Students with Specific Learning Difficulties)
- Ben Sellers, Inclusion specialist and lead of the MEHEM (Music Education Hubs East Midlands) Uprising project

In advance of the meeting, held on Zoom, participants were sent a brief (See Appendix N) describing the aims and methods of the research project, as well as some questions to consider when evaluating the findings. At the meeting, an overview of the findings of this research, as detailed in this chapter and in Reflective Statements 1-7, was presented by the

researcher. A transcript was generated from the discussion, which was sent to participants to review, and the following themes emerged from the analysis of the transcript. These themes are discussed in greater detail in the following sections: the utility of the use of the term 'dyslexia' as opposed to a more general categorisation of 'neurodiversity' (see 11.8.1), the importance of signposting teachers to accurate information about visual stress, implications of high rates of amusia in children and adults with dyslexia, managing teachers' and parents' expectations of outcomes (see 11.8.2), the balance of verbal and nonverbal communication in lessons, implications for students from more diverse ethnic and socioeconomic backgrounds (see 11.8.3) and the suggestion that dissemination might start by providing a training course for music hub inclusion leads (11.8.4). Feedback clarifying various aspects of an earlier version of the model was offered by participants, and this resulted in the model presented above in Figure 11.2.

11.8.1 The utility of the term 'dyslexia'

One participant in the focus group suggested that the term 'neurodiversity' should replace the use of the term 'dyslexia' in my research. Regarding the use of the term 'dyslexia', recent research (Snowling, 2020; Stein, 2023; Wilmot et al., 2023) indicates that although the term is used relative to differing criteria and there is still some ambiguity as to its cause, it still deserves to be studied and understood separately from the more general discussion of neurodivergence. Morrow (2023) uses the more general term of 'learning disabilities', representing the view in which it is seen as a disorder. Upon undertaking this research, I realised that I would need to clarify my use of the term. I did this in presentations and workshops by positioning dyslexia with a number of other learning differences, such as dyscalculia, dysgraphia, dysmusia, dyspraxia/developmental coordination disorder under the wider umbrella of neurodivergence. My rationale for doing this was to acknowledge the high incidence of co-occurring conditions within the dyslexic population but also to clarify that I was not referring to other aspects of neurodivergence, for example, autism or Tourette syndrome, in my study. This further confirms the complexity of dyslexia and the need for more research to clarify the term but also raises the question of the utility of the label of dyslexia for educational purposes. This issue has been debated for a number of years (Elliott & Grigorenko, 2014; Gibbs & Elliott, 2020).

11.8.2 Managing expectations through a variety of outcomes

A participant in the focus group questioned whether the teachers interviewed considered alternative outcomes to attainment standards set, for example, by graded exams. The importance of outcomes based on progress, although those might be different to other students in a classroom or a teaching practice, and the impact of positive outcomes on selfesteem or non-musical goals has been emphasised in these research findings. This links to a need for teachers' flexibility and adaptability as a means of incorporating an inclusive teaching approach in their practice. Reflective Statement 4 discusses my development in reducing the number of verbal instructions and my proximity to the student, which had the impact of improving students' metacognitive skills and relates to the importance of reducing distraction in the learning environment.

11.8.3 Implications for students from diverse backgrounds

One of the limitations of this research (discussed in Chapter 12) is that my students were similar in age, sex and socioeconomic background. An attempt to mitigate this was made by interviewing students from a variety of backgrounds and ages, but as noted by my focus group participants, it might be useful to explore these findings in more diverse contexts and with participants of different ages and genders from a wider variety of cultural and socioeconomic backgrounds, thus extending research to a wider group of teachers and their students and families.

11.8.4 Potential routes for dissemination

Music hubs are currently (as of April 2024) undergoing restructuring, with a greatly reduced number of hubs (from 116 to 43) taking the role of lead hubs from September 2024 (Daubney, 2023). As suggested in the focus group, one means of disseminating this research might be through training targeted at music hub inclusion leads. A participant stated 'For me, the outcome of your research is kind of a thorough, comprehensive training of the nation's instrumental teachers of dyslexia'. By providing the inclusion leads with training, they would be able to support the music educators in their geographical area.

The challenges to this proposal might be enlisting broader support from music hubs with the logistics and costs involved. At this point, the research has been shared in two workshop settings, at the Northwest Music Hubs Music Mark conference (September 2023) and Music

Education Hubs East Midlands (MEHEM Uprising) in January 2024. Additionally, the research has been presented as part of a session on inclusive music teaching for undergraduate students at the University of York (2023-2024), as well as a 'Music and dyslexia' session to cohorts on the MA Music Education: Instrumental and Vocal Teaching pathway (2019-2024). Some participant feedback, as might be collected anonymously to evaluate the delivery and content of these sessions, has been collected (See Appendix O). The feedback indicates that participants found the sessions useful and it correlated with their desire to improve knowledge and teaching techniques, as well as to develop a better understanding of the needs of dyslexic students. By including the risk-resilience model for musical training (Figure 11.3) and utilising the standards framework (Table 11.2) within training, as well as further findings from this research, teachers could be supported in working with dyslexic students and their families, further informed as to relevant aspects within external structures of assessment as provided by music examination boards and aided in working with schools to support learners.

Chapter 12 SUMMARY OF THE STUDY, LIMITATIONS, IMPLICATIONS AND RECOMMENDATIONS

12.1 Summary of the study

By building on the previous chapters in this thesis, Chapter 12 revisits and summarises the findings in response to the main research questions. Furthermore, it discusses the implications which result from this research and articulates possible limitations. The concluding sections present recommendations for future research in music education and dyslexia. This research aimed to improve and enhance instrumental music teaching for dyslexic students. The underlying research questions were:

Research question 1: What are the perceptions of music teachers regarding their experiences and utilisation of pedagogical practices, including strategies, methods and material, in teaching dyslexic students?

Research question 2: Do adjustments and accommodations which music exam boards allow offer dyslexic students unbiased inclusion? What are the perspectives of teachers and students on this topic?

Research question 3: Do teachers, parents and students perceive dyslexic students to have specific strengths? How might these strengths be utilised in the music lesson context?

Research question 4: What are dyslexic students' perceptions of their music learning experiences? How might their voices be nurtured and amplified in the pedagogical process?

Research question 5: What are the teacher, student and parental perceptions of the pedagogical process and roles in the music lesson context? How can these roles collaborate to collectively enable positive outcomes?

Though there is an increase in literature relating to special educational needs (SEND) in music education (Welch et al., 2009; Hammel et al., 2016, Culp & Salvador, 2021) and limited literature related to music education and dyslexia (Flach et al., 2004; Daunt, 2012; Ganschow et al., 1994; Heikkila & Knight, 2012; Hubicki & Miles, 1991; Oglethorpe, 2008: Macmillan,

2005; Miles et al., 2008; Morrow, 2023; Nelson & Hourigan, 2016; Vance, 2004), relatively little material exists to address these specific questions, and research which accounts for the voice of the student with dyslexia in music education is scarce. Furthermore, the limited literature focusing on music education dyslexia research suggests that learning music appears to be regarded as less valuable than learning to read text, although research examining music as a to promote literacy with dyslexic students is becoming more prevalent (Cancer & Antonietti, 2022; Forgeard et al., 2008; Huss et al., 2011; Lorenzo et al., 2014; Overy, 2003; Moritz et al., 2013). However, research also shows that music may have benefits on a variety of levels: socially, emotionally and cognitively (Cooper, 2020; Hallam & Himonides, 2022; Kivijärvi & Poutiainen, 2020).

Accordingly, this project of action research investigated the teaching of two piano students with a formal assessment of dyslexia in the one-to-one lesson context in my teaching practice. Drawing on a literature review of dyslexia and music education and initial lessons with dyslexic students in my teaching practice, these research questions were formulated. Data was collected from an exploratory questionnaire investigating dyslexic students' perceptions of graded music exams. The questionnaire findings indicated a number of areas for further investigation. Semi-structured interviews with a small number of music teachers informed further interviews with a larger group of teachers and illuminated areas of focus for my teaching practice. Data was also gathered from interviews with the parents of the students in my piano teaching practice, in addition to insights from other parents of dyslexic students. Although the initial conception of the research project included more than two dyslexic piano students, the Covid-19 pandemic was a limiting factor in recruiting new students. In order to gain multiple perspectives from different ages and backgrounds, a group of dyslexic individuals with music lesson experiences were interviewed. Analysis of reflective observations from lessons with 'Alex' and 'Ben', supported by video recordings and informed by literature and the interviews, provided data across a number of school terms.

The overall picture emerging from the data indicates a need for greater awareness of dyslexia and training for music teachers in teaching dyslexic students. Furthermore, the findings emphasise that dyslexia does not need to be a barrier to learning about and making music. Rather, these findings appear to demonstrate that negative attitudes and constructs of dyslexia as a deficit, as well as non-inclusive and inflexible exam and educational systems, are

the main obstacles to a high-quality music education experience for dyslexic students. This research reinforces the importance of parental support and the value of positive relationships between teacher-student and parent as promotive factors to mediate risk factors for dyslexic students.

12.2 Reviewing the research questions

12.2.1 (RQ1) What are the perceptions of music teachers regarding their experiences and utilisation of pedagogical practices, including strategies, methods and material, in teaching dyslexic students?

It emerged from the data that negative perceptions of dyslexia outweighed positive perceptions. Misinformation related to dyslexia as a condition and assumptions related to effective teaching strategies were prevalent. Instrumental music teachers often work in isolation and may obtain information from informal routes, although this information may not be research-led. Pedagogical training and support needs were acknowledged and a number of barriers to successful music education for students with dyslexia were noted. Knowledge of dyslexia amongst teachers was correlated with a wider range of strategies, most of which were created by authors of pedagogical material and adapted from strategies used in literacy training. It emerged from this study that specific applications of technology might be beneficial for dyslexic students, particularly those which enable them to compose music and allow for audio playback, to change and adapt scores, and to focus on increasing automaticity with specific aspects of music such as note or pitch recognition, rhythm or music terminology. The high variability of characteristics in dyslexic individuals across the spectrum of neurodivergence, often accompanied by co-occurring conditions, meant that the significance of a flexible approach and sensitivity to individual student needs is needed.

12.2.2 (RQ2) Do adjustments and accommodations which music exam boards allow offer dyslexic students unbiased inclusion? What are the perspectives of teachers and students on this topic?

From the starting point of a lack of research related to dyslexic students' use of graded music exams, indications from this research further illuminate the need for teacher awareness and training. When dyslexic students are prepared in advance to make the best use of reasonable adjustments which remove barriers created by their specific challenges in an exam with an examiner who is sensitive, the experience is considered to be inclusive. Teachers found that the repertoire demands and specific components of exams were disadvantageous for dyslexic students, although they tended to choose exams from different boards depending on the students' needs. However, findings indicate that teachers and students find barriers in obtaining relevant, accessible information from exam boards. This highlights a need for resources to increase awareness of access arrangements and reasonable adjustments, as well as how teachers might prepare students to utilise them effectively during the exam.

12.2.3 (RS3) Do teachers, parents and students perceive dyslexic students to have specific strengths? How might these strengths be utilised in the music lesson context?

Whilst research acknowledges strengths in dyslexia in many areas, for example in entrepreneurial activities, creativity, empathy, resilience and reasoning strengths, scholarship lacked exploration of how these strengths might be utilised in the music lesson context. Despite many participants' focus on the challenges of dyslexia, this research showed that teachers, parents and students with dyslexia recognise specific strengths related to music learning in dyslexic students. It emerged that aural strengths, narrative, interconnected and dynamic reasoning, vivid imagination, creativity, performance-related capabilities and resilience were areas which could be developed and encouraged in lessons. The importance of developing a holistic framework upon which to systematically develop musical concepts was emphasised. The ability to persist in the face of difficulty is needed to sustain repetitive practice over an extended period. Communicating with an audience and playing effectively with an ensemble rely on taking into account another person's perspective and being able to predict the responses of others, or empathy. Connecting with others through mutual understanding is a beneficial performance strength. Many teachers recognise these strengths as compensatory mechanisms for the challenges that dyslexic students experience with sightreading and poor working memory.

The utilisation of these strengths in the music lesson context was explored in greater detail in Reflective Statements 5 and 6. The development of flexible learning profiles with lesson planning and strategies informed by Universal Design for Learning (CAST, 2018) and in collaboration with the student, provided a foundation to identify strengths as well as consider how to remove barriers to learning. The development of a positive relationship with the students, by encouraging their efforts and inviting them to participate in a process of self-

reflection to identify the strategies and methods they found successful, was seen to increase motivation and engagement in the students in my teaching practice.

12.2.4 (RS4) What are dyslexic students' perceptions of their music learning experiences? How might their voices be nurtured and amplified in the pedagogical process?

The perspectives of dyslexic students on their music learning experiences highlight an ability to develop their own ways of learning when empowered and enabled to do so in an environment of empathy, support and compassion. Unfortunately, the results of this research indicate that inflexible curriculum requirements, teachers' lack of adaptability and awareness of accurate knowledge about dyslexia, including potential strengths, often act as impediments in this process. Furthermore, these experiences impact on a student's emotional and mental wellbeing, with some students displaying disruptive behaviour or hiding their difficulties as a coping mechanism in response. The importance of a teacher's role and their development of empathic awareness to create an environment in which the student might flourish cannot be overemphasised. This lends further support for a programme of teacher training with a focus on developing teachers' knowledge about dyslexia, an understanding of a flexible framework (such as UDL) for lesson planning in addition to teaching methods and approaches which are beneficial for dyslexic students.

12.2.5 (RS5) What are the teacher, student and parental perceptions of the pedagogical process and roles in the music lesson context? How can these roles collaborate to collectively enable positive outcomes?

The following findings emerged from examining the interrelationships between teacherstudent-parent. An understanding of the impact of dyslexia on families was a significant factor in being able to relate, as an instrumental teacher, not only to students with dyslexia but also to their parents, who may also in some cases be dyslexic. Knowledge and power imbalances were acknowledged; teachers are somewhat dependent on information being shared by parents regarding assessments or concerns. This includes recognising the effects of past negative educational experiences and how they might influence the parental relationship with their child and communication with the teacher. In order to develop relationships with parents, teachers would benefit from acknowledging the emotional burdens borne by parents, particularly mothers, as they assume roles of advocates and negotiators with schools and teachers in addition to the costs of assessment, extra tuition and assistive technology. Students reported that some teachers gave them extra free tuition and credited success in exams or performance to this.

Furthermore, to access reasonable adjustments and accommodations at school, a formal assessment is required. Obtaining adequate information about reasonable adjustments from music exam boards was time-consuming; knowledge and time are needed to prepare students with dyslexia in the effective use of these reasonable adjustments for music exams as well as school exams. These extra costs are not recognised in the current music provision services and thus are often assumed by teachers and parents. Several risk factors have been identified that impact parental support (adversity and vulnerability factors) including low socioeconomic status, the ability to provide educational resources, assessment costs and exposure to music resources such as instruments or tuition.

Parents require accurate knowledge of dyslexia and how it might affect the student's learning and their legal rights, confidence in their abilities to advocate for students, patience and availability for meetings and the SEN process. Parents, if dyslexic themselves, may require support when faced with the practical or emotional hurdles of their child's challenges and may struggle to help them. They might also find accessing information and filling in text-based forms difficult.

Schools and teachers need an understanding of possible causes of underlying behavioural issues with dyslexic students, accurate knowledge of the use of reasonable adjustments and accommodation and good communication with parents. Factors that support resilience emerge as parental knowledge of dyslexia and how it may affect the student's learning, effective strategies for dealing with organisation or attention challenges, recognising internalising and externalising behaviours and understanding their child's learning profile. If the parent is dyslexic, they may need to keep an open mind about how dyslexia may affect their child differently. Sharing knowledge with the teacher and an openness to partnership with the teacher might enhance the learning process for their child. Finally, by celebrating and praising effort, the parent has the possibility of encouraging their child, in addition to positively promoting role models with dyslexia.

12.3 Summary of my development as a teacher and researcher

I return to McNiff's (2004) fundamental questions of action research here:

- 1. How do I improve what I am doing?
- 2. How do I help you to learn?
- 3. How do I improve what I am doing for our mutual benefit? (p. 19)

12.3.1 Development through analysis of reflective observations

Within Bronfenbrenner's (1979) model (See Reflective Statement 1), the innermost rings, the micro and mesosystems, represent my interactions with my students, their families, and my supervisor. At these levels, analysis of reflective observations, with a broader perspective informed by literature and data collected, led to an evaluation of larger themes. Reflective Statements 1-3 gave context to my position as the researcher, to the students and our lesson environment. The challenges experienced in my teaching practice with dyslexic students motivated this research. Adversities faced during the Covid-19 pandemic and as the result of researching my own teaching practice were described. The reflective statements describe transitional developments regarding power and knowledge balances (RS4), the use of Universal Design for Learning (CAST, 2018) as a framework for lesson planning, selecting strategies from the literature and developing tacit knowledge in the process (RS5), from deficit to strengths-based approaches (RS6) and the need for empathic awareness and the prevention of compassion fatigue, or burnout, (RS7). I describe transitions in my attitude toward dyslexia and my approaches to teaching that occurred over the course of this research. There are a number of teaching strategies which I began to use in this research which I believe have enhanced my teaching for all pupils. My understanding and use of Universal Design for Learning (CAST, 2018) for lesson planning was developed, and I began considering and implementing accessibility principles for all students in my teaching practice.

12.3.2 Development of professional practice

Kember (2002) notes that other outcomes gained through action research might include the 'development of skills, changes in attitudes and the development of revised practices' (Kember, 2002, p. 92). I experienced growth in my understanding of the subject knowledge of music and dyslexia, in knowledge of strategies and the expertise of using them appropriately

(see Reflective Statements 4, 5). There was considerable professional development in my roles as a reflective practitioner and action researcher. Having researched and considered the methodology for this research, I realised that knowledge was socially constructed and arose from the meaning I interpreted from interactions (see Chapter 3). The process of preparing ethical materials for research (see Appendices A-D) allowed me to develop a greater awareness of ethical issues related to research. I attended fourteen research and teachingfocused sessions through the University of York's Building Research and Innovation Capacity Team (BRIC) on topics such as searching for literature, managing data, public engagement, thesis preparation and teaching and learning in Higher Education.

I learned to put data collection methods into practice and developed a number of new skills: designing a survey, interviewing participants, transcribing interviews, video editing, and facilitating a focus group meeting with stakeholders (Chapter 3). At the analysis stage, my understanding of how to organise, code and interpret data was enhanced; I also developed an appreciation for the importance of iteration in this process. Through becoming an Associate Fellow of the Higher Education Academy (2020), I further developed my skills as a reflective practitioner and writer. Additional work as a graduate teaching assistant on the MA Music Education: Instrumental and Vocal Teaching supported my critical analysis skills through the process of marking submissions, both written and lesson videos, according to set criteria. This role also provided further opportunities to develop my teaching skills in small group settings, as well as whole cohort sessions. The opportunity to present seminars to both undergraduate and graduate students at the University of York enabled me to gain experience in developing workshops and delivering presentations effectively; this process was supported by evaluative feedback from my supervisor and session participants. It was also useful that the sessions were developed over a number of years and this helped me to incorporate feedback on delivery and presentation in an iterative process, similar to the process followed in my research.

The following list details various opportunities for sharing an awareness and understanding of my research at the exosystem and macrosystem levels, or areas external to my teaching practice.

1. Research presentations

- a. Presentation at upcoming British Dyslexia Association International Conference (June 2024)
- b. National Conference for Keyboard Pedagogy, USA (June 2023)
- c. British Dyslexia Association Virtual Conference on Core Skills and Creativity (March 2023)
- d. Society for Education and Music Psychology conference (September 2022)
- e. British Dyslexia Association International Conference (May 2021)
- f. International Society for Music Education (August 2020)
- g. European Dyslexia Association's Autumn Research Seminars at Linnaeus
 University, Sweden (September 2019)

2. Workshops and panel

- a. Upcoming presentation York Music Hubs (September 2024)
- b. Music Mark T-Time provocation on inclusive teaching (2024)
- c. Music Education Hubs East Midlands (MEHEM Uprising) for inclusion leads and practitioners (January 2024), Invited to: York Music Hub (2024)
- d. Music Mark Northwest Music Hubs conference (2023)
- e. Music Mark national conference (2023): Panel on inclusivity in music education

3. Consultancy roles

 Associated Board of the Royal Schools of Music (2022) – reviewing Figurenotes (Drake Music Scotland) material for a new project to include alternative notation systems for graded music exam syllabi

4. Committee member

a. British Dyslexia Association music committee (2020 - present)

The workshops and teaching experiences were particularly helpful, as participants' feedback (see Appendix O) further reinforced the need for the findings of my research to be disseminated. This motivated me to adapt and refine Catts and Petscher's (2022) riskresilience model to a new model (Figure 11.3) as a tool to help teachers, parents and dyslexic students, music hubs, schools and music organisations to collaborate in supporting one another in the area of music education. As one of the resilience factors is an equipped and adequately knowledgeable teacher, this provided further impetus for the development of the Knowledge and Teaching standards (Table 11.2) as a foundational tool of assessable objectives for a teacher training programme.

12.4 Limitations

This study has followed the approach of action research with data being collected and analysed from a variety of sources to obtain multiple perspectives. Rich description was provided from the analysis of reflective observations of lessons in my own teaching practice, interviews with parents, teachers and students with dyslexia and from the literature. The period of the Covid-19 global pandemic was a limiting factor in terms of the recruitment of participants and also impacted the way lessons were conducted, including the format and my proximity to the students.

One limitation related to the results of this research is the acknowledgement that the responses of teachers/parents/students may have been influenced by existing literature on the topic. This effect, known as confirmation bias, is described as 'seeking or interpreting evidence in ways that are preferential to existing beliefs, expectations, or hypotheses' (Nickerson, 1998, p. 175). Participants may have viewed literature as representative or normative and formed perceptions that were not supported by the evidence from reality, but rather confirmed and supported what they believed based on these prior assumptions. Although interviews with participants were semi-structured, questions were asked in an open-ended way and care was taken not to influence the answers based on previous responses from other participants.

There are some limitations in this research which relate to participants in my teaching practice and my role as the teacher. The research was limited to my teaching practice; however, I sought to mitigate this limitation by obtaining views from music students, teachers and parents. In examining their views, I also sought to obtain the perspectives of students of a variety of ages. A greater number of student participants in my teaching practice might have enhanced my understanding; however, some potential participants were uncomfortable with lessons being recorded and analysed for research purposes and their views were respected. Though I was limited to two dyslexic students in my teaching practice, the study covered a period of years. This enabled a longitudinal approach in which there was time to develop an indepth relationship with my students and their families. Both participants were male and of a similar age; it would have been advantageous to work with participants of diverse genders, ages and sociocultural backgrounds.

I was also mindful that over a period of time, the students' feelings about lessons being recorded might change and I sought to mitigate this by checking in with them and their families periodically to ensure that they were still comfortable with this. Though the students seemed to enjoy their lessons, it appeared that they did feel uncomfortable or self-conscious at times talking directly about dyslexia and its impact on them even though they both came from homes that were open and positive concerning dyslexia. This may have limited the information that they were willing to share with me. I tried to be sensitive to their needs by adapting my research agenda and at times this meant that we did not record lessons or avoided activities with which they did not want to engage.

There were limitations related to my role as a teacher. The results are based on observational studies and are subject to my bias as a teacher. Also, I was close to the participants and their families in my teaching practice. Inevitably, I was limited by my teacher's lens, often believing and wanting to confirm that what I did was right without examining other possibilities, though this was mediated by my supervisor challenging my critical insights and encouraging further reflective, analytical and research-informed practice. Anderson et al. (2007, p. 162) state that 'Our work takes place in relation to a larger body of knowledge and research that can help us think about and understand our sites and practices but that also is limited in terms of our particular contexts'. In researching my practice, I was not trying to solve problems but rather exploring and accounting for the complexities and ambiguities that I encountered. The results are based on observations from my teaching and may be subject to confirmation and attribution bias as the student's teacher. As noted above, I sought to mitigate this by acknowledging my bias (see Chapter 4) and by taking into consideration other perspectives from my supervisor, colleagues, students and their families and to prioritise their aims and objectives in the lessons. This research was focused on my piano students, but there is scope to explore the connections in the research process to further studies exploring music theory more specifically, and those learning other instruments and voice.

12.5 Implications and recommendations for practice and future research

Various implications arise from this study, but the most important aspects concern supporting an in-depth understanding of the relationships between teachers, parents and students with dyslexia in the music education context as well as raising awareness of the benefits of a strengths-focused approach which is tailored to the individual learner. Despite acknowledging the characteristics of dyslexia that may be regarded as 'weaknesses', these findings also suggest that a personalised approach may benefit students by promoting their confidence, developing their identity and self-concept as a dyslexic person, reducing stigma and empowering them with agency to reach their potential. This approach may also focus and direct the teacher to the student's abilities and may influence their positive attitude towards the learner. Parents, similarly, may benefit from reduced anxiety relating to their child's music education and may have a clearer understanding of the progress they may make.

Another implication from the findings of this research is that although there have been a number of changes to instrumental music exams in recent years, including a shift to performance-only exams during the Covid-19 pandemic, barriers remain. These include the need for ease of finding information about reasonable adjustments and accommodations that can be applied in exams, training for teachers in how to prepare dyslexic students to use reasonable adjustments appropriately, training for examiners and support for parents. Greater versatility in assessments, creating those that relate to student progress rather than attainment might be useful. Another example would be to reduce the number of required pieces for performance assessment and to improve the accessibility of online music theory exams.

These findings highlight the pressures that parents, particularly mothers, face in acting as advocates and negotiators of their child's music education experiences; this begins with the high cost of assessment and follows to include additional burdens of finding a 'dyslexiafriendly' teacher, extra tuition time (if needed), costs associated with assistive technology and other resources. Currently, these costs are not recognised in the music education system; students receive either limited support or none at all from music hubs in the form of additional support such as extra tuition or assistive technology. Therefore, it often falls to individual

teachers to offer free lessons or for parents to meet the cost of these expenses. Additional support for dyslexic students from music hubs is recommended.

These findings suggest that appropriate pedagogical approaches for students with dyslexia may be beneficial; this includes teachers having an accurate knowledge of dyslexia and the assessment process, the strengths and weaknesses a dyslexic student may encounter in learning music, co-occurring conditions and how they may overlap with dyslexia, issues relating to visual disturbances and signposting to relevant professionals, considerations of the learning environment and how to adapt lesson plans and activities in recognition of the variability which can exist amongst dyslexic students. Universal Design for Learning provides a constructive framework for teachers to plan lessons and engage students. Teachers are encouraged to represent the music in a variety of ways and to utilise different methods for how the student puts their knowledge into action. The versatility of music, including the various ways in which it can be transmitted (such as aurally, through modelling and kinaesthetic modes), is particularly useful for students with dyslexia. Sight-reading difficulties, which are prevalent, do not need to prove a barrier to the dyslexic student's progress. This highlights the importance of high-quality training and support for music teachers, which the findings of this research directly support through the new risk-resilience model for music training with dyslexic students and the foundational material for a teacher training course from the Knowledge and Practice Standards for teachers of music (see Table 11.2). These resources offer support in the form of structural documents which identify how collaboration and knowledge might be enhanced.

12.5.1 Implications for practice

As discussed in Chapter 11 (Section 11.8), the logistics and cost involved may be potential impediments to disseminating this research in the form of a training programme for music hub inclusion leads. Other possible pathways of sharing these resources may be through a series of microlearning opportunities on easily accessible sites such as YouTube or Instagram. The model and document created as a result of these findings are currently text-based only, but there are a number of alternative options available for presenting the material in a more accessible manner. These might include audio resources, such as a podcast series, or the use of video, as described above. These resources might be linked to a website including additional

information about suggested method books, repertoire, the use of exam access arrangements and reasonable adjustments, all of which would be presented in an accessible manner.

Additional recommendations might be for music hubs to promote an awareness of dyslexia by highlighting the performances of dyslexic musicians as positive role models. They might also create space for dyslexic students, and their teachers, to share best practices through mentoring projects. Although music organisations and programmes have highlighted musicians with physical disabilities, there seems to be a lack of representation of those with hidden disabilities, like dyslexia. It would be beneficial to have a media campaign with the purpose of promoting awareness and reducing the stigma of dyslexia which prevents some musicians from disclosure in their workplace.

12.5.2 Recommendations for future research

The following recommendations for future research directions deriving from these findings include areas related to pedagogy, support and policy:

- This research has demonstrated how technology might be used to engage dyslexic students in the music lesson context, to build automaticity in pitch and rhythm recognition and to create composition material. Due to the rapidly changing pace of technology, future research might explore the potential of augmented reality or AI (artificial intelligence) in instrumental teaching with dyslexic students.
- The strengths of dyslexia might be studied in different educational settings, for example in secondary schools or higher education, to explore how teaching and assessment practices might be developed to reflect an understanding of these strengths and offer students greater flexibility in the learning process and in representing what they have learned.
- There is scope for further understanding how teachers prepare their students for exams when the student is using reasonable adjustments and access arrangements and for teachers' and exam boards' consideration of best practices in supporting the student.
- Dyslexic students' use of graded music exams is an under-researched area, and the
 accessibility of online music theory exams for dyslexic students deserves further
 investigation. This might explore how offering options in the types of questions on the

exam (for example, open-ended questions as opposed to multiple choice questions which have additional reading requirements) might improve their experience.

- Future research might examine ways in which instrumental teachers can explore the potential of creative strengths seen in dyslexic students, by investigating the different forms this may take and informing teachers of potential applications in music education contexts.
- These research findings have highlighted the importance of the mother's role in advocating for their dyslexic children. Research exploring ways that parents obtain and access information may be beneficial for understanding how organisations like the British Dyslexia Association music committee or music hubs might better support them.
- Research with dyslexic instrumental teachers could inform understanding of the needs
 of these teachers and how they develop and use specific teaching approaches and
 strategies with their students. This could lead to greater support for dyslexic
 instrumental teachers and inform pedagogy.

12.6 Conclusions

This research has explored the dyslexic students' voice in the pedagogical process and highlights the importance for teachers in welcoming and facilitating student voice through the co-creation of strategies and methods in the lesson context, supported by a framework like Universal Design for Learning (CAST, 2018) which enables a highly individualised approach. This is consistent with a growing body of literature on the importance of positive relationships and the benefits of developing students' capabilities rather than focusing on their deficits (Armstrong, 2012; Frederickson, 2004; Patson & Waters, 2015; Rappolt-Schlichtmann et al., 2018, Seligman, 2011).

There are a number of vulnerability and adversity factors related to students with dyslexia and their ability to have high-quality music learning experiences. These include challenges with pitch perception, maintaining a steady pulse, reproducing rhythms, motor coordination, spatial awareness and concentration. There are also a number of characteristics related to their disposition and belief systems which might create vulnerabilities including low confidence, poor motivation, limited understanding and use of metacognitive skills as well as a negative identity construct related to dyslexia. However, promotive factors in the form of parental and

teacher support may serve to mediate these risk factors, although this is dependent on the teacher having the skills and knowledge to adequately support the student and their family. External organisations may support this process through offering training, funding support for students and in the case of music exam boards, greater flexibility in the exams and their components.

Parental involvement has the potential to be a valuable form of support for music students (Ang et al., 2023; Creech, 2010). Findings from this study are in line with research which suggests that parental support is a valuable promotive factor in dyslexic students' lives (Catts & Petscher, 2022; Reid, 2016; Ross, 2019; Wilmot et al., 2023). The findings from this research extend an understanding of the potential impact of constructive interactions between parents, teachers and dyslexic students by identifying not only risk factors related to parental support but also various protective and promotive factors. There may be areas of adversity or vulnerability from a socioeconomic perspective which may influence a student's access to instruments, music books and lessons in addition to parents' limited understanding or belief systems about dyslexia, poor communication with schools, lack of confidence to advocate for their child and a lack of understanding of the SEND process and a lack of musical training; however, these may be mediated by additional external support from the teachers, music organisations or music hubs.

But this relies, in part, on the ability of an instrumental music teacher to be sufficiently equipped to recognise these vulnerabilities and knowledgeable enough to be able to offer support to parents and dyslexic students. Thus, the usefulness of the new risk-resilience model (Figure 11.3) is exemplified in describing the synergistic interactions between parents, students and teachers to mediate risk factors and foster resilience, whether this is in the form of protective factors (strengths) for students such as accurate pitch perception, ability to maintain a steady pulse, focused concentration and attention and a strong identity concept, or in the form of parents who are trained musicians and are able to support the student during practice or can confidently advocate for the student in educational settings.

The aim of this research has been to improve and enhance instrumental music teaching for students with dyslexia. In order to do this, training and support needs for instrumental teachers have been identified. A teacher who is equipped to support dyslexic students will

have knowledge of the complexities of dyslexia as well as the implications of different definitions and criteria used to identify and assess dyslexia. They will have an understanding of co-occurring conditions and the impact of these in the music lesson context. They will be knowledgeable about assessment reports and how these might benefit their understanding of how to support a student, as well as the importance of maintaining safe data storage and the importance of confidentiality. An understanding of Universal Design for Learning (CAST, 2018) will enable them to plan lessons with flexibility, adapting materials and methods, including technology, with the goal of sustaining the student's interest and promoting their overall growth in the process. They will be aware of the challenges that a dyslexic student might face in the music learning process and will be aware of general and specific strategies to support them. An equipped teacher will have an understanding of multisensory teaching approaches and a bank of resources and repertoire which support this type of learning. They will understand the importance of a low-arousal learning environment and a collaborative approach in their relationship with the student and parent. By being aware of and acknowledging the challenges that parents and dyslexic students face, the teacher might build empathic awareness and support for the family through authentic and open conversations. An equipped teacher will recognise the signs and understand how to prevent compassion fatigue, or a sense of burnout.

Overall, the results of this study suggest a need for greater research, training and resources for teachers of dyslexic students. Although music and dyslexia literature (Flach et al., 2016; Morrow, 2023; Oglethorpe, 2008; Vance, 2004) has focused on the challenges students face when learning to read music, the findings indicate that when teachers are adequately equipped with knowledge of dyslexia and an understanding of the benefits of a positive and collaborative strengths-focused approach, this will serve to enhance dyslexic students' music learning experiences.

Reference List:

- ABRSM (Associated Board of the Royal Schools of Music). (2019). *Making music accessible*. London: ABRSM.
- Akyurek, G., & Bumin, G. (2019). An investigation of executive function in children with dyslexia.
 Psychiatry and Behavioral Sciences, 9(1), 10-17.
 https://doi.org/10.5455/PBS.20181128092027
- Albouy, P., Benjamin, L., Morillon, B., & Zatorre, R. J. (2020). Distinct sensitivity to spectrotemporal modulation supports brain asymmetry for speech and melody. *Science*, *367*(6481), 1043-1047. <u>https://doi.org/10.1126/science.aaz3468</u>
- Alden, S., & Pollock, V. L. (2011). Dyslexia and the studio: Bridging the gap between theory and practice. International Journal of Art & Design Education, 30(1), 81-89. <u>https://doi.org/https://doi.org/10.1111/j.1476-8070.2011.01671.x</u>
- Alexander-Passe, N. (2008). The sources and manifestations of stress amongst school-aged dyslexics, compared with sibling controls. *Dyslexia*, 14(4), 291-313. <u>https://doi.org/10.1002/dys.351</u>
- Alfieri, L., Brooks, P., Aldrich, N., & Tenenbaum, H. (2010). Does discovery-based instruction enhance learning? *Journal of Educational Psychology*, 103, 1-18. <u>https://doi.org/10.1037/a0021017</u>
- Alias, N. A., & Dahlan, A. (2015). Enduring difficulties: The challenges of mothers in raising children with dyslexia. *Procedia - Social and Behavioural Sciences*, 202, 107-14 <u>https://doi.org/10.1016/j.sbspro.2015.08.213</u>
- Alkhadim, G. S. (2022). Motivating students with dyslexia: The debilitating effects of normative performance goals on their physiological arousal. *Frontiers in Education*, 7, 882164. <u>https://www.frontiersin.org/articles/10.3389/feduc.2022.882164</u>
- Almahrag, K. M. (2022). Analyzing genetic causes of dyslexia to provide implications for early identification. International Journal of Early Childhood Special Education, 14(1), 8-15. <u>https://doi.org/10.9756/INT-JECSE/V14I1.221002</u>

Alloway, T. P., Rajendran, G., & Archibald, L. M. (2009). Working memory in children with developmental disorders. *Journal of Learning Disabilities*, 42(4), 372-382. <u>https://doi.org/10.1177/0022219409335214</u>

Aloba, D. (2022). How to teach dyslexics music. Shakspeare Editorial.

Aloba, D. (2021). *Deborah Aloba - Singing teacher*. Retrieved on August 15, 2022 from http://deborahaloba.com/deborahalobaprofessionalsingingteacher/

Amber Sound Trust (2024). What playing by ear entails. Retrieved on February 1, 2024 from https://soundtouch.ambertrust.org/services/instrumental-and-vocal-tuition/strategies/playing-by-ear-and-improvising/

- Anderson, G., Herr, K., & Nihlen, A. (2007). Studying your own school: An educator's guide to practitioner action research. Sage. <u>https://doi.org/10.4135/9781483329574</u>
- Angrosino, M. V. (2005). Recontextualizing observation: Ethnography, pedagogy, and the prospects for a progressive political agenda. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage handbook of qualitative research* (3rd ed., pp. 729-745). Sage.
- Andreou, E., & Vlachos, F. (2013) Learning styles of typical readers and dyslexic adolescents, Journal of Visual Literacy (32)2, 1-14. <u>https://doi.org/10.1080/23796529.2013.11674707</u>
- Ang, K., Panebianco, C., & Odendaal, A. (2023). Parents and group music lessons: A role theory perspective for music teacher educators. *Journal of Music Teacher Education*, 32(3), 12-25. <u>https://doi.org/10.1177/10570837221121334</u>
- Apostoli, A. (2008). Can computers help? Matching the inner ear with the outer. In T. R. Miles, J.
 Westcombe & D. Ditchfield (Eds.), *Music and dyslexia: A positive approach* (pp. 101-105).
 Wiley.
- Armstrong, T. (2012). Neurodiversity in the classroom: Strength-based strategies to help students with special needs succeed in school and life. ASCD. Retrieved on February 5, 2024 from <u>https://files.ascd.org/staticfiles/ascd/pdf/siteASCD/publications/books/Neurodiversity-inthe-Classroom-sample-chaps.pdf</u>

- Atkinson, S. (1994). Rethinking the principles and practice of action research: The tensions for the teacher-researcher. *Educational Action Research*, 2(3), 383-401.
 <u>https://doi.org/10.1080/0965079940020306</u>
- Attree, E. A., Turner, Mark. J., & Cowell, N. (2009). A virtual reality test identifies the visuospatial strengths of adolescents with dyslexia. *CyberPsychology & Behavior*, 12(2), 163-168. https://doi.org/10.1089/cpb.2008.0204
- Ayotte, J., Peretz, I., & Hyde, K. (2002). Congenital amusia. *Brain, 125*(2), 238-251. <u>https://doi.org/10.1093/brain/awf028</u>
- Backhouse, G (2001). A pianist's story. In T. R. Miles and J. Westcombe (Eds.), *Music and dyslexia: Opening new doors* (pp. 77-83). Wiley.
- Baeck, E. (2002). The neural networks of music. *European Journal of Neurology*, *9*(5), 449-456. <u>https://doi.org/10.1046/j.1468-1331.2002.00439.x</u>
- Bajaj, D., & Bhatia, D. S. (2020). Psychosocial functioning in children with dyslexia: Perspectives from parents, counsellors and teachers. *Disability, CBR & Inclusive Development, 30*(4), 49-76. https://doi.org/10.5463/dcid.v30i4.847
- Barnes, C. (2019). Understanding the social model of disability: Past, present and future. In N.
 Watson, N., Roulstone, A., & Thomas, C. (Eds.), *Routledge handbook of disability studies* (pp. 14-31). Routledge. <u>https://doi.org/10.4324/978042943081</u>
- Bartolomeo, L. (2023). *Flashnote Derby* [Mobile app]. <u>https://apps.apple.com/us/app/flashnote-</u> <u>derby/id453126527</u>
- Bashan, B., & Holsblat, R. (2017). Reflective journals as a research tool: The case of student teachers' development of teamwork. *Cogent Education*, 4(1), 1-15. <u>https://doi.org/10.1080/2331186X.2017.1374234</u>
- Bate E. (2020). Justifying music in the national curriculum: The habit concept and the question of social justice and academic rigour. British Journal of Music Education, 37(1), 3-15. <u>https://doi.org/10.1017/S0265051718000098</u>
- BDA. (2024). *Definition of dyslexia*. Retrieved on February 1, 2024 from <u>https://www.bdadyslexia.org.uk/news/definition-of-dyslexia</u>

- BDA (British Dyslexia Association). (2020). Accredited Level 5 Certificate in dyslexia; literacy, support and intervention. Retrieved on September 5, 2019 from https://www.bdadyslexia.org.uk/services/training/accredited-level-5-certificate-in-dyslexialiteracy-support-and-intervention-2
- BDA (British Dyslexia Association). (2023). *Dyslexia friendly style guide*. Retrieved on April 8, 2024 from https://www.bdadyslexia.org.uk/advice/employers/creating-a-dyslexia-friendlyworkplace/dyslexia-friendly-style-guide
- BDA (British Dyslexia Association). (2019). Educational cost of dyslexia: Financial, standards and attainment cost to education of unidentified and poorly supported dyslexia, and a policy pathway to end the educational cost of dyslexia. British Dyslexia Association
- BDA (British Dyslexia Association). (2022). *Music and dyslexia*. Retrieved on April 5, 2022 from <u>https://www.bdadyslexia.org.uk/advice/children/music-and-dyslexia</u>
- Bégel, V., Dalla Bella, S., Devignes, Q., Vandenbergue, M., Lemaître, M.-P., & Dellacherie, D.
 (2022). Rhythm as an independent determinant of developmental dyslexia. *Developmental Psychology*, *58*(2), 339-358. <u>https://doi.org/10.1037/dev0001293</u>
- Bera. (2018). Ethical Guidelines for Educational Research, fourth edition, London.
 https://www.bera.ac.uk/researchers-resources/publications/ethicalguidelines-for educational-research-2018. Ethical Guidelines for Educational Research, Fourth Edition.

Berg, C. (2019). *Practicing music by design: Historic virtuosi on peak performance*. Routledge.

Berger, R. (2015). Now I see it, now I don't: Researcher's position and reflexivity in qualitative research. *Qualitative Research*, 15(2), 219-234.

https://doi.org/10.1177/1468794112468475

- Berninger, V., & Abbott, R. D. (2013). Differences between children with dyslexia who are and are not gifted in verbal reasoning. *Gifted Child Quarterly*, *57*(4), 223-233. <u>https://doi.org/10.1177/0016986213500342</u>
- Berninger, V. W., & Wolf, B. J. (2009). *Teaching students with dyslexia and dysgraphia: Lessons from teaching and science*. Paul H. Brookes Publishing Co.

- Besson, M., Chobert, J., & Marie, C. (2011). Transfer of training between music and speech:
 Common processing, attention, and memory. *Frontiers in Psychology, 2,*94. https://doi.org/10.3389/fpsyg.2011.00094
- Besson, M., Schön, D., Moreno, S., Santos, A., & Magne, C. (2007). Influence of musical expertise and musical training on pitch processing in music and language. *Restorative Neurology and Neuroscience*, *25*(3-4), 399-410.
- Bhide, A., Power, A., & Goswami, U. (2013). A rhythmic musical intervention for poor readers: A comparison of efficacy with a letter-based intervention. *Mind, Brain, and Education*, 7(2), 113-123. <u>https://doi.org/10.1111/mbe.12016</u>
- Birt, L., Scott, S., Cavers, D., Campbell, C., & Walter, F. (2016). Member checking: A tool to enhance trustworthiness or merely a nod to validation? *Qualitative Health Research*, 26(13), 1802-1811. <u>https://doi.org/10.1177/1049732316654870</u>
- Bishop-Liebler, P. (2008). Similarities and differences in the dyslexic voice. In T. R. Miles, J.Westcombe & D. Ditchfield (Eds.), *Music and dyslexia: A positive approach* (pp. 117-123).Wiley.
- Bishop-Liebler, P., Welch, G., Huss, M., Thomson, J. M., & Goswami, U. (2014). Auditory temporal processing skills in musicians with dyslexia. *Dyslexia*, 20(3), 261-279. <u>https://doi.org/10.1002/dys.1479</u>
- Blackman, S. (2011), Using pupil perspective research to inform teacher pedagogy: What Caribbean pupils with dyslexia say about teaching and learning. *Journal of Research in Special Educational Needs*, 11, 178-185. <u>https://doi.org/10.1111/j.1471-3802.2011.01197.x</u>
- Blank, I. A., & Fedorenko, E. (2020). No evidence for differences among language regions in their temporal receptive windows. *NeuroImage*, *219*, 116925. <u>https://doi.org/10.1016/j.neuroimage.2020.116925</u>
- Boardman, K. (2020). An exploration of teachers' perceptions and the value of multisensory teaching and learning: A perspective on the influence of Specialist Dyslexia Training in England. *Education*, 48(7), 3-13. <u>https://doi.org/10.1080/03004279.2019.1653349</u>

- Boebinger, D., Norman-Haignere, S. V., McDermott, J. H., & Kanwisher, N. (2021). Music-selective neural populations arise without musical training. *Journal of Neurophysiology*, *125*(6), 2237-2263. <u>https://doi.org/10.1152/jn.00588.2020</u>
- Bogdan, R. C. & Biklin S. K. (1998). *Qualitative research for education: An introduction to theory and methods*. (3rd ed.). Allyn and Bacon.
- Boll-Avetisyan, N., Bhatara, A., & Höhle, B. (2020). Processing of rhythm in speech and music in adult dyslexia. *Brain Sciences*, *10*(5), 261. <u>https://doi.org/10.3390/brainsci10050261</u>
- Bonacina, S., Cancer, A., Lanzi, P. L., Lorusso, M. L., & Antonietti, A. (2015). Improving reading skills in students with dyslexia: The efficacy of a sublexical training with rhythmic background. *Frontiers in Psychology*, *6*, 1510. <u>https://doi.org/10.3389/fpsyg.2015.01510</u>
- Bonifacci, P., & Snowling, M. J. (2008). Speed of processing and reading disability: A cross-linguistic investigation of dyslexia and borderline intellectual functioning. *Cognition*, 107(3), 999-1017. <u>https://doi.org/10.1016/j.cognition.2007.12.006</u>
- Bonifacci, P., Montuschi, M., Lami, L., & Snowling, M. J. (2014). Parents of children with dyslexia:
 Cognitive, emotional and behavioural profile. *Dyslexia*, 20(2), 175-190.
 https://doi.org/10.1002/dys.1469
- Bonner, A., & Tolhurst, G. (2002). Insider-outsider perspectives of participant observation. *Nurse Researcher*, *9*(4), 7-19. <u>https://doi.org/10.7748/nr2002.07.9.4.7.c6194</u>
- Botha, M., Chapman, R., Giwa Onaiwu, M., Kapp, S. K., Stannard Ashley, A., & Walker, N. (2024).
 The neurodiversity concept was developed collectively: An overdue correction on the origins of neurodiversity theory. *Autism: The International Journal of Research and Practice*, 13623613241237871. Advance online publication.
 https://doi.org/10.1177/13623613241237871

Boyle, K. (2020). The instrumental music teacher: Autonomy, identity and the portfolio career in *music*. Routledge.

Boysen, G. A. (2024). Lessons (not) learned: The troubling similarities between learning styles and universal design for learning. *Scholarship of Teaching and Learning in Psychology*, 10(2), 207-221. <u>https://doi.org/10.1037/stl0000280</u>
- Brannick, T., & Coghlan, D. (2007). In defense of being "Native": The case for insider academic research. Organizational Research Methods, 10(1), 59–74. <u>https://doi.org/10.1177/1094428106289253</u>
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2). <u>https://doi.org/10.1191/1478088706qp063oa</u>
- Bresin, R., & Friberg, A. (2011). Emotion rendering in music: Range and characteristic values of seven musical variables. *Cortex*, 47(9), 1068-1081. <u>https://doi.org/10.1016/j.cortex.2011.05.009</u>
- Brimo, K., Dinkler, L., Gillberg, C., Lichtenstein, P., Lundström, S., & Åsberg Johnels, J. (2021). The co-occurrence of neurodevelopmental problems in dyslexia. *Dyslexia*, 27(3), 277-293. <u>https://doi.org/10.1002/DYS.1681</u>
- Bronfenbrenner, U. (1979). *The ecology of human development: Experiments by nature and design*. Cambridge.
- Bronfenbrenner, U., & Evans, G. W. (2000). Developmental science in the 21st century: Emerging questions, theoretical models, research designs and empirical findings. *Social Development*, 9(1), 115-125. <u>https://doi.org/10.1111/1467-9507.00114</u>
- Bronfenbrenner, U., & Morris, P. A. (2006). The bioecological model of human development. In R.
 M. Lerner, W. Damon (Eds.), *Handbook of child psychology: Theoretical models of human development* (pp. 793-828). John Wiley & Sons, Inc.
- Brunswick, N. (2012). Dyslexia: A beginner's guide. Oneworld Publications.
- Burden R. (2008). Is dyslexia necessarily associated with negative feelings of self-worth? A review and implications for future research. *Dyslexia*, 14(3), 188-196. <u>https://doi.org/10.1002/dys.371</u>
- Burns, A. (2010). *Doing action research in English language teaching: A guide for practitioners*. Routledge.
- Burwell, K., Carey, G., & Bennett, D. (2019). Isolation in studio music teaching: The secret garden. Arts and Humanities in Higher Education, 18(4), 372-394. <u>https://doi.org/10.1177/1474022217736581</u>

Buzan, T. (2009). The memory book: How to remember anything you want. Pearson.

- Byrne, E. M., Jensen, H., Thomsen, B. S., & Ramchandani, P. G. (2023). Educational interventions involving physical manipulatives for improving children's learning and development: A scoping review. *Review of Education*, 11(2), 1-42. <u>https://doi.org/10.1002/rev3.3400</u>
- Caccia, M., & Lorusso, M. L. (2021). The processing of rhythmic structures in music and prosody by children with developmental dyslexia and developmental language disorder. *Developmental Science*, 24(1), e12981. <u>https://doi.org/10.1111/desc.12981</u>
- Cain, T. (2008). The characteristics of action research in music education. *British Journal of Music Education*, 25(3), 283-313. <u>http://doi:10.1017/S0265051708008115</u>
- Camilleri, S., Chetcuti, D., & Falzon, R. (2019). 'They labeled me ignorant': Narratives of Maltese youth with dyslexia on national examinations. SAGE Open, 9(2). <u>https://doi.org/10.1177/2158244019855674</u>
- Campbell, L. R. (2021). Doctoral research amidst the Covid-19 pandemic: Researcher reflections on practice, relationships, and unexpected intimacy. *Qualitative Social Work, 20*(1-2), 570-578. <u>https://doi.org/10.1177/1473325020981090</u>
- Cancer, A., & Antonietti, A. (2022). Music-based and auditory-based interventions for reading difficulties: A literature review. *Heliyon*, *8*(4), e09293. <u>https://doi.org/10.1016/j.heliyon.2022.e09293</u>
- Cancer, A., Manzoli, S., Antonietti, A., & Besson, M. (2016). The alleged link between creativity and dyslexia: Identifying the specific process in which dyslexic students excel. *Cogent Psychology*, *3*(1), 1190309. <u>https://doi.org/10.1080/23311908.2016.1190309</u>
- Cantiani, C., Riva, V., Piazza, C., Bettoni, R., Molteni, M., Choudhury, N., & Benasich, A.
 A. (2016). Auditory discrimination predicts linguistic outcome in Italian infants with and without familial risk for language learning impairment. *Developmental Cognitive Neuroscience*, 20, 23–34. <u>https://doi.org/10.1016/j.dcn.2016.03.002</u>
- Carr, W., & Kemmis, S. (1986). *Becoming critical: Education, knowledge and action research*. Falmer.

- Casanova, M. F., El-Baz, A. S., Giedd, J., Rumsey, J. M., & Switala, A. E. (2010). Increased white matter gyral depth in dyslexia: Implications for corticocortical connectivity. *Journal of Autism and Developmental Disorders, 40*(1), 21–29. <u>https://doi.org/10.1007/s10803-009-0817-1</u>
- Casanova, M. F., Buxhoeveden, D. P., Cohen, M., Switala, A. E., & Roy, E. L. (2002). Minicolumnar pathology in dyslexia. *Annals of Neurology*, *52*. <u>https://doi.org/108–110.10.1002/</u>
- CAST (2018). Universal Design for Learning Guidelines version 2.2. [Graphic Organizer]. Retrieved on April 10, 2024 from <u>http://www.udlcenter.org/aboutudl/udlguidelines</u>
- CAST (2020). *Key questions to consider when planning lessons*. (Reprinted from *Universal design for learning: theory and practice*, by A. Meyer, D. H. Rose, & D. Gordon, 2014). Retrieved on April 8, 2024 from <u>http://www.cast.org/products-services/resources/2020/udl-guidelines-key-questions-planning-lessons</u>
- Catts, H. W., & Petscher, Y. (2022). A cumulative risk and resilience model of dyslexia. *Journal of Learning Disabilities*, 55(3), 171-184. <u>https://doi.org/10.1177/00222194211037062</u>

Check, J., & Schutt, R. K. (2012). Research methods in education. Thousand Oaks, CA: Sage.

- Chen, X., Affourtit, J., Ryskin, R., Regev, T. I., Norman-Haignere, S., Jouravlev, O., Malik-Moraleda, S., Kean, H., Varley, R., & Fedorenko, E. (2023). The human language system, including its inferior frontal component in "Broca's area," does not support music perception. *Cerebral Cortex*, 33(12), 7904-7929. <u>https://doi.org/10.1093/cercor/bhad087</u>
- Chern, A., Tillmann, B., Vaughan, C., & Gordon, R. L. (2018). New evidence of a rhythmic priming effect that enhances grammaticality judgments in children. *Journal of Experimental Child Psychology*, *173*, 371-379. <u>https://doi.org/10.1016/j.jecp.2018.04.007</u>
- Chioncel, N. E., Van Der Veen, R. G. W., Wildemeersch, D. & Jarvis, P. (2003). The validity and reliability of focus groups as a research method in adult education. *International Journal of Lifelong Education*, 22(5), 495-517. <u>https://doi.org/10.1080/0260137032000102850</u>
- Cochran-Smith, M. & Lytle, S. (2009). Teacher research as stance. In S. E. Noffke & B. Somekh (Eds). *The SAGE handbook of educational action research* (pp. 39-49). SAGE Publications Ltd, https://www.doi.org/10.4135/9780857021021

- Coghlan, D., & Brydon-Miller, M. (2014). *The SAGE encyclopedia of action research* (Vols. 1-2). SAGE.
- Cogo-Moreira, H., Ávila, C. R. B., de, Ploubidis, G. B., & de Jesus Mari, J. (2013). Effectiveness of music education for the improvement of reading skills and academic achievement in young poor readers: A pragmatic cluster-randomized, controlled clinical trial. *Public Library of Science (PloS) One 8*(3), e59984. https://doi.org/10.1371/journal.pone.0059984
- Cohen, L., Manion, L., & Morrison, K. (2011). *Research methods in education*. Routledge.
- Colenbrander, D., Ricketts, J., & Breadmore, H. L. (2018). Early identification of dyslexia:
 Understanding the issues. Language, Speech, and Hearing Services in Schools, 49(4), 817 828. <u>https://doi.org/10.1044/2018_LSHSS-DYSLC-18-0007</u>
- Colling, L. J., Noble, H. L., & Goswami, U. (2017). Neural entrainment and sensorimotor synchronization to the beat in children with developmental dyslexia: An EEG study. *Frontiers in Neuroscience*, *11*, 360. <u>https://doi.org/10.3389/fnins.2017.00360</u>
- Cook-Sather, A. (2006). Sound, presence, and power: "Student voice" in educational research and reform. *Curriculum Inquiry, 36*(4), 359-390. <u>https://doi.org/10.1111/j.1467-</u> <u>873X.2006.00363.x</u>
- Cooper, P. K. (2020). It's all in your head: A meta-analysis on the effects of music training on cognitive measures in schoolchildren. *International Journal of Music Education, 38*(3), 321-336. https://doi.org/10.1177/0255761419881495
- Cornejo, M., Bustamante, J., del Río, M., de Toro, X., & Latorre, M. S. (2023). Researching with qualitative methodologies in the time of coronavirus: Clues and challenges. *International Journal of Qualitative Methods*, 22, 160940692211501. <u>https://doi.org/10.1177/16094069221150110</u>
- Couvignou, M., Peretz, I., & Ramus, F. (2019). Comorbidity and cognitive overlap between developmental dyslexia and congenital amusia. *Cognitive Neuropsychology, 36*(1-2), 1-17. <u>https://doi.org/10.1080/02643294.2019.1578205</u>
- Couvignou, M., & Kolinsky, R. (2021). Comorbidity and cognitive overlap between developmental dyslexia and congenital amusia in children. *Neuropsychologia*, *155*, 107811. https://doi.org/10.1016/j.neuropsychologia.2021.107811

- Couvignou, M., Tillmann, B., Caclin, A., & Kolinsky, R. (2023). Do developmental dyslexia and congenital amusia share underlying impairments? *Child Neuropsychology: A Journal on Normal and Abnormal Development in Childhood and Adolescence, 29*(8), 1294-1340.
 https://doi.org/10.1080/09297049.2022.2162031
- Couvignou, M., Peyre, H., Ramus, F., & Kolinsky, R. (2024). Do early musical impairments predict later reading difficulties? A longitudinal study of pre-readers with and without familial risk for dyslexia. *Developmental Science*, e13519. <u>https://doi.org/10.1111/desc.13519</u>
- Cox, T., & Kilshaw, H. (2021). Creating a more inclusive classical music: A study of the English orchestral workforce and the current routes to joining it. Arts Council England. <u>https://www.artscouncil.org.uk/sites/default/files/download-file/Executive_Summary.pdf</u>
- Craft, M. (2017). Education for diversity. In M. Craft (Ed.), *Education and Cultural Pluralism* (pp.5-26). Routledge.
- Creech, A. (2010) Learning a musical instrument: the case for parental support. *Music Education Research, 12*(1), 13-32. <u>https://doi.org/10.1080/14613800903569237</u>
- Creech, A., & Hallam, S. (2010). Interpersonal interaction within the violin teaching studio: The influence of interpersonal dynamics on outcomes for teachers. *Psychology of Music*, 38(4), 403-421. <u>https://doi.org/10.1177/03057356093519</u>
- Creswell, J. W. (2013). *Qualitative inquiry and research design: Choosing among five approaches* (3rd ed.). Sage Publications.
- Creswell, J. W., & Miller, D. L. (2000). Determining validity in qualitative inquiry. *Theory Into Practice, 39*(3), 124-130. <u>https://doi.org/10.1207/s15430421tip3903_2</u>
- Crotty, M. (1998). The foundations of social research: Meaning and perspective in the research process. SAGE.
- Culp, M. E. (2017). The relationship between phonological awareness and music aptitude. Journal of Research in Music Education, 65(3), 328-346. https://doi.org/10.1177/0022429417729655
- Cunliffe, A. L. (2004). On becoming a critically reflexive practitioner. *Journal of Management Education, 28*(4), 51-64. <u>https://doi.org/10.1177/1052562904264440</u>

- Dadds, M. (1993). The feeling of thinking in professional self-study. *Educational Action Research*, 1(2), 287-303.
- Dadds, M. (2003). Dissidence, difference and diversity in action research. *Educational Action Research*, 11(2), 265-282. <u>https://doi.org/10.1080/09650790300200209</u>
- Dadds, M. (2008). Empathetic validity in practitioner research. *Educational Action Research*, 16(2), 279-290. <u>https://doi.org/10.1080/09650790802011973</u>
- Dahle, A. E., Knivsberg, A. M., & Andreassen, A. B. (2011). Coexisting problem behaviour in severe dyslexia. Journal of Research in Special Educational Needs, 11(3), 162-170. <u>https://doi.org/10.1111/j.1471-3802.2010.01190.x</u>
- Dan, B. (2019). Developmental coordination disorder continues to evolve. *Developmental Medical Child Neurology, 61*, 240-240. <u>https://doi.org/10.1111/dmcn.14121</u>
- Daniel, B. & Wassell, S. (2002). *Assessing and promoting resilience in vulnerable children*. Jessica Kingsley.
- Darrow, A.-A., & Adamek, M. (2018). Instructional strategies for the inclusive music classroom. General Music Today, 31(3). <u>https://doi.org/10.1177/1048371318756625</u>
- Daubney, A. (2023). *Keynote address for the Westminster Education Forum: The current state of music education*. Retrieved on February 1, 2024 from <u>https://www.ism.org/news/the-current-state-of-music-education-and-the-impact-of-the-national-plan-for-music-education/</u>
- Davidson, J. W., Sloboda, J. A., & Howe, M. J. A. (1995). The role of parents and teachers in the success and failure of instrumental learners. *Bulletin of the Council for Research in Music Education*, 127, 40–44. <u>http://www.jstor.org/stable/40318764</u>
- Davis, R. D. and Braun, E. M. (2003). *The gift of learning*. Berkley Pub.
- Daunt, S. (2012). Issues and strategies that can be applied to most areas of the performing arts. In
 S. Daunt (Ed.), *Music and other performing arts and dyslexia* (pp. 25-45). British Dyslexia
 Association.
- Della Ventura, M. (2019). Exploring the impact of artificial intelligence in music education to enhance the dyslexic student's skills. In *Learning Technology for Education Challenges: 8th*

International Workshop, LTEC 2019, Zamora, Spain, July 15–18, 2019, Proceedings 8 (pp. 14-22). Springer International Publishing.

- Denckla, M. B., & Rudel, R. G. (1976). Rapid 'automatized' naming (R.A.N.): Dyslexia differentiated from other learning disabilities. *Neuropsychologia*, *14*(4), 471-479. https://doi.org/https://doi.org/10.1016/0028-3932(76)90075-0
- Denhart, H. (2008). Deconstructing barriers: Perceptions of students labeled with learning disabilities in higher education. *Journal of Learning Disabilities*, 41(6), 483-497. <u>https://doi.org/10.1177/0022219408321151</u>
- Denscombe, M. (2008). Communities of practice: A research paradigm for the mixed methods approach. *Journal of Mixed Methods Research, 2*(3), 270-283. <u>https://doi.org/10.1177/1558689808316807</u>
- Denscombe, M. (2014). *The good research guide: For small-scale social research projects*. (5th ed). Open University Press.
- Department of Education (2011). The importance of music: A national plan for music education. HMSO
- Department of Education (2022). *The power of music to change lives: A national plan for music education*. Retrieved on January 5, 2024 from <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment</u> <u>data/file/1086619/The Power of Music to Change Lives.pdf</u>
- Department of Education (2023). *Special educational needs in England*. HMSO. Retrieved on February 1, 2024 from <u>https://explore-education-statistics.service.gov.uk/find-</u> <u>statistics/special-educational-needs-in-england</u>
- Diamond, A. (2013). Executive functions. *Annual Review of Psychology, 64*, 135-168. <u>https://doi.org/10.1146/annurev-psych-113011-143750</u>
- Dick, B. (1997). *Approaching an action research thesis: An overview*. Retrieved on August 20, 2023 from <u>http://www.uq.net.au/action_research/arp/phd.html</u>)

- Dixon, M., Lee, S., & Ghaye, T. (2016). Strengths-based reflective practices for the management of change: Applications from sport and positive psychology. *Journal of Change Management*, 16(2), 142-157. <u>https://doi.org/10.1080/14697017.2015.1125384</u>
- Doyle, S. (2007). Member checking with older women: A framework for negotiating meaning. *Health Care for Women International, 28*(10), 888-908. https://doi.org/10.1080/07399330701615325
- Drake Music Scotland. (2024). *Figurenotes*. Retrieved on February 1, 2024 from <u>https://drakemusicscotland.org/figurenotes/</u>
- Ditchfield, D. (2008). The paperwork. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds.), *Music and dyslexia: A positive approach* (pp. 75-81). Wiley.
- DSM (Diagnostic and Statistical Manual of Mental Disorders) (2013). Specific learning disorders. (5th Ed.). American Psychiatric Association.
- Dwyer, S. C., & Buckle, J. L. (2009). The space between: On being an insider-outsider in qualitative research. International Journal of Qualitative Methods, 54-63. <u>https://doi.org/10.1177/160940690900800105</u>
- Dye, V. (2011). Reflection, reflection, reflection, I'm thinking all the time, why do I need a theory or model of reflection? In D. McGregor & L. Cartwright (Eds.), *Developing Reflective Practice* (pp. 217-234). Open University Press.
- Dyment, J. E., & O'Connell, T. S. (2011). Assessing the quality of reflection in student journals: A review of the research. *Teaching in Higher Education*, 16, 81-97. <u>http://dx.doi.org/10.1080/13562517.2010.507308</u>
- Eide, F., & Eide, B. (2011). *The dyslexic advantage: Unlocking the hidden potential of the dyslexic brain*. Hay House.
- Elder, G. H. (1998). The life course as developmental theory. *Child Development, 69*(1), 1-12. <u>https://doi.org/10.1111/j.1467-8624.1998.tb06128.x</u>
- Elliott, J. (1989). Educational theory and the professional learning of teachers. *Cambridge Journal of Education, 19*(1), 81-100.
- Elliott, J. (1991). Action research for educational change. Open University Press.

Elliott, J. G. (2009). The nature of teacher authority and teacher expertise. *Support for Learning,* (24), 197-203. <u>https://doi.org/10.1111/j.1467-9604.2009.01429.x</u>

Elliott, J. G. & Grigorenko, E. L. (2014). The dyslexia debate. Cambridge University Press.

- Enco-Jáuregui, L., Meneses-Claudio, B., & Auccacusi-Kañahuire, M. (2023). Web accessibility for people with dyslexia: A systematic literature review. *EAI Endorsed Transactions on Pervasive Health and Technology*, *9*. <u>https://doi.org/10.4108/eetpht.9.4274</u>
- Everatt, J., Weeks, S., & Brooks, P. (2008). Profiles of strengths and weaknesses in dyslexia and other learning difficulties. *Dyslexia*, 14(1), 16-41. <u>https://doi.org/10.1002/dys.342</u>

Faber, N., & Faber, R. (1998). Accelerated piano adventures book one. Faber Piano Adventures.

- Fautley, M. and Whittaker, A. (2018). Key data on Music Education Hubs 2018. Retrieved on April 1, 2024 from https://www.artscouncil.org.uk/sites/default/files/downloadfile/Music%20Education%20H ubs%2C%20Key%20Data%20-%202018.pdf
- Fawcett, A. (2009). Thomas Richard 'Tim' Miles, psychologist, Born 11 March 1923; Died 11 December 2008. *Dyslexia*, 15, 69-71. <u>https://doi.org/10.1002/dys.387</u>
- Fedorenko, E., Nieto-Castañón, A., & Kanwisher, N. (2012). Syntactic processing in the human brain: What we know, what we don't know, and a suggestion for how to proceed. *Brain and Language*, 120(2), 187-207. <u>https://doi.org/10.1016/j.bandl.2011.01.001</u>
- Ferrer, E., Shaywitz, B. A., Holahan, J. M., Marchione, K. E., Michaels, R., & Shaywitz, S. E. (2015). Achievement gap in reading is present as early as first grade and persists through adolescence. *The Journal of Pediatrics*, *167*(5), 1121-5.e2. <u>https://doi.org/10.1016/j.jpeds.2015.07.045</u>
- Fielding, M. (2004) Transformative approaches to student voice: Theoretical underpinnings, recalcitrant realities. *British Educational Research Journal*, 30, 295-311. <u>http://dx.doi.org/10.1080/0141192042000195236</u>
- Fiveash, A., Bedoin, N., Gordon, R. L., & Tillmann, B. (2021). Processing rhythm in speech and music: Shared mechanisms and implications for developmental speech and language disorders. *Neuropsychology*, 35(8), 771-791. <u>https://doi.org/10.1037/neu0000766</u>

- Flach, N., Timmermans, A., & Korpershoek, H. (2016). Effects of the design of written music on the readability for children with dyslexia. *International Journal of Music Education*, 34(2), 234-246. <u>https://doi.org/10.1177/0255761414546245</u>
- Flaugnacco, E., Lopez, L., Terribili, C., Montico, M., Zoia, S., & Schön, D. (2015). Music training increases phonological awareness and reading skills in developmental dyslexia: A randomized control trial. *Public Library of Science One*, *10*(9), e0138715. <u>https://doi.org/10.1371/journal.pone.0138715</u>
- Foley Nicpon, M., Allmon, A., Sieck, B., & Stinson, R. D. (2011). Empirical investigation of twiceexceptionality: Where have we been and where are we going? *The Gifted Child Quarterly*, 55(1), 3-17. <u>https://doi.org/10.1177/0016986210382575</u>
- Folkestad, G. (2006). Formal and informal learning situations or practices vs formal and informal ways of learning. *British Journal of Music Education*, 23(2), 135-145. <u>https://doi:10.1017/S0265051706006887</u>
- Forgeard, M., Schlaug, G., Norton, A., Rosam, C., Iyengar, U., & Winner, E. (2008). The relation between music and phonological processing in normal-reading and children with dyslexia. *Music Perception*, 25(4), 383-390 <u>https://doi.org/10.1525/mp.2008.25.4.383</u>

forScore. (2022). *forScore* (Version 13.0.7). [Mobile app]. https://apps.apple.com/us/app/forscore/id363738376?ign-mpt=uo%3D4

- Fostick, L., & Revah, H. (2018). Dyslexia as a multi-deficit disorder: Working memory and auditory temporal processing. *Acta Psychologica*, 183, 19-28. <u>https://doi.org/10.1016/j.actpsy.2017.12.010</u>
- Fournier, G., Moreno Sala, M. T., Dubé, F., & O'Neill, S. (2019). Cognitive strategies in sightsinging: The development of an inventory for aural skills pedagogy. *Psychology of Music*, 47(2), 270-283. <u>https://doi.org/10.1177/0305735617745149</u>
- Francis, D. A., Caruana, N., Hudson, J. L., & McArthur, G. M. (2019). The association between poor reading and internalising problems: A systematic review and meta-analysis. *Clinical Psychology Review*, 67, 45-60. <u>https://doi.org/10.1016/J.CPR.2018.09.002</u>
- Franco, M. A. S. (2005). The pedagogy of action research. *Educação e Pesquisa*, *31*(3), 483-502. <u>https://doi.org/10.1590/S1517-97022005000300011</u>

François, C., Chobert, J., Besson, M., & Schön, D. (2013). Music training for the development of speech segmentation. *Cerebral Cortex*, 23(9), 2038-2043. <u>https://doi.org/10.1093/cercor/bhs180</u>

Frank, R. with Livingston, K. (2002). The secret life of the dyslexic child. Rochdale.

- Fraser, S. & Robinson, C. (2004). Paradigms and philosophy. In S. Fraser, V. Lewis, S. Ding, M.Kellett & C. Robinson (Eds.), *Doing research with children and young people* (pp. 59-77).Sage.
- Fredrickson, B. L. (2004). The broaden-and-build theory of positive emotions. *Philosophical transactions of the Royal Society of London. Series B, Biological Sciences*, 359(1449), 1367-1378. <u>https://doi.org/10.1098/rstb.2004.1512</u>
- Frey, A., François, C., Chobert, J., Velay, J. L., Habib, M., & Besson, M. (2019). Music training positively influences the preattentive perception of voice onset time in children with dyslexia: A longitudinal study. *Brain Sciences*, 9(4), 91. <u>https://doi.org/10.3390/brainsci9040091</u>
- Frith, U. (2002). Resolving the paradoxes of dyslexia. In G. Reid and J. Wearmouth (Eds.), *Dyslexia* and literacy, theory and practice. John Wiley & Sons.
- Fujii, S., & Wan, C. Y. (2014). The role of rhythm in speech and language rehabilitation: The SEP hypothesis. *Frontiers in Human Neuroscience*, *8*, 777. <u>https://doi.org/10.3389/fnhum.2014.00777</u>
- Gabay, Y. (2021). Delaying feedback compensates for impaired reinforcement learning in developmental dyslexia. *Neurobiology of Learning and Memory*, *185*, 107518.
 .<u>https://doi.org/10.1016/j.nlm.2021.107518</u>
- Gall, M. (2021). Inclusivity in music education. In S. Vidulin (Ed.), *Proceedings of the seventh international symposium of music pedagogues: Music pedagogy in the context of present and future changes*, Vol. 7 (pp. 47-65).
- Ganschow, L., Lloyd-Jones, J. & Miles, T. R. (1994). Dyslexia and musical notation. *Annals of Dyslexia* 44, 185-202. <u>https://doi.org/10.1007/BF02648161</u>

Gardner, H. (1993). Frames of mind: The theory of multiple intelligences. Basic Books.

- Gardner, J. (2024). Forms and transformations of empathy: Subtleties and complexities of empathic communication. *Psychoanalysis, Self and Context*, 19(1), 80-93. <u>https://doi.org/10.1080/24720038.2023.2266711</u>
- Geiger, O. (2016). "We give you a song and you give us a song"- Reciprocity in action research in the field of music education. *International Journal of Learning, Teaching and Educational Research*, *15*(13), 12-19.
- Geschwind, N., & Galaburda, A. M. (1985). Cerebral lateralization. Biological mechanisms, associations, and pathology: I. A hypothesis and a program for research. Archives of Neurology, 42(5), 428–459. <u>https://doi.org/10.1001/archneur.1985.04060050026008</u>
- Gibbs, G. (1988). *Learning by doing: A guide to teaching and learning methods*. Further Education Unit, Oxford.
- Gibbs, S. J., & Elliott, J. G. (2020). The dyslexia debate: Life without the label. *Oxford Review of Education*, *46*(4), 487-500. <u>https://doi.org/10.1080/03054985.2020.1747419</u>
- Gilger, J. (2017). Beyond a reading disability: comments on the need to examine the full spectrum of abilities/disabilities of the atypical dyslexic brain. *Annals of Dyslexia* 67, 109–113. <u>https://doi.org/10.1007/s11881-017-0142-x</u>
- Glaser, D. (2014). The effects of child maltreatment on the developing brain. *The Medico-Legal Journal*, *82*(3), 97-111. <u>https://doi.org/10.1177/0025817214540395</u>
- Glazzard, J. (2010). The impact of dyslexia on pupils' self-esteem. *Support for Learning, 25*(2), 63-69. <u>https://doi.org/10.1111/j.1467-9604.2010.01442.x</u>

Gobbo, K. (2020). Dyslexia and creativity: Diverse minds. Cambridge Scholars Publisher.

- Goldfus, C. (2012). Intervention through metacognitive development: A case study of a student with dyslexia and comorbid attention deficit disorder (ADD). *Journal of Languages and Culture*, *3*(3). <u>https://doi.org/10.5897/jlc11.042</u>
- Goldman, R., Zahn, C., & Derry, S. J. (2014). Frontiers of digital video research in the learning sciences: Mapping the terrain. In R. K. Sawyer (Ed.), *The Cambridge handbook of the learning sciences* (pp. 213–232). Cambridge University Press.
 https://doi.org/10.1017/CB09781139519526.014

Goodey, S. (2013). *Playing by colour*. Alfred Publishing.

- Gordon, R. L., Fehd, H. M., & McCandliss, B. D. (2015). Does music training enhance literacy skills?
 A meta-analysis. *Frontiers in Psychology*, *6*, 1777.
 https://doi.org/10.3389/fpsyg.2015.01777
- Goswami, U., Thomson, J., Richardson, U., Stainthorp, R., Hughes, D., Rosen, S., & Scott, S. K.
 (2002). Amplitude envelope onsets and developmental dyslexia: A new hypothesis.
 Proceedings of the National Academy of Sciences of the United States of America, 99(16), 10911-10916. https://doi.org/10.1073/pnas.122368599
- Goswami, U., Huss, M., Mead, N., Fosker, T., & Verney, J. P. (2013). Perception of patterns of musical beat distribution in phonological developmental dyslexia: Significant longitudinal relations with word reading and reading comprehension. *Cortex*, 49(5), 1363-1376. <u>https://doi.org/10.1016/j.cortex.2012.05.005</u>
- Goswami, U. (2015). Sensory theories of developmental dyslexia: Three challenges for research. *Nature Reviews Neuroscience*, *16*(1), 43-54. <u>https://doi.org/10.1038/nrn3836</u>
- Goswami, U. (2019). Speech rhythm and language acquisition: An amplitude modulation phase hierarchy perspective. *Annals of the New York Academy of Sciences*, 1453, 67-78. https://doi.org/10.1111/nyas.14137
- Grandey, A. A. (2000). Emotional regulation in the workplace: A new way to conceptualize emotional labor. *Journal of Occupational Health Psychology*, 5(1), 95-110. <u>https://doi.org/10.1037/1076-8998.5.1.95</u>
- Gray, S., Fox, A. B., Green, S., Alt, M., Hogan, T. P., Petscher, Y., & Cowan, N. (2019). Working memory profiles of children with dyslexia, developmental language disorder, or both. *Journal of Speech, Language, and Hearing Research, 62*(6), 1839-1858. https://doi.org/10.1044/2019 JSLHR-L-18-0148

Greenwood, D., & Levin, M. (1998). Introduction to action research. Sage.

Griffith, A. I. (1998). Insider / Outsider: Epistemological privilege and mothering work. *Human* Studies 21, 361-376 . <u>https://doi.org/10.1023/A:1005421211078</u>

- Griffiths, C. B., Norwich, B., & Burden, B. (2004). Parental agency, identity and knowledge:
 Mothers of children with dyslexia. Oxford Review of Education, 30(3), 417-433.
 https://doi.org/10.1080/0305498042000260511
- Groß C., Serrallach B. L., Möhler E., Pousson J. E., Schneider P., Christiner M., & Bernhofs V.
 (2022). Musical performance in adolescents with ADHD, ADD and dyslexia—Behavioral and neurophysiological aspects. *Brain* Sciences, 12(2), 127.
 https://doi.org/10.3390/brainsci12020127
- Guba, E. G., & Lincoln, Y. S. (1994). Competing paradigms in qualitative research. In N. K. Denzin & Y. S. Lincoln (Eds.), *Handbook of qualitative research* (pp. 105-117). Sage.
- Gudmundsdottir, H. R. (2010). Advances in music-reading research. *Music Education Research,* 12(4), 331-338. <u>https://doi.org/10.1080/14613808.2010.504809</u>
- GuitarTabs LLC. (2022). *Rhythm trainer* (Version 2.8.0). [Mobile app]. https://apps.apple.com/us/app/music-rhythm-trainer/id1319997438
- Habib, M., Lardy, C., Desiles, T., Commeiras, C., Chobert, J., & Besson, M. (2016). Music and dyslexia: A new musical training method to improve reading and related disorders. *Frontiers in Psychology*, *7*, 26. <u>https://doi.org/10.3389/fpsyg.2016.00026</u>
- Habib, M. (2021). The neurological basis of developmental dyslexia and related disorders: A reappraisal of the temporal hypothesis, twenty years on. *Brain Sciences*, *11*(6), 708.
 https://doi.org/10.3390/brainsci11060708
- Haft, S. L., Myers, C. A., & Hoeft, F. (2016). Socio-emotional and cognitive resilience in children with reading disabilities. *Current Opinion in Behavioral Sciences*, 10, 133-141. <u>https://doi.org/10.1016/j.cobeha.2016.06.005</u>
- Hall, J., Gaved, M., & Sargent, J. (2021). Participatory research approaches in times of Covid-19: A narrative literature review. *International Journal of Qualitative Methods*, 20. <u>https://doi.org/10.1177/16094069211010087</u>
- Hallam, S., & Himonides, E. (2022). *The power of music: An exploration of the evidence*. Open Book Publishers. <u>https://doi.org/10.11647/OBP.0292</u>

Hammel, A., Hickox, R., & Hourigan, R. (2016). *Winding it back: Teaching to individual differences in music classroom and ensemble settings*. Oxford University Press.

Hammersley, M. (2013). The myth of research-based policy & practice. SAGE Publications.

- Harrar, V., Tammam, J., Pérez-Bellido, A., Pitt, A., Stein, J., & Spence, C. (2014). Multisensory integration and attention in developmental dyslexia. *Current Biology*, *24*(5), 531-535.
 https://doi.org/10.1016/J.CUB.2014.01.029
- Heikkila, E., & Knight, A. (2012). Inclusive music teaching strategies for elementary-age children with developmental dyslexia. *Music Educators Journal, 99*(1), 54-59. <u>https://doi.org/10.1177/0027432112452597</u>
- Heikkinen, H. L. T., Huttunen, R., & Syrjälä, L. (2012). Action research as narrative: Five principles for validation. *Educational Action Research*, *15*(1), 5-19.
 https://doi.org/10.1080/09650790601150709
- Hendren, R. L., Haft, S. L., Black, J. M., White, N. C., & Hoeft, F. (2018). Recognizing psychiatric comorbidity with reading disorders. *Frontiers in Psychiatry*, 9, 101. <u>https://doi.org/10.3389/fpsyt.2018.00101</u>
- Herr, K. & Anderson, G. L. (2005). Quality criteria for action research: An ongoing conversation. In
 K. Herr & G. L. Anderson (Eds.), *The action research dissertation: A guide for students and faculty* (pp. 49-69). Sage.
- Himonides, E. (2012). The misunderstanding of music-technology-education: A meta-perspective,
 In G. McPherson & G. F. Welch (Eds), *The Oxford Handbook of Music Education* (pp. 433-456). Oxford Handbooks. <u>https://doi.org/10.1093/oxfordhb/9780199928019.013.0029</u>
- Hopf, C. (2004). Qualitative interviews: An overview. In U. Flick, E. von Kardorff, & I. Steinke (Eds.), *A companion to qualitative research* (pp. 203-230). Sage.

Hopkins, D. (1993). A teacher's guide to classroom research. Open University Press.

Hornstra, L., Denessen, E., Bakker, J., van den Bergh, L., & Voeten, M. (2010). Teacher attitudes toward dyslexia: Effects on teacher expectations and the academic achievement of students with dyslexia. *Journal of Learning Disabilities, 43*(6), 515-529.
https://doi.org/10.1177/0022219409355479

- Hourigan, R. (2007). Preparing music teachers to teach students with special needs. *Update: Applications of Research in Music Education, 26*(1), 5-14. <u>https://doi.org/10.1177/87551233070260010102</u>
- Hubicki, M., & Miles, T. R. (1991). Musical notation and multisensory learning. *Child Language Teaching and Therapy*, 7(1), 61-78. <u>https://doi.org/10.1177/026565909100700</u>
- Hubicki, M. (2001). A multisensory approach to the teaching of notation. In T. R. Miles & J. Westcombe (Eds.), *Music and dyslexia: Opening new doors* (pp. 85-100). Whurr.
- Hughes, P. (2010). Paradigms, methods and knowledge. In G. MacNaughton, S. Rolfe & I. Siraj-Blatchford (Eds.), *Doing early childhood research* (pp. 31-55). Open University Press.
- Hulme, C., Nash, H. M., Gooch, D., Lervåg, A., & Snowling, M. J. (2015). The foundations of literacy development in children at familial risk of dyslexia. *Psychological Science*, *26*(12), 1877-1886. <u>https://doi.org/10.1177/0956797615603702</u>
- Hülsheger, U. R., & Schewe, A. F. (2011). On the costs and benefits of emotional labor: A metaanalysis of three decades of research. *Journal of Occupational Health Psychology*, 16(3), 361-389. <u>https://doi.org/10.1037/a0022876</u>
- Humphrey, N., & Mullins, P.M. (2002). Self-concept and self-esteem in developmental dyslexia. Journal of Research in Special Educational Needs, 2. <u>https://doi.org/10.1111/j.1471-</u> <u>3802.2002.00163.x</u>
- Huss, M., Verney, J. P., Fosker, T., Mead, N., & Goswami, U. (2011). Music, rhythm, rise time perception and developmental dyslexia: Perception of musical meter predicts reading and phonology. *Cortex*, 47(6), 674-689. <u>https://doi.org/10.1016/j.cortex.2010.07.010</u>
- International Dyslexia Association. (2018). *Knowledge and practice standards for teachers of reading*. Retrieved on September 10, 2022 from <u>https://dyslexiaida.org/knowledge-and-practices/</u>
- Information Commissioner's Office (ICO). (2021). UK GDPR guidelines. Retrieved on April 8, 2024 from https://ico.org.uk/for-organisations/uk-gdpr-guidance-and-resources/
- Jaarsma, B. S., Ruijssenaars A. J. J. M., & Van den Broeck W. (1998). Dyslexia and learning musical notation: A pilot study. *Annals of Dyslexia, 48,* 137-154.

Jäncke L. (2008). Music, memory and emotion. *Journal of Biology*, 7(6), 21. <u>https://doi.org/10.1186/jbiol82</u>

Jantzen, C. (2009). Dyslexia: Learning disorder or creative gift? Floris Books.

- Jones, M. R. (1976). Time, our lost dimension: Toward a new theory of perception, attention, and memory. *Psychological Review*, 83, 323-355. <u>https://doi.org/10.1037/0033-295X.83.5.323</u>
- JSplashApps. (2021). *Music tutor* (Version 3.17). [Mobile app]. Retrieved on September 3, 2021 from <u>https://apps.apple.com/gb/app/music-tutor-sight-reading/id514363426</u>
- Kailaheimo–Lönnqvist, L., Virtala, P., Fandakova, Y., Partanen, E., Leppänen, P. H., Thiede, A., & Kujala, T. (2020). Infant event-related potentials to speech are associated with prelinguistic development. *Developmental Cognitive Neuroscience*, 45, 100831.
 https://doi.org/10.1016/j.dcn.2020.100831
- Kannangara, C. S., Carson, J., Puttaraju, S., & Allen, R. E. (2018). Not all those who wander are lost: Examining the character strengths of dyslexia. *Global Journal of Intellectual and Developmental Disabilities*, 4(5), 86-93. <u>http://doi.org/10.19080/GJIDD.2018.04.555648</u>.
- Kapoula, Z., Ruiz, S., Spector, L., Mocorovi, M., Gaertner, C., Quilici, C., & Vernet, M. (2016).
 Education influences creativity in dyslexic and non-dyslexic children and teenagers. *PLoS ONE*, *11*(3). <u>https://doi.org/10.1371/journal.pone.0150421</u>
- Kember, D. (2002). Long-term outcomes of educational action research projects. *Educational Action Research*, *10*(1), 83-104. https://doi.org/10.1080/09650790200200174
- Kemmis, S., & McTaggart, R. (2005). Participatory action research: Communicative action and the public sphere. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage handbook of qualitative research* (pp. 559-603). Sage.
- Kemmis, S. (2006). Participatory action research and the public sphere. *Educational Action Research*, 14(4), 459-476. <u>https://doi.org/10.1080/09650790600975593</u>
- Kemmis, S., & McTaggart, R. (Eds.) (1988). *The action research planner*. Deakin University Press.
- Kemmis, S., McTaggart, R., & Nixon, R. (2014). *The action research planner: Doing critical participatory action research*. Springer.

- Kere, J. (2014). The molecular genetics and neurobiology of developmental dyslexia as model of a complex phenotype. *Biochemical and Biophysical Research Communications*, 452(2), 236-243. <u>https://doi.org/10.1016/j.bbrc.2014.07.102</u>
- Kerr, H. (2001). Learned helplessness and dyslexia: A carts and horses issue? *Reading*, *35*(2), 82-85. <u>https://doi.org/10.1111/1467-9345.00166</u>
- Kershner, J. R. (2021). An evolutionary perspective of dyslexia, stress, and brain network homeostasis. *Frontiers in Human Neuroscience*, 14. <u>https://doi.org/10.3389/fnhum.2020.575546</u>
- Khalifa, K. (2019). Is Verstehen scientific understanding? *Philosophy of the Social Sciences, 49*(4), 282-306. <u>https://doi.org/10.1177/0048393119847104</u>
- Kidwell, K. E., Clancy, R. L., & Fisher, G. G. (2023). The devil you know versus the devil you don't: Disclosure versus masking in the workplace. *Industrial and Organizational Psychology*, 16(1), 55-60. <u>https://doi.org/10.1017/iop.2022.101</u>
- King, C. (2008). Thirty-Seven oboists. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds), *Music and dyslexia: A positive approach* (pp. 124-136). Wiley.
- Kim-Boyle, D. (2019). 3D Notations and the immersive score. *Leonardo Music Journal, 29* 39-41. <u>https://doi.org/10.1162/lmj_a_01061</u>
- Kirby, P. (2020). Dyslexia debated, then and now: A historical perspective on the dyslexia debate. Oxford Review of Education, 46(4), 472-486. <u>https://doi.org/10.1080/03054985.2020.1747418</u>
- Kirschner, P. A. (2017). Stop propagating the learning styles myth. *Computers & Education*, *106*, 166-171. <u>https://doi.org/10.1016/j.compedu.2016.12.006</u>
- Kivijärvi, S. (2019). Applicability of an applied music notation system: A case study of Figurenotes. International Journal of Music Education, 37(4), 654-666. <u>https://doi.org/10.1177/0255761419845475</u>
- Kivijärvi, S., & Poutiainen, A. (2020). Supplying social capital through music education: A study on interaction in special educational needs students' concerts. *Research Studies in Music Education, 42*(3), 347-367. <u>https://doi.org/10.1177/1321103X19843005</u>

- Knoop-van Campen, C. A. N., Segers E., & Verhoeven L. (2018). The modality and redundancy effects in multimedia learning in children with dyslexia. *Dyslexia*, 24(2), 140-155. https://doi.org/10.1002/dys.1585
- Koelsch, S. (2011). Toward a neural basis of music perception A review and updated model. *Frontiers in Psychology*, *2*, 110. <u>https://doi.org/10.3389/fpsyg.2011.00110</u>
- Kraus, N., & Chandrasekaran, B. (2010). Music training for the development of auditory skills. *Nature Reviews: Neuroscience*, 11(8), 599-605. <u>https://doi.org/10.1038/nrn2882</u>
- Kriss, I., & Evans, B. J. W. (2005). The relationship between dyslexia and Meares-Irlen Syndrome. Journal of Research in Reading, 28(3), 350-364. <u>https://doi.org/10.1111/j.1467-9817.2005.00274.x</u>
- Jenson, J. M., & Fraser, M. W. (2015). A risk and resilience framework for child, youth, and family policy. In J. M. Jenson & M. W. Fraser (Eds), *Social policy for children and families: A risk and resilience perspective* (pp. 6-21). Sage.
- Jentschke, S., Koelsch, S., Sallat, S., & Friederici, A. D. (2008). Children with specific language impairment also show impairment of music-syntactic processing. *Journal of Cognitive Neuroscience*, *20*(11), 1940-1951. <u>https://doi.org/10.1162/jocn.2008.20135</u>
- Jewitt, C. (2012). *An introduction to using video for research*. NCRM Working Paper. NCRM. (Unpublished) Retrieved on April 1, 2024 from <u>https://eprints.ncrm.ac.uk/id/eprint/2259/</u>
- Johnson, M. D., & Bradbury, T. N. (2015). Contributions of social learning theory to the promotion of healthy relationships: Asset or liability? *Journal of Family Theory & Review*, 7(1), 13-27. <u>https://doi.org/10.1111/jftr.12057</u>

Koshy, V. (2009). Action research for improving educational practice. A step-by-step guide. Sage.

- Kous, K., & Polančič, G. (2019). Empirical insights of individual website adjustments for people with dyslexia. *Sensors, 19*(10), 2235. <u>https://doi.org/10.3390/s19102235</u>
- Kriss, I., & Evans, B. J. W. (2005). The relationship between dyslexia and Meares-Irlen Syndrome. Journal of Research in Reading, 28(3), 350–364. <u>https://doi.org/10.1111/j.1467-9817.2005.00274.x</u>

- Ladányi, E., Persici, V., Fiveash, A., Tillmann, B., & Gordon, R. (2020). Is atypical rhythm a risk factor for developmental speech and language disorders? *Wiley Interdisciplinary Reviews: Cognitive Science 5*, e1528. <u>https://doi.org/10.1002/wcs.1528</u>
- Laprise, R. (2017). Empowering the music educator through action research. *Music Educators Journal, 104*(1), 28-33. <u>https://doi.org/10.1177/0027432117708012</u>
- Lea, M. (2008). Sight-reading and memory. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds), *Music and dyslexia: A positive approach* (pp. 92-99). Wiley.
- Lehmann, A., Sloboda, J., & Woody, R. (2007). *Psychology for musicians: Understanding and acquiring the skills*. Oxford University Press.
- Leitão, S., Dzidic, P., Claessen, M., Gordon, J., Howard, K., Nayton, M., & Boyes, M. E. (2017).
 Exploring the impact of living with dyslexia: The perspectives of children and their parents.
 International Journal of Speech-Language Pathology, 19(3), 322-334.
 <u>https://doi.org/10.1080/17549507.2017.1309068</u>
- Leloup, G., Anders, R., Charlet, V., Eula-Fantozzi, B., Fossoud, C., & Cavalli, E. (2021). Improving reading skills in children with dyslexia: Efficacy studies on a newly proposed remedial intervention—repeated reading with vocal music masking (RVM). *Annals of Dyslexia*, *71*, 60-83. <u>https://doi.org/10.1007/s11881-021-00222-4</u>
- Lemon, T. I., & Shah, R. D. (2014). Dyslexia in high performers a study across 4 degree disciplines. *European Psychiatry*, 29, 1. <u>https://doi.org/https://doi.org/10.1016/S0924-9338(14)77546-7</u>
- Leong, V., & Goswami, U. (2014). Impaired extraction of speech rhythm from temporal modulation patterns in speech in developmental dyslexia. *Frontiers in Human Neuroscience*, 96. <u>https://doi.org/10.3389/fnhum.2014.00096</u>
- Leveroy, D. (2013). Locating dyslexic performance: text, identity and creativity. *Research in Drama Education: The Journal of Applied Theatre and Performance, 18*(4), 374-387. <u>https://doi.org/10.1080/13569783.2013.836919</u>

Lewin, K. (1946). Action research and minority problems. Journal of Social Issues, 2, 34-46.

- Liddicoat, A. J. (2022). Teacher education for diversity: Afterword. *Language and Education*, *36*(2), 188-194. <u>https://doi.org/10.1080/09500782.2021.1981929</u>
- Lincoln, Y., & Guba E. (1986). But is it rigorous? Trustworthiness and authenticity in naturalistic evaluation. In D. Williams (Ed.), *Naturalistic Evaluation* (pp. 73-84). Jossey Bass.
- Livingston, E., Siegel L. S., & Ribary, U. (2018). Developmental dyslexia: Emotional impact and consequences. Australian Journal of Learning Difficulties, 23(2), 107-135. <u>https://doi.org/10.1080/19404158.2018.1479975</u>
- Logan, J. (2009). Dyslexic entrepreneurs: The incidence; their coping strategies and their business skills. *Dyslexia*, *15*, 328-346. <u>https://doi.org/10.1002/dys.388</u>
- Lorenzo, O., Herrera, L., Hernández-Candelas, M., & Badea, M. (2014). Influence of music training on language development. A longitudinal study. *Procedia - Social and Behavioral Sciences*, 128, 527-530. <u>https://doi.org/10.1016/j.sbspro.2014.03.200</u>
- Loui, P., Kroog, K., Zuk, J., Winner, E., & Schlaug, G. (2011). Relating pitch awareness to phonemic awareness in children: Implications for tone-deafness and dyslexia. *Frontiers in Psychology*, 2. <u>https://doi.org/10.3389/fpsyg.2011.00111</u>
- Louis, M. C. (2011). Strengths interventions in higher education: The effect of identification versus development approaches on implicit self-theory. *Journal of Positive Psychology*, 6(3), 204-215. <u>https://doi.org/10.1080/17439760.2011.570366</u>
- Lowrey, K. A., Classen, A., & Sylvest, A. (2019). Exploring ways to support preservice teachers' use of UDL in planning and instruction. *Journal of Educational Research and Practice*, 9(1), 261-281. <u>https://doi.org/10.5590/jerap.2019.09.1.19</u>
- Luft, J., & Ingham, H. (1955). The Johari window, a graphic model of interpersonal awareness.
 Proceedings of the western training laboratory in group development. University of
 California Los Angeles.
- Lundetræ, K., & Thomson, J. M. (2018). Rhythm production at school entry as a predictor of poor reading and spelling at the end of first grade. *Reading and Writing*, 31(1), 215-237. <u>https://doi.org/10.1007/s11145-017-9782-9</u>

- Lyon, G. R., Shaywitz, S. E., & Shaywitz, B. A. (2003). A definition of dyslexia. *Annals of Dyslexia*, 53(1), 1-14. <u>https://doi.org/10.1007/s11881-003-0001-9</u>
- Macmillan, J. (2005). *Music and dyslexia and how Suzuki helps*. European Suzuki Association WebJournal. Retrieved on April 2, 2024 from http://www.jennymacmillan.co.uk/Music%20and%20dyslexia%20 for%20ESJ.pdf
- Macmillan, J. (2008). Suzuki benefits for children with dyslexia. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds), *Music and dyslexia: A positive approach* (pp. 137-142). Wiley.

Macmurray, J. (1957). The self as agent. Faber & Faber.

- Madaus, J. W., Foley T. E., McGuire J. M. & Ruban L. M. (2002). Employment self-disclosure of post-secondary graduates with learning disabilities: Rates and rationales. Journal of Learning Disabilities 35(4), 364-369. <u>https://doi.org/10.1177/00222194020350040701</u>
- Magyar-Moe, J. L. (2009). *Therapist's guide to positive psychological interventions*. Elsevier Academic Press.
- Maier, S. F., & Seligman, M. E. (2016). Learned helplessness at fifty: Insights from neuroscience. *Psychological Review*, 123(4), 349-367. <u>https://doi.org/10.1037/rev0000033</u>
- Maiter, S., Simich, L., Jacobson, N., & Wise, J. (2008). Reciprocity: An ethic for community-based participatory action research. *Action Research*, 6(3), 305-325. <u>https://doi.org/10.1177/1476750307083720</u>
- Majeed, N. M., Hartanto, A., & Tan, J. J. X. (2021). Developmental dyslexia and creativity: A metaanalysis. *Dyslexia*, *27*(2), 187–203. https://doi.org/10.1002/dys.1677

Malpas, M. (2017). Self-fulfilment with dyslexia: A blueprint for success. Jessica Kingsley.

- Marchand-Krynski, M. È., Morin-Moncet, O., Bélanger, A. M., Beauchamp, M. H., & Leonard, G. (2017). Shared and differentiated motor skill impairments in children with dyslexia and/or attention deficit disorder: From simple to complex sequential coordination. *Public Library of Science ONE 12*(5): e0177490. <u>https://doi.org/10.1371/journal.pone.0177490</u>
- Marsh, B., & Mitchell, N. (2014). The role of video in teacher professional development. *Teacher Development*, *18*(3), 403-417. <u>https://doi.org/10.1080/13664530.2014.938106</u>

- Maslach, C., & Leiter, M. P. (2016). Understanding the burnout experience: Recent research and its implications for psychiatry. *World Psychiatry*, 15(2), 103-111. <u>https://doi.org/10.1002/WPS.20311</u>
- Masten, A. S. (2018). Resilience theory and research on children and families: Past, present, and promise. *Journal of Family Theory & Review*, 10, 12-31. <u>https://doi.org/10.1111/jftr.12255</u>
- May, H. (2005). Whose participation is it anyway? Examining the context of pupil participation in the UK. *British Journal of Special Education*, 32(1), 29-34. <u>https://doi.org/10.1111/j.0952-3383.2005.00366.x</u>
- Maykut, P., & Morehouse, R. (1994). *Beginning qualitative researchers: A philosophical and practical guide*. Falmer.
- McCarthy, O., Ditchfield, D., Oglethorpe, S. & Westcombe, J. (2008). Winning over the reluctants. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds), *Music and dyslexia: A positive approach* (pp. 55-57). Wiley.
- McCord K. (2004). Moving beyond "That's all I can do:" Encouraging musical creativity in children with learning disabilities. *Bulletin of the Council for Research in Music Education, 159,* 23-32. <u>https://jstor.org/stable/40319205</u>
- McGrath, L. M., Peterson, R. L., & Pennington, B. F. (2020). The multiple deficit model: Progress, problems, and prospects. *Scientific Studies of Reading*, *24*(1), 7-13. <u>https://doi.org/10.1080/10888438.2019.1706180</u>

McLeod, J. (1999). *Practitioner research in counselling*. Sage Publications.

McNiff, J. (2016). You and your action research project. (4th ed.). Routledge.

McNiff, J. with Whitehead, J. (2004). Action research: Principles and practice. (2nd ed.) Routledge.

McNiff, J., & Whitehead, J. (2010). You and your action research project. Routledge.

- McPherson, G. E., & Renwick, J. M. (2011). Self-regulation and mastery of musical skills. In B. J. Zimmerman & D. H. Schunk (Eds.), *Handbook of self-regulation of learning and performance* (pp. 234-248). Routledge/Taylor & Francis Group.
- McRitchie-Pratt, C. (2008). In and around the classroom. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds), *Music and dyslexia: A positive approach* (pp. 19-25). Wiley.

- McTaggart, R. (1994). Participatory action research: Issues in theory and practice. *Educational Action Research*, 2(3), 313-337. <u>https://doi.org/10.1080/0965079940020302</u>
- Melby-Lervåg, M., Lyster, S. A., & Hulme, C. (2012). Phonological skills and their role in learning to read: A meta-analytic review. *Psychological Bulletin*, 138(2), 322-352. https://doi.org/10.1037/a0026744
- Merton, R. K. (1972). Insiders and outsiders: A chapter in the sociology of knowledge. *American Journal of Sociology*, 78(1), 9-47. <u>https://www.jstor.org/stable/2776569</u>
- Merzenich, M. M., Jenkins, W. M., Johnston, P., Schreiner, C., Miller, S. L., & Tallal, P. (1996).
 Temporal processing deficits of language-learning impaired children ameliorated by training. *Science*, 271(5245), 77-81. <u>https://doi.org/10.1126/science.271.5245.77</u>
- Meyer, L., Sun, Y., & Martin, A. E. (2019). Synchronous, but not entrained: exogenous and endogenous cortical rhythms of speech and language processing. *Language, Cognition and Neuroscience*, 35(9), 1089-1099. <u>https://doi.org/10.1080/23273798.2019.1693050</u>
- Miles, T. R., Westcombe, J., Ditchfield, D. (Eds.). (2008). *Music and dyslexia: A positive approach*. Wiley.
- Mills, A. J., Durepos, G. & Wiebe, E. (2010). Encyclopedia of case study research. SAGE.
- Mills, J. (2004) Working in music: Becoming a performer-teacher, *Music Education Research, 6*(3), 245-261, DOI: <u>10.1080/1461380042000281712</u>
- Mills, J. R. (2018). Effective multi-sensory strategies for students with dyslexia. *Kappa Delta Pi Record, 54*(1), 36-40. https://doi.org/10.1080/00228958.2018.1407181
- Miniukovich, A., De Angeli, A., Sulpizio, S., & Venuti, P. (2017, June). Design guidelines for web readability. In Proceedings of the 2017 Conference on Designing Interactive Systems (pp. 285-296). <u>https://doi.org/10.1145/3064663.3064711</u>
- Mitchell, V. W., Harvey, W. S., & Wood, G. (2022). Where does all the 'know how' go? The role of tacit knowledge in research impact. *Higher Education Research and Development*, 41(5), 1664-1678. <u>https://doi.org/10.1080/07294360.2021.1937066</u>

- Mockler, N., & Groundwater-Smith, S. (2015). Seeking for the unwelcome truths: Beyond celebration in inquiry-based teacher professional learning. *Teachers and Teaching: Theory and Practice*, 21(5), 603-614. <u>https://doi.org/10.1080/13540602.2014.995480</u>
- Molinaro, N., Lizarazu, M., Lallier, M., Bourguignon, M., & Carreiras, M. (2016). Out-of-synchrony speech entrainment in developmental dyslexia. *Human Brain Mapping*, *37*(8), 2767-2783. https://doi.org/10.1002/hbm.23206
- Moll, K., Snowling, M. J., & Hulme, C. (2020). Introduction to the special issue "Comorbidities between reading disorders and other developmental disorders". *Scientific Studies of Reading*, 24(1), 1-6. <u>https://doi.org/10.1080/10888438.2019.1702045</u>
- Moojen, S. M. P., Gonçalves, H. A., Bassôa, A., Navas, A. L., de Jou, G. & Miguel, E. S. (2020).
 Adults with dyslexia: How can they achieve academic success despite impairments in basic reading and writing abilities? The role of text structure sensitivity as a compensatory skill.
 Annals of Dyslexia, 70(1), 115-140. <u>https://doi.org/10.1007/s11881-020-00195-w</u>
- Moon, J. A. (2004). A handbook of reflective and experiential learning: Theory and practice. Routledge.
- Morais, J., Periot, A., Lidji, P., & Kolinsky, R. (2010). Music and dyslexia. *International Journal of Arts and Technology, 3*(2-3), 177-194. <u>https://doi.org/10.1504/IJART.2010.032563</u>
- Moreno, S., Lee, Y., Janus, M. & Bialystok, E. (2015). Short-term second language and music training induces lasting functional brain changes in early childhood. *Child Development*, 86(2), 394-406. <u>https://doi.org/10.1111/cdev.12297</u>
- Moritz, C., Yampolsky, S., Papadelis, G., Thomson, J., & Wolf, M. (2013). Links between early rhythm skills, musical training, and phonological awareness. *Reading and Writing: An Interdisciplinary Journal, 26*(5), 739-769. <u>https://doi.org/10.1007/s11145-012-9389-0</u>
- Morrison, K. (1993). *Planning and accomplishing school-centred evaluation*. Peter Francis Publishers.
- Morrow, E. (2023). Music reading for students with learning disabilities. *American String Teacher*, 73(4), 21-26. <u>https://doi.org/10.1177/00031313231197638</u>

- Mortimore, T. & Dupree, J. (2008). *Dyslexia-friendly practice in the secondary classroom*. Learning Matters.
- Musescore (2021). Musescore. [Mobile App]. Retrieved on April 10, 2021 from https://musescore.org/en
- Music First. (2024). *PracticeFirst: Automated practice assessment*. Retrieved on February 1, 2024 from <u>https://www.musicfirst.com/software/practicefirst</u>
- Muter, V. & Snowling, M. J. (2009). Children at familial risk of dyslexia: Practical implications from an at-risk study. *Child and Adolescent Mental Health 14*, 37-41. <u>https://doi.org/10.1111/j.1475-3588.2007.00480</u>.x
- Nancekivell, S. E., Shah, P., & Gelman, S. A. (2020). Maybe they're born with it, or maybe it's experience: Toward a deeper understanding of the learning style myth. *Journal of Educational Psychology*, *112*(2), 221-235. <u>https://doi.org/10.1037/edu0000366</u>
- National Audit Office Report. (2019). Support for pupils with special educational needs and disabilities in England. Retrieved on February 1, 2024 from <u>https://explore-education-statistics.service.gov.uk/find-statistics/special-educational-needs-in-england</u>

Nelson, S. (1998). Flip-a-rhythm. Boosey and Hawkes.

- Nelson, K. P., & Hourigan, R. M. (2016). A comparative case study of learning strategies and recommendations of five professional musicians with dyslexia. Update: Applications of Research in Music Education, 35(1), 54-65. <u>https://doi.org/10.1177/8755123315581341</u>
- NHS (2024). Developmental co-ordination disorder (dyspraxia) in children. Retrieved on April 1, 2024 from <u>https://www.nhs.uk/conditions/developmental-coordination-disorder-</u> <u>dyspraxia/</u>
- Nickerson, R. S. (1998). Confirmation bias: A ubiquitous phenomenon in many guises. *Review of General Psychology*, 2(2), 175-220. <u>https://doi.org/10.1037/1089-2680.2.2.175</u>

Nicolson, R. (2015). *Positive dyslexia*. Rodin Books.

Nicolson, R. I. & Fawcett, A. J. (2005). Developmental dyslexia, learning and the cerebellum. Journal of Neural Transmission. Supplementum, 69, 19-36. https://doi.org/10.1007/3-211-31222-6_2

- Nicolson, R. I., & Fawcett, A. J. (2019). Development of dyslexia: The delayed neural commitment framework. *Frontiers in Behavioral Neuroscience*, 13, 112. https://doi.org/10.3389/fnbeh.2019.00112
- Nightingale, C. (2006). Nothing about me, without me: Involving learners with learning difficulties and disabilities. Learning and Skills Development Agency.
- Noffke, S. E. (1997). Professional, personal, and political dimensions of action research. *Review of Research in Education*, 22(1), 305-343. <u>https://doi.org/10.2307/1167378</u>
- Norton, E., Black, J., Stanley, L., Tanaka, H., Gabrieli, J., Sawyer, C., & Hoeft, F. (2014). Functional neuroanatomical evidence for the double-deficit hypothesis for developmental dyslexia. *Neuropsychologia*, 61, 235-246. <u>https://doi.org/10.1016/j.neuropsychologia.2014.06.015</u>
- Norton, L. S. (2009). Action research in teaching and learning: A practical guide to conducting pedagogical research in universities. Routledge.
- Novita, S. (2016). Secondary symptoms of dyslexia: A comparison of self-esteem and anxiety profiles of children with and without dyslexia. *European Journal of Special Needs Education, 31*(2), 279-288. <u>https://doi.org/10.1080/08856257.2015.1125694</u>
- Nowell, L. S., Norris, J. M., White, D. E. & Moules, N. J. (2017). Thematic analysis: Striving to meet the trustworthiness criteria. *International Journal of Qualitative Methods*, *16*(1), <u>https://doi.org/10.1177/1609406917733847</u>
- Ockelford, A., & Welch, G. (2020). *New approaches in applied musicology: A common framework for music education and psychology research*. Routledge.
- Ockelford, A., & Voyajolu, A. (2020). The development of music-structural cognition in the early years: A perspective from the *Sounds of Intent* model. In A. Ockelford & G. Welch (Eds), *Essays in applied musicology: A common framework for music education and music psychology research* (pp. 13-63). Routledge.
- Ofsted. (2022). The annual report of His Majesty's Chief Inspector of Education, Children's Services and Skills 2021/22. HMSO. Retrieved on February 1, 2024 from https://www.gov.uk/government/publications/ofsted-annual-report-202122-education-childrens-services-and-skills

Oglethorpe, S. (2008). Instrumental music for dyslexics: A teaching handbook (2nd ed.). Wiley.

- Oglethorpe, S. (2008b). Can music lessons help the dyslexic learner? In T. R. Miles, J. Westcombe & D. Ditchfield (Eds.), *Music and dyslexia: A positive approach* (pp. 57-67). Wiley.
- Oglethorpe, S. (2020). *Sheila Oglethorpe obituary*. Retrieved on August 15, 2021 from <u>https://www.theguardian.com/education/2020/aug/09/sheila-oglethorpe-obituary</u>
- Overy, K. (2003). Dyslexia and music: From timing deficits to musical intervention. *Annals of the New York Academy of Sciences, 999,* 497-505. <u>https://doi.org/10.1196/annals.1284.060</u>
- Overy, K. (2008). Classroom games for literacy support. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds.), *Music and dyslexia: A positive approach* (pp. 25-44). Wiley.
- Overy, K. (2008b). Insights from brain imaging. In T. R. Miles, J. Westcombe & D. Ditchfield (Eds), *Music and dyslexia: A positive approach* (pp. 151-161). Wiley.
- Ozernov-Palchik, O., & Gaab, N. (2016). Tackling the 'dyslexia paradox': reading brain and behavior for early markers of developmental dyslexia. *Cognitive Science*, 7(2), 156-176. <u>https://doi.org/10.1002/wcs.1383</u>
- Ozernov-Palchik, O., Wolf, M., & Patel, A. D. (2018). Relationships between early literacy and nonlinguistic rhythmic processes in kindergarteners. *Journal of Experimental Child Psychology*, *167*, 354-368. <u>https://doi.org/10.1016/j.jecp.2017.11.009</u>
- Pajares, M. F. (1992). Teachers' beliefs and educational research: Cleaning up a messy construct. *Review of Educational Research, 62*(3), 307-332. <u>https://doi.org/10.2307/1170741</u>
- Palmer, J. D. (2006). Negotiating the indistinct: Reflections of a Korean adopted American working with Korean born, Korean Americans. *Qualitative Research*, 6(4), 473-495. <u>https://doi.org/10.1177/1468794106068017</u>
- Patel, A. D. (2007). *Music, language and the brain*. Oxford University Press.
- Patel, A. D. (2012). The OPERA hypothesis: assumptions and clarifications. *Annals of the New York Academy of Sciences, 1252*(1), 124-128. <u>https://doi.org/10.1111/j.1749-6632.2011.06426.x</u>
- Patel, A. D., & Iversen, J. R. (2014). The evolutionary neuroscience of musical beat perception: the Action Simulation for Auditory Prediction (ASAP) hypothesis. *Frontiers in Systems Neuroscience*, 8, 57. <u>https://doi.org/10.3389/fnsys.2014.00057</u>

Patston, T., & Waters, L. (2015). Positive instruction in music studios: Introducing a new model for teaching studio music in schools based upon positive psychology. *Psychology of Well-Being* 5(1), 10. <u>https://doi.org/10.1186/s13612-015-0036-9</u>

Patten, M. (2016). *Questionnaire research: A practical guide*. Routledge.

- Paula, V., Vesa, P., Anastasia, G., Anja, T., Laurel J. T., & Teija, K. (2023). Beneficial effects of a music listening intervention on neural speech processing in 0–28-month-old children at risk for dyslexia. *Developmental Science*, 26, e13426. <u>https://doi.org/10.1111/desc.13426</u>
- Pavey, B., Meehan, M., and Davis, S. (2013). The dyslexia-friendly teacher's toolkit. Sage.

Paynter, J. (1982). *Music in the secondary school curriculum*. Cambridge University Press.

- Piascore, Inc. (2022). *Piascore* (Version 6.15.5). [Mobile app]. <u>https://apps.apple.com/us/app/piascore-hd/id406141702</u>
- Perry, C., Zorzi, M., & Ziegler, J. C. (2019). Understanding dyslexia through personalized largescale computational models. *Psychological Science*, *30*(3), 386-395. <u>https://doi.org/10.1177/0956797618823540</u>
- Peter, B., Matsushita, M., & Raskind, W. H. (2011). Global processing speed in children with low reading ability and in children and adults with typical reading ability: Exploratory factor analytic models. *Journal of Speech, Language, and Hearing Research, 54*(3), 885-899. <u>https://doi.org/10.1044/1092-4388(2010/10-0135</u>
- Peterson, C. P., Maier, S. F. M., & Seligman, M. E. P. (1993). *Learned helplessness: A theory for the age of personal control*. Oxford University Press.
- Peterson, R. L., & Pennington, B. F. (2012). Developmental dyslexia. *Lancet, 379*(9830), 1997-2007. <u>https://doi.org/10.1016/S0140-6736(12)60198-6</u>
- Phillips, L., & Cain, M. (2020). 'Exhausted beyond measure': What teachers are saying about COVID-19 and the disruption to education. *The Conversation*. Retrieved on February 1, 2024 from <u>https://theconversation.com/exhausted-beyond-measure-what-teachers-are-sayingabout-covid-19-and-the-disruption-toeducation-01</u>
- Phillips, S., & Kelly, K. (2016). *Teaching literacy to learners with dyslexia: A multi-sensory approach*. Sage.

Polanyi, M. (1966). The tacit dimension. Routledge & Kegan Paul.

- Polischuk, D. K. (2018). *Transformational piano teaching: Mentoring students from all walks of life*. Oxford University Press.
- Politimou, N., Dalla Bella, S., Farrugia, S., N., & Franco, F. (2019). Born to speak and sing: Musical predictors of language development in pre-schoolers. *Frontiers in Psychology*, 10, 948. <u>https://doi.org/10.3389/fpsyg.2019.00948</u>
- Poole, J. (2003). Dyslexia: A wider view. The contribution of an ecological paradigm to current issues. *Educational Research*, 45(2), 167-180. <u>https://doi.org/10.1080/001388032000103253</u>
- Protopapas A. (2013). From temporal processing to developmental language disorders: mind the gap. Philosophical transactions of the Royal Society of London. Series B, *Biological sciences*, 369(1634), 20130090. <u>https://doi.org/10.1098/rstb.2013.0090</u>
- Przekoracka-Krawczyk, A., Brenk-Krakowska, A., Nawrot, P., Rusiak, P., & Naskręcki, R. (2017).
 Unstable binocular fixation affects reaction times but not implicit motor learning in dyslexia.
 Investigative Ophthalmology & Visual Science, 58(14), 6470-6480.
 <u>https://doi.org/10.1167/iovs.16-21305</u>
- Przybylski, L., Bedoin, N., Krifi-Papoz, S., Herbillon, V., Roch, D., Léculier, L., ... & Tillmann, B. (2013). Rhythmic auditory stimulation influences syntactic processing in children with developmental language disorders. *Neuropsychology*, *27*(1), 121.
 https://doi.org/doi:10.1037/a0031277
- Rappolt-Schlichtmann, G., Boucher, A. R., & Evans, M. (2018). From deficit remediation to capacity building: Learning to enable rather than disable students with dyslexia. *Language, Speech, and Hearing Services in Schools, 49*(4), 864-874.
 https://doi.org/10.1044/2018 LSHSS-DYSLC-18-0031
- Ramus, F., Rosen, S., Dakin, S. C., Day, B. L., Castellote, J. M., White, S., & Frith, U. (2003). Theories of developmental dyslexia: Insights from a multiple case study of dyslexic adults. *Brain*, *126*(4), 841-865. <u>https://doi.org/10.1093/brain/awg076</u>

Reid, G. (2004). Dyslexia: A complete guide for parents. Wiley

Reid, G. (2016). Dyslexia: A practitioner's handbook. Wiley.

- Reid, G., & Came, F. (2009). Identifying and overcoming the barriers to learning in an inclusive context. In G. Reid, D. Knight & J. Wearmouth (Eds.), *The Routledge companion to dyslexia* (pp. 193-202). Routledge.
- Reifinger, J. L. (2019). Dyslexia in the music classroom: A review of literature. Update: Applications of Research in Music Education, 38(1), 9-17. <u>https://doi.org/10.1177/8755123319831736</u>
- Register, D., Darrow, A. A., Swedberg, O. & Standley, J. (2007). The use of music to enhance reading skills of second grade students and students with reading disabilities. *Journal of Music Therapy*, 44(1), 23-37. https://doi.org/10.1093/jmt/44.1.23
- Rello, L., & Baeza-Yates, R. (2017). How to present more readable text for people with dyslexia. Universal Access in the Information Society, 16(1), 29-49. <u>https://doi.org/10.1007/s10209-015-0438-8</u>
- Reynolds, A. M., Renzoni, K. B., Turowski, P. L., & Waters, H. D. (2014). "Pssst... over here!": Young children shaping the future of music education. In C. Randles (Ed.) *Music education: Navigating the future* (pp. 201-214). Routledge. <u>https://doi.org/10.4324/9781315777009</u>
- Riddick, B. (1996). Living with dyslexia: The social and emotional consequences of specific learning difficulties. Routledge.
- Riddick, B. (2009). *Living with dyslexia: The social and emotional consequences of specific learning difficulties/disabilities* (2nd ed.). Routledge. <u>https://doi.org/10.4324/9780203864579</u>
- Riddick, B., Farmer, M., Sterling, C. (1997). *Students and dyslexia: Growing up with a specific learning difficulty*. Whurr.
- Riddick, B. (2012). *Living with dyslexia: The social and emotional consequences of specific learning difficulties/disabilities* (2nd ed.). Routledge. <u>https://doi.org/10.4324/9780203864579</u>
- Ridsdale, J. (2004). Dyslexia and self-esteem. In M. Turner & J. Rack (Eds.), *The study of dyslexia* (pp. 249-279). Kluwer Academic/Plenum Publishers.
- Robson, C. (2011). *Real world research: A resource for users of social research methods in applied settings* (3rd ed.). Wiley-Blackwell.

- Rogers, G. L. (1991). Effect of color-coded notation on music achievement of elementary instrumental students. *Journal of Research in Music Education, 39*, 64-73.
- Rolka, E., & Silverman, M. (2015) A systematic review of music and dyslexia. *The Arts in Psychotherapy 46*, 24–32. <u>https://doi.org/10.1016/j.aip.2015.09.002</u>
- Rooke, M. (2015). *Creative, successful, dyslexic: Twenty-three high achievers share their stories.* Jessica Kingsley Publishers.
- Rose, J. (2009). *Identifying and teaching children and young people with dyslexia and literacy difficulties*. Department for Children, Schools and Families.
- Rose, C., & Nicholl, M. J. (1997). Accelerated learning for the 21st Century: The six-step plan to unlock your master-mind. Dell.
- Rosefield, M. (2023). Lost the 'Bounce in your bungee': A grounded theory study into compassion fatigue and burnout in the coaching profession. *International Journal of Evidence Based Coaching and Mentoring, Special Issue 17*. <u>https://doi.org/10.24384/p2wj-7931</u>
- Ross, H. (2019). Supporting a child with dyslexia: How parents/carers engage with school-based support for their children. *British Journal of Special Education*, 46(2), 136-156. <u>https://doi.org/10.1111/1467-8578.12254</u>
- Rostvall, A., & West, T. (2003) Analysis of interaction and learning in instrumental teaching. *Music Education Research*, 5(3), 213-226. <u>https://doi.org/10.1080/1461380032000126319</u>
- Rotjan, M. (2021). Deciding for or deciding with: Student involvement in repertoire selection. *Music Educators Journal, 107*(4), 28-34. <u>https://doi.org/10.1177/00274321211013879</u>
- Rowan, L. (2010). Learning with dyslexia in secondary school in New Zealand: What can we learn from students' past experiences? *Australian Journal of Learning Difficulties, 15*(1), 71-79. https://doi.org/10.1080/19404150903524556
- Rudd, T., Colligan, F., & Naik, R. (2006). *Learner voice*. Futurelab.
- Rutter, M., Caspi, A., Fergusson, D., Horwood, L. J., Goodman, R., Maughan, B., Moffitt, T. E., Meltzer, H., & Carroll, J. (2004). Sex differences in developmental reading disability: new findings from 4 epidemiological studies. *Journal of the American Medical Association*, 291(16), 2007-2012. <u>https://doi.org/10.1001/jama.291.16.2007</u>

Saksida, A., Iannuzzi, S., Bogliotti, C., Chaix, Y., Démonet, J. F., Bricout, L., Billrd, C., Nguyen-Morel, M. A., Le Heuzey, M. F., Soares-Boucaud, I., George, F., Ziegler, J. C., & Ramus, F. (2016).
Phonological skills, visual attention span, and visual stress in developmental dyslexia.
Developmental Psychology, 52(10), 1503-1516. <u>https://doi.org/10.1037/dev0000184</u>

Salkind, N. J. (2010). Encyclopedia of research design. SAGE.

- Salmon, J. (1992). What Brubeck got from Milhaud. American Music Teacher, 41(4), 26-76.
- Sanderson-Mann, J., & McCandless, F. (2006). Understanding dyslexia and nurse education in the clinical setting. *Nurse Education in Practice*, 6(3), 127-133. <u>https://doi.org/10.1016/j.nepr.2005.10.004</u>
- SASC (SpLD Assessment Standards Committee). (2018). Specific Learning Difficulties (SpLDs) and Visual difficulties: A guide for assessors and SpLD practitioners. Retrieved on August 20, 2021 from
 https://www.sasc.org.uk/SASCDocuments/Visual_Difficulties_guidance_for_SpLD_practitio

ners Final June2018.pdf

SASC (SpLD Assessment Standards Committee). (2019). Notes: Visual Difficulties Screening Protocol V2 June 2019 [V1 in Appendix 2 SpLDs and Visual Difficulties: A guide for assessors and SpLD practitioners. Dr Jim Gilchrist, Caroline Holden, Jane Warren 2018]. Retrieved on August 30, 2021 from <u>https://sasc.org.uk/SASCDocuments/Notes-</u> Visual%20Difficulties%20Screening%20Protocol%20V2%20Jun19.pdf

Saunders, K., & Eastap, L. (2018). *The dyslexia handbook 2018*. The British Dyslexia Association.

- Savage, J. (2021). The policy and practice of music education in England, 2010-2020. *British Educational Research Journal*, 47(2), 469-483. <u>https://doi.org/10.1002/berj.3672</u>
- Shaywitz, S. E. (2003). Overcoming dyslexia: A new and complete science-based program for reading problems at any level. Alfred A. Knopf.
- Schiavio, A., Küssner, M. B., and Williamon, A. (2020). Music Teachers' perspectives and experiences of ensemble and learning skills. *Frontiers of Psychology* 11, 291. <u>https://doi.org/10.3389/fpsyg.2020.00291</u>

- Schlaug, G., Norton, A., Overy, K., & Winner, E. (2005). Effects of music training on the child's brain and cognitive development. *Annals of the New York Academy of Sciences*, 1060, 219-230. <u>https://doi.org/10.1196/annals.1360.015</u>
- Schneps, M. H., Brockmole, J. R., Sonnert, G., & Pomplun, M. (2012). History of reading struggles linked to enhanced learning in low spatial frequency scenes. *PLOS ONE*, 7(4), e35724. https://doi.org/10.1371/journal.pone.0035724
- Schneps, M. H., Thomson, J. M., Sonnert, G., Pomplun, M., Chen C., & Heffner-Wong, A. (2013).
 Shorter lines facilitate reading in those who struggle. *Public Library of Science One 8*(8):
 e71161. <u>https://doi.org/10.1371/journal.pone.0071161</u>
- Schön, D. A. (1992). *The reflective practitioner: How professionals think in action* (1st ed.). Routledge. <u>https://doi.org/10.4324/9781315237473</u>
- Schön, D., Gordon, R., Campagne, A., Magne, C., Astésano, C., Anton, J. L., & Besson, M. (2010).
 Similar cerebral networks in language, music and song perception. *NeuroImage*, *51*(1), 450-461. <u>https://doi.org/10.1016/j.neuroimage.2010.02.023</u>

Schreiber, L. M., & Valle, B. E. (2013). Social constructivist teaching strategies in the small group classroom. Small Group Research, 44(4), 395–411. <u>https://doi.org/10.1177/1046496413488422</u>

- Schwabe, L., & Wolf, O. T. (2013). Stress and multiple memory systems: From 'thinking' to 'doing'. *Trends in Cognitive Sciences*, *17*(2), 60-68. <u>https://doi.org/10.1016/j.tics.2012.12.001</u>
- Seligman, M. E. P. (2011). Flourish: A visionary new understanding of happiness and well-being. Free Press.
- Serrallach, B., Groß, C., Bernhofs, V., Engelmann, D., Benner, J., Gündert, N., Blatow, M.,
 Wengenroth, M., Seitz, A., Brunner, M., Seither, S., Parncutt, R., Schneider, P., & SeitherPreisler, A. (2016). Neural biomarkers for dyslexia, ADHD, and ADD in the auditory cortex of
 children. *Frontiers in Neuroscience*, *10*, 324. <u>https://doi.org/10.3389/fnins.2016.00324</u>
- Shaywitz, S. E., & Shaywitz, B. A. (2005). Dyslexia (specific reading disability). *Biological Psychiatry*, 57(11), 1301-1309. <u>https://doi.org/10.1016/j.biopsych.2005.01.043</u>

- Shaywitz, S. E. and Shaywitz, B. A. (2016). *Dyslexia: Profiles of success*. The Yale Center for Dyslexia and Creativity.
- Singer, E. (2005). The strategies adopted by Dutch children with dyslexia to maintain their selfesteem when teased at school. *Journal of Learning Disabilities*, *38*(5), 411-423. <u>https://doi.org/10.1177/00222194050380050401</u>
- Singer, J. A., Blagov, P., Berry, M., & Oost, K. M. (2013). Self-defining memories, scripts, and the life story: Narrative identity in personality and psychotherapy. *Journal of Personality*, 81(6), 569-582. <u>https://doi.org/10.1111/jopy.12005</u>
- SmileyApps LLC. (2020). *Piano Tutor* (Version 7.5.1). [Mobile app]. <u>https://apps.apple.com/us/app/piano-tutor-app/id364898961</u>
- Smith, J. E. (1969). Time, times, and the 'right time'; Chronos and kairos. *The Monist, 53*(1), 1-13. https://doi.org/10.5840/monist196953115
- Smith-Spark, J. H., Henry, L. A., Messer, D. J., Edvardsdottir, E., & Ziecik, A. P. (2016). Executive functions in adults with developmental dyslexia. *Research in Developmental Disabilities*, 53-54, 323-341. <u>https://doi.org/10.1016/j.ridd.2016.03.001</u>
- Snowling, M. J. (2013). Early identification and interventions for dyslexia: A contemporary view. Journal of Research in Special Educational Needs, 13(1), 7-14. <u>https://doi.org/10.1111/j.1471-3802.2012.01262.x</u>
- Snowling, M. J., Gallagher, A., & Frith, U. (2003). Family risk of dyslexia is continuous: Individual differences in the precursors of reading skill. *Child Development*, 74(2), 358–373. <u>https://doi.org/10.1111/1467-8624.7402003</u>
- Snowling, M. J., & Melby-Lervåg, M. (2016). Oral language deficits in familial dyslexia: A metaanalysis and review. *Psychological Bulletin*, 142(5), 498-545. <u>https://doi.org/10.1037/bul0000037</u>
- Snowling, M. J., Muter, V., & Carroll, J. (2007). Children at family risk of dyslexia: A follow-up in early adolescence. *Journal of Child Psychology and Psychiatry 48*, 609-618. <u>https://doi.org/10.1111/j.1469-7610.2006.01725.x</u>

- Snowling, M., & Hulme, C. (2024). Do we really need a new definition of dyslexia? A commentary. *Annals of Dyslexia*, 74(3), 355-362. <u>https://doi.org/10.1007/s11881-024-00305-y</u>
- Snowling, M. J., Hulme, C., & Nation, K. (2020). Defining and understanding dyslexia: Past, present and future. Oxford Review of Education, 46(4), 501-513. https://doi.org/10.1080/03054985.2020.1765756

Snowling, M. J. (2000). Dyslexia. Blackwell.

- Somekh, B. (2006). *Action research: A methodology for change and development*. Open University Press.
- Southcott, J., & de Bruin, L. R. (2022). Being and becoming instrumental musicians and teachers: A post-qualitative exploration. *Frontiers in Education*, *7*. <u>https://doi.org/10.3389/feduc.2022.974184</u>
- Sowa, P. A. (2009). Understanding our learners and developing reflective practice: Conducting action research with English language learners. *Teaching and Teacher Education: An International Journal of Research and Studies, 25*(8), 1026-1032. https://doi.org/10.1016/j.tate.2009.04.008
- Stampoltzis, A., Antonopoulou, E., Zenakou, E. & Kouvava, S. (2010). Learning sensory modalities and educational characteristics of Greek dyslexic and non-dyslexic university students. *Electronic Journal of Research in Educational Psychology, 8*, 561-580. <u>https://doi.org/10.25115/ejrep.v8i21.1383</u>
- Stein, J. (2018). What is developmental dyslexia? *Brain Sciences*, 8(2), 26. <u>https://doi.org/10.3390/brainsci8020026</u>
- Stein J. (2022). The visual basis of reading and reading difficulties. *Frontiers in Neuroscience, 16*, 1004027. <u>https://doi.org/10.3389/fnins.2022.1004027</u>
- Stein J. (2023). Theories about developmental dyslexia. *Brain Sciences*, *13*(2), 208. <u>https://doi.org/10.3390/brainsci13020208</u>
- Steinbrink, C., Knigge, J., Mannhaupt, G., Sallat, S., & Werkle, A. (2019). Are temporal and tonal musical skills related to phonological awareness and literacy skills? Evidence from two
cross-sectional studies with children from different age groups. *Frontiers in Psychology*, *10*, 805. <u>https://doi.org/10.3389/fpsyg.2019.00805</u>

- Stenberg, A., & Cross, I. (2019). White spaces, music notation and the facilitation of sight-reading. Scientific Reports, 9(1), 5299. <u>https://doi.org/10.1038/s41598-019-41445-1</u>
- Stenhouse, L. (1979). Case study in comparative education: Particularity and generalisation. *Comparative Education*, 15(1), 5-10. <u>https://doi.org/10.1080/0305006790150102</u>

Stenhouse, L. (1984). An introduction to curriculum research and development. Heinemann.

- Stewart, L. (2008). Music reading: A cognitive neuroscience approach. In T. R. Miles, J.
 Westcombe & D. Ditchfield (Eds.), *Music and dyslexia: A positive approach* (pp. 162-170).
 Wiley.
- Stewart, D.W., Shamdasani, P.N. & Rook, D.W. (2007). *Focus groups: Theory and practice*. (2nd ed.). Sage.
- Stoodley, C. J., & Stein, J. F. (2013). Cerebellar function in developmental dyslexia. *Cerebellum 12*, 267–276. <u>https://doi.org/10.1007/s12311-012-0407-1</u>
- Strait, D. L., Hornickel, J. & Kraus, N. (2011). Subcortical processing of speech regularities underlies reading and music aptitude in children. *Behavioural Brain Function*, 7, 44. <u>https://doi.org/10.1186/1744-9081-7-44</u>
- Sturm, V., Roy, A.R.K, Datta, S., Wang, C., Sible, I., Holley, S., & Gorno-Tempini, M. L. (2021). Enhanced visceromotor emotional reactivity in dyslexia and its relation to salience network connectivity. *Cortex*, 134, 278-295. <u>https://doi.org/10.1016/j.cortex.2020.10.022</u>

Sudo, P. (1997). Zen guitar. Simon & Schuster.

- Suttle, C. M., Lawrenson, J. G., & Conway, M. L. (2018). Efficacy of coloured overlays and lenses for treating reading difficulty: An overview of systematic reviews. *Clinical & experimental Optometry*, 101(4), 514-520. <u>https://doi.org/10.1111/cxo.12676</u>
- Swanwick, P. K., & Swanwick, K. (1994). Musical knowledge: Intuition, analysis and music education. Routledge. <u>https://doi.org/10.4324/9780203424575</u>

- Swanwick, K. (2012). Understanding music as the philosophical focus of music education. In W.
 Bowman & A. L. Frega (Eds.), *The Oxford handbook of philosophy in music education*. (pp. 328-345). Oxford University Press.
- Tafti, M. A., Hameedy, M. A., & Baghal, N. M. (2009). Dyslexia, a deficit or a difference: Comparing the creativity and memory skills of dyslexic and nondyslexic students in Iran. Social Behavior and Personality: An International Journal, 37(8), 1009-1016. <u>https://doi.org/10.2224/sbp.2009.37.8.1009</u>
- Tallal, P., & Gaab, N. (2006). Dynamic auditory processing, musical experience and language development. *Trends in Neurosciences*, 29(7), 382-390. <u>https://doi.org/10.1016/j.tins.2006.06.003</u>
- Tallal, P. (2012). Improving neural response to sound improves reading. Proceedings of the National Academy of Sciences, 109(41), 16406-16407. <u>https://doi.org/10.1073/pnas.121412210</u>
- Tallentyre, S. G. (ed.). Voltaire in his letters. New York: G. P. Putnam's Sons, 1919. p. 232. French:
 Au prince royal de prusse, le 28 novembre, in M. Palissot (ed.), *Oeuvres de Voltaire: Lettres Choisies du Roi de Prusse et de M. de Voltaire*, Tome II. Paris: Chez Baudoiun, 1802, p. 419.
- Tanner, K. (2009). Adult dyslexia and the 'conundrum of failure'. *Disability and Society, 24*(6), 785-797. <u>https://doi.org/10.1080/09687590903160274</u>
- Te Rietmolen, N., Mercier, M. R., Trébuchon, A., Morillon, B., & Schön, D. (2024). Speech and music recruit frequency-specific distributed and overlapping cortical networks. *ELife*, 13. <u>https://doi.org/10.7554/eLife.94509.3</u>
- Tekin, A. K., & Kotaman, H. (2013). The epistemological perspectives on action research. *Journal of Educational and Social Research*, *3*(1), 81-91. <u>https://doi.org/10.5901/jesr.2013.v3n1p81</u>
- Thomas, D. R. (2006). A general inductive approach for analyzing qualitative evaluation data. *American Journal of Evaluation*, 27(2), 237-246. <u>https://doi.org/10.1177/1098214005283748</u>
- Thomas, I. (2009). Critical thinking, transformative learning, sustainable education, and problembased learning in universities. *Journal of Transformative Education*, 7(3), 245-264. <u>https://doi.org/10.1177/1541344610385753</u>

- Thomson, J. M., & Goswami, U. (2008). Rhythmic processing in children with developmental dyslexia: Auditory and motor rhythms link to reading and spelling. *Journal of Physiology*, *102*(1-3), 120-129. <u>https://doi.org/10.1016/j.jphysparis.2008.03.007</u>
- Tierney, A., & Kraus, N. (2013). Music training for the development of reading skills. *Progress in Brain Research*, 207, 209-241. <u>https://doi.org/10.1016/B978-0-444-63327-9.00008-4</u>
- Tierney, A., & Kraus, N. (2014). Auditory-motor entrainment and phonological skills: precise auditory timing hypothesis (PATH). *Frontiers in Human Neuroscience*, 8, 949. <u>https://doi.org/10.3389/fnhum.2014.00949</u>
- Tillmann, B. (2012). Music and language perception: expectations, structural integration, and cognitive sequencing. *Topics in Cognitive Science*, 4(4), 568-584. <u>https://doi.org/10.1111/j.1756-8765.2012.01209</u>.x
- Townend, J., & Turner, M. (2000). *Dyslexia in practice. A guide for teachers*. Dordrecht.
- Townsend, A. (2010). *Action research: The challenges of understanding and changing practice*. McGraw-Hill.
- Trimble, M., Hesdorffer D. (2017). Music and the brain: The neuroscience of music and musical appreciation. *British Journal of Psychology International,* 14(2), 28-31. http://doi.org/10.1192/S2056474000001720
- Tufford, L., & Newman, P. (2012). Bracketing in qualitative research. *Qualitative Social Work: Research and Practice, 11*(1), 80-96. <u>https://doi.org/10.1177/1473325010368316</u>
- Turker, S., & Reiterer, S. M. (2021). Brain, musicality, and language aptitude: A complex interplay. Annual Review of Applied Linguistics, 41, 95-107. <u>https://doi.org/10.1017/S0267190520000148</u>
- Tutteo. (2022). *Flat.io* (Version 2022). [Web and cloud-based app]. <u>https://flat.io/en-GB/product-overview</u>
- Upitis, R., Abrami, P. C., Brook, J., & King, M. (2017). Parental involvement in children's independent music lessons. *Music Education Research*, 19(1), 74-98. <u>https://doi.org/10.1080/14613808.2016.1202220</u>

- Valdois, S., Roulin, J. L., & Line Bosse, M. (2019). Visual attention modulates reading acquisition. Vision Research, 165, 152-162. <u>https://doi.org/10.1016/j.visres.2019.10.011</u>
- Vance, K. O. B. (2004). Adapting music instruction for students with dyslexia. *Music Educators Journal*, *90*(5), 27-31. <u>https://doi.org/10.2307/3400020</u>
- van Viersen, S., de Bree, E. H., Kroesbergen, E. H., Slot, E. M., & de Jong, P. F. (2015). Risk and protective factors in gifted children with dyslexia. *Annals of Dyslexia*, 65(3), 178-198. <u>https://doi.org/10.1007/s11881-015-0106-y</u>
- Van Weelden, K. & Whipple J. (2005). The effects of field experience on music education majors' perceptions of music instruction for secondary students with special needs. *Journal of Music Teacher Education*, 14(2), 62-68. <u>https://doi.org/10.1177/10570837050140020109</u>
- Vaughn, S., Miciak, J., Clemens, N., & Fletcher, J. M. (2024). The critical role of instructional response in defining and identifying students with dyslexia: A case for updating existing definitions. *Annals of Dyslexia* 74, 325-336. https://doi.org/10.1007/s11881-024-00303-0
- Vellutino, F. R., Fletcher, J. M., Snowling, M. J., & Scanlon, D. M. (2004). Specific reading disability (dyslexia): what have we learned in the past four decades? *Journal of Child Psychology and Psychiatry, and Allied Disciplines, 45*(1), 2-40. <u>https://doi.org/10.1046/j.0021-9630.2003.00305.x</u>
- Vonthron, F., Yuen, A., Pellerin, H., Cohen, D., & Grossard, C. (2024). A serious game to train rhythmic abilities in children with dyslexia: Feasibility and usability study. *JMIR Serious Games*, 12, e42733. <u>https://doi.org/10.2196/42733</u>
- Villacañas de Castro, L. S., & Banegas, D. L. (2020). Philosophical tenets of action research in education. Oxford Research Encyclopedias: Education Oxford University Press. <u>https://doi.org/10.1093/acrefore/9780190264093.013.1429</u>
- Vukovic, R. K., & Siegel, L. S. (2006). The double-deficit hypothesis: A comprehensive analysis of evidence. *Journal of Learning Disabilities, 39,* 25-47. <u>http://dx.doi.org/10.1177/00222194060390010401</u>
- Vygotsky, L. S. (1978). *Mind in society: Development of higher psychological processes*. Harvard University Press.

- Wagner, R. K., Zirps, F. A., Edwards, A. A., Wood, S. G., Joyner, R. E., Becker, B. J., Liu, G., & Beal,
 B. (2020). The prevalence of dyslexia: A new approach to its estimation. Journal of Learning Disabilities, 53(5), 354-365. <u>https://doi.org/10.1177/0022219420920377</u>
- Walker, L. 2008. Learner engagement. A review of learner voice initiatives across the UK's education sectors. Futurelab.
- Ward, A. F., Duke, K., Gneezy, A., & Bos, M. W. (2017). Brain drain: The mere presence of one's own smartphone reduces available cognitive capacity. *Journal of the Association for Consumer Research*, 2(2), 140-154. <u>https://doi.org/10.1086/691462</u>
- Washburn, E. K., Binks-Cantrell, E. S., & Joshi, R. M. (2014). What do preservice teachers from the USA and the UK know about dyslexia?. *Dyslexia*, *20*(1), 1-18. <u>https://doi.org/10.1002/dys.1459</u>
- Watts, M. (2021). Friedrich Froebel: Interpolation, extrapolation. *Early Child Development and Care*, *191*(7–8), 1186-1195. <u>https://doi.org/10.1080/03004430.2021.1881077</u>
- WCAG 2.0 (Web content accessibility guidelines). (2024). What are the Web content accessibility guidelines (WCAG)? Retrieved on April 1, 2024 from <u>https://www.w3.org/WAI/WCAG22/quickref/</u>
- Weiss, A. H., Granot, R. Y., & Ahissar, M. (2014). The enigma of dyslexic musicians. *Neuropsychologia*, *54*, 28–40. <u>https://doi.org/10.1016/j.neuropsychologia.2013.12.009</u>
- Welch, G., Ockelford, A., Carter, F. C., Zimmermann, S. A., & Himonides, E. (2009). "Sounds of Intent": Mapping musical behaviour and development in children and young people with complex needs. *Psychology of Music*, *37*(3), 348-370. https://doi.org/10.1177/0305735608099688
- West, T.G. (2009). In the mind's eye creative visual thinkers gifted dyslexics and the rise of visual technologies. Prometheus Books.
- Westby, C. (2019). The myth of learning styles. *Word of Mouth, 31*(2), 4-7. <u>https://doi.org/10.1177/1048395019879966a</u>

- Whitehead, J. (1989). Creating a living educational theory from questions of the kind, 'How do I improve my practice? *Cambridge Journal of Education*, 19(1), 41-52. https://doi.org/10.1080/0305764890190106
- Whitehead, J. & McNiff, J. (2006). Action research. SAGE Publications, Ltd.
- WHO (World Health Organisation). (1993). *The ICD–10 classification of mental and behavioural disorders: Diagnostic criteria for research*. WHO.
- Williamon, A. (Ed.). (2012). *Musical excellence: Strategies and techniques to enhance performance*. Oxford University Press.
- Wilkins, A. J. (1995). Visual stress. Oxford University Press.
- Wilmot, A., Pizzey, H., Leitão, S., Hasking, P., & Boyes, M. (2023). Growing up with dyslexia: Child and parent perspectives on school struggles, self-esteem, and mental health. *Dyslexia*, 29(1), 40-54. <u>https://doi.org/10.1002/dys.1729</u>
- Wolf, M., & Bowers, P. G. (1999). The double-deficit hypothesis for the developmental dyslexias. Journal of Educational Psychology, 91(3), 415–438. <u>https://doi.org/10.1037/0022-0663.91.3.415</u>
- Wolf, M., Gotlieb, R.J.M., Kim, S.A. et al. (2024). Towards a dynamic, comprehensive conceptualization of dyslexia. *Annals of Dyslexia 74*, 303-324. <u>https://doi.org/10.1007/s11881-023-00297-1</u>
- Wolff, U., & Lundberg, I. (2002). The prevalence of dyslexia among art students. *Dyslexia*, 8(1), 3442. <u>https://doi.org/10.1002/dys.211</u>
- Wolff, P.H. (2002). Timing precision and rhythm in developmental dyslexia. *Reading and Writing 15*, 179-206. <u>https://doi.org/10.1023/A:1013880723925</u>
- Wood, E. (2010). Developing integrated pedagogical approaches to play and learning. In P.
 Broadhead, J. Howard & E. Wood (Eds.), *Play and learning in the early years: From research to practice* (pp. 9-26). Sage.
- Winter, R. (1989). *Learning from experience: Principles and practice in action-research*. The Falmer Press.

- Winter, R., Griffiths, M., & Green, K. (2000). The "academic" qualities of practice: What are the criteria for a practice-based PhD? *Studies in Higher Education*, 25(1), 25-30. <u>https://doi.org/10.1080/030750700115993</u>
- Wristen, B. (2005). Cognition and motor execution in piano sight-reading: A review of literature. Update: Applications of Research in Music Education, 24(1), 44-56. https://doi.org/10.1177/87551233050240010106
- Yu, M., Xu, M., Li, X., Chen, Z., Song, Y., & Liu, J. (2017). The shared neural basis of music and language. *Neuroscience*, 357, 208–219. <u>https://doi.org/10.1016/j.neuroscience.2017.06.003</u>
- Zeserson, K., Welch, G., Burn, S., Saunders, J. & Himonides, E. (2014). *Inspiring music for all: Next* steps in improvement, integration and innovation. Retrieved on February 1, 2024 from <u>https://discovery.ucl.ac.uk/id/eprint/1476037/</u>
- Ziegler, J. C., & Goswami, U. (2005). Reading acquisition, developmental dyslexia, and skilled reading across languages: A psycholinguistic grain size theory. *Psychological Bulletin*, 131(1), <u>3–29. https://doi.org/10.1037/0033-2909.131.1.3</u>
- Ziegler, J. C., Pech-Georgel, C., George, F., & Foxton, J. M. (2012). Global and local pitch perception in children with developmental dyslexia. *Brain and Language*, 120(3), 265–270. <u>https://doi.org/10.1016/j.bandl.2011.12.002</u>
- Zuk, J., Bishop-Liebler, P., Ozernov-Palchik, O., Moore, E., Overy, K., Welch, G., & Gaab, N. (2017).
 Revisiting the "enigma" of musicians with dyslexia: Auditory sequencing and speech abilities. *Journal of Experimental Psychology. General*, 146(4), 495-511.
 https://doi.org/10.1037/xge0000281
- Zuk, J., Perdue, M. V., Becker, B., Yu, X., Chang, M., Raschle, N. M., & Gaab, N. (2018). Neural correlates of phonological processing: Disrupted in children with dyslexia and enhanced in musically trained children. *Developmental Cognitive Neuroscience*, 34, 82-91. <u>https://doi.org/10.1016/j.dcn.2018.07.001</u>

List of appendices

Appendix A: Information and consent form questionnaire

Appendix B: Information and consent forms for parents/guardians of minor participants – piano lessons/ interviews

Appendix C: Information sheet and consent form participants piano lessons/interviews

Appendix D: Focus group participant Information sheet and consent form

Appendix E: Questionnaire questions: Dyslexic students' perceptions of music exams

Appendix F: Interview schedule for teacher interviews

Appendix G: Interview schedule for parents

Appendix H: Interview schedule for students

Appendix I: Sample lesson plan

Appendix J: Accessible text-only version of UDL Checkpoints

Appendix K: Sample reflective journal page

Appendix L: Piano teaching health and safety risk assessment during Covid-19

Appendix M: Student age-appropriate information sheet

Appendix N: Focus group brief

Appendix O: Workshop participant feedback

Appendix P: International Dyslexia Association Knowledge and Practice Standards for Teachers of Reading

Appendix A

Questionnaire: Dyslexic students' perceptions of music exams

Information and consent form

Start of Block: Default Question Block

Q1

Information Sheet

Introduction

I am a PhD candidate from the Department of Music at the University of York who is interested in researching music learning and dyslexia. My goals are to better understand how dyslexic students learn, to improve my piano teaching for dyslexic students as well as to provide resources which other teachers and institutions may find beneficial. I invite you to participate in this research as I believe it may offer a valuable contribution to the knowledge of dyslexia and music.

Please be assured that you are welcome to discuss this with me and to ask for more information and contacts in the field of music and dyslexia which could be provided on request. After you have read through this information and if you agree, then please give your consent in the first question of the survey.

If you have any questions at any time through this process, please contact me at

Purpose

Through my work as a piano teacher, I came to realise that there was much to learn about providing better lessons for dyslexic students. This has motivated my research. In addition to documenting instrumental lessons, I am interested in how dyslexic students perceive the music exam environment and accommodations that are offered by music exam boards.

Participation

You have been invited to take part in this research as data you provide may help me to understand how to improve exam situations for dyslexic students in the future. You do not have to agree to participate. The decision may be one that you want to discuss with someone else. You can ask as many questions as you like and take your time to decide.

Procedures

You will receive the survey through an email link. Once you have agreed your consent to participate, you will answer a series of questions.

Duration

The survey should not take more than 15-20 minutes to complete.

Risks and Discomforts

Some of the questions may pertain to sensitive and personal topics. For instance, it may remind you of an unpleasant experience. You do not have to answer any questions with which you do not feel comfortable. You do not have to provide an answer as to why you refuse.

Benefits

Understanding students' perceptions of music exam accommodations will help me to have a more profound understanding of whether or not students' feel they are fair or effective. This may lead to recommendations which may improve accommodations offered or information provided about the accommodations.

Confidentiality

The survey will be conducted in a way that participants remain anonymous.

Sharing of Research Findings

Research findings may be shared in a timely fashion in journals or conferences but with complete confidentiality maintained. It is my hope that these findings will be applicable and relevant to other interested people who may wish to learn from my research.

Right to refuse or withdraw

You have every right to refuse or withdraw from the research without the need to offer an explanation for the refusal. Your concerns and wishes will be taken very seriously.

Who to Contact

If you have any questions, you may ask them now or later, even after the research has started. This research has been approved by the University Ethics Committee, a group whose task it is to make sure that research participants are protected from harm. For further information, you may contact:

Kristl Kirk

PhD Candidate Department of Music University of York

Or my supervisor:

Dr. Liz Haddon Department of Music University of York

Q2 Please click to indicate that you consent to the following:

I have read the information sheet and understand the content. I understand that as guardian to a minor child I may assist them in completing this survey. (5)

I understand that the purpose of this survey is to better understand dyslexic students' perceptions of music exam access arrangements and reasonable adjustments. (8)

I understand that my responses are anonymous and will only be used for the purposes of this research. (10)

I consent voluntarily to participate in this research study which I understand involves completing a survey. (3)

Skip To: End of Survey If Please click to indicate that you consent to the following: != I have read the information sheet and understand the content. I understand that as guardian to a minor child I may assist them in completing this survey.

Appendix B

Information and consent forms for parents/guardians of minor participants

Kristl Kirk

PhD candidate

Department of Music, University of York

Heslington, York YO10 5DD

Information Sheet and Consent Form for Parents/Guardians of Minor Participants

This information/consent form is for parents of minors who have been invited to participate in the research entitled 'Decoding Music Education with Dyslexic Students: An action research project examining student, teacher and parental interactions in the context of instrumental lessons'.

This Informed Consent Form has two parts:

- Information Sheet (to share information about the study with you)
- Certificate of Consent (for signatures if you agree that your child may participate)

You will be given a copy of the full Informed Consent Form.

Part 1: Information Sheet

Introduction

I am a PhD candidate from the Department of Music at the University of York who is interested in researching music learning and dyslexia. My goals are to better understand how dyslexic students learn, to improve my piano teaching for dyslexic students as well as to provide resources which other teachers and institutions may find beneficial. I invite your child to participate in this research as I believe they may offer a valuable contribution to the knowledge of dyslexia and music.

As their parent/guardian, I am sure that you will have concerns. Please be assured that you are welcome to discuss this with me, and also welcome to ask for more information and contacts in the field of music and dyslexia which could be provided on request. You may take time to reflect on whether or not you would like your child to participate. After you have read through this information and if you agree, then the next thing I will do is ask your child for their permission.

If you have any questions at any time through this process, please contact me at

Purpose

Through my work as a piano teacher, I came to realise that there was much to learn about providing better lessons for dyslexic students. This has motivated my research. In this study, I will make observations of lessons, record lessons on videotape and ask students questions which will enable me to understand how they learn. The research will take place in the context of piano lessons and students will be invited to share their views in a relaxed way. Lesson planning will be guided by their goals in learning to play the piano.

Participation

Your child has been invited to take part in this research as monitoring their learning in piano lessons may enable us to understand and better tailor lessons for them and other dyslexic students in the future. You do not have to agree to your child's participation. The decision may be one that you want to discuss together. You can ask as many questions as you like and take your time to decide.

Procedures

• The following only applies to piano lessons:

Lessons will take place in my home or at the University of York. Before lessons take place, the student and the parent will ideally have a short, informal interview with me. That will be discussed in the next point. This may help me to better understand their goals in learning to play. If your child feels uncomfortable answering any questions during the lesson or in how they are asked to carry out an activity in the lesson, they are free to tell me. We will move on to another activity. The video-tapes and observations made during lessons are confidential and anonymous and will be seen only by myself, members of the University of York Music Department and a selected group of dyslexia specialists. The video-tapes will be kept in a password-protected Google Drive and will be destroyed after 10 years.

• <u>The following applies only to interviews:</u>

If your child does not wish to answer any questions during the interview, he/ she may tell me and I will move on to the next question. I will seek to ensure that the child is comfortable and relaxed, and the only people present will be the interviewer (myself) and the child, unless they wish for someone else to be present. The information recorded will be viewed by myself, members of the University of York Music Department and a selected group of dyslexia specialists.

Duration

The interviews and lessons will be outside of school/work hours. Initially, there will be an interview to assess the student's prior learning experiences, any information on their learning profile and their musical interests. After a series of lessons, another follow-up interview will be scheduled. Lessons are generally booked for half-hour sessions, although this depends on the age and level of the student. This may be something which we will agree to through discussion, depending on what is best for the student. It would be ideal if lessons could continue for a longer period of time (18 months- 2 years), however I do want to reassure you that you and your child can withdraw from the research at any point.

Risks and Discomforts

My goal will be to make the student as relaxed and comfortable during the course of the interview and the lessons. Some of the questions may pertain to sensitive and personal topics. For instance, it may remind them of an unpleasant experience in school. I will make it clear to your child that they do not have to answer any questions or participate in any activity with which they do not feel comfortable. Your child will be made aware that they do not have to provide an answer as to why they refuse. They must also be aware that their responses will not be shared and will remain confidential.

Benefits

First of all, my goal is that your child will enjoy piano lessons and learning about music. I desire to reduce stress during the lesson as much as possible, which is why it is important for me to hear about their learning experiences in the past. I believe that I will improve as a teacher to dyslexic students as I seek to apply recent research to practical learning during lessons. I hope to understand more about your dyslexic child's needs and strengths and that this may bring about an awareness of ways to improve lessons. These lessons may provide contributions to the field of music and dyslexia.

Confidentiality

Each student will be assigned a pseudonym in the research so that they will not be identifiable. The information about your child will only be shared with my supervisor using this pseudonym. Any video-recording will be edited so that participants are not identifiable.

Sharing of Research Findings

Research findings may be shared in a timely fashion in journals or conferences but with complete confidentiality maintained. It is my hope that these findings will be applicable and relevant to other interested people who may wish to learn from my research.

Right to refuse or withdraw

Your child will be asked to agree but has every right to refuse or withdraw from the research without the need to offer an explanation for the refusal. The child's concerns and wishes will be taken very seriously.

Who to Contact

If you have any questions, you may ask them now or later, even after the research has started. This research has been approved by the University Ethics Committee, a group whose task it is to make sure that research participants are protected from harm. For further information, you may contact:

Kristl Kirk, PhD Candidate Department of Music, University of York,

Or my supervisor

Dr. Liz Haddon, Department of Music, University of York

PART II: Certificate of Consent

Certificate of Consent

I have been asked to give consent for my daughter/son to participate in this research study which will involve participating as a student in piano lessons, video-taping and documenting of lessons and an interview. I have read the information. I have had the opportunity to ask questions about it and any questions that I have asked have been answered to my satisfaction. I consent voluntarily for my child to participate as a participant in this study.

Print Name of Parent or Guardian _____

Signature of Parent of Guardian_____

Date _____

Day/month/year

Appendix C

Information sheet and consent form participants

Kristl Kirk PhD candidate Department of Music, University of York Heslington, York YO10 5DD

Information Sheet and Consent Form for Participants

This information/consent form is for those who have been invited to participate in the research entitled 'Decoding Music Education with Dyslexic Students: An action research project examining student, teacher and parental interactions in the context of instrumental lessons'.

This Informed Consent Form has two parts:

- Information Sheet (to share information about the study with you)
- Certificate of Consent (for signatures if you agree that your child may participate)

You will be given a copy of the full Informed Consent Form.

Part 1: Information Sheet

Introduction

I am a PhD candidate from the Department of Music at the University of York who is interested in researching music learning and dyslexia. My goals are to better understand how dyslexic students learn, to improve my piano teaching for dyslexic students as well as to provide resources which other teachers and institutions may find beneficial. I invite your child to participate in this research as I believe they may offer a valuable contribution to the knowledge of dyslexia and music.

Please be assured that you are welcome to discuss this with me and to ask for more information and contacts in the field of music and dyslexia which could be provided on request. After you have read through this information and if you agree, then please sign the certificate of consent.

If you have any questions at any time through this process, please contact me at

Purpose

Through my work as a piano teacher, I came to realise that there was much to learn about providing better lessons for dyslexic students. This has motivated my research. In this study, I will make observations of lessons, record lessons on video-tape and ask students questions

which will enable me to understand how they learn. The research will take place in the context of piano lessons and students will be invited to share their views in a relaxed way. Lesson planning will be guided by their goals in learning to play the piano.

Participation

You have been invited to take part in this research as monitoring your learning in piano lessons may enable me to understand and better tailor lessons for you and other dyslexic students in the future. You do not have to agree to participate. The decision may be one that you want to discuss with someone else. You can ask as many questions as you like and take your time to decide.

Procedures

• The following only applies to piano lessons:

Lessons will only take place in my home or at the University of York. Before lessons take place, the student will ideally have a short, informal interview with me. That will be discussed in the next point. This may help me to better understand their goals in learning to play. It will be made clear that if they feel uncomfortable answering any questions during the lesson or in how they are asked to carry out an activity in the lesson, they are free to tell me. We will move on to another activity. The video-tapes and observations made during lessons are confidential and anonymous and will be seen only by myself, members of the University of York Music Department and a selected group of dyslexia specialists. The video-tapes will be kept in password protected Google Drive and will be destroyed after 10 years.

• The following applies only to interviews:

If you do not wish to answer any questions during the interview, you may tell me and I will move on to the next question. I will seek to ensure that you are comfortable and relaxed, and the only people present will be the interviewer (myself) and you, unless you wish for someone else to be present. The information recorded will be viewed by myself, members of the University of York Music Department and a selected group of dyslexia specialists.

Duration

The interviews and lessons will be outside of school/work hours. The interview may take about one hour. After a series of lessons, another follow-up interview will be scheduled. Lessons are generally booked for half-hour sessions, although this depends on the age and level of the student. This may be something which we will agree to through discussion, depending on what is best for the student. It would be ideal if lessons could continue for a longer period of time (18 months- 2 years), however I do want to reassure you that you can withdraw from the research at any point.

Risks and Discomforts

My goal will be to make the student as relaxed and comfortable during the course of the interview and the lessons. Some of the questions may pertain to sensitive and personal topics. For instance, it may remind you of an unpleasant experience in school. You do not have to answer any questions or participate in any activity with which you do not feel comfortable.

You do not have to provide an answer as to why you refuse. Your responses will not be shared and will remain confidential.

Benefits

First of all, my goal is that you will enjoy piano lessons and that the learning that takes place is guided by your goals. I desire to reduce stress during the lesson as much as possible, which is why it is important for me to hear about past learning experiences. I believe that I will improve as a teacher to dyslexic students as I seek to apply recent research to practical learning during lessons. I hope to understand more about your needs and strengths and how this may bring an awareness of how to improve lessons. These lessons may provide contributions to the field of music and dyslexia.

Confidentiality

Each student will be assigned a pseudonym in the research so that they will not be identifiable. The information about your lessons will only be shared with my supervisor using this pseudonym. Any video-recording will be edited so that participants are not identifiable.

Sharing of Research Findings

Research findings may be shared in a timely fashion in journals or conferences but with complete confidentiality maintained. It is my hope that these findings will be applicable and relevant to other interested people who may wish to learn from my research.

Right to refuse or withdraw

You have every right to refuse or withdraw from the research without the need to offer an explanation for the refusal. Your concerns and wishes will be taken very seriously.

Who to Contact

If you have any questions, you may ask them now or later, even after the research has started. This research has been approved by the University Ethics Committee, a group whose task it is to make sure that research participants are protected from harm. For further information, you may contact:

Kristl Kirk, PhD Candidate, Department of Music, University of York

Or my supervisor

Dr. Liz Haddon, Department of Music, University of York,

PART II: Certificate of Consent

Certificate of Consent

I have been asked to give consent to participate in this research study which will involve participating as a student in piano lessons, documentation of lessons through observations and video-taping, and an interview. I have read the information. I have had the opportunity to ask questions about it and any questions that I have asked have been answered to my satisfaction. I consent voluntarily to be a participant in this study.

Print Name of Parent or Guardian _____

Signature of Parent of Guardian_____

Date _____

Day/month/year

Appendix D Information sheet and consent form participants

UNIVERSITY of York

Decoding instrumental music education for students with dyslexia: An action research project

School of Arts and Creative Technologies

Participant Information Sheet – Non-Anonymous Interviews Project background

The University of York would like to invite you to take part in the following project: Decoding instrumental music education with dyslexic students: An action research project.

Before agreeing to take part, please read this information sheet carefully and let us know if anything is unclear or you would like further information.

What is the purpose of the project?

The work is being conducted according to restrictions that have been subject to approval by the ACT Ethics committee. The Chair of the ACT Ethics committee can be contacted on <u>ACT-ethics@york.ac.uk</u>.

For this research project, we are interested in how to better understand how dyslexic students learn and to improve my piano teaching for dyslexic students as well as to provide resources which other teachers and institutions may find beneficial. I invite you to participate in this research as I believe you may offer a valuable contribution in evaluating the findings of my research.

Your participation in this project will involve attending a Zoom meeting which should last no longer than one and a half hours. The meeting will be recorded for the purposes of transcription.

Please note that to comply with the approved Ethics requirements of this work, we do not intend to discuss sensitive topics with you that could be potentially upsetting or distressing. If you have any concerns about the topics that may be covered in the research study, please raise these concerns with the researcher.

Your participation in this project is voluntary. If you wish, we will provide you with the transcripts of the session. If you would like to receive access to these, you can indicate as such on the consent form.

Why have I been invited to take part?

You have been invited to take part because of your expertise in the area of SEN (special educational needs) music education.

Do I have to take part?

No, participation is optional. If you do decide to take part, you will be given a copy of this information sheet for your records and will be asked to complete a participant consent form. If you change your mind at any point during the research activity, you will be able to withdraw your participation without having to provide a reason. To withdraw your participation, you need to inform the researcher and your data will be deleted as soon as possible.

Will I be identified in any outputs?

Yes. Your participation in this interview is non-anonymous and therefore you will be identified in the following outputs. This is to identify the importance of your role and expertise in qualifying you to participate in this evaluative focus group. You will be named in the introductory text, but any quotes that you make will be anonymised.

Privacy Notice

This section explains how personal data will be used by Kristl Kirk in the Decoding instrumental music with dyslexic students: An action research project at the University of York.

For this project, the University of York is the <u>Data Controller</u>. We are registered with the Information Commissioner's Office. <u>Our registration</u> <u>number</u> is Z4855807.

What is our legal basis for processing your data?

Privacy law (the UK General Data Protection Regulation (GDPR) and Data Protection Act 2018) requires us to have a legal reason to process your personal data. Our reason is we need it to perform a public task.¹

This is because the University has a <u>public function</u>, which includes carrying out research projects.² We need to use personal data in order to carry out this research project.

Information about your health, ethnicity, sexual identity and other sensitive information is called <u>"special category" data</u>. We have to have an additional

legal reason to use this data, because it is sensitive. Our reason is that it is needed for research purposes.³ All research projects at the University follow our <u>research ethics policies</u>.

How do we use your data?

The data collected from this evaluative focus group will be used in Kristl Kirk's PhD thesis. Whilst the participants will not be anonymous, any quoted material will be anonymised in the thesis.

Who do we share your data with?

As well as this, we use computer software or systems to hold and manage data. Other companies only provide the software, system or storage. They are not allowed to use your data for their own reasons.

We have agreements in place when we share data. These agreements meet legal requirements to ensure your data is protected.

How do we keep your data secure?

The University is serious about keeping your data secure and protecting your rights to privacy. We don't ask you for data we don't need, and only give access to people who need to know. We think about security when planning projects, to make sure they work well. Our IT security team checks regularly to make sure we're taking the right steps. For more details see <u>our security webpages.</u>

How do we transfer your data safely internationally?

If your data is stored or processed outside the UK, we follow legal requirements to make sure that the same level of privacy rules still apply.

How long will we keep your data?

The University has rules in place for <u>how long research data can be kept</u> when the research project is finished. For this project, data will be kept for ten years and will then be destroyed by the researcher.

What rights do you have in relation to your data?

You have rights over your data. This sheet explains how you can stop participating in the study, and what will happen to your data if you do. This information is in the section 'Do I have to take part?'

If you want to get a copy of your data, or talk to us about any other rights, please contact us using the details below.

Questions or concerns

If you have any questions or concerns about how your data is being processed, please contact my supervisor, Dr Liz Haddon, at

If you have further questions, the University's Data Protection Officer can be contacted at <u>dataprotection@york.ac.uk</u> or by writing to: **Data Protection Officer**, **University of York, Heslington, York, YO10 5DD**.

Right to complain

If you are unhappy with how the University has handled your personal data, please contact our Data Protection Officer using the details above, so that we can try to put things right.

If you are unhappy with our response, you have a right to <u>complain to the Information</u> <u>Commissioner's Office</u>. You can also contact the Information Commissioner's Office by post to **Information Commissioner's Office**, **Wycliffe House**, **Water Lane**, **Wilmslow**, **Cheshire**, **SK9 5AF** or by phone on **0303 123 1113**. ¹This refers to <u>UK GDPR Article 6 (1) (e)</u>: processing is necessary for the performance of a task carried out in the public interest or in the exercise of official authority vested in the controller

² <u>Our charter and statutes</u> states: 4.f. To provide instruction in such branches of learning as the University may think fit and to make provision for research and for the advancement and dissemination of knowledge in such manner as the University may determine.

³This refers to <u>UK GDPR Article 9 (2) (j)</u>: processing is necessary for archiving purposes in the public interest, scientific or historical research purposes or statistical purposes in accordance with Article 89(1) based on Union or Member State law which shall be proportionate to the aim pursued, respect the essence of the right to data protection and provide for suitable and specific measures to safeguard the fundamental rights and the interests of the data subject

Decoding instrumental music education

for students with dyslexia: An action research project

UNIVERSITY of York

School of Arts and Creative Technologies

Participant Consent Form – Non-Anonymous Interviews

Thank you for your interest in this project. This project aims to obtain evaluative feedback of research findings from a focus group of participants with expertise in the area of SEN (special educational needs) and music teaching.

Please read the following statements carefully and tick the appropriate box:

	YES	NO
I have read the information sheet about this project		
I agree to take part in this project		
I consent to participating in a recorded Zoom meeting		
I understand my right to withdraw and/or destroy my data from this project at any time		
I consent to be identified by name as a member of this focus group in the outputs from this project. I realise that any quotes will be anonymised in the thesis.		
I am over the age of 18		

If you wish to be informed about the outcomes from this project, please provide your email address:

Participant Name:

Researcher Name:

Participant Signature:

Researcher Signature:

Appendix E

Questionnaire questions_Dyslexic students' perceptions of music exams

Q1. Please click to indicate that you consent to the following:

- 1. I have read the information sheet and understand the content.
- 2. I understand that as guardian to a minor child I may assist them in completing this survey, but that their permission is also required.
- 3. I understand that the purpose of this survey is to better understand dyslexic students' perceptions of music exam access arrangements and reasonable adjustments.
- 4. I consent voluntarily to participate in this research study which will involve participating by completing a survey.

Q2. What is your age?

- 5 or under
- 6-10
- 11-18
- 19 or above

Q3. Have you been assessed or diagnosed with dyslexia or any specific learning disability?

- Yes
- No

Q4. Please select as many options which you feel might best describe your specific learning difficulties:

- difficulty with attention or concentration
- difficulties with reading
- difficulties with writing
- struggle with left-right differentiation
- poor hand independence
- poor general motor coordination

- difficulties involving numbers
- difficulties with sequencing
- poor working memory
- visual distraction

Q5. Please select which instruments you have studied and indicate how long you have studied them:

- Piano
- Woodwind instruments
- String instruments
- Brass instruments
- Voice
- Harp
- Organ
- Percussion
- Guitar
- Other, please describe

Q6. Which best describes the type of exam you took ?

- Music theory
- Practical
- Both
- Other, please describe:
- None

Q7. With which music exam board have you taken an exam?

- Associated Board of Royal Schools of Music (ABRSM)
- Trinity College London (TCL)
- London College of Music (LCM)

- Rockschool (RSL) Other
- None (please describe reason for this)

Q8. How did you obtain information regarding the various access arrangements or reasonable adjustments available for dyslexic students during music exams?

- your instrumental teacher
- the exam board website
- contact with the British Dyslexia Association
- your school
- you didn't know there were special arrangements available for dyslexic students for music exams
- Other

Q9. In general, do you feel that the information on the exam board's website was presented in a dyslexia friendly manner?

- Yes
- No
- I don't know

Q10. Which part of the website was not dyslexia friendly?

- font style
- colour combinations or choices
- font size
- spacing of text and images
- lack of customisable options
- design
- content
- absence of a site map
- it was incompatible with text reading software
- lack of alternative methods of obtaining information

Q11. Was information regarding the detail of the exam location and venue presented in a dyslexia friendly manner?

- No
- Yes

Q12. Were access arrangements made clear to you?

- No
- Yes

Q13. Were you requested to provide supporting evidence to document your dyslexia?

- No
- Yes

Q14. If you answered yes to the previous question, please detail the type of evidence you had to supply:

- Letter from an Educational Psychologist or Assessor
- Letter or email from head teacher or principal
- Letter or email from a SENCO
- Letter or email from a Disability Support Unit at a higher education establishment
- Letter or email from a suitable healthcare professional
- Other

Q15. How far in advance of the exam did you have to supply the supporting evidence:

- Four weeks
- One to three months
- Other

Q16. Which adjustments did you use during the exam? Please click all that apply:

- Extra time during the sight-reading portion of the exam
- Assistance with left-right orientation at any point in the exam

- Extra time during the aural test portion of the exam
- Bringing music which had been notated or modified for my reading style
- Modified music during the sight-reading portion of the exam
- Aural repetition in place of the sight-reading portion of the exam
- Use of the scale book to aid processing difficulties
- None
- Request for modified or simpler language used by the examiner
- Other

Q17. Which adjustments did you use during the music theory exam? Please click all that apply:

- Large print exam papers
- Modified exam paper
- Coloured overlay or coloured paper for the exam
- Extra time
- Use of a scribe Use of a reader
- Other

Q18. Were you offered any alternatives to traditional exam models which might have met your adaptive needs better?

- No
- Yes

Q19. If you answered yes to the previous question, please detail the alternatives you were offered:

Q20. Do you feel that the access arrangements were effective in giving you a fair opportunity in your exam?

- No
- Yes

Q21. If you answered no on the previous question, please describe why:

Q22. Do you feel that the reasonable adjustments were effective in giving you a fair opportunity for your exam?

- No
- Yes

Q23. If you answered 'no' to the previous question, please describe why you felt the exam might not have given you a level playing field with other candidates:

Q24. Which recommendations, if any, would you advise for music exams to be more accessible and inclusive for dyslexic students?

Appendix F

Interview schedule for teacher interviews

1. How did you identify dyslexic students in your teaching practice? If you work in an institution, how do they support your teaching of dyslexic students?

2. What do you feel are the greatest challenges facing instrumental teachers when teaching dyslexic students, and why?

What do you think are some of the most significant challenges facing the learner?

3. What are some of the ways that you feel these challenges can be ameliorated? What is the collaboration between you and the student in finding strategies that work?

4. Do you feel that certain teaching strategies are effective when working with dyslexic students? Which ones seem to be effective and why?

Rhythm?

Pitch?

Processing?

Sight-reading?

Exams?

Does your use of these vary with the age of the student and if so, how/why?

What strategies are not effective?

How have you found out which ones are effective (trial and error/reading/training?

5. Dyslexic strengths are beginning to be more widely explored in educational contexts.

What strengths have you identified in dyslexic students?

How did you discover these?

How do you use these in your teaching?

6. How do you feel about the use of technology in teaching dyslexic students?

Do you use it - if so, how?

How is it best used?

Do you see gaps in the technology that is available currently?

How could you envision it might be used more effectively in the future?

7. What are the ways in which you feel teachers can best help dyslexic students to find and nurture their own voice in the process of learning to play a musical instrument?

8. Do you have any experience of dyslexic students being mentored in the educational context?

9. How can you as a teacher respond to and develop awareness of the roles or effects of co-occurring conditions (ADHD, dyspraxia/ motor coordination, autism, dyscalculia) on the learning process?

10. How have you developed your understanding of dyslexia and of how to support students with dyslexia?

Have you been on any courses/training focused on this? (self-funded or provided by your institution?)

Have you read any books about this?

Have you talked to teaching colleagues about this? Etc.

How has this knowledge informed your teaching practices?

11. Are there any resources from fields outside music education that have influenced your understanding and approach to dyslexic students? (E.g. cognitive neuroscience, behavioural, general education)

12. How do you see the role of parents in dyslexic student's lessons or in supporting their practise?

13. Do you consider a dyslexic students' mind-set and the psychological aspects of teaching the dyslexic learner? How does this affect their learning process? How do you support them in challenges?

14. Do you have any further comments?
Appendix G

Interview schedule for parents

- 1. How did you identify that your child had dyslexia?
- 2. How did they feel about having dyslexia?
- 3. What, if any, challenges have they faced at school or with music lessons?
- 4. How do they cope with these challenges?
- 5. Do they face any challenges related to organisation?
 - a. Motor coordination?
 - b. Attention?
- 6. What strengths do you perceive that they have?
- 7. Do you feel they are being supported well at school and in music lessons?
- 8. Are there any ways you see their educational experiences might be improved?
- 9. What helps your child to learn best?
- 10. Is there anything else you would like to add?

Appendix H

Interview schedule for students

- 1. How did you identify that you had dyslexia?
- 2. How did you feel about finding out that you had dyslexia?
- 3. Do you feel that dyslexia has affected your ability to learn music?
- 4. Can you describe your challenges?
- 5. What strategies worked best for you?
- 6. Do you perceive that you have strengths that are related to dyslexia?
- 7. How were your experiences with music teachers?
- 8. Are there any resources that have informed your understanding of dyslexia or how it affects your musical abilities?
- 9. Is there anything else you would like to add?

Appendix I

Sample lesson plan

Lesson plan

Objectives: Ben is preparing for GCSE music.

He needs to become familiarised with music terms. Today, we will focus more on *texture*. Last week, I introduced him to the terms: *monophonic*, *homophonic* and *polyphonic*.

- 1. Ask: Do you know of any music that is monophonic, homophonic or polyphonic?
- 2. Depending on his response, I will ask him to play his piece 'This is amazing', which is melody and accompaniment and see if he can categorise it as homophonic.
- 3. Ask: What do you think the texture of this music might be?
- 4. See if he can connect knowledge to current pieces: Have you played any monophonic pieces?
- 5. What about the bars in Musette in D by Bach that you play? Ask him to play and see if he can connect the monophonic bars with his knowledge.
- 6. Ask him to choose some music samples for me for next week of each type of texture.

Secondly, to reinforce these terms kinaesthetically, I will offer him the opportunity to use some materials at the desk and see if he wants to make a representation of the sounds on paper which he could keep in his notebook.

Alternatives: If he is not interested in doing that, we could have him write the terms down, and then have a listen to 'Cat Patrol' to see how he is progressing with that. Alternatively, we could practise our duet, or he could choose a piece he might like to play.

Appendix J

Accessible text-only version of UDL Checkpoints

Provide multiple means of Representation

Provide options for Perceptions (1) Offer ways of customizing the display of information (1.1) Offer alternatives for auditory information (1.2) Offer alternatives for visual information (1.3) Provide options for Language & Symbols (2) Clarify vocabulary and symbols (2.1) Clarify syntax and structure (2.2) Support decoding of text, mathematical notation, and symbols (2.3) Promote understanding across languages (2.4) Illustrate through multiple media (2.5) Provide options for Comprehension (3) Activate or supply background knowledge (3.1) Highlight patterns, critical features, big ideas, and relationships (3.2) Guide information processing and visualization (3.3) Maximize transfer and generalization (3.4)

Provide multiple means of Action & Expression

Provide options for Physical Action (4) Vary the methods for response and navigation (4.1) Optimize access to tools and assistive technologies (4.2) Provide options for Expression & Communication (5) Use multiple media for communication (5.1) Use multiple tools for construction and composition (5.2) Build fluencies with graduated levels of support for practice and performance (5.3) Provide options for Executive Functions (6) Guide appropriate goal-setting (6.1) Support planning and strategy development (6.2) Facilitate managing information and resources (6.3) Enhance capacity for monitoring progress (6.4)

Provide options for Recruiting Interest

(7) Optimize individual choice and autonomy (7.1) Optimize relevance, value, and authenticity (7.2) Minimize threats and distractions (7.3) Provide options for Sustaining Effort & Persistence (8) Heighten salience of goals and objectives (8.1) Vary demands and resources to optimize challenge (8.2) Foster collaboration and community (8.3) Increase mastery-oriented feedback (8.4) Provide options for Self-Regulation (9) Promote expectations and beliefs that optimize motivation (9.1) Facilitate personal coping skills and strategies (9.2) Develop self-assessment and reflection (9.3) Reference: CAST (2018). Universal design for learning guidelines version 2.2 [graphic organizer]. Wakefield, MA.

Appendix K

Sample reflective journal page

Egyptian Level - Grade 1 warm up

It's nice to see that Alex enjoys the harmonic minor sound of this piece, and he enjoys exploring the keyboard in the lower registers with the piece – which he calls 'the dark side'. He decides to return to it later in the lesson, which I feel is a very positive thing. He has forgotten his other books and blamed his mum again. I have spare copies, so we were able to continue.

Train Ride - Prep test ABRSM

The rhythm is quite uneven, but he is remembering the pitches. The question is how to get him to hear a steady pulse? He played the piece through to the end and plays hands together. I praise him for these things but ask him to try to tap while I play the tune in order for him to hear it played evenly. He does a better job initially when he plays it again and then falls back into habits of the way he was playing it before. I suspect he has practised it this way. I can feel myself thinking 'You can play it as you want to at home, but here we will play it as it is written in the book'. However, he simply does not recognise that he has learned it incorrectly and is very frustrated. And of course, when he is home, he may not remember what we have done together.

Then he tried playing the right hand only, without much rhythmic attention. I know he is proud of himself for being able to play the right-hand pitches. I focused on praising him for playing the right hand through. Next week, we can try segmenting the rhythm into a catchy phrase, which worked well in the past for him.

Behaviour: He was playing the piano while I was talking to him which I don't normally permit. I will have a quiet word at the beginning of our next lesson, explaining again that it's best to do one thing at a time.

G major scale (presented aurally and with a simple line of standard notation)

I prompt on fingering and ask for two octaves with the right hand. I demonstrate fingering for two octaves. He achieves it with multiple prompts. Perhaps too much talking at him while he tries it. I tried blocking out the scale for fingering purposes. I explain the purposes of the scale, to strengthen the fingers.

Lullaby Carol

It took quite a bit of time figuring out where hands should start, and which fingers are on the starting notes. I demonstrate and then he tries but struggles with this. Could this be an issue of finger independence slowing him down? We could try some fun games at the keyboard to work on improving that, perhaps next week we could try some short pieces as exercises for improving this.

Behaviour: Alex was a bit distracted after a student came by to retrieve a coat they left. I try to minimise these distractions, but they happen. I praised him 'You did that on your own and that was brilliant' (trying to encourage self-regulation and get his attention and focus back). Tried demonstration to learn the next section, and that seemed to work very well.

We chat through about the next week's practise and make a copy of a 'Jurassic World' piece he wants to try. Alex chose some goals which I also feel are achievable, and most importantly, he seems very motivated to want to play it.

Action points:

- 1. More basic rhythmic work, segmentation of rhythm with phrases he makes up and will remember
- 2. Finger independence exercises/ pieces/ games
- 3. Try some note finding exercises on the keyboard.
- 4. Ask him if he wants to record his pieces or play for mum next week.

Appendix L

Piano teaching health and safety risk assessment during Covid-19

HEALTH & SAFETY RISK ASSESSMENT – Covid-19 - Example

HEALTH & SAFETY RISK ASSESSMENT - Covid-19 - Example

Site / Location:	Home	Assessor	
Activity:	Private Piano Teaching	Date:	27th June 2020

A. INTRODUCTION This risk assessment considers the risk posed by Coronavirus-19 (SARS-CoV-2) by identifying circumstances where exposure to the virus could occur (see B - 'Sources of Risk'). Existing control measures (that are already in place) for managing the associated risk are defined (see C - 'Control Measures'). Responsibility for these measures are shown in bold. To determine if the sources of risk are managed adequately by the existing control measures, the level of residual risk (taking into account the effect of the control measures) is calculated (see D). Residual risk level is calculated using the scoring matrix shown below. Where the residual risk is scored higher than 6, further action may be required to reduce the risk further (see E). more 4 3 Likelihood 2 ess 1 1 2 3 4 5 better worse Severity **Risk rating:** Monitor to ensure control measures are implemented consistently and that the rating remains valid. 8-12 Amber Try to identify additional controls to reduce the risk. Ensure that control measures are implemented consistently and look to improve by the next review. 15-25 Red Cease this activity until additional controls can be put in place to manage the risk. Circumstances of potential exposure or increased risk of serious illness are identified in the following broad areas:

1. Travel to and from lesson on public transport.

- 2. Contact with common contact surfaces.
- 3. Close proximity activity (air bourne transmission)

	Close proximity between teacher and student could result in airborne transmission of virus:					 Touching common contact surfaces including those listed below could result in virus transmission. Door handles Piano keys Music Practice notes 	 Students who need to use public transport including buses and trains could be in close proximity of other members of the public resulting in potential virus transmission. 		B. SOURCES OF RISK
 b. If / when not practicable to maintain 2m distancing, 1m distancing plus at least one of the following mtigating measures is applied: side to side positioning (no face to face communication) face shield worm face masks or coverings worm. 	a. Seating set up so that teacher and student are 2m apart. Teacher	e. Students makes their own practice notes using their own pen / pencil. Student	d. Students use their own music where possible. Student	c. Teacher and students wash hands immediately before and after lesson. Teacher & Student	b. Where practicable, regularly used doors are propped open to prevent the need for hand contact. Teacher	 a. All common contact surfaces cleaned immediately prior to each student arriving including: Door handle Piano keys Perspex screen Teacher 	a. Check if students who use public transport are able to maintain distancing. If not, consideration made to vary lesson time so student can avoid peak times. Teacher		C. EXISTING CONTROL MEASURES
	4					4	4	s	Þ
	-					-	2	-	RISK
	4					4	8	Ň,	
							Consider conducting lesson by remote connection technology such as Teams, Skype Facetime or other secure platform. Ensure appropriate safeguarding at all times. Teache		E. ADDITIONAL MEASURES (if SxL>6)

Appendix M

Student age-appropriate information sheet for piano lessons





Appendix N Focus group brief

March 15, 2024

Research aim:

The aim of this study was to investigate dyslexia in the instrumental music education context and to understand more about the complexities of the interrelationships between music teachers, students with dyslexia and their parents with the goal of improving and enhancing instrumental music teaching for students with dyslexia. By developing a better understanding of dyslexic students and approaches which might aid them in learning music, a secondary goal would be to generate resources which may be beneficial for stakeholders.

Research questions:

1) How are the strategies, methods and materials that are often recommended for students with dyslexia perceived by teachers and students and utilised in practice?

- 2) How can the strengths of dyslexia be identified and explored in the music lesson context?
- 3) What might be learned from teacher, student and parental interactions?

Reason for doing this research:

Though there is an increase in literature relating to special educational needs (SEN) in music education (Culp & Salvador, 2021; Darrow & Adamek, 2018; Welch & Ockelford, 2015); relatively little material exists to address these questions, and research which accounts for the voice of the student with dyslexia in music education is limited. Furthermore, the scarcity of available literature and funding for music education dyslexia research suggests that learning music is not seen as valuable as learning to read text, although research examining music as a therapy to promote literacy with dyslexic students is becoming more prevalent.

Conceptual frameworks:

My approaches in this work are informed by a mixture of interpretivist and pragmatic views. A socially constructed theoretical framework provides the structural foundation whereby qualitative data is obtained and analysed. The philosophical perspective that knowledge is socially constructed, described by (Crotty, 1998) as the understanding that 'truth, or meaning, comes into existence in and out of our engagement with the realities in our world' (p. 5), establishes the underlying theoretical framework for this research. The ecological and bioecological models of development (Bronfenbrenner, 1977; Bronfenbrenner & Morris, 2007; Bronfenbrenner, 1979) are used to account for the developmental growth of myself as a teacher and that of my students.

Ethical considerations:

Human participants require particular considerations which are set out in the British Educational Research Association's guidelines (Bera, 2018). The following ethical factors had to be considered in my application for ethical approval from the Arts and Humanities Ethics Committee (AHEC) at the University of York: potential conflicts of interest, methods of data collection, participants, informed consent, vulnerabilities, risk analysis, data protection, data storage or transfer, and anonymity. I mitigated risk through the following:

- I obtained ethical approval in advance of the research being carried out.
- I obtained voluntary, informed consent in advance of the research being carried out. In the case of minors, I obtained informed consent from their guardians.
- I maintained ethical and safe boundaries in the teaching setting in order to reduce any risks.
- I maintained openness and transparency about the use of personal information, by guaranteeing anonymity, by safely storing data and by reassurance of trust between the investigator and other participants (Bera, 2018).

 Data protection was secured by the use of pseudonyms and safe storage in passwordprotected files. Data for this research was stored on Google Drive and University of York Filestore, and will be retained for ten years in accordance with the University of York protocol. I will then be responsible for destroying the data.

Methodology and data collection:

This project of action research investigated the teaching of two piano students with a formal assessment of dyslexia in the one-to-one lesson context. Action research is cyclical and change producing. The research took place through four main cycles with a number of smaller cycles in between.

Action research cycles

Reconnaissance Reflecting and Planning Implementing and Observing Evaluating and Reflecting

Drawing on a literature review of dyslexia and music education and informed by semistructured interviews with instrumental music and vocal teachers possessing experience in teaching students with dyslexia, parents of students with dyslexia, and music students with dyslexia, this multi-dimensional approach also included reflective observations from my oneto-one piano lessons with the students.

Data analysis:

Research data was analysed according to Braun & Clarke (2006) thematic analysis.

I acknowledge my role as researcher in collecting the data and also in organising the data in ways which demonstrate how experiences, events and perspectives operate in a socially constructed environment. My aim was to answer the research questions by interpreting the experiences of teachers, dyslexic students and parents and analysing this data, along with my experiences of teaching dyslexia students, to inform my teaching practice and support ongoing pedagogical practice.

The significance of this work:

At the heart of these transitions in my teaching practice has been a greater understanding of dyslexia in the instrumental teaching process. This has been informed by literature and from the perspectives of teachers, parents and students, as well as reflective observations from my own teaching practice. A key finding has been that participants overwhelmingly focused on the deficits of dyslexia and that teachers perceived a lack of research, training and resources to support their teaching for students with dyslexia (Chapter Seven). This has the potential to lead to assumptions and misapplication of strategies which could mean detrimental outcomes for the student. It is crucial that instrumental music teachers have access to high-quality, accessible music training which emphasises an individualised, strengths-focused approach. From this, I have developed a Knowledge and Practice Standards document and have created a new risk-resilience model to better understand how to support students with dyslexia in the instrumental teaching context.

References:

- Bera. (2018). Ethical Guidelines for Educational Research, fourth edition, London.
 https://www.bera.ac.uk/researchers-resources/publications/ethicalguidelines-foreducational-research-2018. Ethical Guidelines for Educational Research, Fourth Edition.
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2). https://doi.org/10.1191/1478088706qp063oa
- Bronfenbrenner, U., & Morris, P. A. (2006). The bioecological model of human development. In
 R. M. Lerner & W. Damon (Eds.), *Handbook of child psychology: Theoretical models of human development* (6th ed., pp. 793–828). John Wiley & Sons, Inc.
- Bronfenbrenner, U. (1977). Toward an experimental ecology of human development. *The American Psychologist*, 32(7). https://doi.org/10.1037/0003-066x.32.7.513
- Crotty, M. (1998). The foundations of social research: Meaning and perspective in the research process. Sage.
- Culp, M. E., & Salvador, K. (2021). Music teacher education program practices: Preparing teachers to work with diverse learners. *Journal of Music Teacher Education*, 30(2). https://doi.org/10.1177/1057083720984365
- Darrow, A.-A., & Adamek, M. (2018). Instructional strategies for the inclusive music classroom. *General Music Today*, *31*(3). https://doi.org/10.1177/1048371318756625

Welch, G. F., & Ockelford, A. (2015). The importance of music in supporting the development of children with learning disabilities. *International Journal of Birth & Parent Education*, *2*(3), 23-25.

Appendix O Workshop participant feedback

Participant feedback Music Mark Northwest Music Hubs Conference workshop September 4, 2023, collected anonymously through Mentimeter

What was your main takeaway from today?

So good to hear about strengths of dyslexic people.

I thought this was really helpful- probably need to think about how I will apply what I learned but I would like more time and training on this.

Helped broaden knowledge and consider strengths of dyslexic students.

Really useful to discuss the practical elements of teaching Dyslexic students but also how these can be applied for teaching all our students.

Lots of recommendations

I will try to teach from my own experiences if what I think is dyslexia.

That strategies need to be individualised for each person and about various strategies that could be put into place.

Useful resources to share with staff.

Thinking about how to teach using students' strengths rather than focus on how to develop weaknesses.

I will make sure to celebrate the strengths of dyslexic pupils and use this to their advantage within lessons using alternative resources.

Learnt a lot about the modern ideas of Dyslexia and a very thorough presentation of what it means to be dyslexia and how to deal with students with dyslexia.

I think it will make me more aware of what to look out for.

It is interesting to gather some information about dyslexia and how it relates to the student's learning.

Found the session really interesting and insightful to see how to support dyslexic music students. Interesting to see an alternate side of the coin as my mum is a BDA assessor.

I learnt a lot of strategies that I can implement in my own learning as well as teaching. I really loved the focus on the positives! Thank you.

I enjoy learning more about dyslexia in a brain and inherited way. I will apply the positive and affirmative points of view of each student.

That dyslexia also comes with many strengths, and I should use these to the students advantage.

I also didn't realise that there were so many different types of dyslexia.

Teaching techniques and how to spot the signs of dyslexia.

Neurodivergence isn't a list of negatives. I will use the positive aspects we have thought about today in lessons.

Signal of rubbing eyes when concentration capacity is reached is interesting to hear - I'll look out for it.

Expanding rhythms, colour overlays not big solution. Increased aural perception.

Going to try colour coding the music with a specific student. Do you know about the Kodály approach? I find that very useful for a multisensory approach to teaching.

More open minded and greater awareness

Very informative thank you.

Helped me to understand the needs of a dyslexic child. Also helpful for myself as a musician who is dyslexic in understanding my own needs.

Drake music and a range of different strategies. I have encountered many of the issues mentioned in my teaching and feel I have a renewed appetite to tackle them.

Very Informative, really useful ideas and strategies. Thankyou.

Really interesting and informative introduction to dyslexia and music education.

Pupil's own perception of what we might consider a musical strength which the perhaps consider not to be ie aural.

Thinking about the particular challenges dyslexic students might face and how to teach them in a way that celebrates their strengths and encourages them rather than discouraging them.

Interesting overview of a complex topic

Books and resources

Great informative presentation. Thank you.

Being so much more flexible with activities and be more aware of their needs in each lesson

Help in recognition of students with dyslexia.

Very interesting. Will look into it further.

Reassessing own teaching

Dyslexia can be a strength if diagnosed and handled appropriately.

Appendix P

International Dyslexia Associations' Knowledge and Practice Standards for

Teachers of Reading



Executive Summary

Reading Difficulties Affect a Large Proportion of the Student Population

Learning to read and write is not natural or easy for many-if not most-students. Learning to read requires mastery of a complex web of underlying language skills that, for the large majority, must be explicitly taught and learned over several years. Reading, spelling, writing, and language abilities exist on a continuum, with only about a third of the students in the United States demonstrating proficient or advanced literacy skills by 4th grade. On the lower end of the distribution, the National Assessment of Educational Progress (NAEP) consistently finds that about 32% of all fourth-graders read at a level described as "below basic," which is not sufficient to support grade-level academic work. The proportion of struggling students is far higher in minority and poorer communities. While most of these students will not qualify for special education under federal laws, their reading and language weaknesses must be addressed. Most of these at-risk students will depend on instruction given in the regular classroom, supplemented by small-group instruction within a response-to-intervention (RTI) framework. The 6–8% of students who may qualify for special education services for suspected learning disabilities will typically demonstrate severe difficulties with language, reading, and writing, but will still be participating in general education classrooms. Clearly, the responsibility for teaching reading and writing to all students resides first with classroom teachers and secondarily with reading specialists, providers of supplementary services, and special education personnel.

Effective Instruction Is Key to Prevention and Intervention

Although dyslexia and related reading and language problems may originate with neurobiological differences, they are mainly treated with skilled teaching. Effective classroom instruction delivered by a knowledgeable teacher, especially in the early grades, can prevent or at least effectively address and limit the severity of reading and writing problems. Potential reading failure can be recognized as early as preschool and kindergarten, if not sooner. A large body of research evidence shows that with appropriate, intensive instruction, all but the most severe reading disabilities can be ameliorated in the early grades, and students can get on track toward academic success. For those students with persistent dyslexia, who need specialized instruction outside of the regular classroom, competent intervention from a specialist can lessen the impact of the disorder and help the student overcome and manage the most debilitating symptoms.

What is the nature of effective instruction for most students who are learning to read? The methods supported by research are explicit, systematic, cumulative, and multisensory in that they integrate listening, speaking, reading, and writing. They are also multilinguistic, as they directly teach the structure of language at all levels, including the speech sound system (phonology), the writing system (orthography), the structure of sentences (syntax), the meaningful parts of words (morphology), word and phrase meanings (semantics), and the organization of spoken and written discourse. The strategies emphasize planning, organization, attention to task, critical thinking, and self-management. Such aspects of instruction are important for all students who are acquiring new concepts, skills, and strategies, but they are especially critical for students with language-learning weaknesses, including dyslexia. A shift to more student-centered, workshop-oriented approaches is only appropriate after students are secure with the fundamentals.

Knowledge and Practice Standards for Teachers of Reading
©2018, The International Dyslexia Association

Structured Literacy

Structured Literacy is an approach to reading instruction that is beneficial for both general education students at risk for reading difficulties due to a variety of factors (e.g., low socioeconomic status, status as an English learner (EL)) *and* for students with disabilities.

This approach is characterized by the provision of systematic, explicit instruction that integrates listening, speaking, reading, and writing and emphasizes the structure of language across the speech sound system (phonology), the writing system (orthography), the structure of sentences (syntax), the meaningful parts of words (morphology), the relationships among words (semantics), and the organization of spoken and written discourse.

The following instructional principles are associated with the provision of Structured Literacy instruction:

- 1. Instructional tasks are modeled, when appropriate.
- 2. Explicit instruction is provided.
- 3. Meaningful interactions with language occur during the lesson.
- 4. Multiple opportunities are provided to practice instructional tasks.
- 5. Corrective feedback is provided after initial student responses.
- 6. Student effort is encouraged.
- 7. Lesson engagement during teacher-led instruction is monitored.
- 8. Lesson engagement during independent work is monitored.
- 9. Students successfully complete activities at a high criterion level of performance.

Are Teachers Prepared?

Teaching language, reading, and writing effectively, especially to students experiencing difficulty, requires considerable knowledge and skill. Regrettably, the licensing and professional development practices currently endorsed by many states are insufficient for the preparation and support of teachers and specialists. Researchers are finding that individuals with reading specialist and special education licenses often know no more about research-based, effective practices than those individuals with general education teaching licenses. The majority of practitioners at all levels have not been prepared in sufficient depth to prevent reading problems, to recognize the early signs of risk, or to teach students with dyslexia and related learning disabilities successfully. Inquiries into teacher preparation in reading have revealed a pervasive absence of rich content and academic rigor in many courses that lead to the certification of teachers and specialists. Analyses of teacher licensing tests show that, typically, very few are aligned with current research on effective instruction for students at risk. When tests are aligned with scientific research, far too many teacher candidates are unable to pass them. To address these gaps and promote more rigorous, meaningful, and effective teacher preparation and professional development, the Center for Effective Reading Instruction (CERI) has adopted this set of knowledge and practice standards.

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

Standards for Informed Practice

Although programs that prepare or support teachers, clinicians, or specialists differ in their methodologies, teaching approaches, and organizational purposes, they should ascribe to a common set of professional standards for the benefit of the students they serve.

The *Knowledge and Practice Standards for Teachers of Reading* explicitly sets forth the knowledge and skills that all teachers of reading are expected to possess to advance students' reading and writing profiles from a Structured Literacy approach in classroom, remedial, and clinical settings.

These standards reflect the current state of the scientific research base and are the result of a rigorous development and vetting process that included the input of a wide range of stakeholders, including researchers, educators, higher education faculty, clinical specialists, parents, and advocates.

Standard 1 addresses foundational concepts, derived from interdisciplinary research, about reading development and reading difficulties. Standard 2 covers knowledge of diverse reading profiles, including dyslexia. Standard 3 pertains to knowledge of assessment. Standard 4 addresses Structured Literacy teaching, offering detailed guidance with regard to the nature of effective instruction in each major domain (phonological sensitivity and phoneme awareness, phonics and word recognition, reading fluency, vocabulary, listening and reading comprehension, and written expression). Standard 4 also offers guidance regarding expectations for teachers engaged in fieldwork or practicum (e.g., in interpretation of assessments, planning differentiated instruction, lesson design, corrective feedback, and so forth). Standard 5 delineates ethical standards for the profession.

Guidance and Support for Preservice and In-service Teachers

Learning to teach reading, language, and writing is a complex undertaking. The competence and expertise of regular classroom teachers is the most important factor in determining who will learn to read, write, and use language well enough to succeed academically. General education teachers, as well as those who provide intervention and specialized instruction, deserve preservice courses that prepare them to teach all students, including those who may be off-track or struggling. Licensed, practicing teachers can still learn these critical skills through substantive in-service coursework. Preparatory and inservice coursework should emphasize the study of reading development, the structure of language, the nature of individual differences, and the methods of Structured Literacy for all those who must be taught how to read. When teachers are better prepared, the impact of reading difficulties, including dyslexia, will be lessened, and many more students will receive the instruction and support that they require to succeed academically.

INTRODUCTION

Purpose of These Standards

The Knowledge and Practice Standards are to be used to guide the preparation, certification, and professional development of those individuals who teach reading and related literacy skills in classroom, remedial, and clinical settings. The term *teacher* is used throughout this document to refer to any person whose responsibilities include reading instruction. The standards aim to specify what individuals responsible for teaching reading should know and be able to do so reading difficulties, including dyslexia, may be prevented, alleviated, or remediated.

Although programs that certify or support teachers, clinicians, or specialists differ in their preparation methodologies, teaching approaches, and organizational purposes, they should ascribe to a common set of professional standards for the benefit of the students they serve. Compliance with these standards should assure the public that individuals who teach in public and private schools, and in clinics, are prepared to implement scientifically based and clinically proven practices.

Background: Why These Standards Are Necessary

Reading difficulties are the most common cause of academic failure and underachievement. The National Assessment of Educational Progress consistently finds that about 31% of all fourth graders read at a level described as "below basic," and only about a third are proficient or advanced. Between 15 and 20% of young students demonstrate significant weaknesses with language processes, including, but not limited to, phonological processing, that are the root cause of dyslexia and related learning difficulties. Of those who are referred to special education services in public schools, approximately 85% are referred because of their problems with language, reading, and/or writing. Informed and effective classroom instruction, especially in the early grades, can prevent and relieve the severity of many of these problems. For those students with dyslexia or language-based learning disorders who need specialized instruction outside of the regular classroom, competent intervention from a specialist can lessen the impact of the disorder and help the student overcome the most debilitating symptoms.

Teaching reading effectively, especially to students who are struggling, requires considerable knowledge and skill. Regrettably, current licensing and professional development practices endorsed by many states are insufficient for the preparation and support of the teachers and specialists who are responsible for enabling *all* students to read and write. Researchers are finding that those individuals with reading specialist and special education licenses often know no more about research-based, effective practices than those individuals with a general education teaching license. The majority of practitioners at all levels have not been prepared in sufficient depth to recognize the early signs of risk, to prevent reading problems, or to teach students with dyslexia and related learning difficulties successfully. Inquiries into teacher preparation in reading have revealed a pervasive absence of substantive content and academic rigor in many courses that lead to the certification of teachers and specialists. Analyses of teacher licensing tests show that, typically, very few are aligned with current research on effective instruction for students at risk. And finally, existing standards for preparation of teachers of reading address literacy very broadly, but in much less detail than is specified here. To address these gaps, CERI has adopted these standards for knowledge, practice, and ethical conduct.

Knowledge and Practice Standards for Teachers of Reading
The International Dyslexia Association

Research-based Assumptions About Reading Difficulties, Including Dyslexia

These standards are broadly constructed to address the knowledge and skill base for teaching reading in preventive, intervention, and remedial settings, especially to students at risk for reading failure. Underlying the standards are assumptions about the nature, prevalence, manifestations, and treatments for reading and writing difficulties that are supported by research and by accepted diagnostic guidelines. These assumptions are as follows:

- Reading skill is distributed on a continuum; students may experience mild, moderate, or severe
 problems with some or all of the essential subskills of reading. Mild and moderate difficulties are most
 likely to be addressed through general education with supplemental small-group support.
- Reading difficulty, especially when manifested early in reading development, involves inaccurate and/or slow recognition of printed words.
- Dyslexia is the appropriate name for disorders of word recognition and spelling that originate from
 core problems in phonological and/or orthographic processing. Many, if not most, students with wordlevel reading and spelling problems will never receive an official diagnosis or be served through special
 education, but they will nonetheless be the responsibility of general education and intervention
 teachers.
- A smaller subgroup of students demonstrates primary difficulties with language comprehension, in conjunction with inadequate word-recognition skills.
- Some students experience a primary problem with the development of fluent, automatic reading, and the slow rate of their reading impairs their academic functioning.
- Different kinds of reading and writing difficulties require different approaches to instruction. One
 program or approach will not meet the needs of all students.
- Although early intervention is the most effective way to prevent and ameliorate learning problems, individuals with dyslexia and other reading difficulties can be helped at any age.
- Dyslexia and related learning difficulties often exist in individuals with aptitudes, talents, and abilities that enable them to be successful in many domains.

Uses for These Standards

The standards outline the (1) content knowledge necessary to teach essential reading and writing skills and strategies to students in general, intervention, and remedial contexts; 2) universal principles and practices of effective instruction; and 3) ethical conduct expected of professional educators and clinicians.

The standards may be used for several purposes, including but not limited to the following:

- educator preparation program accreditation;
- educator certification;
- course design and course sequencing within teacher certification programs;
- delineation of fieldwork requirements and observation checklists; and
- a content framework for the development of licensing or certification examinations.

Knowledge and Practice Standards for Teachers of Reading
©2018, The International Dyslexia Association

How to Read and Cite the Standards

The Knowledge and Practice Standards of Teachers of Reading are organized into five major content areas. Each knowledge standard, in the left column, is numbered to indicate the larger content domain to which it belongs. Examples of how each item within the domain might be observed, tested, or applied are aligned in columns to the right of each standard. The five content domains are as follows:

- Standard 1 addresses foundational concepts about reading development and reading difficulties that are derived from interdisciplinary research.
- Standard 2 covers knowledge of diverse profiles of reading difficulty, including dyslexia, very slow reading, and language comprehension problems.
- Standard 3 pertains to knowledge of assessment relevant to evidence-based practices with a responseto-intervention (RTI) framework.
- Standard 4 addresses Structured Literacy teaching, offering detailed guidance with regard to the
 nature of effective instruction in each major skill domain (phonological sensitivity and phoneme
 awareness, phonics and word recognition, reading fluency, vocabulary, listening and reading
 comprehension, and written expression). Standard 4 also offers guidance regarding expectations for
 teachers engaged in fieldwork or practicum (e.g., in interpretation of assessments, planning
 differentiated instruction, lesson design, corrective feedback, and so forth).
- Standard 5 delineates ethical standards for the profession.

Standards 1, 2, and 3 specify examples of tasks and activities that might demonstrate understanding of the knowledge standard that coursework designers could expect of their students. Standard 4 elaborates the meaning of Structured Literacy instruction by further enumerating examples of the teaching practices that might be expected in a practicum or fieldwork setting. These examples are offered for guidance only; course designers may certainly design other activities and tasks that replace or improve upon those proposed in this document.

When citing the Standards for inclusion on syllabi or training materials, please reference the standard and substandard. For example, KPS 4A.3 (Understand rationale for/Adapt instruction to accommodate individual differences in cognitive, linguistic, sociocultural, and behavioral aspects of learning). The citation for referencing this document follows:

International Dyslexia Association. (2018, March). *Knowledge and Practice Standards for Teachers of Reading*. Retrieved from https://dyslexiaida.org/knowledge-and-practices/

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

	9
	Knowledge and Practice Standards for Teachers of Reading
	Summary Table
	Does Not Include Knowledge and Practice Examples
	Standard I: Foundations of Literacy Acquisition
1.1	Understand the (5) language processing requirements of proficient reading and writing; phonological, orthographic,
	semantic, syntactic, discourse.
1.2	Understand that learning to read, for most people, requires explicit instruction.
1.3	Understand the reciprocal relationships among phonemic awareness, decoding, word recognition, spelling, and vocabulary knowledge.
1.4	Identify and explain aspects of cognition and behavior that affect reading and writing development.
1.5	Identify (and explain how) environmental, cultural, and social factors contribute to literacy development.
1.6	Explain major research findings regarding the contribution of linguistic and cognitive factors to the prediction of literacy outcomes.
1.7	Understand the most common intrinsic differences between good and poor readers (i.e., linguistic, cognitive, and neurobiological).
1.8	Know phases in the typical developmental progression of oral language, phoneme awareness, decoding skills,
	printed word recognition, spelling, reading fluency, reading comprehension, and written expression.
1.9	Understand the changing relationships among the major components of literacy development in accounting for reading achievement.
	Standard 2: Knowledge of Diverse Reading Profiles, Including Dyslexia
2.1	Recognize the tenets of the (2003) IDA definition of dyslexia, or any accepted revisions thereof.
2.2	Know fundamental provisions of federal and state laws that pertain to learning disabilities, including dyslexia and other reading and language disability subtypes.
2.3	Identify the distinguishing characteristics of dyslexia.
2.4	Understand how reading disabilities vary in presentation and degree.
2.5	Understand how and why symptoms of reading difficulty are likely to change over time in response to development and instruction.
	Standard 3: Assessment
3.1	Understand the differences among and purposes for screening, progress-monitoring, diagnostic, and outcome assessments.
3.2	Understand basic principles of test construction and formats (e.g., reliability, validity, criterion, normed).
3.3	Interpret basic statistics commonly utilized in formal and informal assessment.
3.4	Know and utilize in practice well-validated screening tests designed to identify students at risk for reading difficulties.
3.5	Understand/apply the principles of progress-monitoring and reporting with Curriculum-Based Measures (CBMs), including graphing techniques.
3.6	Know and utilize in practice informal diagnostic surveys of phonological and phoneme awareness, decoding skills, oral reading fluency, comprehension, spelling, and writing.
3.7	Know how to read and interpret the most common diagnostic tests used by psychologists, speech-language professionals, and educational evaluators.
3.8	Integrate, summarize, and communicate (orally and in writing) the meaning of educational assessment data for sharing with students, parents, and other teachers.

Knowledge and Practice Standards for Teachers of Reading
©2018, The International Dyslexia Association

	10
	Standard 4: Structured Literacy Instruction
	Substandard A: Essential Principles and Practices of Structured Literacy Instruction
4A.1	Understand/apply in practice the general principles and practices of structured language and literacy teaching, including explicit, systematic, cumulative, teacher-directed instruction.
4A.2	Understand/apply in practice the rationale for multisensory and multimodal language-learning techniques.
4A.3	Understand rationale for/Adapt instruction to accommodate individual differences in cognitive, linguistic,
	sociocultural, and behavioral aspects of learning.
	Substandard B: Phonological and Phonemic Awareness
4B.1	Understand rationale for/identify, pronounce, classify, and compare all the consonant phonemes and all the vowel phonemes of English.
4B.2	Understand/apply in practice considerations for levels of phonological sensitivity.
4B.3	Understand/apply in practice considerations for phonemic-awareness difficulties.
4B.4	Know/apply in practice consideration for the progression of phonemic-awareness skill development, across age and grade.
4B.5	Know/apply in practice considerations for the general and specific goals of phonemic-awareness instruction.
4B.6	Know/apply in practice considerations for the principles of phonemic-awareness instruction: brief, multisensory, conceptual, articulatory, auditory-verbal.
4B.7	Know/apply in practice considerations for the utility of print and online resources for obtaining information about languages other than English.
	Substandard C: Phonics and Word Recognition
4C.1	Know/apply in practice considerations for the structure of English orthography and the patterns and rules that inform
	the teaching of single- and multisyllabic regular word reading.
4C.2	Know/apply in practice considerations for systematically, cumulatively, and explicitly teaching basic decoding and spelling skills.
4C.3	Know/apply in practice considerations for organizing word recognition and spelling lessons by following a structured phonics lesson plan.
4C.4	Know/apply in practice considerations for using multisensory routines to enhance student engagement and memory.
4C.5	Know/apply in practice considerations for adapting instruction for students with weaknesses in working memory, attention, executive function, or processing speed.
4C.6	Know/apply in practice considerations for teaching irregular words in small increments using special techniques.
4C.7	Know/apply in practice considerations for systematically teaching the decoding of multisyllabic words.
4C.8	Know/apply in practice considerations for the different types and purposes of texts, with emphasis on the role of
	decodable texts in teaching beginning readers.
	Substandard D: Automatic, Fluent Reading of Text
4D.1	Know/apply in practice considerations for the role of fluent word-level skills in automatic word reading, oral reading fluency, reading comprehension, and motivation to read.
4D.2	Know/apply in practice considerations for varied techniques and methods for building reading fluency.
4D.3	Know/apply in practice considerations for text reading fluency as an achievement of normal reading development that can be advanced through informed instruction and progress-monitoring practices.
4D.4	Know/apply in practice considerations for appropriate uses of assistive technology for students with serious limitations in reading fluency.
	Substandard E: Vocabulary
4E.1	Know/apply in practice considerations for the role of vocabulary development and vocabulary knowledge in oral and written language comprehension.
4E.2	Know/apply in practice considerations for the sources of wide differences in students' vocabularies.
4E.3	Know/apply in practice considerations for the role and characteristics of indirect (contextual) methods of vocabulary instruction.
4E.4	Know/apply in practice considerations for the role and characteristics of direct, explicit methods of vocabulary instruction.
	Substandard F: Listening and Reading Comprehension
4F.1	Know/apply in practice considerations for factors that contribute to deep comprehension.
4F.2	Know/apply in practice considerations for instructional routines appropriate for each major genre: informational text, narrative text, and argumentation.
4F.3	Know/apply in practice considerations for the role of sentence comprehension in listening and reading comprehension.

Knowledge and Practice Standards for Teachers of Reading
©2018, The International Dyslexia Associatio

	1.
4F.4	Know/apply in practice considerations for the use of explicit comprehension strategy instruction, as supported by
	research.
4F.5	Know/apply in practice considerations for the teacher's role as an active mediator of text-comprehension processes.
	Substandard G: Written Expression
4G.1	Understand the major skill domains that contribute to written expression.
4G.2	Know/apply in practice considerations for research-based principles for teaching letter formation, both manuscript and cursive.
4G.3	Know/apply in practice considerations for research-based principles for teaching written spelling and punctuation.
4G.4	Know/apply in practice considerations for the developmental phases of the writing process.
4G.5	Know/apply in practice considerations for the appropriate uses of assistive technology in written expression.
	Standard 5: Professional Dispositions and Practices
5.1	Strive to do no harm and to act in the best interests of struggling readers and readers with dyslexia and other reading disorders.
5.2	Maintain the public trust by providing accurate information about currently accepted and scientifically supported best practices in the field.
5.3	Avoid misrepresentation of the efficacy of educational or other treatments or the proof for or against those treatments.
5.4	Respect objectivity by reporting assessment and treatment results accurately, and truthfully.
5.5	Avoid making unfounded claims of any kind regarding the training, experience, credentials, affiliations, and degrees of those providing services.
5.6	Respect the training requirements of established credentialing and accreditation organizations supported by CERI and IDA.
5.7	Avoid conflicts of interest when possible and acknowledge conflicts of interest when they occur.
5.8	Support just treatment of individuals with dyslexia and related learning difficulties.
5.9	Respect confidentiality of students or clients.
5.10	Respect the intellectual property of others.

Knowledge and Practice Standards for Teachers of Reading
•©2018, The International Dyslexia Association

	Knowledge and Practice St	andards for Teachers of Reading
	Includes Knowledg	e and Practice Examples
	STANDARD 1: FOUND	TIONS OF LITERACY ACQUISITION
	Substandard	Examples of Coursework Expectations
1.1	Understand the (5) language processing requirements of proficient reading and writing: phonological, orthographic, semantic, syntactic, discourse.	 Explain the domains of language (phonology, morphology, syntax, semantics, pragmatics) and their importance to proficient reading and writing.
1.2	Understand that learning to read, for most people, requires explicit instruction.	• Explain how most people learn to read, how reading acquisition differs from language acquisition, and how writing systems differ from oral language systems.
		 Know that the brain has to establish new neural circuits, linking language and visual regions, to become skilled at reading.
1.3	Understand the reciprocal relationships among phonemic awareness, decoding, word recognition, spelling, and vocabulary knowledge.	 Cite evidence and give practical examples showing how phonemic awareness affects attaining the alphabetic principle, decoding and spelling development, and storage and retrieval of spoken words, and that learning to read affects aspects of language processing, including the extent of phonemic awareness and precision of phonological representations of words in our mental dictionaries.
1.4	Identify and explain aspects of cognition and behavior that affect reading and writing development.	 Cite examples of tasks or tests that measure each general cognitive factor; explain how problems in these areas might be observed in classroom learning.
		 Identify how the following aspects of cognition and behavior affect reading and writing development: attention, automaticity, executive function, verbal memory, processing speed, graphomotor control.
1.5	Identify (and explain how) environmental, cultural, and social factors contribute to literacy development.	 Explain major research findings regarding the contribution of environmental factors to the prediction of literacy outcomes (e.g., language spoken at home, language and literacy experiences, cultural values).
1.6	Explain major research findings regarding the contribution of linguistic and cognitive factors to the prediction of literacy outcomes.	 Identify and explain the contribution of linguistic and cognitive factors to the prediction of literacy outcomes.
1.7	Understand the most common intrinsic differences between good and poor readers (i.e., linguistic, cognitive, and neurobiological).	 Explain the defining characteristics of major types of reading difficulties (i.e., dyslexia, fluency deficits, specific reading comprehension difficulties, mixed reading difficulties).
		 Recognize the major types of reading difficulties when they manifest in a student's developmental history, test performance, and reading behavior.
1.8	Know phases in the typical developmental progression of oral language, phoneme awareness, decoding skills, printed word recognition, spelling, reading fluency, reading comprehension, and written expression.	 Identify the most salient instructional needs of students who are at different points of reading and writing development.
1.9	Understand the changing relationships among the major components of literacy development in accounting for reading achievement.	 Explain the importance of code-emphasis instruction in the early grades and language comprehension once word-recognition skill is established; recognize that vocabulary
		and other aspects of oral language development must be nurtured from the earliest grades through reading aloud and classroom dialogue.

Knowledge and Practice Stan
a d
S.
đ
Teac
F
2
f
teading
•
©2018,
井
-
item
a,
5
5
yslexia
Associ
ation

 research-based differences between good and poor readers, depending on the ki reading disability, with regard to learning word-recognition and decoding skills as compared to listening and reading comprehension. Recognize levels of instructional intensity, frequency, and duration appropriate fo mild, moderate, and severe reading disabilities with the scope of instruction corresponding to the type of reading disabilities with the scope of instruction corresponding to the type of reading difficulties (e.g., dyslexia, specific reading comprehension) to attain catch-up growth and annual growth. Identify how to coordinate regular classroom instruction and other forms of intervention, includin highly specialized settings. Recognize the indicators of a primary disability in reading fluency, including slow processing speed, slow RAN, and nonautomatic word recognition (failure to read words by sight). Recognize how the symptoms of dyslexia or other reading difficulties change as in develops and how instructional priorities and emphases should change according 	2.4 Understand how reading disabilities vary in presentation and degree. 2.5 Understand how and why symptoms of reading difficulty are likely to change over time in response to development and instruction.
 Explain the most fundamental provisions of federal and state laws (IDEA, 504 pertaining to the rights of students with disabilities, especially students' right free, appropriate public education, an individualized educational plan, servic least restrictive environment, and due process. Distinguish IEP goals and objectives that are clear, specific, appropriate to stuneeds, and attainable. Cite research-based prevalence estimates for disorders of word recognition. 	2.2 Know fundamental provisions of federal and state laws that pertain to learning disabilities, including dyslexia and other reading and language disability subtypes. 2.3 Identify the distinguishing characteristics of dyslexia.
 Examples of Coursework Expectations Explain the reasoning or evidence behind key terms in the definition (e.g., neurobiological origin, phonological component of language); distinguish evic based tenets from popular but unsupported beliefs and claims about dyslexia dyslexia is a visual problem; people with dyslexia have unusual talents). 	2.1 Recognize the tenets of the (2003) IDA definition of dyslexia, or any accepted revisions thereof.
SVERSE READING PROFILES, INCLUDING DYSLEXIA	STANDARD 2: KNOWLEDGE OF F

STAND	14 TD 3: ASSESSMENT
Substandard	Examples of Coursework Expectations
3.1 Understand the differences among and purposes for screening, progress- monitoring, diagnostic, and outcome assessments.	 State the major purposes for each kind of assessment and identify examples of each.
3.2 Understand basic principles of test construction and formats (e.g., reliability, validity, criterion, normed).	 Distinguish examples of valid and invalid assessment tools or strategies; demonstrate respect for and fidelity to standardized administration procedures.
3.3 Interpret basic statistics commonly utilized in formal and informal assessment.	 Interpret grade equivalents, age equivalents, normal curve equivalents, percentiles, risk classifications, fluency norms, and standard scores.
	 Recognize the most appropriate types of norm-referenced scores to report and use for interpretation of performance (e.g., percentiles and standard scores rather than grade or age equivalents); interpret grade versus age norms.
3.4 Know and utilize in practice well-validated screening tests designed to identify students at risk for reading difficulties.	 Learn standardized administration of one valid, reliable screening test, administer it to a student or a group of students, and interpret the instructional implications of the results. A valid screening tool that flags students at risk for reading difficulties is likely to selectively, briefly, and efficiently sample subskills such as the following:
	 Letter naming
	 Phoneme isolation and identification, segmentation, blending, and/or manipulation phonics correspondences (sound-symbol relationships)
	 Spelling and phonetic accuracy of spelling attempts
	 Word reading, real and/or nonsense words
	 Oral reading fluency (timed reading of short passages)
	Reading comprehension
3.5 Understand/apply the principles of progress monitoring and reporting with CBMs, including graphing techniques.	 Administer, interpret, and graph or summarize the results of CBMs that directly assess student progress in reading, spelling, and writing and/or the relevant literacy subskills
	that are targeted for instruction.
	 Explain the advantages of CBM for progress monitoring (e.g., ease and speed of
	administration, sensitivity to incremental progress, availability of multiple equivalent
	forms).

ω. 8								3.7													3.6	
Integrate, summarize, and communicate (orally and in writing) the meaning of educational assessment data for sharing with students, parents, and other teachers.							psychologists, speech-language professionals, and educational evaluators.	Know how to read and interpret the most common diagnostic tests used by												comprehension, spelling, and writing.	Anow and utilize in practice informal diagnostic surveys of phonological	
 Explicitly link information from screenings, diagnostic surveys, progress monitoring, and descriptive data to instructional decisions governing the content, entry point, pace, intensity, student grouping, and methods for literacy intervention. 	 Woodcock-Johnson Tests of Cognitive Ability and Achievement (WJR) 	 Test of Word Reading Efficiency (TOWRE) 	 Peabody Picture Vocabulary Test (PPVT) Rapid Automatic Naming Test (RAN) 	 Comprehensive Test of Phonological Processing (C-TOPP) 	 Clinical Evaluation of Language Fundamentals (CELF) 	following:	licensed examiners, including current versions of these instruments, such as the	 Understand and use relevant information from formal assessments administered by 	 Writing performance (punctuation, capitals, syntax, organization, content, spelling, vocabulary) 	 Automatic recognition of high-frequency words 	 Morpheme recognition, interpretation, and spelling 	 Listening comprehension and recall 	 Silent passage reading comprehension and recall 	 Oral passage reading fluency and comprehension 	reading	 Phonics and application of introductory and advanced phonics to spelling and word 	 Accuracy and fluency of letter naming, letter formation, alphabet knowledge 	later)	 Phonological sensitivity (in preschool) and phonemic awareness (in kindergarten and 	inventories for the purpose of purpointing a student's strengths, weaknesses, and instructional needs in the following areas:	 Administer and interpret informal (e.g., not norm-referenced) diagnostic surveys and 	15
~																						
-------	--																					
Mou																						
ē																						
ē																						
đ																						
ad																						
R																						
đ																						
dan.																						
is fo																						
2																						
Re																						
ġ,																						
20																						
R																						
nter																						
nati																						
2																						
2																						
exia																						
Ass																						
g																						

and phonemic awareness.		
 Know a variety of activities for each level of phonological 		
level when mastery is attained on the prior phase.		
student's level of PA development, proceeding to the next	knowledge as the student progresses.	development, across age and grade.
 Select and implement PA activities that correspond with a 	 Plan to link phoneme knowledge with letter (grapheme) 	phonemic-awareness skill
phonological and phonemic-awareness instruction.	phonemic-awareness skills as related to student grade levels.	consideration for the progression of
 Plan and deliver a scope and sequence of systematic 	 Identify the common progression of phonological and 	4B.4 Know/apply in practice
speech sound, not the letter name for spelling a phoneme.	produced and sound.	
 For phonemic-awareness instruction, clearly focus on the 	coarticulation, that alter how certain phonemes are	
nondistorted phonemes (no coarticulation effect).	sounds in natural speech), often resulting from	
 Select key words to illustrate each phoneme that feature 	 Identify common allophonic variations (changes of speech 	awareness difficulties.
minimize coarticulation effects.	phonemic-awareness tasks (e.g., coarticulation effect).	considerations for phoneme
 When introducing a phoneme, select word examples that 	 Identify reasons why students may experience difficulty with 	4B.3 Understand/apply in practice
and rime.		
simple onset that has only one phoneme, blending onset		
sound matching, first sound segmentation) in words with a	 Isolate a given phoneme in a spoken word. 	
recognition and rhyme production, syllable counting, first	 Identify the number of phonemes in a spoken word. 	
early, basic phonological-sensitivity skills (e.g., rhyme	svllables (e.g., mv/pie; mountain/fountain).	
 Know activities that would help children acquire these 	 Recognize and generate rhymes of words with one or more 	
fan/seat vs. fan/pin).	 Blend and segment onset-rime units in one-syllable words. 	
 Choose wide contrasts for beginning rhyme tasks (e.g., 	multisyllabic words.	
rhyming word).	 Identify, count, and separately pronounce the syllables in 	phonological sensitivity.
any phonological-sensitivity lesson (syllable, onset-rime,	phonological sensitivity and provide examples of each.	considerations for levels of
 Explicitly and accurately label the linguistic unit of focus in 	 Explain the general developmental progression of 	4B.2 Understand/apply in practice
represent the phonemes.		
awareness tasks whose spellings do not transparently		
reinforce attention to sound by using words in phonemic-		
knowledge to perform a phonemic-awareness task,		
 For students who may be relying on spelling or letter 		
development.	transparently represent the phoneme (e.g., dogs, sure, ink).	
on the students' phase of phonemic-awareness	 Identify phonemes in words in which the spelling does not 	
/m/, /n/) phoneme contrasts during instruction, depending	sound similar.	
 Deliberately choose wide (e.g., /m/, /z/) or narrow (e.g., 	each other because they share articulatory features and thus	
sound is produced.	 Identify phonemes that are more likely to be confused with 	
gesture or mouth picture to illustrate the way the speech	phonemic awareness, and phonics.	the vowel phonemes of English.
using a mirror, describing the speech sound, or using a hand	 Explain the difference between phonological awareness, 	all the consonant phonemes and all
words during PA lessons by such techniques as modeling,	of alphabetic writing systems.	pronounce, classify, and compare
 Explicitly teach articulatory features of phonemes and 	 Discuss why phonemic awareness is necessary for learners 	4B.1 Understand rationale for/identify,
Examples of Flacticalit of Fleidwork Expectations	Examples of coalsework Expectations	Jupstalluaru
Examples of Dracticum or Eieldwork Expectations	Examples of Coursework Expectations	Substandard
	STANDARD 4: STRUCTURED LITERACY INSTRUCTION B: PHONOLOGICAL AND PHONEMIC AWARENESS	
1/		

/sh/ or with /ch/).		
Spanish, classifying spoken words in English as starting with		
similarly articulated phonemes (e.g., for children who speak	distinguish and produce.	languages other than English.
 Provide practice distinguishing the new phoneme from 	or dialect and are likely to be challenging for the learner to	obtaining information about
language.	sounds in English are not in the student's native language	print and online resources for
nonstandard dialect user may not have in his or her first	with Standard American English to anticipate which speech	considerations for the utility of
 Explicitly teach the phonemes of English that the EL or 	 Compare a student's first language phonological system 	4B.7 Know/apply in practice
	instruction after first grade.	
	goals) in K-1 and for students who need remedial	articulatory, auditory-verbal.
advanced PA activities as appropriate.	for 15–20 weeks (or more, as needed, to reach curricular	brief, multisensory, conceptual,
motions in learning a variety of early, basic, and more	structured literacy classroom teaching and/or intervention	phonemic-awareness instruction:
boxes, body mapping, finger tapping, and left-to-right hand	multisensory phonemic-awareness activities during	considerations for the principles of
 Use tactile and kinesthetic aids, such as blocks, chips, sound 	 Plan to provide brief (5–10 minute), distributed, 	4B.6 Know/apply in practice
	that represents that short vowel in print.	awareness instruction.
	one-syllable words a prerequisite for learning the letter	specific goals of phonemic-
reading, spelling, and vocabulary instruction.	example, making identification of a short vowel in spoken	considerations for the general and
Routinely incorporate phonemic-awareness instruction into	 Align PA instruction to reading and spelling goals, for 	4B.5 Know/apply in practice
		Ĩ

~	
P	
ġ.	
ă.	
Tiop P	
£	
ā.	
an a	
4	
Ξ.	
B	
2	
÷.	
B	
ġ.	
2	
200	
7	
P	
Ð	
2	
ō	
ы Ш	
š	
P	
a)	
2	
ŏ.	
5	

5	
2	
۰.	
ς.	
9	
2	
5	
2	
2	
٤.	
t.	
5	
2	
ξ.	
2	
2	
į.	
ł.	
2	
ł.	
3	
è.	
į.	
١.	
١.	
Ð	
3	
<u>.</u>	
2	
5	
5	
2	
0	
4	
τ	
÷	
ţ.	
;	
3	
t.	
5	
)	
2	
i.	
2	
ŀ.	
2	
ĩ	
5	
ŀ.	
٤.	
п.	

1.7 Know/apply in practice Teach written syllable types in a logical sequence (e.g., considerations for systematically teaching the decoding of multisyllabic r-controlled). Identify the difference between syllable division in natural spech and syllable division in printed words. Clearly distinguish morphemes from syllables while identifying word parts. Each additional strategies for distinguish division in printed words. Explicitly teach written syllable division in natural spech and syllable division in printed words. Clearly distinguish morphemes from syllables while identifying the pronound prefixes, and flexing the decode definition, definitive) if necessarial specification of the decode definition. Explicitly teach students how to to support multisyllable word resplicitly teach students how to the meaning of common on prefixes, and flexing the pronound prefixes. Each the meaning of common on the pronound prefixes. Each the division in the decode definition, definitive) if necessarial strategies for distribution. 	4C.6 Know/apply in practice Distinguish among high-frequency regular/ irregular Introduce high-frequency words (bo words in small increments using special techniques. Define sight words in relation to regular/ irregular words. Place words on a continuum of fully predictable, partially or conditionally pattern-based, and unique (not belonging to a word family). Identify/describe the three factors to consider when determining how to introduce irregular words within a reading program (word frequency, word similarity, word Identify/lescribe the three factors to consider when areading program (word frequency, word similarity, word Identify. Teach truly irregular word's spelling selling regularities, word's spelling. Teach truly irregular word's spelling. Teach truly irregular word's spelling. 	4C.5 Know/apply in practice • Identify how instruction can be modified to increase • Adapt the pace, format, content, s attention, support instruction for students with weaknesses in working memory, attention, executive function, or processing speed. • Identify how instruction can be modified to increase • Adapt the pace, format, content, s	4.2.4 Know Jappy in practice Plan to incorporate multisensory simultaneously employing two or three modalities, including sound-blending techniques cued looking, listening, speaking, touching, moving). Pluently manage and manipulate materials, such as alphabet arcs, s grapheme tiles. Employ signals, so student responses during phonen activities. Builtaneously employing two or three modalities, including objects (chips, tiles, etc.). Pluently manage and manipulate materials, such as alphabet arcs, s student responses during phonen activities. A student responses during phonen activities.
--	--	--	---

Knowledge and Practice Standards for Teachers of Reading
• ©2018. The International Dyslexia Association

Knowle
ğ
and
Practice
Stand
ards f
9
eache
ers of
Reading
• ©201
8, T
፳
Ξ
Interna
Internationa
International Dy
International Dyslexia /
International Dyslexia Associatio

4D.2 Know/apply in practice considerations inde- for varied techniques and methods for read- building reading fluency. flue Des incl sim reading fuency. end sim reading fuency. end sim reading sim sim sim sim sim sim sim sim sim sim	Substandard • Exp 4D.1 Know/apply in practice considerations for the role of fluent word-level skills in automatic word reading, oral reading fluency, reading comprehension, and motivation to read. • Exp • Exp • Exp • read. • Exp • read. • Exp • read. • Exp • read. • Exp • sile • Exp	4C.8 Know/apply in practice • Describ considerations for the different types high-fre and purposes of texts, with emphasis identify on the role of decodable texts in words (in the role of decodable texts in words (in the role of decodable texts) teaching beginning readers. • Analyze edcodal nondec calculat the text the text
scribe the role of and appropriate use of ependent silent reading, assisted reading, repeated ding, and integrated fluency instruction to promote ant reading of text. scribe and role-pial fluency-building techniques, luding brief speed drills, phrase-cued reading, ultaneous oral reading, alternate oral reading, and ultaneous oral reading, alternate oral reading and ultaneous oral reading structure or the structure of the speed ultaneous or the structure of the	D: AUTOMATIC, FUEKIT KRADING OF FERT Examples of Coursework Expectations Jain why all component skills for reading relopment must become accurate and rapid to port more advanced reading skills (e.g., knowledge etter names/sounds, phonemic awareness, coding). Jain how phoneme-grapheme mapping underpins development of accurate, automatic word ognition, oral reading fluency, vocabulary, and ognition, oral reading fluency, vocabulary, and nr reading comprehension.	e how decodable texts differ from predictable and squency word texts in structure and purpose. "And define word types: wholly decodable words, ir words (previously taught), and nondecodable not wholly decodable or previously taught). 2 a decodable text to identify word types (wholly bile, introduced high-frequency words, odable words), and list words identified by type; te percentage of each type of word present in t. STANDARD 4: STRUCTURED UTERACY INSTRUCTION
 Define and identify examples of text at a student's frustration, instructional, and independent reading levels; recognize how requirements for word accuracy in instructional and independent reading increase by grade. Provide ample opportunities for student(s) to read connected text daily, with appropriate feedback on decoding errors. Guide the student to correct his or her reading errors, even when contextually appropriate. Incorporate fluency-building routines and activities into reading, simultaneous oral reading, alternate oral reading, and/or repeated readings, time limits, and scaffolds to enable student success and progress. 	 Examples of Practicum or Fieldwork Expectations Select and use fluency-building routines and activities for both automatic application of literacy subskills and for text reading, as appropriate. Identify relevant apps or computer games for building automaticity in word recognition. Choose instructional materials to build automaticity in subskills/practice reading texts of appropriate difficulty. 	 Effectively develop or select, and utilize, decodable texts to support developing readers in applying taught phonics concepts in context. Select instructional-level texts for student reading that correspond to the content and purpose of students' reading skill lessons. Discern texts that do not support decoding lessons because they contain too many untaught word patterns and high-frequency words.

4E.2 Know/apply in practice considerations for the sources of wide differences in students' vocabularies.	E.1. Know/apply in practice considerations for the role of vocabulary development and vocabulary knowledge in oral and written language comprehension.	Cibetandard	4D.4 Know/apply in practice considerations for appropriate uses of assistive technology for students with serious limitations in reading fluency.	4D.3 Know/apply in practice considerations for text reading fluency as an achievement of normal reading development that can be advanced through informed instruction and progress-monitoring practices.
 Identify the intrinsic and extrinsic (environmental) factors that are causally related to vocabulary growth, including adult-child interaction patterns; school, socioeconomic, and community contexts; first language other than English; and neurodevelopmental differences in language processing. Discuss the vocabulary gap in root word knowledge and the implications for vocabulary instruction. 	 Identify and summarize the evidence that knowledge of word meanings is a major factor in language comprehension and expression. Summarize the findings of the National Reading Panel, the National Early Literacy Panel, and current IES Practice Guides with regard to vocabulary instruction. Identify and discuss the classroom indicators of students' vocabulary strengths and weaknesses, such as limited range of word use, confusion about multiple meanings of words, lack of understanding of idioms, slow word retrieval, and poor-quality definitions. 	STANDARD 4: STRUCTURED LITERACY INSTRUCTION E. VOCABULARY Evamples of Coursework Experiations	 Locate and access assistive technology for students with serious limitations in reading fluency. 	 Identify reading subskills that may be appropriate for brief speed drills (e.g., letter naming, word reading, symbol-sound recall) Identify and define the components of passage reading fluency (accuracy, rate, prosody). Interpret CBMs, including oral-reading fluency norms, to develop fluency-building goals with students.
 Include at least an informal assessment of student vocabulary in screening; refer for speech/language assessment when appropriate. Choose reading materials (read aloud and student reading) that expand vocabulary knowledge. 	 Habitually include vocabulary-building activities and routines during all instruction. Recognize when a particular vocabulary-building activity (e.g., morphemic analysis, contextual analysis) is more or less appropriate depending on the word being taught. 	Examples of Drarticum or Fieldwork Expectations	 Support students in learning to use assistive technology, such as print-to-speech translators, apps, e-books, and audiobooks. 	 Select, administer, and graph appropriate curriculum- based measures of relevant reading subskills. Effectively administer, score, and interpret an oral- reading fluency curriculum-based measure (CBM). Rate the prosodic quality of a student's oral reading. Develop fluency goals and objectives with students and involve students in graphing progress toward those goals.

Knowledge and Practice Standards for Teachers of Reading
©2018. The International Dvslexia Association

andards for Teachers of Reading $ullet$ @2018, The International Dyslexia Associatic	Knowledge and Practice St	
Feachers of Reading ●©2018, The International Dyslexia Associatic	andards for 1	
Reading •©2018, The International Dyslexia Associatic	Feachers of	
32018, The International Dyslexia Associatic	Reading • (
e International Dyslexia Associatic	92018, The	
Dyslexia Associatio	International I	
0)yslexia Associati	

ce considerations
modeling, classroom conversation wide independent reading, indep learning strategies, and word play
narize evidence tha sroom conversation dent reading, indep igies, and word play igies, and word play
 Promote a rich lang quality language in: children's literature activities involving v language use. Plan and deliver less shared storybook pr focus on vocabulary Know the shortcomi
 Promote a rich language environment by sc. quality language in student dialogue, readin children's filterature aloud, engaging student activities involving vocabulary, and modelin language use. Plan and deliver lessons that involve evidenc shared storybook practices, such as Dialogic focus on vocabulary and language enrichme Know the shortcomings for vocabulary build

Substandard Examples of Coursework Expectations Examples of Practicum or Fieldwork
4F.1 Know/apply in practice considerations Articulate a framework for comprehension instruction for factors that contribute to deep including background knowledge, vocabulary, verbal reasoning ability, sentence processing, knowledge of literary structures and conventions, and skills and strategies for close reading of text. Plan and deliver comprehension instruction comprehension instruction interpretation of vocabulary and academ text structure using strategies that fit the
4F.2 Know/apply in practice considerations Contrast the characteristics of the major text genres, for instructional routines appropriate for each major genre: informational Identify text features that characterize each major genre, including logical organization, typical connecting matched to specific informational text st argumentation. Identify text features that characterize each major genre, including logical organization, typical connecting matched to specific informational text st argumentation. Identify text features that characterize each major genre, including logical organization, typical connecting matched to specific informational text st or signal words, and style of language. Match graphic organizers, titles, and topic sentences to various text structures (e.g., description, compare/contrast, reason/evidence, time sequence). Explicitly teach story grammar and use it compare/contrast. Teach students to recognize and interpresent associated with specific informational and use it compare/contrast. Explicitly teach story grammar and use it compare/contrast. Explicitly teach story grammar and use it compare/contrast. Explicitly teach story grammar and use it compare/contrast.
4F.3 Know/apply in practice considerations Define and distinguish among phrases, dependent Teach students how to construct and decomprehension in listening and Know techniques of explicit instruction with sentences, such as sentence elaboration, sentence paraphrase, identifying the function of words within a sentence, and sentence south as sentence, and sentence south as sentence, and sentence south as sentence, and sentence structures in any text that may pose comprehension challenges, such as subject and verb. Teach students how to construct and decomprehension with sentence and sentence elaboration, sentence and sentence and sentence senter the function of words within a sentence, and sentence structures in any text that may pose comprehension challenges, such as figurative language, double negatives, passive voice, embedded clauses, anaphora, and distance between subject and verb. During an oral reading of text, detect an appropriate feedback to students' confur comprehension. Define and distinguish among phrases, dependent Define and distance between During an oral reading of text, detect an appropriate in the students' confur comprehension. During an oral reading of text, detect and appropriate in the students' confur comprehension.

Knowledge
and F
ractice
Standard
sfor
Teachers of
Reading
 ©2018,
The Int
ternational [
) Vslexia
Association

y are most likely offic text, plan w ght be taught, it aphic represent ghting, and sol evels of compro- evels of compro- evels of compro- the surface cod evels of compro- the surface cod evels of compro- the mental moo- heir connection heir connection heir connection heir connection the mental moo- heir connection fic text, identifi fic text, generat s construct a m s construct a m s construct a m

4G.4 Know/apply in practice considerations for the developmental phases of the writing process.	4G.3 Know/apply in practice considerations for research-based principles for teaching written spelling and punctuation.	4G:2 Know/apply in practice considerations for research-based principles for teaching letter formation, both manuscript and cursive.
 Identify the specific subskills of each phase of the writing process so each can be explicitly taught (e.g., planning involves selecting a format, having ideas, and having a goal; drafting requires transcription skill and text/word generation; reviewing requires facility with word choice, sentence editing, mechanics, audience awareness, and so forth). Identify research-based instructional practices to support planning, drafting, and revision. 	 Recognize and explain the influences of phonological, orthographic, and morphemic knowledge on spelling, so instruction will focus on language structures rather than rote memorization. Identify students' levels of spelling development and orthographic knowledge according to a developmental framework. Identify a progression for teaching punctuation that is related to instruction on phrase and sentence structure and sentence types. Analyze student writing samples and spelling tests to refine instructional targets (e.g., development of phonological awareness, knowledge of spelling rules, awareness of inflectional morphemes). 	 Identify and rehearse techniques for building handwriting control and legibility, including modeling basic strokes, using verbal descriptions of motor patterns, using numbered arrows, and using appropriate writing implements, posture supports, and paper. Identify and rehearse techniques for building writing fluency.
 Devote sufficient instructional time to planning, including definition of the goal and expectations, brainstorming of ideas, and anticipation of text format, length, and style. Support transcription with written notes, word banks, graphic organizers, and talking. Support editing and revision with personal or group conferencing, proofreading checklists, and peer-to-peer collaboration. Build a student writing folder and publish selected works in displays or collections. 	 Select instructional targets that match students' levels of spelling development and that follow a scope and sequence of spelling concepts. Explicitly teach spelling concepts (explain concept, lead practice with feedback, support independent practice). Use or develop practice activities that help students generalize learned words and patterns into writing. Identify helpful apps and other technology that support practice support hat would be appropriate for accommodations and modifications. 	 Use multisensory techniques (e.g., saying and writing together) to teach letter formation. Group letters for practice that require similar motor patterns, and explicitly teach those basic pencil strokes. Model letter formation with visual, motor, and verbal support, lead supervised practice, and provide extended practice with feedback. Adapt instruction and writing materials for left-handed students. Build fluency in letter formation, copying, and transcription through frequent, distributed practice and brief timed activities.

		 Critically evaluate specific assistive technology devices/programs and their utility for a specific student. 	
	adjustments depending on individual students' needs.	with organization/composition).	expression.
	 Implement assistive technology for writing; make 	written expression needs (e.g., poor spelling vs. difficulties	use of assistive technology in written
	processing instruction as appropriate.	devices/programs) appropriate to students with varying	considerations for the appropriate
	 Select and provide access to keyboarding and word- 	 Provide examples of specific assistive technology (types of 	4G.5 Know/apply in practice
27		-	

5.10	5.9	5.8	5.7	5.6	5.5	5.4	5.3	5.2	5.1	
lespect the intellectual property of others.	lespect confidentiality of students or clients.	upport just treatment of individuals with dyslexia and related learning difficulties.	wold conflicts of interest when possible and acknowledge conflicts of interest when they occur.	Respect the training requirements of established credentialing and accreditation organizations supported by CERI and IDA.	Avoid making unfounded claims of any kind regarding the training, experience, credentials, affiliations, and degrees of those providing services.	Respect objectivity by reporting assessment and treatment results accurately, and truthfully.	Avoid misrepresentation of the efficacy of educational or other treatments or the proof for or against those treatments.	Maintain the public trust by providing accurate information about currently accepted and scientifically supported best practices in the field.	Strive to do no harm, maintain confidentiality, and act in the best interest of struggling readers and readers with dyslexia and other reading disorders.	STANDARD 5: PROFESSIONAL DISPOSITIONS AND PRACTICES

SUPPORTING EVIDENCE AND REFERENCES¹

STANDARD 1: Foundations of Literacy Acquisition

An extensive research base exists on the abilities that are important in learning to read and write, including how these abilities interact, how they are influenced by experience and instruction, and how the relative importance of various abilities tends to shift across development. Even before formal literacy instruction begins, certain risk indicators, such as poor phonological sensitivity or a history of early language delay, can predict which children are likely to require especially close monitoring and intervention. Moreover, evidence suggests that certain educational practices, such as universal screening, evidence-based general education instruction, and prompt intervention, can prevent or ameliorate many literacy problems. For both general and special educators, knowledge of this research base on literacy development and literacy difficulties forms an essential foundation for the competencies and skills described in subsequent sections of this document.

In addition, familiarity with the systems of language is required to implement Structured Literacy instruction. Formal knowledge about language structures—recognizing, for example, whether words are phonetically regular or irregular; common morphemes in words; common sentence structures in English; and how different types of texts are organized—enables teachers to interpret assessments, present lesson concepts clearly, select appropriate examples of concepts, and provide corrective feedback to students. Teachers' understanding of language structure is essential to providing effective instruction in writing and reading. Research suggests that acquiring an understanding of language structure often requires explicit teaching of this information and more than superficial coverage in teacher preparation and professional development.

- Brady, S., Gillis, M., Smith, T., Lavalette, M., Liss-Bronstein, L., Lowe, E., et al. (2009). First grade teachers' knowledge of phonological awareness and code concepts: Examining gains from an intensive form of professional development. *Reading and Writing: An Interdisciplinary Journal*, 22, 375–510.
- Adams, M. (1990). Beginning to read: Learning and thinking about print. Cambridge, MA: MIT Press. Bickart, T. (1998). Summary report of preventing reading difficulties in young children (National Academy of Sciences). Washington, DC: U.S. Department of Education.
- Carlisle, J. F., Kelcey, B., & Berebitsky, D. (2013). Teachers' support of students' vocabulary learning during literacy instruction in high poverty elementary schools. *American Educational Research Journal*, 50, 1360-1391.
- Crawford, E. C., & Torgesen, J. K. (2006, July). *Teaching all children to read: Practices from Reading First schools with strong intervention outcomes*. Presented at the Florida Principal's Leadership Conference, Orlando. Retrieved from

http://www.fcrr.org/science/sciencePresentationscrawford.ht ningham A. F. & Stanovich K. F. (1997). Facty reading acquisition and its

Cunningham, A. E., & Stanovich, K. E. (1997). Early reading acquisition and its relation to reading experience and ability ten years later. *Developmental Psychology*, 33, 934–945.

Denton, C. A., Fletcher, J. M., Anthony, J. L., & Francis, D. J. (2006). An evaluation of intensive intervention for students with persistent reading difficulties. *Journal of Learning Disabilities*, 39, 447–466.

¹ This reference list is neither comprehensive nor exhaustive. References are offered as examples of the literature supporting the rationale for and the validity of each standard. Many other sources could be referenced in courses and texts aligned with this document.

Knowledge and Practice Standards for Teachers of Reading
• ©2018, The International Dyslexia Association

Fletcher, J. M., Lyon, G. R., Fuchs, L. S., & Barnes, M. A. (in press). Learning disabilities: From identification to Intervention, 2nd Ed. New York: Guilford Press. 29

Foorman, B., Beyler, N., Borradaile, K., Coyne, M., Denton, C. A., Dimino, J., et al. (2016). Foundational skills to support reading for understanding in kindergarten through 3rd grade (NCEE 2016-4008).
 U.S. Department of Education. Institute of Education Sciences, National Center for Education Evaluation and Regional Assistance. Washington, DC: Government Printing Office.

Hudson, R. R., High, L., & Al Otaiba, S. (2007). Dyslexia and the brain: What does current research tell us? The Reading Teacher, 60(6), 506–515.

McCardle, P., & Chhabra, V. (2004). The voice of evidence in reading research. Baltimore: Brookes Publishing.

McCutchen, D., Green, L., Abbott, R. D., & Sanders, E. A. (2009). Further evidence for teacher knowledge: Supporting struggling readers in grades three through five. *Reading and Writing: An Interdisciplinary Journal*, 22, 401-423.

Moats, L. C. (2010). Speech to print: Language essentials for teachers, 2nd Ed. Baltimore: Brookes Publishing.

Mol, S. E., & Bus, A. G. (2011). To read or not to read: A meta-analysis of print exposure from infancy to early adulthood. *Psychological Bulletin*, 137, 267–296.

National Reading Panel. (2000). Teaching children to read: An evidence-based assessment of the scientific research literature on reading and its implications for reading instruction. Washington, DC: National Institutes of Health.

Olson, R. K. (2004). SSSR, environment, and genes. Scientific Studies of Reading, 8(2), 111-124.

Seidenberg, M. (2017). Language at the speed of sight: How we read, why so many can't, and what can be done about it. New York, NY: Basic Books.

Snow, C., Griffin, P., & Burns, S. (2006). *Knowledge to support the teaching of reading*. San Francisco: Jossey- Bass.

Spear-Swerling, L., & Brucker, P. (2004). Preparing novice teachers to develop basic reading and spelling skills in children. Annals of Dyslexia, 54, 332–364.

Spear-Swerling, L., & Cheesman, E. (2012). Teachers' knowledge base for implementing response-tointervention models in reading. *Reading & Writing: An Interdisciplinary Journal*, 25, 1691–1723.

Stanovich, K. E. (2000). *Progress in understanding reading: Scientific foundations and new frontiers*. New York: Guilford Press.

Stone, A. C., Silliman, E. R., Ehren, B. J., & Apel, K. (Eds.). (2014). Handbook of language and literacy: Development and disorders, 2nd Ed. New York: Guilford Press.

Vellutino, F. R., Tunmer, W. E., Jaccard, J. J., & Chen, R. (2007). Components of reading ability: Multivariate evidence for a convergent skills model of reading development. *Scientific Studies of Reading*, 11(1), 3– 32.

Wolf, M. (2007). Proust and the squid: The story and science of the reading brain. New York: HarperCollins.

Zipoli, R., & Merritt, D. (2016). Risk of reading difficulty among children with a history of speech or language impairment: Implications for student support teams. *Preventing School Failure: Alternative Education* for Children and Youth. doi:10.1080/1045988X.2016.1202180

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

STANDARD 2: Knowledge of Diverse Reading Profiles, Including Dyslexia

A well-prepared teacher will expect that students' reading, writing, and language profiles will vary and that a single approach to instruction is unlikely to match the needs of all students. Some students will learn readily and will benefit from more emphasis on centers and independent reading. Some students will have specific problems learning phonemic awareness, phonics, spelling, and decoding, whereas other students may be relatively strong at using phonics but relatively weak in vocabulary, language comprehension, or the text generation aspects of writing. Still others may have a specific and pronounced problem developing automatic recognition of words and may be very slow readers. These subgroups can be supported with small-group instruction delivered under an RTI (MTSS) framework that bolsters their weaker skill areas.

To identify children with dyslexia and other learning disabilities, teachers must understand and recognize the key symptoms of these disorders and how the disorders differ. To plan instruction and detect older students with learning disabilities who may have been overlooked in the early grades, teachers should also understand how students' difficulties may change over time, based on developmental patterns, experience, and instruction, along with increases in expectations across grades.

- Aaron, P. G., Joshi, R. M., Gooden, R., & Bentum, K. (2008). Diagnosis and treatment of reading disabilities based on the component model of reading: An alternative to the discrepancy model of LD. *Journal of Learning Disabilities*, 41, 67–84.
- Catts, H. W., Adlof, S. M., & Weismer, S. E. (2006). Language deficits in poor comprehenders: A case for the simple view of reading. *Journal of Speech, Language, and Hearing Research, 49*(2), 278–293.
- Catts, H. W., Compton, D. L., Tomblin, J. B., & Bridges, M. S. (2012). Prevalence and nature of late-emerging poor readers. *Journal of Educational Psychology*, 104(2), 166–181.
- Compton, D. L., Fuchs, L. S., Fuchs, D., Lambert, W., & Hamlett, C. (2012). The cognitive and academic profiles of reading and mathematics learning disabilities. *Journal of Learning Disabilities*, 45(1), 79–95.
- Ehri, L. C., Cardoso-Martins, C., & Carroll, J. M. (2014). Developmental variation in reading words. Science Direct, 35(5), 1098-1109.
- Fletcher, J. M., Lyon, G. R., Fuchs, L. S., & Barnes, M. A. (in press). Learning disabilities: From identification to intervention, 2nd Ed. New York: Guilford Press.
- Hulme, C., & Snowling, M. (2009) Developmental disorders of language, learning, and cognition. Oxford, England: Wiley-Blackwell.
- Kieffer, M. J. (2010). Socioeconomic status, English proficiency, and late-emerging reading difficulties. Educational Researcher, 39, 484–486.
- Leach, J. M., Scarborough, H. S., & Rescorla, L. (2003). Late-emerging reading disabilities. Journal of Educational Psychology, 95, 211–224.
- Lipka, O., Lesaux, N., & Siegel, L. (2006). Retrospective analyses of the reading development of grade 4 students with reading disabilities: Risk status and profiles over 5 years. *Journal of Learning Disabilities*, 39, 364–378.

Lyon, R., Shaywitz, S., & Shaywitz, B. (2003). A definition of dyslexia. Annals of Dyslexia, 53, 1–14.

- Moats, L. C., & Dakin, K. (2007). Basic facts about dyslexia. Baltimore: The International Dyslexia Association.
- Shaywitz, S. (2003). Overcoming dyslexia: A new and complete science-based program for reading problems at any level. New York: Knopf.
- Spear-Swerling, L. (2015). The power of RTI and reading profiles: A blueprint for solving reading problems. Baltimore, MD: Brookes Publishing.
- Washburn, E. K., Joshi, R. M., & Binks-Cantrell, E. S. (2011). Teacher knowledge of basic language concepts and dyslexia. *Dyslexia*, 17, 165–183.

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

Wolf, M., & Bowers, P. G. (1999). The double-deficit hypothesis for the developmental dyslexias. *Journal of Educational Psychology*, *91*, 415–438.

STANDARD 3: Assessment

Teachers' ability to administer and interpret assessments accurately is essential both to the early identification of students' learning problems and to planning effective instruction. Appropriate assessments enable teachers to recognize early signs that a child may be at risk for dyslexia or other learning disabilities, and the assessments permit teachers to target instruction to meet individual students' needs. Teachers should understand that there are different types of assessments for different purposes (e.g., brief but frequent assessments to monitor progress versus more lengthy, comprehensive assessments to provide detailed diagnostic information) and be able to recognize which type of assessment is called for in a particular situation. Teachers need to know where to find unbiased information about the adequacy of published tests: to interpret this information correctly, they require an understanding of basic principles of test construction and concepts such as reliability and validity. They should also understand how an individual student's component profile may influence his or her performance on a particular test, especially on broad measures of reading comprehension and written expression. For example, a child with very slow reading is likely to perform better on an untimed measure of reading comprehension than on a stringently timed measure; a child with writing problems may perform especially poorly on a reading comprehension test that requires lengthy written responses to open-ended questions. In addition, to implement assessments effectively within an RTI (MTSS) framework, educators must understand certain issues involved in screening and progress monitoring large groups of students. These issues include the value of two-stage screening, appropriate selection and interpretation of progress-monitoring assessments, and signs that a student should be referred for comprehensive evaluation for special education (e.g., early language delay or family history of dyslexia, in a student who is not showing a robust response to intervention). Because fluency is a useful predictor of overall reading competence, especially in elementary-aged students, a variety of fluency tasks have been developed for use in screening and progress monitoring, most notably in CBMs. General and special educators should know how CBMs differ from other types of curriculum-based assessments (e.g., they are quick-timed probes that correlate well with overall competence in a domain), and they should recognize the features that make CBMs particularly useful in screening and progress monitoring (e.g., they come in multiple equivalent forms and are sensitive to incremental progress).

- Compton, D. L., Fuchs, D., Fuchs, L. S., Bouton, B., Gilbert, J. K., Barquero L. A., & Crouch R. C. (2010). Selecting at-risk first-grade readers for early intervention: Eliminating false positives and exploring the promise of a two-stage gated screening process. *Journal of Educational Psychology*, 102, 327–341.
- Cutting, L. E., & Scarborough, H. S. (2006). Prediction of reading comprehension: Relative contributions of word recognition, language proficiency, and other cognitive skills can depend on how comprehension is measured. *Scientific Studies of Reading*, 10, 277–299.
- Farrall, M. L. (2012). *Reading assessment: Linking language, literacy, and cognition*. Hoboken, NJ: John Wiley & Sons.
- Fuchs, D., Fuchs, L. S., & Compton, D. L. (2012). Smart RTI: A next-generation approach to multilevel prevention. *Exceptional Children*, 78, 263–279.
- Fuchs, L. S., & Vaughn, S. (2012). Responsiveness to intervention: A decade later. Journal of Learning Disabilities, 45, 195–203.
- Good, R. H., Simmons, D. C., & Kame'enui, E. J. (2001). The importance and decision-making utility of a continuum of fluency-based indicators of foundational reading skills for third-grade high-stakes outcomes. Scientific Studies of Reading, 5, 257–288.

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

Hasbrouck, J., & Haager, D. (Eds.). (2007). Monitoring children's progress in academic learning. *Perspectives* on Language and Literacy 33(2).

- Hogan, T. P., Catts, H. W., & Little, T. D. (2005). The relationship between phonological awareness and reading: Implications for the assessment of phonological awareness. *Language, Speech, and Hearing Services in Schools*, 36, 285–293.
- Hosp, M. K., Hosp, J. L., & Howell, K. W. (2016). The ABCs of CBM: A practical guide to curriculumbased measurement. New York, NY: Guilford.
- Jenkins, J. R., Johnson, E., & Hileman, J. (2004). When is reading also writing: Sources of individual differences on the new reading performance assessments. *Scientific Studies of Reading*, 8, 125–152.
- Johnson, E. S., Jenkins, J. R., Petscher, Y., & Catts, H. W. (2009). How can we improve the accuracy of screening instruments? *Learning Disabilities Research & Practice, 24*, 174–185.
- Keenan, J. M., Betjemann, R. S., & Olson, R. K. (2008). Reading comprehension tests vary in the skills they assess: Differential dependence on decoding and oral comprehension. *Scientific Studies of Reading*, 12, 281–300.
- Kilpatrick, D. (2015). Essentials of assessing, preventing, and overcoming reading difficulties. Hoboken, NJ: Wiley.
- Pennington, B. (2009). Diagnosing learning disorders (2nd ed.). New York: Guilford Press.

Shinn, M. R. (2008). Best practices in using curriculum-based measurement in a problem-solving model. In A. Thomas, & J. Grimes (Eds.), *Best practices in school psychology* (pp. 671–697). Bethesda, MD: National Association of School Psychologists.

Torgesen, J. K. (2004). Avoiding the devastating downward spiral: The evidence that early intervention prevents reading failure. *American Educator*, 28(3), 6–9, 12–13, 17–19, 45–47.

STANDARD 4, Substandard A: Essential Principles and Practices of Structured Literacy Instruction

Structured Literacy teaching can be contrasted with meaning-emphasis, child-centered, incidental instruction in which foundational skills are generally not emphasized, even for children at the earliest stages of learning to read and write. Structured Literacy involves teaching language concepts in an explicit, systematic, cumulative manner, according to a planned scope and sequence of skill development. Structured Literacy approaches emphasize direct interaction with a teacher who provides clear explanations and modeling of new skills and concepts; prompt, unambiguous, corrective feedback to errors; and application of decoding skills in texts that lend themselves to decoding, rather than texts containing many words that beginners will be unable to decode. In these approaches, spelling instruction is well coordinated with decoding instruction, and higher levels of literacy—such as syntax, paragraph organization, and discourse structure—are also taught systematically. In contrast, most approaches are especially valuable for students with reading disabilities such as dyslexia; however, many other children can also benefit from these approaches, including ELs and children at risk in reading due to limited experiences with literacy and academic language.

- Archer, A., & Hughes, C. A. (2011). *Explicit instruction: Effective and efficient teaching*. New York, NY: Guilford Press.
- Birsh, J. (Ed.) (2011). Multisensory teaching of basic language skills, 3rd Ed. Baltimore: Brookes Publishing.

Knowledge and Practice Standards for Teachers of Reading
©2018, The International Dyslexia Association

Moats, L. C. (2017). Can prevailing approaches to reading instruction accomplish the goals of RTI? Perspectives on Language and Literacy, 43, 15–22.

Moats, L. C., Dakin, K., & Joshi, M. (Eds.) (2012). *Expert perspectives on interventions for reading*. Baltimore, MD: International Dyslexia Association.

Rivera, M. O., Moughamian, A. C., Lesaux, N. K., & Francis, D. J. (2008). Language and reading interventions for English language learners and English language learners with disabilities. Portsmouth, NH: RMC Research Corporation, Center on Instruction.

Spear-Swerling, L. (in press). Structured literacy and typical literacy practices: Understanding differences to create instructional opportunities. *Teaching Exceptional Children*.

STANDARD 4, Substandard B: Structured Literacy Instruction—Phonological Awareness, Phonological Sensitivity, Phonemic Awareness

Phonological sensitivity (awareness of rhyme, alliteration, syllables, and larger chunks of words) and phonemic awareness are essential foundations for reading and writing. All children benefit from explicit teaching of consonant and vowel phonemes apart from, but connected to, the letters that represent them. Without early, research-based intervention, children who struggle with speech-sound awareness are likely to have difficulty learning to use phonics for decoding, remembering the pronunciation of words (especially when they sound similar), and spelling. Furthermore, poor phonological awareness is a core weakness in dyslexia. Ample research exists to inform the teaching of phonological awareness, including research on the phonological skills to emphasize in instruction, appropriate sequencing of instruction, methods to help students identify phonemes, such as the use of articulatory cues, and integrating instruction in phonological awareness with instruction in alphabet knowledge. Educators who understand how to teach these foundational skills effectively can prevent or ameliorate many children's reading problems, including those of students with dyslexia.

Adams, M., Foorman, B. R., Lundberg, I., & Beeler, T. (Spring/Summer, 1998). The elusive phoneme: Why phonemic awareness is so important and how to help children develop it. *American Educator, 22*(1 & 2), 18–29.

Boyer, N., & Ehri, L. C. (2011). Contribution of phonemic segmentation instruction with letters and articulation pictures to word reading and spelling in beginners. *Scientific Studies of Reading*, 15(5), 440–470.

Brady, S. & Shankweiler, D. (Eds.). (1991). *Phonological processes in literacy: A tribute to Isabelle Y. Liberman.* Hillsdale, NJ: Lawrence Erlbaum Associates.

Gillon, G. (2004). Phonological awareness: From research to practice. New York: Guilford Press. Kilpatrick, D. (2015). Essentials of assessing, preventing, and overcoming reading difficulties. Hoboken, NJ: Wiley.

Moats, L.C. & Tolman, C. A. (2018). Language essentials for teachers of reading and spelling (LETRS), 3rd Ed. Dallas, TX: Voyager Sopris Learning.

Neuman, S. B., & Dickinson, D. K. (2002). Handbook of early literacy research. New York: Guilford Press.

Scarborough, H. S., & Brady, S. A. (2002). Toward a common terminology for talking about speech and reading: A glossary of the 'phon' words and some related terms. *Journal of Literacy Research*, 34, 299– 334.

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

STANDARD 4, Substandard C: Structured Literacy Instruction—Phonics and Word Recognition

The development of accurate word-decoding skills is an essential foundation for reading fluency and reading comprehension in all students. Word decoding is the ability to read unfamiliar words by applying knowledge of sounds for letters, letter patterns (e.g., sh, igh, ar), and the alphabetic code. At more advanced stages of word reading, decoding also requires knowledge of syllabication strategies (e.g., dividing between two consonants in a word with a VCCV pattern, such as *lantern*) and the ability to recognize common morphemes in words (e.g., un-, mis-, -ed, -ing, -able). These kinds of skills are often a central weakness for students with poor reading, including those with dyslexia. The ability of both general and special educators to provide explicit, systematic, appropriately sequenced instruction in decoding is indispensable to meet the needs of students with dyslexia and to help prevent reading problems in other at-risk children and beginning readers in general. Educators should know that recent, post-NRP evidence favors synthetic, parts-to-whole approaches to decoding over inductive, whole-word approaches (e.g., word families). They should also recognize the importance of students' opportunities to apply their developing decoding skills in reading connected text, including oral reading with feedback from a teacher. Teacher feedback should emphasize attention to the print and application of decoding skills rather than guessing at words based on pictures or sentence context. Finally, teachers should understand the usefulness of multisensory, multimodal techniques in focusing students' attention on printed words, engaging students, and enhancing memory.

- Blachman, B. A., Schatschneider, C., Fletcher, J. M., Francis, D. J., Clonan, S., Shaywitz, B., et al. (2004). Effects of intensive reading remediation for second and third graders. *Journal of Educational Psychology*, 96, 444–461.
- Brady, S. (2011). Efficacy of phonics teaching for reading outcomes: Implications from post-NRP research. In S. A. Brady, D. Braze, & C. Fowler (Eds.), *Explaining individual differences in reading: Theory and evidence* (pp. 69–96). New York, NY: Psychology Press.
- Calhoon, M. B. (2005). Effects of a peer-mediated phonological skill and reading comprehension program on reading skill acquisition for middle school students with reading disabilities. *Journal of Learning Disabilities*, 38(5), 424–433.
- Carlisle, J. F. (2010). An integrative review of the effects of instruction in morphological awareness on literacy achievement. *Reading Research Quarterly*, 45, 464–487.
- Connor, C. M., Morrison, F. J., & Underwood, P. S. (2007). A second chance in second grade: The independent and cumulative impact of first- and second-grade reading instruction and students' letter-word reading skill growth. *Scientific Studies of Reading*, 11, 199–233.
- Christensen, C. A., & Bowey, J. A. (2005). The efficacy of orthographic rime, grapheme-phoneme correspondence, and implicit phonics approaches to teaching decoding skills. *Scientific Studies of Reading*, 9, 327–349.
- Daane, M. C., Campbell, J. R., Grigg, W. S., Goodman, M. J., & Oranje, A. (2005). Fourth-grade students reading aloud: NAEP 2002 Special Study of Oral Reading (NCES 2006-469). U.S. Department of Education. Institute of Education Sciences, National Center for Education Statistics. Washington, DC: Government Printing Office.
- Ehri, L. C. (2014). Orthographic mapping in the acquisition of sight word reading, spelling memory, and vocabulary learning. Scientific Studies of Reading, 18(1), 5-21. In P. McCardle & V. Chhabra (Eds.), The voice of evidence in reading research (pp. 153–186). Baltimore: Brookes Publishing.
- Goodwin, A. P., & Ahn, S. (2013). A meta-analysis of morphological interventions in English: Effects on literacy outcomes for school-age children. *Scientific Studies of Reading*, *17*, 257–285.

Knowledge and Practice Standards for Teachers of Reading
©2018, The International Dyslexia Association

- Grace, K. (2006). Phonics and spelling through phoneme-grapheme mapping. Longmont, CO: Sopris West. Henry, M. (2010). Unlocking literacy: Effective decoding and spelling instruction, 2nd ed. Baltimore: Brookes Publishing.
- Joseph, L. M., & Schisler, R. (2009). Should adolescents go back to the basics? A review of teaching word reading skills to middle and high school students. *Remedial and Special Education*, *30*(3), 131–147.
- Lovett, M. W., Lacerenza, L., DePalma, M., & Frijters, J. C. (2012). Evaluating the efficacy of remediation for struggling readers in high school. *Journal of Learning Disabilities*, 45(2), 151–169.
- Mathes, P. G., Denton, C. A., Fletcher, J. M., Anthony, J. L., Francis, D. J., & Schatschneider, C. (2005). The effects of theoretically different instruction and student characteristics on the skills of struggling readers. *Reading Research Quarterly*, 40, 148–182.
- McCandliss, B., Beck, I. L., Sandak, R., & Perfetti, C. (2003). Focusing attention on decoding for children with poor reading skills: Design and preliminary tests of the word building intervention. *Scientific Studies of Reading*, 7, 75–104.
- Moats, L. C. (1998). Teaching decoding. American Educator, 22(1&2), 42-49, 95-96.
- Torgesen, J. K. (2004). Lessons learned from research on interventions for students who have difficulty learning to read. In P. McCardle & V. Chhabra (Eds.), *The voice of evidence in reading research* (pp. 355– 381). Baltimore: Brookes Publishing.
- Vadasy, P. F., Sanders, E. A., & Peyton, J. A. (2005). Contributions of reading practice to first-grade supplemental tutoring: How text matters. *Journal of Learning Disabilities*, 38, 364–380.

STANDARD 4, Substandard D: Structured Literacy Instruction—Automatic, Fluent Reading of Text

Reading fluency is the ability to read text effortlessly, quickly, and accurately. Good reading fluency is also characterized by appropriate prosody (e.g., intonation and phrasing) in oral reading. Fluency develops among typical readers in the primary grades and is important because lack of fluency tends to drain students' reading comprehension and motivation to read; poor fluency also makes it difficult for students to keep up with increasing demands for reading volume in the middle and secondary grades. Problems with reading fluency is a very common symptom of dyslexia and other reading disabilities, and these problems can linger even when students' accuracy in word decoding has been improved through effective phonics intervention. Although fluency difficulties may sometimes be associated with processing weaknesses, considerable research supports the role of practice, wide exposure to printed words, and focused instruction in the development and remediation of fluency. To address students' fluency needs, teachers must have a range of competencies, including the ability to interpret fluency-based measures appropriately, place students in appropriate types and levels of texts for reading instruction, stimulate students' independent reading, and provide systematic fluency interventions for students who require them. Teachers should also recognize when a student's fluency difficulties relate to language comprehension factors rather than to decoding, as when a student decodes individual words accurately and automatically but reads text slowly because he or she is struggling to understand meaning. Assistive technology (e.g., text-to-speech software) is often employed to help students with serious fluency difficulties function in general education settings. Therefore, teachers, and particularly specialists, require knowledge about the appropriate uses of this technology.

Carreker, S. (2005). Teaching reading: Accurate decoding and fluency. In J. Birsh (Ed.). *Multisensory teaching of basic language skills* (2nd ed., pp. 213–255). Baltimore: Brookes Publishing.

Knowledge and Practice Standards for Teachers of Reading • ©2018, The International Dyslexia Association

Chard, D., Vaughn, S., & Tyler, B. (2002) A synthesis of research on effective interventions for building fluency with elementary students with learning disabilities. *Journal of Learning Disabilities*, 35, 386– 406.

Cunningham, A. E., & Stanovich, K. E. (1998). What reading does for the mind. American Educator, 22(1&2), 8–15.

- Connor, C. M., Morrison, F. J., & Katch, L. E. (2004). Beyond the reading wars: Exploring the effect of child- instruction interactions on growth in early reading. *Scientific Studies of Reading*, 8, 305–336.
- Cutting, L. E., Materek, A., Cole, C., Levine, T., & Mahone, E. M. (2009). Effects of fluency, oral language, and executive function on reading comprehension performance. *Annals of Dyslexia*, 59, 34–54.
- Ehri, L. C. (1997). Sight word learning in normal readers and dyslexics. In B. Blachman (Ed.). Foundations of reading acquisition and dyslexia (pp. 163–189). Mahwah, NJ: Erlbaum.
- Fuchs, L. S., Fuchs, D., Hosp, M. K., & Jenkins, J. (2001). Oral reading fluency as an indicator of reading competence: A theoretical, empirical, and historical analysis. *Scientific Studies of Reading*, 5(3), 239– 256.
- Hamilton, C., & Shinn, M. R. (2003). Characteristics of word callers: An investigation of the accuracy of teachers' judgments of reading comprehension and oral reading skills. *School Psychology Review*, 32(2), 228–240.
- Hasbrouck, J. E., & Tindal, G. A. (2017). An Update to Compiled Oral Reading Fluency Norms, Technical Report #1702. Eugene, OR: Behavioral Research and Teaching, University of Oregon. http://brt.uoregon.edu
- Hudson, R. F., Lane, H. B., & Pullen, P. C. (2005). Reading fluency assessment and instruction: What, why, and how? *The Reading Teacher, 58*, 702–714.
- Katzir, T., Kim, Y., Wolf, M., O'Brien, B., Kennedy, B., Lovett, M., et al. (2006). Reading fluency: The whole is more than the parts. *Annals of Dyslexia*, *56*(1), 51–82.
- Kuhn, M. (2004/2005). Helping students become accurate, expressive reading: Fluency instruction for small groups. *The Reading Teacher*, 58(4), 338–345.
- Kuhn, M. R., Schwanenflugel, P. J., & Meisinger, E. B. (2010). Aligning theory and assessment of reading fluency: Automaticity, prosody, and definitions of fluency. *Reading Research Quarterly*, 45, 230–251.
- Meyer, M. (Winter, 2002). Repeated reading: An old standard is revisited and renovated. *Perspectives* (The International Dyslexia Association Quarterly Newsletter), 15–18.
- Meyer, M. S., & Felton, R. H. (1999). Repeated reading to enhance fluency: Old approaches and new directions. *Annals of Dyslexia*, 49, 293–306.

Speece, D. L., & Ritchey, K. D. (2005). A longitudinal study of the development of oral reading fluency in young children at risk for reading failure. *Journal of Learning Disabilities*, *38*(5), 387–399.

- Torgesen, J., Alexander, A. W., Wagner, R., Rashotte, C. A., Voeller, K., Conway, T., et al. (2001). Intensive remedial instruction for children with severe reading disabilities: Immediate and long-term outcomes from two instructional approaches. *Journal of Learning Disabilities*, 34, 33–58.
- Valencia, S. W., Smith, A. T., Reece, A. M., Li, M., Wixson, K. K., & Newman, H. (2010). Oral reading fluency assessment: Issues of construct, criterion, and consequential validity. *Reading Research Quarterly*, 45, 270–291.

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

STANDARD 4, Substandard E: Structured Literacy Instruction—Vocabulary

Vocabulary, or knowledge of word meanings, plays a key role in reading comprehension. Knowledge of words is multifaceted, ranging from partial recognition of the meaning of a word to deep knowledge and the ability to use the word effectively in speech or writing. Research supports both explicit, systematic teaching of word meanings and indirect methods of instruction, such as those involving inferring meanings of words from sentence context or from morphology (e.g., word parts, such as common roots and affixes). Teachers should understand the importance of vocabulary to overall reading comprehension, and they should recognize populations of children who are especially likely to be at risk in the area of vocabulary, such as ELs and children with limited exposure to literacy at home. Both general and special educators should know how to develop students' vocabulary knowledge through direct and indirect methods. They should also recognize the importance of a wide exposure to words, both orally and through reading, in students' vocabulary development. For example, although oral vocabulary knowledge frequently is a strength for students with dyslexia, over time, a low volume of reading may tend to reduce these students' exposure to rich vocabulary relative to their typical peers; explicit teaching of word meanings and encouragement of wide independent reading in appropriate texts are two ways to help increase this exposure.

Beck, I. L., McKeown, M. G., & Kucan, L. (2002). Bringing words to life: Robust vocabulary instruction. New York: Guilford Press.

 Biemiller, A. (2009). Words worth teaching: Closing the vocabulary gap. Columbus, OH: SRA/McGraw Hill.
 Biemiller, A., & Boote, C. (2006). An effective method for building meaning vocabulary in primary grades. Journal of Educational Psychology, 98, 44–62.

Carlisle, J. F., Kelcey, B., & Berebitsky, D. (2013). Teachers' support of students' vocabulary learning during literacy instruction in high poverty elementary schools. *American Educational Research Journal*, 50, 1360–1391.

Diamond, L., & Gutlohn, L. (2006). *Vocabulary handbook*. Berkeley, CA: Consortium on Reading Excellence.

- Ebbers, S. (2006). Vocabulary through morphemes. Longmont, CO: Sopris West.
- Gersten, R., Baker, S. K., Shanahan, T., Linan-Thompson, S., Collins, P., & Scarcella, R. (2007). Effective literacy and English language instruction for English learners in the elementary grades: A practice guide (NCEE 2007-4011). Washington, DC: National Center for Education Evaluation and Regional Assistance. Institute of Education Sciences. U.S. Department of Education.
- Graves, M. (2006). *The vocabulary book: Learning and instruction*. New York: Teachers College Press, Columbia University.

Hirsch, E. D. (2006). The knowledge deficit: Closing the shocking education gap for American children. Boston: Houghton Mifflin.

- Kamil, M. (2004). Vocabulary and comprehension instruction: Summary and implications of the National Reading Panel findings. In P. McCardle and V. Chhabra (Eds.), *The voice of evidence in reading research* (pp. 213–234).
- Loftus, S. M., Coyne, M. D., McCoach, B., Zipoli, R., & Pullen, P. C. (2010). Effects of a supplemental vocabulary intervention on the word knowledge of kindergarten students at risk for language and literacy difficulties. *Learning Disabilities Research & Practice*, 25, 124–136.
- Neuman, S. B., & Wright, T. S. (2013). All about words: Increasing vocabulary in the Common Core classroom (PreK-2). New York, NY: Teachers College Press.
- Paynter, D. E., Bodrova, E., & Doty, J. K. (2005). For the love of words: Vocabulary instruction that works, Grades K–6. San Francisco: Jossey-Bass.

Riedel, B. W. (2007). The relation between DIBELS, reading comprehension, and vocabulary in urban firstgrade students. *Reading Research Quarterly*, 42, 546–567.

Stahl, S. A., & Nagy, W. E. (2006) *Teaching word meanings*. Mahwah, NJ: Erlbaum. Tannenbaum, K. R., Torgesen, J. T., & Wagner, R. K. (2006). Relationships between word knowledge and reading comprehension in 3rd grade children. *Scientific Studies of Reading*, 10, 381–398.

STANDARD 4, Substandard F: Structured Literacy Instruction—Listening and Reading Comprehension

Good reading comprehension is the ultimate goal of reading instruction. Reading comprehension depends not only upon the component abilities discussed in previous sections, but also upon other factors, such as background knowledge, comprehension of syntax, and knowledge of text structure. To plan effective instruction and intervention in reading comprehension, teachers must understand the array of abilities that contribute to reading comprehension and use assessments to help pinpoint students' weaknesses. For instance, a typical student with dyslexia, whose reading comprehension problems are associated mainly with poor decoding and dysfluent reading, will need different emphases in intervention than will a student with poor comprehension whose problems revolve around broad weaknesses in vocabulary and oral comprehension. In addition, teachers must be able to model and teach research-based comprehension strategies, such as summarization and the use of graphic organizers, and use methods that promote reflective reading and engagement. Oral comprehension and reading comprehension have a reciprocal relationship; good oral comprehension facilitates reading comprehension, but wide reading also contributes to the development of oral comprehension, especially in older students. Teachers should understand the relationships among oral language, reading comprehension, and written expression, and they should be able to use appropriate writing activities to build students' comprehension. They should also recognize the importance of including oral interventions (and reading interventions) in helping students who have difficulties with comprehension.

- Barnes, M. A., Johnston, A. M., & Dennis, M. (2007). Comprehension in a neurodevelopmental disorder, Spina Bifida Myelomeningocele. In K. Cain & J. V. Oakhill (Eds.), *Children's comprehension problems in* oral and written language: A cognitive perspective (pp. 193–217). New York: Guilford Press.
- Beck, I. L., & McKeown, M. G. (2006). Improving comprehension with questioning the author: A fresh and expanded view of a powerful approach. New York: Scholastic.
- Caccamise, D., & Snyder, L. (Eds.). (2009). Reading comprehension: Issues and instructional applications. Perspectives on Language and Literacy, 35(2).
- Cain, K., & Oakhill, J. V. (2007). Reading comprehension difficulties: Correlates, causes, and consequences. In K. Cain & J. V. Oakhill (Eds.), *Children's comprehension problems in oral and written language: A cognitive perspective* (pp. 81–103) New York: Guilford Press.
- Cardenas-Hagan, E. (2016). Listening comprehension: Special considerations for English learners. Perspectives on Language and Literacy, 42(3), 31–35.
- Carlisle, J. R., & Rice, M. S. (2002). Improving reading comprehension: Research-based principles and practices. Baltimore: York Press.
- Clarke, P. J., Snowling, M. J., Truelove, E., & Hulme, C. (2010). Ameliorating children's reading-
- comprehension difficulties: A randomized controlled trial. *Psychological Science*, *21*, 1106–1116. Cunningham, A. E., & Stanovich, K. E. (1998). What reading does for the mind. *American Educator*, *22*, 8–15.
- Gattardo, A., Stanovich, K., & Siegel, L. (1996). The relationships between phonological sensitivity, syntactic processing, and verbal working memory in the reading performance of third-grade

Knowledge and Practice Standards for Teachers of Reading

©2018, The International Dyslexia Association

children. Journal of Experimental Child Psychology, 63, 563-582.

- Gersten, R., Fuchs, L. S., Williams, J. P., & Baker, S. (2001). Teaching reading comprehension strategies to students with learning disabilities: A review of research. *Review of Educational Research*, 71, 279–320.
- Graesser, A. C. (2008). An introduction to strategic reading comprehension. In D. S. McNamara (Ed.). Reading comprehension strategies: Theories, intervention and technologies (pp. 3–26). New York: Erlbaum.
- Hirsch, E. D. (2006). Building knowledge: The case for bringing content into the language arts block and for a knowledge-rich curriculum core for all children. *American Educator*, 30(1), 8–21, 28–29, 50–51.
- Kamil, M. (2004). Vocabulary and comprehension instruction: Summary and implications of the National Reading Panel findings. In P. McCardle & V. Chhabra (Eds.). *The voice of evidence in reading research* (pp. 213–234). Baltimore: Brookes Publishing.
- Kamil, M. L., Borman, G. D., Dole, J., Kral, C. C., Salinger, T., & Torgesen, J. (2008). *Improving adolescent literacy: Effective classroom and intervention practices: A Practice Guide* (NCEE #2008-4027).
 Washington, DC: National Center for Education Evaluation and Regional Assistance, Institute of Education Sciences, U.S. Department of Education. Retrieved from http://ies.ed.gov/ncee/wwc
- Kintsch, E. (2005). Comprehension theory as a guide for the design of thoughtful questions. *Topics in language disorders, 25*(1), pp. 51–64.
- McKeown, M. G., Beck, I. L., & Blake, R. (2009). Rethinking reading comprehension instruction: A comparison of instruction for strategies and content approaches. *Reading Research Quarterly*, 44, 218–253.
- McMaster, K. L., Fuchs, D., & Fuchs, L. S. (2006). Research on peer-assisted learning strategies: The promise and limitations of peer-mediated instruction. *Reading & Writing Quarterly, 22*, 5–25.
- Nelson, N.W. (2013). Syntax development in the school-age years: Implications for assessment and instruction. *Perspectives on Language and Literacy*, 39, 9–15.
- Oakhill, J., Cain, K., & Elbro, C. (2015). Understanding and teaching reading comprehension: A handbook. New York: Routledge.
- Scarborough, H. S. (2005). Developmental relationships between language and reading: Reconciling a beautiful hypothesis with some ugly facts. In H. W. Catts & A. Kamhi (Eds.), *The connections between language and reading disabilities* (pp. 3–24). Mahwah, NJ: Erlbaum.
- Scott, C. M. (2009). A case for the sentence in reading comprehension. Language, Speech, and Hearing Services in the Schools, 40(2), 184–191.
- Shanahan, T., Callison, K., Carriere, C., Duke, N. K., Pearson, P. D., Schatschneider, C., & Torgesen, J. (2010). Improving reading comprehension in kindergarten through 3rd grade: A practice guide (NCEE 2010-4038). Washington, DC: National Center for Education Evaluation and Regional Assistance, Institute of Education Sciences, U.S. Department of Education.
- Shankweiler, D., Lundquist, E., Katz, L., Stuebing, K. K., Fletcher, J. M., Brady, S., et al. (1999). Comprehension and decoding: Patterns of association in children with reading difficulties. *Scientific Studies of Reading*, 31, 24–35, 69–94.
- Stahl, K. A. D. (2004). Proof, practice, and promise: Comprehension strategy instruction in the primary grades. *The Reading Teacher*, *57*, 598–609.
- Westby, C. (2004). A language perspective on executive functioning, metacognition, and self-regulation in reading. In C. A. Stone, E. R. Silliman, B. J. Ehren, & K. Apel (Eds.). *Handbook of language and literacy: Development and disorders* (pp. 398–427). New York: Guilford Press.
- Williams, J. P. (2006). Stories, studies, and suggestions about reading. *Scientific studies of reading*, 10(2), 121–142.

Willingham, D. T. (2006). How knowledge helps: It speeds and strengthens reading comprehension, learning, Knowledge and Practice Standards for Teachers of Reading • ©2018, The International Dyslexia Association

and thinking. American Educator, 30(1), 30–37. Willingham, D. T. (2006-07). The usefulness of brief instruction in reading comprehension strategies. American Educator, 30(4), 39–45.

STANDARD 4, Substandard G: Structured Literacy Instruction-Written Expression

Just as teachers need to understand the component abilities that contribute to reading comprehension, they also need a componential view of written expression. Important component abilities in writing include basic writing (transcription) skills, such as handwriting, keyboarding, spelling, capitalization, punctuation, and grammatical sentence structure; text generation (composition) processes that involve translating ideas into language, such as appropriate word choice, writing clear sentences, and developing an idea across multiple sentences and paragraphs; and planning, revision, and editing processes. Teachers should understand how, similar to the relationship between reading fluency and reading comprehension, weaknesses in basic writing skills, such as spelling and handwriting, may drain students' abilities and motivation to write. Also, just as in the case of reading, explicit and systematic teaching of important components of writing as part of general education instruction can help prevent or ameliorate many children's writing difficulties. Effective intervention in written expression depends on pinpointing an individual student's specific weaknesses in different component areas of writing and on teachers' abilities to provide explicit, systematic teaching in each area. For instance, a student whose writing difficulties revolve around basic writing skills, such as spelling, will require a different type of intervention than one who has strong foundational writing skills but struggles with text generation processes, such as clarity and word choice. In addition to using assessments to help target individual students' writing weaknesses, both general and special educators should be able to teach research-based strategies in written expression, such as those involving strategies for planning and revising compositions, and they should understand the utility of multisensory methods in both handwriting and spelling instruction. Assistive technology can be especially helpful for students with writing difficulties, especially as they advance into the middle and upper grades and the demands for writing escalate. Teachers should also recognize the appropriate uses of technology in writing (e.g., spell-checkers can be valuable, but do not replace spelling instruction and have limited utility for students whose misspellings are not recognizable). Specialists should have even greater levels of knowledge about technology.

Berninger, V. W., & Wolf, B. J. (2016). Dyslexia, dysgraphia, OWL LD, and dyscalculia: Lessons from science and teaching (2nd ed.). Baltimore: Brookes Publishing.

Edwards, L. (2003). Writing instruction in kindergarten: Examining an emerging area of research for children with writing and reading difficulties. *Journal of Learning Disabilities, 36*, 136.

Englert, C. S., Wu, X., & Zhao, Y. (2005). Cognitive tools for writing: Scaffolding the performance of students through technology. *Learning Disabilities Research & Practice*, 20, 184–198.

Gersten, R., & Baker, S. (2001). Teaching expressive writing to students with learning disabilities: A meta-analysis. *Elementary School Journal*, 101, 251–272.

- Gilbert, J., & Graham, S. (2010). Teaching writing to elementary students in grades 4–6: A national survey. *Elementary School Journal*, 110(4), 494–518.
- Graham, S., & Hebert, M. (2010). Writing to read: Evidence for how writing can improve reading. New York, NY: Carnegie Corporation, Alliance for Excellent Education.

Graham, S., McArthur, C.A., & Fitzgerald, J. (Eds.). (2007). *Best practices in writing instruction*. New York: Guilford Press.

Graham, S., & Perin, D. (2007). Writing next: Effective strategies to improve writing of adolescents in middle and high schools – A report to Carnegie Corporation of New York. Washington, DC: Alliance for Excellent Education.

- Hecker, L., & Engstrom, E. U. (2011). Technology that supports literacy instruction and learning. In J. Birsh (Ed.). Multisensory teaching of basic language skills, (3rd ed.) (pp. 657–683). Baltimore, MD: Brookes Publishing.
- Joshi, M., Treiman, R., Carreker, S., & Moats, L. C. (2008/2009) How words cast their spell: Spelling is an integral part of learning the language, not a matter of memorization. *American Educator*, 32(4), 6– 16, 42–43.
- Masterson, J. J., & Apel, K. (2010). Linking characteristics discovered in spelling assessment to intervention goals and methods. *Learning Disabilities Quarterly*, 33(3), 185–198.
- Moats, L. C. (Winter 2005/06). How spelling supports reading: And why it is more regular and predictable than you think. American Educator, 12–22, 42–43.
- Spear-Swerling, L., & Zibulsky, J. (2014). Making time for literacy: Teacher knowledge and time allocation in instructional planning. *Reading and Writing: An Interdisciplinary Journal*, 27(8), 1353–1378.
- Troia, G. (Ed.). (2009) Instruction and assessment for struggling writers: Evidence-based practices. New York: Guilford Press.
- Wolf, B. (2011). Teaching handwriting. In J. R. Birsh (Ed.). Multisensory teaching of basic language skills (3rd ed., pp 179–206). Baltimore, MD: Brookes Publishing.

Knowledge and Practice Standards for Teachers of Reading
©2018, The International Dyslexia Association